WORLD TRADE ORGANIZATION

WT/DS381/R 15 September 2011

(11-4239)

Original: English

UNITED STATES – MEASURES CONCERNING THE IMPORTATION, MARKETING AND SALE OF TUNA AND TUNA PRODUCTS

Report of the Panel

TABLE OF CONTENTS

| I. | INTRODUCTION | Page 1 |
|------------|--|--------|
| ı. II. | FACTUAL ASPECTS | |
| A. | THE MEASURES AT ISSUE | |
| В. | SCOPE OF APPLICATION OF THE DPCIA PROVISIONS | |
| D. С. | CONDITIONS FOR THE USE OF THE TERM "DOLPHIN-SAFE" OR ANALOGOUS TERMS | |
| D. | THE CERTIFICATION REQUIRED FOR TUNA CAUGHT IN THE ETP | |
| Б. Е. | THE CERTIFICATION REQUIRED FOR TUNA CAUGHT OUTSIDE THE ETP | |
| ъ. F. | OFFICIAL AND ALTERNATIVE "DOLPHIN-SAFE" MARKS AND LABELS | |
| G. | THE "DOLPHIN-SAFE" TRACKING AND VERIFICATION PROGRAM AND OTHER ENFORCEMENT PROVISIONS | |
| Н. | THE "DOLPHIN SAFE" SCHEME ESTABLISHED UNDER THE AIDCP | 14 |
| III. | PARTIES' REQUESTS FOR FINDINGS AND RECOMMENDATIONS | 15 |
| IV. | ARGUMENTS OF THE PARTIES | |
| A. | FIRST WRITTEN SUBMISSIONS | 16 |
| 1. | Mexico | 16 |
| (a) | Introduction | 16 |
| (b) | Summary of facts and background information | 17 |
| (i) | Tuna fishing | 17 |
| | Tuna and dolphins | 17 |
| | Tuna fishing and dolphins mortality in the ETP | 17 |
| | Fishing methods | 17 |
| (ii) | Efforts to protect dolphins | 18 |
| | Multilateral efforts | 18 |
| | Actions of the United States | 18 |
| (iii) | The Mexican fishing fleet | 19 |
| (iv) | Adverse effects of the measure | 20 |
| (c) | Subject products | 20 |
| (d) | Specific measures at issue | 20 |
| (e) | Legal argument | 20 |
| <i>(i)</i> | GATT 1994 | 21 |
| | Mexico's claim under Article III:4 of the GATT 1994 | 21 |
| | Like products | 21 |
| | Law, regulation, or requirement affecting their internal sale, offering for sale, purchase, transportation, distribution, or use | |
| | Less favourable treatment | |
| | Mexico's claim under Article I:1 of the GATT 1994 | 23 |

| | Like products | 23 |
|------------|---|----|
| | Advantage, favour or privilege on a product originating in other WTO Members | 23 |
| | Accorded immediately and unconditionally to the like product originating in the territories of all other Members | 23 |
| | Conclusions | 24 |
| (ii) | Claims under the TBT Agreement | 24 |
| | The US measures constitute a technical regulation | 24 |
| | The measures are inconsistent with Article 2.2 | 25 |
| | The measures do not fulfil a legitimate objective | 25 |
| | The US measures are more trade-restrictive than necessary taking account of the risks non-fulfilment would create | 26 |
| | The US measures are inconsistent with Article 2.4 | 26 |
| | AIDCP standard | 26 |
| | The AIDCP standard is a relevant international standard | 26 |
| | International standard | 26 |
| | Relevant | 26 |
| | The United States failed to base its regulation on the relevant international standard | 27 |
| | The relevant international standard does not constitute an ineffective or inappropriate means for the fulfilment of the legitimate objectives pursued | 27 |
| | The US measures are inconsistent with Article 2.1 | 27 |
| (f) | Conclusions | 27 |
| 2. | United States | 27 |
| (a) | Introduction | 27 |
| (b) | Legal argument | 28 |
| <i>(i)</i> | The US dolphin-safe labelling provisions are not inconsistent with GATT 1994 Article III:4 | 29 |
| (ii) | The US dolphin-safe labelling provisions are not inconsistent with GATT 1994 Article I:1 | 29 |
| (iii) | The US dolphin-safe labelling provisions are not technical regulations and therefore not subject to Article 2 of the TBT Agreement | 30 |
| (iv) | The US measures are not inconsistent with Article 2.1 of the TBT Agreement | |
| (v) | The US dolphin-safe labelling provisions are not inconsistent with Article 2.2 of the TBT Agreement | |
| (vi) | The US dolphin-safe labelling provisions are not inconsistent with Article 2.4 of the TBT Agreement | |
| (c) | Conclusion | |
| В. | OPENING ORAL STATEMENTS AT THE FIRST SUBSTANTIVE MEETING | |
| 1. | Mexico | |
| (a) | Introduction | |
| (b) | Tuna fishing in the Eastern Tronical Pacific | 35 |

| (c) | History of US restrictions on imports of Mexican tuna products | 36 |
|------------|---|----|
| (d) | The US measures | 37 |
| (e) | The US measures are discriminatory in violation of Articles I and III of the GATT 1994 and Article 2.1 of the TBT Agreement | 37 |
| (f) | The US measures are inconsistent with the TBT Agreement | 39 |
| <i>(i)</i> | The US measures are inconsistent with TBT Article 2.2 | 39 |
| (ii) | The US measures are inconsistent with TBT Article 2.4 | 40 |
| (g) | Conclusions | 40 |
| 2. | United States | 40 |
| (a) | Introduction | 40 |
| (b) | Article III:4 of the GATT 1994 | 41 |
| (c) | Article I:1 of the GATT 1994 | 42 |
| (d) | Article 2.1 of the TBT Agreement | 43 |
| (e) | US dolphin-safe labelling provisions are not technical regulations under Articles 2.1, 2.2 or 2.4 of the TBT Agreement | 43 |
| (f) | US provisions are not inconsistent with Articles 2.2 or 2.4 of the TBT Agreement | 44 |
| (g) | Amicus submission | 45 |
| C. | CLOSING ORAL STATEMENTS AT THE FIRST SUBSTANTIVE MEETING | 45 |
| 1. | Mexico | 45 |
| (a) | Discrimination claims | 45 |
| (b) | Impermissible imposition of extraterritorial policies | 45 |
| (c) | Unilateral v. multilateral | 46 |
| 2. | United States | 47 |
| D. | SECOND WRITTEN SUBMISSIONS | 48 |
| 1. | Mexico | 48 |
| (a) | Introduction | 48 |
| (b) | Measures at issue | 49 |
| <i>(i)</i> | The measures must be assessed cumulatively | 49 |
| (ii) | Functioning of the US measures | 49 |
| | Application inside of the ETP | 49 |
| | Application outside of the ETP | 49 |
| (c) | Facts | 50 |
| <i>(i)</i> | The US measures have the effect of excluding Mexican tuna products from the major distribution channels | 50 |
| (ii) | Tuna fishing and effects on dolphins | 50 |
| | US theory of unobserved mortalities | 51 |
| | Purported US goal of no mortality | 51 |

| (iii) | Unilateral action by the United States | 51 |
|------------|--|----|
| (iv) | Effects on other species | 51 |
| (v) | Association between tuna and dolphins outside the ETP | 52 |
| | No definition of "regular and significant" | 52 |
| | Other regional fisheries management organizations. | 52 |
| | Indian Ocean Tuna Commission ("IOTC") | 52 |
| | International Commission for the Conservation of Atlantic Tunas ("ICCAT") | 52 |
| | Western and Central Pacific Fisheries Commission ("WCPFC") | 52 |
| | Other regional fisheries – conclusions | 53 |
| (d) | Legal argument | 53 |
| <i>(i)</i> | General | 53 |
| | The need to address all of Mexico's claims | 53 |
| | Order in which the Panel should address Mexico's claims | 53 |
| | The disciplining of unilateral actions by WTO Members | 54 |
| | "zero tolerance" | 54 |
| (ii) | Mexico's discrimination claims under Articles I and III of the GATT 1994 and Article 2.1 of the TBT Agreement | 54 |
| | Issues that are common to Mexico's discrimination claims | 54 |
| | Like products | 54 |
| | Timeframe for examining the facts related to Mexico's de facto discrimination claims | 54 |
| | The facially neutral US measures de facto discriminate | 55 |
| | Mexico's discrimination claims are not dependent on different treatment for fisheries within and outside the ETP | 56 |
| | US pressure on the Mexican fleet to change fishing areas and/or methods | 56 |
| | Private actions | 56 |
| | The relevance of consumer preferences and perceptions | 57 |
| | Article I:1 of the GATT 1994 | 57 |
| | Advantage | 57 |
| | Shall be accorded immediately and unconditionally | 58 |
| | Article III:4 of the GATT 1994 | 58 |
| | Article 2.1 of the TBT Agreement | 58 |
| (iii) | Other TBT Agreement claims | 59 |
| | The US measures are a technical regulation | 59 |
| | Article 2.2 | 59 |
| | The US measures do not fulfil their stated objectives | 59 |
| | The fulfilment of the objectives using less trade restrictive measures | 60 |
| | Article 2.4 | 60 |
| (e) | Conclusions | 60 |

| 2. | United States | 60 |
|------------|---|----|
| (a) | Introduction | 60 |
| (b) | The US dolphin-safe labelling provisions are not inconsistent with Article III:4 of the GATT 1994 | 61 |
| (c) | The US dolphin-safe labelling provisions are not inconsistent with Article I:1 of the GATT 1994 | 63 |
| (d) | The US dolphin-safe labelling provisions are not technical regulations | 65 |
| (e) | The US dolphin-safe labelling provisions are not inconsistent with Articles 2.1, 2.2 or 2.4 of the TBT Agreement | 66 |
| (f) | Conclusion | 69 |
| E. | OPENING ORAL STATEMENTS AT THE SECOND SUBSTANTIVE MEETING | 69 |
| 1. | Mexico | 69 |
| (a) | Introduction | 69 |
| (b) | The US measures are inconsistent with Article I:1 of the GATT 1994 | 69 |
| (c) | The US measures are inconsistent with Article III:4 of the GATT 1994 | 70 |
| <i>(i)</i> | The US measures have modified the conditions of competition for Mexican tuna products | 70 |
| (ii) | There is no legal basis for the two-part test for proving less favourable treatment under GATT Article III:4 that is proposed by the US | 70 |
| (iii) | It is not necessary for Mexico to prove that the differential treatment that is accorded to imported products is "based on origin" | 70 |
| (iv) | It is not necessary for Mexico to prove the US measures "single out imports for different treatment" | 70 |
| (v) | It is not necessary for Mexico to prove that "almost all" imports are disadvantaged | 71 |
| (vi) | The GATT 1947 Panel Report in Belgian Family Allowances supports Mexico's discrimination claims | 71 |
| (vii) | The US measures afford protection to domestic production | |
| (d) | The US measures are inconsistent with the TBT Agreement | |
| (i) | The US measures are technical regulations | |
| (ii) | The US measures are inconsistent with Article 2.1 of the TBT Agreement | |
| (iii) | The US measures are inconsistent with Article 2.2 of the TBT Agreement | |
| (***) | The meaning of "necessary" | |
| | The US erroneously relies on Article 5.6 and footnote 3 of the SPS Agreement | |
| | The application of Article 2.2 to the facts of this dispute | |
| | The US measures do not fulfil their objectives | 73 |
| | The US measures are more trade restrictive than necessary to fulfil a legitimate objective taking into account of the risks non-fulfilment would create | 73 |
| (iv) | The US measures are inconsistent with Article 2.4 of the TBT Agreement | 73 |
| 2. | United States | 74 |

| (a) | Introduction | 74 |
|------------|---|----|
| (b) | Setting on dolphins adversely affects dolphins | 74 |
| (c) | ETP is unique | 74 |
| (d) | Access to US market | 75 |
| (e) | Article III:4 of GATT 1994 | 75 |
| (f) | Article I:1 of GATT 1994 | 76 |
| (g) | Additional points raised by Mexico | 77 |
| (h) | TBT claims | 77 |
| F. | CLOSING ORAL STATEMENTS AT THE SECOND SUBSTANTIVE MEETING | 78 |
| 1. | Mexico | 78 |
| 2. | United States | 79 |
| v. | ARGUMENTS OF THE THIRD PARTIES | 81 |
| A. | Argentina | 81 |
| 1. | Oral statement of Argentina | 81 |
| (a) | Introduction | 81 |
| (b) | Analysis of the United States measures in the light of the TBT Agreement | 81 |
| <i>(i)</i> | The US regulations on dolphin-safe labelling are technical regulations and are therefore subject to the requirements of the TBT Agreement | 81 |
| | The US regulations on dolphin-safe labelling are de jure technical regulations | 81 |
| | The interpretation proposed by Argentina is supported by the Panel Report in EC – Sardines | 82 |
| | The United States regulations on dolphin-safe labelling have the effect of technical regulations, because they operate "de facto" as such | 83 |
| (ii) | The United States regulations on dolphin-safe labelling are inconsistent with Article 2.1 of the TBT Agreement | 83 |
| | The conditions for awarding the dolphin-safe label laid down by the United States regulations are discriminatory | 83 |
| | The mere existence of a positive label regulated by a State gives rise, per se, to discrimination | 84 |
| (iii) | The AIDCP is not a "recognized body" within the meaning of the second paragraph of Annex 1 to the TBT, nor is it a "standardizing body" | 84 |
| (c) | Conclusion | 85 |
| B. | Australia | 85 |
| 1. | Third party submission and oral statement of Australia | 85 |
| (a) | Mexico's claims under the GATT 1994 | 85 |
| <i>(i)</i> | Mexico's claim under Article III:4 of the GATT 1994 | 85 |
| (ii) | Mexico's claim under Article I:1 of the GATT 1994 | 85 |
| (b) | Mexico's claims under the TBT Agreement | 86 |
| <i>(i)</i> | Technical regulation | |
| (c) | Mexico's claims under Article 2.1 and 2.2 of the TBT Agreement | 87 |

| (i) | Article 2.1 of the TBT Agreement | 87 |
|------------|--|----|
| (ii) | Article 2.2 of the TBT Agreement | 87 |
| | Legitimate objective | 87 |
| | More trade restrictive than necessary to fulfil a legitimate objective | 88 |
| | Taking account of the risks non-fulfilment would create | 89 |
| C. | Brazil | 89 |
| 1. | Third party submission and oral statement of Brazil | 89 |
| (a) | Introduction | 89 |
| (b) | Definition of technical regulation | 89 |
| <i>(i)</i> | "Product characteristics" | 89 |
| (ii) | Whether compliance is mandatory | 90 |
| (c) | Article 2.2 of the TBT Agreement | 91 |
| <i>(i)</i> | Assessment of the pursued objectives | 91 |
| (ii) | The role of regional agreements under Article 2.2 of the TBT Agreement | 91 |
| (iii) | Meaning of the phrase "trade restrictive" | 92 |
| (iv) | "Taking account of the risks non-fulfilment would create" | 92 |
| (d) | "International standard" under Article 2.4 of the TBT Agreement | 93 |
| <i>(i)</i> | "International" | 93 |
| (ii) | The systemic importance of the term "standard" | 93 |
| (e) | Article III:4 of the GATT 1994 | 94 |
| (f) | Likeness | 94 |
| D. | Canada | 94 |
| 1. | Oral statement of Canada | 94 |
| (a) | Technical regulation. | 94 |
| (b) | Relevant international standards | 95 |
| E. | CHINA | 96 |
| 1. | Oral statement of China | 96 |
| F. | European Union | 96 |
| 1. | Third party submission and oral statement of the European Union | 96 |
| (a) | Article III:4 of the GATT 1994: "de facto less favourable treatment" | 96 |
| (b) | Article I:1 of the GATT 1994: the notion of "unconditionally" | 97 |
| (c) | Article I:1 of the GATT 1994: the notion of "advantage, favour, privilege or immunity" | 97 |
| (d) | Articles I:1 and III:4 of the GATT 1994: the notion of "like products" | 97 |
| (e) | Annex 1.1 of the TBT Agreement: labelling requirements and product characteristics | 98 |
| (f) | Annex 1.1 of the TBT Agreement: mandatory compliance | 98 |

| G. | Guatemala | 100 |
|------------|--|-----|
| 1. | Oral statement of Guatemala | 100 |
| (a) | General remarks | 100 |
| (b) | Claims under Articles I:1 and III:4 of the GATT of 1994 | 101 |
| (c) | Claims under the TBT Agreement | 102 |
| H. | JAPAN | 103 |
| 1. | Third party submission of Japan | 103 |
| (a) | Introduction | 103 |
| (b) | The Panel should carefully apply the threshold definition of "technical regulation" in Annex 1 of the TBT Agreement | 103 |
| <i>(i)</i> | The Panel should follow precedents regarding whether the measures at issue are "documents with which compliance is mandatory" | 103 |
| (ii) | The Panel should carefully consider whether the measures at issue are a "document which lays down product characteristics" | 104 |
| (c) | The Panel should treat a Member's identification of its "legitimate objectives" deferentially | 104 |
| (d) | The Panel should take into account the principles regarding "international standards" agreed by the Committee on Technical Barriers to Trade (TBT Committee) | 104 |
| (e) | The Panel should scrutinize carefully whether the measures at issue accord less favourable treatment of imported products within the meaning of Article III:4 of the GATT 1994 | 105 |
| (f) | Conclusion | 105 |
| 2. | Oral statement of Japan | 105 |
| (a) | Introduction | 105 |
| (b) | The Panel must respect the distinction between mandatory and voluntary measures | 105 |
| (c) | The Panel must evaluate the TBT Agreement's negotiating history with care, and as a whole | 105 |
| (d) | The Panel should be aware of the past practice in terms of the order of analysis | 106 |
| (e) | The Panel should not read procedural requirement in Article 2.2 of the TBT Agreement | 106 |
| I. | Korea | 106 |
| 1. | Third party submission and oral statement of Korea | 106 |
| (a) | Mexico's Article III:4 of the GATT1994 claim and trade effect | 106 |
| (b) | Technical regulation of Article 2 of the TBT Agreement | 107 |
| J. | NEW ZEALAND | 108 |
| 1. | Third party submission of New Zealand | 108 |
| (a) | Introduction | |
| (b) | The definition of technical regulation: Annex 1.1 | 108 |
| <i>(i)</i> | Coverage of labelling requirements | 108 |

| (ii) | Is the United States' measure a "mandatory" regulation? | 109 |
|------------|--|-----|
| (c) | The definition of relevant international standard under Article 2.4 | 109 |
| <i>(i)</i> | The requirement that the standard be made by an international body | 110 |
| (ii) | First criteria for the international body: specified activities of standardization | 110 |
| (iii) | Second criteria: openness to at least all Members of the WTO | 110 |
| 2. | Oral statement of New Zealand | 111 |
| K. | THAILAND | 112 |
| 1. | Oral statement of Thailand | 112 |
| L. | Turkey | 112 |
| 1. | Third party submission of Turkey | 112 |
| (a) | Introduction | 112 |
| (b) | Articles 2.2 and 2.4 of the TBT Agreement | 112 |
| (c) | Conclusion | 115 |
| VI. | INTERIM REVIEW | 115 |
| A. | General | 115 |
| B. | DESCRIPTIVE PART (SECTION II OF THE REPORT) | 115 |
| C. | FINDINGS | 116 |
| 1. | General (Section VII.A) | 116 |
| 2. | Whether the measures at issue constitute a "technical regulation" within the meaning of Annex 1.1 of the TBT Agreement (Section VII.B.1) | 116 |
| 3. | Article 2.1 of the TBT Agreement (Section VII.B.2) | 117 |
| 4. | Article 2.2 of the TBT Agreement (Section VII.B.3) | 119 |
| 5. | Article 2.4 of the TBT Agreement (Section VII.B.4) | 120 |
| 6. | Exercise of judicial economy on Mexico's claims under Articles I and III:4 of GATT 1994 (Section VII.C) | 121 |
| VII. | FINDINGS | 123 |
| A. | GENERAL | 123 |
| 1. | Procedural issue: Amicus curiae brief | 123 |
| 2. | Examination of the measures together | 125 |
| (a) | Arguments of the parties | 126 |
| (b) | Analysis by the Panel | 126 |
| 3. | Order of analysis of the claims | 129 |
| (a) | Arguments of the parties | 129 |
| (b) | Analysis by the Panel | 130 |
| B. | MEXICO'S CLAIMS UNDER THE TBT AGREEMENT | 133 |
| 1. | Whether the measures at issue constitute a "technical regulation" within the meaning of Annex 1.1 of the TBT Agreement | 133 |

| whether the measures apply to an identifiable product or group of products | 133 |
|--|---|
| Whether the US dolphin-safe labelling provisions lay down one or more "characteristics" of the products | 136 |
| Arguments of the parties | 136 |
| Analysis by the Panel | 138 |
| Whether compliance with the US dolphin-safe labelling provisions is "mandatory" within the meaning of Annex 1.1 | 140 |
| Arguments of the parties | 140 |
| Analysis by the Panel | 144 |
| Interpretation of the term "mandatory" in Annex 1.1 | 144 |
| Whether the US dolphin-safe provisions are "mandatory" within the meaning of Annex 1.1 | 146 |
| Separate opinion | 152 |
| Whether the US dolphin-safe labelling provisions are inconsistent with Article 2.1 of the TBT Agreement | 162 |
| Overall approach | 162 |
| Main arguments of the parties | 162 |
| Approach of the Panel | 166 |
| Whether the products at issue are like | 166 |
| Meaning of the term "like products" in Article 2.1 of the TBT Agreement | 167 |
| Products at issue | 170 |
| Whether Mexican tuna products are like tuna products originating in the United States or any other country | 171 |
| Whether Mexican tuna products are afforded less favourable treatment than tuna products originating in the United States and other countries in respect of the US dolphin-safe labelling provisions | 175 |
| Arguments of the parties | 175 |
| Analysis by the Panel | 179 |
| Meaning of "less favourable treatment" in Article 2.1 of the TBT Agreement | 179 |
| Whether less favourable treatment is afforded to Mexican tuna products than to tuna products of US or originating in any other country | 181 |
| Whether access to the US dolphin-safe label constitutes an advantage on the US market | 182 |
| Whether Mexican tuna products are denied access to the advantage offered by the dolphin-safe label and are thereby disadvantaged, compared to tuna products originating in the United States or in any other country | 184 |
| The regulatory distinctions at issue: setting on dolphins as opposed to not setting on dolphins | 186 |
| Fishing practices of US and Mexican tuna fleets, and their impact on access to the US dolphin-safe label for Mexican tuna products and tuna products originating in the United States or in any other country | 189 |
| | |
| Relationship between the practices of fleets and access to the label | 191 |
| | Whether the US dolphin-safe labelling provisions lay down one or more "characteristics" of the products |

| | Adaptation costs | 196 |
|------------|---|-----|
| | Fishing practices of fleets and origin of tuna products | 198 |
| | Mexican tuna products on the US market | 199 |
| | "Pressure" to adapt to the US standard | 204 |
| | Conclusion | 205 |
| 3. | Whether the US dolphin-safe labelling provisions are inconsistent with Article 2.2 of the TBT Agreement | 206 |
| (a) | Whether the US dolphin-safe labelling provisions pursue a legitimate objective | 208 |
| <i>(i)</i> | The objectives pursued by the US measures | 209 |
| | Arguments of the parties | 209 |
| | Analysis by the Panel | 210 |
| | Consumer information objective | 211 |
| | Dolphin protection objective | 212 |
| (ii) | Whether the objectives pursued by the United States are legitimate | 215 |
| | Arguments by the parties | 215 |
| | Analysis by the Panel | 217 |
| (b) | Whether the US dolphin-safe provisions are more trade-restrictive than necessary to fulfil their objectives, taking into account the risks that non-fulfilment would create | 219 |
| <i>(i)</i> | Approach of the Panel | 219 |
| | Arguments by the parties | 219 |
| | Analysis by the Panel | 220 |
| (ii) | Whether the US dolphin-safe labelling provisions are more trade-restrictive than necessary to ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught in conditions that are harmful to dolphins | 224 |
| | The contribution of the US dolphin-safe provisions to the US objective of ensuring that consumers are not misled about whether the tuna contained in tuna products was caught in a manner that adversely affects dolphins | 225 |
| | Meaning of the phrase "tuna caught in a manner that adversely affects dolphins" | 227 |
| | Denial of access to the label for tuna caught by setting on dolphins (and high seas driftnet fishing) | 229 |
| | Conditions of access to the label for tuna caught by other methods | 233 |
| | Whether fishing techniques other than setting on dolphins may adversely affect them | 235 |
| | Whether access to the US dolphin-safe label is available to tuna caught in conditions that may adversely affect dolphins | 242 |
| | Adverse effects of fishing activities on dolphins in and outside of the ETP | 246 |
| | Whether the alternative measure proposed by Mexico provides a reasonably available less-trade restrictive means of achieving the same level of protection | 251 |
| (iii) | Whether the US dolphin-safe provisions are more trade-restrictive than necessary to contribute to the protection of dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins | 254 |
| | Arguments by the parties | 254 |

| | Analysis by the Panel | 256 |
|------------|---|-----|
| | The contribution of the US dolphin-safe provisions to protecting dolphins, by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins | 256 |
| | Whether the alternative suggested by Mexico provides a reasonably available less trade restrictive means of achieving the same level of protection | 258 |
| (iv) | Conclusion | 262 |
| 4. | Whether the US dolphin-safe labelling provisions are inconsistent with Article 2.4 of the TBT Agreement | 263 |
| (a) | Overall approach | 263 |
| (b) | Whether the "dolphin-safe" definition contained in the AIDCP Tuna Tracking System Resolution is a relevant international standard | 264 |
| <i>(i)</i> | Arguments of the parties | 264 |
| (ii) | Analysis by the Panel | 273 |
| | The notion of "international standard" in Article 2.4 of the TBT Agreement | 273 |
| | Whether the AIDCP dolphin-safe definition and certification constitutes a "standard" for the purposes of Article 2.4 | 274 |
| | Whether the dolphin-safe definition in the AIDCP tuna tracking and verification resolution has been approved by an international standardizing or standards organization and made available to the public. | 277 |
| | Whether the AIDCP dolphin-safe definition and certification are "relevant" | 281 |
| (c) | Whether the United States used the AIDCP standard as a basis for its dolphin-safe labelling provisions | 283 |
| <i>(i)</i> | Arguments of the parties | 283 |
| (ii) | Analysis by the Panel | 284 |
| (d) | Whether the AIDCP dolphin-safe standard would be an ineffective or inappropriate means for the fulfilment of the legitimate objective(s) pursued by the United States | 285 |
| <i>(i)</i> | Arguments of the parties | 285 |
| (ii) | Analysis by the Panel | 286 |
| C. | MEXICO'S CLAIMS UNDER THE GATT 1994 | 291 |
| VIII. | RULINGS AND RECOMMENDATIONS | 292 |

TABLE OF WTO DISPUTES CITED IN THIS REPORT

| Short Title | Full Case Title and Citation |
|---|--|
| Argentina – Hides and Leather | Panel Report, Argentina – Measures Affecting the Export of Bovine Hides and Import of Finished Leather, WT/DS155/R and Corr.1, adopted 16 February 2001, DSR 2001:V, 1779 |
| Australia – Apples | Appellate Body Report, Australia – Measures Affecting the Importation of Apples from New Zealand, WT/DS367/AB/R, adopted 17 December 2010 |
| Australia – Apples | Panel Report, <i>Australia – Measures Affecting the Importation of Apples from New Zealand</i> , WT/DS367/R, adopted 17 December 2010, as modified by Appellate Body Report WT/DS367/AB/R |
| Australia – Salmon | Appellate Body Report, <i>Australia – Measures Affecting Importation of Salmon</i> , WT/DS18/AB/R, adopted 6 November 1998, DSR 1998:VIII, 3327 |
| Australia – Salmon | Panel Report, <i>Australia – Measures Affecting Importation of Salmon</i> , WT/DS18/R and Corr.1, adopted 6 November 1998, as modified by Appellate Body Report WT/DS18/AB/R, DSR 1998:VIII, 3407 |
| Brazil – Desiccated Coconut | Appellate Body Report, <i>Brazil – Measures Affecting Desiccated Coconut</i> , WT/DS22/AB/R, adopted 20 March 1997, DSR 1997:I, 167 |
| Brazil – Desiccated Coconut | Panel Report, <i>Brazil – Measures Affecting Desiccated Coconut</i> , WT/DS22/R, adopted 20 March 1997, as upheld by Appellate Body Report WT/DS22/AB/R, DSR 1997:I, 189 |
| Brazil – Retreaded Tyres | Appellate Body Report, Brazil – Measures Affecting Imports of Retreaded Tyres, WT/DS332/AB/R, adopted 17 December 2007, DSR 2007:IV, 1527 |
| Brazil – Retreaded Tyres | Panel Report, <i>Brazil – Measures Affecting Imports of Retreaded Tyres</i> , WT/DS332/R, adopted 17 December 2007, as modified by Appellate Body Report WT/DS332/AB/R, DSR 2007:V, 1649 |
| Canada – Aircraft | Appellate Body Report, <i>Canada – Measures Affecting the Export of Civilian Aircraft</i> , WT/DS70/AB/R, adopted 20 August 1999, DSR 1999:III, 1377 |
| Canada – Aircraft | Panel Report, Canada – Measures Affecting the Export of Civilian Aircraft, WT/DS70/R, adopted 20 August 1999, as upheld by Appellate Body Report WT/DS70/AB/R, DSR 1999:IV, 1443 |
| Canada – Autos | Appellate Body Report, <i>Canada – Certain Measures Affecting the Automotive Industry</i> , WT/DS139/AB/R, WT/DS142/AB/R, adopted 19 June 2000, DSR 2000:VI, 2985 |
| Canada – Autos | Panel Report, Canada – Certain Measures Affecting the Automotive Industry, WT/DS139/R, WT/DS142/R, adopted 19 June 2000, as modified by Appellate Body Report WT/DS139/AB/R, WT/DS142/AB/R, DSR 2000:VII, 3043 |
| Canada – Wheat Exports and Grain Imports | Appellate Body Report, Canada – Measures Relating to Exports of Wheat and Treatment of Imported Grain, WT/DS276/AB/R, adopted 27 September 2004, DSR 2004:VI, 2739 |
| Canada – Wheat Exports and Grain Imports | Panel Report, Canada – Measures Relating to Exports of Wheat and Treatment of Imported Grain, WT/DS276/R, adopted 27 September 2004, as upheld by Appellate Body Report WT/DS276/AB/R, DSR 2004:VI, 2817 |
| Chile – Alcoholic Beverages | Appellate Body Report, <i>Chile – Taxes on Alcoholic Beverages</i> , WT/DS87/AB/R, WT/DS110/AB/R, adopted 12 January 2000, DSR 2000:I, 281 |

| Short Title | Full Case Title and Citation |
|---|--|
| Chile – Alcoholic Beverages | Panel Report, <i>Chile – Taxes on Alcoholic Beverages</i> , WT/DS87/R, WT/DS110/R, adopted 12 January 2000, as modified by Appellate Body Report WT/DS87/AB/R, WT/DS110/AB/R, DSR 2000:I, 303 |
| China – Publications and Audiovisual Products | Appellate Body Report, China – Measures Affecting Trading Rights and Distribution Services for Certain Publications and Audiovisual Entertainment Products, WT/DS363/AB/R, adopted 19 January 2010 |
| China – Publications and Audiovisual Products | Panel Report, China – Measures Affecting Trading Rights and Distribution Services for Certain Publications and Audiovisual Entertainment Products, WT/DS363/R and Corr.1, adopted 19 January 2010, as modified by Appellate Body Report WT/DS363/AB/R |
| Colombia – Ports of Entry | Panel Report, Colombia – Indicative Prices and Restrictions on Ports of Entry, WT/DS366/R and Corr.1, adopted 20 May 2009 |
| Dominican Republic – Import and Sale of Cigarettes | Appellate Body Report, <i>Dominican Republic – Measures Affecting the Importation and Internal Sale of Cigarettes</i> , WT/DS302/AB/R, adopted 19 May 2005, DSR 2005:XV, 7367 |
| Dominican Republic – Import and Sale of Cigarettes | Panel Report, <i>Dominican Republic – Measures Affecting the Importation and Internal Sale of Cigarettes</i> , WT/DS302/R, adopted 19 May 2005, as modified by Appellate Body Report WT/DS302/AB/R, DSR 2005:XV, 7425 |
| EC – Asbestos | Appellate Body Report, European Communities – Measures Affecting Asbestos and Asbestos-Containing Products, WT/DS135/AB/R, adopted 5 April 2001, DSR 2001:VII, 3243 |
| EC – Asbestos | Panel Report, European Communities – Measures Affecting Asbestos and Asbestos-Containing Products, WT/DS135/R and Add.1, adopted 5 April 2001, as modified by Appellate Body Report WT/DS135/AB/R, DSR 2001:VIII, 3305 |
| EC – Bananas III | Appellate Body Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, WT/DS27/AB/R, adopted 25 September 1997, DSR 1997:II, 591 |
| EC – Bananas III (Ecuador) | Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, Complaint by Ecuador, WT/DS27/R/ECU, adopted 25 September 1997, as modified by Appellate Body Report WT/DS27/AB/R, DSR 1997:III, 1085 |
| EC – Bananas III (Guatemala and Honduras) | Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, Complaint by Guatemala and Honduras, WT/DS27/R/GTM, WT/DS27/R/HND, adopted 25 September 1997, as modified by Appellate Body Report WT/DS27/AB/R, DSR 1997:II, 695 |
| EC – Bananas III (Mexico) | Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, Complaint by Mexico, WT/DS27/R/MEX, adopted 25 September 1997, as modified by Appellate Body Report WT/DS27/AB/R, DSR 1997:II, 803 |
| EC – Bananas III (US) | Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, Complaint by the United States, WT/DS27/R/USA, adopted 25 September 1997, as modified by Appellate Body Report WT/DS27/AB/R, DSR 1997:II, 943 |

| Short Title | Full Case Title and Citation |
|--|--|
| EC – Bananas III (Article 21.5 – Ecuador II) / EC – Bananas III (Article 21.5 – US) | Appellate Body Reports, European Communities – Regime for the Importation, Sale and Distribution of Bananas – Second Recourse to Article 21.5 of the DSU by Ecuador, WT/DS27/AB/RW2/ECU, adopted 11 December 2008, and Corr.1 / European Communities – Regime for the Importation, Sale and Distribution of Bananas – Recourse to Article 21.5 of the DSU by the United States, WT/DS27/AB/RW/USA and Corr.1, adopted 22 December 2008 |
| EC – Bananas III (Article 21.5 – Ecuador II) | Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas – Second Recourse to Article 21.5 of the DSU by Ecuador, WT/DS27/RW2/ECU, adopted 11 December 2008, as modified by Appellate Body Report WT/DS27/AB/RW2/ECU |
| EC – Hormones | Appellate Body Report, <i>EC Measures Concerning Meat and Meat Products</i> (<i>Hormones</i>), WT/DS26/AB/R, WT/DS48/AB/R, adopted 13 February 1998, DSR 1998:I, 135 |
| EC – Hormones (Canada) | Panel Report, EC Measures Concerning Meat and Meat Products (Hormones), Complaint by Canada, WT/DS48/R/CAN, adopted 13 February 1998, as modified by Appellate Body Report WT/DS26/AB/R, WT/DS48/AB/R, DSR 1998:II, 235 |
| EC – Hormones (US) | Panel Report, EC Measures Concerning Meat and Meat Products (Hormones), Complaint by the United States, WT/DS26/R/USA, adopted 13 February 1998, as modified by Appellate Body Report WT/DS26/AB/R, WT/DS48/AB/R, DSR 1998:III, 699 |
| EC – Bananas III (Article 21.5 – US) | Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas – Recourse to Article 21.5 of the DSU by the United States, WT/DS27/RW/USA and Corr.1, adopted 22 December 2008, upheld by Appellate Body Report WT/DS27/AB/RW/USA |
| EC – Sardines | Appellate Body Report, <i>European Communities – Trade Description of Sardines</i> , WT/DS231/AB/R, adopted 23 October 2002, DSR 2002:VIII, 3359 |
| EC – Sardines | Panel Report, <i>European Communities – Trade Description of Sardines</i> , WT/DS231/R and Corr.1, adopted 23 October 2002, as modified by Appellate Body Report WT/DS231/AB/R, DSR 2002:VIII, 3451 |
| EC – Trademarks and Geographical Indications (Australia) | Panel Report, European Communities – Protection of Trademarks and Geographical Indications for Agricultural Products and Foodstuffs, Complaint by Australia, WT/DS290/R, adopted 20 April 2005, DSR 2005:X, 4603 |
| India – Autos | Appellate Body Report, <i>India – Measures Affecting the Automotive Sector</i> , WT/DS146/AB/R, WT/DS175/AB/R, adopted 5 April 2002, DSR 2002:V, 1821 |
| India – Autos | Panel Report, <i>India – Measures Affecting the Automotive Sector</i> , WT/DS146/R, WT/DS175/R and Corr.1, adopted 5 April 2002, DSR 2002:V, 1827 |
| India – Quantitative Restrictions | Panel Report, <i>India – Quantitative Restrictions on Imports of Agricultural, Textile and Industrial Products</i> , WT/DS90/R, adopted 22 September 1999, as upheld by Appellate Body Report WT/DS90/AB/R, DSR 1999:V, 1799 |
| Indonesia – Autos | Panel Report, <i>Indonesia – Certain Measures Affecting the Automobile Industry</i> , WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R and Corr.1 and 2, adopted 23 July 1998, and Corr. 3 and 4, DSR 1998:VI, 2201 |

| Short Title | Full Case Title and Citation |
|--|--|
| Japan – Agricultural Products II | Appellate Body Report, <i>Japan – Measures Affecting Agricultural Products</i> , WT/DS76/AB/R, adopted 19 March 1999, DSR 1999:I, 277 |
| Japan – Agricultural Products II | Panel Report, <i>Japan – Measures Affecting Agricultural Products</i> , WT/DS76/R, adopted 19 March 1999, as modified by Appellate Body Report WT/DS76/AB/R, DSR 1999:I, 315 |
| Japan – Alcoholic Beverages II | Appellate Body Report, <i>Japan – Taxes on Alcoholic Beverages</i> , WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, adopted 1 November 1996, DSR 1996:I, 97 |
| Japan – Alcoholic Beverages II | Panel Report, <i>Japan – Taxes on Alcoholic Beverages</i> , WT/DS8/R, WT/DS10/R, WT/DS11/R, adopted 1 November 1996, as modified by Appellate Body Report WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, DSR 1996:I, 125 |
| Japan – Apples | Appellate Body Report, <i>Japan – Measures Affecting the Importation of Apples</i> , WT/DS245/AB/R, adopted 10 December 2003, DSR 2003:IX, 4391 |
| Japan – Apples | Panel Report, <i>Japan – Measures Affecting the Importation of Apples</i> , WT/DS245/R, adopted 10 December 2003, as upheld by Appellate Body Report WT/DS245/AB/R, DSR 2003:IX, 4481 |
| Japan – Film | Panel Report, <i>Japan – Measures Affecting Consumer Photographic Film and Paper</i> , WT/DS44/R, adopted 22 April 1998, DSR 1998:IV, 1179 |
| Korea – Alcoholic Beverages | Appellate Body Report, <i>Korea – Taxes on Alcoholic Beverages</i> , WT/DS75/AB/R, WT/DS84/AB/R, adopted 17 February 1999, DSR 1999:I, 3 |
| Korea – Alcoholic Beverages | Panel Report, <i>Korea – Taxes on Alcoholic Beverages</i> , WT/DS75/R, WT/DS84/R, adopted 17 February 1999, as modified by Appellate Body Report WT/DS75/AB/R, WT/DS84/AB/R, DSR 1999:I, 44 |
| Korea – Various Measures on Beef | Appellate Body Report, <i>Korea – Measures Affecting Imports of Fresh, Chilled and Frozen Beef</i> , WT/DS161/AB/R, WT/DS169/AB/R, adopted 10 January 2001, DSR 2001:I, 5 |
| Korea – Various Measures on Beef | Panel Report, <i>Korea – Measures Affecting Imports of Fresh, Chilled and Frozen Beef</i> , WT/DS161/R, WT/DS169/R, adopted 10 January 2001, as modified by Appellate Body Report WT/DS161/AB/R, WT/DS169/AB/R, DSR 2001:I, 59 |
| Mexico – Taxes on Soft Drinks | Appellate Body Report, <i>Mexico – Tax Measures on Soft Drinks and Other Beverages</i> , WT/DS308/AB/R, adopted 24 March 2006, DSR 2006:I, 3 |
| Mexico – Taxes on Soft Drinks | Panel Report, <i>Mexico – Tax Measures on Soft Drinks and Other Beverages</i> , WT/DS308/R, adopted 24 March 2006, as modified by Appellate Body Report WT/DS308/AB/R, DSR 2006:I, 43 |
| US – Anti-Dumping and Countervailing Duties (China) | Appellate Body Report, <i>United States – Definitive Anti-Dumping and Countervailing Duties on Certain Products from China</i> , WT/DS379/AB/R, adopted 25 March 2011 |
| US – Anti-Dumping and Countervailing Duties (China) | Panel Report, <i>United States – Definitive Anti-Dumping and Countervailing Duties on Certain Products from China</i> , WT/DS379/R, adopted 25 March 2011, as modified by Appellate Body Report WT/DS379/AB/R |
| US – Anti-Dumping Measures on PET Bags | Panel Report, <i>United States – Anti-Dumping Measures on Polyethylene Retail Carrier Bags from Thailand</i> , WT/DS383/R, adopted 18 February 2010 |

| Short Title | Full Case Title and Citation |
|---|--|
| US – COOL (Canada/Mexico) | United States – Certain Country of Origin Labelling (COOL) WT/DS384 (Canada) and WT/DS386 (Mexico) |
| US – Corrosion-Resistant Steel Sunset Review | Appellate Body Report, <i>United States – Sunset Review of Anti-Dumping Duties on Corrosion-Resistant Carbon Steel Flat Products from Japan</i> , WT/DS244/AB/R, adopted 9 January 2004, DSR 2004:I, 3 |
| US – Corrosion-Resistant Steel Sunset Review | Panel Report, <i>United States – Sunset Review of Anti-Dumping Duties on Corrosion-Resistant Carbon Steel Flat Products from Japan</i> , WT/DS244/R, adopted 9 January 2004, as modified by Appellate Body Report WTDS244/AB/R, DSR 2004:1, 85 |
| US – Export Restraints | Panel Report, <i>United States – Measures Treating Exports Restraints as Subsidies</i> , WT/DS194/R and Corr.2, adopted 23 August 2001, DSR 2001:XI, 5767 |
| US – FSC (Article 21.5 – EC) | Appellate Body Report, <i>United States – Tax Treatment for "Foreign Sales Corporations" – Recourse to Article 21.5 of the DSU by the European Communities</i> , WT/DS108/AB/RW, adopted 29 January 2002, DSR 2002:I, 55 |
| US – FSC (Article 21.5 – EC) | Panel Report, <i>United States – Tax Treatment for "Foreign Sales Corporations" – Recourse to Article 21.5 of the DSU by the European Communities</i> , WT/DS108/RW, adopted 29 January 2002, as modified by Appellate Body Report WT/DS108/AB/RW, DSR 2002:I, 119 |
| US – Gambling | Appellate Body Report, <i>United States – Measures Affecting the Cross-Border Supply of Gambling and Betting Services</i> , WT/DS285/AB/R, adopted 20 April 2005, DSR 2005:XII, 5663 (Corr.1, DSR 2006:XII, 5475) |
| US – Gambling | Panel Report, <i>United States – Measures Affecting the Cross-Border Supply of Gambling and Betting Services</i> , WT/DS285/R, adopted 20 April 2005, as modified by Appellate Body Report WT/DS285/AB/R, DSR 2005:XII, 5797 |
| US – Gasoline | Appellate Body Report, <i>United States – Standards for Reformulated and Conventional Gasoline</i> , WT/DS2/AB/R, adopted 20 May 1996, DSR 1996:I, 3 |
| US – Gasoline | Panel Report, <i>United States – Standards for Reformulated and Conventional Gasoline</i> , WT/DS2/R, adopted 20 May 1996, as modified by Appellate Body Report WT/DS2/AB/R, DSR 1996:I, 29 |
| US – Poultry (China) | Panel Report, <i>United States – Certain Measures Affecting Imports of Poultry from China</i> , WT/DS392/R, adopted 25 October 2010 |
| US – Section 211 Appropriations Act | Appellate Body Report, <i>United States – Section 211 Omnibus Appropriations Act of 1998</i> , WT/DS176/AB/R, adopted 1 February 2002, DSR 2002:II, 589 |
| US – Section 211 Appropriations Act | Panel Report, <i>United States – Section 211 Omnibus Appropriations Act of 1998</i> , WT/DS176/R, adopted 1 February 2002, as modified by Appellate Body Report WT/DS176/AB/R, DSR 2002:II, 683 |
| US – Shrimp | Appellate Body Report, <i>United States – Import Prohibition of Certain Shrimp and Shrimp Products</i> , WT/DS58/AB/R, adopted 6 November 1998, DSR 1998:VII, 2755 |
| US – Shrimp | Panel Report, <i>United States – Import Prohibition of Certain Shrimp and Shrimp Products</i> , WT/DS58/R and Corr.1, adopted 6 November 1998, as modified by Appellate Body Report WT/DS58/AB/R, DSR 1998:VII, 2821 |

| Short Title | Full Case Title and Citation |
|------------------------------|--|
| US – Softwood Lumber IV | Appellate Body Report, <i>United States – Final Countervailing Duty Determination with Respect to Certain Softwood Lumber from Canada</i> , WT/DS257/AB/R, adopted 17 February 2004, DSR 2004:II, 571 |
| US – Softwood Lumber IV | Panel Report, <i>United States – Final Countervailing Duty Determination with Respect to Certain Softwood Lumber from Canada</i> , WT/DS257/R and Corr.1, adopted 17 February 2004, as modified by Appellate Body Report WT/DS257/AB/R, DSR 2004:II, 641 |
| US – Upland Cotton | Appellate Body Report, <i>United States – Subsidies on Upland Cotton</i> , WT/DS267/AB/R, adopted 21 March 2005, DSR 2005:I, 3 |
| US – Upland Cotton | Panel Report, <i>United States – Subsidies on Upland Cotton</i> , WT/DS267/R, Corr.1, and Add.1 to Add.3, adopted 21 March 2005, as modified by Appellate Body Report WT/DS267/AB/R, DSR 2005:II, 299 |
| US – Wool Shirts and Blouses | Appellate Body Report, <i>United States – Measure Affecting Imports of Woven Wool Shirts and Blouses from India</i> , WT/DS33/AB/R, adopted 23 May 1997, and Corr.1, DSR 1997:I, 323 |
| US – Wool Shirts and Blouses | Panel Report, <i>United States – Measure Affecting Imports of Woven Wool Shirts and Blouses from India</i> , WT/DS33/R, adopted 23 May 1997, as upheld by Appellate Body Report WT/DS33/AB/R, DSR 1997:I, 343 |

TABLE OF GATT DISPUTES CITED IN THIS REPORT

| Short Title | Full Case Title and Citation |
|---------------------------------|---|
| Belgium – Family Allowances | GATT Panel Report, <i>Income Tax Practices Maintained by Belgium</i> , L/4424, adopted 7 December 1981, BISD 23S/127 (See also L/5271, BISD 28S/114) |
| Border Tax Adjustments | Report of the Working Party on Border Tax Adjustments, L/3464, adopted 2 December 1970, BISD 18S/97 |
| EEC – Imports of Beef | GATT Panel Report, European Economic Community – Imports of Beef from Canada, L/5099, adopted 10 March 1981, BISD 28S/92 |
| Italy – Agricultural Machinery | GATT Panel Report, <i>Italian Discrimination Against Imported Agricultural Machinery</i> , L/833, adopted 23 October 1958, BISD 7S/60 |
| Japan – Alcoholic Beverages I | GATT Panel Report, <i>Japan – Customs Duties, Taxes and Labelling Practices on Imported Wines and Alcoholic Beverages</i> , L/6216, adopted 10 November 1987, BISD 34S/83 |
| Japan – Leather (US II) | GATT Panel Report, <i>Panel on Japanese Measures on Imports of Leather</i> , L/5623, adopted 15 May 1984, BISD 31S/94 |
| Japan – Semi-Conductors | GATT Panel Report, <i>Japan – Trade in Semi-Conductors</i> , L/6309, adopted 4 May 1988, BISD 35S/116 |
| Japan – SPF Dimension Lumber | GATT Panel Report, Canada/Japan – Tariff on Imports of Spruce, Pine, Fir (SPF) Dimension Lumber, L/6470, adopted 19 July 1989, BISD 36S/167 |
| Spain – Unroasted Coffee | GATT Panel Report, Spain – Tariff Treatment of Unroasted Coffee, L/5135, adopted 11 June 1981, BISD 28S/102 |
| US – Tuna (Mexico) | GATT Panel Report, <i>United States – Restrictions on Imports of Tuna</i> , DS21/R, 3 September 1991, unadopted, BISD 39S/155 |
| US – Tuna (EEC) | GATT Panel Report, <i>United States – Restrictions on Imports of Tuna</i> , DS29/R, 16 June 1994, unadopted |
| US – Section 337 Tariff Act | GATT Panel Report, <i>United States Section 337 of the Tariff Act of 1930</i> , L/6439, adopted 7 November 1989, BISD 36S/345 |

LIST OF ABBREVIATIONS

| AIDCP | Agreement on International Dolphin Conservation Program |
|-------------------|---|
| ASCOBANS | Agreement on the Conservation of Small Cetaceans of the Baltic and North Seas |
| ASTM | American Society for Testing Materials |
| CHESS | Chase Encirclement Stress Studies |
| Codex | Codex Alimentarius Commission |
| DOC | Department of Commerce |
| DML | Dolphin Mortality Limit |
| DPCIA | Dolphin Protection Consumer Information Act of 1990 |
| DSU | Understanding on Rules and Procedures Governing the Settlement of Disputes |
| EEZ | Exclusive economic zone |
| EII | Earth Island Institute |
| ETP | Eastern Tropical Pacific Ocean |
| FADs | Fish Aggregating Devices |
| GATS | General Agreement on Trade in Services |
| GATT 1994 | General Agreement on Tariffs and Trade 1994 |
| IATTC | Inter-America Tropical Tuna Commission |
| ICCAT | International Commission for the Conservation of Atlantic Tunas |
| IDCPA | International Dolphin Conversation Program Act |
| IEEE | Institute for Electrical and Electronics Engineers |
| IOTC | Indian Ocean Tuna Commission |
| ISO | International Organization for Standardization |
| ISO/IEC | ISO/International Electrotechnical Commission |
| MFN | Most favoured nation |
| MMPA | Marine Mammal Protection Act of 1972 |
| NOAA | National Oceanic Atmospheric Administration |
| NGOs | Non-governmental organizations |
| NMFS | National Marine Fisheries Service |
| PBR | Potential biological removal |
| PPMs | Processes and production methods |
| SPS Agreement | Agreement on Sanitary and Phytosanitary Measures |
| SWFSC | Southwest Fisheries Science Center |
| TBT Agreement | Agreement on Technical Barriers to Trade |
| TTVP | Tuna Tracking Verification Program |
| USC | United States Code |
| Vienna Convention | Vienna Convention on the Law of the Treaties |

| WCPFC | Western and Central Pacific Fisheries Commission |
|-------|--|
| WCPO | Western Central Pacific Ocean |
| WDCS | Whale and Dolphin Conservation Society |

LIST OF US EXHIBITS CITED IN THIS REPORT

| EXHIBIT NO. | TITLE |
|-------------|---|
| US-1C | National Marine Fisheries Service, U.S. Imports of Tuna 2005-2010 |
| US-2 | National Marine Fisheries Service, Total U.S. Imports of Tuna 2009 |
| US-4 | Noren, S.R., and E.F. Edwards. 2007. Physiological and behavioral development in Delphinid calves: implications for calf separation and mortality due to tuna purse-seine sets. Marine Mammal Science 23: 15-29 |
| US-5(05) | Dolphin Protection Consumer Information Act, 16 USC 1385 et seq |
| US-10 | Donahue, M.A. and E.F. Edwards. 1996. An annotated bibliography of available literature regarding cetacean interactions with tuna purse-seine fisheries outside of the eastern tropical Pacific Ocean. NOAA Administrative Report LJ-96-20. |
| US-11 | Myrick, A.C., and P.C. Perkins. 1995. Adrenocortical color darkness and correlates as indicators of continuous acute premortem stress in chased and purse-seine captured male dolphins, Pathophysiology 2:191-204. |
| US-15 | IATTC Purse Seine Vessel Register |
| US-19 | Reilly et al. 2005. Report of the scientific research program under the International Dolphin Conservation Program Act. NOAA-TM-NMFS-SWFSC-372, p. 101 |
| US-20 | Gerrodette, T., G. Watters, W. Perryman, and L. Ballance. 2008. Estimates of 2006 Dolphin Abundance in the Eastern Tropical Pacific, with Revised Estimates from 1986-2003. NOAA Tech. Memo. NMFS-SWFSC-422 |
| US-21 | Wade, P.R., G.M. Watters, T. Gerrodette, and S.B. Reilly, 2007. Depletion of spotted and spinner dolphins in the eastern tropical Pacific: modeling hypotheses for their lack of recovery. Marine Ecology Progress Series 343:1-14. |
| US-22 | Gerrodette, T., and J. Forcada. 2005. Non-recovery of two spotted and spinner dolphin populations in the eastern tropical Pacific Ocean. Marine Ecology Progress Series 291:1-21 |
| US-24 | IATTC. 2010. 2008 IATTC Annual Report. La Jolla, CA. |
| US-27 | Archer, F., T. Gerrodette, S. Chivers, and A. Jackson. 2004. Annual estimates of missing calves in the pantropical spotted dolphin bycatch of the eastern tropical Pacific tuna purse-seine fishery. Fishery Bulletin 102:233-244. |
| US-32 | Starkist, Frequently Asked Questions |
| US-36 | Bumble Bee, Frequently Asked Questions |
| US-37 | Chicken of the Sea, Dolphin Safe Policy |

| EXHIBIT NO. | TITLE |
|-------------|--|
| US-38 | Three companies to stop selling tuna netted with dolphins. N.Y. Times, April 13, 1990. |
| US-39 | Statement of Mr. Francisco Valdez, President of Seafood Emporium, Inc. |
| US-50 | 2010 DML Allocations |
| US-54 | List of vessels |
| US-57 | 50 CFR 216.24 |
| US-58 | 50 CFR 216.93 |
| US-59 | Table of U.S. Dolphin Safe Labeling Conditions |
| US-61 | Gosliner, M.L. 1999. The tuna-dolphin controversy. Pages 120-155 In Twiss and Reeves (eds.) Conservation and management of marine mammals. Smithsonian Institution Press: Washington, D.C. |
| US-62 | Marti L. McCracken and Karin A. Forney, Preliminary Assessment of Incidental Interactions with Marine Mammals in the Hawaii Longline Deep and Shallow Set Fisheries |
| US-66 | 2009 Report on the International Dolphin Conservation Program (2009 AIDCP Report), Document Mop-23-05 (24 September 2010) |
| US-75 | Excerpts of Tuna and Billfishes in the Eastern Pacific Ocean in 2009, IATTC-81-05 |

LIST OF MEXICAN EXHIBITS CITED IN THIS REPORT

| EXHIBIT NO. | TITLE | | | | |
|-------------|---|--|--|--|--|
| MEX-2 | National Research Council, "Dolphins and the Tuna Industry" (National Academy Press: Washington, D.C., 1992) | | | | |
| MEX-3 | Martin A. Hall, "An ecological view of the tuna-dolphin problem: impacts and trade-offs," Reviews in Fish Biology and Fisheries, 8, 1-34 (1998) | | | | |
| MEX-4 | Eric Clua and François Grosvalet, "Mixed-species feeding aggregation of dolphins, large tunas and seabirds in the Azores," Aquatic Living Resources 14 (2001) | | | | |
| MEX-5 | Young and Iudicello, "Worldwide Bycatch Of Cetaceans," U.S. Department of Commerce, National Oceanic and Atmospheric Administration, National Marine Fisheries Service (NOAA Tech. Memo. NMFS-OPR-36) July 2007 | | | | |
| MEX-11 | Agreement on International Dolphin Conservation Program | | | | |
| MEX-21 | International Dolphin Conservation Program Act. Pub. L. No. 105-42, 111 Stat. 1122 (1997) | | | | |
| MEX-22 | 50 C.F.R. §216.24(f) | | | | |
| MEX-29 | Earth Island Inst. v. Evans, 2004 U.S. Dist. LEXIS 15729 (N.D. Cal. 2004) | | | | |
| MEX-30 | Earth Island Inst. v. Hogarth, 484 F.3d 1123 (9th Cir. 2007) | | | | |
| MEX-47(1) | http://www.bumblebee.com/FAQ/#2; http://www.starkist.com/template.asp?section=aboutUs/index.html http://chickenofthesea.com/dolphin_safe.aspx | | | | |
| MEX-50 | http://www.starkist.com/template.asp?section=products/index.html; http://www.bumblebee.com/Products/Family/?Family_ID=1; http://chickenofthesea.com/product_line_list.aspx?FID=3 | | | | |
| MEX-54 | http://www.starkist.com/template.asp?section=products/index.html http://chickenofthesea.com/product_line_list.aspx?FID=3 | | | | |
| MEX-55 | AIDCP Tuna Tracking System resolution Jun 01 | | | | |
| MEX-56 | AIDCP Dolphin Safe certification resolution Jun 01 | | | | |
| MEX-58 | Affidavit (contains BCI) | | | | |
| MEX-64 | Public Opinion Strategies, National Survey Methodology, 9-13 October 2010 | | | | |
| MEX-66 | Government Accountability Office, "National Marine Fisheries Service: Improvements Are Needed in the Federal Process Used to Protect Marine Mammals from Commercial Fishing", GAO 09-78 (December 2008) | | | | |

| EXHIBIT NO. | TITLE | | | | |
|-------------|--|--|--|--|--|
| MEX-67 | Letter from IATTC to Secretary of Commerce enclosing Scientific Report on the Status of Dolphin Stocks in the Eastern Pacific Ocean (October 30, 2002), Scientific Report | | | | |
| MEX-72 | Department of Commerce, "United States Tuna Cannery Receipts January-December 2009 and Comparison" | | | | |
| MEX-84 | NMFS, "U.S. Atlantic and Gulf of Mexico Marine Mammal Stock Assessments - 2009" | | | | |
| MEX-86(A) | Affidavit (contains BCI) | | | | |
| MEX-86(B) | Affidavit (contains BCI) | | | | |
| MEX-86(C) | Affidavit (contains BCI) | | | | |
| MEX-93 | Atlantic Ocean, Caribbean, Gulf of Mexico Large Pelagics Longline Fishery | | | | |
| MEX-97 | ICCAT bycatch species, website: www.iccat.int/en/bycatchspp.htm, accessed 25 November 2010 | | | | |
| MEX-98 | The Western and Central Pacific Tuna Fishery: 2006 Overview and Status of Stocks, Secretariat of the Pacific Community, Oceanic Fisheries Programme, Tuna Fisheries Assessment Report No. 8 | | | | |
| MEX-99 | House of Commons: Environment, Food and Rural Affairs Committee, Caught in the net: by-catch of dolphins and porpoises off the UK coast, 3 rd Report of Session 2003-04 | | | | |
| MEX-105 | WCPFC, Proposed Conservation and Management Measure Mitigating Fishing Impacts on Cetaceans, Seventh Regular Session, Honolulu, Hawaii, USA, 6-10 December 2010 (doc. WCPFC7-2010-DP/17), 15 November 2010 | | | | |
| MEX-109 | Annual Report of the IATTC 1990. Table 5 | | | | |
| MEX-110 | Annual Report of the IATTC 1991. Table 5 | | | | |
| MEX-111 | Pacific Islands Forum Fisheries Agency, Pacific Islands Countries, The Global Tuna Industry and the International Trade Regime – A Guidebook | | | | |

LIST OF AMICUS CURIAE EXHIBITS CITED IN THIS REPORT

| EXHIBIT NO. | TITLE | | | |
|-------------|---|--|--|--|
| 1 | The Tuna Dolphin Issue, Southwest Fisheries Science Center, NOAA Fisheries Service | | | |
| 2 | Collection of News Articles: A Filmmaker Crusades to Make Seas Safe for Gentle Dolphins, PEOPLE MAGAZINE, Vol. 34, No. 5 (August 6, 1990); Sam LaBudde, recipient of the Goldman Environmental Prize, North America 1991; A New Storm Erupts over Saving the Dolphins, The New York Time (Dec. 8, 1990); Epic Debate Led to Heinz Tuna Plan, New York Times (Apr. 16, 1990) | | | |
| 3 | Dolphin Safe Policy: StarKist, Bumblebee and Chicken of the Sea | | | |
| 4 | International Dolphin Conservation Program Act, Hearing before the Subcommittee on Oceans and Fisheries of the Committee on Commerce, Science, and Transportation, United States Senate, S. Hrg. 104-630 at 35-36, 104 th Cong. 2 nd Session (April 30, 1996) (Statement of Senator Barbara Boxer) | | | |
| 28 | Materials from the Earth Island Institute (EII) Website | | | |

I. INTRODUCTION

- 1.1 On 24 October 2008, Mexico requested consultations with the United States pursuant to Article 4 of the Understanding on Rules and Procedures Governing the Settlement of Disputes (DSU), Article XXII of the General Agreement on Tariffs and Trade 1994 (GATT 1994) and Article 14 of the Agreement on Technical Barriers to Trade (TBT Agreement), in relation to certain measures taken by the United States concerning the importation, marketing and sale of tuna and tuna products. Mexico held consultations with the United States on 17 December 2008. Unfortunately, these consultations failed to resolve the dispute.
- 1.2 On 9 March 2009, Mexico requested the establishment of a panel pursuant to Articles 4 and 6 of the DSU, Article XXIII of the GATT 1994 and Article 14 of the TBT Agreement.
- 1.3 At its meeting on 20 April 2009, the DSB established a panel pursuant to the request of Mexico in document WT/DS381/4, in accordance with Article 6 of the DSU.
- 1.4 The Panel's terms of reference are the following:

"To examine, in the light of the relevant provisions of the covered agreements cited by the parties to the dispute, the matter referred to the DSB by Mexico in document WT/DS381/4 and to make such findings as will assist the DSB in making the recommendations or in giving the rulings provided for in those agreements."

- 1.5 On 2 December 2009, Mexico requested the Director-General to determine the composition of the panel, pursuant to paragraph 7 of Article 8 of the DSU.
- 1.6 On 14 December 2009, the Director-General accordingly composed the Panel as follows:

Chairman: Mr Mario Matus

Members: Mr Franz Perrez

Mr Sivakant Tiwari

- 1.7 Argentina, Australia, Brazil, Canada, China, Ecuador, the European Communities¹, Guatemala, Japan, Korea, New Zealand, Chinese Taipei, Thailand, Turkey and Venezuela reserved their rights to participate in the Panel proceedings as third parties.
- 1.8 Following the death of Mr Sivakant Tiwari on 26 July 2010, the parties agreed on a new member of the Panel on 12 August 2010. Accordingly, the panel composition is now as follows:

Chairman: Mr Mario Matus

Members: Ms Elisabeth Chelliah

Mr Franz Perrez

¹ The European Communities reserved its third-party rights at the DSB meeting on 20 April 2009. On 1 December 2009, the Treaty of Lisbon amending the Treaty on European Union and the Treaty establishing the European Community (done at Lisbon, 13 December 2007) entered into force. On 29 November 2009, the WTO received a Verbal Note (WT/L/779) from the Council of the European Union and the Commission of the European Communities stating that, by virtue of the Treaty of Lisbon, as of 1 December 2009, the European Union replaces and succeeds the European Community.

- 1.9 The Panel held its first substantive meeting with the parties on 18, 19 and 20 October 2010. The session with the third parties was held on 19 and 20 October 2010. The second substantive meeting was held on 16 and 17 December 2010.
- 1.10 On 2 February 2011, the Panel issued the descriptive part of its Panel Report to the parties. The Panel issued its interim report to the parties on 5 May 2011. The Panel issued its final report to the parties on 8 July 2011.

II. FACTUAL ASPECTS

A. THE MEASURES AT ISSUE

- 2.1 In its request for the establishment of a panel, Mexico identified the following measures adopted by the United States concerning the importation, marketing and sale of tuna and tuna products:
 - (a) *United States Code*, Title 16, Section 1385 ("Dolphin Protection Consumer Information Act");
 - (b) Code of Federal Regulations, Title 50, Section 216.91 ("Dolphin-safe labeling standards") and Section 216.92 ("Dolphin-safe requirements for tuna harvested in the ETP [Eastern Tropical Pacific Ocean] by large purse seine vessels");
 - (c) The ruling in *Earth Island Institute v. Hogarth*, 494 F.3d 757 (9th Cir. 2007).
- 2.2 This Section describes the main elements of these measures as the Panel understands them, in light of the explanations provided by the parties in their submissions and in response to questions by the Panel. It also briefly describes the "dolphin-safe" scheme established under the Agreement on the International Dolphin Conservation Program (AIDCP).

B. SCOPE OF APPLICATION OF THE DPCIA PROVISIONS

2.3 The Dolphin Protection Consumer Information Act (DPCIA) is codified in Title 16, Section 1385 of the United States Code (USC). Regulations promulgated in accordance with the DPCIA are codified in Title 50, Section 216 of the Federal Regulations. The core of the US "dolphin-safe" labelling scheme is contained in subsection 1385(d)(1)-(3) of the DPCIA. Paragraph (d) of Section 1385 of the DPCIA provisions regulates the use of the term "dolphin-safe" when it appears on tuna products. This provision establishes, in relevant part, the following:

"(1) It is a violation of section 5 of the Federal Trade Commission Act (15 USC 45) for any producer, importer, exporter, distributor, or seller of any tuna product that is exported from or offered for sale in the United States to include on the label of that product the term "dolphin-safe" or any other term or symbol that falsely claims or suggests that the tuna contained in the product were harvested using a method of fishing that is not harmful to dolphins if the product contains tuna harvested--

² Paragraphs (e) through (h) of Section 1835 refer, respectively, to the enforcement of the DPCIA provisions (para. (e)), the issuance of regulations to implement the provisions contained in Section 1835 (para. (f)), the preparation of a study by the Secretary of Commerce on the effect of the fishing technique known as setting on dolphins (intentional deployment on or encirclement of dolphins with purse seine nets) (para. (g)), and the certification requirements for tuna products containing tuna caught in the ETP using a vessel with 363 metric tons or more carrying capacity (para. (h)).

- (A) on the high seas by a vessel engaged in driftnet fishing;
- (B) outside the eastern tropical Pacific Ocean by a vessel using purse seine nets-
 - (i) in a fishery in which the Secretary has determined that a regular and significant association occurs between dolphins and tuna (similar to the association between dolphins and tuna in the eastern tropical Pacific Ocean), unless such product is accompanied by a written statement, executed by the captain of the vessel and an observer participating in a national or international program acceptable to the Secretary, certifying that no purse seine net was intentionally deployed on or used to encircle dolphins during the particular voyage on which the tuna were caught and no dolphins were killed or seriously injured in the sets in which the tuna were caught; or
 - (ii) in any other fishery (other than a fishery described in subparagraph (D)) unless the product is accompanied by a written statement executed by the captain of the vessel certifying that no purse seine net was intentionally deployed on or used to encircle dolphins during the particular voyage on which the tuna was harvested;
- (C) in the eastern tropical Pacific Ocean by a vessel using a purse seine net unless the tuna meet the requirements for being considered dolphin-safe under paragraph (2);

or

- (D) by a vessel in a fishery other than one described in subparagraph (A), (B), or (C) that is identified by the Secretary as having a regular and significant mortality or serious injury of dolphins, unless such product is accompanied by a written statement executed by the captain of the vessel and an observer participating in a national or international program acceptable to the Secretary that no dolphins were killed or seriously injured in the sets or other gear deployments in which the tuna were caught, provided that the Secretary determines that such an observer statement is necessary."³
- 2.4 In turn, paragraph (h) of the DPCIA provisions establishes:
 - "(h) Certification by captain and observer
 - (1) Unless otherwise required by paragraph (2), the certification by the captain under subsection (d)(2)(B)(i) of this section and the certification provided by the observer as specified in subsection (d)(2)(B)(ii) of this section shall be that no dolphins were killed or seriously injured during the sets in which the tuna were caught.
 - (2) The certification by the captain under subsection (d)(2)(B)(i) of this section and the certification provided by the observer as specified under subsection (d)(2)(B)(ii) of this section shall be that no tuna were caught on the trip in which such tuna were harvested using a purse seine net intentionally deployed on or to encircle dolphins, and that no dolphins were killed or seriously injured during the sets in which the tuna were caught, if the tuna were caught on a trip commencing—

³ Mexico's first written submission, Appendix A, Exhibit US-5, para. (d)(1)(A)-(D).

- (A) before the effective date of the initial finding by the Secretary under subsection (g)(1) of this section;
- (B) after the effective date of such initial finding and before the effective date of the finding of the Secretary under subsection (g)(2) of this section, where the initial finding is that the intentional deployment on or encirclement of dolphins is having a significant adverse impact on any depleted dolphin stock; or
- (C) after the effective date of the finding under subsection (g)(2) of this section, where such finding is that the intentional deployment on or encirclement of dolphins is having a significant adverse impact on any such depleted stock."⁴
- 2.5 The DPCIA provisions apply to tuna products. Paragraph (c) of Section 1385 defines the term "tuna product" as "a food item which contains tuna and which has been processed for retail sale, except perishable sandwiches, salads, or other products with a shelf life of less than 3 days".⁵
- 2.6 Subparagraph (d)(1) of Section 1385 of the DPCIA prohibits that tuna products carry the term "dolphin-safe" or "or any other term or symbol that falsely claims or suggests that the tuna contained in the product were harvested using a method of fishing that is not harmful to dolphins", *unless* certain conditions establish by the DPCIA provisions themselves are met.⁶ The conditions that must be fulfilled by tuna products in order to use the term "dolphin-safe" or make any analogous claims vary depending on the manner in which the tuna contained in those products was caught.⁷
- 2.7 The DPCIA provisions refer to *four* criteria to establish *five* basic categories of circumstances in which tuna may be caught.⁸ These criteria are: location (*inside* or *outside* the eastern tropical Pacific Ocean or ETP); fishing gear (*with* or *without* the use of purse seine nets); type of interaction between tuna and dolphins schools (*there is* or *there is no* regular or significant association between tuna and dolphins schools) and the level of dolphin mortalities or injuries (*there is* or *there is no* regular and significant mortality or serious injury). The five categories that result from the combined application of these criteria are described in subparagraphs (A) to (D) of subsection 1385(d)(1) of the DPCIA provisions.
- 2.8 These subparagraphs refer to tuna caught:
 - A) On the high seas by a vessel engaged in *driftnet fishing*;
 - B) Outside the ETP by a vessel using purse seine nets:
 - (i) in a fishery in which the US Secretary of Commerce has determined that there is a regular and significant tuna-dolphin association similar to the association between dolphins and tuna in the ETP;
 - (ii) in any other fishery (other than a fishery described in subparagraph (D)).
 - C) In the ETP by a vessel using purse seine nets; and

⁴ Mexico's first written submission, Appendix A, Exhibit US-5.

⁵ Mexico's first written submission, Appendix A, Exhibit US-5.

⁶ This prohibition is reproduced in Section 216.91(a) of the Federal Code of Regulations.

⁷ Mexico's first written submission, Appendix A, Exhibit US-5.

⁸ Mexico's first written submission, Appendix A, Exhibit US-5.

- D) In a fishery other than the ones described in the previous categories that is identified by the US Secretary of Commerce as having a regular and significant mortality or serious injury of dolphins."
- C. CONDITIONS FOR THE USE OF THE TERM "DOLPHIN-SAFE" OR ANALOGOUS TERMS
- Subsections 1385(d)(1)(2) and (h) of the DPCIA establish specific conditions for the use of the term "dolphin-safe" or any analogous claims for each one of the categories described in subparagraphs (B) through (D) of subsection 1385(d)(1) on tuna products. Tuna products containing tuna caught under the scenario described in subparagraph (A) of subsection 1385(d)(1), i.e. tuna caught on the high seas using driftnet fishing, may under no circumstances be labelled as "dolphin-safe" or display any analogous claims. The documentary evidence required under the DPCIA for the categories (B) through (D) is described below.
- 2.10 With respect to tuna products containing tuna caught outside the ETP by a vessel using purse seine nets in a fishery in which the US Secretary of Commerce has determined that a regular and significant dolphin-tuna association exists (subparagraph (B)(i)), the use of the term "dolphin-safe" or any analogous term is conditional upon a written statement executed by the captain of the vessel and an observer participating in a national or international program acceptable to the Secretary of Commerce, certifying that no purse seine net was intentionally deployed on or used to encircle dolphins during the particular voyage on which the tuna were caught and no dolphins were killed or seriously injured in the sets in which the tuna were caught.
- 2.11 For tuna caught outside the ETP by a vessel using purse seine nets in any fishery, other than a fishery described in subparagraph (D) of subsection 1385(d)(1) of the DPCIA provisions, (subparagraph (B)(ii)), a written statement executed by the captain of the vessel is required, certifying that *no purse seine net was intentionally* deployed *on or used to encircle dolphins* during the particular voyage on which the tuna was harvested.
- 2.12 For tuna harvested *in the ETP* by a vessel using purse seine nets (subparagraph (C)), the conditions are:
 - written statements executed by the captain and an observer approved by the International
 Dolphin Conservation Program (IDCP) certifying that no dolphins were killed or
 seriously injured during the sets in which the tuna were caught; and, if there is a previous
 determination by the Secretary of Commerce that the fishing technique of setting on
 dolphins is having a significant adverse impact on any depleted dolphin stock in the ETP,
 these statements must also certify that no purse seine net was intentionally deployed on or
 used to encircle dolphins;
 - a written statement executed by either the Secretary of Commerce or the Secretary's designee, or a representative of the Inter-American Tropical Tuna Commission (IATTC), or an authorized representative of a participating nation whose national program meets the requirements of the International Dolphin Conservation Program (IDCP) stating that there was an observer approved by the IDCP on board the vessel during the entire trip and that such observer provided the corresponding certification;
 - the written endorsement by each exporter, importer, and processor of the tuna; and

⁹ Mexico's first written submission, Appendix A, Exhibit US-5.

• the above mentioned written statements and endorsements comply with regulations promulgated by the Secretary which provide for the verification of tuna products as dolphin-safe.

Vessels smaller than 363 metric tons carrying capacity fishing in the ETP using purse seine nets are not subjected to these requirements. ¹⁰ Therefore, tuna caught in the ETP by this type of vessels may be labelled "dolphin-safe" without the need to submit any documentary evidence.

- 2.13 Finally, for tuna caught in a fishery *other* than those described in subsection 1385(d)(1)(A)-(C) that is identified by the US Secretary of Commerce as having a *regular* and significant mortality or serious injury of dolphins (subparagraph (D)), the use of the term "dolphin-safe" or any analogous terms is subject to a written statement executed by the *captain* of the vessel and an *observer* participating in a national or international program acceptable to the Secretary of Commerce that *no dolphins were killed or seriously injured* in the sets or other gear deployments in which the tuna were caught, provided that the Secretary of Commerce determines that such an observer statement is necessary.
- 2.14 In response to a question by the Panel, the United States provided the table below¹¹, outlining the specific requirements for access to a "dolphin-safe" label, distinguishing between tuna caught inside and outside the ETP and between different fishing methods.¹²

¹⁰ United States' first written submission, para. 16. Indeed, Section 216.24(a)(2)(i) of Title 50 of the US Code of Federal Regulations establishes:

⁽²⁾⁽¹⁾ It is unlawful for any person using a U.S. purse seine fishing vessel of 400 short tons (st) (362.8 metric tons (mt) carrying capacity or less to intentionally deploy a net on or to encircle dolphins, or to carry more than two speedboats, if any part of its fishing trip is in the ETP.

¹¹ Exhibit US-59.

¹² Question 9 from the Panel.

US Dolphin Safe Labelling Conditions

This table assumes the vessel is not on the high seas engaged in driftnet fishing for tuna. Tuna products containing tuna harvested in this manner cannot be labelled dolphin-safe under any circumstance. In addition to the other requirements reflected in this table, a condition for carrying an alternative dolphin-safe label applicable to all tuna product, regardless of location or fishery/vessel type, is that no dolphins were killed or seriously injured in the sets or other gear deployments in which the tuna were caught. In the sets or other gear deployments in which the tuna were caught.

| Location | Fishery/Vessel type | Tuna-Dolphin (T-D) association/Harm to dolphins | Dolphin Safe Labeling Condition | Authority |
|----------------|----------------------------------|---|---|--|
| Inside the ETP | Large ⁱⁱⁱ purse seine | [ETP has a T-D association] | Written statements executed by the captain and observer providing the certification [required under § 1385 (h)], and endorsed in writing by the exporter, importer, and processor of the product, that - no tuna were caught on the trip in which such tuna were harvested using a purse seine net intentionally deployed on or to encircle dolphins, and that - no dolphins were killed or seriously injured during the sets in which the tuna were caught Written statement by Secretary or designee/IATTC representative/ authorized representative of nation whose nation program meets the requirements of the International Dolphin Conservation Program that - an IDCP-approved observer was on board the vessel during the entire trip and provided the certification required under § 1385 (h) above | § 1385 (d)(1)(C); § 1385 (d)(2)(B); § 1385 (h)(2)(A) |
| | Small purse seine ^{iv} | [ETP has a T-D association] | None | § 1385 (d)(1)(C); § 1385 (d)(2)(A) |
| | Non purse seine | Regular and significant mortality or serious injury to dolphins | Written statement executed by the captain of the vessel and an observer participating in a national or international program acceptable to the Secretary that - no dolphins were killed or seriously injured in the sets or other gear deployments in which the tuna were caught, provided that the Secretary determines that such an observer statement is necessary | § 1385 (d)(1)(D) |
| | | No regular and significant mortality or serious injury to dolphins | None | |

| Location | Fishery/Vessel type | Tuna-Dolphin (T-D) association/Harm to dolphins | Dolphin Safe Labelling Condition | Authority |
|-----------------|------------------------|---|---|----------------------|
| Outside the ETP | Purse seine | Regular and significant T-D association similar to association in the ETP | Written statement executed by the captain of the vessel and an observer participating in a national or international program acceptable to the Secretary, certifying that - no purse seine net was intentionally deployed on or used to encircle dolphins during the particular voyage on which the tuna were caught and - no dolphins were killed or seriously injured in the sets in which the tuna were caught | § 1385 (d)(1)(B)(i) |
| | | No regular and significant T-D association | Written statement executed by the captain of the vessel certifying that - no purse seine net was intentionally deployed on or used to encircle dolphins during the particular voyage on which the tuna was harvested | § 1385 (d)(1)(B)(ii) |
| | Non purse seine | Regular and significant mortality or serious injury to dolphins | Written statement executed by the captain of the vessel and an observer participating in a national or international program acceptable to the Secretary (provided that the Secretary determines that such an observer statement is necessary) that — no dolphins were killed or seriously injured in the sets or other gear deployments in which the tuna were caught | § 1385 (d)(1)(D) |
| | | No regular and significant mortality or serious injury to dolphins | None | |

^{§ 1385 (}d)(1)(A). § 1385 (d)(3)(C)(i).

Vessel greater than 400 short tons (362.8 metric tons) carrying capacity. US Code of Federal Regulations §216.91(a)(1).

These vessels are 400 short tons (362.8 metric tons) or less carrying capacity and are not permitted to set on dolphins to catch tuna. AIDCP, Annex VIII, Exhibit Mex-11; 50 CFR 216.24(a)(2)(i), Exhibit US-23B; see also United States' first written submission, para. 16, fn.10.

D. THE CERTIFICATION REQUIRED FOR TUNA CAUGHT IN THE ETP

- 2.15 As described above, subparagraph (h)(1) of the DPCIA provisions establishes that, unless otherwise required by paragraph (2), tuna harvested in the ETP by a vessel using purse seine nets may be labelled "dolphin-safe" if the captain of the vessel and an approved observer certify that no dolphins were killed or seriously injured during the sets in which the tuna were caught.
- 2.16 However, subparagraph (h)(2) establishes that tuna harvested in the ETP by a large vessel using purse seine nets may be labelled "dolphin-safe" if the captain of the vessel and an approved observer certify that (i) no purse-seine net were intentionally deployed on or used to encircle dolphins during the trip in which the tuna was caught, and (ii) no dolphins were killed or seriously injured during the sets in which the tuna were caught. Subparagraph (h)(2) of the DPCIA provisions conditions the applicability of subparagraph (h)(1) to the existence of a finding by the US Secretary of Commerce that the intentional deployment on or encirclement of dolphins with purse seine nets is *not* having a significant adverse impact on any depleted dolphin stock in the ETP. Subparagraph (g) required the Secretary of Commerce to conduct this task in two stages resulting in an initial and a final finding on the impact of setting on dolphins in the ETP.
- 2.17 On 7 May 1999, the Secretary of Commerce published its initial finding pursuant to the DPCIA provisions. This initial finding concluded that "there [was] insufficient information to conclude that there has been a significant adverse impact on the depleted [dolphin] stocks" in the ETP and that "there [was] no solid evidence in any of the scientific studies to date that links the apparent failure of dolphin stocks to recover at the rate expected based on historical data to the current tuna purse seine fishery practices". 14
- 2.18 On 31 December 2002, the Secretary of Commerce made its final finding that "the intentional deployment on or encirclement of dolphins with purse seine nets [was] not having a significant adverse effect on any depleted dolphin stock in the ETP". As a result of this finding, subparagraph (h)(1) of the DPCIA provisions became momentarily applicable, having the effect of limiting the certification required for tuna caught in the ETP with large vessels using purse seine nets, to the corroboration that no dolphins were killed or seriously injured during the sets in which the tuna were caught, cancelling the requirement contained in subparagraph (h)(2) that the certification of tuna

¹³ Paragraph (g) of the DPCIA provisions establish the characteristics of these findings as follows:

⁽¹⁾ Between March 1, 1999, and March 31, 1999, the Secretary shall, on the basis of the research conducted before March 1, 1999, under section 1414a (a) of this title, information obtained under the International Dolphin Conservation Program, and any other relevant information, make an initial finding regarding whether the intentional deployment on or encirclement of dolphins with purse seine nets is having a significant adverse impact on any depleted dolphin stock in the eastern tropical Pacific Ocean. The initial finding shall be published immediately in the Federal Register and shall become effective upon a subsequent date determined by the Secretary.

⁽²⁾ Between July 1, 2001, and December 31, 2002, the Secretary shall, on the basis of the completed study conducted under section 1414a (a) of this title, information obtained under the International Dolphin Conservation Program, and any other relevant information, make a finding regarding whether the intentional deployment on or encirclement of dolphins with purse seine nets is having a significant adverse impact on any depleted dolphin stock in the eastern tropical Pacific Ocean. The finding shall be published immediately in the Federal Register and shall become effective upon a subsequent date determined by the Secretary.

¹⁴ US Department of Commerce, National Oceanic and Atmospheric Administration, "Taking of Marine Mammals Incidental to Commercial Fishing Operations; Tuna Purse Seine Vessels in the Eastern Tropical Pacific Ocean (ETP), Initial Finding", Federal Register Vol. 64, No. 88, May 7, 1999, Exhibit MEX-26.

¹⁵ Exhibit MEX-29, p. 4.

caught in the ETP by a large purse seine vessel include a statement that no purse seine net was intentionally deployed on or used to encircle dolphins.¹⁶

- 2.19 However, as a result of legal action by Earth Island Institute, a non-governmental organization, alleging that the Secretary of Commerce had abused its powers in reaching the conclusions reflected in the final findings, the US Court of Appeals for the Ninth Circuit vacated these findings on 9 August 2004, declaring them "arbitrary, capricious, and abuse of discretion and contrary to law". The consequence of this judicial decision, in the view of the United States, is that "the findings necessary for the subsection 1385(h)(1) certification to apply do not exist, and therefore the applicable certification for tuna caught using purse seine nets in the ETP remains the one set out in subsection 1385(h)(2)" of the DPCIA provisions, including the requirement that no purse seine net was intentionally deployed on or used to encircle dolphins. Mexico explained that "the U.S. law does not permit revision of the DOC's justification for its findings nor re-consideration by the courts of whether the change in the labeling standard can be implemented". 19
- 2.20 Hence, under the DPCIA provisions that are currently applicable, tuna harvested in the ETP by a large vessel using purse-seine nets may be labelled dolphin-safe if the captain and an observer approved by the IDCP certify that *no dolphins were killed or seriously injured* during the sets in which the tuna were caught and that *no purse seine net was intentionally deployed on or used to encircle dolphins* during the same fishing trip. This certification must be accompanied by a written statement executed by the Secretary of Commerce (or designee), a representative of the Inter-American Tropical Tuna Commission or an authorized representative of a participating nation whose national program meets the requirements of the IDCP, and the endorsement by the exporters, importers and processors required in subparagraphs (d)(2)(B)-(C) of Section 1385 of the DPCIA provisions.²⁰
- E. THE CERTIFICATION REQUIRED FOR TUNA CAUGHT OUTSIDE THE ETP
- 2.21 As explained above, subparagraphs (d)(1)(B) and (D) of the DPCIA provisions establish different categories of tuna harvested *outside* the ETP. These categories are:
 - Tuna caught using purse seine nets in a fishery in which the US Secretary of Commerce has determined that a *regular and significant tuna-dolphin association* exists similar to the association in the ETP (§1385 (d)(1)(B)(i));
 - Tuna caught using purse seine nets in a fishery in which there is no regular and significant association between tuna and dolphins according to the US Secretary of Commerce (§1385 (d)(1)(B)(ii)); and

¹⁶ Subparagraph (h)(2) provided that, where the initial finding is that the intentional deployment on or encirclement of dolphins is having a significant adverse impact on any depleted dolphin stock, or after the effective date of such finding, the certifications required shall be that no tuna were caught on the trip in which such tuna were harvested using a purse seine net intentionally deployed on or to encircle dolphins, and that no dolphins were killed or seriously injured during the sets in which the tuna were caught, if the tuna were caught on a trip commencing before the effective date of the initial finding by the Secretary, after the effective date of such initial finding and before the effective date of the finding of the Secretary.

¹⁷ Exhibit MEX-29, p. 24.

¹⁸ United States' first written submission, para. 22.

¹⁹ Mexico's first written submission, para. 88.

²⁰ Mexico's first written submission, Appendix A, Exhibit US-5.

- Tuna caught in a fishery *other* than the ones described in subparagraphs (d)(1)(A)-(C) that is identified by the US Secretary of Commerce as having a *regular and significant mortality* or *serious injury* of dolphins (§1385 (d)(1)(C)).²¹
- 2.22 As also mentioned above, the DPCIA provisions establish a specific set of conditions that must be fulfilled by each one of these categories of tuna in order to use the term "dolphin-safe" or to make similar claims. In two of these instances (i.e. tuna caught in a fishery in which the US Secretary of Commerce has *determined* that a regular and significant tuna-dolphin association exists, *and* in the case of tuna caught in a fishery other than the ones described in subparagraphs (d)(1)(A)-(C) that is *identified* by the US Secretary of Commerce as having a regular and significant mortality or serious injury of dolphins), the applicability of the relevant requirements is conditioned on the existence of a determination by the Secretary of Commerce that in the fishery in question there is regular and significant tuna-dolphin association similar to the association in the ETP, or regular and significant mortality or serious injury of dolphins.²²
- 2.23 The United States has indicated that no fishery outside the ETP has been determined to have a regular and significant association between tuna and dolphins similar to the association in the ETP.²³ Moreover, the United States has also explained that it has not made a determination that any non-purse seine tuna fishery has regular and significant dolphin mortality.²⁴
- 2.24 Therefore, although it remains a possibility, under the DPCIA provisions, that the Secretary of Commerce may determine that there is regular and significant dolphin-tuna association, or regular and significant mortality or serious injury of dolphins in fisheries outside the ETP, such determinations have not been made until to date. Hence the "dolphin-safe" requirements for tuna caught under the circumstances described in subparagraphs (d)(1)(B)(i) and (d)(1)(D) of Section 1385 are not currently applied with respect to any fishery.
- 2.25 Consequently, the scenarios described in subparagraphs (d)(1) of Section 1385 that are currently applicable are those described in:
 - (a) subparagraph (d)(1)(A), which refers to tuna caught on the high seas by driftnet fishing;
 - (b) subparagraph (d)(1)(C), which refers to tuna caught in the ETP by a large vessel using purse seine nets; and
 - (c) subparagraph (d)(1)(B)(ii), which refers to tuna caught outside the ETP in a fishery that has not been the subject of a determination of regular and significant dolphintuna association or of regular and significant dolphin mortality or serious injury to dolphins by the Secretary of Commerce. The "dolphin-safe" certification required for this type of tuna must be provided by the *captain* of the vessel and, according to subparagraph (d)(1)(B)(ii), must only state that *no purse seine net was intentionally deployed on or used to encircle dolphins* during the particular voyage on which the tuna was harvested.²⁵

²¹ Mexico's first written submission, Appendix A, Exhibit US-5.

²² Mexico's first written submission, Appendix A, Exhibit US-5.

²³ United States' first written submission paras. 38-39, see also United States' response to question 12 from the Panel.

²⁴ United States' response to question 85 from the Panel.

²⁵ Mexico's first written submission, Appendix A, Exhibit US-5.

- 2.26 As further explained in the next section, the parties disagree on whether a certification that no dolphins were killed or seriously injured is required for the use of non-official labels or marks (alternative labels) if they are carried by products containing tuna caught outside the ETP in a fishery with no regular and significant dolphin-tuna association and no regular and significant dolphin mortality or serious injury of dolphins.
- F. OFFICIAL AND ALTERNATIVE "DOLPHIN-SAFE" MARKS AND LABELS
- 2.27 Subparagraph (d)(3) of the DPCIA provisions establishes the following:
 - "(3) (A) The Secretary of Commerce shall develop an official mark that may be used to label tuna products as dolphin-safe in accordance with this Act.
 - (B) A tuna product that bears the dolphin-safe mark developed under subparagraph (A) shall not bear any other label or mark that refers to dolphins, porpoises, or marine mammals.
 - (C) It is a violation of section 45 of title 15 to label a tuna product with any label or mark that refers to dolphins, porpoises, or marine mammals other than the mark developed under subparagraph (A) unless—
 - (i) no dolphins were killed or seriously injured in the sets or other gear deployments in which the tuna were caught;
 - (ii) the label is supported by a tracking and verification program which is comparable in effectiveness to the program established under subsection (f) of this section; and
 - (iii) the label complies with all applicable labeling, marketing, and advertising laws and regulations of the Federal Trade Commission, including any guidelines for environmental labeling."²⁶
- 2.28 Under these provisions, producers wishing to display "dolphin-safe" claims on their tuna products may thus choose between the official "dolphin-safe" mark developed by the Secretary of Commerce under subparagraph (A) or an alternative mark or label of their own design under subparagraph (C). If producers opt to use the official mark, their products may not bear any other label or mark that refers to dolphins, porpoises, or marine mammals.
- 2.29 Moreover, subparagraphs (C)(i)-(iii), establish specific requirements for the use of alternative "dolphin-safe" labels or marks. According to these subparagraphs, tuna products may carry a label or mark that refers to dolphins, porpoises, or marine mammals other than the official mark only if the specific conditions described in subparagraphs (i) to (iii) are met.
- 2.30 The text of subparagraph (d)(3)(C) does not specify whether these requirements for the use of alternative labels are *additional* to the requirements for the use of the official mark as set forth in subparagraph (d)(1). In response to a question by the Panel, the United States asserted that tuna products carrying an alternative label must comply with the specific requirements for the use of alternative labels *in addition* to those set forth in subparagraph (d)(1), which are applicable to the use

²⁶ Mexico's first written submission, Appendix A, Exhibit US-5.

of the official mark or any alternative label.²⁷ The United States has thus indicated that "section 1385(d)(3) provides that if tuna products are labeled with an alternative dolphin-safe label, those tuna products may not contain tuna that was caught in a set in which dolphins were killed or seriously injured (and this conditions [sic] applies regardless of whether the tuna was caught inside or outside the ETP)". ²⁸ Mexico disputed this description of the law, arguing that the United States in practice does not require that alternative labels be supported with a certification that dolphins were not killed or seriously injured for any fishery outside the ETP, that the United States has no tracking or verification program for this certification for tuna harvested outside the ETP, and that the major producers do not claim that their dolphin-safe labels mean anything other than that purse seine nets were not set on dolphins during the voyage in which the tuna were caught.²⁹

- G. THE "DOLPHIN-SAFE" TRACKING AND VERIFICATION PROGRAM AND OTHER ENFORCEMENT PROVISIONS
- 2.31 The US National Marine Fisheries Service (NMFS) has established the Tuna Tracking and Verification Program (TTVP), for tracking and verifying the "dolphin-safe" or "non-dolphin-safe" condition of tuna caught in the ETP. The provisions establishing this program are mainly contained in Title 50, Sections 216.24³¹ and 216.91-216.93 of the US Code of Federal Regulations. Through the use of the TTVP, the US government collects information from domestic tuna processors, US tuna vessels, and importers of tuna products, to verify whether tuna products labelled dolphin-safe meet the statutory conditions. The provisions are mainly contained in Title 50, Sections 216.24³¹ and 216.91-216.93 of the US Code of Federal Regulations.
- 2.32 As explained by the United States and Mexico, every import of every tuna product, regardless of whether the "dolphin-safe" label is intended to be used, must be accompanied by a Fisheries Certificate of Origin (NOAA Form 370). One copy of this form must be submitted to Customs and Border Protection at the time of importation, and a second one to the TTVP. Moreover, US tuna processors must submit monthly reports to the TTVP containing the dolphin-safe status, ocean area of capture, catcher vessel, trip dates, carrier name, unloading dates, and location of unloading of tuna for both imported and domestic receipts. In addition, as part of the TTVP, the NMFS conducts periodic cannery audits and "spot checks" of retail market product.³⁴
- 2.33 According to the United States, if a product is found to be wrongfully labelled during a spot check, the product will most likely be seized as evidence. Later on the US authorities may decide to forfeit, destroy or in the case of imports, have the product re-exported, depending on the facts and circumstances of the case. Moreover, sanctions for offering for sale or export tuna products falsely

³¹ Section 216.24 is not specifically identified by Mexico in its request for establishment of a panel. Paragraph (f) of Section 1385 of the DPCIA, which is challenged by Mexico, instructs the Secretary of Commerce to issue regulations to implement the Dolphin Protection Consumer Information Act, "including regulations to establish a domestic tracking and verification program that provides for the effective tracking of tuna labeled under subsection (d) of [Section 1385]. (Mexico's first written submission, Appendix A, Exhibit US-5.)

²⁷ See United States' response to question No. 8 from the Panel where the United States stated that "Section 1385(d)(1)-(2) applies to all tuna products regardless of whether those tuna products are labelled with the official dolphin-safe label or an alternative dolphin-safe label" and that "[i]n addition, section 1385(d)(3)(C) sets out *additional* conditions for use of an alternative dolphin-safe mark" (emphasis added). See also United States' response to question No. 11 from the Panel, paras. 24-25.

²⁸ See United States' second written submission, paras. 40-41.

²⁹ See Mexico's second written submission, paras. 14-26.

³⁰ Exhibit MEX-73.

³² Mexico's first written submission, Appendix B, Exhibits MEX-22, US-57 and US-58.

³³ United States' response to Panel question No. 4, para. 8.

³⁴ United States' response to Panel question No. 4, paras. 9-10.

labelled "dolphin-safe" may be assessed against any producer, importer, exporter, distributor or seller who is subject to the jurisdiction of the United States.³⁵ Violators may also be prosecuted directly under the DPCIA provisions or under federal provisions establishing false statement or smuggling prohibitions or federal labelling standards.³⁶

H. THE "DOLPHIN SAFE" SCHEME ESTABLISHED UNDER THE AIDCP

- 2.34 The AIDCP establishes a "dolphin-safe" scheme that is separate from the US scheme.
- 2.35 The Inter-American Tropical Tuna Commission (IATTC) began in 1976 multilateral endeavours that led to the creation of the International Dolphin Conservation Program (IDCP).³⁷ These efforts were later reflected in a series of multilateral agreements that were negotiated in response to the evidence that many dolphins were dying in the ETP each year. These agreements were the La Jolla Agreement (1992), the Panama Declaration (1995) and the AIDCP (1999). Both Mexico and the United States are signatories to the La Jolla Agreement and the Panama Declaration and parties to the AIDCP.³⁸
- 2.36 In the Panama Declaration, signatories reaffirm the commitments and objectives of the La Jolla Agreement and state their intention to conclude a binding international agreement on dolphin conservation in the ETP. The preamble to the Panama Declaration thus provides the following:

"Recognizing the strong commitments of nations participating in the La Jolla Agreement and the substantial successes realized through multilateral cooperation and supporting national action under that Agreement, the Governments meeting in Panama (...) announce their intention to formalize by January 31, 1996, the La Jolla Agreement as a binding legal instrument (...). This shall be accomplished by adoption of a binding resolution of the IATTC or other legally binding instrument. The adoption of the IATTC resolution or other legally binding instrument, that utilizes to the maximum extent possible the existing structure of the IATTC, is contingent upon the enactment of certain changes in US law, as envisioned in Annex 1 to this Declaration." ³⁹

2.37 Annex 1 of the Declaration lists certain "envisioned changes in United States law", including the following in relation to labelling:

"the term "dolphin-safe" may not be used for any dolphin caught in the EPO by a purse seine vessel in a set in which a dolphin mortality occurred as documented by observers by weight calculation and well location."

2.38 Mexico argued that the United States, as part of its international obligations under the Panama Declaration was committed to change its "dolphin-safe" definition from "no encirclement of dolphins"

³⁵ United States' response to Panel question No. 4, para. 10. For instance, subsection 1385(e) of the DPCIA provisions establishes:

[&]quot;Any person who knowingly and willfully makes a statement or endorsement described in subsection (d)(2)(B) [referring to the statements and endorsement required in the case of tuna caught in the ETP] of this section that is false is liable for a civil penalty of not to exceed \$100,000 assessed in an action brought in any appropriate district court of the United States on behalf of the Secretary."

³⁶ United States' response to Panel question No. 50, para. 120.

³⁷ Mexico's first written submission, p. 2, para. 7; United States' first written submission, para. 76-80.

³⁸ Mexico's first written submission, pp 15-17, paras. 49-50

³⁹ Declaration of Panama, Exhibit MEX-20, p. 1.

⁴⁰ Declaration of Panama, Annex 1, Exhibit MEX-20, p. 5.

to "no dolphin mortality or serious injury"⁴¹. The United States disagreed with this characterization, stating that the Panama Declaration does not commit the United States to amend its law or to take any other action. In the US view, the Panama Declaration only stated the signatories' intent to adopt a binding dolphin conservation agreement if certain changes were made to US law. 42

- In 1998, signatories to the Panama Declaration concluded negotiation of the AIDCP, which entered into force in 1999. Through a comprehensive program of monitoring, tracking, verification, and certification, featuring the use of independent observers on board tuna fishing vessels, the AIDCP has maintained and reinforced the progress made after the adoption of the La Jolla Agreement and the Panama Declaration, and dramatically reduced observed dolphin mortality in the ETP. 43 The AIDCP is recognized to have made an important contribution to dolphin protection in the ETP. 44
- On 20 June 2001, the Parties to the AIDCP adopted the "Resolution to Adopt the Modified System for Tracking and Verification of Tuna". That resolution includes the following definitions:

"The terms used in this document are defined as follows:

- Dolphin safe tuna is tuna captured in sets in which there is no mortality or serious injury of dolphins;
- Non-dolphin-safe tuna is tuna captured in sets in which mortality or serious injury of dolphins occurs ..."45
- On that same date (20 June 2001), the Parties to the AIDCP also adopted the "Resolution to Establish Procedures for AIDCP Dolphin Safe Tuna Certification". That document states that "[t]he terms used in this document are as defined in the AIDCP System for Tracking and Verification of Tuna". Accordingly the AIDCP Tuna Tracking and Verification Resolution's definitions of "dolphin-safe" and "non-dolphin safe" apply with respect to the AIDCP resolution's rules on dolphin-safe certification. The resolution established the procedures that would enable tuna caught and tracked in accordance with the procedures to receive an "AIDCP dolphin-safe certification". 46 The AIDCP Dolphin Safe Certification Resolution states that application of the procedures "shall be voluntary for each Party, especially in the event that they may be inconsistent with the national laws of a Party".47

PARTIES' REQUESTS FOR FINDINGS AND RECOMMENDATIONS III.

Mexico requests the Panel to find that the US measures are inconsistent with Articles I:1 3.1 and III:4 of the GATT 1994 and Articles 2.1, 2.2 and 2.4 of the TBT Agreement.⁴⁸

⁴¹ Mexico's response to Panel question No. 2, para. 7

⁴² United States' first written submission, para. 77. The United States amended the DPCIA in 1997. This amendment provided for a change in the dolphin-safe standard, contingent on the outcome of certain scientific studies performed by the Department of Commerce. The results of these studies were challenged in US courts, and an opinion was issued by the Court of Appeals for the Ninth Circuit in Earth Island v. Hogarth.

Mexico's first written submission, pp 17-19, paras. 49-50.

⁴⁴ Mexico's first written submission, para. 7; United States first written submission, p. 3, para. 9; p. 74, para. 52.

Mexico's first written submission, para. 229 and Exhibit MEX-55.

⁴⁶ See Exhibit MEX-56, United States' first written submission, para. 83 and United States' second written submission, para. 184.

⁴⁷ AIDCP Dolphin Safe Certification Resolution,. MEX-56.

⁴⁸ Mexico's first written submission, para. 263.

3.2 The United States requests the Panel to reject Mexico's claims that the US dolphin-safe labelling provisions are inconsistent with Articles I:1 and III:4 of the GATT 1994 and Articles 2.1, 2.2 and 2.4 of the TBT Agreement.⁴⁹

IV. ARGUMENTS OF THE PARTIES⁵⁰

A. FIRST WRITTEN SUBMISSIONS

1. Mexico

- (a) Introduction
- 4.1 For over twenty years, yellowfin tuna caught by the Mexican fishing fleet in the Eastern Tropical Pacific ("ETP") have been denied effective access to the US market by virtue of various GATT and WTO inconsistent measures. Sales of Mexican yellowfin tuna in the US market have been severely restricted. Initially, the exclusion took the form of an absolute embargo on the importation of Mexican tuna and tuna products. Notwithstanding that the embargo was lifted, the United States found a new way to prevent Mexican tuna from competing in the US market.
- 4.2 The essence of the current dispute relates to the prohibition of the use of a US dolphin-safe label on imports of tuna products from Mexico, while such a label is permitted to be used on tuna products from other countries, including the United States.
- 4.3 Even though Mexico has maintained a sound and environmentally sustainable method for fishing for tuna and participated in all multilateral initiatives to protect dolphins while fishing for tuna, Mexican tuna products are prohibited by the US measures from using a dolphin-safe label, while tuna caught in other fisheries that have not adopted comparable measures to protect dolphins are able to benefit from a dolphin-safe label.
- 4.4 The US measures are inconsistent with the fundamental obligations of non-discrimination contained in Articles I:1 and III:4 of the GATT, because they grant other foreign tuna and tuna products an advantage in the US marketplace that has not been accorded immediately and unconditionally to Mexican tuna and tuna products, and accord Mexican tuna and tuna products treatment less favourable than that accorded to US tuna and tuna products in the US marketplace.
- 4.5 Furthermore, the US measures are inconsistent with Articles 2.1, 2.2, and 2.4 of the Agreement on Technical Barriers to Trade ("TBT Agreement") because they create unnecessary obstacles to international trade, are not based on the relevant international standard, discriminate against Mexican tuna products and tuna, and address objectives that can be addressed in a less traderestrictive manner.

⁴⁹ United States' first written submission, para. 80.

⁵⁰ This section on the arguments of the parties is based on the executive summaries submitted by the parties to the Panel. The parties also responded to questions from the Panel. Those questions and answers are not reflected in this section.

- (b) Summary of facts and background information
- (i) Tuna fishing

Tuna and dolphins

4.6 In the ETP, there is a natural association between tuna and dolphins. This means that schools of tuna tend to aggregate and swim – to "associate" – near certain species of dolphins in ocean waters. However, the association between tuna and dolphins is not unique to the ETP. Scientific research indicates that there are associations of tuna with dolphins in other oceans of the world, including the Eastern and Western Atlantic, the Indian Ocean, the Central and Western Pacific, and the Gulf of Mexico.

Tuna fishing and dolphins mortality in the ETP

- 4.7 Because the association between dolphins and tuna has long been observed in the ETP, fishermen locate schools of underwater tuna by finding and chasing dolphins on the ocean's surface and intentionally encircling them with purse seine nets to harvest the tuna underneath. In the early years of fishing by setting on dolphins there was considerable incidental dolphin mortality.
- 4.8 When the United States enacted legislation imposing restrictions on fishing in the ETP, the US vessels soon began to fish for tuna elsewhere, outside the ETP. In the case of Mexico, there was no reason for the Mexican fleet to relocate outside its natural and traditional fishing area within the ETP. However, the problem was not ignored by the Mexican fleet. Eventually, multilateral action was taken by the United States, Mexico, and other countries in the region to reduce overall dolphin mortality by establishing an international dolphin conservation program.
- 4.9 There is scientific evidence proving that outside the ETP, fishing results in the killing of many dolphins and other cetaceans (i.e. bycatch), and no measures have been taken in those other regions even remotely comparable to those taken for the ETP. Also, despite the fact that dolphins (and other several species) are affected from fishery bycatch in other oceans, the United States has chosen to regulate fishing methods to protect dolphins exclusively with respect to tuna caught in the ETP.

Fishing methods

- 4.10 The most popular fishing method alternative to the method of setting purse-seine nets around dolphins is "Fish Aggregating Devices", or "FADs". This method consists in casting shadows that attract tuna and other fish species. The FAD sets attract and kill more immature tuna as well as a variety of other bycatch, including sea turtles, sharks and other threatened and endangered species.
- 4.11 In contrast, because only mature yellowfin tuna are able to swim fast enough to "associate" with dolphins, the method of fishing by setting on dolphins produces a large catch of mature tuna appealing to the marketplace and little by-catch.
- 4.12 Furthermore, when using the dolphin set method, it is possible to virtually eliminate dolphin mortalities if certain procedures recommended or required by the AIDCP are observed. Therefore, the dolphin set method, when administered under the AIDCP, is the most environmentally sound method for harvesting tuna.

(ii) Efforts to protect dolphins

Multilateral efforts

- 4.13 The international effort to provide for appropriate conservation of stocks of tuna, dolphins and other species in the fisheries of the ETP had its origins in the Convention for the Establishment of an Inter-American Tropical Tuna Commission ("IATTC"). In 1976, the IATTC broadened its responsibilities to include the treatment of problems arising from the tuna-dolphin relationship in the ETP, which led to the creation of a program that would later be known as the International Dolphin Conservation Program.
- 4.14 The multilateral endeavours of the IATTC were later encompassed in a series of multilateral agreements that were negotiated in the wake of the "tuna-dolphin" controversy in the ETP, namely, the La Jolla Agreement (1992), the Panama Declaration (1995), and the Agreement on the International Dolphin Conservation Program (1998). Both Mexico and the United States are parties to all of these agreements.
- 4.15 The latter of them, the Agreement on the International Dolphin Conservation Program ("AIDCP") comprises the most recent and comprehensive program to protect dolphins while fishing for tuna. The AIDCP, which has been awarded by the United Nations Food and Agriculture Organization, is an unqualified success and has enormously reduced dolphin mortality to statistically insignificant levels while promoting sustainable fishing practices. It was put into force on 15 February 1999.
- 4.16 Through this comprehensive program of monitoring, tracking, verification, and certification with its emphasis on having independent observers on board tuna fishing vessels and former international programs, the participating countries have succeeded in dramatically reducing incidental dolphin mortality to insignificant levels in the ETP.

Actions of the United States

- 4.17 The principal US law relating to the overall issue of the protection of dolphins and other marine mammals is the Marine Mammal Protection Act of 1972, as amended (the "MMPA").
- 4.18 In 1988, the United States amended provisions of the MMPA to prohibit the import into the United States of any marine product, and any fish or fish product harvested where there was not a program comparable to that of the United States in minimizing the incidental taking of marine mammals. Pursuant to these amendments, in August 1990 the United States imposed an import embargo on imports of tuna from Mexico (and other countries) for failure to achieve comparability with US tuna harvesting standards that prohibited setting on dolphins. Mexico challenged this embargo in dispute settlement proceedings under the General Agreement on Tariffs and Trade ("GATT") and prevailed, but the panel was never adopted.
- 4.19 In addition to the direct ban on imports, in 1990 the United States enacted legislation that established a standard for labelling tuna products as "dolphin-safe". That law, known as the Dolphin Protection Consumer Information Act of 1990 ("DPCIA"), amended the MMPA.
- 4.20 According to this standard, with respect to tuna harvested by a vessel using purse-seine nets in the ETP, the DPCIA provided that the product could not be labelled "dolphin-safe" if caught on a trip involving intentional deployment on, or encirclement of, dolphins. Accordingly, Mexican tuna and tuna products harvested while setting on dolphins are denied the "dolphin-safe" label in the US marketplace, even if no mortality or serious injury of a marine mammal was observed.

- 4.21 Later, in 1997, in order to comply with the commitments of the Panama Declaration, the US Congress enacted the International Dolphin Conservation Program Act ("IDCPA"), which became effective on February 15, 1999, The IDCPA amended the MMPA so that a country would be permitted to export tuna fished from the ETP to the United States if it provided documentary evidence that (a) it participates in the IDCP and is a member of the IATTC; (b) it meets its obligations under the IDCP and the IATTC; and (c) it does not exceed specified dolphin mortality limits.
- 4.22 In the 1995 Panama Declaration it was also agreed that the definition of "dolphin-safe" would be changed from "no encirclement of dolphins" to "no dolphin mortality or serious injury. Thus, the legislation also authorized a change in the "dolphin-safe" labelling standard, but not immediately or automatically, but contingent on the outcome of studies it required the Department of Commerce ("DOC") to perform of "whether the intentional deployment on or encirclement of dolphin with purse-seine nets is having a significant adverse impact on any depleted dolphin stock in the eastern tropical Pacific Ocean". Thus, the standard would be changed in accordance with the US international obligation unless there was clear evidence that the method of fishing was actually having a significant adverse impact on dolphin stocks.
- 4.23 On May 1999 and December 2002, the DOC published, respectively, its initial and final findings pursuant to the IDCPA. In both findings, the DOC determined that there was insufficient evidence to conclude that intentional encirclement of dolphins with purse-seine nets was having a significant adverse effect on "depleted dolphin stocks in the ETP".
- 4.24 Certain NGOs, challenged both the Initial and Final Findings. In both cases, the District Court held that the DOC abused its discretion when it triggered a change in the dolphin-safe label standard on the ground that it lacked sufficient evidence of no significant adverse impacts, and in both cases, the Court of Appeals for the Ninth Circuit affirmed the Districts Court's Decision. The last ruling of the Ninth Circuit, in 2007, is known as the *Hogarth* Ruling.
- 4.25 Therefore, and despite the international commitment of the United States, the standard remained unchanged.
- 4.26 It is instructive to note that despite the evidence that dolphins and other marine mammals are killed in fisheries in the US domestic waters, the United States has not adopted requirements for the dolphin-safe label for tuna or other fish caught in those fisheries remotely comparable to the standards applied by the US regulations for the ETP.
- (iii) The Mexican fishing fleet
- 4.27 Between 1980 and 1987, Mexico developed a fleet of 85,000 tons of carrying capacity, along with an associated infrastructure and employment base. A number of coastal communities were effectively built and sustained on the comparative advantage given by the strength of the tuna resource along Mexico's coast, and the markets for that resource around the world.
- 4.28 Aware of the importance of protecting dolphins while fishing tuna, the Mexican fleet invested large amounts of money in the dolphin protection efforts such as training of captains and crews, development of fishing gear less harmful to dolphins, research and monitoring programs, limiting its fishing operations to daylight, and carrying an independent observer in every fishing trip of the Mexican fleet. Those efforts contributed to the reduction of dolphin mortality to insignificant levels in the ETP.

(iv) Adverse effects of the measure

4.29 The US measures have direct and indirect adverse effects. The direct adverse effect relates to tuna products imported, distributed and sold in the US market. The US distribution and retail networks for tuna products are acutely aware of the dolphin-safe issue and the fact that they will encounter actions such as boycotts, promoted by certain economically interested NGOs, if they carry tuna that is not designated as dolphin-safe. Large US grocery chains have indicated that they will be unable to carry any Mexican tuna products unless the tuna products bear a US government approved dolphin-safe label. The indirect adverse is that the three major processors of tuna brands sold in the United States – StarKist, Bumblebee, and Chicken of the Sea – refuse to purchase tuna caught in the ETP including Mexican tuna because tuna products containing such tuna cannot be labelled as dolphin-safe.

(c) Subject products

- 4.30 The subject products are tuna and tuna products.
- 4.31 "Tuna" include all species of tuna purchased by canneries for processing into tuna products including Yellowfin, Albacore, and Skipjack, and "tuna products" are defined in Section 1385(c)(5) of the DPCIA as "a food item which contains tuna and which has been processed for retail sale, except perishable sandwiches, salads, or other products with a shelf life of less than 3 days".

(d) Specific measures at issue

- 4.32 Although several US statutes, regulations and court decisions are involved, the most pertinent measures are: the *United States Code*, Title 16, Section 1385 ("Dolphin Protection Consumer Information Act"); the *Code of Federal Regulations*, Title 50, Section 216.91 ("Dolphin-safe labelling standards") and Section 216.92 ("Dolphin-safe requirements for tuna harvested in the ETP [Eastern Tropical Pacific Ocean] by large purse seine vessels"); and the ruling in *Earth Island Institute v. Hogarth*, 494 F.3d 757 (9th Cir. 2007).
- 4.33 As it stands, because of the *Hogarth* ruling and the underlying statute and regulations, Mexican tuna products containing tuna harvested by vessels using purse-seine nets in the ETP are entitled to use the "dolphin-safe" label only if an AIDCP observer was on board the vessel, and both the observer and the captain of the vessel certify "that no tuna were caught on the trip in which such tuna were harvested using a purse-seine net intentionally deployed on or to encircle dolphins, and that no dolphins were killed or seriously injured during the sets in which the tuna were caught ...".
- 4.34 In contrast, if tuna products contain tuna harvested outside the ETP by a vessel using purse-seine nets, those products are eligible for the "dolphin-safe" label simply if the captain of the vessel executes a written statement certifying that "no purse-seine net was intentionally deployed on or used to encircle dolphins during the particular voyage on which the tuna was harvested". There is no requirement of an observer. There is no requirement of participation in an international program similar to the AIDCP. Moreover, there is no requirement of an attestation that no dolphins were killed or seriously injured during the tuna fishing.

(e) Legal argument

4.35 The US measures are inconsistent with Articles III:4 and I:1 of the GATT 1994 and Articles 2.1, 2.2, and 2.4 of the TBT Agreement.

(i) GATT 1994

Mexico's claim under Article III:4 of the GATT 1994

4.36 The US measures are laws, regulations and requirements affecting the internal sale, offering for sale, purchase, distribution and use of Mexican tuna products and tuna imported into the United States and accord less favourable treatment to Mexican tuna products and tuna than that accorded to the like products of US origin and, therefore, are inconsistent with Article III:4 of the GATT 1994. Accordingly, they afford protection to the production of like US tuna products and tuna.

Like products

- 4.37 In *EC Asbestos*, the Appellate Body stated that "a determination of 'likeness' under Article III:4 is, fundamentally, a determination about the nature and extent of a competitive relationship between and among products". Accordingly, the Mexican and US tuna and tuna products are like product within the meaning of Article III:4 because Mexican and US tuna products directly compete against each other in the US wholesale, distribution and retail market and Mexican and US tuna directly compete against each other in the US cannery market.
- 4.38 In EC Asbestos, the Appellate Body also observed that the Report of the Working Party on Border Tax Adjustments outlined an approach for analysing "likeness" that has been followed and developed since by several panels and the Appellate Body. The approach consists of employing four general criteria outlined in the report. The Appellate Body stated that the panel must examine the evidence relating to each of those four criteria and, then, weigh all of that evidence, along with any other relevant evidence, in making an overall determination of whether the products at issue could be characterized as "like". Accordingly Mexico analyses the four criteria as follows:
 - (i) the properties, nature and quality of the product: The physical properties of Mexican tuna products are identical to those of US tuna products insofar as products from both WTO Members comprise tuna meat in a retail-ready package. The physical properties of Mexican tuna are identical to those of US tuna insofar as they are a marine fish that is purchased by canneries for processing into tuna products.
 - (ii) the end-uses of the products: The end-uses of Mexican tuna products and Mexican tuna are identical to US tuna products and tuna because tuna products from both Mexico and the United States are destined for consumption by final consumers, and tuna from both Mexico and the United States are destined for processing into tuna products at a cannery.
 - (iii) consumers' tastes and habits: But for the regulatory distinction that is at the core of this dispute (i.e., dolphin-safe), consumers' tastes and habits respecting Mexican tuna products and tuna are identical to US tuna products and tuna.
 - (iv) the tariff classification of the products: The Mexican and US tuna products and tuna are categorized under the same tariff classifications.

Law, regulation, or requirement affecting their internal sale, offering for sale, purchase, transportation, distribution, or use

4.39 The US measures are set out in US legislation and regulations as interpreted and applied by the US judiciary. Mandatory criteria have been established for the use of labels that contain the term "dolphin-safe" or any other term or symbol that claims or suggests that the tuna contained in the

products were harvested using a method of fishing that is not harmful to dolphins. Thus, the US measures clearly amount to a law, regulation or requirement within the meaning of Article III:4.

4.40 With regards to the term "affecting" used in Article III:4, the US measures "affect" the internal sale, offering for sale, purchase, transportation, distribution, or use of tuna and tuna products because the participants – processors, wholesalers, retailers and consumers – in the US tuna and tuna product market, are highly sensitive to issues related to dolphin mortality, and most of them will not purchase, offer for sale, distribute or use tuna products and tuna that are not designated as dolphin-safe or cannot be designated as dolphin-safe after processing.

Less favourable treatment

- 4.41 A measure accords less favourable treatment to imported products if it modifies the conditions of competition in the relevant market to the detriment of imported products, or, in other words, if it gives domestic like products a competitive advantage in the market over imported like products.
- 4.42 The less favourable treatment at issue is of a *de facto* rather than *de jure* character because the US measures do not, on their face, discriminate on the basis of the foreign country that is the source of particular tuna and tuna products but rather, they discriminate on the basis of where the tuna is harvested and the fishing method. This has the effect of favouring US tuna and tuna products over like Mexican tuna and tuna products because the fishing fleets of the two countries harvest tuna in different ocean fisheries using different fishing methods.
- 4.43 The *de facto* less favourable treatment in this dispute occurs as follows:
 - Reflecting the longstanding fishing practice of the Mexican fishery, Mexican tuna are almost exclusively caught in the ETP using purse-seine nets that are set upon dolphins. This is the most environmentally responsible way to fish for tuna in the Mexican fleet's traditional tuna fishing grounds. By virtue of the US measures Mexican tuna products can never be designated as dolphin-safe even though the Mexican fleet complies with the stringent dolphin-safe requirements of the AIDCP.
 - In contrast, the US fleet fishes outside of the ETP using other fishing methods such as purseseine nets that are set upon FADs. Under the US measures, US tuna products that contain this tuna can be designated as dolphin-safe even where marine mammal mortalities might, and historically have been shown to occur and where other principles of bycatch reduction might be compromised. Accordingly, by virtue of government intervention in the form of the US measures, Mexican tuna products cannot be designated as dolphin-safe while US tuna products can.
 - It is an established feature of the US tuna product market that participants in the market are sensitive to issues related to dolphin mortality and will make decisions on whether or not to purchase, offer for sale, distribute or use tuna products on the basis of whether they are designated as dolphin-safe. Accordingly, most US market participants will not purchase, offer for sale, distribute or use Mexican tuna products but will purchase, offer for sale, distribute or use US tuna products.
 - Although the US measures apply to the labelling of tuna products, they have an indirect discriminatory effect on Mexican tuna. By virtue of the US measures, tuna products containing Mexican tuna cannot be designated as dolphin-safe. As a consequence, fish canneries located in US territory (e.g., Samoa) or producing tuna destined for the US market

will not accept Mexican tuna for processing into tuna products. These canneries will, however, accept US tuna that can be designated dolphin-safe once processed.

4.44 This *de facto* less favourable treatment between Mexican tuna products and like US tuna products modifies the conditions of competition in the relevant market – i.e., the US tuna products market – to the detriment of imported Mexican tuna products. Similarly, the *de facto* discrimination between Mexican tuna and like US tuna modifies the conditions of competition in the US cannery market to the detriment of Mexican tuna. In this way, the US measures affecting these like products provide less favourable treatment and thus violate the national treatment obligation in Article III:4 of the GATT 1994.

Mexico's claim under Article I:1 of the GATT 1994

- 4.45 The US measures are also an advantage, favour or privilege granted by the United States to tuna products and tuna originating in certain WTO Members that has not been accorded immediately and unconditionally to the like products originating in Mexico and, therefore, are inconsistent with Article I:1 of the GATT 1994.
- 4.46 In the circumstances of this dispute, to determine whether there is a violation of Article I:1, three questions must be answered: (i) are the imported products concerned "like" products; (ii) does the measure at issue confer an advantage, favour or privilege on products originating in any other country; and (iii) was the advantage, favour or privilege granted "immediately and unconditionally" to the like product originating in the territories of all other Members?

Like products

4.47 For the same reasons explained when analysing "likeness" under Article III:4, tuna and tuna products, irrespective of which country they originate, are like products within the meaning of Article I:1 of the GATT 1994.

Advantage, favour or privilege on a product originating in other WTO Members

- 4.48 The advantage, favour or privilege in this dispute is the right to designate tuna products as dolphin-safe. Having a dolphin-safe designation provides a significant commercial advantage and is clearly an "advantage", "favour" or "privilege".
- 4.49 This advantage, favour or privilege is conferred upon tuna products and tuna originating in those countries whose tuna fishing fleets do not fish in the ETP and those countries whose fishing fleets fish in the ETP but use a different fishing method than the Mexican fleet.

Accorded immediately and unconditionally to the like product originating in the territories of all other Members

- 4.50 The US measures do not, on their face, discriminate on the basis of the foreign country that is the source of particular tuna and tuna products, but rather they discriminate on the basis of where the tuna is harvested and the fishing method. This has the effect of favouring tuna and tuna products from some countries over others. Accordingly, the discrimination is *de facto* rather than *de jure* in character, and occurs as follows:
 - Reflecting the longstanding fishing practice of the Mexican fishery, Mexican tuna are almost
 exclusively caught in the ETP using purse-seine nets that are set upon dolphins. This is the
 most environmentally responsible way to fish for tuna in the Mexican fleet's traditional tuna
 fishing grounds. By virtue of the US measures, Mexican tuna products can never be

- designated as dolphin-safe even though the Mexican fleet complies with the stringent dolphin-safe requirements of the AIDCP.
- In contrast, the fleets of other countries fish outside of the ETP using other fishing methods such as purse-seine nets that are set upon FADs. They also fish in the ETP using methods different from that used by the Mexican tuna fleet. Under the US measures, tuna products that contain tuna originating from these countries' fleets can be designated as dolphin-safe and a dolphin-safe label can be affixed to those products. Accordingly, by virtue of government intervention in the form of the US measures, Mexican tuna products cannot be designated as dolphin-safe while tuna products originating in other countries can.
- It is an established feature of the US tuna product market that participants in the market are sensitive to issues related to dolphin mortality and will make decisions on whether or not to purchase, offer for sale, distribute or use tuna products on the basis of whether they are designated as dolphin-safe. Thus, most US market participants will not purchase, offer for sale, distribute or use Mexican tuna products but will purchase, offer for sale, distribute or use tuna products originating in other countries.
- Although the US measures apply to the labelling of tuna products, they have an indirect discriminatory effect on Mexican tuna. By virtue of the US measures, tuna products containing Mexican tuna cannot be designated as dolphin-safe. As a consequence, fish canneries located in other WTO members will not accept Mexican tuna for processing into tuna products. These canneries will, however, accept tuna in other countries that can be designated dolphin-safe once processed.
- 4.51 This *de facto* discrimination between Mexican tuna products and like tuna products originating in other countries modifies the conditions of competition in the US tuna products market to the detriment of imported products originating in a WTO Member, namely tuna products originating in Mexico. Similarly, the *de facto* discrimination between Mexican tuna and like tuna originating in other countries modifies the conditions of competition in the US cannery market to the detriment of tuna originating in Mexico. In this way, the US measures violate the most-favoured-nation obligation in Article I:1 of the GATT 1994.

Conclusions

- 4.52 On the basis of the foregoing, the US measures are inconsistent with Articles III:4 and I:1 of the GATT 1994. Moreover, on the basis of the relevant facts, none of the general exceptions in the GATT 1994 apply to the US measures.
- (ii) Claims under the TBT Agreement
- 4.53 The US measures are a technical regulation that is inconsistent with Articles 2.1, 2.2, and 2.4 of the TBT Agreement.

The US measures constitute a technical regulation

4.54 The US measures constitute a technical regulation within the meaning of the TBT Agreement, pursuant to its Annex 1.1. In this instance, the "document" comprises the statutory and regulatory provisions that make up the labelling provisions. Also, the document meets the three criteria in order to fall within the definition of a technical regulation, as follows:

- (i) It applies to an identifiable product or group of products: The identifiable group of products to which the document applies, is "tuna product", as defined by Section 1385(c)(5).
- (ii) It lays down one or more characteristics of the product: The US measures govern the conditions under which a tuna product can be labelled as "dolphin-safe". This requirement is a product characteristic of the tuna product that is laid down by the US measures.
- (iii) Compliance with the product characteristics is mandatory: Under the DPCIA, it is unlawful to include on the label of any tuna product offered for sale in the United States the term "dolphin-safe" or any analogous term or symbol if the product contains tuna harvested in the ETP by a large purse seine vessel using a purse-seine net intentionally deployed on or to encircle dolphins. This prohibition against the use of the "dolphin-safe" label remains in force even when the international standards of the AIDCP are met. Accordingly, the regulation is mandatory. Furthermore, even if the labelling scheme were not to be considered a priori mandatory, it is de facto mandatory because the market conditions in the United States are such that it is impossible to effectively market and sell tuna products without a dolphin-safe designation.

The measures are inconsistent with Article 2.2

4.55 The US measures are a technical regulation that creates an unnecessary obstacle to international trade, because its objective is not legitimate or, in the alternative, it is more traderestrictive than necessary to fulfil a legitimate objective taking account of the risks non-fulfilment would create. Thus, they are inconsistent with Article 2.2 of the TBT Agreement.

The measures do not fulfil a legitimate objective

- 4.56 The evidence indicates that the US measures do not protect animal life or health or the environment in the general sense. Rather, the measures fulfil the opposite of these legitimate objectives by: (i) promoting the use of alternative fishing methods that result in enormously higher bycatch and therefore wasteful depletion of ocean sealife; and (ii) by undermining the economic incentive for countries and fishing fleets to participate in a very successful multilateral environmental agreement. The objective of the US measures is narrower than the protection of animal life or health or the environment, it is to preserve dolphin stocks in the course of tuna fishing operations in the ETP. In Mexico's view, measures that trade off the life or health of different animal species and which undermine broader environmental objectives that are enshrined in a successful multilateral environmental agreement cannot be found to "fulfil a legitimate objective" within the meaning of Article 2.2 of the TBT Agreement.
- 4.57 In the alternative, even if such measures could in principle be found to fulfil a legitimate objective, the US measures do not fulfil their stated objective of preserving dolphin stocks in the ETP, for the following reasons:
 - The protection of dolphins in the ETP tuna fishery and the consequent preservation of dolphin stocks are governed by the highly successful AIDCP. The US measures do not add to that protection, and all they accomplish is to block imports of tuna products and tuna from the signatory countries and, thereby, protect the US tuna industry and undermine the viability of the AIDCP.

- The Hogarth court ruling goes against the requirements of the AIDCP and against the determination of the US administration, which supported the view that the legitimate environmental objective at issue the preservation of dolphin stocks is fulfilled by the standard for dolphin-safe labelling agreed in the Panama Declaration.
- The statutory condition, unilaterally adopted by the United States requiring that a determination be made by the DOC based on a vague standard, without consultation with the IATTC, and subject to reversal by courts without scientific expertise was designed primarily to fulfil the objective of satisfying domestic political interests.

The US measures are more trade-restrictive than necessary taking account of the risks non-fulfilment would create

- 4.58 In the alternative, if the US measures are found to fulfil a legitimate objective, the US measures are more trade-restrictive than necessary to fulfil that legitimate objective taking into account the risks nonfulfillment would create.
- 4.59 The preservation of dolphin stocks in the ETP tuna fishery is an important objective. According to available scientific and technical information, that objective is being fulfilled by the AIDCP. The US measures do not further contribute to the achievement of this objective. In the absence of the US measures, the preservation of dolphin stocks in the ETP tuna fishery will continue to be accomplished by the requirements and procedures of the AIDCP. Since the US measures are not necessary, the fact they deny competitive opportunities to Mexican tuna products and tuna and thereby create trade restrictions means that they are more trade-restrictive than necessary within the meaning of Article 2.2 of the TBT Agreement.

The US measures are inconsistent with Article 2.4

4.60 The US measures are inconsistent with Article 2.4, because US dolphin-safe labelling measures are not based on an existing relevant international standard, and the relevant international standard is an effective and appropriate means for the fulfilment of the legitimate objectives pursued.

AIDCP standard

4.61 On 20 June 2001, the Parties to the AIDCP adopted the "Resolution to Adopt the Modified System for Tracking and Verification of Tuna" and "the Resolution to Establish Procedures for AIDCP Dolphin Safe Tuna Certification". Both resolutions define dolphin-safe tuna, as tuna captured in sets in which there is no mortality or serious injury of dolphins.

The AIDCP standard is a relevant international standard

International standard

4.62 The AIDCP Standard is a standard for purposes of Annex 1.2 of the TBT Agreement because it provides for common and repeated use rules expressly dealing with the characterization of tuna as dolphin-safe and non-dolphin-safe, compliance with the standard is not mandatory, and it was prepared and approved by the AIDCP member governments, which constitute a recognized body.

Relevant

4.63 The AIDCP standard is "relevant" within the meaning of Article 2.4 because it is used to determine when tuna and tuna product can be certified as dolphin-safe and bear a dolphin-safe label, and thus, it serves the exact same purpose as the US dolphin-safe label measures.

The United States failed to base its regulation on the relevant international standard

4.64 The United States failed to base its regulation on the relevant international standard because it rejected its application in favour of a unilateral standard, a setting of nets on dolphins standard. Also, the US dolphin-safe labelling measures stipulate rules for labelling tuna that directly contradict the guidelines contained in the AIDCP.

The relevant international standard does not constitute an ineffective or inappropriate means for the fulfilment of the legitimate objectives pursued

- 4.65 The US measures do not pursue a legitimate objective. However, in the event the Panel concludes that the measures pursue a legitimate objective, the AIDCP Standard is an effective and appropriate means for the fulfilment of a legitimate objective.
- 4.66 It is an effective means for achieving the pursued objective of protecting dolphins because with this standard, dolphin mortality in the ETP has decreased 99%, and even United States agreed to the standard itself. It is also an appropriate means for achieving the pursued objective of protecting dolphins, because not only it has been effective in decreasing the dolphin mortality, but also it allows the best method of fishing for tuna.

The US measures are inconsistent with Article 2.1

- 4.67 The US measures are inconsistent with Article 2.1 of the TBT Agreement, which imposes national treatment and most-favoured-nation treatment obligations on technical regulations. As discussed above, Mexico argues that the US measures are a technical regulation.
- 4.68 For the same reasons set out for Mexico's claim under GATT Article III:4, the US measures do not accord products imported from the territory of any Member (namely, Mexico) treatment no less favourable than that accorded to like products of national origin.
- 4.69 For the same reasons set out for Mexico's claim under GATT Article I:1, the US measures do not accord products imported from Mexico treatment no less favourable than that accorded to like products originating in any other country.

(f) Conclusions

4.70 On the basis of the foregoing, Mexico respectfully requests that the Panel find that the US measures are inconsistent with Articles I:1 and III:4 of the GATT 1994 and Articles 2.1, 2.2 and 2.4 of the TBT Agreement.

2. United States

(a) Introduction

4.71 The measures at issue in this dispute establish conditions under which the voluntary "dolphin-safe" label may be used on tuna products. These conditions do not allow tuna products to be labelled dolphin-safe if they contain tuna that was caught by intentionally encircling and deploying purse seine nets on dolphins. This fishing technique is commonly referred to as "setting on dolphins" and involves chasing, encircling and deploying purse seine nets on dolphins to catch tuna that swim beneath the dolphins. The US measures apply to tuna caught in any fishery where there is a regular and significant association between tuna and dolphins and apply regardless of the origin of the tuna. The only known fishery where there is a regular and significant association between tuna and dolphins is the Eastern Tropical Pacific Ocean (ETP).

- 4.72 Setting on dolphins to catch tuna has well-documented adverse impacts on dolphins in the ETP. Not only are dolphins killed when encircled with purse seine nets to catch tuna (e.g., over 1000 dolphin were observed killed in sets with purse seine nets in 2009), but research indicates that setting on dolphins to catch tuna causes a number of other adverse effects on dolphins, including separation of mothers and their dependent calves and reduced reproductive success due to stress. Together these adverse effects have resulted in dolphin populations in the ETP that are small fractions of their "pre-fishery" levels. Today, the population level of two main species of dolphins impacted by dolphin fishing in the ETP northeastern offshore spotted and eastern spinner dolphins remain at only 19 and 29 per cent, respectively, of the levels that existed before setting on dolphins became the predominate technique to fish for tuna in the ETP.
- 4.73 The US dolphin-safe labelling provisions ensure that when a dolphin-safe label appears on tuna products in the United States it accurately conveys to consumers that the product does not contain tuna that was caught in a manner that adversely affects dolphins. Any tuna that is caught in the ETP including Mexican tuna during a trip using a technique other than setting on dolphins and in a set in which no dolphins were otherwise killed or seriously injured is eligible to bear the dolphin-safe label. The fundamental premise of Mexico's claims that the US measures prohibit use of the dolphin-safe label on Mexican tuna is simply incorrect.
- 4.74 Mexico is also incorrect that the US measures deny Mexico access to the US market. The US measures set out a voluntary labelling scheme. Mexico can and does sell tuna in the US market that is not labelled dolphin-safe. Moreover, tuna caught by one-third of Mexican purse seine vessels that fish for tuna in the ETP is already eligible for the dolphin-safe label; yet Mexican processors do not to use it. As detailed below, each of Mexico's legal claims fail.
- In addition, although not relevant to the legal claims in this dispute, Mexico is incorrect that the United States failed to abide by its "commitment ... to amend its law" to allow tuna products that contain tuna that was caught by setting on dolphins to be labelled dolphin-safe. The United States agrees that the AIDCP has made an important contribution to dolphin conservation in the ETP, including that it has fostered the continued reduction in *observed* dolphin mortalities in the fishery. However, this tells only part of the story. Dolphins continue to die, or be seriously injured, in the ETP as a result of setting on dolphins to catch tuna. And, their populations remain depleted in the ETP. And this is true even though the parties to the AIDCP and their fleets are generally adhering to their dolphin conservation obligations under the AIDCP. So, while the AIDCP has been successful in significantly reducing the number of observed dolphin mortalities in the ETP, dolphins continue to be adversely affected and dolphin populations have not yet demonstrated the recovery that would be expected if adverse effects of setting on dolphins to catch tuna had been completely alleviated. The United States has a right to ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught by a method that is "dolphin-safe" – that is, a method that does not involve a fishing technique that adversely affects dolphins – and to ensure that the US market is not used to encourage fishing fleets to use fishing techniques that adversely affect dolphins.

(b) Legal argument

4.76 Mexico has failed to establish a prima facie case that the US measures are inconsistent with US obligations under Articles I:1 and III:4 of the GATT 1994 and the TBT Agreement. First, the US dolphin-safe labelling provisions do not discriminate based on origin. Second, the US dolphin-safe labelling are voluntary and therefore do not constitute technical regulations falling within the scope of the TBT Agreement. Accordingly, Mexico cannot even establish that Articles 2.1, 2.2, and 2.4 of the TBT Agreement apply to the US measures. Third, the dolphin-safe labelling provisions serve legitimate objectives; allowing tuna to be labelled "dolphin-safe" that is caught by setting on dolphins would not fulfil these objectives. Accordingly, Mexico cannot sustain its claims under Articles 2.2 and 2.4 of the TBT Agreement. (Although Mexico also included Article 2.3 of the

TBT Agreement it is panel request, Mexico appears to have abandoned that claim as it is not addressed in its submission.)

- (i) The US dolphin-safe labelling provisions are not inconsistent with GATT 1994 Article III:4
- 4.77 The US dolphin-safe labelling provisions do not discriminate based on origin and therefore do not afford less favourable treatment to Mexican tuna products as compared to like domestic products. Because US dolphin-safe labelling provisions do not afford less favourable treatment to Mexican tuna or tuna products, Mexico has failed to establish a prima facie case that the US provisions are inconsistent with Article III:4 of the GATT 1994.
- 4.78 First, because the US provisions do not condition eligibility to use the dolphin-safe label on the origin of the tuna product, the US provisions do not alter the conditions of competition to the detriment of Mexican tuna or tuna products. The US provisions provide that any tuna products regardless of origin may use the dolphin-safe label if they meet the criteria for the label. Second, one-third of the vessels in Mexico's tuna fleet is under 363 metric tons and tuna products containing tuna caught by these vessels are already eligible under the US provisions to use the dolphin-safe label. This in itself is evidence that the US provisions do not afford less favourable treatment to Mexican tuna and tuna products based on origin. Third, the US dolphin-safe labelling provisions afford use of the dolphin-safe label equally to all tuna products that meet the conditions set out in those provisions. It is that Mexican vessels have chosen and continue to choose to catch tuna by setting on dolphins that makes tuna products containing tuna caught by these vessels ineligible for the dolphin-safe label.
- 4.79 The United States would like to point out that Mexico overstates the cost and difficulty of using other techniques to catch tuna and in fact provides no evidence to support its claim that using other techniques would require Mexican vessels to "incur considerable financial and other costs". The same boats and much of the same gear used to set on dolphins to catch tuna may be used to catch tuna using other techniques, specifically sets on floating objects and unassociated schools of tuna. Mexico also overstates that the ETP is its "traditional fishing grounds" and that fishing for tuna in another fishery would be too costly. While Mexican vessels have traditionally fished for tuna in the ETP, so too do vessels flagged to a number of other countries, including the United States. Mexico has provided no evidence to substantiate its contention that it would be too costly for Mexican vessels to fish for tuna in other fisheries. Mexico's statements about the relative environmental impacts of setting on dolphins to catch tuna as compared to other techniques to catch tuna in the ETP are irrelevant and unfounded.
- (ii) The US dolphin-safe labelling provisions are not inconsistent with GATT 1994 Article I:1
- 4.80 The US measures at issue are consistent with Article I:1 of the GATT 1994. The US measures do not accord any advantage to products of any other member that is not also immediately and unconditionally accorded to products of Mexico within the meaning of the GATT 1994 Article I:1. Just as the US measures do not discriminate against Mexico within the meaning of Article III:4, those measures are non-discriminatory within the meaning of Article I:1.
- 4.81 First, the US measures do not modify the conditions of competition because all tuna is subject to the same conditions on use of the dolphin-safe label. The criteria for being eligible to use the dolphin-safe label are origin neutral. Second, the US measures do not afford less favourable treatment to Mexican tuna or tuna products. Tuna products containing tuna caught by some Mexican vessels are already eligible under the US provisions to use the dolphin-safe label. Third, while Mexican vessels have traditionally fished for tuna in the ETP, so too do vessels flagged to a number of other countries, and Mexico does not explain why the origin neutral rules disadvantage it. Mexico's arguments about the relative environmental impacts of various fishing techniques are irrelevant and unfounded. The Article I:1 analysis in *US Tuna (Mexico)* could not be more relevant, as the panel, responding to

essentially the same arguments from Mexico in that dispute, concluded that the dolphin-safe labelling measures were not inconsistent with Article I:1. Finally, contrary to Mexico's assertion, the advantage, favour, or privilege granted by the United States is the opportunity to use the dolphin-safe label if the conditions on use of the dolphin-safe label are met.

- (iii) The US dolphin-safe labelling provisions are not technical regulations and therefore not subject to Article 2 of the TBT Agreement
- 4.82 Measures that fall outside the definition of a technical regulation set out in Annex 1 of the TBT Agreement are not subject to the obligation set out in Article 2 of the TBT Agreement. Because the US dolphin-safe labelling provisions do not meet the definition of a technical regulation, Article 2 does not apply and the US dolphin-safe labelling provisions cannot be inconsistent with Articles 2.1, 2.2 or 2.4 of the TBT Agreement.
- 4.83 The definition of a technical regulation in Annex 1 of the TBT Agreement makes clear that technical regulations are documents with which compliance is mandatory and that "lay down product characteristics or their related processes and production methods" or "deal exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method" or both. Article 4 and Annex 3 sets out Members' obligations with respect to documents with which compliance is voluntary, whereas Article 2 sets out Members' obligations with respect to documents with which compliance is mandatory. Notably, both the definition of standard and the definition of technical regulation encompass "labelling requirements". Whether a "labelling requirement" falls within the scope of standard or technical regulation thus depends on whether compliance with that requirement is mandatory.
- 4.84 Mexico's assertions that the US dolphin-safe labelling provisions set out product characteristics and are mandatory are incorrect. First, the US provisions specify the conditions under which tuna products may be labelled dolphin-safe. The US provisions do not specify the product characteristics (or their related processes or production methods) that tuna products must meet (or not meet) to be sold on the US market. Second, US provisions are not mandatory. The US provisions constitute a voluntary labelling measure and such voluntary labelling measures are not covered by the definition of a technical regulation. It is perfectly legal to sell tuna products in the United States that are not dolphin-safe and that do not bear the dolphin-safe label. In fact, the GATT 1947 panel reached the same conclusion when examining the DPCIA.
- 4.85 A voluntary labelling measure does not become a mandatory labelling requirement simply because the measure requires that what is stated on the label to be truthful. The definition of a standard makes clear that a standard, with which compliance is *not* mandatory, may address "labelling requirements". The difference, then, between a standard that addresses labelling requirements and a technical regulation that addresses labelling requirements is that the former concerns voluntary labelling schemes while the latter concerns mandatory labelling schemes. Confirmation that the definition of a technical regulation only includes mandatory labelling requirements can be found in the decisions and recommendations of the TBT Committee.
- (iv) The US measures are not inconsistent with Article 2.1 of the TBT Agreement
- 4.86 The analysis under Article 2.1 of the TBT Agreement is similar to the analysis of national treatment and most favoured nation provision of the GATT 1994. Mexico relies solely on the arguments it makes regarding the consistency of the US measures with Articles I:1 and III:4 of the GATT 1994 for its arguments under TBT Agreement Article 2.1. The United States has articulated above why Mexico's arguments under Articles I:1 and III:4 of the GATT 1994 fail, and Mexico's arguments under TBT Agreement Article 2.1 fail for the same reasons. Thus, Mexico has failed to establish that the US measures are inconsistent with Article 2.1 of the TBT Agreement.

- (v) The US dolphin-safe labelling provisions are not inconsistent with Article 2.2 of the TBT Agreement
- 4.87 The first sentence of Article 2.2 of the TBT Agreement establishes the general rule that Members shall ensure that technical regulations do not create unnecessary obstacles to international trade, while the second sentence of Article 2.2 explains that "for this purpose, technical regulations shall not be more trade-restrictive than necessary to fulfil a legitimate objective". The preamble to the TBT Agreement makes clear that each Member has the right to decide for itself which legitimate objectives to pursue and to take measures to meet those objectives "at the levels it considers appropriate," including with respect to measures to protect animal life or health or the environment and to prevent deceptive practices.
- 4.88 The US dolphin-safe labelling provisions are to fulfil a legitimate objective within the meaning of Article 2.2 of the TBT Agreement. The objectives of the US dolphin-safe labelling provisions are (1) ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins; and (2) to the extent that consumers choose not to purchase tuna without the dolphin-safe label, the US provisions ensure that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins.
- 4.89 Mexico does not address whether the US provisions fulfil the objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphin and therefore has not established a prima facie case that the US dolphin-safe labelling provisions are more trade-restrictive than necessary to fulfil that objective.
- 4.90 With respect to the objective of protecting dolphins, Mexico appears to believe that the objective of protecting dolphins is not legitimate because the United States should have another objective: preserving other marine species and the environment of the ETP as a whole. It is not for Mexico, however, to decide what policy objectives the United States should pursue. Mexico's argument also ignores the fact that the United States has in place a number of measures to protect other marine species and the environment generally. The US dolphin-safe labelling provisions, however, seek to protect dolphins; the US provisions need not also protect every other marine species and the environment as a whole to serve a legitimate objective. Further, Mexico's argument ignores that Article 2.2 expressly includes "protection of ... animal ... life or health, or the environment" in its illustrative list of legitimate objectives.
- 4.91 The objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins is also one that falls within the illustrative list in Article 2.2, which refers to "prevention of deceptive practices". By setting out conditions under which tuna products may be labelled dolphin-safe, the US provisions are intended to prevent deceptive practices by ensuring that tuna products are not falsely or misleadingly labelled dolphin-safe when they are caught using a fishing practice that adversely affects dolphins.
- 4.92 Mexico argues, in the alternative, that even if the US dolphin-safe labelling provisions "could in principle be found to fulfil a legitimate objective," the US measures do not fulfil that objective. Mexico's argument is without merit. First, by limiting use of the dolphin-safe label to tuna products that contain tuna that was not caught in a manner that adversely affects dolphins, the US dolphin-safe labelling provisions ensure that when consumers purchase tuna products that are labelled dolphin-safe they are not misled or deceived about the effect of the harvesting of that tuna on dolphins. Second, to the extent customers choose not to purchase tuna products without the dolphin-safe label, the US provisions help ensure that the US market is not used to encourage fishing fleets to set on dolphins. As the practice of setting on dolphins to catch tuna in the ETP decreases in frequency, the associated adverse effects on dolphin populations decrease as well.

- 4.93 Mexico's contention that the US dolphin-safe labelling provisions "will not influence or modify the conduct of the ETP fishery" is unfounded. In fact, the demand for tuna products that do not contain tuna that was caught by setting on dolphins is what prompted the US fleet to abandon this fishing technique, as well as what may have prompted Ecuador's fleet to abandon this technique in recent years. Ecuador's fleet continues to fish in the ETP, employing techniques other than setting on dolphins to catch tuna, including yellowfin tuna.
- 4.94 Mexico also argues that the US dolphin-safe labelling provisions have the effect of "withdrawing the economic incentive for countries and fishing fleets to comply with the AIDCP" and undermine the AIDCP, thereby detracting from dolphin protection. This is not supported by the facts. First, the AIDCP was concluded two years after the 1997 amendments to the DCPIA were enacted. Second, the current US dolphin-safe labelling provisions have been in force since 1997, with all parties including Mexico generally acting in compliance with their obligations under the AIDCP.
- 4.95 Article 2.2 of the TBT Agreement provides that technical regulations shall not be "more trade-restrictive than necessary" to fulfil a legitimate objective. Based on the text of Article 2.2, two elements must be shown for a measure to be considered more trade-restrictive than necessary: (1) the measure must be trade-restrictive; and (2) the measure must restrict trade more than is necessary to fulfil the measure's legitimate objective.
- 4.96 With respect to the first element, Mexico concludes that "measures that are 'trade-restrictive' include those that impose any form of limitation of imports, discriminate against imports or deny competitive opportunities to imports". The United States agrees that measures that impose limits on imports or discriminates against them would meet the definition of a measure that is "trade-restrictive". With respect to the second element, the ordinary meaning of the word "necessary" is "that cannot be dispensed with or done without; requisite, essential, needful... requiring to be done; that must be done". A measure that is "more" trade-restrictive than "necessary" is therefore a measure that restricts trade more than is needed or required to fulfil the measure's objective. The word "more" implies a comparison. This comparison in turn implies that other reasonably available measures that fulfil the measure's legitimate objective should be examined to determine whether the measure at issue is "more" than what is required or necessary to fulfil that measure's objective.
- 4.97 Article 5.6 of the SPS Agreement includes a provision similar to Article 2.2 of the TBT Agreement. Article 5.6 provides relevant context for the interpretation of Article 2.2 of the TBT Agreement within the meaning of Article 31(2) of the Vienna Convention and confirms that determining whether a measure is "more trade-restrictive than necessary" within the meaning of Article 2.2 of the TBT Agreement involves determining whether there is an alternative measure that could fulfil the measure's objective that is significantly less trade-restrictive. This interpretation is confirmed by a 15 December 1993, letter from the Director-General of the GATT to the Chief US Negotiator concerning the application of Article 2.2 of the TBT Agreement.
- 4.98 Thus, to establish that the US dolphin-safe labelling provisions are more trade-restrictive than necessary, Mexico must establish that there is a reasonably available alternative measure that fulfils the provisions' objectives that is significantly less trade-restrictive. Mexico has failed to do so.
- 4.99 First, contrary to Mexico's contentions, neither the AIDCP itself nor measures implemented pursuant to it constitute a "reasonably available alternative" that could fulfil the legitimate objectives of the US dolphin-safe labelling provisions. While the United States agrees that the AIDCP has made an important contribution to dolphin protection in the ETP, despite the conservation measures called for under the AIDCP, dolphin populations remain depleted and have not recovered. The US dolphin-safe labelling provisions together with the measures called for under the AIDCP and other provisions of US law form part of a comprehensive US strategy to protect dolphins. Were the United States to substitute one aspect of this comprehensive strategy for another for example forgo

the dolphin-safe labelling provisions in lieu of measures called for under the AIDCP — this would reduce the overall ability of the United States to protect dolphins. In addition, eliminating the US dolphin-safe labelling provisions in lieu of the AIDCP would not fulfil the objective of ensuring that consumers are not misled or deceived about whether or not tuna products contain tuna that was caught in a manner that adversely affects dolphins.

- 4.100 Second, the US dolphin-safe labelling provisions have a minimal impact on trade. As a voluntary labelling scheme, the US provisions do not require tuna or tuna products exported to, or sold in, the United States to be dolphin-safe or to be labelled dolphin-safe. And, nothing in the US provisions prohibits tuna products that are not dolphin-safe and that are not labelled as such from being exported to, or sold in, the United States. In fact, the United States imported US\$ 13 million worth of tuna and tuna products from Mexico 2009. Nor do the US provisions discriminate against imports. To the extent the US dolphin-safe labelling provisions have had an impact on trade, it is not because those measures themselves prohibit or otherwise impose limitations on imports. It is because consumers have a preference for tuna products that contain tuna that is not caught by setting on dolphins.
- 4.101 In sum, Mexico has not established that the objectives of the US provisions are not legitimate nor are more trade-restrictive than necessary. For these reasons, Mexico has failed to establish that the US dolphin-safe labelling provisions are more trade-restrictive than necessary to fulfil a legitimate objective and therefore has failed to establish that the US provisions are inconsistent with Article 2.2.
- (vi) The US dolphin-safe labelling provisions are not inconsistent with Article 2.4 of the TBT Agreement
- 4.102 As the "relevant international standard" for purpose of its claims under Article 2.4, Mexico cites the definition of "dolphin-safe" in an AIDCP resolution: "Resolution to Adopt the Modified System for Tracking and Verification of Tuna". The definition of "dolphin-safe" in the AIDCP tuna tracking resolution does not constitute a relevant international standard within the meaning of Article 2.4 of the TBT Agreement as it is not (1) a standard; (2) international; or (3) relevant.
- 4.103 The definition in the AIDCP tuna tracking resolution does not meet the definition of a "standard" in Annex 1 of the TBT Agreement. First, it does not set out "rules, guidelines or characteristics for products or related processes and production methods;" it sets out a definition for purposes of an intergovernmental agreement. This definition does not itself establish any rules regarding the characterization of tuna; it simply defines a term. Second, the definition of "dolphin-safe" in the tuna tracking resolution is not contained in a "document approved by a ... body". The AIDCP tuna tracking resolution is a document approved by the parties to the AIDCP, and neither the AIDCP nor the parties to it constitute a "body" (i.e. a "legal or administrative entity that has specific tasks and composition"). Third, assuming *arguendo* that the AIDCP was a "body" it does not have recognized activities in standardization and, therefore, would not constitute a "recognized" body.
- 4.104 The definition in the AIDCP tuna tracking resolution also does not qualify as "international". Only a limited number of countries participated in the adoption of the AIDCP resolutions, and the AIDCP by its terms limits those who were eligible to do so. These facts preclude the AIDCP resolutions from qualifying as an "international standard" under the TBT Agreement. The TBT Agreement read together with ISO/IEC Guide 2: 1991 mean that an international standard is one that is adopted by a body whose membership is open to the relevant bodies of at least all Members.

(Consensus is also required for a standard to qualify as an "international standard" under the TBT Agreement.) 51

- 4.105 The definition in the AIDCP tuna tracking resolution is not "relevant". First, the definition does not set out any rules, guidelines or characteristics for the *labelling* of tuna products. It, therefore, does not bear upon, relate to or pertain to the US dolphin-safe labelling provisions. It is also clear from the text of the tuna tracking resolution that the relevance of its definition of "dolphin-safe" is limited to defining that term for purposes of the resolution. Second, the definition of "dolphin-safe" in the tuna tracking resolution defines "dolphin-safe" as tuna caught in a set in which no dolphins were observed killed or seriously injured. The US dolphin-safe labelling provisions, however, seek to ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins, which includes not only whether the tuna was caught in a set in which dolphins were observed killed or seriously injured but whether dolphins were otherwise adversely affected. The definition of "dolphin-safe" in the tuna tracking resolution does not relate or pertain to the objective of the US dolphin-safe labelling provisions.
- 4.106 Use of the definition of "dolphin-safe in the AIDCP tuna tracking resolution would not be effective or appropriate to ensure that consumers are not misled or deceived about whether the tuna products contain tuna that was caught in a manner that adversely affects dolphins or to protect dolphins at the level the United States considers appropriate.
- 4.107 Mexico does not advance any arguments that, that definition is an effective or appropriate means to meet the objective of the US provisions to ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins. For this reason alone, Mexico has failed to establish a prima facie case that the US dolphin-safe labelling provisions are an effective and appropriate means to fulfil the objectives of those provisions.
- 4.108 With respect to protecting dolphins, while the United States agrees that the AIDCP has made an important contribution to protecting dolphins in the ETP, it only addresses part of the problem that is how to reduce dolphin mortality when setting on dolphins to catch tuna. Because it does not prohibit setting on dolphins to catch tuna, it does not ensure that no dolphins are in fact killed or seriously injured when dolphins are used to catch tuna (in fact, the AIDCP contemplates that up to 5000 dolphins may be killed using this technique per year) and it does not address other adverse effects of setting on dolphins to catch tuna. The same is true for the AIDCP resolutions Mexico cites.
- 4.109 The US dolphin-safe labelling provisions have as their objective to protect dolphins in ways that go beyond the protections provided for under the AIDCP. Thus, relying solely on the AIDCP or its resolutions would not be an effective means of fulfilling the objective of the US dolphin-safe labelling provisions to protect dolphins above and beyond minimizing observed mortalities and serious injuries as a consequence of setting on dolphins to catch tuna.
- 4.110 Relying on the AIDCP or the AIDCP resolutions would also not be effective to fulfil the other objective of the US dolphin-safe labelling provisions to ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects

⁵¹ The version of ISO/IEC Guide 2 explicitly referred to in Annex 1 of the TBT Agreement is the sixth edition, 1991. Since then, the sixth edition was cancelled and replaced by the seventh edition published in 1996 (ISO/IEC Guide 2: 1996, Standardization and related activities – General vocabulary); this in turn was cancelled and replaced by the eighth edition – now in force – published in 2004 (ISO/IEC Guide 2: 2004, Standardization and related activities – General vocabulary) (See G/TBT/GEN/38). It should be noted that for the definitions referred to in the panel report, there were no differences between the 2004 edition of ISO/IEC Guide 2, and the 1991 edition. Therefore, references in this report are made to ISO/IEC Guide 2: 1991.

dolphins because tuna caught in accordance with the AIDCP may be caught by setting on dolphins. The AIDCP and the resolution on tuna tracking would also not be appropriate to fulfil the objective of the US provisions as neither the AIDCP nor the resolution on tuna tracking addresses the labelling of tuna – that is, they are not suited for the purpose of ensuring that labels on tuna products contain accurate information about the dolphin-safe status of those products.

- (c) Conclusion
- 4.111 For the reasons stated above, the panel should reject Mexico's claims that the US dolphin-safe labelling provisions are inconsistent with Articles I:1 and III:4 of the GATT 1994 and Articles 2.1, 2.2 and 2.4 of the TBT Agreement.
- B. OPENING ORAL STATEMENTS AT THE FIRST SUBSTANTIVE MEETING

1. Mexico

- (a) Introduction
- 4.112 The essence of this case concerns certain US measures that prohibit the use of a "dolphin-safe" label on imports of tuna products from Mexico notwithstanding that these products meet the definition of dolphin-safe under the governing multilateral environmental agreement to which both Mexico and the United States are party. Accordingly, the goal of protecting dolphins in the Eastern Tropical Pacific (ETP) has long since been accomplished and the US measures are no longer needed for that purpose because dolphins are now protected under a multilateral treaty.
- 4.113 The application of a dolphin-safe label to tuna products has significant commercial value in the United States market. However, Mexican tuna products cannot benefit from that value. The US measures are inconsistent with the fundamental obligations of non-discrimination and constitute an unnecessary obstacle to international trade that is inconsistent with the TBT Agreement.
- 4.114 This dispute reflects an important concern to developing countries such as Mexico about the use by developed countries of non-tariff barriers to impede imports.
- (b) Tuna fishing in the Eastern Tropical Pacific
- 4.115 Where the Mexican fleet fishes in the ETP, mature yellowfin tuna tend to swim underneath groups of dolphins. In the 1950s and 1960s, the US fishing fleet pioneered the method of circling herds of dolphins with purse seine nets. Because mature yellowfin tuna are able to swim fast enough to swim with dolphins, tuna fishing by setting on dolphins produces a large catch of mature tuna, and there is very little bycatch of non-target sea life.
- 4.116 Although Mexican fleet was already using some techniques to protect dolphins, multilateral efforts to reduce dolphin mortality in the ETP started with a non-binding international agreement known as the La Jolla Agreement (concluded in 1992). Recognizing that the formalization of the La Jolla Agreement into a legally binding instrument was the most effective way to normalize global trade in tuna, in 1995 Mexico, the United States and other countries concluded the Panama Declaration, agreeing that if the United States (i) amended US laws on tuna imports to allow yellowfin tuna to be imported from countries participating in the IDCP, and (ii) changed its labelling definition of "dolphin-safe" from a "method of capture" standard to the "non-mortality or serious injury" standard established under the Panama Declaration, they would then enter into a binding international agreement to enhance and continue long-term cooperative multilateral dolphin protection in the ETP. Based on assurances from the United States, in 1998 the participating countries concluded the *Agreement on International Dolphin Conservation Program* (AIDCP), a binding

agreement for dolphin conservation and overall ecosystem management in the ETP. No other fishery in the world has ever adopted comparable techniques to protect dolphins or other bycatch.

- 4.117 Prior to the multilateral efforts to protect dolphins, the US fleet was estimated to be causing the deaths of over 100,000 dolphins per year. Currently, the total mortality for all the fleets in the ETP is approximately 1,000 per year. The available evidence indicates that at least the same amount or more dolphins are being killed outside the ETP in alternative fishing operations.
- 4.118 The techniques used to protect dolphins in the ETP include backing down against the net and causing the edges of the nets to create a channel through which the dolphins may escape; using divers and personnel to assist the remaining dolphins; fitting the top portion of the purse-seine with a skirt of mesh that prevents dolphins from catching their beaks or heads in the net; and prohibiting fishing after sundown to ensure that dolphins within the net can be seen and safely removed. Importantly, every vessel capable of setting purse-seine nets on dolphins has an independent observer on board to monitor whether any dolphin has been harmed and to ensure that all rules of the agreement are followed.
- 4.119 Instead of incurring the costs and responsibilities that have been shouldered by the Mexican fleet, the US fleet gradually abandoned the ETP and moved to unregulated fisheries farther west in the Pacific.
- 4.120 Other methods used for catch tuna by other fleets in the ETP and in other fisheries include: (i) setting purse seine nets on logs or artificial Fish Aggregating Devices; (ii) setting purse seine nets on free-swimming schools of tuna; (iii) long line fishing, and (iv) gillnets. These alternative methods result in highly destructive bycatch of non-target species, including dolphins. In 1996, then-Vice President Al Gore pointed out the deficiencies of other fishing methods highlighting the "unacceptably" bycatch of several species. Yet, the United States wants Mexico to change its fishing method to ones that are more environmentally damaging. The US action is a clear example of a developed country member seeking to impose a unilateral standard on an activity of a developing country occurring outside US borders.
- (c) History of US restrictions on imports of Mexican tuna products
- 4.121 This is only the latest in a series of the restrictions that the United States has imposed on imports of Mexican tuna going back more than 20 years. Over this period the United States has, through various means, denied meaningful access to Mexican tuna and tuna products to the US market.
- 4.122 During the 1980s and 1990s, the United States imposed a complete embargo on imports of Mexican tuna. In 1991, Mexico initiated a GATT 1947 dispute settlement proceeding regarding that embargo, and the panel ruled in Mexico's favour. However, the United States did not lift the embargo on Mexico until 2000, when it acknowledged that Mexico had fulfilled its obligations under the AIDCP. Despite the United States commitment to changing its dolphin-safe labelling regulation, political interests and the US courts blocked the change in the technical regulation. For the past 20 years Mexico has been extremely patient in working cooperatively with the United States by complying with all of the US-imposed conditions and waiting for the US political and court procedures to be completed. In the meantime, the Mexican industry has borne all the costs and burdens of implementing measures to ensure effective protection of dolphins in the ETP, and yet has been unable to benefit from the use of the dolphin-safe label in the US market. Meanwhile, the United States has not even attempted to impose similar obligations on fleets fishing in other ocean areas; yet tuna from other fisheries is allowed to bear the dolphin-safe label.

4.123 By choosing to act unilaterally rather than multilaterally, the United States has undermined the very purpose of the AIDCP, which has been a very successful multilateral environmental agreement. Thus, the US measures do not serve to enhance protection of dolphins and, to the contrary, increase adverse impact on the ecosystem.

(d) The US measures

- 4.124 The measures comprise the legislation, the regulations and the court decision referred to at paragraphs 121 to 135 of Mexico's first written submission. The focus of the US measures is on large purse seine vessels.
- 4.125 In all international fisheries, the greatest portion of tuna caught commercially is captured by large vessels using purse seine nets. In Mexico's case, approximately 95 per cent of its annual tuna catch is from these large vessels. On this point the United States argues that "one third" of the Mexican fleet is small vessels that do not set upon dolphins and can therefore qualify for the label. In fact, this "one third" of Mexico's fleet accounts for less than five per cent of the annual catch.
- 4.126 Under the US measures, for tuna products made from fish caught in the ETP by large vessels with purse seine nets there must be a certification that the tuna was not caught using the dolphin set method and that no dolphins have been killed or seriously injured during the set in which the tuna were caught, among other requirements. This certification must be verified by an independent observer who was onboard the vessel during the voyage. This requirement disqualifies Mexican tuna products from using the dolphin-safe label even though the tuna is caught in full compliance with the requirements of the AIDCP, and an independent observer certifies that no dolphins were harmed or injured in the specific set in which the tuna were caught.
- 4.127 For tuna products produced from tuna caught in any other fishery by large vessels with purse seine nets, the dolphin-safe label can be used if the captain of the vessel has signed a certification that "no purse seine net was intentionally deployed on or used to encircle dolphins during the particular voyage on which the tuna was harvested". This means that no certification is required regarding whether dolphins were killed or injured for fisheries outside the ETP. Therefore, it does not matter to the dolphin-safe designation if dolphins were killed or injured, as long as it is certified that the dolphin set method was not used. The captains' certifications are unverifiable, since there is no requirement for certifications by independent observers.
- 4.128 There is an obvious contradiction in the US approach to the issues in this dispute. The United States, with Mexico and other countries, concluded the AIDCP to make tuna fishing in the ETP dolphin-safe. Today dolphin mortality in the ETP is only a tiny fraction of what it was two decades ago. But the United States now disregards this international success and the multilateral process it helped create. It continues to prevent Mexican tuna products from being labelled dolphin-safe.
- (e) The US measures are discriminatory in violation of Articles I and III of the GATT 1994 and Article 2.1 of the TBT Agreement
- 4.129 Mexico's principal claims concern the discriminatory nature of the US measures. In this regard, the United States cites the unadopted 1991 GATT panel report, claiming that it ruled on this issue already. However, at the time of that dispute, imports from Mexico were subject to a complete embargo, and therefore there was no evidence pertaining to the *de facto* discriminatory effect of the labelling measures in the US market. Mexico have that evidence now. In the US market for tuna products, a dolphin-safe label is commercially valuable. Mexico has submitted evidence regarding the necessity of a dolphin-safe label to market Mexican tuna products through the major distribution channels, and the value to the Mexican tuna industry of being able to affix the label to its tuna

products. This was confirmed by the then US Assistant Secretary of State for Oceans and International Environment in a statement submitted to the US court. He recognized that, without the dolphin-safe label, the ability of fishermen to sell tuna in the US market is severely curtailed.

- 4.130 Mexico has a comparative advantage in fishing for tuna because mature yellowfin are found near the Mexican coastline and adjacent waters. The large size of these tuna significantly increases the efficiency of capture and processing. Furthermore, Mexican yellowfin tuna products are superior, in terms of taste and quality, to products made from most other tuna species. Therefore, the Mexican tuna industry has clear economic and quality advantages over its competitors. Notwithstanding this comparative advantage, Mexican tuna products are excluded from the major distribution channels in the US market.
- 4.131 The US measures discriminate *de facto* against Mexican tuna products. De facto discrimination occurs because of the economic and other circumstances related to certain products and the application of a government measure in those circumstances. This type of discrimination tilts the balance of competitive opportunities against the imported products. The US measures discriminate against Mexican tuna products because the Mexican fleet fishes for tuna using dolphin sets and therefore its tuna products are prohibited from having the label while the fleets of other countries, including the United States, fish using other methods and therefore are permitted to use the label. Thus, the US measure is attempting to pressure Mexico to make the choice between fishing grounds and methods in order to be eligible for a dolphin-safe label.
- 4.132 The main distinguishing facts between this dispute and the many *de facto* discrimination findings that have been made is that the measure in question relates to the production process methods underlying the product rather than relating directly to the product and that the label is necessary to access the principal distribution channels.
- 4.133 The same observations apply to Mexico's *de facto* most-favoured-nation discrimination claims under GATT Article I:1 and Article 2.1 of the TBT Agreement when the treatment accorded to Mexican tuna products is compared to that accorded to tuna products from other countries.
- 4.134 The United States' narrow interpretation of these non-discrimination provisions would allow developed country Members to structure their internal measures so as to unilaterally impose their regulations on other WTO Members, in particular developing country Members. One of the stated objectives of the challenged measure is to encourage fishing fleets such as those of Mexico to change their fishing practices.
- 4.135 In fact, the United States actually imposes more liberal conditions for the use of the labelling standard in all fisheries other than the ETP because, in those other fisheries, no certification is required that no dolphin were killed or seriously injured and independent observers are not required. This further magnifies the discrimination against Mexican tuna products because costs that must be incurred by the Mexican fleet do not have to be incurred by the fleets of other countries.
- 4.136 The United States suggests that to be eligible for a dolphin-safe label the Mexican fleet could use different fishing methods, minimizing the cost and ignoring the implication of switching to alternative fishing methods, such as: (i) huge adverse environmental implications and (ii) financial costs.
- 4.137 The United States also argues that Mexico is a net importer of tuna products, that Mexico could not export to the United States, and therefore cannot be discriminated against. This ignores the fact that the yellowfin tuna caught by Mexico in the ETP is a high-quality tuna that could attract a premium price in the US market and therefore make exports to the US market commercially attractive.

- 4.138 Mexico has presented prima facie evidence that Mexican tuna products are excluded from the major distribution channels in the United States because of the prohibition against using the dolphin-safe label, while the tuna products of the United States and other countries are allowed to use the label and therefore are not excluded from those channels. That is sufficient for the panel to conclude that the US measures do not provide national and MFN treatment.
- (f) The US measures are inconsistent with the TBT Agreement
- 4.139 The US tuna measures are mandatory within the meaning of the TBT Agreement. The US measures make it unlawful to include on the label of any tuna product offered for sale the term "dolphin-safe" or any analogous term or symbol if the product contains tuna harvested in the ETP by a vessel fishing with the dolphin set method, even when the international standards of the AIDCP have been met.
- 4.140 In EC-Asbestos, the Appellate Body determined that product characteristics may be prescribed or imposed in either a positive or a negative form. The dolphin-safe labelling rules meet the latter criterion and therefore are mandatory. The situation is similar to that addressed by the Appellate Body in EC-Sardines. However, the situation in this dispute is more extreme than in EC-Sardines because if the label is not used, tuna products are largely excluded from the US market. For these reasons, Mexico's approach is consistent with that of the Appellate Body in Asbestos and Sardines.
- (i) The US measures are inconsistent with TBT Article 2.2
- 4.141 The US tuna measures do not fulfil the objective of ensuring the consumers are not misled or deceived about whether tuna products contain tuna that was caught by a method that adversely affects dolphins, because they allow tuna products to be labelled as dolphin-safe even if they contain tuna from fishing sets in which dolphins were killed or injured, when the tuna was caught anywhere outside the ETP. Most consumers would expect the term dolphin-safe to mean that no dolphins were killed or injured in harvesting the tuna, but the label does not mean that for tuna caught outside the ETP. The US measures therefore mislead consumers. This is demonstrated by a recent national poll conducted by a highly regarded polling firm. The poll indicated that 48% of the public believes that dolphin-safe means no dolphins were injured or killed in the course of capturing the tuna (AIDCP standard), 22% believes it means there is no dolphin meat in the can, and only 12% believe it means dolphins were not encircled and then released in the capture of the tuna (US standard).
- 4.142 Moreover, the US measures encourage fishing fleets to use techniques that are not necessarily safe for dolphins. There are substantial dolphin mortality rates associated with those alternative methods. Some of the evidence comes from the US government itself. It is well established that dolphins are regularly killed in gillnets, both in US waters and elsewhere. Plainly dolphins are at risk all over the world, from a variety of different fishing methods. In truth, the ETP is the fishery where dolphins are best protected.
- 4.143 Thus, the US measures have the effect of disadvantaging Mexican tuna products and the Mexican industry, which the United States agrees is in full compliance with the AIDCP, while allowing tuna products from many other countries to be misleadingly labelled dolphin-safe even if dolphins were killed in the sets in which the tuna was caught. Thus, the US measures are more trade restrictive than necessary to accompany the stated objectives. If the objective is to encourage fishing methods that do not cause adverse effects to dolphins the AIDCP standard would accomplish the US goal in a less restrictive manner. Indeed, the US government long agreed. It was the court decision that prevented the United States from implementing its multilateral commitments.

- (ii) The US measures are inconsistent with TBT Article 2.4
- 4.144 Mexico has shown that there is an international standard defining the term dolphin-safe, established through the AIDCP. The United States is a founding member of the AIDCP and agreed to that standard. Yet the United States ultimately failed to base its own domestic standard on the international standard to which it has agreed. Accordingly, the US measures are inconsistent with Article 2.4 of the TBT Agreement.

(g) Conclusions

4.145 From the facts of this dispute and the evidence provided by Mexico it is clear that the tuna measures maintained and adopted by the United States discriminate against Mexican exports of tuna products. The measures modify the conditions of competition, and do not fulfil a legitimate objective.

2. United States

- (a) Introduction
- 4.146 The measures at issue in this dispute collectively referred to as the US dolphin-safe labelling provisions establish conditions under which tuna products may voluntarily be labelled dolphin-safe. These conditions ensure that when a dolphin-safe label appears on a tuna product in the United States it accurately conveys to consumers that the product does not contain tuna that was caught in a manner that adversely affects dolphins.
- 4.147 Mexico alleges that the US dolphin-safe labelling provisions are inconsistent with US obligations under the General Agreement on Tariffs and Trade 1994 (GATT 1994) and the Agreement on Technical Barriers to Trade (TBT Agreement). The Panel should reject Mexico's claims.
- 4.148 First, Mexico has not adduced evidence sufficient to demonstrate that the US dolphin-safe labelling provisions afford less favourable treatment to Mexican tuna products as compared US tuna products or tuna products of any other country. This is not surprising as the US dolphin-safe labelling provisions do not discriminate based on origin. Mexico, therefore, has not established that the US provisions are inconsistent with Articles I:1 or III:4 of the GATT 1994.
- 4.149 Second, the US dolphin-safe labelling provisions establish a voluntary labelling scheme. Because the US provisions do not set out labelling requirements with which compliance is mandatory, they do not meet the definition of a technical regulation under the TBT Agreement and, therefore, are not subject to Articles 2.1, 2.2, or 2.4 of the TBT Agreement.
- 4.150 Third, even if the US dolphin-safe labelling provisions were considered technical regulations, they fulfil legitimate objectives that could not be fulfilled if the provisions permitted tuna caught by setting on dolphins to be labelled dolphin-safe. Therefore, even aside from the fact that they are not considered technical regulations, the US provisions would not breach Articles 2.1, 2.2 or 2.4 of the TBT Agreement.
- 4.151 Before turning to these points, we highlight a point that is central to this dispute: setting on dolphins to catch tuna adversely affects dolphins. Intentionally setting on dolphins to catch tuna results in both observed and unobserved dolphin mortalities. The effects on dolphins caused by this fishing technique include death by starvation or from predation when dependent calves are separated from their mothers during high-speed chases and acute cardiac and muscle damage caused by the exertion of avoiding pursuing speedboats and helicopters for prolonged periods. At least 5 million dolphins were killed from 1959 to 1976 in the Eastern Tropical Pacific Ocean (or ETP) as a result of

being chased and encircled to catch tuna. Despite conservation measures adopted since that time, populations of two primary species of dolphins in the ETP remain depleted, at only 19 and 35 per cent of their pre-1959 levels. Moreover, there are no clear signs that these depleted dolphin populations are recovering, and the best available science tells us that setting on dolphins to catch tuna is the most probable reason that these populations remain depleted and show no clear signs of recovery.

(b) Article III:4 of the GATT 1994

- 4.152 Mexico has acknowledged that US dolphin-safe labelling provisions do not, on their face, afford less favourable treatment to imported tuna products, but instead claims that the US provisions do so in fact.
- 4.153 Mexico has failed to show that the US provisions use the manner in which tuna is caught as a means in fact to single out imports for treatment that is different than the treatment afforded domestic products, let alone treatment that is less favourable. In this regard, Mexico wrongly identifies the Appellate Body report in *Korea Various Measures on Beef* as setting out the legal approach the Panel should take in analysing Mexico's claim under Article III:4.
- 4.154 In this dispute, the US dolphin-safe labelling provisions on their face afford the same treatment to imported and domestic tuna products. There is no reason to evaluate whether those provisions cause a change in the conditions of competition to the detriment of imported products without first examining whether those provisions in fact afford treatment that is different for imported and domestic products.
- 4.155 Indeed, rather than the *Korea Various Measures on Beef* report, the United States suggests that the Panel may find it instructive to consider the panel report in *Mexico Taxes on Soft Drinks* as well as the reports in the *Korea Alcoholic Beverages*, *Chile Alcoholic Beverages*, and *Dominican Republic Import and Sale of Cigarettes* disputes. In those disputes, the challenged measures did not on their face distinguish between domestic and imported products, but allegedly discriminated against imports in fact.
- 4.156 In this dispute, Mexico has not adduced similar evidence to show that the US dolphin-safe labelling provisions although origin neutral on their face in fact use the manner in which the tuna was caught to single out imports. It is not credible to argue that the US conditions for labelling tuna dolphin-safe act as a proxy to distinguish between domestic and imported tuna products, when most imported products contain tuna that was caught by methods other than setting on dolphins and are eligible for, and in fact, use a dolphin-safe label.
- 4.157 While Mexico asserts that its fleet "almost exclusively" sets on dolphins to catch tuna, this is incorrect. One-third of Mexico's purse seine fleet exclusively uses techniques other than setting on dolphins to catch tuna and therefore tuna caught by these vessels is eligible to use the dolphin-safe label. The remaining two-thirds of Mexico's purse seine fleet also opportunistically uses techniques other than setting on dolphins to catch tuna, and the tuna caught by these vessels using those techniques are also eligible to use the dolphin-safe label.
- 4.158 We also recall that the Appellate Body has found that the absence of a clear relationship between the stated objectives of a measure and distinctions it draws between like products can be a factor in determining whether those distinctions which are on their face origin neutral in fact single out imports. In this dispute, however, there is a clear relationship between the objectives of the US dolphin-safe labelling provisions and the conditions under which tuna products may be labelled dolphin-safe.

- 4.159 Mexico also contends that because the Mexican fishing fleet primarily fishes for tuna in the ETP, the US provisions afford less favourable treatment to Mexican tuna and tuna products. This is also incorrect. First of all, it is the choice of fishing method and whether any dolphins were killed or seriously injured, not the place where the tuna was caught, that determines whether tuna products are eligible to be labelled dolphin-safe. Second, even if where the tuna was caught determined eligibility to label tuna as dolphin-safe, there were 46 US purse seine vessels, of which 31 were full-time, that fished for tuna in the ETP in the year the statute was enacted in 1990. By comparison, in 1990 Mexico had 52 vessels that fished for tuna in the ETP. Further, vessels from a number of countries fish for tuna in the ETP. Tuna caught in the ETP therefore cannot be equated with tuna of Mexican origin.
- 4.160 Mexico has also failed to show that the US provisions modify the conditions under which domestic and imported tuna and tuna products compete. The US provisions allow domestic and imported tuna products the same opportunities to compete in the US market.
- 4.161 In this regard, the US provisions provide producers a choice. They can set on dolphins to catch tuna, in which case they cannot label tuna products containing that tuna dolphin-safe, or they can use other methods and ensure that no dolphins are killed or seriously injured in the set, in which case they are eligible to label tuna products containing that tuna dolphin-safe.
- 4.162 Mexico appears to suggest that its proximity to the ETP gives it a competitive advantage relative to the US and other countries in terms of fishing for tuna by setting on dolphins. Other countries, including the United States, are similarly close to the ETP, including those areas of the ETP where setting on dolphins to catch tuna occurs.
- 4.163 The possibility that Mexico's fleet may incur some costs to switch from setting on dolphins to using other techniques to catch tuna is not evidence that the US provisions afford less favourable treatment to Mexican tuna products.
- (c) Article I:1 of the GATT 1994
- 4.164 Mexico has also failed to establish that the US dolphin-safe labelling provisions are inconsistent with Article I:1 of the GATT 1994. In examining whether the US dolphin-safe labelling provisions were inconsistent with Article I:1, a 1991 panel under the GATT 1947 rejected Mexico's claims. In particular, the panel found the US provisions "applied to all countries whose vessels fished in the [ETP] and thus did not distinguish between products originating in Mexico and products originating in other countries".
- 4.165 Analysing whether a measure complies with Article I:1 of the GATT 1994 involves among other things consideration of (1) whether the measure accords an advantage to products originating in any Member and (2) whether that advantage is accorded immediately and unconditionally to products originating in any other Member.
- 4.166 With respect to the first consideration, Mexico wrongly identifies the "advantage" at issue in this dispute. The US provisions grant the advantage of the opportunity to use the dolphin-safe label to products that meet the conditions for using the dolphin-safe label. With respect to the second consideration, Mexico has not established that the conditions the US provisions establish for labelling tuna products dolphin-safe while origin neutral on their face in fact act as a proxy to single out imports from some countries over others as eligible to be labelled dolphin-safe.
- 4.167 In this regard, one-third of Mexico's purse seine fleet exclusively uses techniques other than setting on dolphins to catch tuna and the remaining two-thirds of Mexico's fleet also opportunistically uses techniques other than setting on dolphins to catch tuna. Additionally, the technique of setting on

dolphins to catch tuna is not unique to the Mexican fishing fleet. The fishing fleets of Colombia, El Salvador, Guatemala, Nicaragua, Panama, and Venezuela also have vessels that set on dolphins, among other techniques, to catch tuna in the ETP.

- 4.168 As in the case with Mexico's argument under Article III:4, Mexico's argument that the US provisions afford less favourable treatment to Mexican tuna and tuna products because the Mexican fleets primarily fish for tuna in the ETP should be rejected.
- 4.169 In this connection, it may be helpful for the panel to consider another dispute where complainants argued that a measure that was origin neutral on its face in practice discriminated against imports from certain countries as compared to others. In *Canada Autos*, for example, the Panel found that limiting eligibility for an import duty exemption to certain importers in practice discriminated against imports originating in certain countries and therefore breached Article I:1 of the GATT 1994. In contrast to the situation in *Canada Autos*, Mexican fishing vessels can choose to meet the conditions that would make products containing their tuna eligible for the dolphin-safe label. The fact that a significant portion of Mexico's fleet has chosen not to do so, cannot be attributed to the US provisions or any failure of those provisions to afford Mexican tuna products an advantage they accord to like products originating in other countries.
- 4.170 Mexico's arguments that it would be costly for Mexican vessels to adopt alternative fishing techniques should be rejected for the same reasons as they should be under Mexico's Article III:4 claim.
- (d) Article 2.1 of the TBT Agreement
- 4.171 Mexico relies on the same evidence and argument to support its claim that the US provisions afford less favourable treatment to Mexican tuna products as compared to domestic tuna products and tuna products originating in other countries in breach of Article 2.1 as it does in respect of its GATT 1994 Article III:4 and I:1 claims. As already reviewed in today's statement, and in the US first written submission, the Panel should reject Mexico's claims under Articles III:4 and I:1 of the GATT 1994. Moreover, US dolphin-safe labelling provisions are not technical regulations and therefore cannot breach Article 2.1.
- (e) US dolphin-safe labelling provisions are not technical regulations under Articles 2.1, 2.2 or 2.4 of the TBT Agreement
- 4.172 With respect to Mexico's claims under the TBT Agreement, there are several reasons why the Panel should reject Mexico's claims, and we detail those reasons in the US first written submission. Today, we will focus on the two primary reasons. One, the US dolphin-safe labelling provisions are not technical regulations and therefore are not subject to Article 2 of the TBT Agreement. As a consequence, they cannot be inconsistent with Articles 2.1, 2.2 or 2.4 of the TBT Agreement. Two, the US dolphin-safe labelling provisions fulfil a legitimate objective that cannot be fulfilled by allowing tuna products that contain tuna caught by setting on dolphins to be labelled dolphin-safe.
- 4.173 First, the US dolphin-safe labelling provisions are not subject to Article 2 of the TBT Agreement. Article 2 of the TBT Agreement concerns "technical regulations". The US dolphin-safe labelling provisions, however, are not "technical regulations". Annex 1 of the TBT Agreement defines a technical regulation as a "document that lays down product characteristics or their related processes or production methods ... with which compliance is mandatory". As elaborated in the US first written submission, under this definition two requirements must be met for a measure to be a technical regulation: (1) the measure must be either a document that lays down product characteristics or their related processes or production methods, or a document that deals exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to

- a product, processes or production method; and (2) compliance with the aforementioned product characteristics, labelling requirements, etc. must be mandatory. The US dolphin-safe labelling provisions do not meet the second element of this definition.
- 4.174 In this regard, it is useful to consider what "labelling requirements" as the term is used in Annex 1 of the TBT Agreement means. It does not mean that labelling is required in order that the product can be sold; if it did that would render the phrase "with which compliance is not mandatory" in the definition of a standard *inutile*. Instead, ISO/IEC Guide 2:1991 defines "requirement" as "a provision that conveys criteria to be fulfilled". Thus, in the context of Annex 1 "labelling requirements" means criteria or conditions that must be met in order for the labelling of a product to conform with the standard or technical regulation.
- 4.175 Marketers of tuna products are free to choose whether to participate in the US labelling scheme and regardless of that choice continue to sell their products in the United States. Compliance with the US dolphin-safe labelling provisions is, thus, not mandatory within the meaning of Annex 1 of the TBT Agreement.
- (f) US provisions are not inconsistent with Articles 2.2 or 2.4 of the TBT Agreement
- 4.176 The objectives of the US dolphin-safe labelling provisions are (1) ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins; and (2) contributing to the protection of dolphins. The prevention of deceptive practices and the protection of animal life or health are expressly identified as legitimate objectives in Article 2.2 of the TBT Agreement, and the objectives of the US dolphin-safe labelling provisions squarely fall within these two objectives. As we explained in the US first written submission, the United States is concerned about the ETP as a whole and has a number of programs and measures in place to address conservation and management of marine resources in the ETP that go beyond dolphin conservation. The US dolphin-safe labelling provisions cannot be "illegitimate" simply because other environmental concerns also merit attention.
- 4.177 With regard to Mexico's claims under Article 2.2, we disagree with Mexico's contention that the objectives of the US dolphin-safe labelling provisions could be fulfilled in the absence of the US provisions, in particular because the AIDCP and measures implemented pursuant to it fulfil those objectives. While the AIDCP has made an important contribution to dolphin conservation in the ETP, setting on dolphins to catch tuna continues to adversely affect dolphins. If the US provisions permitted tuna products containing tuna caught by setting on dolphins to be labelled dolphin-safe, the US provisions would no longer fulfil the provisions' objectives.
- 4.178 We also note that Mexico's presentation of its Article 2.2 claims appears to be based not on the text of Article 2.2 but instead on application of the legal approach used in deciding whether a measure is "necessary" within the meaning of Article XX of the GATT 1994. The elements that go into answering these respective questions differ and it would not be appropriate to apply the same legal approach to both.
- 4.179 With regard to Mexico's claims under Article 2.4, Mexico cannot establish that allowing tuna caught by setting on dolphins to be labelled dolphin-safe would be effective and appropriate in fulfilling the objectives of those provisions, even assuming for the sake of argument that the definition of "dolphin-safe" in the AIDCP resolutions constituted a "relevant international standard".
- 4.180 Under the AIDCP resolutions which Mexico wrongly cites as "relevant international standards" tuna caught by setting on dolphins may be considered dolphin-safe, notwithstanding the evidence that setting on dolphins to catch tuna adversely affects dolphins. Allowing tuna products to

be labelled dolphin-safe based on the AIDCP resolution definitions would therefore not be effective or appropriate in fulfilling the objective of the US dolphin-safe labelling provisions.

4.181 Further, Mexico's efforts to elaborate the relative ecosystem impacts of various methods to catch tuna are also not relevant to whether the US provisions are consistent with its WTO obligations. The fact that methods of catching tuna other than setting on dolphins impact the ecosystem does not mitigate the fact that setting on dolphins to catch tuna adversely affects dolphins.

(g) Amicus submission

- 4.182 We have reviewed the submission filed by the Humane Society International and the American University, Washington College of Law and believe that it contains a number of pieces of relevant and useful information that could assist the Panel in understanding the issues in this dispute.
- C. CLOSING ORAL STATEMENTS AT THE FIRST SUBSTANTIVE MEETING

1. Mexico

- 4.183 The US measures are a clear example of a developed country member seeking to impose a unilateral standard on an activity of a developing country occurring outside the US borders and render meaningless a multilateral effort that is indeed achieving astonishing results in protecting dolphins.
- 4.184 A number of issues have been discussed during our meeting. I would like to focus on a few key points in my closing statement.
- (a) Discrimination claims
- 4.185 Mexico's discrimination claims under Article I:1 and Article III:4 of the GATT 1994 and Article 2.1 of the TBT Agreement are at the core of Mexico's case.
- 4.186 As we have discussed, this is a classic example of *de facto* discrimination against a like product. The dolphin-safe labelling measure has upset the balance of competitive opportunities for imports of Mexican tuna products in favour of like US origin tuna products and tuna products of other countries.
- 4.187 Whether some small portion of Mexico's production of tuna products could be eligible for the dolphin-safe label is the not the applicable test under the WTO's rules. The test is whether the measure has upset the balance of competitive opportunities. We have submitted prima facie evidence that Mexico's high quality tuna products are excluded from the major distribution channels in the US, while US-origin tuna products and tuna products from other countries have open access to those channels.
- 4.188 Further, we have demonstrated that the dolphin-safe label has substantial value in the marketplace. And when I say dolphin-safe label, I mean a label that conveys that dolphins were not killed or seriously injured in the harvesting of the tuna in the can, which is what consumers believe it means.
- (b) Impermissible imposition of extraterritorial policies
- 4.189 As we have heard from the US the past several days, it wants to make access to its market conditional on Mexico changing its fishing methods and/or fishing areas. This goes against fundamental principles underlying the WTO because it would take away the comparative advantage of the Mexican tuna industry.

- 4.190 As Mexico has explained, different fishing methods require fishing in different areas, for different types of fish. Mexico currently focuses on catching mature dolphin-safe tuna near its coastline and in adjacent waters. FAD fishing would require moving to a different fishing area and concentrating on catching lower value skipjack or less mature yellowfin. FAD fishing would result in destructive bycatch of both immature tuna and non-target species including dolphins. It is not economical or practical for vessels to shift back and forth from one system and region to another.
- 4.191 Yesterday the US suggested Mexico should sell skipjack in the US market. But other tuna products companies sell yellowfin tuna products as "premium" products in the US market. Apparently the US thinks it is acceptable for the Mexican brands to be excluded from that premium market.
- 4.192 Regarding the negative impact of FAD fishing on tuna stocks, the US says Mexico could just limit or suspend entirely fishing to preserve the fishery. In other words, the US thinks that the Mexican fleet should substantially reduce its fishing which would be the result if the entire Mexican fleet shifted to FAD fishing. So when the US says the objective of the labelling measure is to stop dolphin sets, it seems to be saying that the Mexican industry should simply abandon tuna entirely.
- 4.193 Fundamentally, the US is trying to force "choices" on the Mexican industry through its labelling measure. That should not be permissible under the TBT Agreement.
- 4.194 On the other hand, we also heard the US say yesterday that it is fine with Mexico continuing to set on dolphins for tuna to be sold in Mexico. If the objective of the measure is to discourage setting on dolphins, we don't understand how the US measure accomplishes that. The only thing it accomplishes very well is to exclude Mexican tuna products from competing in the US market.

(c) Unilateral v. multilateral

- 4.195 One of the factors that makes this case unusual, and which highlights the egregious nature of the US measures, is there is already an extremely successful multilateral agreement that addresses the precise objective of the US labelling measure. Under this agreement, dolphin deaths have already been reduced to a rate of 0.02 per cent of the overall population-- which is well below what is considered acceptable under the US law that regulates the killing of marine mammals in fishing operations in US domestic waters. (We reject the highly exaggerated US claims about unobserved mortalities and will submit in writing the reasons that backs Mexico's position.)
- 4.196 As we have explained in our written submission and during this hearing, the US has long agreed that the AIDCP was successful, and that alternative methods of fishing were highly destructive and undesirable. It ratified the treaty and approved legislation to implement its international commitments. We have shown you former Vice President Gore's letter and court testimony from the then responsible US government official. There is considerably more evidence available that the US government agreed with Mexico over an extended period. Indeed, the US even changed its dolphin-safe standard to match the AIDCP standard for a short time in 2002, before court litigation led it to revert back to the original standard.
- 4.197 Now the US is telling you that it is not satisfied with the AIDCP. But it never came to the IATTC and asked to use the AIDCP structure to investigate dolphin populations in the ETP, or whether dolphin sets were causing stress to dolphins. With regard to this issue, it has simply ignored the multilateral organization that the US played a leading role in establishing.
- 4.198 Unilateral trade-related actions such as the dolphin-safe labelling measures that have the objective of pressuring developing countries into conforming their behaviour to the dictates of developed countries cannot be not tolerated under the WTO system. Thus, this case is important not

only to Mexico and the Mexican tuna industry, but also for the future operation of the WTO system. Multilateral solutions – especially those that are as successful as the AIDCP – should not be ignored.

2. United States

- 4.199 As we've discussed these past days, it is clear that Mexico is able, and does sell, their tuna products in the United States. Moreover, Mexican tuna products can be labelled dolphin-safe in the United States. The label is not barrier to the US market.
- 4.200 The US provisions establish a voluntary origin-neutral labelling scheme. These provisions apply the same conditions on use of the dolphin-safe label for tuna products regardless of origin. In any fishery where there is a regular and significant association between tuna and dolphins, or regular and significant dolphin mortality, the documentation required to substantiate dolphin-safe labels is the same: the observer and the captain must certify that no dolphins were killed or seriously injured in the set and no nets were set on dolphins during the fishing trip.
- 4.201 The purse-seine tuna fishery in the ETP is a fishery where not only is there a regular and significant tuna dolphin association, but one that is intentionally exploited on a wide-scale, commercial basis by setting on dolphins to catch tuna. Dolphin deaths are a foreseeable and expected consequence of this fishing technique, and this is why under AIDCP procedures each vessel that fishes for tuna in this way is assigned a set number of dolphins that each vessel may be observed killing each year.
- 4.202 And, observed dolphin mortalities only tell a small portion of the story. Setting on dolphins to catch tuna has other detrimental impacts on dolphins: separation of dependent calves from their mothers, reduced reproductive success due to stress induced foetal mortality, acute cardiac and muscle damage, to name a few. The best available science tells us that setting on dolphins to catch tuna is the most probable reason why dolphin stocks in the ETP remain depleted, at less than 30 per cent of their abundance levels before this fishing practice began, and is also why they are showing no clear signs of recovery. In this case, Mexico has not shown that any difference in documentation is in fact a means to afford different or less favourable treatment to Mexican imports.
- 4.203 In that regard, Mexico argues that the US provisions use fishing method and location as a way to single out Mexican imports. The evidence however does not support that claim. At the time the US provisions were adopted, 46 US and 52 Mexican purse seine vessels were registered to fish for tuna in the ETP, and most of these vessels set on dolphins to catch tuna. Since then, US vessels abandoned setting on dolphins to catch tuna to meet the conditions to label their tuna dolphin-safe. On the other hand, the Mexican vessels that were setting on dolphins chose to continue that practice, and therefore not to meet the conditions to label their products dolphin-safe. Mexico cannot rely on its vessels' choice not to meet the conditions to label their products dolphin-safe a choice the US fleet equally had to make as a basis to argue the US provisions discriminate against Mexican products.
- 4.204 Further undermining Mexico's contentions are that one-third of its fleet does not set on dolphins to catch tuna, and therefore products containing its tuna are eligible to be labelled dolphin-safe. Up until 2002, tuna caught by these vessels was sold to US canneries and sold in the US as dolphin-safe.
- 4.205 We also draw attention to the maps in Exhibit MEX-65 that Mexico provided today. That exhibit shows that Mexican vessels are fishing just off the coast of Mexico using methods other than setting on dolphins to catch tuna, and the Mexican representative said today that 20 per cent of Mexico's catch is caught using methods other than setting on dolphins. Exhibit MEX-65 also appears to show that Mexican vessels are traveling just as far west to set on dolphins to catch tuna as it claims

it would have to travel south to fish for tuna not in association with dolphins. This evidence supports the US position that the US provisions and the conditions they establish for use of the dolphin-safe label are not singling out Mexican imports.

- 4.206 Mexico has also referenced the fact that there is dolphin by-catch and mortality in other fisheries. The United States does not deny that dolphins may sometimes interact with a fishery and dolphin mortality may occur outside the ETP. The United States is concerned about this, and as discussed these past days and in the US first written submission, the United States works with other regional fisheries management organization to monitor and address these concerns. But other fisheries where dolphin morality may occur do not involve the exploitation of a regular and significant association between the target species and dolphins, and there is no information indicating mortality levels on any similar order of magnitude to what has occurred in the ETP.
- 4.207 Finally, we would like to highlight a few of the key legal points discussed today.
- 4.208 First, the US provisions are not technical regulations. They establish conditions under which tuna may be labelled dolphin-safe, compliance with which is not mandatory.
- 4.209 Second, even if the US provisions were technical regulations, they fulfil a legitimate objective. Those objectives are ensuring consumers have accurate information about whether tuna products were caught in a manner that adversely affects dolphins, and ensuring that the US market is not used to encourage the setting on dolphins to catch tuna.
- 4.210 Third, the AIDCP definitions Mexico cites are not international standards. They were not adopted by a body within the meaning of that term under the TBT Agreement, and they were not adopted by a recognized body that is open to all WTO Members.
- 4.211 Fourth, as Mexico acknowledges, the US provisions are origin-neutral on their face, and Mexico has not shown that those provisions in fact single out imports for less favourable treatment.
- 4.212 Mr Chairman, this concludes my closing statement. We thank the Panel, members of the Secretariat, and the interpreters for their time and attention in these proceedings.
- D. SECOND WRITTEN SUBMISSIONS

1. Mexico

- (a) Introduction
- 4.213 This dispute concerns US measures that unilaterally impose conditions related to fishing methods in order for tuna products to be labelled "dolphin-safe" and thereby enable those products to access major distribution channels in the US market. These conditions conflict with and undermine a highly successful multilateral environmental agreement, the AIDCP, which already addresses concerns about dolphin mortality and injury. Notwithstanding Mexico's compliance with the AIDCP, the conditions imposed by the US measures prohibit the use of a dolphin-safe label on imports of Mexican tuna products while permitting the label to be used on like tuna products from other countries including the United States. As a consequence, Mexican tuna products that qualified as dolphin-safe under the AIDCP are not allowed to be labelled dolphin-safe in the US market and, therefore are not able to access the major distribution channels while tuna products of the United States and other countries are.
- 4.214 The US measures are intended to extraterritorially pressure the Mexican tuna fleet to change where it fishes for tuna and/or to change its fishing method to alternative methods that are recognized

to cause substantial environmental harm. As its core, the protection of dolphins is what the "dolphin-safe" label is all about. However, the US measures are not effective in providing additional protection for dolphins in the ETP and, in fact, provide no protection for dolphins outside the ETP where other fishing methods are used. Thus, the US consumers cannot be sure that tuna that is labelled dolphin-safe and sold in the United States is actually dolphin-safe.

- 4.215 The circumstances that created the need for the US measures have been remedied by the AIDCP and no longer exist. To the extent that the US arguments regarding other circumstances that necessitate the US measures have any merit, the United States should have addressed those circumstances multilaterally. Finally, there are less trade-restrictive alternative measures available to the United States that can achieve the objectives that it seeks.
- (b) Measures at issue
- (i) The measures must be assessed cumulatively
- 4.216 In Mexico's view the US measures identified in its panel request and its First Written Submission should be analysed as a whole. The application of all the measures prohibits in a discriminatory manner the use of a "dolphin-safe" label on imports of Mexican tuna products and modifies conditions of competition in the relevant market.
- (ii) Functioning of the US measures

Application inside of the ETP

4.217 Under the US measures, for tuna products produced from fish caught in the ETP by large vessels with purse seine nets there must be a certification that (i) the tuna was not caught using the dolphin set method at any time during a particular fishing trip, and (ii) no dolphins have been killed or seriously injured during the set in which the tuna were caught. This certification must be verified by an independent observer who was onboard the vessel during the voyage. The requirement that the tuna must not be caught using the dolphin set method automatically disqualifies Mexican tuna products from using a dolphin-safe label even though such tuna is dolphin-safe under the AIDCP.

Application outside of the ETP

- 4.218 The dolphin-safe standard that the United States applies to tuna harvested outside the ETP is that "no purse seine net was intentionally deployed on or to encircle dolphins during the fishing trip". Outside the ETP there is no requirement to certify that no dolphins were killed or seriously injured while fishing for tuna.
- 4.219 16 U.S.C. § 1385(d) separates tuna caught outside the ETP into the following categories: (i) caught on the high seas by a vessel engaged in driftnet fishing; (ii) caught using purse seine nets "in a fishery in which the Secretary has determined that a regular and significant association occurs between dolphins and tuna; (iii) caught using purse seine nets "in any other fishery", and (iv) caught without use of purse seine nets in a fishery "identified by the Secretary as having a regular and significant mortality or serious injury of dolphins". Although not expressly mentioned in the statute, for tuna caught without using purse seine nets in a fishery not identified as having a regular and significant mortality or serious injury of dolphins, no certification at all is required. Also, for vessels using purse seine nets that are smaller than 400 short tons no certification is required.
- 4.220 The key point is that, in the 20 years since the law was enacted, the United States has not designated any non-ETP fishery as having a regular and significant association between tuna and

dolphins, or regular and significant dolphin mortalities. Thus, with respect to fishing by vessels using purse seine nets, the only relevant category for non-ETP fisheries is the one for "any other fishery."

- 4.221 The United States asserts that the standard for alternative labels in section 1385(d)(3)(C) ("no dolphins killed or seriously injured") is being applied cumulatively with the standard for the official label in section 1385(d)(1)(B)(ii) ("no purse seine net was intentionally deployed on or used to encircle dolphins"). That assertion is contradicted by Exhibit US-59, Form 370, and the practices of the US producers of tuna products. In particular, neither Exhibit US-59 nor Form 370 describes a standard for tuna caught in "any other fishery" that includes the "no dolphins killed or seriously injured" requirement, and each of the three major US producers of tuna products define "dolphin-safe" as not setting on dolphins, with no mention of dolphins not being killed or seriously injured.
- (c) Facts
- (i) The US measures have the effect of excluding Mexican tuna products from the major distribution channels
- 4.222 It is clear that the US dolphin-safe labelling measures have altered the conditions of competition for Mexican tuna products. The US measures have created a market distinction for tuna products in the US market where the dolphin-safe label has an important commercial value. By virtue of the US measures which prohibit the label on Mexican tuna products, those products are excluded from the major distribution channels.
- 4.223 US imports have grown from \$259 million in 2000 to \$613 million in tuna products in 2009. Mexico's exports have remained minimal throughout this period. In 2009 US imports from Mexico were only \$7.5 million (1% of the US market for imported tuna products). This clearly shows an adverse impact of the US measures on Mexican tuna products.
- 4.224 Although the United States asserts that imports' share of the US market of tuna products has steadily increased since the US labelling provisions were enacted, in Mexico's view this assertion applies only with respect to tuna caught outside the ETP and/or using a fishing method other than the dolphin set method. For tuna caught in association with dolphins there has not been any significant growth or gains in market share.
- 4.225 The United States also asserts that Mexico has vessels under 363 metric tons and therefore tuna harvested by these vessels already meets the conditions to be labelled dolphin-safe under the US measures. Yet, those vessels amount to less than 4% of the total catch of the Mexican tuna fleet and these tuna is sold in the domestic market. With respect to the vessels above 363 metric tons that sometime uses other techniques than setting on dolphins; the tuna caught with these techniques cannot be labelled as dolphin-safe given that the US measures require that the dolphin set method is not used during an entire voyage.
- (ii) Tuna fishing and effects on dolphins
- 4.226 In explaining its justification for the measures at issue, the United States relies on two primary themes: first, that tuna fishing is resulting in the deaths of many thousands of dolphins more than those that are recorded by the independent observers onboard each large purse seine vessel in the ETP, and second, that even the approximately 1,000 deaths observed annually in the ETP are too many (and by implication should be reduced to zero).

US theory of unobserved mortalities

4.227 The entire premise of the US theory of unobserved mortalities is that the populations of two dolphin species that it has unilaterally determined as "depleted" are not growing at the rate that would be expected. To support this assertion, the United States continues to cite the old estimates as though they were conclusive facts, and inexplicably fails to mention that those estimates were superseded by a more recent report by the US Department of Commerce (2008), which not only had the benefit of more recent field research, but also found errors in the underlying data relied upon by earlier reports. The 2008 Report indicates that all three of the officially "depleted" dolphin stocks in the ETP were estimated to be growing at rates considered to be near the 4-8 per cent maximum possible for dolphins. This finding contradicts the United States assertion that setting upon dolphins in a manner consistent with the AIDCP is adversely affecting dolphin stocks through unobserved dolphin mortality.

Purported US goal of no mortality

- 4.228 The United States makes the following statement: "[M]exico may be considering 1,239 dolphin deaths as "virtually" none, but with 1,239 dolphin deaths in 2009 it is not accurate to say that dolphin mortality in the purse seine tuna fishery in the ETP has been virtually eliminated." The implication of this assertion is that anything more than zero or a few mortalities is unacceptable.
- 4.229 The impact of bycatch on populations is measured in relation to the size of the affected populations. When asked by the Panel if a bycatch of 1000 dolphins per year was significant, in terms of population recovery or conservation, the United States recognized that given the relative size of ETP dolphin populations, the death of 1,000 dolphins annually was not believed to be significant from a population recovery perspective.
- 4.230 Finally, it is important to note that, although the US goal in the ETP is no mortality at all, the United States tolerates the deaths of many marine mammals each year in fisheries that are under its own control. Mexico notes that there is no scientific basis for the United States to claim that the death of a marine mammal in a tuna fishery is more harmful than the death of a marine mammal in a monkfish or other non-tuna fishery, or that dolphin mortalities are irrelevant if the tuna is not destined for canning.

(iii) Unilateral action by the United States

4.231 The United States has adopted its own dolphin-safe labelling regime which diverges from and undermines the regime adopted and implemented under the AIDCP. For Mexico and other countries, the incentive to create an international binding agreement was to gain commercially meaningful access to the US A fundamental premise of the negotiation was that the United States would change its law to provide that tuna products harvested in the ETP in accordance with the AIDCP would be eligible for the US "dolphin-safe" label. Now, twelve years later, under current US legislation, complying with the requirements of the AICDP is still not sufficient to allow Mexico to use the "dolphin-safe" label. Mexico is not aware of any action of any kind by the United States to address this issue in the AIDCP, in an IATTC meeting, or in any other international regional fisheries organization.

(iv) Effects on other species

4.232 Mexico understands that the United States is claiming that fishing on FADs is the most ecologically responsible manner to harvest tuna. This is the first time that the United States has officially promoted such a position, which conflicts with the position of the United States when it adopted the IDCPA. Mexico observes that the Western and Central Pacific Fisheries Commission

implemented a ban on FAD fishing during August and September 2009, and intended to extend the ban to three months in future years. The US claim of support for increased FAD sets in the ETP cannot be reconciled with its actions in the ICCAT and its concurrence with the FAD limitations imposed by the WCPFC.

- (v) Association between tuna and dolphins outside the ETP
- 4.233 The United States asserts that there is no regular and significant association between dolphins and tuna outside the ETP and therefore that it is unnecessary for the United States to pay attention to whether dolphins are being killed or seriously injured outside the ETP.

No definition of "regular and significant"

4.234 The United States concedes that there is no definition of "regular and significant". If "regular and significant" means "no mortalities", many fisheries, including the Western and Central Pacific tuna fishery, should be designated. Thus, it is obvious that there is no scientific basis for the decision not to evaluate whether other fisheries have a "regular and significant association" with dolphins or "regular and significant" dolphin mortalities.

Other regional fisheries management organizations

4.235 The United States greatly overstates the monitoring activities of the non-ETP regional fisheries management organizations. Mexico has produced evidence that no regional fisheries management organization other than the IATTC has had a comprehensive program for many years to monitor bycatch of marine mammals.

Indian Ocean Tuna Commission ("IOTC")

4.236 The IOTC is still in the initial stages establishing programs to monitor bycatch. The October 2010 report of the IOTC Working Party on Ecosystems and Bycatch recognizes that the paucity of data makes any attempt to estimate levels of bycatch very difficult. Thus, it is obvious that the United States cannot justify relying on data from the IOTC as a basis for concluding that marine mammals are unharmed in the Indian Ocean tuna fishery.

International Commission for the Conservation of Atlantic Tunas ("ICCAT")

4.237 The ICCAT also has very limited information on bycatch. It publishes a webpage listing species that have been caught as bycatch, which includes 26 species of dolphins and whales. Of those, 15 are indicated to have been caught in purse seine nets. Thus, the United States cannot justify relying on data from the ICCAT as a basis for concluding that marine mammals are unharmed in the Atlantic tuna fishery – especially knowing that the fishery "interacts" with at least 26 species of marine mammals.

Western and Central Pacific Fisheries Commission ("WCPFC")

4.238 Almost all US purse seine vessels fish for tuna in the Western Central Pacific Ocean. The United States asserts that their independent observers witnessed 1,500 sets made by US vessels and saw few interactions with dolphins, but left unexplained whether the observers were specifically trained to watch for dolphin interactions, where the observed vessels were fishing, whether the vessels were using purse seine nets or longlines, and whether or not the dolphins were killed or seriously injured. To put the issue of bycatch in the WCPO in perspective, it is important to note that there were approximately 100,000 purse seine sets in that fishery in 2009, and there has been limited observer coverage.

- 4.239 The WCPFC Secretariat has published an evaluation of the impact of the WCPO fishery based on more extensive data. Several points are apparent from this analysis. First, no data have been publicly available on the overall interaction of this fishery with marine mammals. Second, observers witnessed dolphin and whale sets being made, indicating that there is an association between tuna and marine mammals in at least some parts of the WCPO. Third, the total number of purse seine sets observed over eleven years (33,319) is approximately one third of the total number of purse seine sets currently being made in the WCPO. As the 33,319 observed sets represent about one third of current total sets, the implication is that the WCPO purse seine fishery causes almost 2,200 marine mammal mortalities annually. Because the observers were not trained for dolphin observation, and apparently did not count serious injuries, it is likely that this figure underestimates the total.
- 4.240 Mexico is not aware that efforts have been made to analyse the impact this mortality might have on dolphin populations in the WCPO by conducting abundance surveys, evaluate whether dolphin populations in the WCPO are growing at expected rates, or study the indirect effects on the WCPO dolphin populations caused by separation of mothers from calves and stress to dolphins resulting from their interactions with WCPO fishing vessels.

Other regional fisheries – conclusions

- 4.241 Other regional fisheries management organizations are still in the early stages in monitoring the impact of fishing operations on marine mammals. Accordingly, the evidence of dolphin mortalities in these other fisheries presented by Mexico cannot be refuted by citing to inaction by those fisheries management organizations. Also, there is considerable other evidence available of significant marine mammal associations and mortalities in non-ETP fisheries, both for tuna and other target species. Thus, it is clear that protections for marine mammals are needed globally and that it is arbitrary for the United States to single out the ETP and Mexico's tuna products industry for special negative treatment when the AIDCP remains the world's only successful multilateral agreement for the protection of marine mammals, while there is substantial evidence of harm to marine mammals in other fisheries at least comparable to that in the ETP.
- (d) Legal argument
- (i) General

The need to address all of Mexico's claims

4.242 It is essential to the effective resolution of this dispute that the Panel avoids judicial economy and rules on all of the claims raised by Mexico under the GATT 1994 and the TBT Agreement because of: (i) the nature of the measures at issue; (ii) the fact that this is the first time that such measures have been subject to dispute settlement under the DSU; (iii) the differences in the wording and potential application of the provisions of the GATT 1994 and the TBT Agreement to the measures; and (iv) the importance of the effective discipline of such non-tariff measures to developing country Members such as Mexico.

Order in which the Panel should address Mexico's claims

4.243 It is logical for the Panel first to address Mexico's claims of discrimination under GATT Articles I:1 and III:4 and Article 2.1 of the TBT Agreement. An examination of Mexico's other claims under Articles 2.2 and 2.4 of the TBT Agreement should then follow.

The disciplining of unilateral actions by WTO Members

4.244 The US measures challenged in this dispute *de facto* condition access to the principal US distribution channels for tuna products on compliance with a fishing method that has been unilaterally imposed by the United States. These measures pressure the Mexican tuna fleet to use alternative fishing methods in the ETP that are very destructive to the marine ecosystem. The obligations in the WTO Agreements must not be interpreted so as to allow a WTO Member to condition access to its domestic market based on compliance with that Member's unilateral policy relating to actions outside its territory including unincorporated process and production methods. The only circumstances where such actions should be permitted are where they can be justified under one of the specific exceptions to the WTO obligations. None of those exceptions have been invoked by the United States nor do any of them apply to the US measures.

"zero tolerance"

- 4.245 The United States justifies its measure by saying, in essence, that a dramatic reduction of dolphin mortality to a number that is already statistically insignificant is not enough. Noting that the TBT Agreement gives WTO members the right to set their own standards, the United States defends its right to pursue a policy where no dolphin would die (zero tolerance). The argument by the United States raises an important systemic issue. Members of the WTO are free under both the TBT Agreement and the SPS Agreement to consider various risk factors as they determine where to set their standards for health and safety. But a standard of zero tolerance would be unworkable in the real world
- (ii) Mexico's discrimination claims under Articles I and III of the GATT 1994 and Article 2.1 of the TBT Agreement

Issues that are common to Mexico's discrimination claims

Like products

- 4.246 Both Mexico and the United States agree that the relevant products for the like product analysis are "tuna products". The sole issue that has been raised regarding the like products issue is whether, in the light of the US consumer preferences tuna products that are labelled dolphin-safe are "like" tuna products that are not labelled dolphin-safe. Mexico has explained why consumer preferences support Mexico's position that the products are like.
- 4.247 In addition, for a brief period in January 2003 when the United States changed the definition of dolphin-safe, a US distributor of Mexican tuna products was able to import Mexican tuna products with a dolphin-safe label to a large US grocery chain. Later, when the dolphin-safe label was prohibited for Mexican tuna products, the sales were no longer possible. The ability to sell Mexican tuna products with the dolphin-safe label confirms that Mexican AIDCP dolphin-safe tuna products and tuna products from the United States and other countries that are dolphin-safe under the US measures are "like" from the perspective of US consumers and retailers.

Timeframe for examining the facts related to Mexico's *de facto* discrimination claims

4.248 Mexico's discrimination claims concern *de facto* discrimination. The US measures violate Articles I:1, III:4 and 2.1 because their effect is to adversely modify the conditions of competition in the US marketplace for Mexican tuna products when compared to like tuna products from the United States and other countries. This denial of competitive opportunities is shown by the relevant facts as they existed at the time of the establishment of the Panel.

- 4.249 The United States and European Union in its third party submission argue that the facts as they existed when the US measures were first introduced in 1990 are relevant to Mexico's *de facto* discrimination claim. In Mexico's view the Panel must examine the facts as of the date of the Panel's establishment and determine whether there is *de facto* discrimination.
- 4.250 The unadopted GATT 1947 panel report in *US Tuna (Mexico)* illustrates the importance of the Panel examining the facts as they exist at the time of its establishment. As explained by Mexico, at the time of the GATT 1947 panel, the United States had a trade embargo in place against Mexican tuna. Thus, there was no evidence of the *de facto* discrimination and trade restrictive effects of the challenged measures. The *de facto* discriminatory and trade restrictive effects are clearly occurring today and it is the existence of those effects that gives rise to Mexico's request that the Panel rule upon its claims of *de facto* discrimination in this dispute.

The facially neutral US measures de facto discriminate

- 4.251 Mexico and the United States agree that the US measures are, on their face, origin neutral. They disagree, however, on whether the measures *de facto* discriminate. Basically, the United States argues that the US measures provide that tuna products of any origin that contains tuna that was caught by setting on dolphins may not be labelled dolphin-safe and that the use of the dolphin-safe label is equally prohibited for US tuna and tuna from other countries if it was caught by setting on dolphins. To support its arguments, the United States refers to the Panel Report in *EC Approval and Marketing of Biotech Products* and the Appellate Body Report in *Dominican Republic Import and Sale of Cigarettes*.
- 4.252 The basic argument of the United States is that a measure that is "origin neutral" on its face is by definition consistent with the national treatment obligation. The Appellate Body made it clear in *Korea Various Measures on Beef* that a measure can treat domestic and like imported products differently without violating the national treatment obligation. However, the Appellate Body did not state that a measure that does NOT treat domestic and imported like products differently is, for that reason, consistent with the national treatment obligation. That is precisely the effect of the supposedly "origin neutral" measures at issue in this dispute. Although presented in the context of national treatment, this principle also applies to Mexico's most-favoured-nation discrimination claims.
- 4.253 The EC Approval and Marketing of Biotech Products panel report to which the United States refers does not support its argument. In the Panel's view, Argentina did not provide sufficient evidence of the alleged de facto discrimination. In this dispute, the issue of insufficient evidence does not arise because Mexico has provided prima facie evidence of the de facto discriminatory effect of the US measures. The Dominican Republic Import and Sale of Cigarettes Appellate Body Report does not imply that a measure will necessarily be consistent with the national treatment obligation if it has a detrimental effect unrelated to the foreign origin of the product. That ruling must be understood in the context of the facts of that dispute. Those facts are readily distinguishable from the facts in this case. In this dispute the discriminatory effect is not between certain producers and importers but on the group of Mexican tuna products overall compared to the group of like tuna products from the United States and, in the case of Mexico's MFN claims, other countries. Also, the discrimination in this dispute does not depend on the characteristics of individual importers but, rather, on the fishing practices of the fleet that caught the tuna for exportation to the United States in the form of tuna products.
- 4.254 Rather, the situation in this dispute is analogous to the facts in one of the seminal GATT 1947 non-discrimination reports, *Belgium Family Allowances*. In that instance, the Panel found that a Belgian law that levied a charge on foreign goods that were purchased by Belgian public bodies and that originated in countries *whose system of family allowances did not meet specific requirements*, was

in violation of the non-discrimination obligations Articles I:1 and possibly III:2. There, as here, the discriminatory treatment related to conditions that the foreign country did not meet.

Mexico's discrimination claims are not dependent on different treatment for fisheries within and outside the ETP

4.255 Mexico raised the difference in regulation in the ETP and non-ETP fisheries to demonstrate that the US measures do not fulfil either the first objective stated by the United States or the objective that is reflected in the design, structure and character of the measures which is the protection of dolphins. While differences in treatment may further exacerbate the discrimination being faced by Mexican tuna products, the factual basis of Mexico's discrimination claims is that the *prohibition* against the use of the dolphin-safe label on most Mexican tuna products denies competitive opportunities to those products compared to like products from the United States and other countries.

US pressure on the Mexican fleet to change fishing areas and/or methods

- 4.256 The United States argues that Mexico overstates the cost and difficulty of using other techniques to catch tuna and in fact provides no evidence to support its claim that using other techniques would require Mexican vessels to "incur considerable financial and other costs". The European Union understands that there are no technical or legal barriers preventing any of the WTO Members bordering the ETP to fish tuna outside the ETP. Whether or not and to what magnitude costs would be incurred by the Mexican tuna fleet to change fishing areas or methods is immaterial to Mexico's *de facto* discrimination claims.
- 4.257 Notwithstanding that there is no evidentiary burden on Mexico to prove the costs associated with changing fishing areas and/or methods, Mexico has done so. These costs include: (i) catching juvenile tuna which is both less commercially viable and environmentally sustainable because of smaller catches and depletion of stocks; (ii) cutting production dramatically or move to another region to fish for different species of tuna, which is more commonly found in association with FADs to avoid destroying the yellowfin fishery close to Mexico; (iii) traveling longer distances; (iv) catching less economically valuable tuna (skipjack), and (v) obtaining licences which are expensive and, to date, unobtainable.

Private actions

- 4.258 The United States is incorrect in asserting that the "limited demand for non dolphin-safe tuna is a result of the retailers and consumer preferences for dolphin-safe tuna, not the US dolphin-safe labelling provisions". On this specific issue, the United States has misconstrued the ruling of the Appellate Body in *Korea Various Measures on Beef*.
- 4.259 The facts of this dispute are different from those in *Korea Various Measures on Beef;* however, the principles set out in this statement by the Appellate Body are applicable to this dispute. The US measures are restricting the conditions under which the "dolphin-safe" label may be used and, therefore, are restricting the nature of the choice that can be made by US consumers. As a direct consequence of the prohibition in the US measures, dolphin-safe labels other than those meeting the specific conditions of the US measures cannot be used. Thus, US consumers are denied the option of choosing Mexican tuna products that are labelled with the international AIDCP dolphin-safe label. In this way, it is the government intervention in the form of the US measures that adversely affects the conditions of competition.

The relevance of consumer preferences and perceptions

- 4.260 Consumer preferences and perceptions are relevant in this dispute to the extent that having a label that displays the words or a symbol signifying "dolphin-safe" is required to have access to the major distribution channels for tuna products in the US market. Mexico has presented evidence showing that a significant portion of the final consumers of tuna in the United States identify the term dolphin-safe with the notion that "no dolphins were injured or killed in the course of capturing tuna" (AIDCP standard) rather than with the idea that the tuna was caught in a manner that does not adversely affect dolphins (US standard).
- 4.261 Although the United States argues that consumers' preferences are identical in all segments of the market, in Mexico's view there are differences between the preferences of canneries retailers and final consumers. The US canneries do not buy tuna that is caught in the ETP using the dolphin set method due to the regulatory distinctions created by the US measures. In April, 1990 the US canneries announced their policies of not purchasing tuna caught in association with dolphins knowing of the imminent enactment of the DPCIA. The continuation of these policies is in the interests of the canneries. Their major tuna operations are outside the ETP. Therefore, under the US measures, they are able to apply the dolphin-safe label on their tuna products with the only requirement being that the captain of the vessel sign a certification that "no purse seine net was intentionally deployed on or used to encircle dolphins during the particular voyage on which the tuna was harvested."
- 4.262 With respect to the preferences of retailers and final consumers, both are identical in the sense that they look for dolphin-safe labelled tuna products. The retailers' interest is to sell tuna products which is why they require a dolphin-safe label. With respect to final consumers, they look for tuna products that are labelled as "dolphin-safe" which to them means that "no dolphins were killed or seriously injured during the capture of tuna".
- 4.263 During the discussions of the IDCPA, Senator Ted Stevens recognized that the new US legislation would guarantee US consumers that no dolphins were killed during the harvest of tuna that is labelled as dolphin-safe. This was because under the existing US law, dolphins may have been killed but as long as the tuna was not harvested by intentionally encircling dolphins it could be labelled as dolphin-safe.

Article I:1 of the GATT 1994

Advantage

- 4.264 The United States argues that the advantage, favour, or privilege at issue in this dispute is not merely the right to label tuna products as dolphin-safe and that no Member has the right to unconditionally label its products dolphin-safe under the US law. In the view of the United States, the advantage, favour, or privilege is the opportunity to use the dolphin-safe label if the conditions on use of the dolphin-safe label are met. These arguments conflates the concept of an "advantage, favour or privilege" with the conditions that are imposed on the granting of that advantage, favour or privilege. Mexico has established that the dolphin-safe label, whether meeting the US conditions or the international AIDCP standard, has commercial value in the US market in the sense that access to the label enables tuna products to be sold in the principal distribution channels of the US market. It is the application of a dolphin-safe label that has value, not the conditions attached to that label.
- 4.265 The United States also to the GATT 1947 panel report in *US Tuna (Mexico)* in support of its argument. As explained by Mexico, the facts that were before the GATT 1947 panel were very different from those before this panel. At that time, the nature of the "advantage" was not apparent. Today, the right to designate tuna products as dolphin-safe is an "advantage" because it provides access to the major distribution channels in the US tuna products market.

Shall be accorded immediately and unconditionally

4.266 In its third party submission and its responses to the question of the Panel, the European Union argues that "the requirement of 'unconditionally' in Article I:1 of the GATT requires the domestic measure to extend the advantage to all goods meeting the *conditions prescribed in the measure*" and that it "requires the grant of the advantage only to those products of all WTO Members that satisfy the same conditions". The European Union bases this interpretation on the report of the panel in *Canada – Autos*.

4.267 As explained by the panel in *Colombia – Ports of Entry*, the panel in *Canada – Autos* considered that the issue of whether an advantage within the meaning of Article I:1 is accorded "unconditionally" cannot be determined independently of whether it involves discrimination between like products of different countries. Whether conditions attached to an advantage granted in connection with the importation of a product offend Article I:1 depends on whether or not such conditions discriminate. In this dispute, the US measures make the advantage subject to conditions with respect to the situation or conduct of Mexico. These conditions discriminate against tuna products from Mexican in favour of tuna products from other countries. Accordingly, contrary to Article I:1, the United States has granted an advantage to tuna products of other WTO Members and has not accorded that advantage immediately and unconditionally to like tuna products of all other Members, namely Mexico.

Article III:4 of the GATT 1994

4.268 The United States argues that the fact that one-third of Mexico's purse seine fleet exclusively uses techniques other than setting on dolphins to catch tuna and therefore tuna caught by these vessels is eligible to be labelled dolphin-safe under the US provisions, is evidence that US provisions do not use origin to distinguish between tuna products that are eligible to be labelled dolphin-safe. Mexico has clarified these facts. Thus, even if the United States' interpretation of the applicable law and the test that it proposes is accepted, approximately 96 per cent of Mexican tuna could not be labelled as dolphin-safe when processed into tuna products and, therefore, could not access the principal distribution channels in the US market. A similar high percentage, if not 100 per cent, of US tuna products could be labelled as dolphin-safe and could access those channels. These ratios demonstrate that, even under the proposed US test, Mexican tuna products are singled out for less favourable treatment.

4.269 However, the United States' interpretation of the applicable law and the proposed test is incorrect. It is legally irrelevant that some Mexican tuna products or even tuna products from other WTO Members could be labelled as dolphin-safe and access the principal US distribution channels. Such access does not eliminate the *de facto* discrimination that has been caused by the upsetting of the balance of competitive opportunities between Mexican tuna products and like US products. A Member cannot balance more favourable treatment of imported products in some instances against less favourable treatment of imported products in other instances. For these reasons the *de minimis* access that Mexican tuna products may have to the US market is immaterial to Mexico's discrimination claim under Article III:4.

Article 2.1 of the TBT Agreement

4.270 Mexico has presented detailed arguments on its claim under Article 2.1 of the TBT Agreement. Mexico addresses all of the points raised in the United States' rebuttal arguments above.

(iii) Other TBT Agreement claims

The US measures are a technical regulation

- 4.271 The position of the United States that the US dolphin-safe provisions identified by Mexico in this dispute are "voluntary measures" and therefore do not meet the definition of a technical regulation under Annex 1.1 of the TBT Agreement is without merit. By virtue of the US measures, it is not possible to label tuna products as dolphin-safe under more than one standard. In this way, the US measures are mandatory and amount to a technical regulation within the meaning of Annex 1.1.
- 4.272 In simple terms, in order to label a tuna product in the US with a dolphin-safe label, it is necessary to comply with specific requirements provided in the US measures. These requirements are imposed in negative form (i.e. dolphin-safe tuna products offered for sale in the United States must not possess certain characteristics). What makes the US measures in this dispute mandatory is not whether a label is *de jure* required in order to sell tuna products in the US market. Rather, it is the fact that the US measures restrict retailers, consumers and producers to a single choice for labelling tuna products as dolphin-safe. There is no available option for US consumers to buy tuna products that have been produced from tuna caught in accordance with the international AIDCP standard for the protection of dolphins and is labelled as dolphin-safe under that standard.
- 4.273 The situation in this dispute is very similar to that addressed by the Appellate Body in EC Sardines, in which the challenged measure provided that only products prepared from Sardina pilchardus could be marketed as preserved sardines. In other words, only products of that species could have the word "sardines" as part of the name on the container. In this dispute, the US measures cover two different types of tuna based on the fishing method used to harvest them. Much like the European measures in EC Sardines, the US measures provide that the only products that can have the term dolphin-safe or a similar term or symbol are those that meet the US legal provisions.
- 4.274 It is also important to note in this regard that the US dolphin-safe labelling measure for the ETP is backed by an extensive conformity assessment procedure to ensure that ETP tuna and tuna products made from such tuna comply with the US requirements. The fact that there is such a conformity assessment mechanism reinforces Mexico's position that the measure is a technical regulation and not a standard.

Article 2.2

The US measures do not fulfil their stated objectives

- 4.275 The US asserts that the US measures have two objectives: (i) to ensure that consumers are not mislead or deceived about whether the tuna product contains tuna that was caught in a manner that adversely affects dolphins; and (ii) to contribute to the protection of dolphins by ensuring that the US market is not used to encourage fishing fleets to set upon dolphins. There are several reasons why the US measures do not fulfil these stated objectives.
- 4.276 First, the US measures are based on the *assumption* that the fishing method utilized by the Mexican fishing fleet and regulated by the AIDCP adversely affects dolphins, while there is no scientific evidence for that. As noted above, the most recent study of the Department of Commerce indicates that dolphin stocks are recovering. Second, tuna products that are labelled as dolphin-safe under the US measures might contain tuna that was caught in a set where dolphins were killed or seriously injured. Finally, the objective of encouraging foreign fishing fleets not to set upon dolphins applies only to fleets fishing in the ETP. The fishing method of those fleets is governed by the AIDCP and the US measures do not go any further than the AIDCP.

The fulfilment of the objectives using less trade restrictive measures

4.277 To the extent that the US measures fulfil any objectives, taking into account the risks of non-fulfilment, those objectives could be fulfilled using less trade restrictive measures. A less trade restrictive way of fulfilling the objectives would be to create dolphin-safe standards rather than a technical regulation whereby the AIDCP standard could be recognized and a label complying with the AIDCP standard used. In this way US consumers will be fully informed of all aspects of dolphin-safe fishing methods and they can choose accordingly when purchasing tuna products from US retailers.

Article 2.4

- 4.278 The US argument that the AIDCP organization is not a "body" within the meaning of Article 2.4 is without merit. National governments are "bodies," and the AIDCP is based on their membership. The AIDCP has its own administration, which is implemented by the Secretariat of the IATTC. The core purpose of the AIDCP organization is to establish standards for tuna fishing to protect marine mammals. There is no support for the US claim that the AIDCP organization is not a "recognized" or "standardizing" body for establishing when tuna caught in the ETP is dolphin-safe. The United States is a founding member of the AIDCP. It fully participated in the creation and establishment of the AIDCP's dolphin-safe standard. Indeed, the very purpose of that standard was to facilitate access to the US market.
- 4.279 Accordingly, the potential problems with applying Article 2.4 to standards in which a Member has not participated raised by the United States do not arise in this dispute. The issue raised in this dispute is whether it is inconsistent with WTO obligations for a Member to maintain a unilateral technical regulation that conflicts with an international standard to which the Member expressly agreed.

(e) Conclusions

4.280 On the basis of the foregoing, Mexico respectfully requests that the Panel find that the US measures are inconsistent with Articles I:1 and III:4 of the GATT 1994 and Articles 2.1, 2.2 and 2.4 of the TBT Agreement.

2. United States

(a) Introduction

- 4.281 The US dolphin-safe labelling provisions establish origin-neutral conditions under which tuna products may voluntarily be labelled dolphin-safe. These conditions ensure that when a dolphin-safe label appears on a tuna product that it does not contain tuna that was caught in a manner harmful to dolphins. It is well-documented, and virtually uncontested by Mexico, that setting on dolphins to catch tuna adversely affects dolphins. These well-documented adverse effects lie at the core of why the US provisions condition the labelling of tuna products "dolphin-safe" on such products not containing tuna caught by setting on dolphins.
- 4.282 The Panel should reject Mexico's claims that the US dolphin-safe labelling provisions are inconsistent with Articles I:1 or III:4 of the GATT 1994 and Articles 2.1, 2.2 or 2.4 of the TBT Agreement.

- (b) The US dolphin-safe labelling provisions are not inconsistent with Article III:4 of the GATT 1994
- 4.283 To establish its Article III:4 claim, Mexico must first establish that the US dolphin-safe labelling provisions accord different treatment to imported and domestic tuna products and that any such different treatment is based on origin. Then, if it establishes that the US provisions accord any different treatment to imported and domestic tuna products, it must establish that the treatment accorded imported tuna products is less favourable than the treatment accorded domestic tuna products. Mexico may establish that the US dolphin-safe labelling provisions accord different treatment to imported products based on origin either by demonstrating that the US provisions on their face accord such different treatment or by demonstrating that the US provisions while origin-neutral on their face in fact accord such different treatment. As elaborated below, Mexico has not established, either in law or in fact, that US dolphin-safe labelling provisions accord different treatment, let alone less favourable treatment, to imported tuna products.
- 4.284 First, Mexico has failed to show that the US provisions accord any different treatment to Mexican tuna products than the treatment accorded domestic products. Mexico's legal analysis skips the threshold issue of different treatment and jumps immediately to the issue of whether the treatment the US provisions accord alters the conditions of competition to the detriment of imported products. Second, in order to establish that a measure accords less favourable treatment within the meaning of Article III:4, it must be shown that any different treatment accorded to imported products is based on origin and that any different treatment is less favourable. Simply offering evidence that some imported products are accorded different treatment than some like domestic products is insufficient to support an Article III:4 claim.
- 4.285 In other disputes where a party has claimed that a facially origin neutral measure in fact discriminates based on origin, the complaining party has presented substantial evidence that what may appear to be origin-neutral criteria in fact single out imports for different treatment. In this dispute, Mexico has not adduced similar evidence to show that the US dolphin-safe labelling provisions although origin neutral on their face in fact use the manner or the place in which the tuna was caught to single out imports. In fact, as reviewed below, the evidence on the record leads to the opposite conclusion.
- 4.286 First, approximately 84 per cent of the US market for canned tuna products is accounted for by a combination of imported tuna products and domestic tuna products that contain imported tuna. Of the \$1.2 billion of US imports of tuna and tuna products, the vast majority contained tuna that was caught by methods other than setting on dolphins and are eligible to be labelled dolphin-safe. Second, one-third of Mexico's purse seine fleet exclusively uses techniques other than setting on dolphins to catch tuna and therefore tuna products that contain tuna caught by these vessels are eligible to be labelled dolphin-safe. The remaining two-thirds of Mexico's purse seine fleet also opportunistically uses techniques other than setting on dolphins to catch tuna. In fact during the first meeting with the Panel, Mexico acknowledged that 20 per cent of its fleet's catch is caught by techniques other than setting on dolphins. Tuna caught by those vessels using those techniques are also eligible to use the dolphin-safe label. It is therefore incorrect that Mexican vessels "almost exclusively set on dolphins" to catch tuna. It is also important to emphasize that Mexican vessels have a choice of whether to set on dolphins to catch tuna or to use other techniques to catch tuna. Mexico's large purse seine could also catch tuna using other techniques and do so in the ETP. Third, at the time the US dolphin-safe labelling provisions were enacted, there were 46 US purse seine vessels along with 52 Mexican vessels that fished for tuna in the ETP. Most of the 46 US purse seine vessels authorized to fish for tuna in the ETP that year set on dolphins to catch tuna and did not fully discontinue the practice until years later. Thus, at that time the US dolphin-safe labelling provisions were enacted, tuna products that contained tuna caught by US vessels were not eligible to be labelled dolphin-safe.

- 4.287 Mexico focuses on the fact that its fleet fishes for tuna in the ETP to argue that the US provisions discriminate against Mexican tuna products. This argument should be rejected. First, the United States imports significant amounts of tuna products that contain tuna caught in the ETP and are labelled dolphin-safe. For example, in 2009 the United States imported \$48 million worth of canned tuna products (i.e., tuna in airtight containers) from Ecuador that contained tuna caught using purse seine nets in the ETP. All of these imports were eligible to be labelled dolphin-safe. Second, tuna caught in the ETP cannot be equated with tuna of Mexican origin. The ETP is not a Mexican fishery, but is a geographic region that encompasses a fishery where Mexican vessels fish for tuna along with vessels from many other countries. Further, the origin of tuna is not determined by where it was caught but the flag of the vessel that caught it. Tuna caught in the ETP could be of Mexican origin or of an origin of any country that has vessels fishing for tuna in the ETP.
- 4.288 In addition to failing to establish that the US dolphin-safe labelling provisions use the conditions under which tuna products may be labelled dolphin-safe as a means to single out imports for different and less favourable treatment, Mexico has also failed to establish that the US provisions reflect any intent to afford protection to domestic production of tuna products. Article III:1 of the GATT 1994 provides relevant context for Article III:4 and sets forth that "the broad and fundamental purpose of Article III is to avoid protectionism in the application of internal tax and regulatory measures." Yet, when directly asked by the Panel for any evidence that the US provisions were introduced with the objective of disturbing competition between imported and non-imported tuna or affording protection to US tuna products, Mexico had no evidence to offer. Looking at the design, structure and characteristics of the provisions, also reveals that the US dolphin-safe labelling provisions do not reflect any intent to afford protection to domestic production or discriminate against Mexican tuna products. Further, unlike in *Chile Alcoholic Beverages* there is a clear relationship between the objectives of the US dolphin-safe labelling provisions and the conditions under which tuna products may be labelled dolphin-safe.
- 4.289 Mexico argues that the US dolphin-safe labelling provisions "impose more liberal conditions for use of the labelling standard in all fisheries other than the ETP because, in those fisheries, no certification is required that no dolphin[s] were killed or seriously injured and independent observers are not required." Mexico claims that "at least the same amount or more dolphins are being killed outside the ETP in alternative fishing operations" as inside the ETP as a result of fishing operations there. Mexico further claims that fisheries outside the ETP are "U.S. fisheries" while the ETP is a Mexican fishery and therefore that the different conditions that apply with respect to those fisheries supports its claim that the US provisions discriminate against Mexican tuna products. Mexico's arguments should be rejected.
- 4.290 First, to the extent there are any differences in documentation to substantiate dolphin-safe claims they are calibrated to the risk that dolphins will be killed or seriously injured when tuna is caught and are not evidence that the US dolphin-safe labelling provisions discriminate against Mexican tuna products. Second, the so-called available scientific evidence Mexico cites to support its assertion that the extent of dolphin mortality as a result of tuna fishing operations outside the ETP are the same or greater outside the ETP do not in fact support that assertion. Third, to the extent Mexico's claim relies on the different standards applied under the US dolphin-safe labelling provisions as compared to those under the Marine Mammal Protection Act, such arguments are inapposite. The issue before the Panel is whether the US dolphin-safe labelling provisions accord imported tuna products less favourable treatment. Comparing how fisheries are managed under the MMPA as compared to how tuna products may be labelled dolphin-safe under the US dolphin-safe labelling provisions does not shed light on that issue.
- 4.291 Mexico has also not shown that the US measures have modified the conditions of competition to the detriment of imported tuna products. Unlike in *Korea Beef*, the US dolphin-safe labelling provisions do not limit the marketing opportunities for imported tuna products. Imported tuna

products comprise a substantial share of the US market for tuna products, and the US provisions do not impose any choice on marketers of tuna products in terms of selling tuna products in the United States. The limited demand for non-dolphin-safe tuna products is a result of retailer and consumer preferences for dolphin-safe tuna products, not the US dolphin-safe labelling provisions.

- 4.292 Mexico offers several arguments that purport to show the dolphin-safe labelling provisions modify the conditions of competition to the detriment of Mexican tuna products. These should be rejected. Countries other than Mexico, including the United States, are similarly close to the ETP, including those areas of the ETP where setting on dolphins to catch tuna occurs. Evidence submitted by Mexico shows that Mexico would not have to travel long distances, change its target species, or significantly alter the duration of its trips in order to seize the opportunity to fish without setting on dolphins in its own backyard. Mexico has not substantiated its assertion that switching fishing techniques would involve "considerable financial and other costs," particularly in light of the fact that the same boats and fishing gear that is used to catch tuna by setting on dolphins can be used to catch tuna using other techniques. To the extent there are costs associated with adopting alternative techniques to catch tuna those would not be unique to Mexican vessels. Furthermore, producers often must shoulder additional costs in conjunction with compliance with a government measure, and such costs are not evidence that the US measures are consistent with Article III:4.
- 4.293 Mexico states that if its vessels were to expand its fishing for yellowfin tuna in a manner other than by setting on dolphins, they would catch juvenile tuna rather than the mature tuna it currently catches and exhaust the tuna stocks. However, this argument assumes that expanded fishing operations by these vessels would be done in contravention of fisheries management measures maintained under the auspices of the multilateral IATTC.
- 4.294 Mexico asserts that the US dolphin-safe labelling provisions benefit US producers because they "exclude[] Mexican brands from competing in the US market" and allow US canneries to "avoid having to ensure that tuna they purchase from non-ETP sources was captured without killing or seriously injuring dolphins." There is no basis for Mexico's assertion. First, the US provisions do not exclude Mexican brand tuna products from the US market. Second, US canneries use an alternative dolphin-safe label on their tuna products and are therefore subject to the condition in section 1385(d)(3) that tuna labelled dolphin-safe must not contain tuna caught in a set in which dolphins were killed or seriously injured. Third, there is no benefit to US canneries of avoiding having to ensure that the tuna they purchase is not caught in a set in which dolphins were killed or seriously injured. Fourth, US canneries supported the DPCIA because consumers were concerned about dolphins being harmed when tuna was caught and wanted assurances that tuna products did not contain tuna that was caught in a manner harmful to dolphins.
- 4.295 When the DPCIA was enacted, the United States had 46 US purse seine vessels that fished for tuna in the ETP of which 31 were doing so full-time. Both Mexican and US vessels fished in the ETP by setting on dolphins at the time. Therefore, to the extent that the conditions of competition were altered by the US dolphin-safe labelling provisions, they were not changed to the detriment of imports.
- 4.296 In sum, the US dolphin-safe labelling provisions accord no less favourable treatment to Mexican tuna or tuna products than that accorded to tuna and tuna products of the United States. Therefore, the US provisions are not inconsistent with Article III:4 of the GATT 1994.
- (c) The US dolphin-safe labelling provisions are not inconsistent with Article I:1 of the GATT 1994
- 4.297 Mexico has also failed to establish that the US dolphin-safe labelling provisions are inconsistent with Article I:1 of the GATT 1994. Mexico both wrongly identifies the advantage at

issue in this dispute and fails to establish that the US dolphin-safe labelling provisions accord an advantage to imported tuna products of other countries that they fail to accord to imported tuna products of Mexico.

- 4.298 The issue of whether the US provisions fail to accord such an advantage was heard and decided two decades ago when a 1991 panel under the GATT 1947 rejected Mexico's claims that the US dolphin-safe labelling provisions were inconsistent with Article I:1 of the GATT 1994. The legal and factual conclusions of the panel in *US Tuna Dolphin I* were well reasoned and sound, and nothing in the intervening time has changed to support a different conclusion.
- 4.299 Mexico appears to believe that the advantage Mexican products are being denied is the right to carry the dolphin-safe label. That is incorrect. No product (whether of the United States or any other Member) is entitled unconditionally to be labelled dolphin-safe under US law. Rather, the advantage at issue in this dispute is the opportunity under US law to label tuna dolphin-safe if certain conditions are met.
- 4.300 Mexico also appears to believe that a measure may be found inconsistent with Article I:1 of the GATT 1994 simply by virtue of the fact that imported products from some countries qualify for an advantage while others do not. This is not a correct reading of Article I:1 of the GATT 1994. As prior reports have expressed, whether conditions attached to an advantage granted by a measure are inconsistent with Article I:1 depends upon whether or not such conditions discriminate with respect to the origin of products. Conditions that are origin-neutral are not inconsistent with the obligation in Article I:1 that any advantage granted to imported products originating in any Member shall immediately and unconditionally be granted to like products originating in any other Member.
- 4.301 Mexico argues the US dolphin-safe labelling provisions, while origin neutral on their face, in practice discriminate against Mexican tuna products as compared to imports from other countries. Yet, Mexico has not put forth evidence sufficient to substantiate its claim. In particular, Mexico has not established that the conditions the US provisions establish for labelling tuna products dolphin-safe which Mexico acknowledges are origin neutral on their face in fact act as a proxy to single out imports from some countries over others as eligible to be labelled dolphin-safe. To the contrary, the information on record in this dispute demonstrates that the US provisions do not fail to accord an advantage to Mexican tuna products that they accord to imported tuna products originating in other countries.
- 4.302 As reviewed in connection with Mexico's claims under Article III:4 of the GATT 1994, one-third of Mexico's purse seine fleet exclusively uses techniques other than setting on dolphins to catch tuna and as Mexico acknowledged during the first meeting with the Panel, approximately 20 per cent of Mexican tuna catch is caught by techniques other than setting on dolphins. Additionally, the technique of setting on dolphins to catch tuna is not unique to the Mexican fishing fleet. The fishing fleets of Colombia, El Salvador, Guatemala, Nicaragua, Panama, and Venezuela also have vessels that set on dolphins, among other techniques, to catch tuna in the ETP. Setting on dolphins was a technique used by US vessels at the time the US provisions were adopted. These facts are evidence that US provisions do not use the manner in which the tuna was caught as a proxy to distinguish between tuna products that are eligible to be labelled dolphin-safe and those that are not based on origin. It is the manner in which the tuna was caught not origin that determines whether tuna products containing it may be labelled dolphin-safe.
- 4.303 Further, nothing prevents Mexico's fleet from expanding its use of techniques other than setting on dolphins to catch tuna, including on account of the costs as noted above. To the extent Mexico relies on the same arguments it made in the Article III:4 context to prove that the US labelling provisions discriminate against Mexican, because they call for different documentation to substantiate dolphin-safe claims, those arguments are without merit for the same reasons as discussed above. As in

the case with Mexico's argument under Article III:4, Mexico's argument that the US provisions discriminate against Mexican tuna products because the Mexican fleets primarily fish for tuna in the ETP should likewise be rejected in the context of Mexico's Article I:1 claim.

- 4.304 In this case, unlike in *Canada Autos*, limiting the use of the dolphin-safe label to tuna products that do not contain tuna that was caught by setting on dolphins or in a set in which dolphins were killed or seriously injured does not have the effect of limiting eligibility to use the dolphin-safe label to imports originating in only certain countries. It is not the origin of the product, but whether that product was caught in a manner that adversely affected dolphins that determines eligibility to use the dolphin-safe label. In contrast to the situation in *Canada Autos*, Mexican fishing vessels can choose to meet the conditions that would make products containing their tuna eligible for the dolphin-safe label.
- 4.305 In sum, the US measures do not accord an advantage, favour, or privilege to tuna or tuna products originating in any other country that is not also accorded to Mexico. Therefore, the US provisions are not inconsistent with Article I:1 of the GATT 1994.
- (d) The US dolphin-safe labelling provisions are not technical regulations
- 4.306 Mexico has failed to establish that the US dolphin-safe labelling provisions constitute technical regulations within the meaning of Annex 1 of the TBT Agreement. In particular, Mexico has failed to establish that compliance with the US dolphin-safe labelling provisions is mandatory. A labelling requirement with which compliance is mandatory is a measure that establishes conditions under which a product may be labelled in a certain way *and* requires the product to be labelled in that way in order to be marketed.
- 4.307 The principal flaw in Mexico's interpretation of the definition of technical regulation in Annex 1 of the TBT Agreement is that it conflates the meaning of the term "labelling requirement" with the phrase "with which compliance is mandatory." In doing so, Mexico would render the phrase "with which compliance is mandatory" and the phrase "with which compliance is not mandatory" in the definition of a technical regulation and the definition of a standard, respectively, without effect. Mexico's approach is inconsistent with the fundamental rule of treaty interpretation that an interpretation of the terms of a treaty is to be preferred that gives full effect and meaning to each of its terms.
- 4.308 Applying the correct interpretation of a technical regulation to the facts of this dispute reveals that compliance with the US dolphin-safe labelling provisions is not mandatory within the meaning of Annex 1 of the TBT Agreement. While the US dolphin-safe labelling provisions set out conditions under which tuna products may be labelled dolphin-safe, they do not require tuna products to be labelled dolphin-safe to be marketed. In fact, tuna products that are not labelled dolphin-safe are readily available on the US market.
- 4.309 Mexico advances that one way to distinguish between a labelling requirement that is voluntary and one that is mandatory is whether the label contemplated in the labelling requirement is the only label that may be used in the market. There is no basis for Mexico's theory. First, it is not based on the text of the TBT Agreement. Second, Mexico's theory conflates the meaning of the term a "labelling requirement" with the meaning of the phrase "with which compliance is mandatory" and renders the latter without effect. Mexico also argues that "the fact [the U.S.] measures established surveillance and enforcement procedures" is an additional point supporting its position that the US dolphin-safe labelling provisions are mandatory. The information collected by the United States and the surveillance activities it undertakes to ensure that tuna products labelled dolphin-safe are in fact dolphin-safe are simply mechanisms that support the underlying labelling requirement established in the US dolphin-safe labelling provisions. They do not change the fact that the US provisions do not

require tuna products to be labelled in a certain way to be marketed in the United States and therefore compliance with the labelling requirements set out in the US provisions is not mandatory within the meaning of Annex 1 of the TBT Agreement.

- 4.310 The US dolphin-safe labelling provisions are not like the measures in EC Asbestos or EC Sardines, which concerned measures that fell within the scope of the first sentence of the definition of a technical regulation; neither concerned labelling requirements.
- 4.311 Mexico's alternative argument that the US provisions are *de facto* mandatory should be rejected. Contrary to Mexico's assertions, a labelling requirement cannot be "*de facto*" mandatory simply based on private actors' preference for products labelled in a certain way. Some form of government action must make it compulsory or obligatory that for products to be marketed they must be labelled in a certain way in order for compliance with a labelling requirement to be mandatory. In this dispute, Mexico identifies no government action that makes compliance with the US dolphin-safe labelling provisions mandatory.
- 4.312 For the forgoing reasons, Mexico has failed to establish that the US dolphin-safe labelling provisions are technical regulations within the meaning of Annex 1 of the TBT Agreement and accordingly has failed to establish that the US dolphin-safe labelling provisions are subject to Article 2 of the TBT Agreement. As a consequence, the US dolphin-safe labelling provisions cannot be found inconsistent with Article 2.1, 2.2 or 2.4 of the TBT Agreement, and the Panel should therefore reject Mexico's claims under those articles.
- (e) The US dolphin-safe labelling provisions are not inconsistent with Articles 2.1, 2.2 or 2.4 of the TBT Agreement
- 4.313 Despite textual and contextual differences between Articles I:1 and III:4 of the GATT 1994 and Article 2.2 of the TBT Agreement, Mexico relies solely on the arguments it makes regarding the consistency of the US dolphin-safe labelling provisions with Articles I:1 and III:4 of the GATT 1994 for its arguments under TBT Article 2.1. The United States has articulated why Mexico's arguments under Articles I:1 and III:4 of the GATT 1994 fail, and Mexico's arguments under TBT Article 2.1 fail for the same reasons.
- 4.314 To establish a breach of Article 2.2 of the TBT Agreement, a complaining party must establish that the measure at issue is "more trade-restrictive than necessary to fulfil a legitimate objective". A measure is "more trade-restrictive than necessary to fulfil a legitimate objective" if (1) there is a reasonably available alternative measure (2) that measure fulfils the objectives of the measure at the level that the Member imposing the measure has determined is appropriate and (3) is significantly less trade-restrictive. Mexico has not established any of these elements with respect to the US dolphin-safe labelling provisions.
- 4.315 It would not be appropriate to apply the same interpretive approach panels and the Appellate Body have undertaken in connection with the word "necessary" as it appears in Article XX of the GATT 1994 in analysing whether a measure is "more trade restrictive than necessary" within the meaning of Article 2.2 of the TBT Agreement.
- 4.316 Applying the proper interpretation of Article 2.2 of the TBT Agreement to the facts of this dispute, it is clear the US dolphin-safe labelling provisions are no more trade-restrictive than necessary to fulfil their legitimate objectives. The US dolphin-safe labelling provisions fulfil the objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins by establishing conditions under which tuna products may be labelled dolphin-safe that are based on whether the tuna product contains tuna that was caught in a manner that adversely affects dolphins. The US provisions also fulfil the

objective of protecting dolphins by ensuring that the US market is not used to encourage setting on dolphins to catch tuna. As the practice of setting on dolphins to catch tuna decreases, the associated adverse effects on dolphins decrease as well.

- 4.317 With respect to the situation Mexico describes where a tuna product might contain tuna caught in a set in which a dolphin was killed yet still be labelled dolphin-safe, this does not support Mexico's contention that the US dolphin-safe labelling provisions fail to full their objective of ensuring consumers are not misled or deceived about whether the product contains tuna that was caught in a manner that adversely affects dolphins. First, it is a hypothetical situation. Second, Mexico has presented no evidence that such a hypothetical actually exists. In fact, the likelihood of any such products being on the US market is low.
- 4.318 Rather than demonstrating that the US dolphin-safe labelling provisions do not fulfil their objectives of ensuring consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins, the documentation to support dolphin-safe claims reflect that the US provisions took a balanced approach that weighed the risk of products containing tuna caught in a manner that adversely affects dolphins against the burden of ensuring that by conditioning use of dolphin-safe labelling on an observer's statement that no dolphins were killed or seriously injured. Such an approach that weighs costs and benefits is consistent with well-established approaches to the introduction of government measures. It is also consistent with the TBT Agreement.
- 4.319 Mexico has failed to establish that there is a reasonably available alternative measure that fulfils the provisions' objectives that is significantly less trade-restrictive. Neither the AIDCP nor the AIDCP resolutions are alternatives that would fulfil the objectives of the US dolphin-safe labelling provisions. The AIDCP is an agreement that seeks to reduce observed dolphin mortalities and serious injuries when dolphins are set upon to catch tuna. Thus, the AIDCP is not a substitute for provisions that seek to protect dolphins by discouraging the practice of setting on them to catch tuna. In addition, application of the procedures called for under the AIDCP would not fulfil the objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins, since those procedures do not address the labelling of tuna products or dolphin-safe claims on tuna products.
- 4.320 Use of the definition of "dolphin-safe" referred to in the AIDCP resolutions would also not fulfil the objective of the US dolphin-safe labelling provisions. First, that definition if used as a basis for the conditions under which tuna products may be labelled dolphin-safe would not ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins because it would allow tuna products to be labelled dolphin-safe if dolphins were set upon to catch the tuna. Second, these definitions if used as a basis for the conditions under which tuna products may be labelled dolphin-safe would not contribute to the protection of dolphins by ensuring that the US market is not used to encourage the practice of setting on dolphins to catch tuna, since they would permit tuna products to be labelled dolphin-safe that were caught by setting on dolphins.
- 4.321 The Panel should also reject Mexico's claims that US dolphin-safe labelling provisions are inconsistent with Article 2.2 of the TBT Agreement, because Mexico has not demonstrated that the US dolphin-safe labelling provisions restrict trade, much less that they restrict trade more than necessary within the meaning of Article 2.2 of the TBT Agreement.
- 4.322 Mexico claims that the US dolphin-safe labelling provisions are inconsistent with Article 2.4 of the TBT Agreement because they are not based on relevant international standards. However, to substantiate an Article 2.4 claim, the complaining party must establish that (1) there is a relevant international standard; (2) that standard would not be an ineffective or inappropriate in fulfilling the

legitimate objective of the measure; and (3) the measure at issue is not based on that standard. Mexico has not established, however, that the AIDCP resolutions Mexico identifies – resolution on tuna tracking and dolphin-safe certification – are relevant international standards within the meaning of Article 2.4 of the TBT Agreement or that they would not be ineffective and inappropriate to fulfil the objectives of the US dolphin-safe labelling provisions.

- 4.323 First, Mexico has not made any showing that the AIDCP resolutions meet the definition of a standard set out in Annex 1 of the TBT Agreement. In particular, Mexico has not established that the AIDCP resolutions provide for rules, guidelines or characteristics for products or related process and production methods or for an aspect covered under the second sentence such as a labelling requirement, or that the definition of dolphin-safe referred to in the AIDCP resolutions is for "common and repeated use." It has also not established that the AIDCP resolutions were adopted by a recognized body. The definition of dolphin-safe in the AIDCP resolutions a definition for purposes of those resolutions. Neither resolution purports to establish a definition of "dolphin-safe" for general application outside the context of the AIDCP resolutions. To construe the definition of "dolphin-safe" in the AIDCP resolutions as standards for "common and repeated use" would vastly expand the scope of the term standard in the TBT Agreement and have serious implications with respect to Members' rights and obligations under any intergovernmental resolution or agreement. The AIDCP resolutions were adopted by the parties to the AIDCP, and Mexico has not established that the parties to the AIDCP constitute a body as that term is defined for purposes of the TBT Agreement nor that the parties to the AIDCP could be considered to have recognized activities in standardization.
- 4.324 The AIDCP resolutions are also not "international" within the meaning of Article 2.4 of the TBT Agreement. In particular, the AIDCP resolutions were not adopted by a body whose membership is open to the relevant bodies of at least all Members and therefore do not qualify as "international" for purposes of Article 2.4 of the TBT Agreement. Further, the definition of "dolphin-safe" in the AIDCP resolutions are not "relevant." For example, that definition does not relate or pertain to the objectives of the US dolphin-safe labelling provisions aimed at ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins and ensuring the US market is not used to encourage the practice of setting on dolphins to catch tuna and thereby contribute to dolphin protection.
- 4.325 Mexico has also not put forth any evidence or argument that the definition of "dolphin-safe" in the AIDCP resolutions would not be ineffective or inappropriate to fulfil the objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins. For this reason alone, Mexico has failed to establish a prima facie case under Article 2.4 of the TBT Agreement. With respect to the objective of the US dolphin-safe labelling provisions of ensuring that the US market is not used to encourage the practice of setting on dolphins to catch tuna and thereby contributing to the protection of dolphins, Mexico has also not established that the definition of dolphin-safe in the AIDCP resolutions would not be ineffective or inappropriate in fulfilling that objective. If the US dolphin-safe labelling provisions were based on the definition of dolphin-safe in the resolutions then tuna products that contain tuna caught by setting on dolphins could be labelled dolphin-safe, and consumers would no longer know whether tuna products labelled "dolphin-safe" contain tuna that was caught by setting on dolphins and their purchases of tuna products that contain tuna caught by setting on dolphins could serve to encourage a practice that adversely affects dolphins. Basing the US provisions on the definition of "dolphin-safe" in the AIDCP resolutions would therefore not be effective or appropriate in fulfilling the objective of the US provisions of contributing to dolphin protection.

(f) Conclusion

- 4.326 For the reasons stated above, the panel should reject Mexico's claims that the US dolphin-safe labelling provisions are inconsistent with Articles I:1 and III:4 of the GATT 1994 and Articles 2.1, 2.2 and 2.4 of the TBT Agreement.
- E. OPENING ORAL STATEMENTS AT THE SECOND SUBSTANTIVE MEETING

1. Mexico

- (a) Introduction
- 4.327 This dispute concerns US measures that unilaterally impose conditions related to fishing methods in order for tuna products to be labelled "dolphin-safe" and thereby enable those products to access major distribution channels in the US market.
- 4.328 Despite everything that has happened in the last 20 years, and all the lessons learned in the Eastern Tropical Pacific (ETP), the US continues to pursue unilateral policies put in place in 1990. The Agreement on the International Dolphin Conservation Program (AIDCP) has been highly successful at reducing dolphin mortalities to levels that scientists describe as statistically insignificant. In addition to protecting dolphins, the AIDCP also protects tuna stocks and the overall ecosystem of the ETP by, *inter alia*, reducing bycatch associated with alternative fishing methods including FAD fishing. Notwithstanding Mexico's compliance with the AIDCP, the US measures prohibit the use of a dolphin-safe label on 96% of Mexican tuna production while permitting the label to be used on like tuna products from other countries including the US resulting in 96% of Mexican tuna products being effectively denied access to the US market.
- 4.329 The US measures are intended to unilaterally and extraterritorially pressure the Mexican tuna fleet to change where it fishes for tuna and/or to change its fishing method to environmentally unsustainable alternative methods. This undermines the environmental objectives of the AIDCP and, if the US changes were implemented, would threaten the economic viability of the Mexican fishing fleet which is an important part of Mexico's economy as a developing country member.
- 4.330 Mexico has presented a *prima facie* case with respect to all of its claims and requests that the Panel rule on each of Mexico's claims and avoid the exercise of judicial economy.
- (b) The US measures are inconsistent with Article I:1 of the GATT 1994
- 4.331 Tuna products from virtually all other countries, including the two largest exporters Thailand and the Philippines, are allowed to label their products dolphin-safe; by contrast, Mexican tuna products are not. Due to the US measures, current imports of Mexican tuna products into the US account for less than one per cent of all US imports of tuna products. The "advantage" in this dispute is the right to designate tuna products as dolphin-safe. As set out by the Panel in *Canada Autos*, the obligation to accord "unconditionally" to third countries which are WTO Members an advantage which has been granted to any other country means that the extension of that advantage may not be made subject to conditions with respect to the situation or conduct of those countries. The US measures make the advantage subject to conditions with respect to the situation or conduct of Mexico (i.e., fishing methods for tuna) and thereby discriminate against tuna products from Mexico in favour of like tuna products from other countries.
- 4.332 The facts in this dispute are analogous to those in *Belgium Family Allowances*, in which the panel condemned a measure which discriminated against imports depending on the type of family allowance system that was in place in the exporting country. Similarly, in this dispute the

US measures discriminate against imports depending on the type of fishing method employed by the exporting country.

- (c) The US measures are inconsistent with Article III:4 of the GATT 1994
- 4.333 The US arguments concerning Article III:4 improperly attempt to transform the discrete facts of particular cases of less favourable treatment into supposed general requirements for all cases of less favourable treatment. Moreover, the arguments regarding less favourable treatment raise profound systematic implications for the WTO trading system. If embraced, these arguments would empty the concept of *de facto* discrimination of all meaning and undermine the national treatment principle that is one of the fundamental cornerstones of the multilateral trading system.
- (i) The US measures have modified the conditions of competition for Mexican tuna products
- 4.334 The US measures have severely restricted the ability of Mexican tuna products to be sold in the US. The principal US distribution channels will not carry Mexican tuna products because they are prohibited by the US measures from being labelled as dolphin-safe. Thus, the US measures modify the conditions of competition to the detriment of imported Mexican tuna products. 96% of Mexico's tuna production is, by virtue of these measures, denied access to the principal distribution channels in the US market.
- (ii) There is no legal basis for the two-part test for proving less favourable treatment under GATT Article III:4 that is proposed by the US
- 4.335 There is no legal basis for the proposed US test that Mexico must establish that the US dolphin-safe labelling provisions accord different treatment to imported and domestic tuna products and that any such different treatment is "based on origin". After fulfilling this first step, Mexico must then establish that the treatment accorded to imported tuna products is less favourable than the treatment accorded to domestic tuna products. The only authority that the US cites is the Appellate Body report in *Korea Various Measures on Beef* and that report does not support the proposed test.
- (iii) It is not necessary for Mexico to prove that the differential treatment that is accorded to imported products is "based on origin"
- 4.336 The US argues that it is necessary for Mexico to prove that differential treatment is "based on origin" in order for there to be less favourable treatment under Article III:4. The US misreads EC Approval and Marketing of Biotech Products and Dominican Republic Import and Sale of Cigarettes and there is no legal basis for its assertion.
- (iv) It is not necessary for Mexico to prove the US measures "single out imports for different treatment"
- 4.337 The US argues that it is necessary for Mexico to demonstrate that what may appear to be origin-neutral criteria in fact single out imports for different treatment. The reports cited by the US (Mexico Taxes on Soft Drinks, Chile Alcoholic Beverages, and Korea Alcoholic Beverages) do not stand for the proposition that imports must always be singled out for there to be less favourable treatment. While singling out imports is one way to accord less favourable treatment it is not the only way. The issue is not whether a measure singles out imports but rather whether a measure modifies the conditions of competition in the relevant market to the detriment of imported products.

- (v) It is not necessary for Mexico to prove that "almost all" imports are disadvantaged
- 4.338 The US argues that Mexico must prove that "almost all" imports receive the less favourable treatment and almost all like domestic products do not. While *de facto* discrimination may be obvious in circumstances where almost all imports receive less favourable treatment and almost all like domestic products do not, there is no legal requirement for Mexico to prove discrimination to this extent.
- 4.339 From the perspective of imports, the "almost all" standard posed by the US is inconsistent with the established principle that not all imports must receive less favourable treatment in order for there to be discrimination: (i) a measure does not have to give rise to less favourable treatment for like imported products in each and every case in order to accord treatment no less favourable; (ii) the possible absence of less favourable treatment of like imported product in some instances does not detract from the fact that there might be less favourable treatment in other instances; and (iii) a Member cannot balance more favourable treatment of imported products in some instances against less favourable treatment of imported products in other instances. From the perspective of like domestic products it is equally clear that not all like domestic products have to be free of the less favourable treatment in order for there to be discrimination.
- (vi) The GATT 1947 Panel Report in Belgian Family Allowances supports Mexico's discrimination claims
- 4.340 Citing the panel report in *Indonesia Autos* as authority, the US argues that *Belgian Family Allowances* stands for the proposition that if conferral of an advantage within the meaning of Article I:1 of the GATT is made conditional on any criteria then those criteria must be related to the imported product itself and, on this basis, argues that it does not support Mexico's claims. Mexico disagrees. *Belgian Family Allowances* stands for the proposition that a measure cannot condition an advantage (e.g., access to the principal distribution channels for a product) upon certain actions by the exporting WTO Member. The panel condemned a measure which discriminated against imports depending on the type of family allowance system that was in place in the exporting country. In this dispute, the US measures discriminate against imports depending on the type of fishing method employed by the exporting country. Thus, the facts of the two cases are analogous. Both relate to actions undertaken by the exporting country.
- (vii) The US measures afford protection to domestic production
- 4.341 Contrary to the argument of the US, Mexico does not have to establish that the US provisions reflect any intent to afford protection to domestic production of tuna products. The question of whether a measure affords protection to domestic production is not a question of protectionist intent but, rather, one of application of the measure. For purposes of assessing whether there is less favourable treatment, the legal issue is what the effect of a measure is in the market, not why it is enacted. Under Article III:4, the Appellate Body has specifically ruled that if there is less favourable treatment of the group of like imported products, there is, conversely, protection of the group of like domestic products.
- (d) The US measures are inconsistent with the TBT Agreement
- (i) The US measures are technical regulations
- 4.342 The US argues that the principal flaw in Mexico's interpretation of the meaning of "technical regulation" is that it conflates the meaning of the term "labelling requirement" with the phrase "with which compliance is mandatory". This argument is completely without merit. The US misses the key point of Mexico's argument. Simply stated, it is the limitation of "dolphin-safe" to a single standard

that makes the US measures a technical regulation. The US measures can be contrasted with a situation where there are two *standards* for dolphin-safe—the US standard and the international AIDCP standard—and either can be used provided that the requirements of the standard being used are met and the designation of dolphin-safe that is being used is identified with the specific standard.

- 4.343 The US takes the narrow position that to be a technical regulation, a measure must require that the product be labelled in a certain way in order to be marketed or sold in the domestic market. Mexico agrees that this is one example of a technical regulation. However, it is not the only example. The Appellate Body reports in EC Asbestos and EC Sardines make this clear. The situation in this case is very similar to that addressed in EC Sardines. In that case, sardines could not be marketed as "preserved sardines" in the EU unless they comprised certain species of fish; in this case tuna products cannot be marketed in the US as "dolphin-safe" unless the dolphin set fishing method was not used to catch tuna in the ETP. In both EC Sardines and in this dispute, the products at issue—sardines and tuna products—could be sold in the domestic markets without the designations of "sardines" or "dolphin-safe" although the market opportunities were limited, particularly in the circumstances of this dispute.
- 4.344 Finally, if the US measures were not considered *a priori* mandatory, they have a mandatory nature in a *de facto* manner. As Mexico has explained in detail, having a dolphin-safe label is essential in order to market tuna products in the principal distribution channels in the US market.
- (ii) The US measures are inconsistent with Article 2.1 of the TBT Agreement
- 4.345 The US measures modify the conditions of competition in the relevant market to the detriment of imported Mexican tuna products in favour of like tuna products from the US and other countries. This is the basis for Mexico's claims that Article 2.1 has been violated.
- (iii) The US measures are inconsistent with Article 2.2 of the TBT Agreement

The meaning of "necessary"

4.346 The GATT and GATS jurisprudence illuminates the ordinary meaning of "necessary" in three important ways: (i) with regard to the continuum of necessity, (ii) with regard to reasonableness; and (iii) with regard to the importance of the interests and values at stake. All three of these elaborations apply to the meaning of the same term in Article 2.2. Contrary to the argument of the US, the fact that the term "necessary" in Articles XX and XIV is used in the context of whether it is necessary to breach a provision of the GATT 1994 or the GATS rather than whether it is "more trade-restrictive than necessary" in Article 2.2 does not render these interpretations inapplicable. Although the subject matter of what is necessary differs between these provisions, the concept of necessity is the same. This follows unavoidably from the appropriate use of the customary rules of treaty interpretation, as mandated by Article 3.2 of the Dispute Settlement Understanding.

The US erroneously relies on Article 5.6 and footnote 3 of the SPS Agreement

4.347 When attempting to interpret the term "necessary," the US erroneously relies on Article 5.6 of the SPS Agreement. The language of Article 5.6 and its associated footnote differs significantly from that of Article 2.2 of the TBT Agreement and, therefore, it is inappropriate to read into Article 2.2 the concept that the measure must fulfil a legitimate objective "at the level that the Member imposing the measure has determined appropriate". Moreover, reading into Article 2.2 this concept is inconsistent with the last phrase in the second sentence of Article 2.2, which reads "taking account of the risks non-fulfilment would create". This is because the phrase contemplates a "necessary" measure meeting a lower level of protection than determined by the Member if such a lower level of protection is appropriate in the light of the risks of non-fulfilment of the objective.

The application of Article 2.2 to the facts of this dispute

4.348 According to the US the stated objectives of the US measures relate solely to adverse effects on dolphins that occur when nets are set upon dolphins. Under this approach, the US measures have no objectives concerning adverse effects on dolphins resulting from other fishing methods or occurring in ocean regions other than the ETP. Mexico's position is that Mexico's fishing methods do not adversely affect dolphins and that the fishing methods of other fishing fleets do adversely affects dolphins.

The US measures do not fulfil their objectives

4.349 Mexico continues to question the legitimacy of the objectives of the US measures, in particular the objective to impose unilaterally and extraterritorially the fishing methods preferred by the US as a condition for access to the principal distribution channels in the US tuna products market. It is not "legitimate" for the US to use this means to get foreign tuna fleets to change their fishing Mexico also questions the "zero tolerance" standard that appears to underlie the US measures. However, even assuming arguendo that the objectives of the US measures are legitimate, the measures do not fulfil those objectives. The fundamental premise of the US measures has not been substantiated. Thus, it cannot be said that the US measures "ensure that consumers are not mislead or deceived about whether the tuna product contains tuna that was caught in a manner that adversely affects dolphins". Thus, contrary to the stated objectives of the U.S., it cannot be said that the US measures "ensure that tuna products labelled as dolphin-safe are in fact dolphin-safe" and that the label "accurately conveys to consumers that the product does not contain tuna that was caught in a manner that adversely affects dolphins". Finally, since the US measures essentially block from the principal distribution channels tuna caught only in the ETP, the objective of encouraging foreign fishing fleets not to set upon dolphins applies only to fleets fishing in the ETP. The fishing method of those fleets is governed by the AIDCP and the US measures do not go any further than the AIDCP. Thus, the US measures have no effect on encouraging foreign fishing fleets not to set upon dolphins.

The US measures are more trade restrictive than necessary to fulfil a legitimate objective taking into account of the risks non-fulfilment would create

4.350 To the extent that the US measures fulfil any objectives, taking into account the risks of nonfulfilment, those objectives could be fulfilled using less trade restrictive measures. The risks of nonfulfilment of the objectives is low because there is a very low if non-existent risk of adverse consequences should the objectives not be carried out. This is because all of the objectives of the US measures are already being fulfilled by the AIDCP, at least for the tuna products made from tuna harvested in the ETP. For tuna products produced from tuna harvested outside the ETP, the stated objectives are not being fulfilled. A less trade restrictive way of fulfilling the objectives would be to create dolphin-safe *standards* rather than a technical regulation whereby the AIDCP standard could be recognized and a label complying with the AIDCP standard used. At the same time, the different US standard could be recognized and a label complying with that standard used. In this way US consumers will be fully informed of all aspects of dolphin-safe fishing methods and they can choose accordingly when purchasing tuna products from US retailers.

(iv) The US measures are inconsistent with Article 2.4 of the TBT Agreement

4.351 Focusing on the key points, the AIDCP resolution on certifying tuna as dolphin-safe addresses the exact same issue as the US dolphin-safe measure. The resolution is relevant within the meaning of Article 2.4 and it is effective to meet the objective of informing consumers that dolphins were not adversely affected. The US participated in the creation and adoption of the AIDCP resolution and when it first announced its official label in 1999 it considered the international standard to be relevant. The AIDCP is a group of national governments that acts collectively to set standards

for protecting dolphins in the ETP tuna fishery, and which uses the IATTC Secretariat as its administering body. Finally, the US agreed to this standard, and is now trying to avoid it. For purposes of this dispute, the AIDCP dolphin-safe standard is the relevant international standard.

2. United States

(a) Introduction

- 4.352 Mexico is essentially asking the Panel to find that Articles I:1 and III:4 of the *General Agreement on Tariffs and Trade 1994* ("GATT 1994") and Articles 2.1, 2.2 and 2.4 of the *Agreement on Technical Barriers to Trade* ("TBT Agreement") require the United States to allow tuna products that contain tuna caught by setting on dolphins to be labelled dolphin-safe. Setting on dolphins to catch tuna, however, is not safe for dolphins. It is for this reason that the US voluntary dolphin-safe labelling provisions do not permit tuna products that contain tuna caught by setting on dolphins to be labelled dolphin-safe.
- 4.353 With respect to GATT 1994 Article III:4, Mexico fails to present any evidence that the US provisions afford any different treatment to imported tuna products as compared to domestic tuna products. With respect to Article I:1, Mexico fails to present any evidence that the US provisions afford any different treatment to Mexican products as compared to tuna products from other countries.
- 4.354 Mexico has also failed to establish that the US provisions set out labelling requirements with which compliance is mandatory and has therefore failed to establish that the US provisions are even technical regulations subject to Articles 2.1, 2.2 and 2.4 of the TBT Agreement. It also has not established the other elements necessary to demonstrate a breach of those articles.
- (b) Setting on dolphins adversely affects dolphins
- 4.355 The United States has provided ample evidence in its earlier submissions that setting on dolphins has significant adverse effects on individual dolphins. Dependent nursing dolphin calves are often separated from their mothers during the high-speed chases. Without their mothers, these young dolphins often starve to death or are killed by predators. Dolphins may also suffer other harms, some of them cumulative, that may not be observed at the time of the set but manifest at some point later. And these conclusions are uncontested by Mexico.
- 4.356 There is clear evidence that these adverse effects result in unobserved mortality and serious injury of dolphins in the ETP and are attributable to the practice of setting on dolphins to catch tuna. For example, Mexico does not refute evidence that dolphin mortality is at least 14% greater than observed dolphin mortality due to dependent calves that are separated from their mothers.
- 4.357 In terms of the effects on dolphin populations, the United States has also offered substantial evidence that setting on dolphins is adversely affecting dolphins stocks. Mexico is wrong when it argues that dolphin stocks are growing at rates that indicate population recovery. The most recent assessment model was published in 2007. That report shows median annual growth rates at 1.7 and 1.4 per cent, respectively, for offshore spotted and eastern spinner dolphin stocks in the ETP.

(c) ETP is unique

4.358 The ETP is the only ocean in the world where there is a regular and significant tuna-dolphin association that is commercially exploited on a wide scale commercial basis to catch tuna. And, it is the only ocean in the world for which the best available science indicates that the most probable reason dolphin populations have remained depleted for over 20 years and show no clear signs of recovery is because of that fishing method.

4.359 Mexico's assertions that there is substantial evidence of tuna-dolphin associations and harm to marine mammals in other fisheries at least comparable to that in the ETP are simply unfounded. The additional information Mexico cites in its second written submission does not support that dolphin mortality outside the ETP is comparable to dolphin mortality inside the ETP or that tuna and dolphins associate outside the ETP in any way comparable to the way they do in the ETP.

(d) Access to US market

- 4.360 Mexico asserts that the US dolphin-safe labelling provisions "de facto" condition access to principal US distribution channels for tuna products on those products containing tuna caught using a "fishing method unilaterally imposed by the United States". There is no basis for Mexico's assertion. The US dolphin-safe labelling provisions do not require tuna products to be labelled dolphin-safe to be exported to or sold in the US market and in fact tuna products that are not labelled dolphin-safe are sold in the United States.
- 4.361 Further, under the US provisions, there is no reason Mexico could not sell tuna products in the United States that contain tuna caught by that portion of its fleet that does not set on dolphins with a dolphin-safe label. In fact, at one point in time, Mexico did exactly this.
- 4.362 Mexico appears to equate the US provisions as "pressure" amounting to a requirement. There is no basis for this. Further, it is retailers' and consumers' preference for dolphin-safe tuna, not the US provisions, that affects the demand in the United States for tuna products that contain tuna caught by setting on dolphins. Mexico's arguments that the US provisions created consumers' preference for dolphin-safe tuna products are unsupported and ignore the history leading up to enactment of the US dolphin-safe labelling provisions.

(e) Article III:4 of GATT 1994

- 4.363 Regarding Mexico's claims under Article III:4, Mexico misconstrues the US argument. The United States is not arguing that simply because a measure is origin neutral on its face, it is consistent with the obligation under Article IIII:4 to provide national treatment. Instead, what we have explained in our previous submissions is that in order to establish that a measure affords less favourable treatment to imported products either on its face or in fact it must be shown that the treatment afforded imported products is different than the treatment afforded like domestic products and that any such different treatment is less favourable. Article III:4 does not prohibit different treatment of products based on factors unrelated to the foreign origin of the product. As Article III:1 makes clear, the general principle under Article III is that measures shall not be applied so as to afford protection to domestic production.
- 4.364 Thus, in this dispute where Mexico concedes that the US dolphin-safe labelling provisions do not on their face afford less favourable treatment to imported products, Mexico must show that the US provisions use what appears to be an origin neutral condition, for example, fishing technique, to single out imported tuna products for treatment that is different and less favourable than treatment accorded domestic tuna products.
- 4.365 Mexico fails to acknowledge the significance of the findings in *EC Approval and Marketing of Biotech Products* and *Dominican Republic Import and Sale of Cigarettes*. In particular, Mexico misses the key point made by the panel and Appellate Body in those reports: establishing that a measure affords less favourable treatment to imported products than to like domestic products requires showing that any different treatment afforded some imported products as compared to some like domestic products is attributable to the foreign origin of the imported products.

- 4.366 As noted, Mexico has failed to establish that the facially origin-neutral conditions under which tuna products may be labelled dolphin-safe in fact accord different treatment to imported tuna products.
- 4.367 First, no tuna product may be labelled dolphin-safe if it contains tuna caught by setting on dolphins. It is true that Mexico sets on dolphins to catch tuna, and that tuna products containing that tuna cannot be labelled dolphin-safe. But those facts do not establish that the US dolphin-safe labelling provisions use fishing technique as a proxy to treat imported tuna products differently (or less favourably) than domestic tuna products.
- 4.368 Second, Mexico has expressly stated that its national treatment and most-favoured nation claims are not based on any differences in documentation to support dolphin-safe claims.
- 4.369 Third, in our previous submissions, the United States reviewed the type of evidence put forth in other disputes where seemingly origin-neutral criteria were in fact used to afford different treatment to imported products. For example, in the *Mexico Taxes on Soft Drinks* dispute, the facts demonstrated that "almost 100 per cent" of imported products were subject to a higher tax rate than like domestic products. In the *Chile Alcoholic Beverages* dispute, the facts showed that 95 per cent of imports were subject to the higher tax rate. In this dispute, the vast majority of imported tuna products do not contain tuna that was caught by setting on dolphins and are eligible to be labelled dolphin-safe.
- 4.370 Mexico has also failed to show that the US provisions modify the conditions of competition to the detriment of imported products. First, to establish that a measure has modified the conditions of competition to the detriment of imported products it is insufficient to show that the measure introduced some change in the market; instead, it must be shown that the change the measure introduced modified the conditions of competition as between imported products and like domestic products.
- 4.371 Further, the US measures do not affect the ability of Mexican tuna products to be marketed in the United States. That many retailers choose not to stock non-dolphin-safe tuna, and that there is limited demand for non-dolphin-safe tuna products generally, is a result of retailer and consumer preferences for dolphin-safe tuna products, not the US dolphin-safe labelling provisions.
- 4.372 Second, Mexico is wrong that the impact of the US provisions on Mexican tuna products differed from their impact on domestic tuna products. For example, at the time the US provisions were enacted 46 US purse seine vessels fished for tuna in the ETP as compared to 52 Mexican purse seine vessels, and both primarily set on dolphins to catch tuna.

(f) Article I:1 of GATT 1994

- 4.373 With respect to its claim under Article I:1, first, Mexico appears to be conflating its Article III:4 and Article I:1 claims. In particular, Mexico assumes that the "conditions of competition" analysis that panels and the Appellate Body have taken in examining whether a measure affords "less favourable treatment" under Article III:4 is equally applicable to examining whether a measure fails to accord an advantage under Article I:1. This is not the case.
- 4.374 Second, Mexico has failed to put forward any evidence of the advantage the US provisions allegedly accord tuna products originating in other countries that the US provisions allegedly fail to accord Mexican tuna products. It has also not explained why, if the US provisions "pressure" Mexico's fishing fleet to change its fishing practices or location, the US provisions do not similarly pressure fishing fleets of other countries to change fishing methods or location.

- 4.375 The sole basis for Mexico's claim under Article I:1 is that Mexican tuna products contain tuna caught by setting on dolphins in the ETP, while tuna products originating in other countries contain tuna caught using alternative methods in other oceans. However, evidence submitted by the United States shows that vessels of many countries fish for tuna in the ETP both by setting on dolphins and by using other methods, and that the origin of tuna is determined by the flag of the vessel that caught it, not where it was caught.
- 4.376 In this regard, Mexico misunderstands the reports in *Canada Autos* and *Colombia Ports of Entry*. Those reports very clearly state even in the passages Mexico quotes that Article I:1 permits Members to grant an advantage subject to conditions, provided such conditions are not based on origin. Contrary to Mexico's assertion, the facts in this dispute are not analogous to those in *Belgian Family Allowances*: the conduct of the government of Mexico, including the laws it has adopted, has nothing to do with whether tuna products may be labelled dolphin-safe.
- 4.377 We also recall that the GATT 1947 panel in *US Tuna (Mexico)* already considered and decided whether the US dolphin-safe labelling provisions discriminate against Mexican tuna products by conditioning the use of a dolphin-safe label on tuna products not containing tuna caught by setting on dolphins.
- (g) Additional points raised by Mexico
- 4.378 Mexico has not identified how the text of Article I:1 or III:4 of the GATT 1994 prohibit the type of "pressure" Mexico claims the US provisions exert on its fishing fleet. Evaluating whether or not a measure might amount to "extra-territorial regulation" is neither useful nor relevant to analysing whether a measure is or is not consistent with these obligations.
- 4.379 Moreover, Mexico's assertion that the WTO Agreement prohibits a Member from conditioning access to its market on products having been produced in a particular way, if true, would have significant implications that go way beyond the issues in this dispute.
- 4.380 Mexico asserts, without any reference to provisions of the WTO Agreement or other relevant sources, that the Panel should only consider facts as proposed at the time the Panel was established when assessing if the US provisions violate Articles I:1 and III:4 of the GATT 1994. Mexico's argument should be rejected. There is no reason that the panel should ignore the evidence and analysis that is clearly relevant to this dispute.

(h) TBT claims

- 4.381 *Technical Regulation*. Regarding Mexico's claims under the TBT Agreement, Mexico has not established that the US dolphin-safe labelling provisions are technical regulations and therefore that the US provisions are even subject to the TBT provisions that Mexico alleges they breach.
- 4.382 Mexico continues to focus on the notion that because the US provisions make it unlawful to label tuna products dolphin-safe if they contain tuna caught by setting on dolphins, compliance with the US provisions is mandatory. Mexico's argument however conflates the meaning of the term "labelling requirement" with the meaning of the phrase "with which compliance is mandatory". The term "labelling requirement" appears in both the definition of a standard and the definition of a technical regulation, with the difference between a labelling requirement that is a standard and a labelling requirement that is a technical regulation being that compliance with the former is not mandatory while compliance with the latter is. Thus, to establish that a labelling requirement is a technical regulation, compliance with the conditions under which a product may be labelled in a certain way must be mandatory.

- 4.383 Compliance with a labelling requirement is mandatory where the measure not only sets out the conditions under which a product may be labelled in a particular way, but also requires the product to be labelled in that way to be imported, sold or otherwise placed on a market. Were it otherwise, as Mexico suggests, this would render either the term "labelling requirement" or the phrase "with which compliance is not mandatory" in the definition of a standard without effect. It would also render the phrase "with which compliance is mandatory" in the definition of a technical regulation without effect.
- 4.384 Mexico's arguments that compliance with the US dolphin-safe labelling provisions is mandatory because they "restrict retailers, consumers and producers to a single choice for labelling tuna products dolphin-safe" finds no basis in the text of the TBT Agreement. Allowing products to be labelled in a certain way that do not meet those conditions would defeat the purpose of the labelling requirement.
- 4.385 Articles 2.2 and 2.4 of the TBT Agreement. Setting aside that Mexico has not established that the US dolphin-safe labelling provisions are technical regulations, Mexico has also not established the basic elements necessary to substantiate its claims under Articles 2.2 and 2.4 of the TBT Agreement. For example, with respect to its claim under Article 2.2 of the TBT Agreement, Mexico has not identified a reasonably available alternative measure that would fulfil the objectives of the US provisions. There are a number of additional flaws in Mexico's arguments under Article 2.2 and 2.4.
- 4.386 First, Mexico's assertion that the US dolphin-safe labelling provisions are based on the "assumption" that setting on dolphins to catch tuna adversely affects dolphins flatly contradicts the evidence before this Panel as reviewed earlier in our statement. By conditioning the labelling of tuna products on such products not containing tuna caught by setting on dolphins, the US provisions fulfil their objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna caught in a manner harmful to dolphins.
- 4.387 Second, the hypothetical situation that Mexico describes under which a tuna product might be sold as dolphin-safe that contains tuna caught in a set in which a dolphin was killed or seriously injured is not evidence that the US provisions fail to fulfil their objective. Mexico's argument ignores the reality that in adopting measures to fulfil legitimate objectives it is appropriate for Members to consider the costs of such measures in light of their benefits. Indeed, this is a means by which Members can help ensure that their measures are not more trade-restrictive than necessary to fulfil their legitimate objectives.

F. CLOSING ORAL STATEMENTS AT THE SECOND SUBSTANTIVE MEETING

1. Mexico

- 4.388 In this dispute Mexico has presented claims under GATT Articles I.1 and III.4 and Articles 2.1, 2.2 and 2.4 of the TBT Agreement.
- 4.389 We respectfully requests that the Panel rule on *all* of its claims, in particular the discrimination claims under both Articles I:1 and III.4 of the GATT 1994.
- 4.390 Mexico reiterate that a lot of the discussion we had during this second substantive meeting with the panel related to the scientific issues of this dispute including the factual aspects of fishing outside the ETP is not relevant to Mexico's discrimination claims. It is relevant only to Article 2.2 of the TBT Agreement.

- 4.391 The evidence that Mexico has put forward is sound and accurate. Mexico has presented a prima facie case with respect of all of its claims.
- 4.392 As discussed earlier today, we have presented an effective less trade restrictive alternative to the US measures.
- 4.393 We would finally like to thank the Panel for your attention in this meeting, the Secretariat staff for your support, and also the interpreters for your assistance.

2. United States

- 4.394 Mr Chairman, members of the Panel, thank you for the opportunity for closing remarks. We will start by reviewing a couple of facts central to this dispute.
- 4.395 First, is the fact that it is not a requirement to access the US market to have tuna products labelled dolphin-safe, nor for them to be dolphin-safe to enter US market. In fact, there is substantial evidence that there are non-dolphin-safe tuna products on the US market and dolphin-safe products not labelled dolphin-safe on the US market. Thus, there is clear evidence that a dolphin-safe label is not a condition to access the US market.
- 4.396 A second factual element of critical importance is that setting on dolphins is harmful to dolphins. There is clear evidence that it is harmful to dolphins, and Mexico has not contested this evidence. The only evidence that Mexico has contested is at what rate dolphin populations are growing. Mexico has concluded from the information it cites that dolphin populations are recovering. We have discussed why this is not correct. And, this relates to a broader issue regarding the facts presented by Mexico in this dispute.
- 4.397 Throughout this proceeding the United States has cited a suite of published scientific papers, which underwent rigorous peer-review and scrutiny, to describe the abundance and trends of ETP dolphin stocks and the impacts of intentional chase and encirclement on these stocks. The results of these papers, and the conclusions contained within, are robust and regarded throughout the scientific community as setting a research standard, most notably for abundance estimation of dolphins using ship-based survey methods.
- 4.398 We have observed during these hearings and in its written submissions that Mexico has presented or interpreted available information in many cases without the necessary and proper context provided by the authors. Instead, Mexico has selectively used portions of papers or data that, when taken out of context or expressed without important uncertainties or caveats, appear to lend support to its arguments, but in fact do not.
- 4.399 One example that we reviewed today was the 2008 abundance estimate report, that Mexico cited while omitting very clear caveats included in that report by its authors. Mexico has attempted to devalue the scientific evidence presented by the United States in this dispute, but has not provided a basis for these criticisms that finds support in the scientific literature. In an instance where Mexico did present such criticism, in the form of a National Research Council (NRC) Report regarding pre-fishery abundance, Mexico omitted the fact that subsequent estimates of pre-fishery abundance have been produced that addressed and accounted for potential deficiencies in these earlier estimates.
- 4.400 The third important factual point is that the ETP is unique. The degree of occurrence and exploitation of dolphins when fishing for tuna and the impact on dolphins of that is orders of magnitude higher in the ETP than in other oceans. This is because unlike in any other ocean in the world, millions of dolphins are intentionally targeted to catch tuna; this simply does not occur in any other ocean in the world.

- 4.401 Turning to a few legal points, regarding GATT 1994 Article III.4, we have talked a lot about what less favourable treatment means. Mexico has not argued that the US provisions on their face discriminate against Mexican tuna products. And, while we do not disagree that a facially neutral measure can discriminate, Mexico has not shown that *de facto* discrimination exists in this dispute. The United States has pointed to other disputes where *de facto* discrimination was found and pointed out that the type of evidence that lead to the conclusions in those disputes has not been presented by Mexico in this dispute.
- 4.402 We have also provided the Panel with an analytical approach that we think is helpful in looking at these issues, and one that is consistent with approaches taken in prior WTO reports. In determining whether a measure affords less favourable treatment, one needs to determine what treatment is being afforded imported products, and what treatment is being afforded like domestic products. That treatment can then be looked at under a conditions of competition analysis to determine if the treatment afforded imported products is less favourable than the treatment afforded domestic products.
- 4.403 Mexico suggests starting with the conditions of competition analysis. Even if you followed Mexico's approach, the results would be the same. The US provisions do not alter the conditions of competition to the detriment of imported products. It is true, that prior to the US labelling provisions, no standard for labelling tuna products dolphin-safe existed, and thus, the US provisions when adopted introduced a change in the market. However, that measure did introduce a change that treated imported products any differently than like domestic products. The US provisions afforded imported and domestic products the same opportunity to compete under the same conditions on use of the dolphin-safe label. Mexico instead focuses on the impact of the US measures. But even if that were an appropriate approach, the evidence on the record shows that the number of US and Mexican purse seine vessels in the ETP that caught tuna by setting on dolphins was relatively the same at the time the US provisions were adopted. The fact that US vessels and producers changed their fishing method or location, and that Mexican vessels and producers did not, is not evidence that the US provisions altered the conditions of competition to the detriment of imported products.
- 4.404 With regards to Article I:1 of the GATT 1994, we have discussed in the context of Article III:4 why the approach Mexico has taken in attempting to show the US provisions discriminate against Mexican tuna products is not valid, and Mexico's arguments under Article I:1 are without merit for similar reasons.
- 4.405 With respect to Mexico's claims under the TBT Agreement, we spent a lot of time discussing the definition of a technical regulation, what it means for compliance to be mandatory, and what it means to be a labelling requirement. For compliance with a labelling requirement to be mandatory, a measure must set out conditions under which a product may be labelled in a certain way, and require the product to be labelled to be placed on the market. Mexico on the other hand asserts that a labelling requirement is converted to a technical regulation when there is only a single set of conditions under which a product may be labelled in a particular way. Mexico is essentially saying that in order for the US dolphin labelling provisions to be voluntary, the US must permit products to be labelled dolphin-safe even if they do not meet the conditions to be labelled dolphin-safe. This leaves no room for a measure to be a "labelling requirement" and be voluntary. Yet, labelling requirements are covered under both the definition of a standard, which provides that standards are voluntary, and the definition of a technical regulation, which provides that technical regulations are mandatory.
- 4.406 Regarding Mexico's claims under Articles 2.2 and 2.4 of the TBT Agreement, I will not discuss those in detail now, but do reiterate that Mexico has failed to establish that the US provisions are inconsistent with those articles and refer the Panel to the US written submissions.

4.407 Mr Chairman, members of the Panel, and members of the Secretariat assisting you, thank you for your time and attention throughout these proceedings. This concludes my closing statement.

ARGUMENTS OF THE THIRD PARTIES⁵² V.

- A. ARGENTINA
- Oral statement of Argentina⁵³ 1.
- Introduction (a)
- 5.1 The Argentine Republic has expressed the wish to participate as a third party in these proceedings given its systemic interest in the correct interpretation of the provisions at issue in this dispute, under both the Agreement on Technical Barriers to Trade (TBT Agreement) and the GATT 1994.
- Analysis of the United States measures in the light of the TBT Agreement (b)
- The US regulations on dolphin-safe labelling are technical regulations and are therefore *(i)* subject to the requirements of the TBT Agreement

The US regulations on dolphin-safe labelling are *de jure* technical regulations

- Argentina considers that the regulations on dolphin-safe labelling constitute "technical regulations" and come within the scope of application of the TBT Agreement inasmuch as they meet the three criteria laid down by the Appellate Body⁵⁴ for definition as technical regulations.
- The United States argues that the measure under consideration does not constitute a "technical 5.3 regulation" since one of these requirements is not met, namely the requirement of being mandatory. According to the United States, its dolphin-safe labelling legislation is a "standard", not subject to the disciplines of Articles 2.1, 2.2 and 2.4 of the TBT, but to Article 4 thereof.⁵⁵
- The Panel must determine the scope of the requirement of "mandatory" compliance under paragraph 1 of Annex 1 to the TBT, and thus establish the boundary line between "technical regulations" and "standards", which will have serious systemic implications.
- 5.5 A decision that excluded the possibility of applying the disciplines established by the TBT for the issuance and implementation of technical regulations in respect of any labelling requirement. where the use thereof was voluntary, could open the way for the indiscriminate use of this type of technical requirement for protectionist purposes, thereby defeating the object and purpose of the TBT. It should be remembered that one of the objectives of the TBT is to prevent technical standards being "applied in a manner which would constitute a means of arbitrary or unjustifiable discrimination between countries where the same conditions prevail or a disguised restriction on international trade".56

⁵² This section of the arguments of the third parties is based on the executive summaries submitted by the third parties to the Panel, where available. Footnotes in this section are those of third parties.

This oral statement was originally made in Spanish.

⁵⁴ Appellate Body Report, EC – Asbestos, paras. 67-70 and Appellate Body Report, EC – Sardines, para. 176.

55 United States' first written submission, paras. 125-138.

⁵⁶ Preamble, TBT Agreement.

- 5.6 The Appellate Body in *EC Asbestos* elaborated on the scope of the term "mandatory" as used in the first sentence of the definition of "technical regulation" contained in Annex 1 to the TBT. In the same case, the Appellate Body also made it clear that the second sentence of the definition of "technical regulation" contains a list of examples of the concept of "product characteristics" used in the first sentence. The sentence of the term "mandatory" as used in the first sentence. The sentence of the definition of the term "mandatory" as used in the first sentence. The sentence of the term "mandatory" as used in the first sentence. The sentence of the definition of the term "mandatory" as used in the first sentence of the definition of the term "mandatory" as used in the first sentence of the definition of the term "mandatory" as used in the first sentence of the definition of the term "mandatory" as used in the first sentence of the definition of the term "mandatory" as used in the first sentence of the definition of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the definition of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandatory" as used in the first sentence of the term "mandator
- 5.7 From a combined reading of these two provisions, it is apparent that <u>compliance with the</u> "labelling requirements" contained in a document must be mandatory in order for the document to be <u>considered a "technical regulation"</u>. In particular, a document which establishes "labelling requirements", "with which compliance is mandatory", shall constitute a "technical regulation".
- 5.8 A distinction should be made between two separate situations to which compliance with "labelling requirements" may apply: (a) the mandatory or non-mandatory use of a label; and (b) the requirements that must be met for the use of the label. Argentina considers that the words "labelling requirements" and "with which compliance is mandatory" cover both situations in accordance with the ordinary meaning of those terms.
- 5.9 There is no reason whatsoever to limit the ordinary meaning of the words analysed to the obligation to affix a label, excluding the "mandatory requirements" that have to be met if it is decided to affix a label. Therefore, provided that at least one of these two elements is of a "mandatory" nature, the necessary requirement of mandatoriness will be met so that the document in question can be characterized as a "technical regulation".
- 5.10 In the case under consideration, the dolphin-safe label acts as a sort of "certification" that the marketed product fulfils certain requirements laid down by law. With labels of this type, the requirements that must be met by the product in order for it to be eligible for the label are always mandatory.
- 5.11 Argentina considers that an interpretation of Annex 1 to the TBT Agreement of the kind proposed by the United States, which would require the use of a dolphin-safe type label to be mandatory in order for a document to be characterized as a "technical regulation", would turn any label of this kind the use of which, by definition, is voluntary into a "standard", thereby depriving the definition of "technical regulation" of any content.
- 5.12 In addition to the notion of mandatoriness, there is another element which, in Argentina's view, distinguishes standards from technical regulations: "standards" must be "approved by a recognized body". The United States has not demonstrated that this element is present in its legislation on dolphin-safe labelling.
- 5.13 For the above reasons, Argentina considers that the dolphin-safe labelling requirements established in US legislation constitute a "technical regulation" within the meaning of Annex 1 to the TBT and that, consequently, they are governed by the provisions contained in Articles 2.1, 2.2 and 2.4 of that Agreement.

The interpretation proposed by Argentina is supported by the Panel Report in EC – Sardines

5.14 Argentina considers that the instant case presents similarities with the EC – Sardines case, and that the decisions of the Panel and the Appellate Body in that case confirm the view that

⁵⁷ Appellate Body Report, EC – Asbestos, para. 68.

⁵⁸ Appellate Body Report, *EC – Asbestos*, para. 67.

⁵⁹ Paragraph 2 of Annex 1 to the TBT Agreement.

dolphin-safe labelling requirements constitute a "technical regulation". In that case, there was no obligation on Peruvian producers to market their fish ("*Sardinops sagax*") as "preserved sardines". On the contrary, they could market their product freely, provided that it did not bear the mark "preserved sardines".

- 5.15 As in the EC Sardines case, Mexican producers are free to market their products in the United States, provided that they do not use a label indicating that they are dolphin-safe. On the other hand, if they wished to use the mark "dolphin-safe", the product would have to satisfy a mandatory requirement: that of having been caught in accordance with the conditions laid down in United States law.
- 5.16 In both cases, the use of the mark (whether it be "dolphin-safe" or "preserved sardines") entails enforcement of a document laying down the required characteristics of the product, compliance with which is mandatory; in other words, a technical regulation.
- 5.17 On that occasion, the Panel was in no doubt about its decision that the regulation in question did indeed constitute a "technical regulation", and that decision was not reversed by the Appellate Body.

The United States regulations on dolphin-safe labelling have the effect of technical regulations, because they operate "de facto" as such.

- 5.18 Argentina believes that the analysis of the "mandatoriness" of the document under consideration should not be confined to a determination of "de jure" mandatoriness, but should also look at whether the use of that label is mandatory in fact ("de facto" mandatoriness). Argentina is of the view that the document has the practical effect of a technical regulation and operates as such in relation to market actors.
- 5.19 The labelling authorized on the basis of compliance with regulations laid down by a State gives rise to a positive perception on the part of the consumer, who will then tend to allow his/her purchasing decisions to be influenced by the labelling, even if it is voluntary.
- 5.20 Given the existence of labelling regulations, it is an established feature of the US tuna product market that "most participants will not purchase, offer for sale, distribute or use tuna products" that fail to display the dolphin-safe label, but that they will purchase, offer for sale, distribute or use local tuna products, and products from other sources, which are labelled dolphin-safe.
- 5.21 Thus, as Mexican tuna is a like product, the regulations on positive dolphin-safe labelling operate as a *de facto* technical regulation, compliance with which is a "mandatory" requirement for the viable marketing of the product in the market where the regulation is in force.
- (ii) The United States regulations on dolphin-safe labelling are inconsistent with Article 2.1 of the TBT Agreement

The conditions for awarding the dolphin-safe label laid down by the United States regulations are discriminatory

5.22 The regulations under consideration lay down conditions for awarding the dolphin-safe label that discriminate between like products, whether of national origin or originating in other Members.

⁶⁰ Panel Report, EC – Sardines, paras. 7.3-7.4.

⁶¹ Mexico's first written submission, para. 165.

Argentina maintains that a method of granting a positive label based on a system or conditions of production, as in the present case with regard to the use of a specific type of fishing gear, results in discrimination between like products.

5.23 The US regulations discriminate on the basis of where the tuna has been caught and the fishing method employed. As the fishing fleets of the United States and other countries fish in different seas and use types of fishing gear different from those used by Mexico, the conditions for awarding the dolphin-safe label established by the United States regulations have the effect of favouring tuna products from the US and other countries over Mexican products. Consequently, these regulations adversely alter the conditions of competition in the US market, so that Mexican tuna is de facto afforded less favourable treatment than that granted to the identical product of national origin or originating in other countries. Thus, the regulations are inconsistent with Article 2.1 of the TBT Agreement.

The mere existence of a positive label regulated by a State gives rise, per se, to discrimination

- 5.24 Argentina would emphasize that the mere existence of a positive label regulated by a State gives rise, *per se*, to a perception which leads the consumer to incline towards the *like* product bearing the label, whether it be of national origin or from some other WTO Member, to the detriment of the imported *like* product which is not authorized to bear that label.
- (iii) The AIDCP is not a "recognized body" within the meaning of the second paragraph of Annex 1 to the TBT, nor is it a "standardizing body"
- 5.25 Argentina considers that the Agreement on the International Dolphin Conservation Program (AIDCP) should not be regarded as a "recognized body" under the terms of the second paragraph of Annex 1 to the TBT Agreement. The agreement in question could not have that status given that it does not meet the definition of an "international body" because it is not "open" to participation by the relevant bodies of at least all WTO Members. On the contrary, the Agreement provides for a specific accession procedure with specific criteria or conditions.
- 5.26 According to the "Decision of the [TBT] Committee on principles for the development of international standards, guides and recommendations with relation to Articles 2, 5 and Annex 3 of the Agreement" (G/TBT/9), openness is one of the principles that should be observed in drafting "international standards". The decision states that membership of an international standardizing body should be open on a non-discriminatory basis to the relevant bodies of all WTO Members.
- 5.27 The decision also states that, in the preparation of standards, transparency, impartiality and consensus, effectiveness and relevance, coherence and the development dimension should be observed in addition to openness. The development dimension refers to the fact that the constraints on developing countries' effective participation in the elaboration of international standards should be taken into account. Hence, the impartiality and openness of any international standardization process requires that developing countries are not excluded *de facto* from the process.
- 5.28 Argentina notes that the membership of the AIDCP is neither open nor oriented towards universality as befitting an institution authorized to approve "international standards". Consequently, the fact that only a limited number of countries are parties to the AIDCP and that it is a "semi-closed" agreement which does not provide for the participation of the relevant bodies of at least all WTO Members, and that it imposes conditions on the participation of new States or regional organizations, means that it cannot be categorized as a "recognized body" under the terms of the TBT Agreement.

- 5.29 Similarly, the AIDCP is not a "standardizing body"⁶² with recognized standardization activities. In fact, the development of *standards* is not the principal function of the AIDCP, as it is in other standardizing bodies such as the FAO/WHO Codex Alimentarius
- 5.30 In view of the foregoing, AIDCP resolutions could not constitute "relevant international standards" within the meaning of Article 2.4 of the TBT Agreement, and WTO Members are not obliged to bring their domestic legislation into line with those parameters.
- (c) Conclusion
- 5.31 Argentina hopes that the above comments may be useful in the Panel's deliberations and help arrive at an interpretation that is harmonious and in accordance with the aims and objectives of the TBT Agreement and the GATT, avoiding forms of discrimination in international trade flows that are unjustifiable in the light of the principles of the multilateral trading system.
- B. AUSTRALIA
- 1. Third party submission and oral statement of Australia
- (a) Mexico's claims under the GATT 1994
- (i) Mexico's claim under Article III:4 of the GATT 1994
- 5.32 Australia considers that whether a measure is "origin-neutral", while indicative, is not determinative of whether a measure accords less favourable treatment to imported products. Rather, the question the Panel must address for the purposes of Article III:4 of the GATT 1994 is whether the US dolphin-safe labelling measures modify the conditions of competition to the detriment of Mexican tuna and tuna products. The Appellate Body has said the focus of this investigation should be on what appears to be the fundamental thrust and effect of the measure itself.
- 5.33 Australia submits that the Panel may look to a range of criteria to determine whether an equality of competitive conditions exists, and should examine those criteria in the light of the fundamental effect the measure has on competition. 65
- (ii) Mexico's claim under Article I:1 of the GATT 1994
- 5.34 In *Canada Autos* the Appellate Body confirmed that Article I:1 of the GATT 1994 covers *de facto* as well as *de jure* discrimination.⁶⁶ The Appellate Body in that case refused to accept the proposition that measures which were on their face "origin-neutral" could not otherwise be in breach of Article I:1 of the GATT 1994.⁶⁷
- 5.35 In this dispute, the question to be addressed by the Panel is whether *in practice* the US dolphin-safe labelling measures accord an advantage to the tuna products of any other Member that is not also immediately and unconditionally accorded to the tuna products of Mexico. More particularly, can the US dolphin-safe labelling measures, which accord an advantage to tuna products

 $^{^{62}}$ "International standardization body" according to the Panel Report, \textit{EC}-Sardines, para. 221.

⁶³ Appellate Body Report, Korea – Various Measures on Beef, paras. 135 and 137.

⁶⁴ Appellate Body Report, *Korea – Various Measures on Beef*, para. 142.

⁶⁵ Appellate Body Report, *Korea – Various Measures on Beef*, paras. 142-147; Appellate Body Report, *EC – Asbestos*, para. 123.

⁶⁶ Appellate Body Report, *Canada – Autos*, para. 78.

⁶⁷ Appellate Body Report, *Canada – Autos*, para. 78.

of all countries regardless of origin, provided those countries act in a certain way, nevertheless be considered *de facto* discriminatory vis-à-vis Mexico? In this regard, Australia notes the statement of the panel in *Canada – Autos* that "an advantage can be granted subject to conditions without necessarily implying that it is not accorded 'unconditionally' to the like product of other Members". ⁶⁸

- (b) Mexico's claims under the TBT Agreement
- (i) Technical regulation
- 5.36 In Australia's view, the US dolphin-safe labelling measures set out "requirement(s) concerning a product label" and are therefore a "labelling requirement" that applies to a product, process or production method within the definition of a technical regulation in paragraph 1 of Annex 1 to the TBT Agreement.⁶⁹
- 5.37 Australia notes the US dolphin-safe labelling measures do not *require* tuna products to be labelled or to contain certain information on the label; nor do they *prevent* the sale in the United States of tuna products containing tuna harvested in a particular manner or tuna products that do not bear the "dolphin-safe" label. The measures regulate the circumstances in which the "dolphin-safe" label may be used on tuna products. Australia therefore considers that the US dolphin-safe labelling provisions are not mandatory on their face and thus on this basis do not fall within the definition of a "technical regulation" in Annex 1.1 of the TBT Agreement.
- 5.38 Australia notes however Mexico's claim that the US dolphin-safe labelling measures, if not *a priori* mandatory, are nonetheless *de facto* mandatory. Australia shares New Zealand's view⁷¹ that there may be circumstances where a government's actions in conjunction with an otherwise voluntary measure effectively make compliance with the measure mandatory. Australia also agrees with New Zealand that "a measure that is not *a priori* mandatory will only constitute a 'technical regulation' in cases where it is clearly warranted by the facts'"⁷² In Australia's view, there must be some factor in the measure itself or the governmental actions surrounding the measure which mean for the relevant industry that a measure which appears to be voluntary on its face is effectively made "binding or compulsory".⁷³
- 5.39 In the circumstances of this dispute, Mexico's argument would appear to require that the effects in the market of consumer purchasing preferences flowing from information provided via the "dolphin-safe" label be attributable to government as a *mandatory* measure. However, Australia submits that consumer preferences *alone* cannot determine whether a labelling requirement is voluntary or mandatory. Such a proposition would result in all labelling requirements falling within the definition of a "technical regulation" and render meaningless the definition and disciplines of the TBT Agreement applying to standards.

⁶⁸ Panel Report, *Canada – Autos*, para. 10.24.

⁶⁹ See Panel Report, EC – Trademarks and Geographical Indications (Australia), para. 7.449; Appellate Body Report, EC – Asbestos, para. 67.

⁷⁰ Mexico's first written submission, para. 203.

⁷¹ New Zealand's third party submission, para. 23.

⁷² New Zealand's third party submission, para. 23.

⁷³ Appellate Body Report, *EC – Asbestos*, para. 68.

- (c) Mexico's claims under Article 2.1 and 2.2 of the TBT Agreement
- (i) Article 2.1 of the TBT Agreement
- 5.40 In Australia's view, the analysis of "treatment no less favourable" developed under GATT Article III:4 would assist in the interpretation of Article 2.1 of the TBT Agreement. ⁷⁴ Australia therefore considers that, like Article III:4 of the GATT 1994, Article 2.1 of the TBT Agreement "obliges Members of the WTO to provide *equality of competitive conditions* for imported products ..." in respect of technical regulations. ⁷⁵
- 5.41 Australia submits that when considering whether Mexican tuna products have been accorded "treatment less favourable" under Article 2.1 of the TBT Agreement, the Panel should have regard to the test set out by the Appellate Body in *Korea Various Measures on Beef* in the context of Article III:4 of the GATT 1994, by examining whether the measures at issue modify the conditions of competition in the US market to the detriment of imported Mexican tuna products. Australia considers that whether a measure is "origin-neutral", while indicative, is not in itself determinative of whether a measure accords less favourable treatment to imported products.
- 5.42 Australia submits that the Panel could also obtain guidance from the Appellate Body's views in *Korea Various Measures on Beef* that "the intervention of some element of private choice does not relieve [a Member] of responsibility under the GATT 1994 for the resulting establishment of competitive conditions less favourable for the imported product ...". Thus, Australia considers a relevant question for the Panel is whether, having regard to the evidence, the (alleged) inability of Mexican tuna products to access the US "dolphin-safe" label is the result of *governmental* action modifying the conditions of competition in the US market to the detriment of Mexican tuna products, having regard to the fundamental thrust and effect of the measures at issue.
- (ii) Article 2.2 of the TBT Agreement
- 5.43 In Australia's view, the first sentence of Article 2.2 establishes the fundamental obligation of Members with respect to "technical regulations" to "ensure that technical regulations are not prepared, adopted or applied with a view to or with the effect of *creating unnecessary obstacles to international trade*" The second sentence of Article 2.2 explains that "[f]or this purpose, technical regulations shall not be more trade-restrictive than necessary ...". In other words, the second sentence of Article 2.2 sets out the conditions technical regulations must meet in order to satisfy the fundamental obligation contained in the first sentence of Article 2.2.

Legitimate objective

5.44 Australia recalls the statement of the Panel in EC – Sardines that "Article 2.2 and [the] preambular text affirm that it is up to the Members to decide which policy objectives they wish to pursue and the levels at which they wish to pursue them". 80 Australia submits that under Article 2.2 of the TBT Agreement, as under Article 2.4, "there must be an examination and a determination on the

⁷⁴ See also Panel Report, EC – Trademarks and Geographical Indications (Australia), para. 7.464.

⁷⁵ Appellate Body Report, *Korea – Various Measures on Beef*, para. 135, citing Appellate Body Report, *Japan – Alcoholic Beverages II*, pp. 16-17.

⁷⁶ Appellate Body Report, *Korea – Various Measures on Beef*, para. 137.

⁷⁷ Appellate Body Report, *Korea – Various Measures on Beef*, para. 146.

⁷⁸ Appellate Body Report, *Korea – Various Measures on Beef*, para. 149.

⁷⁹ Appellate Body Report, *Korea – Various Measures on Beef*, para. 142.

⁸⁰ Panel Report, *EC – Sardines*, para. 7.120.

legitimacy of the objectives of the measure".81 There must be an assessment of whether the stated objective(s)⁸² of a technical regulation put forward by the respondent can be considered "legitimate" within the meaning of Article 2.2. However, Australia submits it is not relevant to this assessment whether the objective(s) put forward by the United States for the US dolphin-safe labelling measures are considered appropriate.83

More trade restrictive than necessary to fulfil a legitimate objective

- Australia considers that the interpretation of the phrase "more trade restrictive than necessary 5.45 to fulfil a legitimate objective" under Article 2.2 calls for a "weighing and balancing" of the elements contained in that phrase, similar to that required in a "necessity" analysis under GATT Article XX.
- Australia submits that the Panel's examination should focus on: (a) whether the measure is trade restrictive; (b) whether the measure is "to fulfil" a legitimate objective; and (c) whether there are other reasonably available alternatives that may be less trade restrictive while still fulfilling the legitimate objective at the level of protection the Member considers appropriate.
- Australia agrees with Mexico that measures that are "trade restrictive" include those that impose any form of limitation on imports, discriminate against imports or deny competitive opportunities to imports.⁸⁴
- In examining whether the US dolphin-safe labelling measures are "to fulfil a legitimate objective", Australia submits the relevant question is whether the measures carry out, or have the capacity to carry out, their stated objectives. 85 In addressing this question, the evidence before the Panel concerning the perceptions and expectations of US consumers relating to the meaning of the "dolphin-safe" label and the criteria behind its use will be crucial. Australia submits a relevant consideration is whether the evidence shows that consumers in fact understand the criteria behind the "dolphin-safe" label and accordingly base their purchasing decisions not only on whether dolphins were killed or seriously injured during harvesting, but also on whether the tuna contained in tuna products was caught in a particular manner.
- Australia submits that the question of whether there are other reasonably available, less trade restrictive alternatives⁸⁶ is pertinent to the Panel's consideration of whether a measure is "more *trade*" restrictive than necessary". Australia also considers that the degree of trade-restrictiveness of the US dolphin-safe labelling measures (and possible alternatives to those measures) is a relevant consideration for the Panel.⁸⁷ Australia considers the level of protection considered appropriate by the United States in relation to the US dolphin-safe labelling measures⁸⁸ and an examination of whether the US dolphin-safe labelling measures form "part of a comprehensive strategy" to address the

Appellate Body Report, EC – Sardines, para. 286.
 Appellate Body Report, EC – Sardines, para. 289.

⁸³ See Mexico's first written submission, paras. 207-208.

Mexico's first written submission, para. 217. See also Panel Report, *India – Autos*, para. 7.265; Panel Report, EC - Bananas III (Article 21.5 - Ecuador II), para. 7.330; Panel Report, EC - Bananas III (Article 21.5 - US), para. 7.677.

See also United States' first written submission, para. 154.

⁸⁶ See e.g. Appellate Body Report, *EC – Asbestos*, paras. 170-174; Appellate Body Report, *Korea –* Various Measures on Beef, paras. 165-166; Panel Report, US – Gasoline, para. 6.24; GATT Panel Report, US – Section 337 Tariff Act, para. 5.26.

⁸⁷ Appellate Body Report, *Korea – Various Measures on Beef*, para. 163.

Appellate Body Report, Brazil – Retreaded Tyres, para. 15, citing Appellate Body Report, US – Gambling, para. 311; sixth preambular paragraph, TBT Agreement.

protection of dolphins⁸⁹ are also relevant factors in the Panel's consideration of whether there are other reasonably available. less trade restrictive alternatives to the measures at issue.

Taking account of the risks non-fulfilment would create

- 5.50 Australia submits that a finding as to whether a technical regulation is "more trade restrictive than necessary to fulfil a legitimate objective" must be weighed against the risks non-fulfilment of the particular legitimate objective would create. Such risks may differ depending on the *nature* of the legitimate objective the measure is designed to fulfil and the level of protection a Member considers appropriate. If the risks associated with non-fulfilment of a particular objective would be high, then the measure may still be justified regardless of its trade-restrictiveness.
- 5.51 While the terms of GATT Article XX do not call expressly for an assessment of the "risks non-fulfilment would create", Australia submits there are parallels with the "necessity" test adopted by the Appellate Body in its consideration of GATT Article XX, which includes consideration of the importance of the interests or values at stake. 90 Australia believes such consideration is relevant to the determination of the issues in this dispute.
- C. Brazil
- 1. Third party submission and oral statement of Brazil
- (a) Introduction
- 5.52 Brazil offers comments on aspects of: (i) the definition of technical regulation in the TBT Agreement; (ii) Article 2.2 of the TBT Agreement; (iii) Article 2.4 of the TBT Agreement; (iv) less favourable treatment under Article III:4 of the GATT 1994; and (v) likeness.
- (b) Definition of technical regulation
- 5.53 The parties disagree as to whether the challenged measures are a "technical regulation" as defined in Annex 1(1) of the TBT Agreement. This definition is three-pronged. Brazil will comment on two prongs of this definition, i.e.: (1) a document that lays down product characteristics, or their related processes and production methods, and may also include or deal exclusively with certain requirements including labelling requirements as they apply to a product, process or production method; and (2) compliance with the requirements laid down in the document is mandatory.
- (i) "Product characteristics"
- 5.54 The United States argues that labels are not "product characteristics" within the meaning of the definition of "technical regulation". Conversely, Mexico argues that means of identification such as labels are product characteristics.

⁸⁹ Appellate Body Report, *Brazil – Retreaded Tyres*, paras. 154, 172, 211; see also United States' first written submission (corrected version), para. 171.

⁹⁰ Appellate Body Report, *Brazil – Retreaded Tyres*, para. 178.

⁹¹ TBT Agreement, Annex 1(1); and Appellate Body Report, EC – Asbestos, paras. 67, 68 and 70.

- 5.55 Brazil notes that the Appellate Body has stated plainly that for purposes of the definition of a technical regulation in Annex 1(1) of the TBT Agreement, "labelling requirements" are an "example[] of 'product characteristics'". 92
- 5.56 The suggestion was made that the panel and Appellate Body reports in *EC Sardines* and *EC Trademarks and Geographical Indications (Australia)* support the proposition that to establish whether a measure on labelling is a technical regulation, it is necessary to examine further whether the label relates to product characteristics. Brazil believes this is an incorrect reading. In both cases, the EU argued that the measure at issue did not lay down a labelling requirement. Addressing the EU's arguments, the panels and Appellate Body also proceeded to consider whether the measure could *otherwise* be seen as laying down product characteristics. Thus, they did *not* rule that a measure had to lay down product characteristics *in addition to* laying down labelling requirements. To the contrary, the panel in *EC Trademarks and Geographical Indications (Australia)* stated that "[t]he issue is not whether the content of the label refers to a product characteristic: the label on a product *is* a product characteristic".
- (ii) Whether compliance is mandatory
- 5.57 Mexico and the United States disagree on whether compliance with the labelling requirements at issue is mandatory. The Panel must examine this question with particular care, because it bears on the distinction between technical regulations and standards, which is central to the TBT Agreement.
- 5.58 Two cases may provide guidance for examining the question of mandatory *versus* voluntary labelling requirements. *First*, in *EC Sardines*, the challenged measure provided that only products consisting of *Sardina pilchardus* could be sold in the EU as "preserved sardines". Therefore, products consisting of *Sardinops sagax* could not be sold as "preserved sardines", but could be sold without this label. The EU did not even dispute that its measure was "mandatory". But in arguing that the measure did not apply to *Sardinops sagax*, it asserted that "[t]he only legal consequence of the [EU] Regulation for preserved *Sardinops sagax* [was] that they may not be called 'preserved sardines'." In rejecting the EU's argument, the Appellate Body explained that a measure on labelling, or naming, has an effect beyond the products that are allowed to bear a label or name. It said: "preserved products made, for example, of *Sardinops sagax* are, by virtue of the EC Regulation, *prohibited* from being identified and marketed under an appellation including the term 'sardines'."
- 5.59 Second, in EC Trademarks and Geographical Indications (Australia), the EU argued that the labelling requirement was "not ... mandatory ... in order to market the relevant products in the European Communities". The panel noted that the EU had tried and failed to pursue a similar argument in EC Sardines. It found that, similar to EC Sardines, the "negative implication" that followed from the challenged labelling requirement was that "products with a GI identical to a

⁹⁴ See Appellate Body Report, EC – Sardines, paras. 187 and 190; Panel Report, EC – Sardines, paras. 7.37 and 7.39; and Panel Report, EC – Trademarks and Geographical Indications (Australia), paras. 7.435, 7.437 and 7.448-7.450. In EC – Trademarks and Geographical Indications (Australia), the EU also suggested that labelling requirements had to apply to "characteristics of a product", a suggestion rejected by the panel. Panel Report, EC – Trademarks and Geographical Indications (Australia), paras. 7.437 and 7.449.

⁹² Appellate Body Report, *EC – Asbestos*, para. 67.

⁹³ Japan's third party submission, paras. 12-14.

⁹⁵ Panel Report, *EC – Trademarks and Geographical Indications (Australia)*, para. 7.449 (emphasis in the original).

⁹⁶ EU's Appellant's Submission as quoted in Appellate Body Report, EC – Sardines, para. 185.

⁹⁷ Appellate Body Report, *EC – Sardines*, para. 184 (emphasis in the original). *See* also Appellate Body Report, *EC – Sardines*, para. 185; and Appellate Body Report, *EC – Asbestos*, para. 69.

Panel Report, EC – Trademarks and Geographical Indications (Australia), para. 7.454.

Community protected name" that did not satisfy that requirement could not be marketed in the EU bearing that GI. It followed that the requirement at issue was "an obligatory or mandatory requirement". 99

5.60 There appear to be parallels between these two cases and the present one. In the present case, it appears that tuna caught in the Eastern Tropical Pacific with certain fishing methods cannot be sold in the US bearing the "dolphin-safe" label, irrespective of whether these fishing methods could also be dolphin-safe.

(c) Article 2.2 of the TBT Agreement

5.61 In Brazil's view, the following sequence of analysis may be useful in assessing conformity of a measure with Article 2.2: (i) first, examining whether the objective stated by the regulating Member is "legitimate"; (ii) second, if the stated objective is found to be legitimate, evaluating whether the measure bears a rational link with the stated objective. If this is the case, (iii) the measure should be subject to the "necessity test" that is provided under Article 2.2. Such a test entails an assessment of whether the measure is more trade-restrictive than necessary to fulfil the legitimate objective in point. This assessment includes consideration of the measure's effectiveness in fulfilling the asserted objective; its degree of trade-restrictiveness; and reasonably available less trade-restrictive alternatives. Article 2.2 also expressly provides that this "necessity test" must be informed by "the risks non-fulfilment would create". Brazil comments on some of these elements below.

(i) Assessment of the pursued objectives

- This does not mean that a panel has no role in determining whether the objectives are "legitimate". In *EC Sardines*, under Article 2.4, the Appellate Body agreed that "there must be an examination and a determination on the legitimacy of the objectives". Article 2.2 contains a non-exhaustive list of legitimate objectives. In this dispute, the United States asserts only objectives that fall within the list of Article 2.2.
- 5.63 A panel should also examine whether the asserted objective or objectives are actually those being pursued by the Member with the measures at issue. This involves examining the link between the measure and the stated objective. It might also involve, e.g. considering whether an objective has been identified *post hoc* in the course of a WTO dispute.
- (ii) The role of regional agreements under Article 2.2 of the TBT Agreement
- 5.64 In discussing Article 2.4 of the TBT Agreement, Brazil cautions that the Panel should exercise care in deciding whether an "international" standard exists. At the same time, where there is an agreement between the parties to a dispute that pursues the same objectives pursued by a challenged unilateral measure, this must be given importance, even if the agreement is only regional, and not open to at least all WTO Members. Such a fact would be relevant, in particular, under Article 2.2 of the TBT Agreement.

⁹⁹ Panel Report, EC – Trademarks and Geographical Indications (Australia), paras. 7.454-7.456.

This is, of course, a different question from that of which party bears the burden of proof under Article 2.2. See Appellate Body Report, EC – Sardines, paras. 275-282.

Appellate Body Report, EC – Sardines, paras. 286; and Panel Report, EC – Sardines,

Appellate Body Report, EC – Sardines, paras. 286; and Panel Report, EC – Sardines, paras. 7.121-7.122.

- Under Article 2.2, first, if the regulating Member's stated objective is already being pursued through a regional agreement, and yet the Member chooses to adopt a unilateral, distinct measure to pursue the same objective, this might cast some doubts on the rational link between the unilateral measure and the objective in question. Second, when a less trade restrictive alternative is embodied in a regional agreement between, among others, the parties to a dispute, such a fact must be given This would be consonant with the recognition, in the special weight under Article 2.2. TBT Agreement, that the harmonization of technical regulations and standards can be beneficial, by facilitating the conduct of international trade. 102
- (iii) Meaning of the phrase "trade restrictive"
- First, the United States imports the "significantly" less trade restrictive standard from 5.66 footnote 3 to Article 5.6 of the SPS Agreement, into Article 2.2 of the TBT Agreement. However, this standard is dictated by the specific wording of footnote 3 to Article 5.6 of the SPS Agreement. 103 The United States relies on a letter sent by the GATT Director-General to its Chief Negotiator in support of its reading that this standard also applies to the TBT Agreement. Brazil disagrees that a letter sent by the then Director-General to one party to the negotiations may, by itself, inform the meaning of a multilateral agreement.
- Second, the United States argues that because the challenged measures do not impede all tuna trade from Mexico, they are not trade restrictive. It is well-established that a measure may be "trade restrictive" without impeding all trade (or even any actual trade). The context provided by Article XI of the GATT 1994 offers useful interpretive guidance. Panels have observed that the ordinary meaning of "restriction" includes "a thing that restricts someone or something, a limitation on action, a limiting condition or regulation". ¹⁰⁴ They have found that a restriction can exist when there is an impact on competitive opportunities, and not necessarily on trade flows. ¹⁰⁵
- 5.68 Third, the United States argues that the only trade effects of the measures are caused by consumers' choices. Yet by disciplining labelling schemes, the TBT Agreement recognizes the importance of measures that operate through consumers' choices.
- "Taking account of the risks non-fulfilment would create" (iv)
- 5.69 Another interpretive issue raised by Article 2.2 is the meaning of the ending clause of the second sentence, "taking account of the risks non-fulfilment would create", accompanied by an illustrative list of "relevant elements of consideration" "in assessing such risks".
- This clause appears to require, in particular, an examination of the nature and degree of the risks that could occur if the legitimate objective were not fulfilled. In Brazil's view, an essential part of this examination is to assess the *importance* of the legitimate objective. ¹⁰⁶ For example, if the legitimate objective were the protection of human life and health, then the risks of non-fulfilment would likely be extremely high. If the objectives were different, the risks could be lower.

¹⁰² See TBT Agreement, Preamble, Article 2.4-2.6, and Annex 3, paras. F to H.

The US acknowledges that parties could not agree to include it in the TBT Agreement. United States' first written submission, para. 168.

Panel Report, Colombia - Ports of Entry, para 7.235; Panel Report, India - Quantitative Restrictions, para. 5.128. See also Panel Report, India – Autos, para. 7.270 (a restriction is not the same as a prohibition).

¹⁰⁵ Panel Report, Argentina – Hides and Leather, para. 11.20. See also Panel Report, Brazil – Retreaded Tyres, para. 7.370; Panel Report, Colombia – Ports of Entry, para. 7.240; and GATT Panel Report, Japan – Leather (US II), para. 55.

106 See, e.g., Appellate Body Report, Korea – Various Measures on Beef, para. 164.

- The text of Article 2.2 (in particular, the present conditional "would create") indicates that the risks of non-fulfilment must be taken into account at the time of preparing or applying a technical regulation. The context of Article 2.3 supports this view. Article 2.3 provides that technical regulations "shall not be maintained if the circumstances or objectives giving rise to their adoption no longer exist ...", presupposing that an assessment of the circumstances "giving rise to" the adoption was already conducted in the past.
- The type of factors that Members must consider when conducting this assessment of risks includes "available scientific and technical information". In a case such as the present one, this could include, for example, scientific studies of fishing practices and their effects, a properly conducted statistical sampling of consumer preferences, etc.
- (d) "International standard" under Article 2.4 of the TBT Agreement
- Article 2.4 of the TBT Agreement requires that the Panel interpret, among others, the phrase 5.73 "international standards". Brazil offers comments on the interpretation of the adjective "international" in this phrase, and on the systemic importance of the interpretation of the term "standards".
- *(i)* "International"
- The ordinary meaning of the term "international" suggests that at a minimum, "different nations" must have participated in the adoption of the standards. Relevant context is provided in Annex 1(4) of the TBT Agreement, which defines an "international body or system" as a "[b]ody or system whose membership is open to the relevant bodies of at least all Members". By contrast, Annex 1(5) defines a "regional body or system" as one "whose membership is open to the relevant bodies of only some of the Members". This suggests that an international standard is one adopted by a body "open to the relevant bodies of at least all Members".
- The issue of membership has also been addressed in a Decision of the Committee on Technical Barriers to Trade. The terms of the Decision further indicate that "international standards" under Article 2 are those that emanate from bodies whose membership is open to the relevant bodies of at least all Members. 107
- Other relevant context for interpreting the scope of "international standard" is Article 2.5 of the TBT Agreement. This provision lays down a rebuttable presumption that technical regulations prepared, adopted or applied "in accordance with relevant international standards", do not create an unnecessary obstacle to international trade. The existence of this presumption lends support to the view that for a standard to be "international", and therefore, to form the basis for a presumption of conformity vis-à-vis all WTO Members, it must have been adopted by a body "open to the relevant bodies of at least all Members".
- The systemic importance of the term "standard" (ii)
- There has been limited analysis in the case law of the definition of the term "standard" in 5.77 Annex 1(2) of the TBT Agreement. Yet, this term is a threshold one in the TBT Agreement. Beyond the requirements of Articles 2.4, 2.5 and 2.6, which govern the relationship between international standards and technical regulations, the interpretation of the term "standards" will determine, crucially, the scope of application of the obligations in Article 4, and also, in part, the

¹⁰⁷ Decision of the Committee on Principles for the Development of International Standards, Guides and Recommendations with Relation to Articles 2, 5 and Annex 3 of the Agreement, paras. 3 and 6.

108 See Panel Report, EC – Sardines, para. 7.65.

scope of application of the provisions on conformity assessment in Articles 5 through 9, and transparency in Article 10. Brazil therefore urges the Panel to interpret the term "standards" with extreme care, taking into account the context provided by other provisions of the TBT Agreement, as required by the Vienna Convention on the Law of Treaties.

(e) Article III:4 of the GATT 1994

- 5.78 As panels and the Appellate Body have repeatedly held, a measure violates Article III:4 when it "modifies the conditions of competition in the relevant market to the detriment of imported products". A measure that produces effects by influencing the choices of consumers may modify the conditions of competition to the detriment of imported products.
- 5.79 It is for the complainant to provide evidence that the challenged measures have modified the conditions of competition to the detriment of imported products. This evidence may relate to actual trade effects, but it need not do so.¹¹¹

(f) Likeness

5.80 The parties in this case do not dispute that the products at issue are like. However, the EU suggests that consumer preferences, alone, might cause products to be not like. Brazil does not concur with the view expressed by the EU. The Appellate Body has set out a number of criteria that have been used in analysing likeness. In doing so, it has made clear that no single criterion can be decisive. Instead, panels must establish whether products are like based on "all of the pertinent evidence". 113

D. CANADA

1. Oral statement of Canada

(a) Technical regulation

5.81 The second sentence of the definition of "technical regulation" sets out categories of subject matter that are not examples of "product characteristics". Rather, these categories are independent of those found in the first sentence. Thus, the Appellate Body's requirement that a "technical regulation" must lay down one or more product characteristics should be expanded. The construction of the definition of "technical regulation" supports this interpretation.

5.82 First, the two sentences of the definition each contain a number of categories of subject matter that address different aspects or characteristics of the product at issue. The categories in the first sentence address characteristics intrinsic to the product itself and process or production methods (PPMs) that are related to those intrinsic product characteristics. The categories in the second sentence address characteristics extrinsic to the product; they encompass presentational or informational aspects of the product.

¹⁰⁹ Appellate Body Report, *Korea – Various Measures on Beef*, para. 137 (emphasis removed). *See* also, e.g., GATT Panel Report, *US – Section 337*, para. 5.11.

also, e.g., GATT Panel Report, US – Section 337, para. 5.11.

110 See Appellate Body Report, Korea – Various Measures on Beef, para. 146; and Korea's third party submission, paras. 4-5.

Appellate Body Report, US – FSC (Article 21.5 – EC), para. 215.

European Union's third party submission, para. 29.

¹¹³ Appellate Body Report, *EC – Asbestos*, paras. 101-103 (emphasis in the original).

- 5.83 Second, the use of "also" and "deal exclusively with" at the beginning of the second sentence indicate that the categories covered by the second sentence are additional to, and distinct from those covered by the first sentence.
- 5.84 Third, the use of the word "it" as the subject of the second sentence instead of the word "they" indicates that the subject of the second sentence is the "document" rather than the "product characteristics". Thus, there is no connotation that the subject matter categories covered by the second sentence are in some way related to or constitute "product characteristics".
- 5.85 Finally, the use of the disjunctive word "or" in the first sentence indicates that the first sentence also refers to categories of subject matter that could be separate and distinct from one another. Thus, there are categories within the first sentence, such as some PPMs, that may be the subject of a technical regulation without constituting product characteristics.
- (b) Relevant international standards
- 5.86 For the purposes of Article 2.4 of the TBT Agreement, the documents that may be considered "relevant international standards" should not be limited to only those documents emanating from bodies with recognized activities in standardization
- 5.87 First, the ordinary meanings of the terms "body" and "recognize" in no way limit "recognized bodies" to bodies with particular activities, such as the adoption of standards.
- 5.88 Second, the scope of what constitute "relevant bodies" capable of adopting "international standards" must reflect the fact that the drafters of the TBT Agreement chose the term "relevant bodies" and not alternative terms such as "standardizing bodies" or "standards bodies" that were already defined in the ISO/IEC Guide 2: 1991.
- 5.89 Third, the explanatory note to the definition of "standard" in the TBT Agreement stating that the "[TBT] Agreement covers also documents that are not based on consensus" suggests that the TBT Agreement's definition of "standard" may be broader than just documents adopted by international standardizing bodies.
- 5.90 Finally, a broader interpretation of the types of entities may be considered "recognized bodies" is supported by the object and purpose of the TBT Agreement, which seeks to ensure that Members' technical regulations do not become unnecessary obstacles to international trade or a form of arbitrary or unjustifiable discrimination.
- 5.91 Limiting international "standards" to only those documents emanating from bodies with recognized activities in standardization would be unduly restrictive, particularly in light of this object and purpose.
- 5.92 However, this does not mean that any body may be considered a "recognized body". In instances where the body that has created the alleged standard is not specifically a standardizing body, the Panel should evaluate whether this body constitutes a "recognized body" in accordance with the six principles for the development of international standards espoused in the TBT Committee's Decision on Principles for the Development of International Standards, Guides and Recommendations.

E. CHINA

1. Oral statement of China

- 5.93 China thanks the Panel for giving this opportunity to present its views in this proceeding. China makes this third party oral statement because of its systemic interest in the correct and consistent interpretation of the *General Agreement on Tariffs and Trade 1994* (GATT 1994) and the *Agreement on Technical Barriers to Trade* (the TBT Agreement). China reserves its right to make further comments in the future.
- 5.94 First, China notes that although Mexico held that "the US measures are inconsistent with the fundamental obligations of non-discrimination contained in Articles I:1 ..."¹¹⁴, the United States claimed that its measures "do not accord any advantage to products of any other member that is not also immediately and unconditionally accorded to products of Mexico". ¹¹⁵ Therefore, China would welcome the present Panel's clarification of the issue of whether voluntary labels to origin-neutral conditions constitutes "grant an advantage, favour, privilege or immunity to a product originating in another country" within the meaning of Article I:1 of the GATT 1994.
- 5.95 Secondly, China notes that Mexico asserted that the US failed to base its regulation on the "Relevant International Standard". However, the United States claimed that the definition in the AIDCP is not "standard", "international" or "relevant". Consequently, China would appreciate if the present Panel could clarify the definition of "relevant international standard" in Article 2.4 of the TBT Agreement.
- 5.96 Thirdly, China notes that while Mexico claimed "the association between tuna and dolphins is not unique to the ETP", but also "including the Eastern and Western Atlantic, the Indian Ocean ..." the United States keep a contrary attitude on that. Therefore, China invites the present Panel to address the question whether there are enough evidences demonstrating that ETP (Eastern Tropical Pacific Ocean) is the only place where there is a regular and sustained association between dolphins and yellowfin tuna.
- F. EUROPEAN UNION

1. Third party submission and oral statement of the European Union

- (a) Article III:4 of the GATT 1994: "de facto less favourable treatment"
- 5.97 The European Union considers that the panel should not lightly assume that a measure having an impact on the quantity of products imported from another WTO Member necessarily implies a *de facto* "less favourable treatment". The Appellate Body in *Dominican Republic Import and Sale of Cigarettes* held that detrimental effects of a measure on the quantity of imported products is not sufficient for finding a *de facto* "less favourable treatment" if such effects can be explained "by factors or circumstances unrelated to the foreign origin of the product". The European Union notes that all regulatory measures at least potentially affect existing trade flows or structures. The GATT 1994 and the TBT Agreement are not meant to eliminate regulatory diversity in itself. The Panel needs to examine whether the measure affects the competitive conditions to the detriment of imported products. This requires the Panel to look at a range of criteria.

116 Mexico's executive summary of its first written submission, para. 6.

¹¹⁴ Mexico's executive summary of its first written submission, para. 4.

¹¹⁵ United States' first written submission, para. 120.

¹¹⁷ Appellate Body Report, *Dominican Republic – Import and Sale of Cigarettes*, para. 96.

5.98 In the present case, the European Union notes a number of facts that may support the view that the potential effects of the "dolphin-safe" labelling conditions on tuna imports from Mexico are "unrelated to the foreign origin of the product". First, the United States introduced these labelling conditions in 1990 when its domestic tuna fishing fleet was still using the "setting on dolphins" fishing method. Second, the subsequent commercial decision of certain market actors outside of the United States to maintain this fishing method should not affect the analysis. Third, it seems that tuna imported from several other countries can be designated as "dolphin-safe" in the United States. Thus, any potential negative effects on tuna imported from Mexico would be the result of the fishing method applied and its perceived environmental impact and not the result of the foreign origin of the product. Fourth, should the US assertion that a third of tuna imports from Mexico qualifies for the US "dolphin-safe" label prove accurate, it would provide additional support to the view that the measure does not accord "less favourable treatment" to imported products. Fifth, it seems that the US fishing industry incurred costs when it switched to different fishing methods until around 1994 which Mexico's fishing industry did not incur. The latter advantage could offset potential disadvantages Mexico's fishing industry could potentially suffer if it switched fishing methods today.

(b) Article I:1 of the GATT 1994: the notion of "unconditionally"

5.99 Contrary to Mexico's argument, the European Union considers that Article I:1 GATT 1994 obliges a WTO Member to extend to the products of all WTO Members an advantage granted to the like products of a specific country *only on the same terms and conditions* as those with which the advantage is granted to the products of that country. Otherwise, the WTO Member would be obliged to grant to the like products of all WTO Members *more advantages* than the advantage granted to the products of the specific country, i.e., that country would receive the advantage only for its products that satisfy the conditions, while all WTO Members would receive the advantage even for like products that do not satisfy the conditions. This view is supported by the panel report in *Canada – Autos* which stated that an advantage can be granted subject to conditions without necessarily implying that it is not accorded "unconditionally" to the like product of other Members. In the present case, the United States would comply with its obligations under Article I:1 vis-à-vis Mexico if it allowed Mexico's tuna the possibility to carry the "dolphin-safe" label on the same terms and conditions as those imposed on the tuna of other countries.

(c) Article I:1 of the GATT 1994: the notion of "advantage, favour, privilege or immunity"

5.100 In the European Union's view, it is not clear whether subjecting the use of the voluntary "dolphin-safe" label to specific labelling conditions constitutes "granting an advantage, favour, privilege or immunity to a product originating in another country". The European Union notes that the use of the label depends on the free choice of market operators and that the labelling conditions do not seem to differentiate on the basis of the origin of the goods.

(d) Articles I:1 and III:4 of the GATT 1994: the notion of "like products"

5.101 The European Union considers that the facts of this case may indicate that, to consumers' minds and purchasing patterns, tuna that has been fished in a manner that harms dolphins may be considered as a different product from tuna that has been fished in a manner that does not harm dolphins. The segment of consumer demand for "dolphin-safe" tuna appears to be very important and to probably outweigh by far the segment of consumer demand for "non-dolphin-safe" tuna. Mexico recognizes that only "dolphin-safe" tuna can be commercialised successfully in the United States. This is the reason why Mexico wishes to have tuna fished through "setting on dolphins" labelled as "dolphin-safe".

¹¹⁸ See Panel Report, *Canada – Autos*, para. 10.24.

- 5.102 If, on the basis of its analysis of the facts, the Panel reaches the conclusion that consumers tend to perceive "dolphin-safe" tuna as being different from "non-dolphin-safe" tuna and that their consumption patterns confirm this difference in the perceptions, then the Panel should draw the logical conclusion that one of the four criteria for the identification of "like products" has not been satisfied.
- (e) Annex 1.1 of the TBT Agreement: labelling requirements and product characteristics
- 5.103 The question whether the US labelling scheme is a "technical regulation" within the meaning of Annex 1.1 of the TBT Agreement raises several interpretative issues.
- 5.104 The first is the relationship between the first and the second sentence of Annex 1.1 of the TBT Agreement. The European Union would note that it seems uncontested that the US labelling scheme is a document dealing with "labelling requirements as they apply to a product" within the meaning of the second sentence of Annex 1.1. The parties present diverging views on whether the matters addressed in the second sentence are examples of "product characteristics" within the meaning of the first sentence (Mexico) or alternatives thereto (United States). In the view of the European Union, both interpretative approaches would seem to lead to the same conclusion, i.e. that a document dealing with "labelling requirements" (if it applies to identifiable products and is mandatory) is a technical regulation. The wording of Annex 1.1 of the TBT Agreement indicates that the term "technical regulations" comprises documents dealing with three categories of subject matter: (1) documents laying down "product characteristics", (2) documents laying down "processes and production methods" relating to product characteristics and (3) documents including or dealing exclusively with "terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method". In the view of the European Union, it seems irrelevant whether subject matter (3) is considered to be a sub-set of subject matter (1), as suggested by the Appellate Body in EC – Asbestos, or – with a view to the use of the words "it" and "also" in the second sentence – an alternative category of subject matter which equally qualifies a document as a "technical regulation".
- 5.105 The second issue is whether "labelling requirements" must have a certain substantive content in order to qualify the document as a "technical regulation". Japan suggested that a panel must, in addition to finding that a document deals with "labelling requirements as they apply to a product", also determine whether the labelling requirement lays down "product characteristics". The European Union finds it difficult to square this interpretation with the wording of Annex 1.1 of the TBT Agreement. The second sentence of Annex 1.1 only requires that "labelling requirements" "apply to a product, process or production method", and not that they lay down product characteristics. Such an additional substantive requirement does not seem to follow from the context within Annex 1.1, either. If "labelling requirements" are seen as an independent subject matter, the term "product characteristics" of the first sentence does not apply in the second sentence. If "labelling requirements" are considered to be examples of "product characteristics", a document dealing with "labelling requirements" will be a "document which lays down product characteristics" within the meaning of the first sentence, without the need to examine its substantive content.
- (f) Annex 1.1 of the TBT Agreement: mandatory compliance
- 5.106 The question is whether compliance with labelling requirements contained in a law is "mandatory" within the meaning of Annex 1.1 of the TBT Agreement if the labelling scheme leaves economic operators the choice to market a product with or without the label, but obliges them to comply with the labelling conditions once they choose to use the label. In the view of the European Union, the following considerations are pertinent.

- 5.107 With regard to the *ordinary meaning* of the term "mandatory", compliance with labelling requirements is "mandatory" if the labelling requirements are "compulsory", regulated "in a binding or compulsory fashion" or effectively "prescribed or imposed". The European Union would be reticent to deduce the "mandatory" nature of the US labelling scheme from the mere fact that economic operators in the United States are prohibited "by law" from using the label if they do not meet its conditions. Exclusively focusing on the form of the labelling scheme would appear formalistic since the US labelling scheme is optional in the sense that it leaves economic operators the freedom to choose whether they want to use the label or not. Whereas the conditions under which economic operators may bear the label are binding, the use of the US labelling scheme is not compulsory. In that sense, the measures at issue differ from labelling provisions which a law forces economic operators to use for certain specified products.
- 5.108 In the view of the European Union, the measures at issue differ significantly from those addressed in *EC Sardines*. Whereas a product could not be marketed as "preserved sardines" under the EC Regulation if it contained fish from species other than *Sardina pilchardus*, the US labelling scheme does not prevent economic operators from marketing their products as "tuna" irrespective of whether the fish was harvested in accordance with the "dolphin-safe" criteria of the DPCIA or not. Consumers in the United States can identify tuna that was not harvested in accordance with the DPCIA as "tuna".
- 5.109 The European Union would also caution against including within the scope of the term "mandatory" labelling schemes which Mexico considers to be "de facto mandatory" because of market conditions. Consumers in a given market may prefer, or even overwhelmingly prefer, purchasing products that bear a certain label. But such consumer preferences do not make the use of the label "compulsory" or effectively "prescribed or imposed". Certain legal concepts in the WTO Agreements may extend to de facto situations. The problem with Mexico's proposed concept of "de facto mandatory" labelling schemes is, however, that it is not based on the design or structure of the measure, but on consumer preferences in the US market. As these factors are entirely outside the control of the US legislator, the European Union would be reticent to extend the term "mandatory" to such scenarios (i.e. consumer preferences) without any clear textual guidance from WTO provisions.
- 5.110 Within the *context* of Annex 1.1 of the TBT Agreement, Annex 1.2 of the TBT Agreement defines a standard as a "document ... with which compliance is not mandatory" and which may deal with "labelling requirements". The European Union would note that standards inherently contain rules, guidelines or characteristics to be met by those adhering to the standard. If the requirement to meet the specifications of a standard rendered the standard "mandatory", the definition of standards would become devoid of effectiveness. Standards would, rather, be "voluntary" as long as economic operators can choose whether they want to follow a particular standard or not. Thus, the mere fact that economic operators choosing to use the US labelling scheme must comply with its conditions does not necessarily render the labelling scheme mandatory. Another contextual argument follows from Explanatory Note to Annex 1.2 of the TBT Agreement which characterizes "technical regulations as mandatory documents". This could indicate that technical regulations are documents the application or use of which is mandatory. Finally, the TBT Committee decision of 1995 setting out that "all mandatory labelling requirements" must be notified as technical regulations under Article 2.9 of the TBT Agreement¹¹⁹ could be relevant context within the meaning of Article 31(3)(b) of the Vienna Convention on the Law of Treaties. It is likely that the TBT Committee understood "mandatory labelling requirements" in the same way as the GATT Secretariat which had stated in 1992 that

¹¹⁹ See Decisions and Recommendations adopted by the WTO Committee on Technical Barriers to Trade since 1 January 1995, G/TBT/1/Rev. 9 of 8 September 2008, p. 17.

labelling requirements need to be notified if "it is mandatory for a product to bear a label if it is to enter the domestic market". 120

5.111 The European Union considers that the *object and purpose* of the TBT Agreement of reducing technical barriers to trade does not mandate an overly broad interpretation of the term "mandatory". The TBT Agreement only covers measures that fall within the scope of one of three categories "technical regulations", "standards" or "conformity assessment procedures" which are defined with some detail. Even if a measure is no technical regulation because it is "voluntary", it may still be a standard. It may also have to comply with the GATT 1994.

G. GUATEMALA

1. Oral statement of Guatemala

5.112 At the outset, on behalf of the delegation of Guatemala, I would like express our appreciation for this opportunity to present our views on the matters at issue.

(a) General remarks

- 5.113 Mexico claims that the essence of the current dispute relates to the prohibition of the use of a US "dolphin-safe" label on imports of tuna products from Mexico, while such a label is permitted to be used on tuna products from other countries, including products from the United States. ¹²¹ According to Mexico, this prohibition adversely affects the importation of Mexican tuna and tuna products and its internal sale and distribution in the US market. ¹²² Moreover, Mexico argues that without a "dolphin-safe" label, the ability to sell Mexican tuna products into the US market is restricted. ¹²³
- 5.114 In accordance with the US regulations at issue, tuna products may not be labelled "dolphin-safe" if they contain tuna that was caught during a fishing trip in which purse seine nets were intentionally deployed on or used to encircle dolphins or dolphins were killed or seriously injured in the set in which the tuna was caught.¹²⁴
- 5.115 Mexico vessels catch tuna using the "purse-seine" net fishing method (also known as "setting on dolphins"). Mexico asserts that this method is a sound and environmentally 1 sustainable method for fishing tuna. According to Mexico, the efforts of the Mexican fishing fleet contributed substantially to the reduction of dolphin mortality to the levels less than 0.004%. 126
- 5.116 On the contrary, the United States submits that the "purse-seine" net fishing method has important adverse effects on dolphins. Among other arguments, the United States submits that this method is harmful to dolphins because: dolphins are chased for up to 90 minutes before vessels are able to surround the dolphins and the tuna below; dolphins may drown when trapped under the net; or may be subjected to serious injury or death through entanglement in the net. Moreover, dolphin mothers and nursing calves are often separated during the high-speed chase and encirclement, and

TBT/Spec/23, quoted in: Negotiating History of the Coverage of the Agreement on Technical Barriers to Trade with Regard to Labelling Requirements, Voluntary Standards, and Processes and Production Methods Unrelated to Product Characteristics, WT/CTE/W/10, G/TBT/W/11 of 29 August 1995, para. 19.

¹²¹ Mexico's first written submission, para. 4.

¹²² Mexico's first written submission, para. 6.

¹²³ Mexico's first written submission, para. 8.

¹²⁴ United States' first written submission, para. 14.

¹²⁵ Mexico's first written submission, para. 7.

¹²⁶ Mexico's first written submission, para. 109.

these calves die as a result of starvation, predation and other causes, even when their mothers are released from sets alive.

- 5.117 The United States agrees with Mexico that the magnitude of observed dolphin mortality due to setting on dolphins with purse seine nets has decreased significantly. However, according to the US, observed mortality and serious injury fail to measure the full impact that setting on dolphins to catch tuna has on dolphins.¹²⁷
- 5.118 The United States also submits that the US measures at issue apply to tuna caught in any fishery where there is a regular and significant association between tuna and dolphins and apply regardless of the origin of the tuna.
- 5.119 Guatemala understands that the essence of this dispute is not a prohibition on the use of a dolphin-safe label on tuna products from Mexico, as Mexico appears to characterize it. Guatemala believes that the problem is of a factual nature. While Mexico believes that its fishing method (i.e. setting on dolphins) is a sound and environmentally sustainable method for fishing tuna, the United States, on the contrary, considers that such method has adverse effects on dolphins and, consequently, the use of a dolphin-safe label would mislead consumers and encourage fishing practices that the US considers prejudicial to dolphins.
- 5.120 Both Parties have submitted evidence supporting their respective positions in this regard. Some of the evidence includes inconclusive studies. While Guatemala has not a final position on this issue, my delegation recognizes that the task before the Panel is not simple at all. In any event, Guatemala favours a solution that is both, consistent with the WTO Agreements and respectful of the environment.
- 5.121 Now, I would like to make some brief comments on the legal arguments submitted by the parties.
- (b) Claims under Articles I:1 and III:4 of the GATT of 1994
- 5.122 Regarding Mexico's claims on Articles I:1 and III:4 of the GATT of 1994, Guatemala observes that Mexico recognizes that the US measures do not, on their face, discriminate "on the basis of the foreign country that is the source of particular tuna and tuna products". Rather, Mexico says, "they discriminate on the basis of where the tuna is harvested and the fishing method". 128 Mexico adds that "the less favourable treatment at issue is of a *de facto* rather than *de jure* character". 129
- 5.123 In the view of Guatemala, a *de facto* claim implies a higher burden of proof for the complaining Party. In this particular case, Guatemala believes that Mexico should demonstrate better, among other aspects, how the competitive opportunities have been *de facto* modified as a consequence of the measures at issue and why the fishing method used by Mexican vessels is equivalent or better than the method used by US vessels (both, in terms of dolphin and other marine mammals protection).
- 5.124 With respect to the place where the tuna is harvested, both Mexico and the United States agree that, for reasons that are not fully understood by scientists, there is a natural association between tuna and dolphins. However, the United States contests that, contrary to Mexico's assertions, the

¹²⁷ United States' first written submission, paras. 41, 53, 54 and 55.

¹²⁸ Mexico's first written submission, para. 164 and 185.

¹²⁹ Mexico's first written submission, para. 164 and 185.

¹³⁰ Mexico's first written submission, para. 11. United States' first written submission, para. 37.

Eastern Tropical Pacific (ETP) is fundamentally different from all other oceans in that it is the only ocean where tuna and dolphins have a known regular and significant association. ¹³¹

- 5.125 On this particular issue, Guatemala does not understand why the US measures contain different provisions for fisheries in different oceans. Even if it were true that association between dolphins and tuna only occurs in the ETP, Guatemala does not see the need to have different regulations for fisheries in different oceans. The basic requirement is that a tuna product may not be labelled "dolphin-safe" if it contains tuna that was caught during a fishing trip in which purse seine nets were intentionally deployed on or used to encircle dolphins or dolphins were killed or seriously injured in the set in which the tuna was caught. If there is no association, then there is no possibility that purse seine nets be intentionally deployed on or used to encircle dolphins.
- 5.126 Therefore, Guatemala considers that the United States should make an effort to demonstrate that the design of its law was not intended to make a differentiation in treatment. Specially, in the light of some inconclusive studies regarding the possible association between dolphins and tuna in other oceans besides the ETP.
- (c) Claims under the TBT Agreement
- 5.127 In addition, there is disagreement between the parties as to whether the US measures at issue are to be considered as technical regulation or as a standard.
- 5.128 Guatemala understands that the arguments turn around the mandatory or voluntary character of these regulations, among others.
- 5.129 Guatemala observes that the US "dolphin-safe" labelling measures do not require tuna products to be labelled or to indicate on the label the fishing method utilized. Guatemala also observes that these measures do not prevent the sale in the United States, even if the use of a fishing method prevents the use of a "dolphin-safe" label.
- 5.130 In addition, Mexico makes the argument that the US "dolphin-safe" labelling measures, if not *a priori* mandatory, are nonetheless *de facto* mandatory "because the marketing conditions in the US are such that it is impossible to effectively market and sell tuna products without a dolphin-safe designation". ¹³²
- 5.131 Guatemala is not persuaded that there is a possibility that a standard, because of its effects, can be considered as a technical regulation. In any event, Guatemala is of the view that Mexico does not appear to have demonstrated that the measure is *de facto* mandatory.
- 5.132 Moreover, Guatemala coincides with Australia that the actions of private individuals (including consumers) cannot alter the nature of a measure. In any event, if Mexico would like to pursue the claim that the measure is *de facto* mandatory, Mexico should demonstrate that the mandatory nature of a measure is attributable to governmental actions; not to decisions by private individuals.
- 5.133 Finally, Mr Chairman, Guatemala observes that the United States argued that mandatory labelling requirements and voluntary labelling requirements are set out in separate provisions of the TBT Agreement. While mandatory labelling requirements must be notified under Article 2.9 of the

¹³¹ United States' first written submission, para. 38.

¹³² Mexico's first written submission, para. 203.

TBT Agreement, which addresses technical regulations, voluntary labelling requirements must be published under Paragraph L of Annex 3 of the TBT Agreement, which addresses standards. ¹³³

- 5.134 Guatemala recognizes that notifications in the WTO do not prejudge the legal character of a measure. However, Guatemala notes that the United States appears to have notified the regulations at issue under Article 2.9.2 of the TBT Agreement. Guatemala does not know whether notification G/TBT/Notif.00/5 of 10 January 2000 refers specifically to the measures at issue. If it does, Guatemala wonders why the US would notify a standard under the provisions foreseen for notification of technical regulations. Did the United Stated consider at that time that these measures were mandatory?
- H. JAPAN

1. Third party submission of Japan

- (a) Introduction
- 5.135 Japan's third party submission focuses on issues of principle, rather than factual issues. This submission is limited to four issues before the Panel: factors determining whether the measure at issue is a "technical regulation" as defined in Annex 1 of the Agreement on Technical Barriers to Trade ("TBT Agreement"); the proper interpretation and application of Article 2.2 of the TBT Agreement; factors relevant to determining whether an "international standard" exists for the purposes of Article 2.4 of the TBT Agreement; and issues regarding "less favourable treatment" under Article III:4 of the General Agreement on Tariffs and Trade 1994 ("GATT 1994").
- (b) The Panel should carefully apply the threshold definition of "technical regulation" in Annex 1 of the TBT Agreement
- 5.136 To be a "technical regulation", a document must fall within the definition in Annex 1 of the TBT Agreement. As the Appellate Body stated in EC-Sardines, a document must: (i) apply to an identifiable product or group of products; (ii) lay down one or more characteristics of the product (intrinsic or related; positive or negative); and (iii) compliance with the product characteristics must be mandatory.
- (i) The Panel should follow precedents regarding whether the measures at issue are "documents ... with which compliance is mandatory"
- 5.137 Mexico has presented only summary allegations regarding whether the Dolphin Protection Consumer Information Act (DPCIA) is a "document with which compliance is mandatory". To determine whether the US measures at issue are "technical regulations" for the purpose of Annex 1 of the TBT Agreement, the Panel should apply the criteria in Annex 1, as interpreted in *EC –Sardines –* that is, it should determine whether these measures have the effect of prescribing or imposing product characteristics, such as features, qualities, attributes, or other distinguishing marks.
- 5.138 As the Appellate Body correctly noted in EC Asbestos, a technical regulation can specify product characteristics in either a positive or negative form. In EC Sardines, the Panel reasoned that by positively requiring the use of $Sardina\ pilchardus$ in products with the trade description "sardines", the EC regulation at issue also laid down product characteristics in a negative form for all other fish, by excluding them from being marketed as "sardines". Japan would like to note that the key issue before the Panel is whether the DPCIA and the other measures at issue legally prohibit the marketing

¹³³ United States' first written submission, para. 137.

of certain tuna products under a particular trade description, even if they are entitled to use that trade description, or even if their use of that trade description is legitimate. The Panel should examine carefully whether the US measures at issue merely require the "dolphin-safe" indication on a label to be truthful, or whether they preclude use of the "dolphin-safe" indication for tuna products that were in fact harvested in a dolphin-safe manner.

- (ii) The Panel should carefully consider whether the measures at issue are a "document which lays down product characteristics"
- 5.139 Mexico has made the blanket assertion that "a label on a product is a product characteristic", arguing that the US measures "govern the conditions under which a product can be labelled as 'dolphin-safe'" and that this requirement is a "product characteristic". But the analysis seen in the prior TBT disputes to date contradicts such position; in determining whether a measure was a technical regulation, they found that the mandatory labelling requirements at issue were technical regulations *because* these requirements laid down product characteristics. The Panel cannot simply determine that the measure is a labelling requirement and stop its analysis there.
- 5.140 Japan observes that the current dispute is the first to apply the TBT Agreement to labelling of a product in respect of the conditions under which the product or its content was produced, where those conditions do not necessarily affect the product as such. Therefore the Panel should carefully consider to what extent it should mechanically apply the findings in prior cases even if the facts and circumstances of this case do not match with those findings.
- (c) The Panel should treat a Member's identification of its "legitimate objectives" deferentially
- 5.141 Japan shares the view in paragraph 7 of the US first written submission that "It is not for Mexico to choose for the United States which legitimate objectives it should pursue." The TBT Agreement, like the GATT 1994, accords a degree of deference with respect to the domestic policy objectives which Members wish to pursue.
- 5.142 A party claiming a violation of Article 2.2 bears the burden of proof in respect of all elements of its claim. However, Mexico has not addressed the issue of the DPCIA as a truth-in-labelling measure for the prevention of deceptive practices. Instead, Mexico argues that the US measures do not fulfil a legitimate objective because they "do not protect animal life or health or the environment in the general sense", and because they favour dolphins over other marine species.
- 5.143 Article 2.2 concerns the choice of means used to accomplish a legitimate objective. Article 2.2 does not authorize a complaining party or the panel to substitute its own policy priorities for those of the responding party. Nor does Article 2.2 authorize a panel to rule that the responding party's objectives are not legitimate because they do not match those of the complaining party or the panel.
- (d) The Panel should take into account the principles regarding "international standards" agreed by the Committee on Technical Barriers to Trade (TBT Committee)
- 5.144 Mexico also asserts that the US measures are inconsistent with Article 2.4 of the TBT Agreement because they are not based on the "AIDCP Standard", two definitions appearing in a Resolution adopted on 20 June 2001 by the Parties to the Agreement on the International Dolphin Conservation Program (AIDCP).
- 5.145 Japan submits that in this connection, the Panel should draw on the guidance provided by the TBT Committee's Decision on "Principles for the Development of International Standards, Guides and Recommendations with Relation to Articles 2, 5 and Annex 3 of the Agreement"

(G/TBT/1/Rev.9, Annex B)which makes it clear that membership of an international standardizing body should be open on a non-discriminatory basis to relevant bodies of at least all WTO Members and at every stage of standards development. No purported international standard should be recognized as such if the principles recognized in the Decision were disregarded in its elaboration.

- (e) The Panel should scrutinize carefully whether the measures at issue accord less favourable treatment of imported products within the meaning of Article III:4 of the GATT 1994
- 5.146 Mexico has also claimed that the US measure is inconsistent with Article III:4 of the GATT 1994, and bears the burden of proving that claim. The US response to this claim focuses entirely on arguing that Mexican tuna and tuna products are not accorded "less favourable treatment", asserting that the measures at issue do not modify the conditions of competition because all tuna is subject to the same conditions on use of the dolphin-safe label. However, even measures that apply facially identical treatment to domestic and imported like products may still discriminate *de facto* in violation of Article III:4. Japan notes that the US measures may block certain Mexican tuna products from using the words "dolphin-safe" or similar wording even if tuna harvested by purse-seine fishing is, at least in Mexico's view, not harming the dolphin population.

(f) Conclusion

5.147 Japan appreciates the opportunity to present its views in this dispute between Mexico and the United States, and respectfully requests the Panel to consider these views of a systemic nature in examining the facts and arguments presented by the parties.

2. Oral statement of Japan

- (a) Introduction
- 5.148 Japan's oral statement focuses on three points that the Panel (1) must respect the distinction in the TBT Agreement between mandatory technical regulations and voluntary measures; (2) should examine the negotiating history of the TBT Agreement with care and as a whole, (3) should be aware of the past practice in terms of order of analysis; and (4) the Panel should not read procedural requirement in Article 2.2 of the TBT Agreement.
- (b) The Panel must respect the distinction between mandatory and voluntary measures
- 5.149 The TBT Agreement provides separate and distinct obligations applying to *mandatory* technical regulations on the one hand, and to *voluntary* standards on the other. The distinctive characteristic of a "standard" is that private persons may choose freely whether to use the standard at all, but if they use the standard, they must conform to the specifications that the standard provides. The use the voluntary standard without meeting the specification of the standard may constitute deceptive description. However, the fact that a Member has a law prescribing that a deceptive description is liable does not mean that all of the voluntary product standards in that Member's territory are suddenly mandatory technical regulations. The Panel must respect the distinction in the TBT Agreement between mandatory technical regulations and voluntary standards.
- (c) The Panel must evaluate the TBT Agreement's negotiating history with care, and as a whole
- 5.150 In discussing the TBT Agreement's application to labelling requirements, the US first written submission (in footnote 141) relies in part on a selective and unbalanced reading of the negotiating history of the TBT Agreement. The negotiating history simply demonstrates that on these fundamental issues, a variety of views were expressed during the Uruguay Round, as they remain divided today. In order to interpret the provision on which there is no established interpretation, the

Panel can only follow the rules of interpretation in Article 31 of the Vienna Convention on the Law of Treaties, and have recourse to the preparatory work to confirm the meaning resulting from the application of Article 31 or to determine the meaning when such a reading leaves the meaning ambiguous or obscure. If the Panel does use the preparatory work, it must use it not selectively, but as a whole.

- (d) The Panel should be aware of the past practice in terms of the order of analysis
- 5.151 The Appellate Body in EC Bananas III first addressed the issue of the order of the analysis. It found (in paragraph 204) that Article X:3(a) of the GATT 1994 and Article 1.3 of the Agreement on Import Licensing both applied to the measures at issue, but that the panel in that dispute "should have applied the Licensing Agreement first, since this agreement dealt specifically, and in detail, with the administration of import licensing procedures". In EC Asbestos, both the panel and the Appellate Body examined Canada's TBT claims first. The same practice should be followed in this case.
- (e) The Panel should not read procedural requirement in Article 2.2 of the TBT Agreement
- 5.152 Brazil argues that the phrase in Article 2.2 of the TBT Agreement, "taking account of the risks non-fulfilment would create" read together with the last sentence of Article 2.2 imposes a positive procedural requirement that a Member proposing a technical regulation must *first* assess the risks that would arise if the legitimate objective were not fulfilled. In addition, in order to support its argument, Brazil suggests that Articles 5.1 and 5.2 of the Agreement on the Application of Sanitary and Phytosanitary Measures("SPS Agreement") provide the right model for Article 2.2 of the TBT Agreement.
- 5.153 Japan is of the opinion that Brazil's arguments are not in line with the text of the TBT Agreement because Article 2.2 simply requires that the technical regulation shall not be trade restrictive than necessary to fulfil a legitimate objective, taking account of the risks non-fulfilment of the legitimate objective would create. It does not require that a Member proposing a technical regulation conduct a prior risk assessment of the type argued by Brazil.
- 5.154 As for the contextual argument in respect of the SPS Agreement, Articles 5.1 and 5.2 of the Agreement do not support the Brazil's argument because of the clear differences of languages between Article 5.1 of the SPS Agreement and Article 2.2 of the TBT Agreement. Japan urges the Panel to take a cautious approach to interpreting Article 2.2, and refrain from adding an obligation not provided in the covered agreements.
- I. KOREA

1. Third party submission and oral statement of Korea

- 5.155 Korea addresses in its submission and oral statement (1) Mexico's Article III:4 of the GATT 1994 claim; and (2) Article 2 of the TBT Agreement with regard to the interpretation of technical regulation.
- (a) Mexico's Article III:4 of the GATT1994 claim and trade effect
- 5.156 Mexico states that the United States Dolphin Safe measures "do not, on their face, discriminate on the basis of the foreign country that is the source of particular tuna and tuna products. Rather, they discriminate on the basis of where the tuna is harvested and the fishing method". Thus, Mexico's Article III:4 claim, is a *de facto* claim, that the US measures have the effect of favouring US tuna and tuna products over like Mexican tuna and tuna products.

- 5.157 Both party agree that in order to determine whether the United States measures at issue are in violation of Article III:4 of the GATT 1994, the Panel must consider whether these measures "modif[y] the conditions of competition in the relevant market to the detriment of imported products". Korea also notes that a measure, while origin-neutral on its face, may modify the conditions of competition in the relevant market to the detriment of imported products, which means a *de facto* discrimination.
- 5.158 One of the leading Appellate Body cases in the area of *de facto* Article III:3 violation claim is *Korea Various Measures on Beef*. While the circumstances and the structure of the measures examined by the Appellate Body in that case are different from the US measures at issue in this case, the reasoning formulated by the Appellate Body in *Korea Various Measures on Beef* is equally applicable to this case.
- 5.159 The critical factor in *Korea Various Measures on Beef* was the Appellate Body's analysis of the "thrust and effect" of measures at issue. The Appellate Body added that when the "effect" of the measure manifested in a substantial disparity, the intervention of some element of private choice does not relieve of responsibility under the GATT 1994 for the resulting establishment of competitive conditions less favourable for the imported product than for the domestic product.
- 5.160 Based on the foregoing, in order to determine whether the United States' Dolphin Safe measures constitute a *de facto* violation of Article III:4 of the GATT 1994, the Panel should ascertain what was the "thrust and effect" of the adoption of these measures. And, to properly carry out such an analysis, the Panel should inquire as to the prevailing market conditions for Mexican tuna and imported tuna just prior to the introduction of the Dolphin Safe measures and how the introduction of such measures affected the sales of these two like products in the US market.

(b) Technical regulation of Article 2 of the TBT Agreement

- 5.161 Both parties agree that Annex 1.1 of TBT Agreement prescribes a three part test for determining whether the measure at issue constitutes a technical regulation: (1) it must apply to an identifiable product or group of products; (2) it must lay down one or more characteristics of the product; (3) compliance with the product characteristic must be mandatory. In this case, there is no disagreement that the measures in question satisfy the first criterion. The dispute is over the second and third elements of Annex 1.1: (1) whether the labelling requirement at issue constitutes a "product characteristic"; and (2) whether compliance with same is mandatory.
- 5.162 Korea submits that the second sentence of Annex 1.1 should not be interpreted as providing examples of "product characteristics" mentioned in the first sentence. As a grammatical matter, the singular "it" used in the second sentence of Annex 1.1 cannot refer back to the plural "product characteristics". Korea, however, does not agree with the view that the second sentence of Annex 1.1 should be interpreted as adding other "aspects" into the first sentence of Annex 1.1.
- 5.163 In Korea's view, the second sentence should instead be interpreted as providing further elaboration on how the first sentence should be interpreted. In other words, rather than broadening the range of measures that may fall under the definition of "technical regulation," it notes that certain measures that deal with "terminology, symbols, packaging, marking or labelling requirements" may have the result of laying down product characteristics or their related processes and production methods. As the Appellate Body pointed out in *EC Asbestos* and *EC Sardines*, the term 'technical regulation" is not limited only to "feature and qualities intrinsic to the product itself, but may also related 'characteristics', such as the means of identification, the presentation and the appearance of a product".

- 5.164 Mexico claims that the compliance with the United States Dolphin Safe regime is mandatory because "it is unlawful to include on the label of any tuna product offered for sale in the United States the term 'dolphin-safe' or an analogous term or symbol if the product contains tuna harvested in the ETP by a large purse-seine vessel using a purse-seine net intentionally deployed on or to encircle dolphins". Korea is of the view that Mexico misconstrues the mandatory requirement set forth under Annex 1.1 because the critical inquiry there is not whether the label requirement at issue set forth mandatory components, but rather whether the obligation to affix a label regarding product, process or production is mandatory. The US Dolphin Safe measures do require that certain conditions must be met in order to affix the "Dolphin Safe" label, but affixing such label is not mandatory under the US scheme and as such does not meet the third element of technical regulation.
- 5.165 Mexico also claims that "[e]ven if the labelling scheme were not to be considered *a priori* mandatory, it is *de facto* mandatory because the market conditions in the United States are such that it is impossible to effectively market and sell tuna products without the dolphin-safe designation". Korea notes that there is no indication in the text of Annex 1.1 of the TBT Agreement that members intended to inject a *de facto* mandatory requirement element to the concept of technical regulations. Moreover, even if that were to be the case, in Korea's view, Mexico appears not to be successful in adducing evidence relevant to support its claim that "it is impossible to effectively market and sell tuna products without the dolphin-safe label designation".
- J. NEW ZEALAND

1. Third party submission of New Zealand

- (a) Introduction
- 5.166 New Zealand's participation as a third party in this dispute reflects its systemic interest in the legal issues arising from the United States' measures that stipulate how tuna and tuna products qualify for a dolphin-safe label for use in the United States. As an exporting nation, the proper implementation of the WTO Agreement on Technical Barriers to Trade (the TBT Agreement) is of fundamental importance to New Zealand. The potential for technical regulations, standards and conformity assessment procedures to constitute unnecessary barriers to trade is of great concern.
- 5.167 New Zealand recognizes the rights of WTO Members to take measures necessary to ensure, *inter alia*, the protection of animal health or the environment, or for the prevention of deceptive practices, at the levels they consider appropriate. However, any such measures are subject to the disciplines of the WTO Agreements. With respect to technical regulations and standards, and procedures for assessment of conformity with technical regulations and standards, New Zealand notes the particular importance the TBT Agreement places on ensuring that such measures do not create unnecessary obstacles to trade.
- (b) The definition of technical regulation: Annex 1.1
- (i) Coverage of labelling requirements
- 5.168 New Zealand considers that Annex 1.1 of the TBT Agreement provides for a wide range of labelling measures to fall within the definition of "technical regulation". Such measures have the potential to have a significant impact on trade, and allowing a large body of potentially trade restrictive measures to escape the strictest disciplines of the TBT Agreement (those disciplines applicable to technical regulations) would undermine the objectives of the Agreement.
- 5.169 The definition of "technical regulation" is set out in Annex 1.1 of the TBT Agreement. Mexico and the United States differ in their interpretation of this provision, particularly with regard to

the meaning of "product characteristic" and the relationship between the two sentences of Annex 1.1. New Zealand considers that the Appellate Body's decisions in EC – Asbestos and EC – Sardines support a broad interpretation of "product characteristic", and therefore, of the term "technical regulation".

- (ii) Is the United States' measure a "mandatory" regulation?
- 5.170 New Zealand notes that there is a distinction between a mandatory requirement to label (where a product must be labelled in order to be sold in a market) and a mandatory labelling requirement (such as here, where regulations prescribe requirements that are mandatory in the event that a producer wishes to use a particular label). The measure in question in the present dispute is an example of the latter type of requirement: there is no formal legal requirement that tuna be labelled "dolphin-safe" in order to be sold on the United States market. However, if such a label is used, it must comply with certain conditions as set out in the Dolphin Protection and Conservation Information Act. The measure is therefore not *a priori* mandatory.
- 5.171 Annex 1.1 makes no mention of whether measures must be *a priori* mandatory, or whether a *de facto* mandatory measure will qualify as a technical regulation. However, the TBT Agreement divides the measures covered into those that are mandatory (technical regulations), and those that are not (standards). Regulations and standards are subject to different obligations.
- 5.172 Given this difference, it is central to the application of the Agreement that a distinction be maintained between mandatory and voluntary measures. In New Zealand's view, therefore, a measure that is not *a priori* mandatory will only constitute a "technical regulation" in cases where it is clearly warranted by the facts. For example, there may be situations where a Government's conduct is such that it essentially makes compliance obligatory with what would otherwise be a 'standard' under the TBT Agreement. New Zealand submits that the Panel should find that such an assessment would require a close examination of the facts of each situation to ensure that the principles of the TBT Agreement are being upheld.
- (c) The definition of relevant international standard under Article 2.4
- 5.173 Pursuant to Article 2.4 of the TBT Agreement, Members must use international standards as a basis for their technical regulations except where such standards would be an "ineffective or inappropriate means for the fulfilment of the legitimate objectives pursued, for instance because of fundamental climatic or geographical factors or fundamental technological problems".
- 5.174 New Zealand notes that the management of living marine resources is subject to a specialized international legal regime. It considers that Regional Fisheries Management Organisations/Arrangements are the appropriate bodies for managing shared fisheries, and reaffirms the ability of countries to enact trade-related measures based on the decisions of those bodies.
- 5.175 The TBT Agreement does not include a definition of the term "international standard". However, in accordance with relevant and related definitions used in the TBT Agreement and relevant ISO guidelines, New Zealand considers that, in order to constitute an "international standard" under Article 2.4, a standard must be made by an international body. Further, that international body must fulfil two criteria. First, it must be an international body whose specified activities include standardization. Second, it must be open to at least all members of the WTO in order to provide for their participation in the preparation and establishment of standards.

- (i) The requirement that the standard be made by an international body
- 5.176 The conclusion that an international standard must be made by an international body derives from the definition of "standard" in Annex 1.2 of the TBT Agreement, in particular the requirement that it be a "document approved by a *recognized body*" [emphasis added]. The definition of "standard" is elaborated upon in an explanatory note to Annex 1.2, which refers to the topic of international standards. It provides that "standards prepared by the *international standardization community* are based on consensus ..." [emphasis added]. An inference may be drawn from this explanatory note that in the context of international standards as contemplated in Article 2.4, the recognized body in question must be an international one. The concept of a "standardizing body" is also internationalized in other provisions of the TBT Agreement, including in Article 2.6 and the Code of Good Practice in Annex 3, which use the term "international standardizing body".
- 5.177 In light of Article 31(1) of the Vienna Convention on the Law of Treaties, the ordinary meaning of the words "international standard" in Article 2.4 must be considered in their context, which, in this situation, includes Annex 1.2, Article 2.6 and Annex 3 of the TBT Agreement. New Zealand considers that the context within which Article 2.4 is situated therefore supports an interpretation that an "international standard" is one that is prepared by an "international standardizing body", even though this phrase was not specifically used in Article 2.4.
- (ii) First criteria for the international body: specified activities of standardization
- 5.178 The term "body" is defined in the ISO/IEC Guide 2: 1999 as a "legal or administrative entity that has specific tasks and composition". With standardizing bodies in particular, the defining specific task of the body is that it has "recognized activities in standardisation". New Zealand notes that some international organizations that are not specifically established for the sole purpose of setting standards may also have this function as part of their role, and considers that these organizations may also act as standardizing bodies in some discrete situations. In the majority of cases, however, standardizing bodies will have been established, at least partially, for the express purpose of creating international standards.
- (iii) Second criteria: openness to at least all Members of the WTO
- 5.179 The TBT Committee has provided a clear indication of the type of bodies that it considers to be international standardizing bodies and the principles that such bodies should embrace. The principles considered to be important for international standards development are set out in a 2000 TBT Committee Decision and include transparency, openness, impartiality and consensus, effectiveness and relevance, coherence, and to address the concerns of developing countries". ¹³⁶ In particular, the TBT Committee Decision notes that "Membership of an international standardizing body should be open on a non-discriminatory basis to relevant bodies of at least all WTO Members".
- 5.180 Support for the requirement of open membership can also be found in the TBT Agreement. Annex 1.4 defines "international body or system" as a "body or system whose membership is open to the relevant bodies of at least all Members". Article 2.6 requires Members to play a full part in "the preparation by appropriate international standardizing bodies of international standards for products for which they either have adopted, or expect to adopt, technical regulations". If membership of

135 See the definition of "standardising body" in ISO/IEC Guide 2: 1991, paragraph 4.3.

¹³⁴ ISO/IEC Guide 2: 1991, paragraph 4.1.

Decision on Principles for the Development of International Standards, Guides and Recommendations with Relation to Articles 2, 5 and Annex 3 of the TBT Agreement, G/TBT/9, 13 November 2000, at paragraph 20 and Annex 4.

standardizing bodies is not open to all WTO Members then this hinders their ability to meet their obligations under Article 2.6.

- 5.181 Requiring international standards to be made by international bodies with a specified activity of standardization and to be open to all WTO Members supports the object and purpose of the TBT Agreement, which includes harmonization of standards. Harmonization is facilitated where all Members can be involved in the preparation of standards. The international nature of standards also reduces the risk of a proliferation of inconsistent standards set by bodies with limited memberships. A proliferation of such standards would also impact upon a Member's ability to participate in standard-setting activities across all the necessary bodies, thereby undermining the Article 2.6 requirement.
- 5.182 Considering the sum of these various definitions and requirements, read in the light of the context, object and purpose of the TBT Agreement, New Zealand submits that an "international standard" under Article 2.4 must be a document approved by a recognized body, such body being an international standardizing body within the definition provided by the ISO/IEC Guide 2:1991, and whose membership is open to the relevant bodies of at least all WTO Members.

2. Oral statement of New Zealand

- 5.183 New Zealand's oral statement at the third party session at the first panel hearing focused on the issues of whether a *de facto* mandatory measure could fall within the definition of technical regulation in Annex 1.1 of the TBT Agreement, and also on the definition of international standard in Article 2.4 of the TBT Agreement. New Zealand commenced by outlining its interest in the current dispute. As an exporting nation, the proper implementation of the TBT Agreement is of fundamental importance to New Zealand, in light of the potential for technical regulations, standards and conformity assessment procedures to constitute unnecessary obstacles to trade.
- 5.184 The key point made by New Zealand was that it considers that Annex 1.1 provides for a wide range of labelling measures to fall within the definition of "technical regulation" and that are therefore subject to the relevant TBT disciplines. Given the potential for labelling measures to significantly impact on trade it is important that they are subject to the strictest disciplines in the TBT Agreement, so as not to undermine the objectives of that Agreement.
- 5.185 With regard to the question of whether the measure at issue in the present dispute is mandatory, New Zealand noted its view that Annex 1.1 does not exclude the possibility of a measure being *de facto* mandatory. However, New Zealand recognized that the Panel should not make such a finding lightly as to do so would risk rendering null and void the intended distinction between mandatory technical regulations and voluntary standards provided for in the TBT Agreement. In light of this, New Zealand submitted that a measure should only be deemed *de facto* mandatory where it is warranted on the facts. For example, where a Government's conduct is such that it essentially makes compliance obligatory with what would otherwise be a "standard" under the TBT Agreement.
- 5.186 On the definition of 'international standard' in Article 2.4, New Zealand noted that the management of living marine resources is subject to a specialized international legal regime. New Zealand considers that Regional Fisheries Management Organisations are the appropriate bodies for managing shared fisheries, and reaffirmed the ability of countries to enact trade-related measures based on the decisions of those bodies.
- 5.187 New Zealand explained that in accordance with relevant and related definitions in the TBT Agreement and ISO guidelines, New Zealand considers that an "international standard" under Article 2.4 must be a standard made by an international body. Furthermore, that international body must be an international body that has recognized activities in standardization and that must be open to at least all Members of the WTO. New Zealand explained that these criteria support the

harmonization of international standards, which is part of the object and purpose of the TBT Agreement.

K. THAILAND

1. Oral statement of Thailand

- 5.188 Thailand will be brief in its intervention today.
- 5.189 Thailand makes this third party oral statement because of its interest in the interpretation and application of the Agreement on Technical Barriers to Trade ("TBT Agreement").
- 5.190 We observe that there is a limited amount of jurisprudence on the proper interpretation and application of the TBT Agreement. Because this Panel may have to make findings on TBT issues for which there are currently no WTO rulings, and because Thailand is a developing-country Member increasingly subject to other Members' TBT measures, we place much importance in this dispute. In particular, we are interested in the proper interpretation and application of the phrase "more traderestrictive than necessary taking into account of the risks non-fulfilment would create" as set out in Article 2.2 of the TBT Agreement. Also of importance is the consideration of whether any specific criteria must be applied by WTO Members seeking to implement TBT measures deviating from existing or imminent international standards, as set forth in Article 2.4 of the TBT Agreement.

L. Turkey

1. Third party submission of Turkey

- (a) Introduction
- 5.191 Turkey thanks the Panel for giving this opportunity to present its views in this proceeding. Turkey is participating in this case because of its systemic interest in the interpretation and implementation of the Agreement on Technical Barriers to Trade (TBT Agreement). Turkey will present its views on particular issues raised by the complaining parties of this dispute on the interpretation and implementation of the TBT Agreement.
- 5.192 Turkey is not in the intention to present an opinion on the specific factual context of this dispute and takes no position what so ever as to the defence and allegations presented by the parties on whether the specific measure at issue is inconsistent to the subject provisions of the WTO Agreements. Turkey wishes to contribute, by expressing its opinion on some systemic issues regarding the interpretation of the provisions of the TBT Agreement.
- (b) Articles 2.2 and 2.4 of the TBT Agreement
- 5.193 Within the WTO framework, the TBT Agreement is intended to ensure that technical regulations, standards and conformity assessment procedures do not constitute unnecessary barriers to international trade, while recognizing the right of members to take regulatory measures to achieve their legitimate objectives.
- 5.194 First of all, for a regulation to be recognized as a technical regulation it has to contain the criteria defined in the Annex 1 of the TBT Agreement. Annex 1 of the TBT Agreement, which contains terms and definitions for the purpose of the Agreement, describes a technical regulation as "a document which lays down product characteristics or their related processes and production methods, including the applicable administrative provisions, with which compliance is mandatory. It may also

include or deal exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method."

- 5.195 As the Appellate Body identified in EC Asbestos and EC Sardines, order to mention a technical regulation, a document must meet three criteria: (i) a document must apply to a specific product or product group, (ii) the document must state one or more characteristics of the product and (iii) compliance with the stated characteristics of the product must be mandatory. 137 The requirements regulated in this provision are conditions which the regulation must carry in order to be recognized as a technical regulation.
- 5.196 The first criterion determined in Annex 1 of the TBT Agreement is that the regulation must apply to a specific product or product group. Therefore if a regulation applies a determinable specific product or a product group then it can be assumed that the first criterion is met.
- 5.197 In regard to the second criterion that is the document must state one or more characteristics of the product, the Appellate Body in EC – Asbestos Annex explained the "product characteristics" as follows:
 - "... The 'characteristics' of a product include, in our view, any objectively definable 'features', 'qualities', 'attributes', or other 'distinguishing mark' of a product. Such 'characteristics' might relate, inter alia, to a product's composition, size, shape, colour, texture, hardness, tensile strength, flammability, conductivity, density, or viscosity. In the definition of a 'technical regulation' in Annex 1.1, the TBT Agreement itself gives certain examples of "product characteristics" -'terminology, symbols, packaging, marking or labelling requirements'. These examples indicate that 'product characteristics' include, not only features and qualities intrinsic to the product itself, but also related 'characteristics', such as the means of identification, the presentation and the appearance of the product. In addition, according to the definition in Annex 1.1 of the TBT Agreement, a 'technical regulation' may set forth 'applicable administrative provisions' for products which have certain characteristics."¹³⁸
- 5.198 In this framework, Turkey is of the view that this criterion can be deemed as met if a regulation solely obliges labelling requirements.
- 5.199 The last criterion determined in the Annex is that compliance with the stated characteristics of the product must be mandatory. This means that a technical regulation must regulate the "characteristics" of products in a binding or compulsory manner. Mandatory means; required or commanded by authority; obligatory. In this regard, in order to state that a characteristic defined in the regulation is mandatory, it sufficient to show that without carrying that characteristic, it is legally not possible to sale, export, import or produce that product.
- 5.200 On the other hand, 2.2 and 2.4 of the TBT Agreement are the provisions which define conditions that a technical regulation must be in conformity with.
- 5.201 Article 2.2 states that technical regulations are not to be prepared with the aim or effect of causing unnecessary obstacles to trade. It is also stated that they shall not be more trade restrictive than they need to be, to fulfil a legitimate objective.

¹³⁷ Appellate Body Report, EC – Asbestos, paras. 66-72; Appellate Body Report, EC – Sardines, para. 176.

138 Appellate Body Report, *EC – Asbestos*, para 67.

- 5.202 What is to be understood from the term "legitimate objective" is also determined in the same provision as; national security requirement, the prevention of deceptive practices; protection of human health or safety, animal or plant life or health, or the environment. Given that the list of "legitimate objectives" stated in Article 2.2 is not exhaustive, members may protect other legitimate objectives while using the measures prescribed under the TBT Agreement. Nevertheless, regulatory measures undertaken with the purpose to protect a member's legitimate objectives shall neither hinder the rights of other WTO members nor what has been achieved through market access liberalization.
- 5.203 With regard to the interpretation of this provision, Turkey is in the view that unnecessary obstacles to trade can result when a regulation is more restrictive than necessary to achieve a given policy objective, or when it does not fulfil a legitimate objective.
- 5.204 As highlighted in various Panel and Appellate Body reports, in interpreting the necessity test within the context of Article XX of the GATT 1994, a regulation is more restrictive than necessary when the objective pursued can be achieved through alternative measures which have less trade-restricting effects. In this context, this interpretation could be considered in interpreting the Article 2.2 of the TBT Agreement.
- 5.205 On the other hand, according to the provision in deciding whether a technical regulation is more trade restrictive than necessary, it should also be taken into account "the risks non-fulfillment would create". In assessing such risks, relevant elements of consideration to be determined are available scientific and technical information, related processing technology or intended end-uses of products. In Turkey's view, although the burden of proof rests upon the claimant who asserts a violation of the Article 2.2, this would not be high threshold if a WTO Member do not indicate in a sufficient manner, how and which scientific information has been taken into account in adopting such technical regulation.
- 5.206 Last but not least, preamble of the TBT Agreement should also be considered in the interpretation of TBT Agreement in general and Article 2.2 in particular. Therefore, Regulatory measures are permitted as long as they are not applied in a manner which constitutes a means of "arbitrary or unjustifiable discrimination between countries where the same conditions prevail" or a "disguised restriction on international trade", and are otherwise in accordance with the provisions of the TBT Agreement.
- 5.207 On the other hand, Article 2.4 states that "Where technical regulations are required and relevant international standards exist or their completion is imminent, members shall use them, or the relevant parts of them, as a basis for their technical regulations except when such international standards or relevant parts would be an ineffective or inappropriate means for the fulfilment of the legitimate objectives pursued, for instance because of fundamental climatic or geographical factors or fundamental technological problems."
- 5.208 With this in mind, Article 2.4 contains an important obligation assumed by WTO Members under the TBT Agreement. It also carries a right, as WTO Members are authorized to deviate from international standards (or from parts of it) if these are *ineffective or inappropriate* for the achievement of the legitimate objective pursued.
- 5.209 The term "relevant international standards" has been interpreted by the Appellate Body in the EC Sardines. As suggested by the Appellate Body, to be relevant the standard needs "to bear upon, relate, or be pertinent to a technical regulation".
- 5.210 With regard to the meaning of the term "ineffective or inappropriate" (Article 2.4 of the TBT Agreement), the Appellate Body noted that it referred to two issues, effectiveness and appropriateness. These two elements, although closely related, were considered different in nature.

The Panel, as upheld by the Appellate Body, pointed out that the term "ineffective" referred to something which was not "having the function of accomplishing", "having a result", or "brought to bear", whereas the term "inappropriate" referred to something which was not "specially suitable", "proper", or "fitting".

(c) Conclusion

5.211 Turkey appreciates this opportunity to present its views to the Panel. Turkey requests this Panel to review the comments stated in this submission, in interpreting the TBT Agreement.

VI. INTERIM REVIEW

A. GENERAL

- 6.1 On 5 May 2011, we transmitted our interim report to the parties. On 26 May 2011, Mexico and the United States requested the Panel to review precise aspects of the Interim Panel Report pursuant to Article 15.2 of the DSU and the Working Procedures adopted by the Panel. On 14 June 2011, both parties presented comments on each other's requests for review.
- 6.2 In its comments on the US request for review, Mexico argued that "many of the changes proposed by the United States appear to be efforts to change the substantive findings of the Panel, to enter into debate with the Panel, to have the Panel place special emphasis on US arguments, or to alter the Panel's descriptions of Mexico's arguments. Moreover, many of the US comments are more in the nature of requests for reconsideration which are not appropriate for this phase of the proceedings". 139
- As stated on previous occasions by the Appellate Body, the interim review stage is not an appropriate moment to introduce new and unanswered evidence. However, in our view, requests to review precise aspects of the Panel's report may legitimately include requests for "reconsideration" of specific factual or legal findings, provided that such requests are not based on the presentation of new evidence. We therefore did not find it necessary to exclude *a priori* from consideration any request for review from either party on the sole basis that it would seek reconsideration by the Panel of some of its determinations. We note in this respect that Mexico itself requested the Panel to reconsider its decision to exercise judicial economy in relation to Mexico's claims under the GATT 1994 and sought a review of certain aspects of the Panel's determinations.
- 6.4 With this preliminary observation in mind, we explain below how we took into account the parties' specific requests for review in our final Report.

B. DESCRIPTIVE PART (SECTION II OF THE REPORT)

6.5 The United States requested some adjustments to the factual description of the US dolphin-safe measures and of the AIDCP regime, in paragraphs 2.7, 2.8, 2.12, 2.15, 2.16, 2.19, 2.20, 2.21, 2.22, 2.23, 2.25, 2.35, and 2.38 in the Descriptive Part of the Report. Mexico objected to the proposed changes, which it found either unnecessary or inappropriate. We adjusted the text of these paragraphs as relevant in light of the parties' comments, with a view to clarifying the description of the relevant facts.

¹³⁹ Mexico's comments on the United States' comments on the interim report, para. 2.

¹⁴⁰ See Appellate Body Report, EC – Sardines, para. 301.

C. FINDINGS

1. General (Section VII.A)

6.6 The United States proposed some adjustments to paragraphs 7.13 (and the associated footnote) and 7.23 of the interim report. The Panel adjusted the language in these paragraphs for greater clarity and accuracy, taking into account also Mexico's comments on the US request.

2. Whether the measures at issue constitute a "technical regulation" within the meaning of Annex 1.1 of the TBT Agreement (Section VII.B.1)

- 6.7 The United States requested adjustments to paragraphs 7.60, 7.72, 7.116 of the interim report to improve the accuracy of the descriptions contained therein. The Panel accepted these changes to the extent that they constituted clarifications. The United States also suggested additions to paragraph 7.13 of the interim report to reflect in more detail its arguments concerning US commitments under the Panama Declaration. The Panel addressed this point in paragraphs 2.16 to 2.19 and referred the reader to it through footnote 196.
- Mexico requested the Panel to review paragraphs 7.86, 7.165, 7.176, 7.180 of the interim report "in accordance with the arguments and direct evidence presented by Mexico". Specifically, Mexico suggests that references to evidence concerning canneries and tuna in these paragraphs be eliminated, in light of the fact that Mexico narrowed down its claims in the course of the proceedings to tuna products only and do not include tuna. ¹⁴¹ In light of this, Mexico requests that the Panel focus on the retailer and consumer consumption stages only. Mexico argues that it has put forward evidence concerning the preferences of retailers and consumers suggesting that if the AIDCP label could be placed on its products, they would be acceptable to major US retailers and a substantial proportion of US consumers.
- 6.9 The United States commented that Mexico's request should be rejected. The United States observed that, while Mexico did state in response to a question from the Panel that the like product analysis should be limited to tuna products, it made arguments and presented evidence throughout the proceedings that addressed the processing of tuna, not just tuna products. In the US view, the Panel's findings properly reflect the scope of Mexico's claims, and Mexico does not explain why its narrowed claims would preclude the Panel from taking evidence relating to the practices of canneries and tuna producers into account. In the US view, such evidence is relevant not only in its own right but also as indirect evidence of the preferences of consumers.
- 6.10 We first note that, as described in paragraphs 7.232 and 7.233, Mexico clarified in the course of the proceedings that it was seeking findings under Articles III:4, I:1 of GATT and Article 2.1 of the TBT Agreement only in relation of tuna products, and not tuna. For that reason, we made no findings of law under these provisions in relation to tuna as such, and limited our findings in this respect to tuna products. 143
- 6.11 We also note, however, that the section of the Report containing the paragraphs referred to above addresses the question of whether compliance with the measures at issue is "mandatory" within the meaning of Annex 1 of the TBT Agreement. This question is not, as such, dependent on the scope of Mexico's non-discrimination claims under Article 2.1 of the TBT Agreement (or under Article III:4 and I:1 of GATT 1994).

¹⁴¹ Mexico's comments on the United States' comments on the interim report, paras. 7-21.

¹⁴² See the United States' comments on Mexico's comments on the interim report, para. 17.

¹⁴³ See paras. 7.227 and 7.228.

- 6.12 We have adjusted the text of paragraph 7.87 and the associated footnote 265 in order to reflect more clearly the manner in which Mexico presented its arguments concerning canneries and tuna in relation to the "mandatory" character of the measures.
- 6.13 Other references to the practices of canneries in this Section that Mexico seeks the removal of (in paragraphs 7.165, 7.176 and 7.177 to 7.180 of the interim report) relate to the separate opinion contained in paragraphs 7.146 to 7.190 in the final report. Modifications made to this part of the Report should therefore be understood as having been made by the author of the separate opinion.
- 6.14 Paragraph 7.167, which refers to Mexico's arguments, has been modified to clarify the manner in which Mexico referred to the practices of tuna processors in the context of its arguments that the US measures are *de facto* "mandatory".
- 6.15 Paragraphs 7.176 and 7.177 to 7.180 of the interim report have not been modified because, in the view of their author, references to the practices of canneries in relation to the purchase of tuna are relevant to the determination made in this section, notwithstanding Mexico's decision to narrow down the scope of its non-discrimination claims to tuna products only. This section of the separate opinion addresses whether compliance with the US dolphin-safe provisions is *de facto* mandatory. As observed above, an assessment of this question is not dependent on the scope of Mexico's non-discrimination claims.
- 6.16 The specific question that the evidence relating to the practices of tuna processors helps to inform in this context is whether the measures are mandatory by virtue of US actions rather than as a result of the practices of the market. The evidence relating to tuna processors makes clear that processors do not wish to purchase tuna that would not be eligible for the label, although it would be consistent with the terms of the measures to do so, and that this is a choice that they voluntarily make. In light of Mexico's observation that the analysis should focus on the retailer and consumer level, additional language has been included in paragraph 7.185 to clarify the author's view on the arguments highlighted by Mexico in its request for review. As explained in that paragraph, a consideration of the decision of retailers not to carry tuna products that are not eligible for the dolphin-safe label confirms that it is the decisions of private actors themselves that give weight to the dolphin-safe label on the US market and does not modify the conclusion reached in this section. As explained in paragraph 7.180, in the view of the author of the separate opinion, such decisions of private actors on the market in relation to a voluntary standard do not turn such standard into a "technical regulation" with which compliance is mandatory.
- 6.17 The author of the separate opinion declined to introduce an additional reference to Mexico's arguments at paragraph 7.162 of the interim report, which reflects the author's assessment of the EC Sardines rulings.

3. Article 2.1 of the TBT Agreement (Section VII.B.2)

- 6.18 The United States suggested changes to paragraphs 7.224, 7.250, 7.303, 7.307, 7.309, 7.315, 7.316, 7.323 and 7.325 of the interim report to improve the accuracy of the descriptions contained therein. Both parties also requested a number of insertions in order to reflect in more detail their arguments in the findings of the Panel. The United States suggested additions to paragraphs 7.251, 7.253 and 7.342 of the interim report to reflect in more detail its arguments. Mexico objected to such changes arguing that the United States was highlighting arguments that were irrelevant to the point discussed, either seeing to convert them into Panel's findings or changing the focus of the paragraphs making them confusing to read.
- 6.19 The Panel accepted such requests where it considered that they contributed to a better understanding of the issues and added clarity to the findings. On that basis, the Panel made

adjustments to paragraphs 7.230, 7.256, 7.312, 7.314, 7.324, 7.329, 7.329, 7.331 and 7.353 of the final report to reflect the arguments made in the course of the proceedings by both parties and clarify where necessary its treatment of these arguments. The Panel declined, however, to make the requested adjustments to paragraphs 7.303 and 7.253 of the interim report, where it did not consider that the proposed changes were useful to clarify or improve the treatment of the issue addressed in paragraphs 7.308 and 7.257 of the final report.

- 6.20 Mexico also requested the Panel to review paragraphs 7.258, 7.284, 7.285, and 7.350 to 7.354 of the interim report, "in accordance with the arguments and direct evidence presented by Mexico", for the reasons outlined in paragraph 6.8 above. As described above, we acknowledge that Mexico narrowed its non-discrimination claims, in the course of the proceedings, to tuna products, and not tuna. However, we are not persuaded that this implies that we may not take into consideration relevant evidence presented by the parties in the course of the proceedings in relation to the practices of tuna processors and tuna, to the extent that it may inform our assessment of whether the measures at issue afford Mexican tuna *products* less favourable treatment within the meaning of Article 2.1 of the TBT Agreement. Indeed, should we have excluded from consideration any evidence relating to tuna rather than tuna products, we would also not have been able to take into account some of the main facts at the core of Mexico's claims, including the fishing practices of its tuna fleet.
- 6.21 Paragraph 7.289 of the final report has been modified to clarify the manner in which the Panel understands Mexico's arguments concerning the practices of tuna processors and how these should be taken into consideration when analysing the US dolphin-safe labelling provisions consistency with Article 2.1 of the TBT Agreement.
- 6.22 In paragraphs 7.362 to 7.364, reference is made to the practice of tuna processors for the purposes of determining whether access to the label constitutes an advantage on the US market (for tuna products). As explained in paragraph 7.364, the fact that major tuna processors respond to dolphin-safe concerns by ensuring that they source only tuna that will make their tuna products eligible for a dolphin-safe label is, in our view, evidence that a dolphin-safe designation is considered valuable on the US market, and that, consequently, access to the label regulated under the US dolphin-safe measures constitutes an advantage for tuna products on the US market. We see no need to modify our determinations in this respect. However, in order also to address Mexico's observation that the analysis should address the direct evidence provided by Mexico in relation to the preferences of retailers and final consumers, we have added paragraph 7.290.
- 6.23 In paragraphs 7.350 to 7.354, reference is made to the practices of major tuna processors to inform the relationship between the measures and the practices of operators on the market in relation to the dolphin-safe issue. As described in paragraph 7.353, this contributes, in combination with other elements, to the Panel's assessment of the situation of Mexican tuna *products* on the US market, in particular the role of the measures and other factors in the limited presence of tuna products caught by setting on dolphins, including Mexican tuna products, on the US market. For this reason, we do not consider it necessary to eliminate these references on account of the fact that Mexico's claims are limited to tuna products and do not extend to tuna. Nonetheless, we considered it useful to elaborate further also on the elements highlighted by Mexico in relation to the preferences of retailers and consumers of tuna products to clarify the basis for our findings in this respect. We therefore added paragraph 7.364 to the final report.
- 6.24 Mexico also sought a review of paragraphs 7.330, 7.334 and 7.335 of the interim report, to reflect more fully its arguments on the costs of adaptation to the US measures. The United States commented that the interim report already included Mexico's arguments about its efforts to adapt and

that it saw no reason to reiterate them.¹⁴⁴ We reviewed the relevant section in paragraphs 7.336 to 7.346 to reflect more fully the arguments presented Mexico and accordingly also explained in more detail our treatment of these arguments.

4. Article 2.2 of the TBT Agreement (Section VII.B.3)

- 6.25 We adjusted the text of paragraph 7.346 to improve its clarity, further to a request by the United States and taking into account Mexico's comments.
- 6.26 The United States also requested adjustments to paragraphs 7.301, 7.302, 7.326, 7.327 of the interim report. We declined to make these changes, which would have altered our assessment of the relevant issues in a manner that we did not agree with.
- 6.27 The United States requested changes to paragraphs 7.381, 7.385, 7.397 and 7.403 of the interim report that would have modified the Panel's assessment of the objective of the US dolphin-safe labelling provisions in relation to dolphin protection, to define it in terms of "ensuring that the US market is not used to encourage fishing fleets to set on dolphins" instead of "ensuring that the US market is not used to catch tuna in a manner that adversely affects dolphins". For the reasons explained in paragraphs 7.414 to 7.425, we have understood this objective in the manner in which the United States itself initially defined it, i.e. "protecting dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins". We therefore declined to make the changes requested by the United States. However, we made some adjustments to paragraphs 7.397, 7.402, 7.414 and 7.567 of the final report to clarify the manner in which the United States described this objective in its different submissions to the Panel.
- 6.28 The United States requested the Panel to cite additional factual information in a footnote concerning its arguments on dolphin mortalities in the Western Central Pacific Ocean (WCPO) (footnote 676 in the interim report). In support of this insertion, the United States cites an exhibit presented by Mexico, which was not referred to by the United States in the part of its submissions that the footnote addressed.
- 6.29 We note that it would have been appropriate for the United States to refer to the relevant evidence in support of its factual assertions at the time of making such assertions, rather than in the interim phase of the proceedings. Nonetheless, we also note that the United States had referred to this particular piece of evidence and the factual information contained therein in one of its comments on responses to questions by Mexico on a related issue. For that reason, we accepted to refer to and

"Reports on that fishery reveal that marine mammal take is relatively small. Based on reporting from independent observers on U.S. vessels fishing for tuna in the Western Central Pacific Ocean (WCPO), of the 1500 sets observed in 2008, there were 5 interactions with false killer whales and one interaction with a short-finned pilot whale observed; there were no observed interactions with other marine mammals."

This statement is accompanied by the following text in a footnote "Communication from Dr. Charles Karnella, International Fisheries Administrator, NOAA Fisheries Pacific Islands Regional Office, to Brad Wiley, Foreign Affairs Specialist, NOAA Fisheries Office of International Affairs on October 29, 2010". The communication in question is not provided or otherwise explained.

146 See US comments to Mexico's response to Panel question No. 101 ("For example, Mexico cites data on marine mammal interactions in the Western Central Pacific Ocean, yet that data show that over an eleven

¹⁴⁴ United States' comments on Mexico's comments on the interim report, paras. 42-43.

¹⁴⁵ In its response to the Panel question No. 15, para. 49, the United States noted that:

address in our final Report the relevant evidence cited by the United States. This is addressed in paragraphs 7.524 to 7.529.

- 6.30 The United States requested that a new paragraph be added immediately following paragraph 7.490 of the interim report to reflect its response to Mexico's arguments that there are substantial dolphin mortality rates associated with fishing techniques other than setting on dolphins, that scientific research indicates that there are associations of tuna with dolphins outside the ETP and that despite the evidence that dolphins and other marine mammals are being killed in significant numbers in ocean regions other than the ETP, no measures have been taken in those other regions comparable to those taken for the ETP. 147
- 6.31 Mexico objected to the proposed insertion, stating that the United States has proposed to add almost an entire new page reciting arguments from its written submissions, repeating points addressed elsewhere in the interim report. Nonetheless Mexico noted that, if the Panel decides to include the new paragraph proposed by the United States, for purposes of maintaining balance in the manner in which the Parties' arguments are described, it would expect the Panel to also repeat in detail Mexico's responses to those US arguments. 148
- 6.32 The Panel agreed to incorporate additional references to the parties' arguments (at paragraphs 7.512 to 7.514) and consequently found it necessary to clarify also its treatment of these arguments (at paragraphs 7.524 to 7.530 and 7.559 to 7.561).
- 6.33 The United States requested a modification of paragraph 7.510 of the interim report to clarify the requirements for access to an alternative label. The Panel clarified the language of this paragraph and accordingly also clarified its treatment of this issue, at paragraphs 7.540 and 7.541.

5. Article 2.4 of the TBT Agreement (Section VII.B.4)

- 6.34 With respect to the Panel's findings under Article 2.4 of the TBT Agreement, the Panel adjusted the text of paragraphs 7.604, 7.605, 7.606 and 7.612 of the interim report in order to clarify the description of United States' arguments in these paragraphs, at their request. The Panel also edited the text of paragraphs 7.638 to 7.640 of the final report to improve their clarity, further to a request of the United States and taking into account comments by Mexico. 149
- 6.35 The United States also sought changes to paragraphs 7.630 to 7.639 of the interim report in relation to the interpretation of the terms "international standard" in Annex 1 of the TBT Agreement. Mexico considered this to amount to a request for reconsideration of the Panel's legal findings. However, as described above, we do not consider that it is inappropriate for a party to seek a review of factual or legal determinations made by the Panel in the interim report, provided

year period 33,319 purse seine sets were observed and bycatch of 1315 marine mammals was reported, 46 of which were reported killed.")

¹⁴⁸ Mexico's comments on the United States comments on the interim report, paras. 75-76.

¹⁴⁷ United States' comments on the interim report, para. 80.

¹⁴⁹ United States' comments on the interim report, paras. 129, 130 and 134. Mexico objected to the modification proposed in paragraph 7.641, considering that the proposed insertion would not provide further clarity because the meaning was already clear (Mexico comments on the United States' comments on the interim report, para. 108). Given the differences of view between the parties throughout the proceedings on the meaning of the term "dolphin-safe", we saw value in clarifying the context in which the terms "dolphin-safe" and "non-dolphin safe" are used in this paragraph..

¹⁵⁰ United States' comments on the interim report, paras. 131-133, 135-137.

¹⁵¹ Mexico comments on the United States' comments on the interim report, paras. 107 and 109-110.

that such request is based on evidence previously presented to the Panel. Accordingly, without prejudice to its substantive merits, we do not consider the United States' request to be inadmissible.

- 6.36 As we understand it, the United States is seeking a review by the Panel of its determination that the "international standard" in Article 2.4 of the TBT Agreement should be understood with reference to the definition of a "standard" contained in the TBT Agreement rather than with reference to the definition of the same term in the ISO/IEC Guide 2.
- 6.37 For the reasons explained in paragraph 7.670, we are not persuaded that the definition of the term "standard" in Annex 1.2 should be the primary basis for interpreting this term as contained in the expression "international standard" in Article 2.4. Rather, we consider that the meaning of the term "international standard" in that provision should be understood with reference to its definition in the ISO/IEC Guide 2, including the term "standard" as also defined in the same Guide. We have therefore declined to modify paragraphs 7.630 to 7.639 of the interim report as requested by the United States, but adjusted the language of paragraphs 7.670 to clarify the basis for our determination.
- 6.38 However, we also note that, should the term "international standard" be interpreted with reference to the of the terms "standard" and "international body" as defined in Annex 1 of the TBT Agreement as suggested by the United States, this would not alter our conclusion, for the purposes of this dispute, that the AIDCP dolphin-safe definition constitutes an international standard, insofar as it is voluntary and is approved by parties to the AIDCP, whose membership is open to the relevant bodies of at least all WTO Members.
- 6.39 The United States also sought a modification to paragraph 7.658 and footnote 852 of the interim report, to remove the reference to the membership of the Antigua Convention. We declined to make this change, on the basis that, as noted by Mexico, this reference is pertinent to the Panel's consideration of whether membership in the AIDCP is open to the relevant national body of every country. To the extent the membership of the AIDCP is linked to the Membership of the Antigua Convention, the reference to the latter is pertinent.
- 6.40 The United States recommended the addition of a sentence in paragraph 7.667 of the interim report to clarify the conditions of application of the "dolphin safe" certification under the AIDCP resolutions. Mexico objected to this insertion on the basis that it does not directly relate to the issue discussed in this paragraph and seeks to introduce new arguments. We have clarified the language in paragraph 7.640 to clarify the point raised by the United States. In paragraph 7.686 of the interim report, the United States requested the addition of a reference to specific DMLs permitted in the AIDCP. We did not find this addition to be necessary and therefore declined to introduce it.

6. Exercise of judicial economy on Mexico's claims under Articles I and III:4 of GATT 1994 (Section VII.C)

6.41 Mexico requested the Panel to reconsider its decision to exercise judicial economy in regard to findings under Article I:1 and III:4 of the GATT. First, Mexico considered that in this case the report has only resulted in a partial resolution of the dispute and that the Panel has exercised false judicial economy with respect to its discrimination claims because despite finding that the measures were consistent with Article 2.1 of the TBT Agreement, the Panel declined to rule on Articles I:1

¹⁵² United States' comments on the interim report, para.138

¹⁵³ Mexico comments on the United States' comments on the interim report, para. 110

¹⁵⁴ United States' comments on the interim report, para. 139.

¹⁵⁵ United States' comments on the interim report, para. 140.

¹⁵⁶ Mexico's comments on the interim report, para. 67.

and III:4 and therefore left open the question of discrimination.¹⁵⁷ The United States responded that such assertion misses the point because the question as to the proper exercise of judicial economy is not whether the complainant feels its concerns have been resolved but rather whether the Panel believes it has addressed the claims necessary to resolve the dispute. The United States considered that the Panel acted within the scope of its discretionary authority in deciding to exercise judicial economy on Mexico's claims under the GATT 1994.¹⁵⁸

6.42 Further, Mexico argued that finding that the US measures are inconsistent with Article 2.2 of the TBT Agreement does not, in itself fully resolve this dispute and referred to the panel's findings in *US – Poultry (China)*, where despite its finding of inconsistency under the SPS Agreement the Panel refrained from exercising judicial economy with respect of a claim under Article I of the GATT 1994 on the basis that there were different obligations embodied in both provisions and a finding only on Article XI would not provide a positive solution to the dispute with respect to discrimination. Mexico considered the same situation arises here with the Panel's findings under Article 2.2 of the TBT Agreement does not provide a positive solution to this dispute in respect of discrimination. The United States considers that the Panel's finding that the measures are inconsistent with Article 2.2 of the TBT Agreement does resolve the dispute because it says it does not understand the Panel to have considered that these findings alone did but rather that the Panel appears to have fully examined Mexico's non-discrimination claims, as well as its claims under Articles 2.2 and 2.4 of the TBT Agreement, and on this basis the Panel considered that it had resolved the dispute.

6.43 Mexico argued that although the three provisions deal with non-discrimination obligations, each of them is different in language, nature, scope and application and that that the TBT provisions cannot be taken to replace, subsume or exclude provisions of the GATT and that thus, the Panel's finding of consistency with Article 2.1 cannot be assumed to mean that the US measures are also consistent with Articles I:1 and III:4. 161 In response the United States observed that Mexico consistently referred the Panel to its arguments under the GATT provisions as the basis for its TBT Article 2.1 claims and that it agreed with the Panel that Mexico has not explained how its assertion of differences in nature, scope, and application of the provisions result in different rights and obligations which would have different implications during implementation. It also noted that Mexico made this same argument earlier in the proceedings and that it appeared from the report that the Panel had already taken these arguments into consideration in deciding to exercise judicial economy. 162 The United States considered Mexico misunderstood the Panel's statement which it says was relying on the fact that Mexico's arguments under Article 2.1 of the TBT Agreement derived directly from its arguments under the GATT 1994 provisions, and so the Panel could be comfortable in concluding that in light of its findings under Article 2.1 of the TBT Agreement it need not reach Articles I:1 and III:4 of the GATT 1994. 163 In Mexico's view, the dissent by one member of the Panel on the threshold issue of whether the US measures constitute a technical regulation and are covered by Article 2.2

¹⁵⁷ Mexico argued that if a Panel decides to exercise judicial economy, it must be consistent with the aim of the dispute settlement system which is to resolve the matter referred to the DSB and to secure a positive solution to a dispute. (Mexico's comments on the interim report, paras. 69-71). The United States noted that "[t]he Appellate Body has consistently ruled that panels are not required to address all the claims made by a complaining party but rather a panel has discretion to determine which claims it must address in order to resolve the dispute between the parties." (footnote omitted) (United States' comments on Mexico's comments on the interim report, fn 25).

¹⁵⁸ United States' comments on Mexico's comments on the interim report, para. 59.

^{159 (}footnote omitted) (Mexico's comments on the interim report, fn 24).

¹⁶⁰ Mexico's comments on the interim report, para. 73.

¹⁶¹ Mexico's comments on the interim report, paras. 71-72.

¹⁶² United States' comments on Mexico's comments on the interim report, para. 60.

¹⁶³ United States' comments on Mexico's comments on the interim report, para. 62.

further highlights the importance of the Panel ruling on Mexico's claims under the GATT 1994. The United States noted that where the Panel had already resolved the threshold issue in Mexico's favour, the exercise of judicial economy was within the Panel's discretion. 164

6.44 For the reasons explained in Section VII:C of this Report, we did not consider it necessary to modify our decision to exercise judicial economy in respect of Mexico's claims under Articles I.1 and III:4 of the GATT 1994. As explained in paragraph 7.748, we consider that, in addressing all aspects of Mexico's claims under the TBT Agreement, including, but not limited to, its discrimination claims, we have addressed Mexico's claims in a manner sufficient to resolve the dispute. We further note in this respect that, despite having found a violation of Article 2.2 of the TBT Agreement, the Panel proceeded to examine Mexico's claim under Article 2.4 of the TBT Agreement, in order to address adequately the alleged breaches of the different legal obligations invoked by Mexico under the TBT Agreement.

VII. FINDINGS

A. GENERAL

1. Procedural issue: Amicus curiae brief

- 7.1 Before the first substantive meeting, the Panel received an unsolicited *amicus curiae* brief from Humane Society International and American University's Washington College of Law, accompanied by a letter dated 6 May 2010.
- 7.2 On 28 June 2010, the Chairman of the Panel informed the parties and third parties that it considered, in light of the Appellate Body's rulings in *US Shrimp*, that it had the discretionary authority either to accept and consider or to reject information and advice submitted to it, and that it would accordingly treat this submission as it deemed appropriate. The parties and third parties were invited to provide any views they had in relation to the submission at the first substantive meeting. The parties and third parties were also reminded of the possibility of incorporating part or all of the information contained in the submission into their respective submissions and/or oral statements, subject to the provisions of the DSU and the Panel's Working Procedures. ¹⁶⁵

"I would like to inform you that the Dispute Settlement Registrar has received an unsolicited *amicus curiae* brief from Humane Society International and American University's Washington College of Law. The brief has been forwarded to the Panel. A copy of the brief is attached.

Taking into account the determinations of the Appellate Body in *US – Shrimp* (WT/DS58/AB/R), the Panel considers that it has the discretionary authority either to accept and consider or to reject information and advice submitted to it, and it will accordingly treat this brief as it deems appropriate. The Parties and Third Parties are invited to offer any views they may have in relation to the brief at the first substantive meeting.

The Panel reminds Parties and Third Parties that, if they so wish, they have the possibility of incorporating part or all of the information contained in the brief into their respective submissions and/or oral statements, subject to the provisions of the Dispute Settlement Understanding and the Panel's working Procedures."

¹⁶⁴ United States' comments on Mexico's comments on the interim report, para. 61.

¹⁶⁵ Letter from the Chairman of the Panel to the Parties dated 28 June 2010:

- 7.3 At the first substantive meeting, the United States requested the Panel to review and consider the submission in its deliberations, in light of the relevant and useful information it contained which it believed could assist the Panel in understanding the issues in this dispute. ¹⁶⁶ In addition, in its responses to the Panel's questions, the United States referred to some of the information contained in the *amicus curiae* brief in support of its arguments. ¹⁶⁷ Mexico did not comment on the filing of the *amicus curiae* brief or on its content at the first substantive meeting.
- 7.4 The Panel subsequently invited both parties to comment on certain information presented as an exhibit to the *amicus curiae* brief. In its response, Mexico observed that factual information submitted through an *amicus curiae* brief could not be properly treated as part of the record of this dispute. The United States responded that, although a panel is not required to consider information submitted by *amici*, it may consider such information as it sees fit. In particular, the United States noted that the Chairman of the Panel had invited the parties and third parties in this dispute to offer views in relation to the amicus curiae brief filed. The particular information are relation to the amicus curiae brief filed.
- 7.5 With respect to the question of whether the information contained in the brief may properly be considered by the Panel, we first note that the Appellate Body in *US Shrimp* observed that:

"The thrust of Articles 12 and 13, taken together, is that the DSU accords to a panel established by the DSB, and engaged in a dispute settlement proceeding, ample and extensive authority to undertake and to control the process by which it informs itself both of the relevant facts of the dispute and of the legal norms and principles applicable to such facts. That authority, and the breadth thereof, is indispensably necessary to enable a panel to discharge its duty imposed by Article 11 of the DSU to "make an objective assessment of the matter before it, including an objective assessment of the facts of the case and the applicability of and conformity with the relevant covered agreements ... "¹⁷⁰

7.6 In light of the "amplitude of the authority vested in panels to shape the processes of fact-finding and legal interpretation" the Appellate Body ruled that:

"If, in the exercise of its sound discretion in a particular case, a panel concludes inter alia that it could do so without 'unduly delaying the panel process', it could grant

166 The United States mentions that the submission discusses the adverse impact on dolphins and dolphin populations in the ETP resulting from the practice of setting on dolphins to catch tuna; explains that fleet capacity in the ETP is the largest threat to tuna stocks in the ETP rather than particular fishing techniques; reviews the substantial retailer and consumer interest in a dolphin-safe label that ensures that consumers are not misled as to whether tuna products contain tuna that was caught in a manner that adversely affects dolphins. It also emphasized Humane Society International's nearly three decade involvement in the issues surrounding this dispute. United States' opening oral statement of the first substantive meeting, para. 54.

167 In particular when responding to the Panel's question as to whether consumers' preferences were determined by the fishing method, the United States responded that at the time the US dolphin-safe labelling provisions were enacted, there was strong consumer sentiment that setting on dolphins to catch tuna was unacceptable and that the amicus curiae brief highlighted this sentiment. United States' response to Panel question No. 40(a), para. 98. The United States also referred to the amicus curiae brief to support its assertion that the ETP is fundamentally different from all other oceans in that it is the only ocean where tuna and dolphins have a known regular and significant association and the only ocean where such an association is exploited as the foundation for a commercial fishery. See United States' response to Panel question No. 12(b), fn 18.

¹⁶⁸ Mexico's response to Panel question No. 88, para. 5.

¹⁶⁹ United States' comments on Mexico's response to Panel question No. 88. None of the third parties offered any views with respect to the amicus curiae brief.

170 Appellate Body Report, *US – Shrimp*, para. 106.

Appellate Body Report, *US – Shrimp*, para. 106. Appellate Body Report, *US – Shrimp*, para. 108.

permission to file a statement or a brief, subject to such conditions as it deems appropriate. The exercise of the panel's discretion could, of course, and perhaps should, include consultation with the parties to the dispute." ¹⁷²

7.7 Taking into account these determinations, as described above, the Panel informed the parties prior to the first meeting that:

"Taking into account the determinations of the Appellate Body in *US – Shrimp* (WT/DS58/AB/R), the Panel considers that it has the discretionary authority either to accept and consider or to reject information and advice submitted to it, and it will accordingly treat this brief as it deems appropriate."

- 7.8 As also described above, the Panel invited the parties to comment on the brief at issue, in accordance with the requirements of due process. 173
- 7.9 The Panel therefore considers that it has the authority to consider the information contained in the submission filed by Humane Society International and American University's Washington College of Law, and has done so to the extent that it deemed it relevant to the examination of the claim before it. Where the Panel considered the information presented in and the evidence attached to the *amicus curiae* brief relevant, it has sought the views of the parties in accordance with the requirements of due process. In addition, to the extent that one of the parties has cited the amicus curiae brief or cross-referenced to the exhibits presented with such brief in its reasoning or responses to questions, as described in paragraph 7.3 above, these elements form part of the submissions of that party in these proceedings and the Panel deems appropriate refer to such information in its findings.

2. Examination of the measures together

7.10 In its request for the establishment of a panel, Mexico identified three separate legal instruments: (i) United States Code, Title 16, Section 1385 –the "Dolphin Protection Consumer Information Act – DPCIA"–, (hereafter "the DPCIA") (ii) the Code of Federal Regulations, Title 50, Section 216.91 –"Dolphin-safe labelling standards"– and Section 216.92 –"Dolphin-safe requirements for tuna harvested in the ETP [Eastern Tropical Pacific Ocean] by large purse seine vessels"–(hereafter "the implementing regulations") and (iii) the judicial ruling in *Earth Island Institute v. Hogarth*, 494 F. 3d 757 (9th Cir. 2007) (hereafter "the *Hogarth* ruling") as basis for its claims. However, in the course of the proceedings, Mexico referred to the measures as a whole under the generic expression "US measures"¹⁷⁴, "labelling provisions"¹⁷⁵ or "labelling measures"¹⁷⁶.

¹⁷² Appellate Body Report, *US – Shrimp*, para. 107.

We also note that the Appellate Body recently followed a comparable procedure in similar circumstances:

[&]quot;On 15 December 2010, the Appellate Body received an unsolicited *amicus curiae* brief. After giving the participants and the third participants an opportunity to express their views, the Division hearing the appeal did not find it necessary to rely on this *amicus curiae* brief in rendering its decision" (Appellate Body Report, *US – AD/CVD on certain products from China*, para. 18).

¹⁷⁴ See for instance Mexico's first written submission, paras. 155, 164, 165, 168, 169 and throughout the written submission.

¹⁷⁵ See for instance Mexico's first written submission, para. 195.

¹⁷⁶ See for instance Mexico's first written submission, para. 240.

"US dolphin-safe labelling measures" 177 and "labelling scheme". 178 The United States has also referred to the measures as the "US dolphin-safe labelling provisions" 179 or the "labelling scheme". 180

7.11 The Panel therefore asked Mexico to clarify whether it considered that each measure individually leads to each of the violations it identified, or whether it considered that the violations arise from the measures in combination, and whether it was seeking from the Panel separate determinations in relation to each measure. The Panel also afforded the United States an opportunity to comment on this matter.

(a) Arguments of the parties

- 7.12 In response to the Panel's question, Mexico explained that it had identified three separate legal instruments under this dispute, but that each instrument is part of to the same measure: the US dolphin-safe labelling regime. Mexico contends that "[a] measure can be made of up more than one instrument. It is common in the domestic legal systems of many WTO Members for a measure to comprise legislative provisions, regulatory provisions and other kinds of legal instruments". 182
- 7.13 Mexico further explained that the DPCIA is a law enacted by the US Congress which contains the requirements to obtain a dolphin-safe label. The regulations codified under 50 CFR Section 216.91 and 216.92 provide, Mexico says, the regulatory conditions imposed by the US Department of Commerce for the use of dolphin-safe label on tuna and tuna products in accordance with the DPCIA. Finally, Mexico noted that, as the United States had recognized, the judicial ruling in *Earth Island Institute vs. Hogarth* "vacated the Secretary of Commerce's final finding under Section 1385(g) that would have permitted tuna products that contained tuna that was caught by setting on dolphins to be labelled dolphins safe". ¹⁸³ As a consequence of this interpretation by the US Court, the US measure prohibits, Mexico argues, the use of a dolphin-safe label if the tuna was caught in association with dolphins, and it does not matter that no dolphins were killed or injured in the set in which the tuna was caught. ¹⁸⁴
- 7.14 In light of these observations, Mexico considers that the measures should be analysed as a whole through a "comprehensive analysis" leading to the conclusion that there are violations of Articles I:1 and III:4 of the GATT 1994 and Articles 2.1, 2.2 and 2.4 of the TBT Agreement. Mexico also specifies that it is neither asking for separate rulings for each measure, nor is it suggesting that the Panel undertake an independent analysis of each measure. 186
- 7.15 The United States did not comment on Mexico's proposed approach.
- (b) Analysis by the Panel

7.16 As described above, Mexico has identified three distinct legal instruments in its request for establishment of a panel (the DPCIA, the implementing regulations, and the *Hogarth* ruling). In the

¹⁷⁷ See for instance Mexico's first written submission, paras. 226 and 247.

¹⁷⁸ See for instance Mexico's first written submission, para. 203.

¹⁷⁹ See for instance United States' first written submission, paras. 97, 98, 100, 102, 103 and throughout the written submission.

¹⁸⁰ See for instance United States' first written submission, paras. 4, 16 and 55.

¹⁸¹ Mexico's response to Panel question No. 2, paras. 6-7.

¹⁸² Mexico's response to Panel question No. 2, para. 8.

¹⁸³ Mexico's response to Panel question No. 2, para. 7 (footnote omitted).

¹⁸⁴ Mexico's response to Panel question No. 2, para. 7.

¹⁸⁵ Mexico's response to Panel question No. 2, para. 9.

¹⁸⁶ Mexico's response to Panel question No. 2, para. 10.

course of the proceedings, Mexico clarified that it seeks findings in relation to the combined operation of these three instruments, rather than for each instrument individually. We must therefore consider whether these various instruments taken together may be described as constituting the measures before us.

7.17 We first recall that, pursuant to Article 3.3 of the DSU, dispute settlement proceedings under the DSU may be initiated in "situations in which a Member considers that any benefits accruing to it directly or indirectly under the covered agreements are being impaired by measures taken by another Member" (emphasis added). With respect to the notion of "measure" in this context, the Appellate Body has observed that:

"This phrase identifies the relevant nexus, for purposes of dispute settlement proceedings, between the 'measure' and a 'Member'. In principle, any act or omission attributable to a WTO Member can be a measure of that Member for purposes of dispute settlement proceedings. The acts or omissions that are so attributable are, in the usual case, the acts or omissions of the organs of the state, including those of the executive branch."187

- 7.18 In principle, therefore, any act or omission attributable to a Member can be a measure for the purposes of dispute settlement proceedings. We also note the Appellate Body's determination that "the parties, and, in particular, the complainant, and the panel enjoy a certain latitude in defining the relevant measures". 188
- In this instance, Mexico's claims relate to certain provisions of a law (the DPCIA) and of a regulation (sections of the Federal Code), as well as to a federal court ruling (the *Hogarth* ruling), and Mexico invites the Panel to consider these measures as a whole and to make findings in relation to them taken together. The United States has not objected to this approach.
- In addressing this issue, we first note that it has not been suggested in these proceedings that any of these legal instruments taken in isolation would not constitute an "act or omission of the organs of the state" attributable to the United States. We further note that the DPCIA and the implementing regulations constitute legislative or regulatory acts of the federal authorities, while the court ruling constitutes an act of the judicial branch. Each of these normative instruments is a priori capable of constituting a measure attributable to the United States, which may be challenged in dispute settlement proceedings under the DSU.
- 7.21 The question we must consider, however, is whether it is appropriate to consider these measures jointly in our analysis of Mexico's claims, and make findings based on their combined operation, rather than on the basis of each individual measure separately.

¹⁸⁷ Appellate Body Report, US - Corrosion-Resistant Steel Sunset Review, para. 81 and footnote 79. See also Appellate Body Report, *Australia – Apples*, para. 171.

Appellate Body Report, *Australia – Apples*, para. 181.

189 We note that in *Brazil – Retreaded Tyres*, the panel examined whether discrimination arose from the importation of used tyres through court injunctions. In US – Section 211 Appropriations Act the challenged law contained provisions mandating that no US court shall recognize, enforce or otherwise validate any assertion of rights or trademarks made under certain circumstances. The panel noted that the only application of the challenged law by US courts to date of which the parties were aware was the Havana Club Holdings, S.A. v Galleon S.A. (Havana Club Holdings, S.A. v. Galleon S.A., 203 F.2d 116 (2d Cir. 2000)). Panel Report, US -Section 211 Appropriations Act, para. 2.13.

- In considering this issue, we find it useful to review how the various instruments cited by Mexico function and relate to each other. First, with respect to the DPCIA, the United States has explained that the DPCIA, codified at 16 U.S.C. 1385, establishes the conditions for use of a "dolphin-safe" label on tuna products. The DOC is legally bound to ensure, through the NOAA and the NMFS, that the criteria set out in the statute are satisfied. In addition, the United States has defined the two sections of the Code of Federal Regulations as the "DPCIA implementing regulations" and has stated that "[T]he DPCIA and implementing regulations are the source and authority for understanding the conditions on labelling tuna products dolphin-safe; NOAA Form 370 merely reflects what the law and regulations provide. As explained by the United States, Section 216.91 sets out conditions for the use of the dolphin-safe label and Section 216.92 contains provisions to ensure that tuna caught by certain vessels is labelled dolphin-safe only if the conditions set out in the DPCIA have been met. Given the hierarchical and operational link between the DPCIA and the relevant sections of the Code of Federal Regulation, it is not clear that the latter could be operational on their own or totally independently without the authority of the DPCIA.
- 7.23 Finally, the Court ruling challenged by Mexico, the *Hogarth* ruling, finds its origin in a challenge to a Secretarial finding mandated by a provision in the DPCIA, which required a determination as to whether the intentional deployment on or encirclement of dolphins with purse seine nets was having a significant adverse impact on any depleted dolphin stock in the ETP. Therefore, in the absence of the DPCIA and actions taken under it, the court ruling would not have existed. In addition, the outcome of that Court ruling has had a direct impact on the rules that are applied under the DPCIA, because the fulfilment of certain conditions foreseen in the DPCIA was dependent on the results of the secretarial findings that were rendered void by the rulings. ¹⁹⁵
- 7.24 To summarize, together and collectively, the various provisions in the different legal instruments identified by Mexico, including the *Hogarth* ruling, set out the terms of the US "dolphin-safe" labelling scheme, as currently applied by the United States. We also note that the United States does not object to Mexico's request to consider the various instruments together and that it has articulated its defence in these proceedings on the basis of the measures taken together. In light of these elements, we see merit in considering these closely related instruments together as a single measure for the purposes of this dispute.
- 7.25 We also note that a comparable issue has arisen in two cases relating to SPS measures (*Japan Apples* and *Australia Apples*), where the panels considered whether various requirements imposed by Japan and Australia respectively, and embodied in different instruments, should be treated

¹⁹¹ United States' response to Panel question No. 3, paras. 3-6.

¹⁹⁰ Panel question No. 3.

¹⁹² In that regard see Panel Report, *US – Export Restraints*, para. 8.91.

¹⁹³ United States' response to Panel question No. 7, paras. 19.

¹⁹⁴ The United States has explained that: "Regulations pertaining to the use of dolphin-safe label are set out in the U.S. Code of Federal Regulations (CFR). Mexico challenges the provisions set out at 50 CFR 216. 91 and 216.92. These provisions reflect the conditions for use of the dolphin-safe label on tuna products set out in the DPCIA. Consistent with the DPCIA, section 216.91 sets out the conditions for use of the dolphin-safe label based on whether the tuna was caught in a fishery where there is a regular and significant association between tuna and dolphins or regular and significant mortalities or injuries of dolphins. Section 216. 91 also clarifies that these conditions only apply to vessels in the ETP that have a carrying capacity greater than 362.8 metric tons, and section 216.92 contains provisions to ensure that tuna caught by such vessels is labelled dolphin-safe only if the conditions set out in the DPCIA have been met. Section 216.92 sets out the provisions applicable to domestic and imported tuna separately, although the basic requirements are the same and seek to ensure that claims that tuna is dolphin-safe comply with U.S. law." United States' first written submission, para. 31.

195 See paras. 2.16-2.19.

as a single measure or as a combination of several individual measures. ¹⁹⁶ In these cases, in addition to considering whether the different requirements might constitute a single measure for the purposes of dispute settlement under the DSU¹⁹⁷, the panel also had to consider whether they constituted a "phytosanitary measure" within the meaning of the *SPS Agreement*, an issue that is not before this Panel. Nonetheless, we find that the test developed by the panel in *Japan – Apples* provides useful guidance for our analysis. The panel in that case considered that the various requirements were interrelated and cumulatively constituted the measures actually applied by Japan to the importation of US apple fruit to protect against the entry, establishment or spread of fire blight within its territory. ¹⁹⁸ That panel therefore saw no legal, logical or factual obstacle to treating the requirements identified by the United States as a single phytosanitary measure within the meaning of the SPS Agreement. ¹⁹⁹

7.26 Similarly, we see no "legal, factual or logical obstacle"²⁰⁰ to treating the various interrelated legal instruments identified by Mexico as the basis for its claims in these proceedings as a single measure for the purposes of our findings. Accordingly, we will consider them together throughout these findings. These measures taken together are hereafter referred to as "the US dolphin-safe labelling provisions".

3. Order of analysis of the claims

7.27 Mexico raises claims under two of the covered agreements: GATT 1994 and the TBT Agreement. We must therefore determine in the first instance the order in which it is appropriate for us to address these claims.

(a) Arguments of the parties

- 7.28 In its written submissions, Mexico presented its arguments concerning GATT 1994 first, and then its arguments under the TBT Agreement. The United States presented its arguments in the same order.
- 7.29 In response to a question by the Panel, Mexico clarified that its principal claims concern the discriminatory nature of the US measures and therefore it found it logical for the Panel to first address its claims of discrimination under GATT Articles I:1 and III: 4 and Article 2.1 of the TBT Agreement; followed by an examination of its other claims under Articles 2.2 and 2.4 of the TBT Agreement.²⁰¹
- 7.30 Mexico also acknowledged that it is an accepted practice, when addressing claims under both the GATT 1994 and the TBT Agreement, to address the claims under the TBT Agreement first. However, it considered the order of examination of a complaining Member's claims important in situations where a panel exercises judicial economy in respect of one or more of the claims. Mexico also held that although a panel has discretion to exercise judicial economy, if the panel fails to make findings that are necessary to resolve the dispute it will constitute a false judicial economy and an error of law. 202

¹⁹⁶ Panel Report, *Japan – Apples*, para. 4.17. See also Panel Report, *Australia – Apples*, paras. 7.113-7.115.

¹⁹⁷ The Panel therefore had to interpret Articles 6.2 and 19. 1 of the DSU to verify whether a matter brought before the DSB could refer to several "measures". Panel Report, *Japan – Apples*, para. 8.11.

¹⁹⁸ Panel Report, *Japan – Apples*, para. 8.16.

¹⁹⁹ Panel Report, *Japan – Apples*, para. 8.17.

²⁰⁰ Panel Report, *Japan – Apples*, para. 8.17.

²⁰¹ Mexico's response to Panel question No. 1, paras. 1-2.

²⁰² Mexico's response to Panel question No. 114, para. 68.

- 7.31 Citing the Appellate Body Report in *EC Bananas III*, Mexico considered that in the case at hand it was essential for an effective resolution of the dispute that the Panel rule on all its claims because of: "(i) the nature of the measures at issue; (ii) the fact that this is the first time that such measures have been subject to dispute settlement under the DSU; (iii) the differences in the wording and potential application of the provisions of the GATT 1994 and the TBT Agreement to the measures; and (iv) the importance of the effective discipline of such non-tariff measures to developing country Members such as Mexico." Mexico emphasized that this final reason was particularly important to developing countries who are, in its view, more exposed to the adverse effects of non-tariff measures. Mexico concluded that for those reasons, the order of its claims was not important and the Panel should examine all the claims in the logical order it had exposed. ²⁰³
- 7.32 Mexico further explained that it was necessary for the Panel to rule on its discrimination claims under both the GATT 1994 and the TBT Agreement because the nature, scope and application of the claims under Articles I:1, III:4, and 2.1 are different and address different rights and obligations which, in turn, will have different implications during the implementation phase of this dispute. With respect to its claims under Articles 2.2 and 2.4 of the TBT Agreement, Mexico specified that these claims relate to aspects of the US measures other than discrimination and are necessary to address the trade restrictive effects of the US measures that exist independently of the discrimination. It concluded that due to their different nature, scope and application, none of its claims overlap and all are necessary to an effective resolution of the dispute.
- 7.33 The United States, in line with its reasoning that the US dolphin-safe labelling provisions are not technical regulations and therefore are not subject to the provisions of the TBT Agreement, considered it would be appropriate for the Panel to analyse first Mexico's claims under the GATT 1994.²⁰⁷
- 7.34 The United States agreed with Mexico that judicial economy gives panels discretion to refrain from making findings which are not necessary to resolve a dispute. However, it considered that the reasons advanced by Mexico in this respect appeared unrelated to whether exercising judicial economy on any of Mexico's claims would fail to resolve the dispute.²⁰⁸
- (b) Analysis by the Panel
- 7.35 The TBT Agreement and GATT 1994 are both found in Annex 1A of the WTO Agreement, which contains all the multilateral agreements on trade in goods that form part of the results of the Uruguay Round.
- 7.36 The Appellate Body has commented on the task of the interpreter when considering two agreements contained in Annex 1A, *in casu*, the Agreement on Agriculture and the SCM Agreement:

"[T]he Agreement on Agriculture and the SCM Agreement "are both Multilateral Agreements on Trade in Goods contained in Annex 1A of the Marrakesh Agreement Establishing the World Trade Organization (the 'WTO Agreement'), and, as such, are both 'integral parts' of the same treaty, the WTO Agreement, that are 'binding on all Members'". Furthermore, as the Appellate Body has explained, "a treaty interpreter

²⁰³ Mexico's response to Panel question No. 1, paras. 4-5.

Mexico's response to Panel question No. 114, para. 69.

²⁰⁵ Mexico's response to Panel question No. 114, para. 70.

²⁰⁶ Mexico's response to Panel question No. 114, para. 71.

²⁰⁷ United States' response to Panel question No. 1, para. 1.

²⁰⁸ United States' comments on Mexico's response to Panel question No. 114, para. 46.

must read all applicable provisions of a treaty in a way that gives meaning to *all* of them, harmoniously". ²⁰⁹

- 7.37 Similarly, the TBT Agreement and GATT 1994 are both "integral parts" of the WTO Agreement and binding on all Members. The question we must consider, in reading the provisions of the two agreements harmoniously, is whether it is appropriate to address the claims presented under these two agreements in any particular order.
- 7.38 We first note that panels are in principle free to consider the claims before them in the manner that they consider most appropriate to the resolution of the matter before them. Nonetheless, "panels must ensure that they proceed on the basis of a properly structured analysis to interpret the substantive provisions at issue". As the Appellate Body also expressed it:

"[I]n each case it is the nature of the relationship between two provisions that will determine whether there exists a mandatory sequence of analysis which, if not followed, would amount to an error of law. In some cases, this relationship is such that a failure to structure the analysis in the proper logical sequence will have repercussions for the substance of the analysis itself."²¹²

- 7.39 Thus, in light of the Appellate Body's guidance on this matter, we will examine the nature of the relationship between GATT 1994 and the TBT Agreement so as to ascertain whether a particular order of analysis of the claims is appropriate.
- 7.40 We note in this respect the Appellate Body's determination that:

²⁰⁹ Appellate Body Report, *US – Upland Cotton*, para. 549. We also note the Appellate Body's analysis of the relationship between the SCM Agreement and GATT 1994:

The authors of the new WTO regime intended to put an end to the fragmentation that had characterized the previous system. This can be seen from the preamble to the WTO Agreement which states, in pertinent part:

Resolved, therefore, to develop an integrated, more viable and durable multilateral trading system encompassing the General Agreement on Tariffs and Trade, the results of past trade liberalization efforts, and all of the results of the Uruguay Round of Multilateral Trade Negotiations.

Article II:2 of the *WTO Agreement* also provides that the Multilateral Trade Agreements are "integral parts" of the *WTO Agreement*, "binding on all Members". The single undertaking is further reflected in the Articles of the *WTO Agreement* on original membership, accession, non-application, acceptance and withdrawal. Furthermore, the *DSU* establishes an integrated dispute settlement system which applies to all the "covered agreements", allowing all the provisions of the *WTO Agreement* relevant to a particular dispute to be examined in one proceeding. Appellate Body Report, *Brazil – Desiccated Coconut*, p.18.

The Appellate Body has thus determined that:

"As a general principle, panels are free to structure the order of their analysis as they see fit. In so doing, panels may find it useful to take account of the manner in which a claim is presented to them by a complaining Member. Furthermore, panels may choose to use assumptions in order to facilitate resolution of a particular issue or to enable themselves to make additional and alternative factual findings and thereby assist in the resolution of a dispute should it proceed to the appellate level".(Appellate Body Report, *Canada – Wheat Exports and Grain Imports*, paras. 126).

As the Appellate Body found in *US – Shrimp* and *Canada – Autos*, panels that ignore or jump over a prior logical step of the analysis run the risk of compromising or invalidating later findings. This risk is compounded in the case of two legally interrelated provisions, where one of those provisions must, as a matter of logic and analytical coherence, be analysed before the other, as is the case with subparagraphs (a) and (b) of Article XVII:1 of the GATT 1994.

Appellate Body Report, *Canada – Wheat Exports and Grain Imports*, para. 127. ²¹² Appellate Body Report, *Canada – Wheat Exports and Grain Imports*, para. 109.

"[A] provision of an agreement included in Annex 1A of the WTO Agreement (including the SCM Agreement), and a provision of the GATT 1994, that have identical coverage, both apply, but ... the provision of the agreement that 'deals specifically, and in detail' with a question should be examined first."²¹³

- 7.41 In the present dispute, provisions of the TBT Agreement and of the GATT 1994 are both invoked. As expressed in its Preamble, the TBT Agreement reflects Members' desire to "further the objectives of GATT 1994". The Appellate Body observed that "it does so through a specialized legal regime that applies solely to a limited class of measures. For these measures, the TBT Agreement imposes obligations on Members that seem to be different from, and additional to, the obligations imposed on Members under the GATT 1994". 214
- 7.42 In addition, the Interpretative Note to Annex 1A to the WTO Agreement clarifies that, in the event of a conflict between a provision of the GATT 1994 and a provision of another Annex 1A Agreement, the latter will prevail to the extent of the conflict. In the present dispute, the provisions of the TBT Agreement would therefore prevail over those of GATT 1994 in such a situation.
- 7.43 These considerations suggest to us that the TBT Agreement "deals in detail, and specifically" with the matters that it addresses. Therefore, where claims under GATT 1994 are presented in parallel with claims under the TBT Agreement, claims under the TBT Agreement should be considered first. This is not modified, in our view, by the fact that there is some uncertainty about the extent to which the provisions invoked under the TBT Agreement are in fact applicable.
- 7.44 We note that this was also the approach followed by the panel in EC Asbestos:

"According to the Appellate Body in *European Communities – Regime for the Importation, Sale and Distribution of Bananas*, when the GATT 1994 and another Agreement in Annex 1A appear a priori to apply to the measure in question, the latter should be examined on the basis of the Agreement that deals 'specifically, and in detail,' with such measures. In this particular case, as the parties do not agree on the legal nature of the measure itself (technical regulation covered by the *TBT Agreement* or general ban coming under the scope of the GATT 1994 alone), it is difficult at this stage to determine which Agreement, either the GATT 1994 or the *TBT Agreement*, deals with the measure in question most specifically and in the most detailed manner without undertaking an in-depth examination of the measure in the light of each Agreement.

In order to decide upon the order in which our consideration should proceed, in the way suggested by the Appellate Body, the hypothesis should be that, if the Decree is a 'technical regulation' within the meaning of the *TBT Agreement*, then the latter would deal with the measure in the most specific and most detailed manner. Consequently, in our view it must first be determined whether the Decree is a technical regulation within the meaning of the *TBT Agreement*. If this is the case, we shall start considering this case by examining the ways in which the Decree violates

²¹³ Appellate Body Report, US – Softwood Lumber IV, para. 134.

²¹⁴ Appellate Body Report, *EC – Asbestos*, para. 80.

²¹⁵ (footnote original) Adopted on 25 September 1997, WT/DS27/AB/R, hereinafter "European Communities – Bananas", para. 204.

the *TBT Agreement*. If we find that the Decree is not a 'technical regulation', we shall then immediately start to consider it in the context of the GATT 1994."²¹⁶

- 7.45 We agree with the reasoning and approach reflected in these findings and adopt them for the purposes of our examination of Mexico's claims. In the present case, the GATT 1994 and the TBT Agreement have both been invoked, and as in EC Asbestos, the parties disagree whether the measures at issue constitute technical regulations within the meaning of the TBT Agreement.
- 7.46 Accordingly, taking into account the specificity of the TBT Agreement and its precedence over GATT 1994 in the event of a conflict between provisions of the two agreements, we find it appropriate to consider first Mexico's claims under the TBT Agreement.
- 7.47 This determination is without prejudice to the question of whether the provisions of the TBT Agreement invoked by Mexico are applicable to the measures at issue. We will consider this question in the context of our examination under the TBT Agreement in Section B below. This determination is also without prejudice to the question of exercise of judicial economy, which Mexico has referred to in relation to the order of analysis of the claims, but which is, in our view, a distinct matter, to be decided at a later stage of our analysis.

B. MEXICO'S CLAIMS UNDER THE TBT AGREEMENT

- 7.48 Mexico has presented claims under Articles 2.1, 2.2 and 2.4 of the TBT Agreement. All three of these provisions relate to "technical regulations". The United States, however, considers that the measures at issue do not constitute "technical regulations" within the meaning of the TBT Agreement.
- 7.49 We must therefore determine, as a threshold matter, whether the measures at issue constitute a "technical regulation", as this question determines the applicability of the provisions invoked by Mexico under the TBT Agreement. If we determine that the US dolphin-safe labelling provisions do constitute a "technical regulation", we will then consider further Mexico's claims under each of the three provisions that it has invoked.

1. Whether the measures at issue constitute a "technical regulation" within the meaning of Annex 1.1 of the TBT Agreement

7.50 Annex 1.1 of the TBT Agreement defines a "technical regulation" as follows:

"Document which lays down product characteristics or their related processes and production methods, including the applicable administrative provisions, with which compliance is mandatory. It may also include or deal exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method".

7.51 Pursuant to Article 3.2 of the DSU, we must interpret this term in accordance with customary rules of interpretation of public international law. As recognized by the Appellate Body, these customary rules of treaty interpretation have been codified in Articles 31 to 33 of the *Vienna Convention on the Law of Treaties* ("Vienna Convention").²¹⁷

²¹⁶ Panel Report, EC – Asbestos, paras. 8.16–8.17.

²¹⁷ Appellate Body Report, *US – Gambling*, para. 159; Appellate Body Report, *Japan – Alcoholic Beverages II*, p. 10; Appellate Body Report, *US – Gasoline*, p. 17. Articles 31 and 32 of the *Vienna Convention* read as follows:

- 7.52 The general rule of interpretation contained in Article 31 of the *Vienna Convention* has been construed by the Appellate Body as requiring that the interpretation of a treaty provision be primarily based on the text of the treaty provision itself.²¹⁸ The words contained in that text must be given their ordinary meaning in their context and in light of the object and purpose of the treaty in question.²¹⁹ We also take note of the Appellate Body's conclusion that "[o]ne of the corollaries of the general rule of interpretation in the *Vienna Convention* is that interpretation must give meaning and effect to *all* the terms of a treaty".²²⁰
- 7.53 The Appellate Body has interpreted the definition of a "technical regulation" in EC Asbestos and EC Sardines. These rulings provide useful guidance for our analysis in this case. In particular, the Appellate Body established in EC Asbestos a three-tier test for determining whether a measure is a "technical regulation" under the TBT Agreement. This test was then followed in EC Sardines. The three elements under this test are derived from the wording of the definition in Annex 1.1 of the TBT Agreement. Accordingly, a measure is a "technical regulation", if:
 - (a) the measure applies to an identifiable product or group of products;
 - (b) it lays down one or more characteristics of the product; and

Article 31

General rule of interpretation

- 1. A treaty shall be interpreted in good faith in accordance with the ordinary meaning to be given to the terms of the treaty in their context and in the light of its object and purpose.
- 2. The context for the purpose of the interpretation of a treaty shall comprise, in addition to the text, including its preamble and annexes:
- (a) any agreement relating to the treaty which was made between all the parties in connection with the conclusion of the treaty;
- (b) any instrument which was made by one or more parties in connection with the conclusion of the treaty and accepted by the other parties as an instrument related to the treaty.
- 3. There shall be taken into account, together with the context:
- (a) any subsequent agreement between the parties regarding the interpretation of the treaty or the application of its provisions;
- (b) any subsequent practice in the application of the treaty which establishes the agreement of the parties regarding its interpretation;
- (c) any relevant rules of international law applicable in the relations between the parties.
- 4. A special meaning shall be given to a term if it is established that the parties so intended.

Article 32

Supplementary means of interpretation

Recourse may be had to supplementary means of interpretation, including the preparatory work of the treaty and the circumstances of its conclusion, in order to confirm the meaning resulting from the application of article 31, or to determine the meaning when the interpretation according to article 31:

- (a) leaves the meaning ambiguous or obscure; or
- (b) leads to a result which is manifestly absurd or unreasonable.

²¹⁸ Appellate Body Report, *Japan – Alcoholic Beverages II*, p. 11.

²¹⁹ Appellate Body Report, *US – Gasoline*, p. 17.

²²⁰ Appellate Body Report, *US – Gasoline*, p. 23.

²²¹ Appellate Body Report, *EC – Asbestos*, paras. 66-70.

²²² Appellate Body Report, *EC – Sardines*, para. 176.

²²³ Appellate Body Report, *EC – Sardines*, para. 176.

- (c) compliance with the product characteristics is mandatory.
- 7.54 We therefore now consider whether the US measures challenged by Mexico constitute a technical regulation within the meaning of Annex 1.1 of the TBT Agreement, on the basis of these three elements, as articulated by the Appellate Body.
- 7.55 Accordingly, we will consider the following issues in turn:
 - (a) Whether the US dolphin-safe labelling provisions apply to an identifiable group of products;
 - (b) Whether they lay down one or more characteristics of these products;
 - (c) Whether compliance with them is mandatory within the meaning of Annex 1.
- (a) Whether the measures apply to an identifiable product or group of products
- 7.56 Mexico notes that the term "tuna product" is defined in Section 1385(c)(5) of the DPCIA. Mexico also observes that the provisions of the DPCIA apply specifically to a "tuna product" as defined by such provisions. Therefore, according to Mexico, the list of items identified by this definition, constitute an "identifiable group of products" to which the document applies. 224
- 7.57 The United States does not disagree with Mexico that the US dolphin-safe provisions apply to an identifiable group of products (tuna products). However, as further discussed in the following subsections, the United States disputes Mexico's assertion that such provisions set out product characteristics and that they are mandatory regulations. 225
- 7.58 We note that in EC Asbestos, the Appellate Body clarified that while a technical regulation must be applicable to an *identifiable* product or group of products, this did not mean that the product or group of products needed to be expressly *identified* in the document:
 - "A 'technical regulation' must, of course, be applicable to an *identifiable* product, or group of products. Otherwise, enforcement of the regulation will, in practical terms, be impossible. ... However, in contrast to what the Panel suggested, this does not mean that a 'technical regulation' must apply to 'given' products which are actually named, identified or specified in the regulation. Although the TBT Agreement clearly applies to 'products' generally, nothing in the text of that Agreement suggests that those products need be named or otherwise expressly identified in a 'technical regulation'. Moreover, there may be perfectly sound administrative reasons for formulating a 'technical regulation' in a way that does not expressly identify products by name, but simply makes them identifiable for instance, through the 'characteristic' that is the subject of regulation."
- 7.59 We must therefore determine whether the US dolphin-safe labelling provisions apply to an identifiable product or group of products.
- 7.60 As described in Section II.A above, the DPCIA establishes the conditions under which tuna products may be labelled dolphin-safe. The provisions of the DPCIA relate specifically to two types

²²⁴ Mexico's first written submission, para. 196.

²²⁵ United States' first written submission, para. 128.

Appellate Body Report, EC – Asbestos, para. 70. See also Appellate Body Report, EC – Sardines, paras. 176 and 180.

of goods: "tuna" and "tuna products". They regulate the "tuna products" containing "tuna" that can be labelled as dolphin-safe. As noted above, the DPCIA defines "tuna products" in Section 1385(c)(5) as "food items which contains tuna and which has been processed for retail sale, except perishable sandwiches, salads, or other products with a shelf life of less than 3 days". 228

- 7.61 Section 216.3 of Title 50 of the Code of Federal Regulations identifies the definitions applicable to Section 216, and makes the following cross-reference: "in addition to the definitions contained in the MMPA and unless the context otherwise require, in this part 216 ... 'Tuna product' means any food product processed for retail sale and intended for human or animal consumption that contains an item listed in Section 216.24(f)(2)(i) or (ii), but does not include perishable items with a shelf life of less than 3 days." Pursuant to Section 216.24(f)(2)(i) and (ii), tuna products are those products containing one of the listed species of tuna. By virtue of Section 216.3, this definition applies to the regulations challenged by Mexico, i.e. Title 50, Sections 216.91 and 216.92.
- 7.62 In light of the above, the Panel agrees with Mexico that the US dolphin-safe labelling provisions apply to an "identifiable" product or group of products, that is, "tuna products", as defined in the DPCIA and in Section 216.3 of Title 50 of the Code of Federal Regulations.
- (b) Whether the US dolphin-safe labelling provisions lay down one or more "characteristics" of the products
- (i) Arguments of the parties
- 7.63 Mexico argues that the US measures govern the conditions under which a tuna product can be labelled as "dolphin-safe". According to Mexico, this requirement is a product characteristic of the tuna product that is laid down by the United States' measures.²²⁹
- 7.64 Mexico observes that the definition of "technical regulation" contained in Annex 1.1 expressly includes "marking or labelling requirements". Mexico also notes that the Appellate Body has established that "a 'labelling requirement' is a product characteristic". Based on the Appellate Body's ruling in EC Asbestos, Mexico submits that "product characteristics" include, not only features and qualities intrinsic to the product itself, but also related "characteristics", such as means of identification, the presentation and the appearance of a product. Mexico also recalls that the panel in EC Trademarks and Geographical Indications (Australia) came to a similar conclusion and considered that the label on a product is itself a product characteristic. 230
- 7.65 Mexico further clarifies that in presenting its claims under the TBT Agreement, it has applied the interpretation of "technical regulation" set out by the Appellate Body in EC Asbestos and EC Sardines. Mexico recalls that on those occasions, the Appellate Body ruled that the "heart of the definition" of "technical regulation" under Annex 1.1 is that a "document" must "lay down" "product

²²⁷ Exhibit US-5 and Appendix A of Mexico's first written submission.

Exhibit US-5 and Appendix A of Mexico's first written submission. The United States specifies in its first written submission that tuna products include canned or pouched tuna, whole frozen tuna, frozen tuna steaks, frozen tuna filets, tuna loins for canning, tuna in glass jars, tuna burgers, fish balls and fish cakes that contain tuna, and sushi grade frozen tuna. (United States' first written submission, fn 4). Mexico contends that the most common form of tuna products is tuna in retail-ready cans or pouches. (Mexico's first written submission, para. 116).

²²⁹ Mexico's first written submission, para. 198. ²³⁰ Mexico's first written submission, para. 197.

characteristics".²³¹ Therefore, Mexico's position is that the measures at issue "relate ... to a labelling requirement that is a product characteristic".²³²

- 7.66 The United States contends that the US dolphin-safe labelling provisions do not set out product characteristics for tuna products. Instead, they specify the conditions under which tuna products may be labelled dolphin-safe. Therefore, although the United States recognizes that the US dolphin-safe provisions "set out requirements that must be met for tuna to be labeled dolphin-safe", it submits that the US dolphin-safe provisions do not specify the product characteristics that tuna products must meet (or not meet) to be sold on the United States' market.²³³
- 7.67 In this context, the United States argues that reading together the first and the second sentences of the definition of "technical regulation" in Annex 1.1 of the TBT Agreement makes it clear that technical regulations are "documents with which compliance is mandatory and that 'lay down product characteristics or their related processes and production methods' or 'deal exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method' or both".
- 7.68 The United States submits that the Appellate Body in *EC Asbestos* "appears to mistakenly have read the second sentence of Annex 1.1 of the TBT Agreement as providing examples of 'product characteristics' covered by the first sentence of the definition". Moreover, the United States submits that the panel in *EC Trademarks and Geographical Indications (Australia)* "repeated this same mistake". According to the United States, the second sentence of the definition of "technical regulation" does not contain examples of "product characteristics"; it rather sets out aspects other than product characteristics that may be the subject of a document with which compliance is mandatory and thus fall within the definition of a "technical regulation".²³⁵
- 7.69 The United States discusses the relationship between the two sentences of Annex 1.1 in light of the phrase "it may also or exclusively deal with". According to the United States, the word "it" in this phrase refers back to the word "document" in the first sentence, "such that the document may in addition or instead deal with aspects that are not considered product characteristics such as terminology or labelling requirements". Furthermore, the United States claims that the Appellate Body's interpretation of the second sentence of Annex 1.1 appears to ignore the word "also". According to the United States, the use of the word "also" supports the view that the second sentence is additional to the first. Hence, the United States submits that the Appellate Body's approach "appears not to give full effect to the terms of the relevant provisions of the TBT Agreement" and, therefore, should not be followed. 236
- 7.70 The United States notes that the US dolphin-safe provisions "set out the requirements that must be met for tuna to be labeled dolphin-safe". However, the United States submits that this fact and the fact that these provisions make it unlawful to sell tuna products that are labelled "dolphin-safe" without complying with the applicable conditions under the DPCIA labelling scheme, does not prove that the US dolphin-safe provisions constitute a technical regulation. According to the United States, these facts prove "[a]t most... that the U.S. dolphin-safe labelling provisions are

²³¹ Mexico's response to Panel question No. 46, para. 118.

²³² Mexico's response to Panel question No. 47, para. 127.

²³³ United States' first written submission, para. 129.

²³⁴ United States' first written submission, para. 126.

²³⁵ United States' first written submission, fn 141.

²³⁶ United States' first written submission, fn 141; *see also*, United States' first oral statement, para. 40.

²³⁷ United States' first written submission, para. 129, *see also* United States' second written submission, para. 96.

'labelling requirements'''. 238 It also submits that "the term 'labelling requirement' should be construed to have the same meaning in both the definition of a technical regulation and the definition of a standard". 239 Therefore, in the United States' view, the US dolphin-safe provisions "establish labelling" requirements with which compliance is not mandatory". 240

Analysis by the Panel (ii)

7.71 The Panel recalls the definition of a "technical regulation" in Annex 1.1 of the TBT Agreement, as set out in paragraph 7.50 above. We recall also the Appellate Body's finding that a document meets the second element of its three-tier test for a "technical regulation" if it "lays down one or more characteristics of the product". ²⁴¹ Although this finding was made with reference only to the core concept of "product characteristics" ²⁴², we understand this second element of the test as relating more generally to the subject matter of the measure, as defined in Annex 1.1. What we must ascertain, therefore, in this part of our analysis, is whether the matter addressed in the US dolphin-safe labelling provisions falls within the scope of the subject matters of a technical regulation, as defined in Annex 1.1.

7.72 We note in this respect that both sentences of Annex 1.1 provide indications as to the subject matter, or contents, of technical regulations. The first sentence establishes that a technical regulation may lay down "product characteristics or their related processes and production methods" (emphasis added). The second sentence ("It may also deal with ..."), further elaborates on the subject-matter of technical regulations, and enumerates some specific items that technical regulations may also "include or deal exclusively with". As described by the Appellate Body:

"In addition, according to the definition in Annex 1.1 of the TBT Agreement, a 'technical regulation' may set forth the 'applicable administrative provisions' for products which have certain 'characteristics'. Further, we note that the definition of a 'technical regulation' provides that such a regulation 'may also include or deal exclusively with terminology, symbols, packaging, marking or labelling requirements', (emphasis added) The use here of the word 'exclusively' and the disjunctive word 'or' indicates that a 'technical regulation' may be confined to laying down only one or a few 'product characteristics'." (emphasis original)²⁴³

From these elements, it is clear, in our view, that the subject-matter of a technical regulation may be "confined", as the Appellate Body expressed it, to one of the elements enumerated in the second sentence, including "labelling requirements", "as they apply to a product, process or production method".

²³⁸ United States' second written submission, para. 89.

²³⁹ United States' second written submission, para. 91.

²⁴⁰ United States' second written submission, para. 96.

²⁴¹ Appellate Body Report, *EC – Sardines*, para. 176.

[&]quot;The heart of the definition of a 'technical regulation' is that a 'document' must 'lay down' – that is, set forth, stipulate or provide - 'product characteristics. The word characteristic has a number of synonyms that are helpful in understanding the ordinary meaning of that word, in this context. Thus, the characteristics of a product include, in our view, any objectively definable 'features', 'qualities', 'attributes', or other 'distinguishing mark' of a product. Such 'characteristics' might relate, inter alia, to a product's composition, size, shape, colour, texture, hardness, tensile strength, flammability, conductivity, density, or viscosity. In the definition of a technical regulation in Annex 1.1, the TBT Agreement itself gives certain examples of 'product characteristics' – 'terminology, symbols, packaging, marking or labelling requirements'." Appellate Body Report, EC - -Sardines, para. 189.

²⁴³ Appellate Body Report, *EC – Asbestos*, para. 67.

- In the present case, Mexico and the United States agree that the measures at issue establish the conditions under which tuna products may be labelled "dolphin-safe". 244 Both parties also acknowledge that the US dolphin-safe labelling provisions establish "labelling requirements" within the meaning of Annex 1 of the TBT Agreement.²⁴⁵ Moreover, they both consider that the term "labelling requirements" has the same meaning in the definition of a "technical regulation" and the definition of a "standard", ²⁴⁶ and that the meaning to be ascribed to the term "labelling requirement" in this context is that of a set of criteria or conditions that must be met before a label can be used.²⁴⁷
- We note that Annex 1 of the TBT Agreement does not contain a definition of the term "labelling requirement". However, it provides that the terms used in the TBT Agreement have the same meaning as the terms defined by the sixth edition of the ISO/IEC Guide 2: 1991, except if otherwise provided in Annex 1. The ISO/IEC Guide 2: 1991 defines the term "requirement" as a "provision that conveys criteria to be fulfilled". ²⁴⁸ The parties' own definition of the term "labelling" requirements" as explained in the previous paragraph is consistent with this definition.
- In the present case, the US dolphin-safe labelling provisions define the conditions that must be met in order to bear a "dolphin-safe" label. In so doing, they "convey criteria to be fulfilled" in order to qualify for such label. They therefore lay down "labelling requirements" within the meaning of Annex 1.1. Further, we note that the second sentence of Annex 1.1 refers to labelling requirements "as they apply to a product, process or production method". We must therefore also determine whether the labelling requirements laid down in the US dolphin-safe labelling provisions apply to "a product, process or production method".
- In response to a question by the Panel in this respect, Mexico observed that this phrase means "labelling requirements as they make use of or administer a product, process or production method", and that if the second sentence is referred to, "the labelling requirement in the US measures clearly applies to a "product", namely tuna products" The United States, for its part, considers that the phrase "they apply to a product, process or production method" clarifies that the second sentence covers terminology, symbols, packaging, marking and labelling requirements as they apply to a product, process or production method", and that the words "as they apply to" may be understood as meaning terminology, symbols, packaging and marking requirements that "refer to, concern or relate to a product, process or production method". In this dispute, the United States considers that its dolphin-safe labelling provisions "concern both a product - tuna products - and a production method".250
- We agree with the United States' characterization of the terms "as they apply to" as meaning that the labelling requirements and other elements listed in the second sentence must relate to and concern "a product, process or production method". We also note that there is no disagreement among the parties that the labelling requirements laid down in the US dolphin-safe labelling provisions "apply to" a product, namely tuna products. We agree with this determination. We are

Mexico's first written submission, para. 198; Mexico's second written submission, para. 1; United States' first written submission, para. 129; United States' second written submission, para. 89.

²⁴⁵ The United States also considers, however, that, these labelling requirements are not "mandatory" within the meaning of Annex 1.1 and thus do not fall within the scope of this provision. This question is considered separately in the next section.

²⁴⁶ Mexico's response to Panel question No. 48, para. 128; United States' second written submission,

para. 91.

247 Mexico's response to Panel question No. 48, para. 131; United States' second written submission, para. 91.

248 See also para. 7.5 of ISO/IEC 2004 Guide 2, eight edition.

Mexico's response to Panel question No. 119, para. 86.

²⁵⁰ United States' response to Panel question No. 119, para. 43.

therefore satisfied that the measures at issue lay down labelling requirements, as they apply to a product, process or production method and that the subject-matter of the measures therefore falls within the scope of the second sentence of Annex 1.1.

- 7.79 We do not find it necessary to consider in addition whether the labelling requirements in the US dolphin-safe labelling provisions also fall within the scope of the first sentence as "product characteristics or related production or processing methods", since, as the Appellate Body has observed, the terms of the second sentence make it clear that the subject-matter of a technical regulation may be confined to one of the items enumerated in the second sentence.²⁵¹
- Whether compliance with the US dolphin-safe labelling provisions is "mandatory" within the (c) meaning of Annex 1.1
- *(i)* Arguments of the parties
- Mexico submits that the US dolphin-safe provisions constitute a mandatory regulation.²⁵² 7.80 According to Mexico, the statutory and regulatory provisions that make up the dolphin-safe labelling provisions, as interpreted by a court decision, contain mandatory language that specifies that it is unlawful to include on the label of any tuna product offered for sale in the United States the term "dolphin-safe" or any analogous term or symbol if the product contains tuna harvested in the ETP by a large purse-seine vessel using a purse-seine net intentionally deployed on or to encircle dolphins. 253 Mexico argues that the US dolphin-safe labelling provisions involve a prohibition on the use of the "dolphin-safe" label, or the use of similar terms or symbols, on Mexican tuna products marketed in the United States, and that this prohibition remains if force even when the international standards of the AIDCP are met.²⁵⁴ Thus, in Mexico's view, the DPCIA labelling requirements are imposed in a negative form, i.e. the United States' measure require that "[t]una products offered for sale in the United States must not possess certain characteristics (i.e., distinguishing marks – the dolphin-safe label or any other analogous label or mark) unless the prescribed requirements are met". 255
- 7.81 According to Mexico, the mandatory language of the US dolphin-safe provisions is found in the provisions that specify that it is unlawful to include on the label of any tuna product offered for sale in the United States the term "dolphin-safe" or any analogous term or symbol if the product contains tuna harvested in the ETP by a large purse-seine vessel using a purse-seine net intentionally deployed on or to encircle dolphins.²⁵⁶
- Mexico clarifies that what makes the United States' measures in this dispute mandatory is not whether a label is de jure required in order to sell tuna products in the US market, but that the measures restrict retailers, consumers and producers to a single choice for labelling tuna products as dolphin-safe. According to Mexico, by virtue of the US dolphin-safe provisions it is not possible to label tuna products as dolphin-safe under more than one standard.²⁵⁷

²⁵¹ We note in this respect that the Appellate Body has also determined that "labelling requirements", as well as other items described in the second sentence of Annex 1.1, constitute "product characteristics" within the meaning of the first sentence.

Mexico's first written submission, para. 200; Mexico's second written submission, paras. 192-199.

²⁵³ Mexico's response to Panel question No. 46, para. 120.

²⁵⁴ Mexico's first written submission, para. 200; Mexico's first oral statement, paras. 3, 5, 23; Mexico's second written submission, para. 195.

²⁵⁵ Mexico's first written submission, para. 202; Mexico's second written submission, para. 195.

²⁵⁶ Mexico's second written submission, para. 194; Mexico's response to Panel question No. 46, para. 120. 257 Mexico's second written submission, para. 196.

- 7.83 Mexico submits that the US dolphin-safe provisions exclusively allow the application of the dolphin-safe label on tuna products when certain requirements established by those provisions are met. Therefore, "compliance with those requirements is the only way to be able to use a dolphin-safe label". In Mexico's view, "there are no alternative means to certify that a tuna product is dolphin-safe". According to Mexico "[t]he fact that the U.S. measures prohibit the utilization of any other standards to certify that a tuna product is dolphin-safe gives these measures a *de jure* mandatory distinction". ²⁵⁸
- 7.84 Mexico adds that the mandatory language of the US dolphin-safe provisions is also found in the enforcement provisions of the DPCIA.²⁵⁹ Mexico contends that additional support for the argument that the United States' measures are *de jure* mandatory is the fact that these measures establish surveillance and enforcement procedures if an importer tries to import tuna products bearing the dolphin-safe label when such products do not meet the requirements set out in US dolphin-safe provisions.²⁶⁰Alternatively, Mexico argues that even if the labelling scheme were not to be considered *a priori* mandatory, it is *de facto* mandatory because the market conditions in the United States are such that it is impossible to effectively market and sell tuna products without a dolphin-safe designation.²⁶¹
- 7.85 On this basis, Mexico submits that the US dolphin-safe provisions satisfy the third condition recognized by the Appellate Body for a technical regulation to be considered as such. 262
- 7.86 According to Mexico, the application of a dolphin-safe label to tuna products has "significant commercial value" in the US market "given that consumers at each consumption stage demand this label in order to sell or buy tuna products". 263
- 7.87 Mexico argues that the measures have a direct adverse effect on tuna products imported and sold in the US market, as US distribution and retail networks for tuna products are acutely aware of the dolphin safe issue and the fact that they will encounter actions such as boycotts, promoted by certain economically interested NGOs, if they carry tuna that is not designated as dolphin safe. Mexico argues that large US grocery chains have indicated that they will be unable to carry any Mexican tuna products unless the tuna products bear a US government approved dolphin safe label. Mexico also identifies indirect adverse effects on tuna caught by the Mexican tuna fleet. Hexagonal description of the description of the dolphin safe label. Hexagonal description of the dolphin safe label. Hexagonal description of the dolphin safe label and the dolphin safe label. Hexagonal description of the dolphin safe label and the dolphin safe label. Hexagonal description of the dolphin safe label and the dolphin safe label and the dolphin safe label.
- 7.88 Mexico acknowledges that the United States' argument that carrying the dolphin-safe label is voluntary and that it is legal to sell tuna products in the United States that do not bear this label, "may be correct" from a *de jure* perspective. However, Mexico submits that, from a *de facto* perspective, the label is essential if the tuna products are to be sold in the principal distribution channels in the US market.²⁶⁶
- 7.89 Thus, Mexico claims, the US dolphin-safe provisions have "the effect of excluding Mexican tuna products from the major distribution channels in the U.S. market and creating new barriers to

²⁵⁸ Mexico's response to Panel question No. 52, paras. 141-142.

²⁵⁹ Mexico's second written submission, para. 194.

²⁶⁰ Mexico's response to Panel question No. 52, para. 143.

²⁶¹ Mexico's first written submission, para. 203; Mexico's response to Panel question No. 52, para. 147.

²⁶² Mexico's first written submission, para. 202.

²⁶³ Mexico's response to Panel question No. 52, para. 148.

²⁶⁴ Mexico's first written submission, para. 111.

²⁶⁵ Mexico indicates that the owners of the three major processed tuna brands sold in the United States "refuse to purchase tuna caught in the ETP including Mexican tuna because tuna products containing such tuna cannot be labeled as dolphin-safe". Mexico's first written submission, para. 112.

²⁶⁶ Mexico's second written submission, para. 195.

Therefore, Mexico argues that "compliance with the requirements imposed by the United States becomes mandatory for the Mexican tuna industry in order to access the U.S. market" 267

- 7.90 Mexico further submits that the situation in this dispute is very similar to that addressed by the Appellate Body in EC – Sardines. Mexico recalls that on that occasion, the challenged measure provided that only products prepared from Sardina pilchardus could be marketed as preserved sardines. 268 Mexico argues that "[m]uch like the European measures in Sardines, the U.S. measures provide that the only products that can have the term dolphin-safe or a similar term or symbol are those that meet the U.S. legal provisions". 269
- The United States responds that the US dolphin-safe provisions are not mandatory. Instead, 7.91 the United States argues, they constitute a voluntary labelling measure not covered by the definition of a "technical regulation" set out in Annex 1 of the TBT Agreement. According to the United States, the US dolphin-safe provisions do not require tuna products to be labelled "dolphin-safe".²⁷¹
- In the US view, a document dealing with labelling requirements with which compliance is mandatory is one that has the effect of prescribing or imposing labelling requirements affirmatively or by negative implication.²⁷² In other words, "a labelling scheme would be mandatory if the labelling scheme - or some other government action - prohibited products from being sold, imported, distributed or otherwise marketed that were not labeled in that way". 273 United States submits that the US dolphin-safe provisions do not prescribe or impose labelling requirements because they do not require tuna products to be labelled or to contain certain information on a label to be sold, imported, distributed or otherwise marketed. ²⁷
- The United States acknowledges that the US dolphin-safe provisions require that the label may be used only for tuna products containing tuna that was not caught during a trip in which purseseine nets were deployed on or to encircle dolphins and in a set in which no dolphins were killed or seriously injured. However, according to the United States, a voluntary labelling measure does not become a mandatory labelling requirement "simply because the measure requires what is stated on the label to be truthful". 275
- The United States further submits that in its view, "inherent in the idea of a standard is that there are certain conditions to be met in order to meet the standard". ²⁷⁶ In the United States' view, if marketers of products were permitted to claim in a label that their products conformed to a particular standard when those products did not meet the conditions to be labelled in that way, the utility of that standard would be lost.²⁷⁷ The requirement to meet these conditions does not result in the standard becoming mandatory or a technical regulation. According to the United States, arguing otherwise

²⁶⁷ Mexico's response to Panel question No. 52, para. 149.

²⁶⁸ Mexico's second written submission, para. 196.

²⁶⁹ Mexico's second written submission, para. 198.

²⁷⁰ United States' first written submission, para. 130.

²⁷¹ United States' first oral statement, para. 40; United States' second written submission, paras. 88-95.

²⁷² United States' first written submission, para. 131.

²⁷³ United States' response to Panel question No. 52, para. 125; United States' second written

submission, para. 88.

274 United States' first written submission, para. 132; United States' response to Panel question No. 52, para. 125; United States' second written submission, para. 96.

275 United States' first written submission, para. 134.

²⁷⁶ United States' first written submission, para. 135.

²⁷⁷ United States' first oral statement, para. 44.

would turn all labelling standards into technical regulations and render the definition of standard inutile, because there would be no labelling requirement with which compliance could be considered not mandatory.²⁷⁸

In the United States' view, Mexico's position "conflates the meaning of the term 'labelling requirement' with the phrase 'with which compliance is mandatory'". 27 According to the United States, "[w[hether compliance with a labelling requirement is mandatory must turn on something more than the mere fact that the labelling requirement sets out conditions under which products may (or may not) be labelled in a certain way". 280 Therefore, the United States submits that an approach that gives full effect and meaning to the term "labelling requirement" and the phrase "with which compliance is mandatory" defines the term "labelling requirement" with which compliance is mandatory as "a measure that establishes conditions under which a product may be labelled in a certain way and requires the product to be labelled in that way in order to be marketed" (emphasis in the original).²⁸¹

In relation to Mexico's alternative claim that the DPCIA labelling requirements are *de facto* mandatory, the United States submits two arguments. First, according to the United States, whether compliance with a labelling requirement is mandatory depends on whether some government action makes compliance mandatory. 282 The United States contends that Mexico's argument that the US dolphin-safe provisions have prevented Mexican tuna products from accessing the United States' market is not based on these provisions or on any other measure or government action, but on retailer and consumer preferences for tuna products that were not caught in a manner that adversely affects dolphins.²⁸³ According to the United States, "consumer or retailer preferences alone cannot determine whether a labelling requirement is mandatory".²⁸⁴ Second, the United States argues that the fact that some distribution channels in the United States do purchase and sell tuna products that are not labelled "dolphin-safe" demonstrates that the DPCIA labelling provisions are not mandatory. ²⁸⁵

Consequently, the United States also rejects Mexico's alternative argument that the 7.97 US dolphin-safe provisions are de facto mandatory because major distribution channels for tuna products would only purchase and sell dolphin-safe labelled tuna products.²⁸⁶

The United States also submits that the US dolphin-safe provisions are not like the measures in EC - Asbestos and EC - Sardines. According to the United States, compliance with those measures was mandatory because the products that did not possess the product characteristics set out in the measures challenged in those cases, could not be sold in the territory of the European Union.²⁸⁷ The United States emphasizes that under the US dolphin-safe provisions it is not prohibited to sell tuna products that fail to meet the DPCIA labelling conditions. ²⁸⁸ Furthermore, the United States

²⁷⁸ United States' first written submission, para. 135, United States' first oral statement, para. 42; United States' response to Panel question No. 52, para. 125; United States' response to Panel question No. 56, para. 128.

279 United States' second written submission, paras. 90 and 98.

²⁸⁰ United States' second written submission, para. 92.

United States' second written submission, paras. 93.

²⁸² United States' second written submission, paras. 108-09.

²⁸³ United States' second written submission, para. 110; United States' response to Panel question

United States' response to Panel question No. 57, para. 130.

285 United States' second written submission, para. 113; United States' response to Panel question

²⁸⁶ United States' response to Panel question No. 57, para. 130.

²⁸⁷ United States' second written submission, para. 104.

²⁸⁸ United States' second written submission, para. 103.

submits that a further distinguishing factor is that the US dolphin-safe provisions do not prevent marketers of tuna products from marketing them as tuna products. Finally, the United States observes that in the earlier disputes, the Appellate Body did not address the issue of mandatory compliance with a labelling requirement.²⁹⁰

Accordingly, the United States concludes that compliance with the DPCIA labelling provisions is not mandatory within the meaning of Annex 1 of the TBT Agreement.²⁹¹ Thus, the United States submits, such provisions are not technical regulations and cannot be inconsistent with the obligations set out in Article 2 of the TBT Agreement.²⁹²

(ii) Analysis by the Panel

7.100 As described in paragraph 7.53 above, the Appellate Body has determined that the third condition for a document to be considered a "technical regulation" is that compliance with the product characteristics laid down in the document is "mandatory". 293 We must therefore now determine whether compliance with the labelling requirements established by the US dolphin-safe labelling provisions is "mandatory", as claimed by Mexico, or whether, as the United States argues, they constitute "voluntary" labelling requirements.

7.101 To make this determination, we must first consider the meaning to be given to the term "mandatory" in Annex 1.1.

Interpretation of the term "mandatory" in Annex 1.1

7.102 To recall, Annex 1.1 of the TBT Agreement defines a "technical regulation" as a:

"Document which lays down product characteristics or their related processes and production methods, including the applicable administrative provisions, with which compliance is mandatory. It may also include or deal exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method." (emphasis added)

7.103 We first note that dictionary definitions of the term "mandatory" include "binding" as well as "obligatory, compulsory, not discretionary" or "required by law or mandate; compulsory". This suggests that the notion of "mandatory" may encompass the legally binding and enforceable character of the instrument, and may also relate to its contents, prescribing/imposing a certain behaviour. We also note that the ISO/IEC Guide 2 establishes that the expression "mandatory requirement", should be used to mean only "a requirement made compulsory by law or regulation". 296

²⁹⁵ Concise Oxford Dictionary (10th ed.), p. 865.

²⁸⁹ United States' second written submission, para. 104.

²⁹⁰ United States' second written submission, para. 105.

²⁹¹ United States' first oral statement, para. 45.

²⁹² United States' first written submission, para. 138.

²⁹³ Appellate Body Report, *EC – Asbestos*, paras. 66-70; Appellate Body Report, *EC – Sardines*, para. 176.

294 (OED online)

²⁹⁶ We also note the following definitions in the ISO/IEC Guide 2:

[&]quot;exclusive reference (to standards)": "reference to standards that states that the only way to meet the relevant requirements of a technical regulation is to comply with the standards referred to" (emphasis added).

[&]quot;mandatory standard": "standard the application of which is made compulsory by virtue of a general law or exclusive reference in a regulation".

7.104 In Annex 1.1, the term "mandatory" appears in the phrase "with which compliance is mandatory". In interpreting this expression, the Panel finds useful guidance in the following passage of the Appellate Body's decision in EC – Asbestos:

"The definition of a technical regulation in Annex 1.1 of the TBT Agreement also states that 'compliance' with the 'product characteristics' laid down in the 'document' must be 'mandatory'. A 'technical regulation' must, in other words, regulate the 'characteristics' of products in a binding or compulsory fashion. It follows that, with respect to products, a 'technical regulation' has the effect of prescribing or imposing one or more 'characteristics' – 'features', 'qualities', 'attributes', or other 'distinguishing mark'." (italics in the original, underlining added)

In this and subsequent cases, the Appellate Body further clarified that the product characteristics laid down in a technical regulation may "be prescribed or imposed ... in either a positive or a negative form", which means that, for instance, "the document may require, positively, that products must possess certain 'characteristics', or the document may require, negatively, that products must not possess certain 'characteristics''. ²⁹⁷

7.106 As we understand it therefore, and as interpreted by the Appellate Body, the notion of "mandatory" "compliance" in Annex 1.1 relates both to the "binding" character of the instrument and the fact that it prescribes certain characteristics or other features that the product must or must not possess.

7.107 Taking the terms of Annex 1.1 in their context, we also note that the subject-matter of a "technical regulation" and the subject-matter of a "standard" are defined in very similar terms, and it is essentially their "not mandatory" or "mandatory" character that distinguishes the two categories of instruments. For instance, labelling requirements may be equally prescribed by technical regulations and standards.²⁹⁸ However, while technical regulations may prescribe such requirements in a compulsory fashion, standards would only do so on a "not mandatory" basis. This difference is expressed by the words "with which compliance is mandatory" in the definition of a "technical regulation" in Annex 1.1 and the opposite language, i.e. "with which compliance is *not* mandatory" (emphasis added), in the definition of a "standard" in Annex 1.2.

7.108 The Explanatory Note to Annex 1.2 further emphasizes that "[f]or the purpose of this Agreement standards are defined as voluntary and technical regulations as mandatory documents" (emphasis added). This Explanatory Note clarifies the drafters' intention to make a distinction between the definition of "standard" in the TBT Agreement and that contained in the ISO/IEC Guide 2, according to which standards may be voluntary or mandatory. The use of the term "voluntary" in the Explanatory Note also suggests that the expression "not mandatory" in Annex 1.2 may be understood as meaning in essence "voluntary". By implication therefore, a document that sets out "voluntary" requirements would also not be considered to be "mandatory" within the meaning of Annex 1.1.²⁹⁹ Our interpretation of the term "mandatory" in Annex 1.1 should be duly informed by this context.

para. 176.

The Panel notes that according to the definitions provided by Annex 1 of the TBT Agreement, "terminology" "symbols", "packaging", "product characteristics", "related processes and production methods", "terminology", "symbols", "packaging", "marking" and "labelling requirements" may be the subject of either technical regulations or standards.

²⁹⁷ Appellate Body Report, EC – Asbestos, para. 69; see also Appellate Body Report, EC – Sardines,

²⁹⁹ We note that the reference to "mandatory" compliance is not repeated in the second sentence of Annex 1.1. Nonetheless, it is undisputed that this requirement applies to all instruments that would constitute "technical regulations". As we understand it, the second sentence merely clarifies that the specific elements

- 7.109 Against this context, the Panel is mindful of the fact that the term "mandatory" expresses the single characteristic that defines the key conceptual distinction between two of the three types of measures covered under the TBT Agreement (technical regulations and standards) and therefore plays a central role in preserving the balance between the different sub-regimes coexisting within that Agreement.
- 7.110 We also bear in mind the object and purpose of the TBT Agreement. This Agreement establishes, in the words of the Appellate Body, "a specialized legal regime" that aims at furthering the objectives of the GATT 1994. 300 This specialized set of provisions is intended to discipline very specific categories of measures³⁰¹, namely, technical regulations, standards and conformity assessment procedures. The TBT Agreement establishes clear thresholds that define the type of measures that fall within the scope of each of these three categories of measures, and attach different types and levels of obligations to each one. These thresholds are embodied by the definitions of "technical regulation", "standard" and "conformity assessment procedure" contained in Annex 1 of the TBT Agreement.
- 7.111 In sum, we consider that compliance with product characteristics or their related production methods or processes is "mandatory" within the meaning of Annex 1.1, if the document in which they are contained has the effect of regulating in a legally binding or compulsory fashion the characteristics at issue, and if it thus prescribes or imposes in a binding or compulsory fashion that certain product must or must not possess certain characteristics, terminology, symbols, packaging, marking or labels or that it must or must not be produced by using certain processes and production methods. By contrast, compliance with the characteristics or other features laid out in the document would not be "mandatory" if compliance with them was discretionary or "voluntary".
- 7.112 In light of these general observations, we now consider whether the US dolphin-safe labelling provisions impose or prescribe product characteristics, or more specifically, in this case, labelling requirements, in a binding or compulsory fashion, such that compliance with these should be considered to be "mandatory" within the meaning of Annex 1.1. In approaching this determination, we also take note of the Appellate Body's observation, in EC-Asbestos and EC-Sardines, that when examining whether a measure is a technical regulation, "the proper legal character of the measure at issue cannot be determined unless the measure is examined as a whole". 302

Whether the US dolphin-safe provisions are "mandatory" within the meaning of Annex 1.1

7.113 Mexico submits two alternative arguments on this point. First, Mexico submits that the US dolphin-safe provisions de jure establish mandatory labelling requirements. Alternatively, Mexico asks the Panel to decide that, even if the US dolphin-safe provisions do not on their face require certain product characteristics, they do so in a de facto manner. We will therefore consider these two

listed in it may form the subject-matter of a technical regulation, but is not intended to imply that a document laying down such elements, with which compliance would not be mandatory, would be a technical regulation. Indeed, such a reading would render meaningless the corresponding text in Annex 1.2. To the extent that there is any ambiguity in this respect from the structure of Annex 1.1, the clear terms of the Explanatory Note resolve it by clarifying that a technical regulation is defined as a mandatory document. This explanation applies to the entirety of the definition of a technical regulation and provides direct guidance as well as context to the interpretation of the terms of Annex 1.1.

 $^{^{300}}$ See second paragraph of the Preamble of the TBT Agreement. 301 We find support for our understanding in the Appellate Body's decision in EC – Asbestos, it concluded that "although the TBT Agreement is intended to 'further the objectives of GATT 1994', it does so through a specialized legal regime that applies solely to a limited class of measures", para. 80. Appellate Body Reports in EC - Asbestos, para. 64, and EC - Sardines, paras. 192-193.

aspects in turn, without prejudice, at this stage of our analysis, to the question of whether *de facto* "mandatory" compliance is covered by Annex 1.1.

- 7.114 At the outset, we find it useful to clarify the scope of our enquiry into whether the measures establish *de jure* mandatory labelling requirements. In our view, compliance would be *de jure* "mandatory" in a situation in which the mandatory character of the provisions would be discernable from an examination of the terms or structure of the measures themselves. In order to determine whether the United States' measures *de jure* establish labelling requirements for tuna products, we therefore need to determine whether the alleged mandatory character of the dolphin-safe labelling requirements can be derived from the terms of the US dolphin-safe provisions themselves.
- 7.115 In approaching this question, we first recall our conclusion under the previous subsection that, as acknowledged by both parties, the US dolphin-safe labelling provisions lay down *labelling requirements* within the meaning of Annex 1 of the TBT Agreement. The particular nature and functioning of this type of instrument should, in our view, inform our understanding of what "mandatory" compliance within the meaning of Annex 1.1 implies in relation to such requirements.
- 7.116 By hypothesis, both technical regulations and standards laying down "labelling *requirements*" would establish the conditions that a product must comply with before being able to carry a certain

"Article 3.1(a) prohibits *any* subsidy that is contingent upon export performance, whether that subsidy is contingent 'in law or in fact'. The Uruguay Round negotiators have, through the prohibition against export subsidies that are contingent *in fact* upon export performance, sought to prevent circumvention of the prohibition against subsidies contingent *in law* upon export performance. In our view, the legal standard expressed by the word 'contingent' is the same for both *de jure* or *de facto* contingency. There is a difference, however, in what evidence may be employed to prove that a subsidy is export contingent. *De jure* export contingency is demonstrated on the basis of the words of the relevant legislation, regulation or other legal instrument" (italics in the original, underlined added). Appellate Body Report, *Canada – Aircraft*, para. 167.

Similarly, in *Canada – Autos*, the Appellate Body considered that:

"We start with what we have held previously. In our view, a subsidy is contingent 'in law' upon export performance when the existence of that condition can be demonstrated on the basis of the very words of the relevant legislation, regulation or other legal instrument constituting the measure. The simplest, and hence, perhaps, the uncommon, case is one in which the condition of exportation is set out expressly, in so many words, on the face of the law, regulation or other legal instrument. We believe, however, that a subsidy is also properly held to be de jure export contingent where the condition to export is clearly, though implicitly, in the instrument comprising the measure. Thus, for a subsidy to be de jure export contingent, the underlying legal instrument does not always have to provide expressis verbis that the subsidy is available only upon fulfillment of the condition of export performance. Such conditionality can also be derived by necessary implication from the words actually used in the measure." Appellate Body Report, *Canada – Autos*, para. 100.

We note that these two determinations by the Appellate Body were made in relation to provisions contained in a different agreement, i.e. the Agreement on Subsidies and Countervailing Measures. However, we consider that they nonetheless provide useful guidance, because it seems to us that the Appellate Body's reasoning is not determined by the text of the particular provisions involved, but rather by the inherent differences between a *de jure* and a *de facto* analysis.

 $^{^{303}}$ We are guided in this determination by certain rulings concerning the *de facto* vs. *de jure* distinction in other contexts. We thus find the following passage of the Appellate Body's decision in *Canada – Aircraft* very illustrative:

label. However, as also recognized by both parties, this characteristic, which is common to mandatory and voluntary labelling requirements, does not in itself make compliance with a labelling requirement "mandatory" or "not mandatory". The definitions described above make clear a basic distinction between a "requirement", which refers to the conditions or criteria to be fulfilled in order to comply with a document, and the notion of "mandatory" requirement as a condition made *compulsory by law*.

- 7.117 Thus, a conclusion that compliance with certain labelling requirements is mandatory within the meaning of Annex 1.1 of the TBT Agreement must be based on considerations other than, or beyond, the mere fact that such document establishes criteria for the use of a certain label. As determined in paragraph 7.111 above, compliance with product characteristics or their related production methods or processes is "mandatory" within the meaning of Annex 1.1, if the document in which they are contained has the effect of regulating in a legally binding or compulsory fashion the characteristics at issue, and if it thus *prescribes* or *imposes* in a *binding* or *compulsory* fashion that certain product *must* or *must not* possess certain characteristics, terminology, symbols, packaging, marking or labels or that it *must* or *must not* be produced by using certain processes and production methods. In the context of a labelling requirement, therefore, the question we must consider is not only whether the document lays down certain conditions for the use of a label, or prescribes a certain content for a given label, but whether the document at issue regulates in a binding fashion these conditions or content.
- 7.118 In the present dispute, Mexico does not allege that the US dolphin-safe provisions *positively* require, *de jure*, the use of the dolphin-safe label. Indeed, it is undisputed that the measures at issue do not impose a positive requirement to label tuna products for sale on the US market as dolphin-safe. Neither the statutory and regulatory provisions nor the court decision challenged by Mexico³⁰⁵ contain language that imposes *the use of* the dolphin-safe label for tuna products as a condition for these products to be marketed in the United States.
- 7.119 Mexico argues, however, that these measures *negatively* require that "tuna products offered for sale in the United States must not possess certain characteristics" unless certain conditions are met. Specifically, Mexico submits that the measures at issue involve a prohibition on the use of a "dolphin-safe" label on Mexican tuna products marketed in the United States. In Mexico's view, this prohibition may be expressed as a requirement that "[t]una products offered for sale in the United States must not possess certain characteristics (*i.e.*, distinguishing marks the dolphin-safe label or any other analogous label or mark) unless the prescribed requirements are met".
- 7.120 Mexico further submits that "there are no alternative means to certify that a tuna product is dolphin-safe". According to Mexico "[t]he fact that the U.S. measures prohibit the utilization of any other standards to certify that a tuna product is dolphin-safe gives these measures a *de jure* mandatory distinction". 309
- 7.121 We find it necessary to consider in the first instance the terms of the measures, in order to determine whether they support the view that they prescribe the labelling requirements at issue "in a compulsory manner".

³⁰⁴ Mexico's response to Panel question No. 48, para. 131; United States' second written submission, para. 91.

³⁰⁵ Mexico's first written submission, paras. 121-36.

³⁰⁶ Mexico's first written submission, para. 202.

³⁰⁷ Mexico's first oral statement, para. 23; Mexico's first written submission, para. 200.

³⁰⁸ Mexico's first written submission, para. 202.

³⁰⁹ Mexico's response to Panel question No. 52, paras. 141-142; Mexico's second written submission, para. 196.

7.122 The introductory paragraph of subsection (d)(1) of the Dolphin Protection Consumer Information Act, Title 16, Section 1385, establishes the following:

"(d) LABELLING STANDARD

- (1) It is a violation of section 5 of the Federal Trade Commission Act (15 U.S.C. 45) for any producer, importer, exporter, distributor, or seller of any tuna product that is exported from or offered for sale in the United States to include on the label of that product the term 'dolphin-safe' or any other term or symbol that falsely claims or suggests that the tuna contained in the product were harvested using a method of fishing that is not harmful to dolphins if the product contains tuna harvested ..." (emphasis added)
- 7.123 The fishing techniques in connection with specific geographical areas that do not qualify for the use of a dolphin-safe designation on the product label are then described in various subparagraphs.
- 7.124 We understand this provision as establishing requirements in relation to the terms that may be used "on the label of [any tuna product that is exported from or offered for sale in the United States]". Specifically, it sets out the conditions for the use of a dolphin-safe designation on such labels: producers may only identify tuna as "dolphin-safe", or more generally represent that their tuna has been caught in conditions that are not harmful to dolphins, if they comply with the conditions set out in subsection (d). In other words, in order for tuna products to be "exported from or offered for sale in the United States" with a dolphin-safe designation, they need to comply with the requirements set out in subsection (d) of the DPCIA.
- 7.125 In addition, Section 1385 (d)(3) of the DPCIA provides as follows:
 - "(A) The Secretary of Commerce shall develop an official mark that may be used to label tuna products as dolphin-safe in accordance with this Act.
 - (B) A tuna product that bears the dolphin-safe mark developed under subparagraph (A) shall not bear any other label or mark that refers to dolphins, porpoises, or marine mammals.
 - (C) It is a violation of section 45 of title 15 to label a tuna product with any label or mark that refers to dolphins, porpoises, or marine mammals other than the mark developed under subparagraph (A) unless ..." (emphasis added).
- 7.126 This provision therefore further defines the conditions under which reference may be made to certain terms, i.e. "dolphins, porpoises, or marine mammals", on a label for tuna products.
- 7.127 We also note that violations of these provisions are subject to specific enforcement measures. Section 1385 (e) of the DPCIA, entitled "enforcement", reads: "Any person who knowingly and wilfully makes a statement or endorsement described in subsection (d)(2)(B) that is false is liable for a civil penalty of not to exceed \$ 100, 000 assessed in an action brought in any appropriate district court of the United States on behalf of the Secretary."
- 7.128 Together, the provisions at issue therefore regulate the use of the term "dolphin-safe" and of other related terms on labels for tuna products offered for sale on the US market. They do so in a binding and exclusive manner. The provisions at issue imply in particular that there is no possibility for competing or alternative definitions of what is "dolphin-safe" on the US market, or for any claims to be made relating to "dolphins, porpoises or marine animals", except in compliance with the criteria set out in the measures.

- 7.129 We recall in this respect the Appellate Body's determination that a document laying down product characteristics is "mandatory" when it "regulates" these characteristics "in a binding and compulsory fashion", and "have the effect of prescribing or imposing" the product characteristics at issue, either in a positive or in a negative fashion. We bear in mind also the Appellate Body's determination that "labelling requirements" constitute "product characteristics" in this context.
- 7.130 In the present case, the question to be addressed is therefore whether the US measures "regulate" the labelling requirements at issue "in a binding or compulsory fashion", and whether they "have the effect of prescribing or imposing" the labelling requirements at issue, either in a positive or in a negative fashion.
- 7.131 In our view, to the extent that they prescribe, in a binding and legally enforceable instrument, the manner in which a dolphin-safe label can be obtained in the United States, and disallow any other use of a dolphin-safe designation, the US tuna labelling measures "regulate" dolphin-safe labelling requirements "in a binding or compulsory fashion". It is not compulsory to *meet* these requirements and to bear the label, in order to sell tuna on the US market. The US measures therefore do not impose any *requirement to label*, in a "positive" manner. However, they do prescribe and impose *the conditions under which* a product may be labelled dolphin-safe. In particular, the measures prescribe "in a negative form", to use the Appellate Body's terms, that no tuna product may be labelled dolphin-safe or otherwise refer to dolphins, porpoises or marine mammals if it does not meet the conditions set out in the measures, and thus *impose* a prohibition on the offering for sale in the United States of tuna products bearing a label referring to dolphins and not meeting the requirements that they set out.
- 7.132 We see a difference, in this respect, between the fact that compliance with the underlying *standard* that provides access to the label (i.e. the use of certain fishing methods to harvest tuna) is not obligatory, and the fact that the measures prescribe in a binding manner the conditions for the use of certain terms on labels for tuna products, on the basis of compliance or absence of compliance with that underlying standard.
- 7.133 In order to be marketed as dolphin-safe tuna, tuna products must be prepared exclusively from fish caught in the specific conditions laid down in the US dolphin-safe labelling provisions. Tuna products containing tuna caught in a manner not complying with the specific conditions identified in the regulation are, by virtue of the measures at issue, prohibited from being identified and marketed under an appellation including the term "dolphin-safe" or other related designations.
- 7.134 We note that this situation bears a close resemblance to that addressed by the Appellate Body in *EC Sardines*. The measure at issue in that case was an EC regulation on "preserved sardines", which set out a number of prescriptions for the sale of such sardines, including the requirement that they contain only one named species of sardines (*sardina pilchardus*), to the exclusion of others (including *sardinops sagax*). In that case, as explained by the European Communities and expressly noted by the Appellate Body, "the only legal consequence of the [EC] Regulation for preserved *sardinops sagax* is that they may not be labeled 'preserved sardines'". In other words, it was legally possible to sell *sardinops sagax* on the EC market, as long as it was not sold under the appellation "preserved sardines".
- 7.135 In that case, the Appellate Body observed that "preserved products made, for example, of *sardinops sagax* are, by virtue of the EC Regulation, *prohibited* from being identified and marketed under an appellation including the term "sardines" (emphasis in original).³¹¹ The Appellate Body

³¹⁰ Appellate Body Report, EC – Sardines, para. 185.

Appellate Body Report, EC – Sardines, para. 184.

further observed that "Article 2 of the EC Regulation provides that, to be marketed as "preserved sardines", products must be prepared exclusively from fish of the species *Sardina pilchardus*", and determined that this requirement was a "product characteristic" intrinsic to preserved sardines that was laid down in the EC Regulation. The Appellate Body then turned to the fact that compliance with the Regulation was "mandatory" to conclude, overall, that it was a "technical regulation".

7.136 Similarly, the panel in EC – Trademarks and Geographical Indications (Australia), in a finding that was not appealed, determined that:

"The second indent of Article 12(2) makes a distinction between those products using a GI identical to a Community protected name that satisfy the labelling requirement, and those which do not. The negative implication that follows from this requirement is that products with a GI identical to a Community protected name that do not satisfy this labelling requirement must *not* use the indications PDO, PGI or equivalent national indications and, to the extent that they fall within the protection granted to a prior identical Community protected name, must *not* be marketed in the European Communities using that GI. Therefore, the second indent of Article 12(2) is an obligatory or mandatory requirement".³¹²

- 7.137 These rulings clearly suggest that the mere fact that it is legally permissible to place the product on the market without using the designation that is regulated by the measures at issue does not compel the conclusion that these measures are not "mandatory" within the meaning of Annex 1.1, where the measures effectively regulate in a binding manner the use of the appellation. Similarly, in the present case, the legal consequence of the measures for tuna products not meeting the requirements of the regulation is that they may not be labelled dolphin-safe. This implies that such products are *prohibited* from being identified and marketed under this appellation.
- 7.138 We also note in this respect that, although the EC "preserved sardines" Regulation set out a "naming" requirement rather than "labelling requirements", the Appellate Body commented that it effectively saw no difference between the two notions, both being "essentially "means of identification" of a product" coming within the scope of the definition of "technical regulation". This suggests that this is not a distinguishing factor in the analysis, that would modify its conclusions.
- 7.139 We also note the emphasis placed by the Appellate Body in EC Sardines, on the role of such measures in providing consumer information. Similarly in the present case, the measures serve the function of informing consumers about the manner in which the tuna was caught. They regulate in a binding fashion the information that may be conveyed in this respect, as well as the manner in which it may be conveyed.
- 7.140 Our consideration of these prior rulings thus confirms further our view that compliance with the labelling requirements laid out in the US dolphin-safe labelling provisions is "mandatory" within the meaning of Annex 1.1.
- 7.141 In making this determination, we are mindful, as discussed in paragraph 7.109 above, that a proper interpretation of the expression "mandatory" compliance in Annex 1.1 of the TBT Agreement must preserve the conceptual and functional distinction between technical regulations and standards.

³¹² Panel Report, EC – Trademarks and Geographical Indications (Australia), para. 7.456.

³¹³ The Appellate Body noted that the European Communities itself underscores the important role that a "name" plays as a "means of identification" when it argued before the Panel that one of the objectives pursued by the European Communities through the EC Regulation is to provide precise information to avoid misleading the consumer. Appellate Body Report, *EC–Sardines*, para. 191.

In this respect, we find it useful to highlight that we are not suggesting that *any* situation in which access to a label reflecting compliance with a particular standard is reserved for products that comply with the specific requirements of that standard would amount to a situation in which a mandatory technical regulation exists. In our view, the measures at issue in the present case go significantly beyond that.

- 7.142 First, the measures at issue are legally enforceable and binding under US law (they are issued by the government and include legal sanctions). This is an important component of the "mandatory" character of the measures. This alone, however, might not necessarily distinguish them from any standard that is protected against abusive or misleading use under general law, such as trademark protection or laws against deceptive practices.
- 7.143 In addition, however, the measures at issue prescribe certain requirements that must be complied with in order to make *any* claim relating to the manner in which the tuna contained in tuna product was caught, in relation to dolphins. The measures at issue regulate not only the use of the particular label at issue, but more broadly the use of a range of terms for the offering for sale of tuna products, beyond even the specific "dolphin-safe" appellation. The measures at issue thus prohibit the use of other terms such as "porpoise" or "marine mammal" or any statement relating to dolphins, porpoises or marine animals, whether misleading or otherwise, if the conditions set out in the regulation are not met.
- 7.144 Furthermore, the measures embody compliance with a specific standard as the exclusive means of asserting a "dolphin-safe" status for tuna products. The measures leave no discretion to resort to any other standard to inform consumers about the "dolphin-safety" of tuna than to meet the specific requirements of the measure. Effectively, the "dolphin-safe" standard reflected in the measures at issue is, by virtue of these measures, the *only* standard available to address the issue. Through access to the label, the measures thus effectively regulate the "dolphin-safe" status of tuna products in a binding and exclusive manner and prescribe, both in a positive and in a negative manner, the requirements for "dolphin-safe" claims to be made. This distinguishes this situation from one in which, for example, various competing standards may co-exist in relation to the same issue, with different but related claims, each of which may be protected in its own right.³¹⁴
- 7.145 In light of all the above, we find that the measures at issue establish labelling requirements, compliance with which is mandatory. In light of our conclusion that the measures at issue establish *de jure* mandatory labelling requirements, we do not find it necessary to consider further Mexico's argument that they are also *de facto* mandatory.

Separate opinion

- 7.146 One of the panelists is unable to agree with the reasoning and conclusions contained in paragraphs 7.128 to 7.145 above. This section reflects the views of that panelist.
- 7.147 While I agree with the three-tier test for determining whether a measure is a "technical regulation" under the TBT Agreement as established in principle by the Appellate Body in EC Asbestos and followed in EC Sardines, I do not agree with the conclusions of the majority that the measures at issue meet the requirements of this test. According to that test, in order to be a technical

³¹⁴ We note in this respect the following definitions of the ISO/IEC Guide 2: "exclusive reference (to standards)": "reference to standards that states that *the only way* to meet the relevant requirements of a technical regulation is to comply with the standards referred to" (emphasis added).

[&]quot;mandatory standard": "standard the application of which is made compulsory by virtue of a general law *or exclusive reference* in a regulation" (emphasis added).

regulation, (i) a measure has to apply to an identifiable product or group of products, (ii) it has to lay down one or more characteristics of the product or related processes and production methods, and (iii) compliance with the prescribed product characteristics or process and production method is mandatory. Thereby, the second sentence of Annex 1.1 of the TBT Agreement should not be read as independent of the first sentence, it should rather be read as ensuring a broad understanding of the term "technical regulation". The second sentence makes clear that according to the second criteria, technical regulations are not limited to a narrow understanding of "product characteristics" but that they also include prescriptions with regard to "terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method". Thus, a labelling requirement that applies to a process or production method that is not directly related to the product can also be a technical regulation as long as it meets the other two criteria of the test. I agree with the majority that the measures at issue meet the first two requirements. However, I do not agree that the measures at issue require mandatory compliance with the prescribed product characteristics or process and production method.

- 7.148 A treaty has to be interpreted in good faith in accordance with the ordinary meaning to be given to its terms in their context and in the light of its object and purpose (Article 31.1 of the *Vienna Convention*). Moreover, a treaty should not be interpreted in a manner that leads to an unreasonable result (Article 32(b) of the *Vienna Convention*). It is therefore important to construe Annex 1.1 of the TBT Agreement in accordance with the ordinary meaning to be given to its terms in their context and in the light of the objective and purpose of the TBT Agreement and in a manner that does not make void other provisions of the Agreement.
- 7.149 Annex 1.1 and 1.2 of the TBT Agreement use similar language, the second sentence is even identical. Thus, both technical regulations and standards can include terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method. The main difference is that while compliance with technical regulations is mandatory, compliance with standards is not mandatory. In interpreting the phrase "with which compliance is mandatory", it is important to reflect the context of that formulation, the objective and purpose of the TBT Agreement, and the existence of Annex 1.2. An interpretation that conflates the requirement that compliance with a technical regulation has to be mandatory with the term "labelling requirements" of the second sentence or an interpretation that leaves no space to Annex 1.2 would not be reasonable.
- 7.150 According to the ordinary meaning of the term, labelling requirements are requirements that must be fulfilled in order to be allowed to use a certain label. Any labelling scheme foresees such requirements in fact, if such requirements would not exist and if a certain label could be used independent of whether specific requirements are fulfilled, the label would become meaningless. Labelling schemes can be compulsory when the use of a certain label is compulsory to access the market, or they can be voluntary when products can be marketed with or without the label. But "labelling requirements", i.e. the requirements that are formulated to allow the use of a label, must be met also within a voluntary labelling scheme. However, in difference to a compulsory labelling scheme, within a voluntary scheme products have not to be labelled and have not to fulfil these labelling requirements in order to be marketed, they can also be put on the market without that label and without fulfilling these labelling requirements. In a voluntary labelling scheme, labelling requirements are thus not mandatory for marketing products.
- 7.151 Both Annex 1.1 and Annex 1.2 refer to labelling requirements. Labelling requirements can thus be technical regulations or standards. The criteria whether labelling requirements are a technical regulation or a standard relates to the fact whether compliance is mandatory or not. In order to give any sense to the term "labelling requirement" as used both in Annex 1.1. and 1.2, the requirement that compliance is mandatory cannot relate to the obligation to meet certain requirements to be allowed to use the label, but to the question whether a labelling scheme is compulsory i.e. whether products must use a label in order to be marketed or voluntary i.e. products may be marketed with or

without the label. As indicated by the Appellate Body in *EC – Asbestos*, a technical regulation must "regulate the characteristics of products in a binding or compulsory fashion" and "a 'technical regulation' has the effect of prescribing or imposing one or more 'characteristics' – 'features', 'qualities', 'attributes', or other 'distinguishing mark'". A labelling requirement which is a technical regulation would thus impose to a product the obligation to use the label and to fulfil the related labelling requirements. If however compliance with the labelling requirement and use of the label is not mandatory, the labelling requirement has to be seen as a standard. Arguing that the mere fact that a product is prohibited from using a label if it does not fulfil these standards makes compliance compulsory would leave no space for voluntary labelling schemes as standards.

- 7.152 From the terms of the measures at issue, it appears that they are legally binding, in that the conditions set out in subsection (d) must be met if a "dolphin-safe" label is used, and the use of a label in violation of these conditions would be subject to enforcement measures. The prohibition on the use of alternative labels unless the conditions set out in subsection (d)(3)(C) are met is also legally enforceable. This is a typical situation common to both voluntary and mandatory labelling schemes. However, neither the use of the "dolphin-safe" label nor the use of the specific fishing techniques and locations that condition access to the label is compulsory, either in a positive or negative manner (i.e. their use is neither obligatory nor prohibited).
- 7.153 In other words, the measures do not impose a general requirement to label or not to label tuna products as "dolphin-safe". It remains a voluntary and discretionary decision of operators on the market to fulfil or not fulfil the conditions that give access to the label, and whether to make any claim in relation to the dolphin-safe status of the tuna contained in the product. If an operator wishes to make such claim, however, it must abide with the conditions laid down in the DPCIA and other related measures. In short, the measures at issue set out the requirements for dolphin-safe labelling, but they impose no obligation to label (or not to label) tuna as "dolphin-safe". For this reason, compliance with the labelling requirements at issue is not "mandatory" within the meaning of Annex 1.1.
- 7.154 The notion of "mandatory compliance" within the meaning of Annex 1.1 implies the obligatory character of *the characteristics* prescribed in the measures, and not simply the legal enforceability of the measures. A distinction should be made, in this context, between on the one hand, the *legally binding* character of a document laying down labelling requirements and, on the other hand, *mandatory compliance* with the conditions advertised through the use of the label. While may well be necessary for the measures at issue to be legally enforceable, this is not necessarily sufficient, in order to make them "mandatory" within the meaning of Annex 1.1. Rather, the notion of "mandatory compliance" relates more fundamentally to the fact that the measure at issue prescribes or *imposes* compliance with specific requirements to allow a product to be marketed, without allowing discretion to depart from them.
- 7.155 This distinction is consistent with the essence of the core distinction made in the Agreement between "mandatory" technical regulations and "voluntary" standards. The obligatory character of compliance with the characteristics laid out in a technical regulation contrasts with the discretion that exists, under a "voluntary" (or "not mandatory") standard, for operators to comply or not comply with the standard. The requirements to be allowed to use a label have to be clearly distinguished from the obligation to use a label.
- 7.156 As described above, under the measures at issue, it is the *voluntary* decision of those producers to label their products in a certain way what triggers the obligation to comply with such

 $^{^{315}}$ Appellate Body Report, EC – Asbestos, paras. 68-69; see also Appellate Body Report, EC – Sardines, para. 176.

standards. The fact that operators may be legally accountable for any misleading or false declarations in the event that they choose to advertise compliance with a standard without in fact meeting its requirements does not modify the essentially voluntary nature of such standard.

- 7.157 I note in this respect that governments may want to ensure that whenever a certain product makes a claim in a label, such claim is reflective of the true characteristics of that product or the actual related processes and methods involved in its production. For this reason, governments may enact laws and regulations requiring that *if* a producer claims that its products are in compliance with certain voluntary specifications, i.e. standards, that producer is under a duty to ensure that its products actually meet those standards. For the same reasons (i.e. consumer protection and fair competition), those laws and regulations may require that producers do not use labels or symbols that may induce the public to erroneously believe that the products carrying such labels or symbols comply with the conditions for the use of *other* labels. The existence of such a legally binding norm that compels producers to fulfil the promises they make in relation to compliance with voluntary standards should not transforms such standards into mandatory technical regulations. If it would, no space would be left for labelling requirements as standards as foreseen in Annex 1.2.
- 7.158 For these reasons, the fact that compliance with the terms of the labelling requirements is legally enforceable alone is not sufficient to render compliance with these requirements "mandatory" within the meaning of Annex 1.
- 7.159 In the present case, the *chapeau* of subsection (d)(1) of the DPCIA, establish that it is a violation of the Federal Trade Commission Act to export from or to offer for sale in the United States tuna products that "*include on the label* of that product the term '*dolphin-safe*' or any other term or symbol that *falsely claims or suggests* that the tuna contained in the product were harvested using a method of fishing that is not harmful to dolphins", unless the conditions set out by that section itself are met (emphasis added). In addition, subsection (d)(3) of this statute provides that "[i]t is a violation of section 45 of title 15 to label a tuna product with any label or mark that refers to dolphins, porpoises, or marine mammals other than the mark developed under subparagraph (A)"³¹⁶ unless certain conditions are met.³¹⁷.
- 7.160 The *chapeau* of subsection (d)(1) of the DPCIA is intended to ensure that *when a claim is made* on a label that certain conditions have been met, the producers of the goods carrying that label are bound to meet those conditions. In turn, subsection (d)(3), in combination with subsection (d)(1) of the DPCIA, may be understood as a provision aimed at preventing deceptive practices in relation to the use of the "dolphin-safe" label.
- 7.161 This conclusion is not modified by the limitations placed by the measures on the use of alternative "dolphin-safe" marks or labels and on the use of certain terms. These aspects of the measures should be understood in connection with the overall architecture of the measure, which aims to ensure that any claims made in relation to the impact on dolphins of the catching of the tuna contained in tuna products are consistent with the conditions set out in the measures, and do not mislead consumers. Most importantly, these provisions do not render the "labelling requirement" mandatory in the sense that the label must be used and that the requirements to use the label must be

³¹⁶ Subparagraph (A) establishes: "The Secretary of Commerce shall develop an official mark that may be used to label tuna products as dolphin-safe in accordance with this Act". Therefore, we believe that this subsection may be read as follows: "[i]t is a violation of section 45 of title 15 to label a tuna product with any label or mark that refers to dolphins, porpoises, or marine mammals other than [the 'dolphin-safe' label]".

317 See paras. 2.27 to 2.30 above.

met in order to put tuna on the US market. In fact, as evidenced by the United States, tuna without the label is marketed within the United States.³¹⁸

7.162 Mexico relies, in support of its claim that the measures at issue constitute a "mandatory" technical regulation, on two prior decisions of the Appellate Body, namely EC – Asbestos and EC – Sardines. I wish to point out important distinctions between the situations addressed in those disputes and the present case.

7.163 In EC – Asbestos, the measure at issue did not contain any voluntary aspect. Asbestos and asbestos-containing products were simply banned by the measure at issue. According to the Appellate Body, this ban negatively imposed on all products the requirement not to contain asbestos. Therefore, the producers and importers of asbestos and asbestos-containing products were not given an alternative in order to market their products in the territory of the European Union. The requirement not to include asbestos in their products was imposed directly by the regulations at issue, and not by the producers' or importers' own decision to become bound by such prohibition. Therefore, although the decision of the Appellate Body in EC – Asbestos provides useful guidance in relation to several points in discussion in the present dispute, it does not offer conclusive assistance in relation to whether legally enforceable provisions setting out the conditions of use of a label to advertise compliance with a voluntary standard, or prohibiting the use of deceptive labels, turn such standard into a technical regulation.

7.164 In EC – Sardines, the mandatory nature of the measures challenged in that case was not in question. 319 In that case, the Appellate Body was therefore not called upon to make a determination concerning the interaction between mandatory and voluntary requirements. The measures in that case required the use of only the species Sardina pilchardus in products marketed as "preserved sardines". 320 The Appellate Body found that "preserved products made, for example, of Sardinops sagax [were], by virtue of the EC Regulation, prohibited from being identified and marketed under an appellation including the term 'sardines'" (emphasis in the original).³²¹ In other words, the exporters of preserved sardines containing any species of sardines other than Sardina pilchardus were not allowed to market their products as "sardines". EC - Sardines did thus not involve a labelling requirement but a naming requirement. And by not allowing to market certain preserved sardines as sardines, these products were prohibited to enter the sardine market at all. However, in difference to EC - Sardines, Mexico is not prohibited from selling its tuna as tuna to the United States. It is a fundamental difference between the facts of that case and those of the present dispute that in this case the measures lay down labelling requirements, which allow the operator to make claims reflecting compliance with a particular standard. The US dolphin-safe provisions are intrinsically linked to voluntary labelling scheme, which, in turn, is intended to reflect compliance with a particular standard (a "dolphin-safe" standard). This is a factual circumstance with which neither the panel nor the Appellate Body in EC – Sardines was confronted. They could, therefore, conclude that the measures in question laid down product characteristics in a negative form without affecting the conceptual differences between technical regulations and standards. For these reasons, the Appellate Body's decision in EC - Sardines does not offer conclusive guidance on whether legally enforceable provisions imposing compliance with particular requirements in order to claim compliance with a standard though the use of a label, or prohibiting the use of deceptive labels relating to the same matter, have the effect of turning a voluntary standard into a technical regulation.

³¹⁸ See para. 7.345 below.

³¹⁹ Appellate Body Report, EC – Sardines, para. 194.

Appellate Body Report, EC – Sardines, para. 179, quoting the Panel Report in that same dispute, para. 7.45.

321 Appellate Body Report, *EC – Sardines*, para. 184.

- 7.165 As observed earlier, the US measures need to be considered as a whole. This requirement, along with the duty to interpret the terms of the definition of the TBT Agreement in a way that does not deprive the definition of a "standard" of its meaning, compels to conclude that the legally binding components of the US dolphin-safe provisions do not transform voluntary "dolphin-safe" labelling requirements into mandatory technical regulations. In light of all the above, I conclude that the US dolphin-labelling provisions do not establish de jure mandatory labelling requirements.
- 7.166 Mexico alternatively argues that the labelling scheme established by the US dolphin-safe provisions is *de facto* mandatory "because the market conditions in the United States are such that it is impossible to effectively market and sell tuna products without a dolphin-safe designation". 322 In this respect, as described above. Mexico has observed that the US distribution and retail networks for tuna products are acutely aware of the dolphin safe issue and the fact that they will encounter actions such as boycotts, promoted by NGOs, if they carry tuna that is not labelled as dolphin safe. Mexico argues that large US grocery chains have indicated that they will be unable to carry any Mexican tuna products unless they bear a US government-approved dolphin safe label. Mexico has also noted that the three major tuna processors in the United States refuse to purchase tuna caught in the ETP including Mexican tuna because tuna products containing such tuna cannot be labelled as dolphin safe.323
- 7.167 Mexico observes that the use of the "dolphin-safe" label has "significant commercial value" in the United States' market because "consumers at each consumption stage demand this label in order to sell or buy tuna products". 324 Therefore, Mexico contends, the prohibition on having the dolphin-safe label on Mexican tuna products has the effect of excluding Mexican tuna products from the major distribution channels in the US market, and compliance with the requirements imposed by the US dolphin-safe provisions becomes mandatory for the Mexican tuna industry in order to access the US market.³²⁵
- 7.168 The United States responds that Mexico's contention that the US dolphin-safe provisions have prevented Mexican tuna products from accessing the United States' market is not based on the US dolphin-safe provisions themselves or on any other measure or government action that limits the opportunity for Mexican tuna products to be imported, sold, distributed or otherwise marketed in the United States. According to the United States, Mexico's argument is based on retailer and consumer preferences for tuna products that were not caught in a manner that adversely affects dolphins.³²⁶ Moreover, the United States argues that some distribution channels in the United States do purchase and sell tuna products that are not labelled "dolphin-safe", and that consumer or retailer preferences alone cannot determine whether a labelling requirement is mandatory.³²⁷
- 7.169 In addressing this part of Mexico's claim, it needs to be considered, as a threshold matter, whether the definition of a "technical regulation" in Annex 1.1 of the TBT Agreement encompasses situations in which the document at issue would be de facto rather than de jure mandatory.
- 7.170 Issues relating to the *de facto* or *de jure* coverage of specific obligations have been discussed in the context of various provisions contained in the GATT 1994³²⁸, the GATS³²⁹ and other covered

³²² Mexico's first written submission, para. 203.

³²³ Mexico's first written submission, paras. 111-112.

³²⁴ Mexico's response to Panel question No. 52, para. 148.

³²⁵ Mexico's response to Panel question No. 52, para. 149.

³²⁶ United States' second written submission paras. 110-111; United States' response to Panel question No. 52, para. 127.

327 United States' response to Panel question No. 57, para. 130.

³²⁸ See e.g. GATT panel reports, Belgium – Family Allowances; EEC – Imports of Beef, GATT panel report, Spain - Unroasted Coffee and Japan - SPF Dimension Lumber (in relation to Article I of the

agreements, such as the SCM Agreement. 330 Although these precedents must be considered with caution since they refer to provisions that notably differ in scope and nature from those contained in the TBT Agreement, the reasoning behind those decisions may shed light on the question at issue here.

When the Appellate Body and previous panels have addressed the question of whether a particular provision equally applies to de jure and de facto situations, they appear to have focused both on the text of the provision in question and on the potential consequences of excluding the possibility of applying that norm to *de facto* scenarios. The following passage of the Appellate Body's decision in EC – Bananas III illustrates this approach:

"The question here is the meaning of 'treatment no less favourable' with respect to the MFN obligation in Article II of the GATS. ... The obligation imposed by Article II is unqualified. The ordinary meaning of this provision does not exclude de facto discrimination. Moreover, if Article II was not applicable to *de facto* discrimination, it would not be difficult -- and, indeed, it would be a good deal easier in the case of trade in services, than in the case of trade in goods -- to devise discriminatory measures aimed at circumventing the basic purpose of that Article" (italics in the original).³³¹

- In the present case, the first task is therefore to assess whether the ordinary meaning of the language used in the definition of technical regulation in the TBT Agreement allows or excludes the possibility of de facto mandatory technical regulations. The relevant language is the term "mandatory", or more broadly, the expression "with which compliance is mandatory" in Annex 1.1. This formulation seems not to exclude the possibility that compliance has to be *de facto* mandatory.
- 7.173 Namely, it does not exclude that the mandatory character of the measure may result from a combined effect of various measures or actions attributable to the Member in question, and not necessarily exclusively from the specific instrument in which the "product characteristics or related processes and production methods" are laid down. What is required is that compliance with the characteristics laid down in the document be mandatory. This does not necessarily prejudge whether this mandatory character is contained in that instrument itself. It does not exclude that voluntary and non-mandatory labelling requirements may become mandatory as a result of "some other governmental action"³³² or more generally, some other action attributable to the Member concerned.³³³
- 7.174 The question is therefore whether, despite the absence of a de jure requirement in the measures at issue to use the "dolphin-safe" label in order to market tuna products in the United States, tuna products are nonetheless *compelled* to carry that label as a result of some other action attributable to the United States.

GATT 1994); GATT panel report, Italy - Agricultural Machinery (in relation to Article III.4); GATT panel report, Japan - Semi-Conductors and panel report Argentina - Hides and Leather (in relation to Article XI:1 of the GATT 1994).

³²⁹ See e.g. Appellate Body Report, EC – Bananas III (in relation to Article II of GATS).

³³⁰ See e.g. Appellate Body Report, Canada – Autos (in relation to Article 3.1(b) of the SCM Agreement).

331 Appellate Body Report, *EC – Bananas III*, para. 233; see also, Appellate Body Report, *Canada –*

³³² United States' response to Panel question No. 52, para. 125.

³³³ It would be conceivable, for example, that if a Member in fact grants certification in relation to one particular standard as a means of complying with a regulatory requirement, but not in relation to other standards, its actions may turn that formally voluntary standard, into a *de facto* technical regulation.

- 7.175 In summary, the "dolphin-safe" label may be considered *de facto* mandatory in order to market tuna products in the United States, if doing otherwise becomes impossible, not because it would contradict a mandatory provision in the measures, but because it would be prevented by a factual situation that is sufficiently connected to the actions of the United States. Thus, this analysis is two-fold. First, the impossibility of marketing tuna products in the United States without the "dolphin-safe" label must be established. Second, such impossibility must arise from facts sufficiently connected to the US dolphin-safe provisions or to another governmental action of the United States.
- 7.176 First, it has to be noted that tuna products are sold in the United States without the "dolphin-safe" label. Not only hast the US submitted evidence in this regard, Mexico has also not challenged this fact. However, Mexico maintains that the dolphin-safe labelling scheme "is *de facto* mandatory because *the market conditions* in the United States are such that it is impossible to *effectively* market and sell tuna products without a dolphin-safe designation" (emphases added). As non-labelled tuna is sold in the US to a certain limited extent, to "effectively" market has to mean having access to the major distribution channels and not being limited to the limited market segment that exists in the US for non-labelled Tuna.
- 7.177 To the extent that Mexico's argument is based on an impossibility of "effectively" marketing and selling tuna products without a dolphin-safe designation, this would not, in my view, provide a sufficient basis for a determination that compliance with the US dolphin-safe labelling requirements is "mandatory" within the meaning of Annex 1.1. Compliance with a voluntary technical document such as a standard may substantially increase the chances of a product being *effectively* sold in a given market. Conversely, failure to comply with such standard may have negative consequences for the competitiveness of a product in that market. However, this fact alone would not alter the voluntary or "not mandatory" nature of that standard, within the meaning of the *TBT Agreement*.
- 7.178 In explaining the adverse effects of the measures, Mexico argued that they have direct effects on tuna products, because major retailers in the United States refuse to buy tuna products that cannot be labelled dolphin safe, and indirect effects on tuna caught by the Mexican fleet, because major producers of tuna in the United States also refuse to purchase Mexican or other tuna caught in the ETP because tuna product containing such tuna could not be included in tuna products labelled dolphin safe.³³⁵ I agree with the United States, however, that these are decisions made by private actors that do not necessarily involve the participation of the State. Such private actions alone should not be able to turn an otherwise voluntary norm into a technical regulation.
- 7.179 Therefore, Mexico would need to demonstrate that the decision made by these companies not to buy tuna or tuna products that cannot be labelled dolphin-safe is the result of the application of the US dolphin-safe provisions or of some other action by the United States. However, Mexico has failed to substantiate such contention.
- 7.180 On the contrary, the United Sates has shown that the three major processors of tuna products in the United States have adopted as their own commercial policies not to use tuna that has been obtained by setting on dolphins". In particular, the United States has presented as evidence the dolphin-safe policy of the major US tuna companies as it is published on their website. Bumble Bee's policy is instructive in this respect:

³³⁴ Mexico's first written submission, para. 202.

³³⁵ Mexico's first written submission, paras. 111 and 112.

³³⁶ Exhibit US-32; Exhibit US-36; Exhibit US-37.

"What is Bumble Bee's Dolphin safe policy? Bumble Bee Foods, LLC remains fully committed to the 100% dolphin-safe policy we implemented in April 1990. This policy guarantees the following:

 Bumble Bee will not purchase tuna from vessels that net fish associated with dolphins. Our purchasing agreements require certification of dolphin-safe fishing practices from all tuna suppliers ...

The U.S. Government Dolphin-Safe Regulations are in the process of being modified through the International Dolphin Conservation Program. The new regulations will allow for a less stringent U.S. Dolphin-Safe Regulation. Despite the new, less stringent "compliance" criteria, Bumble Bee remains committed to and in compliance with a "no encirclement" policy. The commitment of Bumble Bee to dolphin-safety will remain unchanged regardless of any changes to the dolphin-safe law.

We continue to adhere to our 100% dolphin-safe policy."³³⁷

7.181 This evidence also suggests that these companies have been willing to maintain such policies regardless of the potential legislative changes to the US dolphin-safe provisions, and in particular regardless of the potential change of the US definition of "dolphin-safe", to adopt the definition endorsed by the AIDCP, which Mexico argued the United States should apply. Mexico submits that what prompted those policies was the large number of dolphins being killed annually in the ETP at the time these policies were first announced, and that this figure has substantially decreased.³³⁸ However, even if Mexico's assertion is correct, it would not prove that these private actors' commercial policies are determined by the existence or content of the US dolphin-safe labelling scheme. If Mexico's allegation is correct, it would only prove that these companies define their purchase policies in consideration of their own perception of the risks faced by the dolphin populations in the ETP.

7.182 In addition, it is relevant to note that the dolphin-safe policies of the major companies were all implemented prior to the enactment of the first version of the DPCIA which is here challenged, that is to say, prior to the United States having adopted the "strict" definition of dolphin-safe as absence of setting on dolphins.³³⁹ These policies were implemented in April 1990, seven months before the enactment of the DPCIA. The United States has noted that the amicus curiae brief submitted to the Panel highlights that at the time of the enactment of the US dolphin-safe labelling provisions, there was a strong consumer sentiment that setting on dolphins to catch tuna was unacceptable and that something should be done to ensure that consumers had a choice not to purchase products that contained tuna caught in association with dolphins.³⁴⁰ Indeed, the paragraphs of the *amicus curiae* brief to which the United States refer and the exhibits attached to the brief quoted in such paragraphs support the conclusion that such policies were prompted by the lobbying exerted by environmentalists rather than by the enactment of the DPCIA itself. According to a press article the chairman of Heinz's StarKist Seafood Company, the companies' dolphin-safe policy was attributable to both "external influences -lobbying by environmentalists and a consumer boycott of tuna- and an internal corporate debate. ... The consumer boycott, which included a growing number of schoolchildren, seemed to argue for a dolphin-safe policy."³⁴¹ This evidence also explains the

³³⁸ Mexico's response to Panel question No. 43, para. 105.

³³⁷ Exhibit US-37.

³³⁹ Exhibit US-32; Exhibit US-36; Exhibit US-37, Exhibit US-38.

³⁴⁰ See United States' response to Panel question No. 40(a), para. 98 which cites paragraphs 18-20, 25, 62-64 of the amicus curiae brief which in turn refer to exhibits 1, 2, 3 and 4 attached to the amicus submission.

³⁴¹ See United States' response to Panel question No 40(a), para. 98 which cites paragraphs 18-20, 25 and 62-64 of the *amicus curiae* brief which in turn refer to Exhibit Amicus EX-2.

nationwide boycott of tuna supported by schoolchildren, celebrities and business leaders was due to a "dolphin campaign video". A biologist footage shot during an undercover operation from October 1987 to January 1988 on a Panamanian vessel provided film evidence that some purse-seiners were indiscriminately slaughtering dolphins while harvesting the yellowfin tuna that swims beneath the mammals. The evidence presented to the Panel suggests that the broadcast of this documentary on television triggered consumers' and civil society reactions.

7.183 Mexico argues that the three major tuna processors in the United States have a commercial interest in maintaining the US dolphin-safe provisions as they stand today because they would otherwise face competition from Mexican tuna product brands.³⁴² However, even assuming that such economic interest *does* exist, this would not change the fact that this interest and the business decisions taken on the basis of such interest remain within the sphere of private actions that are not necessarily attributable to a measure by the State. Private companies may have strong economic interests that relate to voluntary standards, they may even make important business decisions on the assumption that such standards will or will not remain applicable to their products. However, the existence of those interests and the decisions based upon them do not change the voluntary character of the standard.

7.184 Mexico also observes that large US retailers have indicated that they will be unable to carry any Mexican tuna products unless the tuna products bear a US government approved dolphin safe label.³⁴³ In support of this contention, Mexico has presented an affidavit from an officer of a US company selling Mexican canned tuna in the US market and supporting material. This affidavit states that for several years the company attempted to sell Mexican tuna products in major US grocery chains, which have refused to buy the products because they were not eligible for the dolphin-safe label. The affidavit provides evidence that some of the major US chains have expressly indicated that if the tuna product qualified to be labelled "dolphin-safe", they would sell it and that the company's inability to place the "dolphin-safe" label on the cans has directly affected its ability to sell the Mexican brand in the United States.³⁴⁴

7.185 However, this evidence does not, in my view, support the conclusion that the decision of retailers not to carry tuna products not eligible for a dolphin-safe label is the result of actions attributable to the United States. Rather, this evidence confirms that it is the decision of retailers on the market not to carry tuna products unless they are eligible for US dolphin safe labelling. I note in this respect that the affidavit refers to instances of removal of the products from the shelves following interventions by an NGO, which suggests that retailers react to the dolphin-safe concern in a manner comparable to that of the tuna processors referred to above, that is, that they make their decisions in consideration of their perception of the acceptability of the products to consumers rather than by reason of the US measures themselves. One of the supporting documents provided by Mexico expressly states that this is not a legal issue but an issue of "consumer acceptance". 345

7.186 Based on the above, I find that Mexico has failed to demonstrate that the US dolphin-safe provisions establish either *de jure* or *de facto* mandatory labelling requirements. Thus, I conclude that the third condition for a document to be considered a technical regulation, i.e. that compliance with the labelling requirements at issue is mandatory, has not been met in the present case. Therefore, the US dolphin-safe labelling provisions should not be seen as constituting technical regulations within

³⁴⁵ See Exhibit MEX-58.

³⁴² Mexico's first written submission, para. 113; *see also* Mexico's response to Panel question No. 43, para. 109.

³⁴³ Mexico's first written submission, para. 111.

³⁴⁴ Exhibit MEX-58.

the meaning of the TBT Agreement. Consequently, the provisions of Article 2 of the TBT Agreement, including its paragraphs 1, 2 and 4 should not be applicable to the measures at issue.

7.187 Finally, I wish to make it clear that, in making this determination, no determination is made as to whether the measures at issue might otherwise fall within the scope of the TBT Agreement. However, since Mexico did not submit any claims based on other provisions of the TBT Agreement, no determinations in this regard can be made.

7.188 Notwithstanding this conclusion, I consider it appropriate to pursue the analysis of the measures at issue as a technical regulation, on the basis of the opinion of the majority of the Panel. Without prejudice to my views with respect to this initial determination, I therefore associate myself with the remainder of these findings.

2. Whether the US dolphin-safe labelling provisions are inconsistent with Article 2.1 of the TBT Agreement

7.189 Article 2.1 of the TBT Agreement provides:

"Members shall ensure that in respect of technical regulations, products imported from the territory of any Member shall be accorded treatment no less favourable than that accorded to like products of national origin and to like products originating in any other country."

7.190 Mexico claims that the US dolphin-safe labelling provisions are inconsistent with Article 2.1 because they afford treatment less favourable to Mexican tuna products than to US tuna products and tuna products originating in other countries. We first set out our overall approach to the analysis of this claim, in order to determine what must be established in order for Mexico to prevail in this claim.

- (a) Overall approach
- (i) Main arguments of the parties

7.191 In its first written submission, with respect to Article 2.1 of the TBT Agreement, Mexico refers the Panel to the arguments it made in the context of its claims under the GATT 1994. According to Mexico, Article 2.1 of the TBT Agreement imposes national treatment and most-favoured-nation treatment obligations on technical regulations. It is, Mexico argues, similar to the national treatment obligation in Article III:4 of the GATT 1994, except that it applies to both national treatment and most-favoured-nation obligations. Mexico also observes that like Articles I:1 and III:4 of the GATT 1994, Article 2.1 prohibits both *de jure* and *de facto* discrimination.³⁴⁶

7.192 Mexico noted the following essential elements of an inconsistency with Article 2.1: (i) that the measure at issue is a "technical regulation"; (ii) that products of national origin and products originating in any other country are "like products" with respect to the imported products within the meaning of that provision; and (iii) that the imported products are accorded "less favourable"

³⁴⁶ (footnote original) In Canada – Autos, the Appellate Body observed that Article III:4 of the GATT 1994 covered both de jure and de facto inconsistency and, on this basis, found that a similar provision that disciplined measures that favoured the use of domestic over imported goods (i.e., Article 3.1(a) of the SCM Agreement) would also apply not only to de jure inconsistency but to de facto inconsistency. The same reasoning applies to Article 2.1 of the TBT Agreement which is even more analogous to Article III:4 than Article 3.1(a) of the SCM Agreement. See Appellate Body Report, Canada – Certain Measures Affecting the Automotive Industry, WT/DS139/AB/R, WT/DS142/AB/R, adopted 19 June 2000, para. 140. (Mexico's first written submission, fn 164).

treatment than that accorded to like products of national origin and to like products originating in any other country.³⁴⁷

7.193 With regard to the first element, Mexico refers to earlier arguments and observes that the US dolphin-safe labelling provisions are a technical regulation. 348

7.194 As far as the second element is concerned, Mexico specifies that although the meaning of the term "like products" in Article 2.1 has not been elaborated upon in WTO jurisprudence, given that the language in Article 2.1 is similar to that in Article III:4 of the GATT 1994, it is reasonable to use the four criteria used to determine likeness under Article III:4 discussed above, reviewed in the proper context and in consideration of the object and purpose, to assist in determining likeness under Article 2.1. In light of its conclusions in the context of Article III:4, Mexico asserts all criteria indicate that tuna and tuna products are like products, irrespective of their originating country.³⁴⁹

7.195 Finally, Mexico also contends that for the reasons set out in its claim under Article III:4, the US dolphin-safe labelling provisions do not accord products imported from the territory of any Member (namely, Mexico) treatment no less favourable than that accorded to like products of national origin³⁵⁰ and for the reasons set out in its claim under GATT Article I:1, the measures do not accord products imported from the territory of any Member (namely, Mexico) treatment no less favourable than that accorded to like products originating in any other country³⁵¹ and therefore concludes the US dolphin-safe labelling provisions are inconsistent with Article 2.1 of the TBT Agreement.³⁵²

7.196 The United States presents its defence in line with Mexico's order of analysis. In addition, the United States notes that Mexico relies solely on the arguments it makes regarding the consistency of the measures with Articles I:1 and III:4 of the GATT 1994 for its arguments under Article 2.1 of the TBT Agreement. Observing that it had already articulated why Mexico's arguments under Articles I:1 and III:4 of the GATT 1994 the United States contends that Mexico's arguments under Article 2.1 of the TBT Agreement also fail for the same reasons.

7.197 In light of the parties' references to their analyses under Article I:1 and Article III:4 of the GATT 1994 in the context of their discussion of Article 2.1 of the TBT Agreement, the Panel asked them to elaborate on whether an analysis of "likeness" and "less favourable treatment" under Article 2.1 of the TBT Agreement should be assumed to have exactly the same contours as the same analysis under Article III:4 of the GATT 1994. Further, if this is the case, the Panel asked the parties to confirm whether this would imply that all interpretations developed by panels and the Appellate Body in the context of Article III:4 of the GATT 1994 are entirely transposable to Article 2.1. The Panel also asked the parties to comment on the differences in the wording used under each provision.

7.198 Mexico responded that although the language used in Article 2.1 is different from that used in Articles III:4 and I:1, both of these GATT 1994 provisions offer guidance on how to interpret Article 2.1.³⁵³ Further, Mexico explained that the essence of the non-discrimination obligations in the GATT 1994 and other covered agreements is that like products should be treated equally, irrespective of their origin. It also noted the importance of these provisions for Members, in the context of non-tariff barriers such as the US dolphin-safe labelling provisions. Accordingly, it is essential that the

³⁴⁷ Mexico's first written submission, para. 257.

³⁴⁸ Mexico's first written submission, para. 258.

³⁴⁹ Mexico's first written submission, para. 259.

Mexico's first written submission, para. 260.

³⁵¹ Mexico's first written submission, para. 261.

³⁵² Mexico's first written submission, para. 262.

³⁵³ Mexico's response to Panel question No. 58(a) and (b), paras. 172.

elements of these provisions, including the definitions of "like products" and "less favourable treatment" be carefully interpreted and applied. Mexico argues that the long history of interpreting and applying these terms in the context of Article III:4 and other provisions of the GATT 1994 should be relied upon when interpreting Article 2.1 of the TBT Agreement in order to maintain, Mexico argues that the integrity of the non-discrimination obligations contained in that provision.³⁵⁴

7.199 As for the term "like products", Mexico recalled the Appellate Body's rulings on the need to interpret this term as it appears in different provisions of the covered agreements in light of the context and the object and purpose of the provision at issue and the object and purpose of the covered agreement in which the provision appears. In light of this jurisprudence, Mexico concludes that:

"Both Article III:4 and Article 2.1 apply to technical regulations so, at a general level, their context is similar. Given that the specific language used in both Articles when referring to the term 'like products' – i.e., 'shall be accorded treatment no less favourable than that accorded to like products of national origin' – is identical, the immediate context of the term when used in the two provisions is also identical. Accordingly, it is appropriate to give the term the same meaning under both provisions. Thus, as stated in Mexico's First Written Submission, the four criteria used to determine likeness under Article III:4 should be used to determine likeness under Article 2.1."

7.200 Mexico considers it appropriate to give the term "treatment no less favourable" the same meaning under both Article III: 4 of the GATT and Article 2.1 of the TBT Agreement. In that regard it noted it is crucial that the term "treatment no less favourable" in Article 2.1 is interpreted and applied with reference to the equality of competitive opportunities for imported products as compared to like domestic products. Such competitive opportunities can be affected Mexico says, in a variety of ways, not only by measures directly regulating products and restricting imports but also by those that indirectly affect imports.³⁵⁷ Therefore Mexico concludes that the assessment for "treatment no less favourable" in Article 2.1 of the TBT Agreement should be the same as under Article III:4 of the GATT 1994 i.e. whether a measure modifies the conditions of competition in the relevant market to the detriment of imported products.³⁵⁸

7.201 Regarding the differences in wording between Article 2.1 of the TBT Agreement and Article I:1 of the GATT 1994, Mexico noted that the language of Article I:1 provides that "any advantage, favour, privilege or immunity granted by any [Member] to any product originating in ... any other country shall be accorded immediately and unconditionally to the like product originating in ... the territories of all other [Members]" whereas Article 2.1 refers to "products imported from the territory of any Member shall be accorded treatment no less favourable than that accorded ... to like products originating in any other country". 359

³⁵⁴ Mexico's response to Panel question No. 58(a) and (b), paras. 173.

³⁵⁷ Mexico's response to Panel question No. 58(a) and (b), para. 177.

^{355 (}footnote original) Appellate Body Report, European Communities – Measures Affecting Asbestos and Asbestos-Containing Products, WT/DS135/AB/R, adopted 5 April 2001, paragraphs 88-89. (Mexico's response to Panel question No. 58(a) and (b), fn 77).

³⁵⁶ Mexico's response to Panel question No. 58(a) and (b), para. 175.

^{358 (}footnote original) Appellate Body Report, Dominican Republic – Measures Affecting the Importation and Internal Sale of Cigarettes, WT/DS302/AB/R, adopted 19 May 2005, paragraphs 91, 93 and 96; Appellate Body Report, Korea – Measures Affecting Imports of Fresh, Chilled and Frozen Beef, WT/DS161/AB/R, WT/DS169/AB/R, adopted 10 January 2001, paragraphs 135, 137 and 144. (Mexico's response to Panel question No. 58(a) and (b), fn 81).

³⁵⁹ Mexico's response to Panel question No. 58(a) and (b), para. 179.

7.202 Although the language in the two provisions differs, Mexico notes that both set out most-favoured-nation treatment obligations. It also observes that:

"The Appellate Body has observed that '[a]part from Article I:1, several 'MFN-type' clauses dealing with varied matters are contained in the GATT 1994' and '[t]he very existence of these other clauses demonstrates the pervasive character of the MFN principle of non-discrimination'. The provisions of the GATT 1994 referred to by the Appellate Body when making this statement use very different language to express the MFN obligation including language similar to Article 2.1 of the *TBT Agreement* as illustrated in GATT Article IX:1—i.e., 'treatment ... no less favourable than the treatment accorded to like products of any third country'."

7.203 Mexico concludes that Article I:1 protects the competitive opportunities of imported products, not trade flows, and imposes upon WTO Members the obligation to treat like foreign products equally, irrespective of their origin. In its view, the jurisprudence interpreting the obligation in Article I:1 of the GATT 1994 is relevant to the interpretation of Article 2.1 of the TBT Agreement.³⁶¹

7.204 In its second written submission, Mexico addresses several issues it considers common to all its discrimination claims, the TBT Agreement and the GATT 1994, including the issue of likeness. However, Mexico addresses separately the question of the "advantage accorded immediately and unconditionally" under Article I:1 of the GATT 1994. Finally, as for Article 2.1 of the TBT Agreement, Mexico refers again to the arguments presented to support its claims under Articles I:1 and III:4 of the GATT 1994. ³⁶²

7.205 The United States observes that, notwithstanding the inapplicability of Article 2.1 of the TBT Agreement to the present dispute, an analysis of likeness and less favourable treatment under Article 2.1 of the TBT Agreement should not be exactly same as under Article III:4 of the GATT 1994 as there are important textual and contextual differences between the two. 363 One such difference the United States observed is that Article 2.1 states that "Members shall require that in respect of technical regulations, products imported from the territory of any Member shall be accorded treatment no less favourable than that accorded to like products of national origin". The United States emphasizes that Article 2.1 applies "in respect" of a technical regulation. 364

7.206 The United States further observes that the language in Article I:1 of the GATT 1994 and Article 2.1 of the TBT Agreement differs regarding the treatment afforded to imports from one Member as compared to the treatment afforded imports of another Member. Article 2.1 of the TBT Agreement refers to not affording "less favourable treatment" to imported products as compared to like products originating in other countries, whereas Article I:1 of the GATT 1994 states that any "advantage" or "privilege" granted to products originating in one country "shall be accorded immediately and unconditionally" to like products originating in other countries. These textual differences should be taken into account when interpreting Article I:1 of the GATT 1994 and Article 2.1 of the TBT Agreement respectively.

³⁶⁰ Mexico's response to Panel question No. 58(a) and (b), para. 180 (original footnote omitted)

³⁶¹ Mexico's response to Panel question No. 58(a) and (b), para. 181.

³⁶² Mexico's second written submission, paras. 125-191.

³⁶³ United States' response to Panel question No. 58(a) and (b), para. 132.

³⁶⁴ United States' response to Panel question No. 58(a) and (b), para. 132.

(ii) Approach of the Panel

7.207 Article 2.1 reads as follows:

"Members shall ensure that in respect of technical regulations, products imported from the territory of any Member shall be accorded treatment no less favourable than that accorded to like products of national origin and to like products originating in any other country."

- 7.208 This provision must be read in conjunction with the introductory passage of Article 2 ("With respect to their central government bodies:"), which makes clear that the provisions of Article 2, and thus the obligations it contains, relate only to the "central government bodies" of WTO Members. This is further confirmed by the title of Article 2, which reads "Preparation, adoption and application of technical regulations by central government bodies".
- 7.209 The terms of this provision thus suggest that a violation of Article 2.1 exists if two sets of conditions are met:
 - (a) the measure is a technical regulation prepared, adopted or applied by a Member's central government bodies; and
 - (b) products imported from the territory of a Member are accorded "less favourable treatment" than like products of national origin or originating in any other country in respect of this technical regulation.
- 7.210 We have already determined above that the US dolphin-safe labelling provisions constitute a technical regulation. We must now, in addition, determine whether they fall within the purview of Article 2 as a technical regulation prepared, adopted and applied by one or more "central government body" of the United States.
- 7.211 Annex 1.6 of the TBT Agreement defines a "central government body" as a "central government, its ministries and departments or any body subject to the control of the central government in respect of the activity in question". The measures at issue in this dispute are a federal US law (the DPCIA), a federal regulation (Sections of the US Federal Code), and a ruling by a federal court of the United States (the *Hogarth* ruling). It has not been disputed that these constitute acts of central government bodies of the United States.
- 7.212 We must now determine whether less favourable treatment is accorded to Mexican imported products than to like products of US origin or like products originating in any other country with respect to these measures. For that purpose, we consider first whether the products at issue are "like products". If we determine that they are, we will then need to consider whether "less favourable treatment" is being afforded by the United States, in respect of the US dolphin-safe labelling provisions, to Mexican products than to like products originating in the United States or in any other country.
- (b) Whether the products at issue are like
- 7.213 As described above, Mexico considers that Mexican tuna products are "like" tuna products of US origin and tuna products originating in any other country. The United States does not dispute this determination. Nonetheless, we must ascertain that this condition is met, to discharge our duty of

making an objective assessment of the matter before us in accordance with Article 11 of the DSU. 365 We also note that it is for the party asserting a fact or claim to bear the burden of proving this fact or claim³⁶⁶, and that it is therefore for Mexico to put forward sufficient evidence to raise a presumption that the tuna products at issue are "like".

7.214 In order to ascertain whether Mexico has demonstrated that Mexican tuna products and tuna products originating in any other country are "like", we must first clarify the meaning of the terms "like products" in Article 2.1 of the TBT Agreement.

Meaning of the term "like products" in Article 2.1 of the TBT Agreement

- 7.215 Pursuant to Article 3.2 of the DSU and as described earlier, we must interpret the terms "like products" in Article 2.1 of the TBT Agreement in accordance with the customary rules of interpretation of public international law.
- 7.216 Mexico recalls the Appellate Body's rulings on the need to interpret the term "like products" in light of the context and of the object and purpose of the provision at issue and of the object and purpose of the covered agreement in which the provision appears. ³⁶⁷ In light of these rulings, Mexico concludes that it is appropriate to give the term the same meaning under Article III:4 of the GATT and Article 2.1 of the TBT Agreement. Thus, as stated in Mexico's First Written Submission, the four criteria used to determine likeness under Article III:4 should be used to determine likeness under Article 2.1."368
- 7.217 Mexico also observes³⁶⁹ that the Appellate Body in EC Asbestos explained that "a determination of 'likeness' under Article III:4 is, fundamentally, a determination about the nature and extent of a competitive relationship between and among products"³⁷⁰ and that "the scope of 'like' in Article III:4 is broader than the scope of 'like' in Article III:2, first sentence.³⁷¹ Further, Mexico notes that the Appellate Body in the same dispute also "observed that the Report of the Working Party on Border Tax Adjustments outlined an approach for analyzing 'likeness' that has been followed and developed since by several panels and the Appellate Body."³⁷²
- 7.218 The United States observes that an analysis of "likeness" and "less favourable treatment" under Article 2.1 of the TBT Agreement should not be exactly the same as under Article III:4 of the

³⁶⁵ We note that this approach has been adopted by previous panels. In *US – Anti-Dumping Measures* on PET Bags, the Panel thus noted:

[&]quot;Notwithstanding the United States' decision not to contest Thailand's claim, we consider that we are still bound by Article 11 of the DSU to make an "objective assessment of the facts of the case and the applicability of and conformity with the relevant covered agreements, and make such other findings as will assist the DSB in making the recommendations or in giving the rulings provided for in the covered agreements". *US – Anti-Dumping Measures on PET Bags*, para. 7.5.

³⁶⁶ "Once that party has put forward sufficient evidence to raise a presumption that what is claimed is true, the burden of producing evidence shifts to the other party to rebut the presumption." Appellate Body Report, US – Wool Shirts and Blouses, p. 14.

³⁶⁷ (footnote original) Appellate Body Report, European Communities – Measures Affecting Asbestos and Asbestos-Containing Products, WT/DS135/AB/R, adopted 5 April 2001, paragraphs 88-89. (Mexico's response to Panel question No. 58(a) and (b), fn 77).

³⁶⁸ Mexico's first written submission, para. 259; Mexico's response to Panel question No. 58(a) and (b), para. 175.

Mexico's first written submission, para. 146.

³⁷⁰ (footnote original) Appellate Body Report, EC –Asbestos, para. 99.

³⁷¹ (footnote original) Id.

^{372 (}footnote original) Id., para. 101.

GATT 1994 as there are important textual and contextual differences between the two.³⁷³ The United States acknowledges that a determination of likeness under Article III:4 is essentially a determination about the nature and extent of a competitive relationship between and among products. However, it offers no specific views on how the interpretation of the term "like products" in Article 2.1 of the TBT Agreement might differ or not from that of the same term in Article III:4 of the GATT 1994.

7.219 We first note that the terms "like products" appear in a number of provisions in the covered agreements, including the non-discrimination obligations of Articles I:1, II:2, III:2 and III:4 of the GATT 1994 and that panels and the Appellate Body have interpreted these terms in some of these provisions. We note in this respect the Appellate Body's observation that:

"[W]hile the meaning attributed to the term 'like products' in other provisions of the GATT 1994, or in other covered agreements, may be relevant context in interpreting Article III:4 of the GATT 1994, the interpretation of 'like products' in Article III:4 need not be identical, in all respects, to those other meanings."³⁷⁴

Similarly, the interpretation of the term "like products" in other provisions of the covered agreements may provide relevant context and guidance in interpreting the same term in Article 2.1 of the TBT Agreement, but it need not be assumed that this term must be interpreted identically in all respects in both provisions. Rather, as expressed by the Appellate Body, "[i]n each of the provisions where the term "like products" is used, the term must be interpreted in light of the context, and of the object and purpose, of the covered agreement in which the provision appears". 375

7.221 Dictionary definitions of the term "like" point to "[h]aving the same characteristics or qualities as some other person or thing; of approximately identical shape, size, colour, character, etc., with something else; similar; resembling; analogous". This suggests that products would be "like" when they share common "characteristics or qualities" and resemble each other. However, as observed by the Appellate Body, this dictionary definition alone does not resolve all issues of interpretation, as it does not indicate which characteristics or qualities are important, the degree to which products must share such characteristic or qualities, or from whose perspective this is to be assessed. The Appellate Body has thus compared the concept of "likeness" to an "accordion":

"The concept of 'likeness' is a relative one that evokes the image of an accordion. The accordion of 'likeness' stretches and squeezes in different places as different provisions of the WTO Agreement are applied. The width of the accordion in any one of those places must be determined by the particular provision in which the term 'like'

³⁷³ United States' response to Panel question No. 58(a) and (b), para. 132.

Appellate Body Report, *EC – Asbestos*, para. 89. Appellate Body Report, *EC – Asbestos*, para. 88.

³⁷⁶ Oxford English Dictionary online, at www.oed.com.

³⁷⁷ "First, this dictionary definition of 'like' does not indicate which characteristics or qualities are important in assessing the 'likeness' of products under Article III:4. For instance, most products will have many qualities and characteristics, ranging from physical properties such as composition, size, shape, texture, and possibly taste and smell, to the end-uses and applications of the product. Second, this dictionary definition provides no guidance in determining the degree or extent to which products must share qualities or characteristics in order to be 'like products' under Article III:4. Products may share only very few characteristics or qualities, or they may share many. Thus, in the abstract, the term 'like' can encompass a spectrum of differing degrees of 'likeness' or 'similarity'. Third, this dictionary definition of 'like' does not indicate from whose perspective 'likeness' should be judged. For instance, ultimate consumers may have a view about the 'likeness' of two products that is very different from that of the inventors or producers of those products." Appellate Body Report, EC – Asbestos, para. 92.

is encountered as well as by the context and the circumstances that prevail in any given case to which that provision may apply. ..."³⁷⁸

- 7.222 Accordingly, our understanding of how narrowly or broadly the accordion of likeness "stretches and squeezes" for the purposes of our examination of Mexico's claims under Article 2.1 of the TBT Agreement in this case or, in other words, the "degree or extent to which products must share qualities or characteristics" and the perspective from which this is to be examined in the case at hand, must be informed by the fact that our examination takes place under Article 2.1 of the TBT Agreement, as well as by the context and circumstances that prevail in this case.
- 7.223 We also note that the terms of this provision very closely mirror those of Article III:4, the national treatment obligation in the GATT 1994, so that it may be possible to seek guidance from the interpretation of that provision. At the same time, we are mindful that Article 2 of the TBT Agreement does not contain an introductory paragraph comparable to Article III:1 of the GATT 1994, setting out a "general principle" that would inform our understanding of the exact degree or extent to which products must share qualities or characteristics so as to be considered like in the context of Article 2, and the perspective from which this is to be examined. In the context of Article III:4, the Appellate Body determined that:

"As products that are in a competitive relationship in the marketplace could be affected through treatment of *imports* 'less favourable' than the treatment accorded to *domestic* products, it follows that the word 'like' in Article III:4 is to be interpreted to apply to products that are in such a competitive relationship. Thus, a determination of 'likeness' under Article III:4 is, fundamentally, a determination about the nature and extent of a competitive relationship between and among products." ³⁷⁹

- 7.224 Although this statement was made in the context of Article III:4 of the GATT 1994, we find it pertinent also to an interpretation of the terms "like products" in Article 2.1 of the TBT Agreement.
- 7.225 The TBT Agreement applies to a limited set of measures, and our understanding of its terms, including the terms "like products" must be informed by this context. As expressed in the preamble of the TBT Agreement, this Agreement reflects the intention of the negotiators to:

"[E]nsure that technical regulations and standards, including packaging, marking and labelling requirements, and procedures for assessment of conformity with technical regulations and standards do not create unnecessary obstacles to trade."

To the extent that Article 2.1 contributes to avoiding "unnecessary obstacles to trade" arising from undue discrimination with respect to technical regulations, it seeks to preserve the competitive opportunities of products originating in any Member, in relation to technical regulations. Thus, the term "like products" under Article 2.1 of the TBT Agreement may be similarly understood as relating to "the nature and extent of a competitive relationship" between and among products.

7.226 We further note, as the Appellate Body did in relation to Article III:4 of the GATT 1994, that this does not necessarily imply that Members may not draw any regulatory distinctions, under Article 2.1 of the TBT Agreement, between products that have been determined to be like products.³⁸⁰

³⁷⁸ Appellate Body Report, *Japan – Alcoholic Beverages II*, p. 21 (DSR 1996:I, p. 97 at 114).

Appellate Body Report, *EC – Asbestos*, para. 99.

³⁸⁰ In *EC – Asbestos* the Appellate Body found that "... a Member may draw distinctions between products which have been found to be 'like', without, for this reason alone, according to the group of 'like'

The question of the treatment to be given to products that are like is addressed separately in the requirement of not affording treatment less favourable, which we consider in the next Section of our Report.

7.227 With these determinations in mind, we now consider whether Mexican tuna products are like tuna products of other origins, within the meaning of Article 2.1 of the TBT Agreement. As a preliminary matter, we must clarify what products are to be compared for the purposes of this analysis.

Products at issue

- 7.228 In its request for the establishment of a panel, Mexico refers to measures concerning the importation, marketing and sale of tuna and tuna products.³⁸¹ In its first written submission, Mexico similarly identified the subject products as tuna and tuna products, and analysed the likeness under Article III:4 of Mexican and US tuna and tuna products.³⁸²
- 7.229 In this context, Mexico explained that "tuna" includes all species of tuna purchased by canneries for processing into tuna products including yellowfin, albacore, and skipjack, and "tuna products" are defined in Section 1385(c)(5) as "a food item which contains tuna and which has been processed for retail sale, except perishable sandwiches, salads, or other products with a shelf life of less than 3 days". 383 The most common form of tuna products, Mexico explained, is "tuna in retailready cans or pouches."³⁸⁴
- 7.230 With reference to a US court decision, Mexico further explained that fresh tuna caught by US flag vessels in international waters is of US origin, tuna caught by Mexican vessels is of Mexican origin, and tuna caught by the vessels of other countries take the origin of the flag country in question.³⁸⁵ With reference to NAFTA rules of origin, Mexico also explained that the country where processing occurs is the country of origin of a tuna product.³⁸⁶ In response to a question by the Panel, the United States confirmed that in general, for tuna processed into canned or pouched tuna products. the country where the tuna is processed is the country of origin for that tuna product under the US law. 387 For *tuna*, the United States stated that the origin of tuna is not determined by where it was caught but by the flag of the vessel that caught it. 388
- 7.231 The Panel sought a clarification from both parties whether the comparison for the likeness analysis was between US and Mexican tuna in general, between Mexican tuna caught in the ETP by

imported products 'less favourable treatment' than that accorded to the group of 'like' domestic products. "Appellate Body Report, EC – Asbestos, para. 100.

Mexico's request for the establishment of a Panel, WT/DS381/4.

³⁸² Mexico's first written submission, paras. 114-120.

- 383 (footnote original) See Appendix A.
 384 (footnote original) See e.g.: Starkist http://www.starkist.com/template.asp?section=products /index.html (Exhibit MEX-54); Bumble Bee - http://www.bumblebee.com/Products/Family/?Family ID=1; and http://chickenofthesea.com/product line list.aspx?FID=3 Chicken Sea http://chickenofthesea.com/product_line_list.aspx?FID=11 (Exhibit MEX-50).
 - ³⁸⁵ Mexico's first written submission, para. 118.
 - ³⁸⁶ Mexico's first written submission, para. 119.
- ³⁸⁷ United States' response to Panel question No. 112, fn 48.(footnote original) Rules of origin for fish and fish products are codified in the U.S. tariff schedule. See, e.g., Harmonized Tariff Schedule of the United States (2010), General Note 12(b) and 12(n)(v) for North American Free Trade Agreement rule of origin for fish, Exhibit US-56.
 - ³⁸⁸ United States' response to Panel question No.15 (a), para, 44.

setting on dolphins and US tuna caught otherwise, or between US dolphin-safe tuna and Mexican dolphin-safe tuna. 389

- 7.232 Mexico responded that the relevant products are Mexican and US tuna products in general. In addition it observed that the method of fishing and geographic region in which the tuna are caught are unincorporated PPMs that are not relevant to the like products determination.³⁹⁰ The United States in turn clarified that the like products analysis under Article III:4 should compare US tuna products in general and imported tuna products in general.³⁹¹ In its second written submission, Mexico also referred to *tuna products* rather than tuna and tuna products. In light of Mexico's response, the Panel asked Mexico to clarify further whether it was no longer seeking findings on tuna as distinct from tuna products.³⁹² Mexico confirmed that its claims are limited to findings concerning tuna products, specifying that the great majority of Mexican tuna products are made from tuna caught by the Mexican fleet.³⁹³
- 7.233 In light of these clarifications, we understand Mexico to be seeking findings in relation to tuna products and not in relation to tuna as such. Accordingly, the products to be compared for the purposes of determining their likeness are US tuna products and Mexican tuna products, as well as tuna products originating in any other country.
- 7.234 We also observe that Mexico has explained that although its challenge applies in respect of all tuna products, for the purpose of demonstrating the violation, it would use the most common tuna product, which is tuna meat packaged in retail ready cans and pouches.³⁹⁴ Accordingly, we consider Mexico's analysis in respect of these specific tuna products and our findings relate to these products.

Whether Mexican tuna products are like tuna products originating in the United States or any other country

- 7.235 To demonstrate that Mexican tuna products and tuna products originating in the United States or any other country are like, Mexico has followed the approach derived from the GATT Working Party Report on *Border Tax Adjustments*. This approach is based on an analysis of four general criteria, reflecting "four categories of 'characteristics' that the products involved might share: (i) the physical properties of the products; (ii) the extent to which the products are capable of serving the same or similar end-uses; (iii) the extent to which consumers perceive and treat the products as alternative means of performing particular functions in order to satisfy a particular want or demand; and (iv) the international classification of the products for tariff purposes".
- 7.236 From this analysis, Mexico concludes that Mexican and US tuna products directly compete against each other in the US wholesale, distribution and retail market and that similarly, Mexican and US tuna directly compete against each other in the US cannery market.³⁹⁵ It states that all relevant evidence supports a finding that Mexican and US tuna products and tuna are like.³⁹⁶ Mexico underlines the panel's duty to "examine the evidence relating to each of those four criteria and, then,

³⁸⁹ Panel question No. 74.

³⁹⁰ Mexico's response to Panel question No. 74, paras. 272-273.

³⁹¹ United States' response to Panel question No. 74, para. 160.

³⁹² Panel question No. 144.

³⁹³ Mexico's response to Panel question No. 144, paras. 119-120.

³⁹⁴ Mexico's first written submission, para. 147

³⁹⁵ Mexico's first written submission para. 153.

³⁹⁶ Mexico's first written submission para. 154.

weigh all of that evidence, along with any other relevant evidence, in making an overall determination of whether the products at issue could be characterized as 'like'."³⁹⁷

7.237 The United States does not address Mexico's assertion that US and Mexican tuna and tuna products are "like" for the purposes of Article 2.1 of the TBT Agreement. Rather, it argues that the US dolphin-safe labelling provisions neither modify the conditions of competition to the detriment of Mexican tuna nor give a competitive advantage to US tuna and tuna products, insofar as they apply to any tuna and tuna products regardless of their origin. 398

7.238 We first note that the four general criteria used by Mexico as basis for its likeness analysis have been endorsed by the Appellate Body as "tools to assist in the task of sorting and examining the relevant evidence" for the purposes of determining "likeness" in the context of Article III:4 of the GATT 1994. In that context, the Appellate Body also observed that "in this determination, '[n]o one approach ... will be appropriate for all cases'. Rather, an assessment utilizing 'an unavoidable element of individual, discretionary judgment' has to be made on a case-by-case basis." We note in this respect that the Appellate Body has also determined that:

"The kind of evidence to be examined in assessing the 'likeness' of products will, necessarily, depend upon the particular products and the legal provision at issue. When all the relevant evidence has been examined, panels must determine whether that evidence, as a whole, indicates that the products in question are 'like' in terms of the legal provision at issue."

7.239 Similarly, in the context of an examination of likeness for the purposes of Article 2.1 of the TBT Agreement, not only the provision to be applied, but also "the context and the circumstances that prevail in any given case to which that provision may apply" have a bearing on the identification of the appropriate approach.

7.240 In light of our earlier conclusions in paragraph 7.223 above, and in the circumstances of this case, we find it *a priori* appropriate to consider the four general criteria used by Mexico, to determine whether Mexican and other tuna products are like within the meaning of Article 2.1 of TBT Agreement. These elements are apt to provide information about the extent and nature of the similarities between these products, so as to ascertain the nature and extent of their (actual or expected) competitive relationship.

7.241 We take note also of the Appellate Body's observation, in the context of Article III:4 of the GATT 1994 that "the adoption of a particular framework to aid in the examination of evidence does not dissolve the duty or the need to examine, in each case, all of the pertinent evidence". This consideration applies similarly in the context of Article 2.1 of the TBT Agreement.

7.242 Mexico first submits that the physical properties of Mexican tuna products are identical to those of US tuna products insofar as the products from both WTO Members comprise tuna meat in a retail-ready package. Mexico further observes that canned and pouched tuna meat from the various tuna species compete against each other in the US tuna market, confirmation of this is that the largest

³⁹⁸ United States' first written submission, para. 104.

³⁹⁷ (footnote original) Id., para. 109.

³⁹⁹ Appellate Body Report, EC – Asbestos, para. 102.

⁴⁰⁰ (footnote original) Appellate Body Report, Japan – Alcoholic Beverages II, p.114.

^{401 (}footnote original) Appellate Body Report, Japan – Alcoholic Beverages II, p.113.

⁴⁰² Appellate Body Report, *EC – Asbestos*, paras. 101–103.

⁴⁰³ Appellate Body Report, *EC – Asbestos*, para. 102.

seller of tuna products in the United States packages various species of tuna meat. 404 Mexico also observes that, to the extent that there are physical differences in the species of the tuna meat, such differences do not materially affect the competitive relationship between Mexican and US tuna products because Mexican and certain US tuna products contain tuna meat from identical tuna species such as yellowfin tuna and canned and pouched tuna meat from the various tuna species compete against each other in the US market. 405

7.243 It is not disputed that the physical characteristics and properties of Mexican tuna products and of tuna products of US origin and tuna products originating in any other country are identical, in that they all similarly contain tuna. The information cited in Mexico's submission suggests that tuna products may be made from a variety of tuna species 406 We note in this respect that, in other parts of its arguments, Mexico suggested that some species of tuna have more commercial value than others. 407 However, neither party has suggested or demonstrated that these various products would not be in competition on the same market as a result, or that such variations would have an impact on the extent to which Mexican and US tuna products compete with each other on the US market, such as to make them unlike for the purposes of Article 2.1 of the TBT Agreement. 408

7.244 Mexico further observes that the end uses of Mexican tuna products and tuna products of US or other origin are identical, insofar as tuna products are destined for consumption by final consumers. 409 We note that it is not disputed that US and Mexican tuna products have the same end uses. We also note that it is not disputed that Mexican tuna products and tuna products from third countries have the same end uses.

7.245 Mexico also observes that Mexican and US tuna products and tuna are classified under the same tariff subheading 1604.14 of the Harmonized System, which relates to "Tunas, Skipjack and Bonito (Sarda Spp.) (Prepared or Preserved)". 410 That the tariff classification is identical for prepared

http://www.starkist.com/template.asp?section=products/index.html. (footnote original) SeeExhibit MEX-54.

⁴⁰⁵ Mexico's first written submission, para. 148.

^{406 &}quot;Classic" tuna products are made of chunk light tuna, chunk or solid white albacore tuna; "gourmet" tuna products are made of select chunk light tuna. solid light tuna, yellowfin tuna and solid white albacore, see http://www.starkist.com/template.asp?section=products/index.html. (Mexico's first written submission, fn 100).

⁴⁰⁷ See in particular Mexico's response to Panel question No.24 and 38 of the Panel, at fn 26 and

para. 43.

We note in this respect the observations of the Appellate Body, in the context of the second sentence of Article III.2 of the GATT 1994, to the effect that a degree of variation is inevitable in the context of grouping products for the purposes of a determination of whether they are "directly competitive or substitutable" under that provision. Similarly, an assessment of "likeness" may involve a grouping of products with some degree of variation in features. The key point, however, is to ascertain whether these products are "like" the domestic products to which they are being compared:

[&]quot;... Some grouping is almost always necessary in cases arising under Article III:2, second sentence, since generic categories commonly include products with some variation in composition, quality, function and price, and thus commonly give rise to sub-categories. From a slightly different perspective, we note that "grouping" of products involves at least a preliminary characterization by the treaty interpreter that certain products are sufficiently similar as to, for instance, composition, quality, function and price, to warrant treating them as a group for convenience in analysis. But, the use of such "analytical tools" does not relieve a panel of its duty to make an objective assessment of whether the components of a group of imported products are directly competitive or substitutable with the domestic products....

Whether, and to what extent, products can be grouped is a matter to be decided on a case-by-case basis." (*Korea – Alcoholic Beverages*, paras. 142-143).

409 Mexico's first written submission para. 150.

⁴¹⁰ Mexico's first written submission, para. 152 and fn 102.

or preserved tuna of all species confirms that tuna products made from different species are, for commercial purposes, in essence the same product.

7.246 In sum, an analysis of the first three *Border Tax Adjustment* criteria suggests that the products at issue share common physical characteristics and properties, end uses and tariff classifications. Indeed, they are in essence the same products, processed in a different country.

7.247 The fourth and final "general criterion" considered by Mexico is "consumer preferences", which the Appellate Body has defined in the context of Article III:4 of the GATT 1994 as the "extent to which consumers are – or would be – willing to choose one product instead of another to perform those end-uses". Mexico considers that, but for the regulatory distinction that is at the core of the dispute, consumers' tastes and habits are identical with respect to Mexican and US tuna products and tuna. The United States does not challenge this conclusion.

7.248 We note that the European Union, in its third party submission, observed that the elements presented to the Panel suggest that there may be different consumer perceptions and preferences with respect to dolphin-safe and not dolphin-safe tuna and tuna products, and that this may have an impact on whether the two types of products are like.⁴¹³

7.249 The information presented to the Panel does suggest that US consumers have certain preferences with respect to tuna products, based on their dolphin-safe status, and we do not exclude that such preferences may be relevant to an assessment of likeness. To the extent that consumer preferences, including preferences relating to the manner in which the product has been obtained, may have an impact on the competitive relationship between these products, we consider it *a priori* relevant to take them into consideration in an assessment of the likeness. However, we are not persuaded that, in the circumstances of this case, a consideration of US consumer preferences relating to the dolphin-safe status of tuna products should lead us to modify our conclusion with respect to the likeness of US and Mexican tuna products and tuna products originating in any other country.

7.250 The basis for our analysis is a comparison between Mexican tuna products and tuna products of US origin and tuna products originating in any other country, not between dolphin-safe and not dolphin-safe tuna. A comparison on the basis of dolphin-safe status would imply that Mexican tuna products are assumed not to be dolphin-safe while US tuna products and tuna products originating in any other country would be assumed to be dolphin-safe. However, we see no basis for making such

⁴¹³ European Union's third party written submission, para. 28.

⁴¹¹ Appellate Body Report, *EC—Asbestos*, para. 117.

⁴¹² Mexico's first written submission para. 151.

⁴¹⁴ In *EC – Asbestos*, the European Communities had requested the Appellate Body to find the Panel's approach had misconstrued the relationship between Articles III:4 and XX of the GATT 1994 by requiring the likeness of two products be determined solely on the basis of commercial factors. In the EC's view, "If noncommercial considerations may only be considered at the Article XX stage of the analysis, then the list of policy for which regulators may distinguish between products is unduly limited to the categories listed in Article XX." The Appellate Body disagreed with the Panel that considering evidence relating to the health risks associated with a product, under Article III:4, nullifies the effect of Article XX(b) of the GATT 1994. It stated that "The scope and meaning of Article III:4 should not be broadened or restricted beyond what is required by the normal customary international law rules of treaty interpretation, simply because Article XX(b) exists and may be available to justify measures inconsistent with Article III:4. The fact that an interpretation of Article III:4, under those rules, implied a less frequent recourse to Article XX(b) does not deprive the exception in Article XX(b) of effet utile." Evaluating evidence relating to the health risks arising from the physical properties of a product does not, in the Appellate Body's opinion, prevent a measure which is inconsistent with Article III:4 from being justified under Article XX(b). The Appellate Body found the Panel erred in excluding the health risks associated with chrysotile asbestos fibres from its examination of the physical properties of that product. Appellate Body Report in EC – Asbestos, paras. 34, 115 and 116.

an assumption at this stage of our analysis.⁴¹⁵ We also note that it has not been suggested that, to the extent that US consumers would distinguish between different tuna products based on their dolphin-safe status, they would not apply this distinction to all tuna products, whatever their origin. As observed by Mexico, the preferences of US consumers are identical, in respect of US and Mexican tuna products and indeed, tuna products originating in any other country. In light of these elements, we find that an examination of US consumer preferences in relation to the dolphin-safe status of tuna products does not modify our conclusion that Mexican tuna products are like tuna products of US origin and tuna products originating in any other country.

- 7.251 We therefore conclude that Mexico has established that Mexican tuna products are like tuna products of US origin and tuna products originating in any other country within the meaning of Article 2.1 of the TBT Agreement.
- (c) Whether Mexican tuna products are afforded less favourable treatment than tuna products originating in the United States and other countries in respect of the US dolphin-safe labelling provisions
- 7.252 Having determined that Mexican tuna products and tuna products of US and other origins are like products within the meaning of Article 2.1, we must now determine whether Mexican tuna products are afforded less favourable treatment than tuna products of US and other origins in respect of the US dolphin-safe labelling provisions.
- (i) Arguments of the parties
- 7.253 In its first written submission, 416 Mexico explained that the less favourable treatment arising from the measures is a result of the following:
 - (a) Mexican tuna are almost exclusively caught in the ETP using purse seine nets set on dolphins; Mexican tuna products containing this tuna cannot be designated as dolphin-safe although the Mexican fleet complies with the stringent requirements of AIDCP;
 - (b) the US fleet fishes outside the ETP using other methods such as setting on FADs; tuna products containing this tuna can be labelled dolphin-safe under the measures, even where marine mammal mortality might have occurred and bycatch reduction might be compromised;
 - (c) accordingly, Mexican tuna products cannot be labelled dolphin-safe while US tuna products can;
 - (d) it is established that participants in the US market are sensitive to "issues related to dolphin mortality" and will make decisions on the basis of whether the products are designated as dolphin-safe. Most participants will not purchase tuna products that are not designated as dolphin-safe;

what the US measures at issue define it to mean, i.e. inter alia, tuna caught without setting on dolphins. This definition is, however, disputed between the parties. Furthermore, as observed by the United States and acknowledged by Mexico, not all of the tuna of Mexican origin is caught by setting on dolphins.

416 Mexico's first written submission, para. 165.

- (e) accordingly, most US market participants will not purchase, offer for sale, distribute or use Mexican tuna products but will purchase US tuna products. The US measures prevent Mexico and Mexican tuna industry from taking action to re-balance the competitive opportunities between Mexican and like US tuna products by labelling their tuna products as dolphin-safe and promoting the integrity and legitimacy of the AIDCP standard;
- (f) although the measures apply to tuna products, they have an indirect discriminatory effect on Mexican tuna, because canneries will not accept Mexican tuna for processing.
- 7.254 Subsequently, as we have clarified, Mexico expressly limited its claims to tuna products and not tuna. As we understand it, therefore, the final element identified in (f) is no longer relevant to Mexico's claim.
- 7.255 In its rebuttal submission, Mexico also clarifies that its discrimination claims "are not dependant on demonstrating that the treatment of ETP and non-ETP fisheries is different" and that "the factual basis of Mexico's discrimination claims is that the *prohibition* against the use of the dolphin-safe label on most Mexican tuna products denies competitive opportunities to those products compared to like product from the United States and other countries". 417
- 7.256 The United States argues that Mexico has failed to establish that the origin-neutral conditions (the manner or the place in which the tuna was caught) under which tuna products may be labelled dolphin-safe do actually accord different treatment to imported tuna products. Specifically, the United States argues that Mexico has not adduced evidence to show that the US dolphin-safe labelling provisions although origin neutral on their face in fact use the manner or the place in which the tuna was caught to single out imports. On the contrary the evidence leads to the opposite conclusion. 418
- 7.257 The United States considers that Mexico has failed to establish that the origin-neutral conditions under which tuna products may be labelled dolphin-safe in fact accord different treatment to imported tuna products. The United States considers in particular the following aspects:
 - (a) the overwhelming majority of tuna products on the US market are imported and the vast majority of those products are not caught by setting on dolphins⁴¹⁹;
 - (b) Mexican vessels use methods other than setting on dolphins to catch tuna⁴²⁰;
 - (c) US vessels set on dolphins to catch tuna at the time the US dolphin-safe labelling provisions were enacted⁴²¹;
 - (d) that Mexican vessels fish in the ETP is not a basis to argue that the US dolphin-safe labelling provisions accord different treatment to Mexican tuna products; 422 and finally

⁴¹⁷ Mexico's second written submission, para. 150.

⁴¹⁸ United States' second written submission, para. 21.

⁴¹⁹ United States' second written submission, paras. 22-23.

⁴²⁰ United States' second written submission, paras. 24-27.

⁴²¹ United States' second written submission, para. 28.

⁴²² United States' second written submission, paras. 29-32.

(e) there is no evidence that the objective of the US dolphin-safe labelling provisions is to afford protection to domestic production.

7.258 The United States also explains why, in its view, to the extent that there are any differences in documentation required to substantiate dolphin-safe claims, they are calibrated to the risk that dolphins may be killed or seriously injured when tuna is caught. In the US view, dolphin mortalities outside the ETP are not comparable to dolphin mortalities and injury in the ETP. The United States further considers that the structure of the statute shows a clear relationship with its stated objectives and does not support the conclusion that the dolphin-safe labelling conditions are applied in a manner so as to afford protection.

7.259 As far as the MFN component of its claim is concerned, Mexico refers to its arguments presented under Articles I:1 and III: 4 of the GATT on the basis that "[t]his provision of the TBT Agreement imposes national treatment and most-favoured nation treatment obligations on technical regulations. It is similar to the national treatment obligation in Article III:4 of the GATT 1994, except that it applies both national treatment and most-favoured-nation obligations. Like Articles I:1 and III:4 of the GATT 1994, Article 2.1 prohibits both *de jure* discrimination and *de facto* discrimination".

7.260 In that respect, Mexico observes, as in respect of its claim under Article III:4, that the US measures do not, on their face, discriminate on the basis of the foreign country that is the source of particular tuna and tuna products; rather, they discriminate on the basis of where the tuna is harvested and the fishing method. But this has the effect of favouring tuna and tuna products from some countries over others because different countries harvest tuna in different ocean fisheries using different fishing methods. In its opinion, by reflecting the longstanding fishing practice of the Mexican fishery, whilst the fleets of other countries fish outside of the ETP using other fishing methods such as purse-seine nets that are set upon FADs, the US dolphin-safe labelling provisions de facto discriminate against Mexican tuna insofar as tuna products that contain tuna originating from these countries' fleets can be designated as dolphin-safe and a dolphin-safe label can be affixed to those products whilst Mexican tuna products cannot be designated as dolphin-safe.

7.261 Given the preference of the US market for products labelled as safe, Mexico asserts that the US measures prevent Mexico and the Mexican tuna industry from taking action to re-balance the competitive opportunities between Mexican tuna products and like products originating in other countries by labelling their tuna products as dolphin-safe and promoting the legitimacy and integrity of the AIDCP standards, which in its view, has led to a situation whereby tuna and tuna products from certain countries are prevalent in the US market, while it is not commercially viable to sell imported Mexican tuna products in the US market.⁴²⁸

7.262 Mexico concludes that due to the lack of dolphin-safe designation, fish canneries located in other WTO Members will not accept Mexican tuna for processing into tuna products. These canneries will, however, accept tuna in other countries that can be designated dolphin-safe once processed. Therefore, for the same reasons as the one alleged under Article I:1 of the GATT, the US measures do

⁴²³ United States' second written submission, para. 38.

⁴²⁴ United States' second written submission, paras. 33-36.

⁴²⁵ United States response to Panel question No. 150, para. 106.

⁴²⁶ Mexico's first written submission, para.256.

⁴²⁷ Mexico's first written submission, para. 185.

⁴²⁸ Mexico's first written submission, para. 185.

not accord products imported from the territory of any Member (namely, Mexico) treatment no less favourable than that accorded to like products originating in any other country. 429

7.263 According to the United States, the US dolphin-safe labelling provisions do not afford less favourable treatment to Mexican tuna products *vis-à-vis* products of other origin insofar as they afford the use of the dolphin-safe label equally to all tuna products that meet the conditions set out in those provisions and deny that possibility equally to all tuna products that fail to meet those conditions. In the United States' view, the fact that the US dolphin-safe labelling provisions prohibit tuna products containing tuna caught by setting on dolphins – a technique that some Mexican vessels happen to use – does not mean that the US provisions afford less favourable treatment to Mexican tuna and tuna products. Instead, it means that Mexican vessels have chosen to set on dolphins to catch tuna and, because of that choice, tuna products that contain tuna caught by setting on dolphins by those vessels do not meet the conditions necessary to use the dolphin-safe label, just as tuna products containing tuna caught by vessels flagged to any other country, including the United States, do not meet the conditions necessary to use the dolphin-safe label.

7.264 In response to a question by the Panel on whether less favourable treatment under Article 2.1 of the TBT Agreement should be assumed to have exactly the same contours as the same analysis under Article III:4 of GATT 1994, Mexico noted that the provision contains two non-discrimination obligations applicable to technical regulations, one that is similar to the national treatment obligation in Article III:4 and the other that is similar to the most-favoured-nation obligation in Article I:1, although the language used in Article 2.1 is different from that used in Articles III:4 and I:1. ⁴³¹ It observed however that the immediate context of the term "treatment no less favourable" is identical in Articles III:4 and 2.1 and that therefore, it is appropriate to give the term the same meaning under both provisions. ⁴³² In its view, it is crucial that the term "treatment no less favourable" in Article 2.1 is interpreted and applied with reference to the equality of competitive opportunities for imported products as compared to like domestic products. Mexico contends that such competitive opportunities can be affected in a variety of ways, not only by measures directly regulating products and restricting imports but by those that indirectly affect imports. In that regard, the concept of the equality of competitive opportunities is flexible and protects the integrity of the non-discrimination obligations in a broad range of factual circumstances, Mexico says.

7.265 For that reason, Mexico concludes that the assessment for "treatment no less favourable" in Article 2.1 of the TBT Agreement should be the same as under Article III:4 of the GATT 1994 i.e., whether a measure modifies the conditions of competition in the relevant market to the detriment of imported products. Mexico acknowledges the different language used in Article I:1 of the GATT 1994 and Article 2.1 of the TBT Agreement but however notes that while the language in the two provisions differs, both set out the MFN treatment obligations which is a cornerstone of the GATT and one of the pillars of the WTO trading system. Mexico finally recalls that Article I:1 protects the competitive opportunities of imported products, not trade flows and it imposes upon WTO Members the obligation to treat like foreign products equally, irrespective of their origin and that it is in the sense of the protection of competitive opportunities for imported products from one WTO Member vis-à-vis like products from other WTO Members that the jurisprudence interpreting

⁴²⁹ Mexico's first written submission, para. 261.

⁴³⁰ United States' first written submission, para. 108.

⁴³¹ Mexico's response to Panel's question No. 58(a), para 172.

⁴³² Mexico's response to Panel's question No. 58(a), para 176.

⁴³³ Mexico's response to Panel's question No. 58(a), para 177.

⁴³⁴ Mexico's response to Panel's question No. 58(a), para. 178.

⁴³⁵ Mexico's response to Panel's question No. 58(a),para. 180.

the obligation in Article I:1 of the GATT 1994 is relevant to the interpretation of Article 2.1 of the TBT Agreement. 436

7.266 In response, the United States emphasizes the textual differences between Article I:1 of the GATT 1994 and Article 2.1 of the TBT Agreement which it argues should be taken into account when interpreting Article I:1 of the GATT 1994 and Article 2.1 of the TBT Agreement respectively. It holds that with respect to the U.S. arguments in connection with Mexico's claim under Article 2.1 of the TBT Agreement, Mexico did not present any evidence or argument in support of that claim that differed from the evidence and argument it presented in connection with its claims under Article I:1 or Article III:4 of the GATT 1994, whereas the United States has demonstrated in its response to Mexico's claims under Articles I:1 and III:4 of the GATT 1994 that those claims are without merit. 437

(ii) Analysis by the Panel

7.267 We must first clarify the meaning of the terms "less favourable treatment" in Article 2.1 of the TBT Agreement, before turning to a consideration of whether, in the present case, such less favourable treatment is afforded to Mexican tuna products with respect to the US dolphin-safe labelling provisions.

Meaning of "less favourable treatment" in Article 2.1 of the TBT Agreement

7.268 To recall, Article 2.1 of the TBT Agreement provides:

"Members shall ensure that in respect of technical regulations, products imported from the territory of any Member shall be accorded *treatment no less favourable* than that accorded to like products of national origin and to like products originating in any other country." (emphasis added).

7.269 As previously, we must interpret the terms "less favourable treatment" in this provision in accordance with the customary rules of interpretation of public international law, which direct the interpreter to determine the ordinary meaning of the terms, taken in their context, and in light of the object and purpose of the treaty.

7.270 We note that the term "less favourable treatment", like the term "like products", appears in more than one provision of the covered agreements, and has previously been interpreted in the context of some of these provisions, including Article III:4 of the GATT 1994 and, to some extent, Article II of the GATS. We also note that these terms fulfil, in these various provisions, a comparable function, i.e. they define the content of a non-discrimination obligation (e.g. MFN in the case of Article II of the GATS, and national treatment in the context of Article III:4). Interpretations of the same term in these other provisions may therefore provide useful guidance for its interpretation in the context of the TBT Agreement.

7.271 At the same time, as previously, we are mindful that the same terms may not have exactly the same contours in various provisions of different covered agreements, and that we must interpret the

⁴³⁶ Mexico's response to Panel's question No. 58(a),paras, 181-182.

⁴³⁷ United States' response to Panel's question No. 58 (a), para. 133.

⁴³⁸ See *US – Gasoline* (Panel Report, *US – Gasoline*, para. 6.10), *Japan – Film* (Panel Report, *Japan – Film*, para. 10.379), *Korea – Beef* (Appellate Body Report, *Korea – Various Measures on Beef*, paras. 135-137.), *EC – Asbestos* (Appellate Body Report, *EC – Asbestos*, para. 100), *EC – Bananas III* (Appellate Body Report, *EC – Bananas III*, paras. 231-234), *Canada – Autos* (Panel Report on *Canada – Autos*, para. 10.254).

terms "less favourable treatment" in Article 2.1 of the TBT Agreement taking account of the specific context in which they appear in this particular covered agreement. We note in this respect the Appellate Body's observations concerning the interpretation of the MFN obligation contained in Article II of the GATS, which is also expressed in terms of "less favourable treatment":

"The Panel interpreted Article II of the GATS in the light of panel reports interpreting the national treatment obligation of Article III of the GATT. The Panel also referred to Article XVII of the GATS, which is also a national treatment obligation. But Article II of the GATS relates to MFN treatment, not to national treatment. Therefore, provisions elsewhere in the GATS relating to national treatment obligations, and previous GATT practice relating to the interpretation of the national treatment obligation of Article III of the GATT 1994 are not necessarily relevant to the interpretation of Article II of the GATS. The Panel would have been on safer ground had it compared the MFN obligation in Article II of the GATS with the MFN and MFN-type obligations in the GATT 1994".

7.272 This observation suggests that the object of the specific provision at issue, as well as its wording, must be given due consideration in interpreting the terms "less favourable treatment". This is consistent with the requirements of the customary rules of interpretation, which require a consideration of the terms of the treaty in their context and in light of its object and purpose.

7.273 The relevant dictionary meanings of the word "treatment" suggest that it refers to the manner in which something is addressed. Dictionary definitions of the adjective "favourable" point to something "positive", and a synonym is "advantageous". The plain meaning of the term therefore suggests that imports of any Member must not be dealt with, in respect of technical regulations, in a manner less advantageous, than the like products of national or any other foreign origin. Read in light of the broader context of Article 2, less favourable treatment would arise "in respect of technical regulations", if imported products originating in any Member were placed at a disadvantage, compared to like domestic products and imported products originating in any other country, with respect to the preparation, adoption or application of technical regulations.

7.274 This is consistent with the general concept articulated by the Appellate Body that: "[t]he essence of the non-discrimination obligations is that like products should be treated equally, irrespective of their origin". Although this statement was made in the context of an analysis of Article I of the GATT 1994, in our view it sheds light on the essence of non-discrimination obligations in the covered agreements under Annex 1A generally (i.e. those agreements concerning trade in goods), both MFN and national treatment.

7.275 In our view, such "equality" of treatment does not necessarily imply *identity* of treatment for all products, but rather an *absence of inequality* to the detriment of imports from any Member. Here, we see some commonality between this requirement and the non-discrimination obligations embodied in Articles III:4 and Article I:1 of the GATT 1994. Under Article III:4, the Appellate Body thus observed that:

⁴³⁹ (*footnote original*) In addition to Article I (the fundamental MFN provision of the GATT), Articles III:7, IV(b), V:2 and V:5, IX:1 and XIII:1 are also MFN-type obligations in the GATT 1994. Appellate Body Report, *EC – Bananas III*, para. 231.

⁴⁴⁰ Oxford English Dictionary Online, at www.oed.com.

⁴⁴¹ Oxford English Dictionary Online, at www.oed.com.

⁴⁴² Appellate Body Report, *EC – Bananas III*, para. 190.

"[A] Member may draw distinctions between products which have been found to be 'like', without, for this reason alone, according to the group of 'like' imported products 'less favourable treatment' than that accorded to the group of 'like' domestic products."⁴⁴³

7.276 The same reasoning extends also to Article 2.1 of the TBT Agreement. The essence of the measures covered under this provision is to set out certain product characteristics or their related processes and production methods (or terminology, symbols, packaging, marking or labelling requirements as they apply to products or related processes and production methods), that must be complied with. Distinctions in treatment may therefore arise in this context, but they must not be designed or applied to the detriment of imports or imports of certain origins. In the context of Article 2.1 of the TBT Agreement, this question is also informed by the terms of the preamble, which makes it clear that measures covered by the TBT Agreement must not be "applied in a manner which would constitute a means of arbitrary or unjustifiable discrimination between countries where the same conditions prevail".

7.277 We also note that the obligation placed on Members is to "*ensure that* products ... *shall* be accorded treatment no less favourable". This language requires the achievement of "treatment no less favourable", and not only "reasonable measures" to that end, as is the case under Article 3 in relation to technical regulations by local government bodies and non-governmental bodies.

7.278 In the light of these determinations, we now consider whether Mexican tuna products are treated *less favourably* than US and/or other imported tuna products originating in any other country, i.e. in essence, whether Mexican tuna products are at a disadvantage compared to tuna products originating in the United States or in any other country, in respect of the US dolphin-safe labelling provisions.

Whether less favourable treatment is afforded to Mexican tuna products than to tuna products of US or originating in any other country

7.279 Mexico explains that the factual basis for its discrimination claim is that the "prohibition" against the use of the dolphin-safe label on most Mexican tuna products denies competitive opportunities to those products compared to like products from the United States and other countries. As described above, Mexico explains that its products do not have access to the label regulated by the measures, because Mexican tuna are caught "almost exclusively" in the ETP by setting on dolphins, while US tuna products have access to the label, because the US fleet fishes outside the ETP by other fishing methods. The "prohibition" that Mexico alleges therefore rests on an assumption that its products fall under one (less desirable) regulatory category, i.e. tuna caught by setting on dolphins, which is not eligible for the label under any circumstances, while those of the United States and a number of other countries fall under another (more desirable) one, i.e. tuna caught outside the ETP using other methods, which is eligible for the label.

7.280 In response to a question by the Panel concerning the basis for its claims, Mexico clarified that its non-discrimination claims "are not dependant on demonstrating that the treatment of ETP and non-ETP fisheries is different". As we understand it, therefore, Mexico does not challenge any differences in treatment arising from different regulatory categories for tuna caught in different fishing zones. Rather, Mexico's discrimination claim is based on the requirement of "no setting on

⁴⁴³ Appellate Body Report, *EC – Asbestos*, para. 100.

⁴⁴⁴ Mexico's second written submission, para. 150.

⁴⁴⁵ Mexico's response to Panel question No 145, para. 124.

dolphins" that conditions access to the US dolphin-safe label, wherever the fish is caught, and its implications in practice for Mexican tuna products.

7.281 What Mexico argues, in essence, is that this distinction *in fact* operates so as to exclude most Mexican tuna products from access to the label, while most US tuna products and those of a number of other countries, will benefit from it. As we understand it, this argument rests on the following key elements:

- (a) that access to the label has a value on the marketplace, such that it is an advantage to have access to the label, and a disadvantage not to have access to it;
- (b) that Mexican tuna products have no access (or virtually no access) to this added value on the market, while US and other imported tuna products do, by virtue of the different fishing practices of their respective fleets; and
- (c) that, as a result, the measures modify the conditions of competition on the marketplace to the detriment of Mexican tuna products.

7.282 With respect to the MFN component of its claim, Mexico argues that the measures at issue make the advantage of access to the label "subject to conditions with respect to the situation or conduct of Mexico, and that these conditions discriminate products from Mexico in favour of tuna products of other countries". It further argues that tuna products from "virtually all other countries, including the two largest exporters Thailand and the Philippines", are allowed to label their products dolphin-safe, and that, by contrast, exporters of Mexican tuna products are not. 447

7.283 As we have determined above, "less favourable treatment" would be afforded to Mexican tuna products in respect of the measures if they were placed at a disadvantage compared to US and/or other imported products with respect to the preparation, adoption or application of the US dolphin-safe measures. Mexico's claim in the present case is that it is *de facto* deprived of the benefit of access to the label, and thus at a competitive disadvantage on the US market because it fishes in the ETP by setting on dolphins while the US and other fleets fish outside the ETP by other methods.

7.284 We agree that, to the extent that access to the label would be an advantage granted by the measures, and that Mexican tuna products were to be deprived of access to this advantage while US or other imported tuna products originating in any other country had access to it, this would place Mexican tuna products at a disadvantage on the US market and thus constitute "less favourable treatment" within the meaning of Article 2.1 of the TBT Agreement. We therefore must consider whether (a) access to the label is an advantage and (b) Mexican tuna products are denied access to it under the measures, so that they are disadvantaged on the US market as compared to US or imported tuna products originating in any other country.

Whether access to the US dolphin-safe label constitutes an advantage on the US market

7.285 As described in Section 7.53(c) above, the measures at issue do not *require* any tuna products to be labelled "dolphin-safe", nor do they subject the right to market tuna products on the US market to meeting the conditions for the label. 448 Nonetheless, access to the label is controlled by compliance

⁴⁴⁷ Mexico's opening oral statement at the second substantive meeting (Article I).

⁴⁴⁶ Mexico's second written submission (Article I).

We note that the GATT 1947 US-Tuna panels examined embargoes on importation of tuna products in addition to the dolphin safe labelling scheme.

with the terms of the measures. Therefore, to the extent that access to the label is an advantage on the marketplace, this advantage is provided by the measures themselves. The exact value of the advantage provided by access to the label on the marketplace will depend on the commercial value attributed to it by operators on the market, including retailers and final consumers.

7.286 The United States observes that its dolphin-safe labelling provisions do not impose any choice on marketers of tuna products in terms of selling tuna products in the United States, and that Mexico has not identified anything in the provisions that would limit the marketing of tuna products that are not dolphin-safe or are not labelled dolphin-safe. It argues that the limited demand for non-dolphin-safe tuna products is a result of retailer and consumer preferences for dolphin-safe tuna products, not the US dolphin-safe labelling provisions.

7.287 We agree with the United States that US consumers' decisions whether to purchase dolphin-safe tuna products are the result of their own choices rather than of the measures. However, as observed above, it is the measures themselves that control access to the label and allow consumers to express their preferences for dolphin-safe tuna. An advantage is therefore afforded to products eligible for the label by the measures, in the form of access to the label.⁴⁵⁰

7.288 We further note that it is undisputed that US consumers are sensitive to the dolphin-safe issue. This is acknowledged by both Mexico and the United States⁴⁵¹, and is also confirmed by the evidence presented with the *amicus curiae* brief to which the United States has referred to in its answers to questions.⁴⁵² This evidence suggests that, following public campaigning by the environmental organization "Earth Island Institute" in the late 1980s (including through film footage shot in 1987-88 showing the capture and killing of dolphins during a fishing trip where setting on dolphins was used), tuna processors were under pressure to stop purchasing tuna caught in conditions that were harmful to dolphins.⁴⁵³ The evidence presented to the Panel also shows that major tuna processors reacted to these dolphin-safe concerns, and that this led to changes in their purchasing policies as of April 1990. These policies are still in place: such companies will not purchase tuna from vessels that fish in association with dolphins.⁴⁵⁴

7.289 These elements suggest that the dolphin-safe label has a significant commercial value on the US market for tuna products, as the only means through which dolphin-safe status can be claimed. Indeed, the evidence that canners refuse to buy tuna caught in association with dolphins suggest that the pressure is sufficient to induce processors of tuna products to avoid altogether tuna that would make their final products ineligible for the label. While this is only indirect evidence as to the final consumers' behaviours, it suggests that the producers themselves assume that they would not be able to sell tuna products that do not meet dolphin-safe requirements, or at least not at a price sufficient to warrant their purchase.

7.290 In addition, Mexico has presented evidence concerning retailers' and final consumers' preferences regarding tuna products. With respect to retailers, Mexico has presented an affidavit from an officer of a US company selling Mexican canned tuna in the US market, in which it stated that for

450 See the panel's findings in *Colombia – Ports of Entry*, in the context of Article I, where an "advantage" was granted in the form of "flexibility" to present import declarations in more favourable conditions. See Panel Report, *Colombia – Ports of Entry*, paras. 7.339-7.347.

⁴⁵¹ Mexico's first written submission, para. 160 and response to Panel question No. 40 (b) United States' response to Panel question No. 40 (a).

⁴⁴⁹ United States' second written submission, para. 61.

United States' response to Panel question No. 40(a), para. 98 which cites paras. 18-20, 25, 62-64 of the amicus curiae brief which in turn refer to Exhibits 1, 2, 3 and 4 attached to the amicus submission.

⁴⁵³ Exhibit Amicus curiae brief EX-2.

⁴⁵⁴ Exhibit US-32, Exhibit US-36 and Exhibit US-37.

several years the company attempted to sell Mexican tuna products in major US grocery chains which have refused to buy the products because they were unable to sell the brand that was not eligible for the dolphin-safe label. The affidavit provides evidence that some of the major US chains have expressly indicated that if the tuna product qualified to be labelled "dolphin-safe" they would sell it and that the company's inability to place the "dolphin-safe" label on the cans has directly affected its ability to sell the Mexican brand in the US. 455 This evidence confirms the value of the dolphin safe label on the US market.

7.291 We therefore agree with Mexico that access to the label provides an advantage on the US market. With this initial determination in mind, we now consider whether, as Mexico argues, its tuna products are effectively denied access to the advantage provided by the label, while US and other imported tuna products originating in any other country benefit from access.

> Whether Mexican tuna products are denied access to the advantage offered by the dolphin-safe label and are thereby disadvantaged, compared to tuna products originating in the United States or in any other country

7.292 As described above, Mexico analyses the existence of less favourable treatment in terms of comparing the treatment afforded to Mexican tuna products made from tuna caught in the ETP by setting on dolphins (not eligible for the label) and the treatment afforded to US tuna products made from tuna caught by other methods (eligible for the label) outside the ETP. Mexico's analysis is therefore based on comparing the treatment currently received by most of its own tuna (which is caught in conditions that do not entitle products made from it to the label) and that currently received by most of the US tuna (caught in conditions that entitle products made from it to bear the label).

7.293 The United States, on the other hand, considers that "the US dolphin-safe labelling provisions afford use of the dolphin-safe label equally to all tuna products that meet the conditions set out in those provisions and deny that possibility equally to all tuna products that fail to meet those conditions". 456 This approach appears to assume that the relevant comparison is whether within the categories of tuna products entitled or not entitled to the label, there is any differentiated treatment between Mexican tuna products and those products originating in the United States or in any other country. However, Mexico's complaint lies in the difference of treatment between products falling on either side of the regulatory distinction, i.e. between products that are entitled to the label and those that are not.

7.294 A comparison of treatment within each of the regulatory categories, although it may reveal differences in treatment, cannot therefore form the sole basis of the assessment of Mexico's claim in this case. The analysis to be conducted must necessarily, in this case, encompass a comparison of the treatment afforded to imported products that have access to the label and domestic products that do not have access to it. On the other hand, a comparison that considered *only* those imports that have no access to the label and those domestic products that do have access to it may also be insufficient, if it were to only reveal that some imports do not receive the better treatment afforded to some domestic products.

7.295 In this respect, we find that the Appellate Body's suggestion, in EC - Asbestos, that an enquiry into less favourable treatment involves a comparison of how the group of domestic like products and the group of like imports are treated, 457 provides useful guidance. It suggests that the starting point for the analysis should be the entire groups of both products identified as like products.

⁴⁵⁵ Exhibit MEX-58.

⁴⁵⁶ United States' first written submission, para. 109.

⁴⁵⁷ Appellate Body Report, *EC – Asbestos*, para. 100.

Accordingly, we approach this analysis on the basis of a comparison between the treatment afforded to the groups of US and Mexican tuna products as a whole, as well as Mexican tuna products compared to tuna products originating in any other country, in order to assess the relative situation of these products in respect of access to the dolphin-safe label regulated by the US dolphin-safe provisions.

7.296 The parties have also debated the relevant timeframe for the assessment, and we therefore find it useful to clarify this also as a preliminary matter.

7.297 The United States cites the panel report in Mexico – Taxes on Soft Drinks in support of an analysis of the measure at the time of its adoption, therefore comparing the impact the measure had on domestic and imported like products at the time it was enacted rather than the actual trade effect the measure has as of today on imported compared to domestic like products. Mexico objects to this approach, arguing that the US measures violate the non-discrimination provisions because their effect is to adversely modify the conditions of competition in the US marketplace for Mexican tuna products when compared to like tuna products from the United States and other countries, and that such denial of competitive opportunities is shown by the relevant facts as they existed at the time of the establishment of the Panel. 458 In particular, Mexico argues that it is not necessary for the Panel to determine whether the US measures when introduced in 1990 de facto discriminated against Mexican tuna products but that it must examine the facts as of the date of the Panel's establishment and determine whether, on the basis of those facts, there is *de facto* discrimination. ⁴⁵⁹ Mexico also argues that it is not possible, without a complete assessment of all relevant facts to determine whether there was de facto discrimination at that time and hence it cannot be assumed that they were not de facto discriminatory in 1990. 460 Mexico also observes that the European Union's position as a third party in the present dispute differs from its position in the context of the US – COOL dispute (DS386). 461

7.298 Finally, Mexico considers that the unadopted GATT 1947 panel report in *US – Tuna (Mexico)* further illustrates the importance of the Panel examining the facts as they existed at the time of its establishment insofar as, at the time of the GATT 1947 panel, the United States had a trade embargo in place against Mexican tuna. As a result, Mexican tuna imports were blocked so there was no factual basis upon which to assess the effects of the dolphin-safe labelling provisions in the US market. In other words, there was no evidence of the *de facto* discrimination and trade-restrictive effects of the challenged measures. 462 In Mexico's view, the *de facto* discriminatory and trade restrictive effects are clearly occurring today and it is the existence of those effects that gives rise to Mexico's request that the Panel rule upon its claims of *de facto* discrimination in this dispute. 463

7.299 We first recall that our mandate requires us to consider the consistency of the measures in accordance with our terms of reference. These terms of reference are defined by Mexico's request

⁴⁵⁸ Mexico's second written submission, para. 132.

⁴⁵⁹ Mexico's second written submission, paras. 133134.

⁴⁶⁰ Mexico's second written submission, paras. 135.

Mexico explains that in US - COOL, the European Union argued, in response to an argument that the Panel should consider historic facts that existed long before a Panel's establishment when assessing whether de facto discrimination exists today, that the Panel must look at future facts insofar as "the immediate regulatory shock" of a regulation does not, in itself, necessarily demonstrate less favourable treatment (see Mexico' rebuttal submission, fn 88) (The Panel has not been provided with the European Union's third party submission in that dispute).

462 Mexico's second written submission, para. 138.

⁴⁶³ Mexico's second written submission, para. 139.

⁴⁶⁴ See Appellate Body Report, *Brazil - Desiccated Coconut*, p. 22 ("The matter referred to a panel for consideration consists of the specific claims stated by the parties to the dispute in the relevant documents specified in the terms of reference").

for establishment of a panel. Accordingly, we are required to determine the existence of a violation as of that time, and to make recommendations for compliance on that basis. ⁴⁶⁵ In the present case, our task is therefore to determine whether, as of the date of the establishment of this Panel, the US dolphin-safe labelling provisions violate the relevant provisions of the covered agreements.

- 7.300 However, this does not necessarily imply that past events are irrelevant to an assessment of whether such a violation exists. It is quite possible that past events will shed light on the Panel's assessment of the situation as of its establishment, and an understanding of the situation as it presents itself today may legitimately be informed by its history. There is, in our view, no reason to exclude *a priori* such aspects from consideration, to the extent that they may inform a proper understanding of the situation as of establishment of the Panel.
- 7.301 With these initial clarifications in mind, we now consider whether Mexican tuna products are at a disadvantage, compared to tuna products of US and any other origin, with respect to access to the US dolphin-safe label under the measures at issue.
- 7.302 Mexico argues that its fleet primarily catches tuna in the ETP by setting on dolphins (thereby making it ineligible for the US label), while the US and other fleets catch tuna primarily outside the ETP using other fishing methods. It is therefore the combination of the fishing method used and geographical fishing zone exploited by different fleets with the requirements of the measures that results in what Mexico calls a "prohibition" on using the label for Mexican tuna products.
- 7.303 We therefore consider in turn the regulatory distinction upon which Mexico's claims rest (i.e. the requirement of not setting on dolphins), the fishing practices of the Mexican and other fleets, and the relative situation of Mexican and other tuna products originating in any other country on the US market, in order to determine whether Mexican tuna products are at a disadvantage in respect of the US dolphin-safe labelling provisions. Finally, we also address an additional argument by Mexico to the effect that less favourable treatment is demonstrated by the very nature of the measures at issue and the "pressure" that they exert on Mexico to modify its fishing practices.

The regulatory distinctions at issue: setting on dolphins as opposed to not setting on dolphins

7.304 As described in Section II above, the US dolphin-safe labelling provisions involve a number of regulatory distinctions, that form the basis for different substantive and documentary requirements for the obtention of a "dolphin-safe" label, depending on the fishery involved, the type of vessel, and the fishing method employed. As also described above, the key regulatory distinction of relevance to Mexico's discrimination claims is the distinction between the treatment of tuna products containing tuna caught by setting on dolphins and the treatment of tuna products containing tuna caught by other fishing methods. Both inside and outside the ETP, tuna is required not to be caught by setting on dolphins for tuna products made from it to qualify for the US dolphin-safe label. Because its fleet sets on dolphins while the United States and a number other fleets do not, Mexico considers that this requirement amounts to a "prohibition" on bearing the label for Mexican tuna. We therefore first consider this regulatory distinction in itself.

⁴⁶⁵ In accordance with Article 19.1 of the DSU, when a panel concludes that a measure is inconsistent with a covered agreement, it shall recommend that the Member concerned bring the measure into conformity with that agreement.

466 We note that in the context of an assessment of "less favourable treatment" under Article III:4 of

⁴⁶⁶ We note that in the context of an assessment of "less favourable treatment" under Article III:4 of GATT 1994, the Appellate Body's reference to whether the measure at issue "modifies" the conditions of competition suggest that a consideration of the situation today as compared to what it was prior to the enactment of the measure may be pertinent in assessing its impact. The same considerations apply in the present context.

⁴⁶⁷ Paragraph (d) of Section 1385 of the DPCIA.

7.305 We first note that, the fact that the measures distinguish between fish based on its capture method rather than its origin, this distinction is not inherently tied to the "national" origin of the fish. The fishing method at issue, setting on dolphins, is accessible to any fleet operating in an area where such method can be practised. Therefore, denying the label to tuna caught by "setting on dolphins" does not, in itself, imply that "less favourable treatment" is afforded to Mexican tuna products. Indeed, any fleet operating anywhere in the world must comply with the requirement.

7.306 The evidence before the Panel suggests that setting on dolphins occurs especially in the ETP, because of the regular association observed between tunas and dolphins in that area. However, we note that Mexico itself has suggested that association between tunas and dolphins, and setting on dolphins as a fishing method, is also practised outside of the ETP. The evidence before us confirms that such association and the practice of "setting" on dolphins has been observed in other areas of the world, though not on the commercial scale observed in the ETP. To the extent that setting on dolphins is or can be practised outside the ETP, the impact of the requirement not to set on dolphins would be felt also in those other fisheries.

7.307 We acknowledge that the requirement of not intentionally setting on dolphins as a condition for access to the label may be more onerous in practice for those fleets operating in the ETP, to the extent that the method is known to be regularly employed on a commercial scale in the ETP. However, even assuming that access to the label may be more onerous for purse seine fleets fishing in the ETP as compared to those fishing outside the ETP, this would in principle affect equally all fleets fishing in the ETP. It is undisputed that, as a fishery, the ETP is accessible to – and is in fact used by – a number of fleets, including those of Bolivia, Colombia, Ecuador, El Salvador, Guatemala,

⁴⁶⁸ Although Mexico has pointed out that there may be association that occurs outside of the ETP and setting on dolphins causing mortality or serious injury it has not challenged the United States' assertion that association occurs in the ETP. Mexico's second written submission, para. 3 and Mexico's response to Panel question No. 86.

469 "With regard to the U.S. argument that non-ETP tuna is not caught by setting nets on dolphins, Mexico has already explained that there is considerable evidence to the contrary. Mexico has already submitted evidence that dolphins and other marine mammals are being killed in substantial numbers outside the ETP by other fishing methods. Mexico has further explained that for non-ETP tuna, the United States has no mechanism to verify the accuracy of captains' certifications or the precision of non-U.S. canneries' tracking systems. If the goal of the U.S. measure genuinely is to give consumers accurate information on whether the specific tuna products they are purchasing were caught in a manner harmful to dolphins, the U.S. measure cannot fulfil that objective for tuna products made from non-ETP tuna." Mexico's response to Panel question No. 156, para. 147.

⁴⁷⁰ "Vessels outside the ETP that may intentionally or incidentally trap dolphins in their nets do not possess the training, the gear or the nets to safely release those dolphins." Mexico's response to Panel question No. 28, para. 54.

"There is substantial evidence of significant interactions between fishing vessels and dolphins outside the ETP, but the United States has devoted no resources to investigating or monitoring those interactions." Mexico's response to Panel question No. 86, para. 3.

Exhibit MEX-4: The association of yellowfin tuna with dolphins has been observed in all oceans of the world.

Exhibit MEX-97: List of bycatch species recorded as being ever caught by any major tuna fishery in the Atlantic/Mediterranean proves that there is dolphin bycatch with the use of purse seine nets (however it does not explicitly say whether the purse seine was set on dolphins).

Exhibit MEX-98: On the Western and Central Pacific tuna fishery: "There is a relatively high level of observer coverage in the equatorial purse-seine fishery, with 33,319 sets of observed in the last 11 years 1995-2005. Marine mammals were caught in a very small proportion of these observed sets, mainly from sets targeting tuna schools associated with either whales or dolphins.

Exhibit MEX-105 "... several interactions with species of special interest including ... pilot whales ... This could indicate that vessels are, intentionally or unintentionally, setting their nets around these species dolphins". (note: pilot whales are dolphins)

Honduras, Mexico, Nicaragua, Panama, Peru, Spain, the United States, Vanuatu, and Venezuela⁴⁷¹, In addition, the technique of fishing in association with dolphins in the ETP is not exclusive to Mexico since, as Mexico itself admits, it was the US fishing fleet who pioneered the method of circling herds of dolphins with purse seine nets and a number of other nations' vessels set on dolphins to catch tuna in the ETP. Therefore, any fleet deciding to fish for tuna in the ETP could set on dolphins.

7.308 Mexico explained that it has developed its tuna fishery in the ETP and that almost the entirety of its vessels have operated within that region. Mexico has explained that the ETP is a "traditional fishing ground" for it, due to its geographical proximity to the zone and that its fleet catches tuna "almost exclusively" in the ETP (by setting on dolphins). We note however that, as described above, a number of other fleets have also been fishing in the ETP for tuna. In fact, it appears that there are vessels of more nationalities fishing for tuna in the ETP today than there were at the time of enactment of the measures. The second statement of the measures.

7.309 Therefore, to the extent that the requirement of not setting on dolphins is based on a fishing method that may be used by vessels of any nationality operating where this method can be practiced, tuna of any nationality, including US and Mexican, as well as others, could potentially meet (or not meet) the requirements for dolphin-safe labelling. As described above, the measures at issue involve further distinctions, in particular in the certification requirements for different areas, depending on the status of the fishery, the type of vessel and fishing method used. However, Mexico has indicated that these differences in treatment are not the factual basis for its discrimination claims. We therefore do not address them in this context.⁴⁷⁶

7.310 We also note that the eligibility of a tuna *product* of a given nationality to bear (or not) the label does not necessarily depend on the manner in which the fleet bearing the flag of the same country operates, and on whether the fish caught by that fleet is eligible to be included in dolphin-safe tuna products. As mentioned earlier, the origin of a tuna product is determined not by the origin of the fish that it contains but by its place of processing. 477 Accordingly, Mexican tuna products need not be composed of Mexican tuna. Therefore, even assuming, for the sake of argument, that tuna of Mexican origin might more likely not be eligible for the label because it would be caught in the ETP by setting on dolphins, this would not necessarily imply that products processed in Mexico would be less likely to qualify for the label. This is because Mexican processors could choose to make their products from tuna of other origins meeting the requirements of the label. Mexico has asserted that Mexican canners use Mexican tuna. 478 The evidence presented to the Panel suggests that this is the case. According to the information submitted by Mexico, the Mexican tuna cannery industry is vertically integrated, and the major Mexican tuna products producers and canneries own their vessels, which operate in the ETP. Because of such integration Mexico contends that the effect of the US measure is to limit the ability of Mexican brands of tuna products from competing with the brands produced by canneries and canneries of third countries. 479 However, the integration of the Mexican

⁴⁷¹ "The Active Purse Seine Vessel Register, which lists all purse seine vessels authorized to fish for tuna in the ETP, includes vessels from Bolivia, Colombia, Ecuador, El Salvador, Guatemala, Honduras, Mexico, Nicaragua, Panama, Peru, Spain, the United States, Vanuatu, and Venezuela." (United States' response to Panel Question No 15 para. 44 and fn 36 referring to Exhibit US-15).

⁴⁷² Mexico's first oral statement, para. 7.

⁴⁷³ United States' response to Panel question No. 91, para. 24.

⁴⁷⁴ See Mexico's first written submission, paras. 165, 167, 186 and 188.

⁴⁷⁵ See Mexico's and the United States' responses to Panel questions Nos. 89 and 90.

⁴⁷⁶ Mexico's response to Panel question No. 145, para. 122.

⁴⁷⁷ See para. 7.230 above.

Mexico's response to Panel question No. 144, para. 120.

⁴⁷⁹ Mexico's response to Panel question No. 44, para. 108.

tuna industry does not mean that it must use Mexican tuna, or that it could not choose to require such tuna to be caught in conditions that would make it eligible for the US label.

7.311 In sum, an examination of the requirement of not setting on dolphins embodied in the US dolphin-safe provisions as a condition for access to the label does not suggest that this requirement in itself, places Mexican tuna products at a disadvantage as compared to US and other imported tuna products. Bearing in mind this initial determination, we now consider whether less favourable treatment nonetheless arises from the application of the measures, by reason of the practices of the fleets.

Fishing practices of US and Mexican tuna fleets, and their impact on access to the US dolphin-safe label for Mexican tuna products originating in the United States or in any other country

Practices of Mexican and other fleets

7.312 Mexico has explained that its tuna fleet fishes almost exclusively in the ETP by setting on dolphins, in conditions meeting the requirements for "dolphin-safe" designation under the AIDCP, but not meeting the requirements of the US measures for dolphin-safe labelling. The United States observes that Mexico's assertion that its fleet operates almost exclusively by setting on dolphins is incorrect insofar as one third of Mexico's purse seine fleet exclusively uses techniques other than setting on dolphins to catch tuna (and therefore tuna products that contain tuna caught by these vessels are eligible to be labelled dolphin-safe)⁴⁸¹ and the remaining two thirds of Mexico's purse seine fleet also opportunistically uses techniques other than setting on dolphins to catch tuna. The United States also notes that Mexico has several non-purse seine vessels that fish for tuna in the ETP that do not set on dolphins.

7.313 The United States also argued that during the first meeting with the Panel, Mexico acknowledged that 20 per cent of its catch is caught by techniques other than setting on dolphins. The United States observes that tuna caught by those vessels using those techniques are also eligible to use the dolphin-safe label. The United States also holds that Mexico does not substantiate its assertion that the third of its fleet that comprises vessels of 363 metric tons carrying capacity or less accounts for only 5 per cent or less of the Mexican fleet's total catch and that it is incorrect for Mexico to assert that "vessels [with 363 metric tons carrying capacity or less] are not economically viable". Mexico responded that its fleet under 363 metric tons of carrying capacity amounts to less than 4 per cent of the total catch of the Mexican tuna fleet and this tuna is sold in the domestic market. With respect to the vessels above 363 metric tons that sometimes use techniques other than

⁴⁸¹ (footnote original) U.S. first written submission, paras. 108. These are vessels that have a carrying capacity of 363 metric tons or less and for which the AIDCP prohibits the setting on dolphins to catch tuna. See id. para. 45, 91. (United States' second written submission, fn 24).

⁴⁸² (footnote original) U.S. first written submission, paras. 68-69. (United States' second written

462 (footnote original) U.S. first written submission, paras. 68-69. (United States' second written submission, fn 25).

⁴⁸³ See US response to Panel question No. 91, para. 14, noting that in 2010, the Mexican fleet comprised 69 purse seine vessels and 158 non-purse seine vessels in the ETP, and some non-purse seine vessels in the Caribbean, citing Exhibit US-16.

⁴⁸⁴ (*footnote original*) U.S. Closing Statement at the First Panel Meeting, para. 7. (United States' second written submission, fn 26).

⁴⁸⁵ (*footnote original*) Mexico Opening Statement at the First Panel Meeting, para. 25; Mexico Answers

to the First Set of Questions from the Panel, para. 40. (United States' second written submission, fn 27).

⁴⁸⁶ United States' second written submission, fn 30 (footnote omitted).

⁴⁸⁰ Mexico's first written submission, para. 165.

⁴⁸⁷ Mexico's second written submission, para. 40.

setting on dolphins, Mexico explains that this tuna cannot be labelled as dolphin-safe given that the US measures require that the dolphin set method not be used during an entire voyage. 488

- 7.314 Based on the above, we note that it is undisputed that at least two thirds of Mexico's purse seine tuna fleet fishes in the ETP by setting on dolphins (therefore fishing for tuna that would not be eligible to be contained in a "dolphin-safe" tuna product under the US dolphin-safe labelling provisions). We also take note of Mexico's indication that the portion of its fleet that the United States identifies as catching tuna eligible to be included in dolphin-safe tuna products (i.e. vessels under 363 metric tons) represents only a limited share of the total catch of the fleet, although we also note that the exact figures are disputed. 489
- With respect to the US fleet, the United States has indicated that US vessels used to set on 7.315 dolphins to catch tuna at the time the US dolphin-safe labelling provisions were enacted. The United States explains that there were, at the time, 46 US purse seine vessels along with 52 Mexican vessels that fished for tuna in the ETP⁴⁹⁰, most of which were authorized to set on dolphins to catch tuna. 491 The United States further indicates that US vessels did not fully discontinue the practice until years later, in the mid-1990s. 492
- 7.316 From these undisputed elements, it appears that the US fleet currently does not practice setting on dolphins in the ETP. We note also that two US full time purse seine vessels are currently registered to fish in the ETP for 2010 but neither has sought or obtained a Dolphin Mortality Limit (or DML) under the AIDCP, which implies that they are not allowed to set on dolphins. Mexico has also observed that these two vessels are also registered in the WCPFC, which suggests that they are not operating exclusively in the ETP.
- 7.317 From the above, it can be inferred that, as the practices of the US and Mexican tuna fleets currently stand, most tuna caught by Mexican vessels, being caught in the ETP by setting on dolphins, would not be eligible for inclusion in a dolphin-safe product under the US dolphin-safe labelling provisions. However, most tuna caught by US vessels is potentially eligible for the label, provided that it otherwise complies with the requirements of the measures.
- 7.318 Mexico also contends that US tuna products manufacturers source less than 1 per cent of their supplies of tuna from the ETP and that virtually all tuna used in canned and pouched tuna products is harvested by purse seine vessels. 493 Mexico accordingly concludes that the US measures do not impact the US fleet, or challenge existing commercial arrangements by the large canneries, and makes it much easier for the United States to maintain the measures. 494
- 7.319 However, for the reasons explained below, we are not persuaded that it follows from these facts that the United States affords Mexican tuna products "less favourable treatment" than to tuna products originating in the United States or any other country in respect of the measures at issue.

⁴⁸⁹ United States' comments on Mexico's response to Panel question No. 148, para. 78.

⁴⁸⁸ Mexico's second written submission, para. 40.

^{490 (}footnote original) U.S. Opening Statement at the First Panel Meeting, para. 20; IATTC, 1990 Annual Report, Exhibit US-54. (United States' second written submission, fn 37).

⁴⁹¹ (footnote original) U.S. Opening Statement at the First Panel Meeting, para. 20. (United States' second written submission, fn 38).

^{492 (}footnote original) U.S. first written submission, para. 43. (United States' second written submission, fn 39).

493 Mexico's second written submission, fn 14 and para. 37. Exhibit MEX-72.

⁴⁹⁴ Mexico's comments to United States' response to Panel question No. 92 (b), para. 16.

Relationship between the practices of fleets and access to the label

7.320 It is undisputed that, at the time of enactment of the measures, there were a number of vessels of different fleets fishing in the ETP by setting on dolphins, including a number of US and Mexican vessels. Each vessel operating in the ETP by setting on dolphins was therefore faced with the choice of either continuing to fish in these conditions, and renouncing the benefit of the US dolphin-safe label on the US market, or discontinuing this practice in favour of another fishery or another fishing method in the ETP, and thus having access to the market for dolphin-safe tuna in the United States. In this respect, as discussed earlier, the measures modified the conditions of operation on the market equally for purse seine vessels of all origins.

7.321 In light of the information provided by the parties, it appears that both the US and Mexican fleets were fishing in the ETP at the time of the enactment of the measures and that both were setting on dolphins. Mexico has stated that in 1990, its fleet dominated the purse seine fishery in the ETP, both in terms of carrying capacity and in terms of number of vessels. Mexico had an annual total capacity of 35 to 40 per cent followed by the United States, between 20 and 25 per cent. However, following the adoption of the dolphin-safe policy, the number of major US vessels operating in the ETP decreased. According to the United States, in 1990, US purse seine vessels set on dolphins and on unassociated sets (FADs and free swimming schools) in order to catch tuna in the ETP. Based on NMFS staff knowledge, it suggests that in addition to the United States, Mexico was amongst the countries that had purse seine vessels in the ETP fishing for tuna by setting on dolphins and that foreign vessels also used other techniques such as unassociated purse seine sets, pole and line, troll, longline and gillnet.

7.322 The United States has also explained that in 1990, 49 per cent of its purse seine tuna fleet was inside the ETP and in addition a large portion of the US non-purse seine vessels were outside the ETP. Mexico has not provided exact data on the portion of purse seine and non-purse seine vessels it had operating in the ETP at the time of the enactment of the measure but has stated that since it has developed its tuna fishery in the ETP, almost the entirety of its vessels have operated within that region and that this remains today. In light of the information provided by the United States on the fishing techniques used in the ETP in 1990, it appears that Mexico also had some non-purse seine vessels operating in the ETP, such as baitboats. A baitboat is a vessel that fishes for tuna using hooks and bait and can use among other things pole and line and troll to catch tuna.

7.323 According to Mexico, in 1989, it had a total of 52 purse seiners operating in the ETP and a total of 16 baitboats, while the United States had 53 purse seiners, 8 baitboats and 1 jigboat. In 1990, Mexico had 52 purse seiners and 11 baitboats and the United States had 46 purse seiners and 6 baitboats. In 1991, Mexico had 49 purse seine vessels operating in the ETP and 9 baitboats and the United States had 23 purse seiners and 7 baitboats. From this information we observe that at the time of the enactment of the measure, the majority of the US and Mexican fleets operating in the ETP was composed of purse seiners. To the extent that such vessels were capable of setting on dolphins of them would have similarly had to adapt to the requirements of the US dolphin-safe provisions if they wished to catch tuna eligible for inclusion in a dolphin-safe tuna product on the US market.

⁴⁹⁵ Mexico's response to question 89 (b) from the second substantive meeting, paras. 21-22.

⁴⁹⁶ United States' response to question 89 (b) from the second substantive meeting, para. 7.

⁴⁹⁷ United States' response to question 91 from the second substantive meeting, para. 11.

⁴⁹⁸ Mexico's response to question 91 from the second substantive meeting, para. 24.

⁴⁹⁹ United States' response to question 89 (b) from the second substantive meeting.

⁵⁰⁰ Exhibit MEX-109.

⁵⁰¹ Exhibit MEX-110.

 $^{^{502}}$ The prohibition for purse seiners smaller than 363 metric tons of carrying capacity to set on dolphins is set forth in Annex VIII.6 of the AIDCP

7.324 To summarize, as of 1990 (enactment of the first version of the DPCIA), Mexico and the United States had a comparable number of purse seine vessels operating in the ETP, although it remains unclear what proportion of those purse seiners was setting on dolphins (as opposed to unassociated sets). In addition, the Mexican fleet had a higher number of baitboats (non-purse seine vessels) than the US fleet had. On that basis, it appears that the United States and Mexico were in a comparable position with regard to their fishing practices in the ETP, in that both of them had the majority of their fleet operating in the ETP composed of purse seine vessels potentially setting on dolphins. Both of these fleets had therefore to adapt their fishing methods in order to catch tuna eligible for the US dolphin-safe label. We now consider how the practices of the different fleets subsequently evolved in this respect.

7.325 As Mexico describes its own fishing techniques, it appears that its fleet concentrated on improving the dolphin friendliness of its fishing methods by continuing to set on dolphins, but respecting the conditions of the AIDCP, which entered into force in February 1999 as a successor of the non-binding La Jolla Agreement, with the objective of reducing incidental dolphin mortalities in tuna purse seine fishery rather than by abandoning this method and switching to another fishing method. Mexico explains that its fleet fishes almost exclusively by setting on dolphins in accordance with the AIDCP requirements. The United States challenges Mexico's assessment that its fleet fishes almost exclusively by setting on dolphins, but does not dispute that most Mexican tuna is caught in this manner.

7.326 It is worth noting also that some of the evidence presented to the Panel suggests that at least some Mexican companies also abandoned setting on dolphins and sought to meet the "no setting on dolphins" requirement (and defended the maintenance of the current US standard before the US Congress). 503

⁵⁰³ The United States submitted as evidence a statement which it states was made by the President of Seafood Emporium Inc. to the US Congress, in which it supported the definition of the US dolphin-safe label and expressed opposition to the proposed redefinition of dolphin-safe tuna, which includes the following:

"Passage of S. 1420 and H.R. 2823 will undermine the efforts of Mexican tuna canneries that have gone dolphin-safe by allowing companied and fishing owners in Mexico that do not have dolphin-safe policies to flood the US market with their cheap tuna caught on dolphins. There are currently several tuna companies in Mexico that are considering adopting dolphin-safe policies, but are hesitant due to concern that the current US dolphin-safe definition will be weakened.

By allowing tuna caught by net setting on dolphins to be labeled dolphin-safe, you remove the incentive for those Mexican Flag tuna seiners that are currently operating dolphin-safe to fish without setting nets on dolphins. Why should they make the effort not to set nets on dolphins when their competition can intentionally set nets on dolphins and call their tuna dolphin-safe?

We fully support the environmental leads in Congress, especially Senator Boxer and Biden, and Representatives Studds and Miller, who are totally opposed to weakening the current U.S. definition of dolphin-safe. On behalf of the tuna companies in Mexico that have adopted dolphin-safe policies which prohibit the setting of nets on dolphins, I strongly urge President Clinton and the Members of Congress to oppose S. 1420 and H.R. 2823 and to co-sponsor S. 1460 and H.R. 2856."

United States' response to Panel's question No. 41, para. 105 and Exhibit US-39.

We note that Mexico disputed the reliability of this evidence, stating that it was "an undated letter from a purported trader in Mexican tuna products and that two of the cited companies were not in the tuna business, and the third had gone out of business (See Mexico's second written submission, para. 41 and footnote 23). As

- 7.327 US vessels, on the other hand, gradually discontinued setting on dolphins to catch tuna, and abandoned the practice entirely in 1994, four years after the enactment of the measures. The United States indicated that some of its vessels reflagged and continued to set nets on dolphins to catch tuna, while others began using other techniques to catch tuna either in the ETP or in other oceans, mainly in the Western Central Pacific Ocean. 504 Mexico observed in this respect that the US fleet had already begun transitioning to the Western Pacific. 505 A report published in 1992 by the US National Research Council notes in this respect that "the US tuna fleet in the ETP has become very small, probably because of a variety of factors, probably including the recent decision by major canners to buy only "dolphin-safe" tuna. 506
- 7.328 The information presented to the Panel suggests that other fleets operating in the ETP also adapted in different ways. 507 The United States explains that non-US vessels also either continued to set on dolphins to catch tuna or began to use other techniques to catch tuna either in the ETP or in other oceans but that Mexican vessels remained in the ETP and continued to set on dolphins. 508 It appears that certain fleets fishing in the ETP for tuna have chosen to catch tuna using other techniques and do so in the ETP.
- 7.329 According to the evidence provided by Mexico, in 1989 Ecuador had 34 purse seiners and no baitboats operating in the ETP⁵⁰⁹, in 1990, it had 34 purse seiners and 7 baitboats⁵¹⁰, and in 1991 it had 33 purse seiners and 6 baitboats; Spain had 2 purse seiners. Thus, at the time of the enactment of the measures, the majority of Ecuador's fleet operating in the ETP was composed of purse seiners and given that the AIDCP had not yet entered into force and that therefore there was no prohibition for small purse seine vessels to set on dolphins, arguably all of Ecuadorian purse seine vessels were capable of setting on dolphins. 511 Thus, all of them would have had to adapt to the requirements of the US dolphin-safe labelling provisions. According to the IATTC fourth quarter report for 2009 presented by Mexico, the Ecuadorian fleet represents 39 per cent of the ETP total fleet in terms of number of vessels and 28 per cent in terms of carrying capacity. The Mexican fleet has 23% of carrying capacity, and Venezuela and Panama each had 14%. The ETP fleet is therefore currently dominated by the Ecuadorian and Mexican fleets. However, during the years 1989-1990 the ETP fleet

we understand it, the United States represents that this statement was presented in hearings leading to the congressional passage of the DPCIA, similar to the statement by Representative Boxer, also referred to by the United States (United States response to question No. 39, para. 99) and by the statement. We make no determination relating to the activities of the companies referred to in this statement and their interest in this respect, but we have no reason to assume that the evidence presented by the United States does not reflect a genuine statement to the US Congress as it purports to do.

⁵⁰⁴ United States' response to Panel question No. 18, para. 53.

505 Mexico argues that, in order to fish in that area it is necessary to fish in the EEZ's of the member nations of the Commission, which requires requesting licenses and paying fees. Mexico argues that despite the high cost of the licenses, such licenses are very difficult to obtain from the relevant Island States and that Mexican vessels have tried for 3 consecutive years to obtain a license with no success. Moreover, Mexico observes that it has not been accepted as a Party in the WCPFC where its current status is "cooperating non member". (Mexico's response to Panel question No 38, para. 89). 506 Exhibit MEX-2, p. 3.

⁵⁰⁷ Mexico notes that, of the various fleets operating in the ETP, Belize, Bolivia, Colombia, Guatemala, Honduras, Nicaragua, Panama, Vanuatu, Venezuela, and Peru, are subject to a U.S. embargo on imports of yellowfin tuna, imposed independently of the measures at issue here (Mexico's response to Panel question No. 78, para. 275).

⁵⁰⁸ United States' response to Panel question No. 18, para. 53.

⁵⁰⁹ Exhibit MEX-109.

⁵¹⁰ Exhibit MEX-110.

511 The prohibition for purse seiners smaller than 363 metric tons of carrying capacity to set on dolphins is set forth in Annex VIII.6 of the AIDCP.

512 See Mexico's response to Panel question No. 89(a), para. 18. See also Exhibit MEX-108.

was dominated by the Mexican and the US fleets.⁵¹³ As pointed out by Mexico, the Mexican fleet was the largest with 35-40 per cent of the annual total capacity.⁵¹⁴

7.330 As far as the adaptation to the US dolphin-safe labelling provisions is concerned, Ecuador decided to fish for tuna in the ETP using techniques other than setting on dolphins⁵¹⁵, exporting \$76.4 million of tuna products in airtight containers to the United States in 2009⁵¹⁶, all of which was dolphin-safe⁵¹⁷ and \$48 million of which contained tuna caught in the ETP.⁵¹⁸ For comparison with other years, the United States indicated that in 2006, tuna products in airtight containers from Ecuador totalled \$94.0 million (all dolphin-safe except for 11 importations; in 2007, \$89.1 million dollars (all of which was dolphin-safe) and \$101.9 million dollars in 2008 (all dolphin-safe except for one importation).⁵¹⁹ The United States explained that Ecuador's fleet made the choice in 2010 to catch tuna in the ETP exclusively using techniques other than setting on dolphins, and that for years it had been using those other techniques to catch tuna and exporting those tuna products to the United States that are labelled as dolphin-safe. 520 During the years 1989-1990, Ecuador's fleet represented around 19 per cent of the total ETP fishing fleet in terms of number of vessels, however it did not dominate the ETP fleet as it does today. Ecuador itself explained, as a third party, that its fleet has done very little to fish by setting on dolphins in the ETP, due to market restrictions on tuna caught by using this technique, the high costs of the equipment required, the limited number of schools of tuna associated with dolphins at the moment, and the low DMLs. 521

7.331 These elements suggest that the various fleets operating in the ETP by setting on dolphins adapted in different ways to dolphin-safe concerns, the Mexican fleet having concentrated its efforts on complying with the AIDCP requirements on observer coverage and fishing gear and equipment. As a result, Mexican (and any other) vessels that chose to continue to set on dolphins under the AIDCP requirements were not eligible for dolphin-safe labelling under the existing US measures, while tuna caught without setting on dolphins remained eligible.

7.332 We note, in this respect, that countries party to the AIDCP and the fleets that sought to respond to dolphin-safe concerns by complying with AIDCP requirements may have expected the US dolphin-safe labelling provisions eventually to evolve towards accepting tuna caught in accordance with these requirements as "dolphin-safe". Indeed, this possibility was foreseen in the Panama Declaration and the DPCIA itself. However, as described above, as a result of the rulings in

⁵¹⁴ Mexico's response to Panel question No. 89(b), para, 21.

⁵¹³ See Exhibits MEX-109 and MEX-110.

⁵¹⁵ (footnote original) U.S. Answers to the First Set of Questions from the Panel (Question 17), para. 52; NMFS, Foreign Trade and TTPV databases. In 2010, Ecuador made the choice to fish for tuna in the ETP exclusively using techniques other than setting on dolphins, and has again made this choice for 2011. U.S. Opening Statement at the First Panel Meeting, para. 32; U.S. first written submission, para. 40 &n.40. (United States' second written submission, fn 34).

^{516 (}footnote original) US Imports of Tuna 2005-2009 (Ecuador), Exhibit US-1C. (United States' second written submission, fn 35).
517 United States' response to Panel question No. 17, para. 52 and United States' second written

United States' response to Panel question No. 17, para. 52 and United States' second writter submission, fn 35.

⁵¹⁸ (footnote original) NMFS, Foreign Trade and TTPV databases. (United States' second written submission, fn 36).

⁵¹⁹ (footnote original) Imports of Tuna from Ecuador, Exhibit US-1C; NMFS, Tuna Tracking and Verification Program Databases. (United States' response to Panel question No. 17 fn 49).

⁵²⁰ United States' second written submission, para. 79.

⁵²¹ See Ecuador's response to Panel question No. 2.

Hogarth, this did not happen because the conditions foreseen in the DPCIA for this change to occur were ultimately not fulfilled. 522

7.333 We acknowledge that this circumstance may have had a significant impact on the choice of countries party to the AIDCP to enter into that agreement, and to require their fleets to adhere to its provisions and on the choices made by fleets in adapting to dolphin-safe concerns, to the extent that they had an interest in catching tuna that would be eligible for inclusion in dolphin-safe tuna products for sale on the US market. However, the question before us is whether the US measures at issue placed Mexican tuna products at a disadvantage, compared to US or other imported tuna products. In this respect, we note that the choice facing the fleets of the United States, of Mexico and other foreign origins was the same, and that US and other fleets operating in the ETP could equally have chosen to continue to set on dolphins in the ETP under the conditions set out in the AIDCP, on the assumption that the US measures may evolve towards an acceptance of tuna caught in accordance with these requirements as eligible for inclusion in a dolphin-safe tuna product in the United States. In that respect, the situation arising from the measures was the same for both fleets. Similarly, other tuna fleets operating in international waters face the choice of determining, where to operate and in what conditions, based on a variety of factors, including economic, geographical and environmental factors, and the markets to which they would have access.

7.334 Based on the evidence before us, and taking into account the evolution in the practices of tuna fleets over time, including since the enactment of the US dolphin-safe provisions, as described above, we are not persuaded that any current discrepancy in their relative situations is a result of the measures rather than the result of their own choices. In this respect, we note the observation of the Appellate Body in *Korea – Various Measures on Beef* that where it is the decision of private actors rather than the governmental measure that results in the segregation of imported and domestic like products, this would not be a breach of Article III:4 insofar as what is addressed by this provision is merely the governmental intervention that affects the conditions under which like goods, domestic and imported, compete in the market. Similarly, in the context of Article 2.1 of the TBT Agreement, what must be considered is the treatment arising from the preparation, adoption and

⁵²² See paras. 2.16 to 2.19 above.

The Panel is aware of the fact that from 1994 to 1997, US purse seine vessels were prohibited from setting on dolphins in the ETP (United States' first written submission, fn 34). Indeed, the International Dolphin Conservation Act (IDCA) imposed a general moratorium on the taking or importation of marine mammals or marine mammal products and contemplated an exception for incidental taking of marine mammals in the course of commercial fishing operations conducted under a permit issued by the National Marine Fisheries Service. Such permit would be issued allowing up to a maximum allowable quota of dolphins to be taken annually. It is in the context of a permit that had been accorded to the American Tunaboat Association that the prohibition of setting on dolphins was imposed. (Exhibit US-12) While we note that such prohibition to set on dolphins was imposed only to the US fleet because a determination had been made that the US purse seine fleet had reached the maximum allowable quota mandated by the IDCA (International Dolphin Conservation Act), we observe that the IDCA also imposed a ban on the importation of fish or products from fish caught with commercial fishing technology that resulted in incidental mortality or serious injury to ocean mammals in excess of the United States' standards. In particular, the importation of yellowfin tuna harvested with purse-seine nets in the ETP and products therefrom was prohibited unless the Secretary of Commerce made an affirmative finding for the nation that it complied with the United States' standards. On 10 October 1990, the US Government, pursuant to court order, imposed an embargo on imports of such tuna from Mexico until the Secretary made a positive finding based on documentary evidence that it did not exceed the percentage of admitted dolphin mortality. Therefore, while the United States was prohibited from setting on dolphins, Mexican fleet was barred from importing its tuna products to the United States. For that reason, we believe the US and the Mexican fleet were both on equal footing with respect to the choice they had to make in terms of adapting their fishing techniques to US labelling requirements. (See US – Tuna (Mexico), GATT Report, paras.2.3-2.7) 524 United States cites the Appellate Body Report in Korea – Various Measures Beef, para. 149.

application of the technical regulation by the Member taking the measure, rather than differences in the impact of the measure that are attributable to the behaviour of private actors on the market.

Adaptation costs

- 7.335 Mexico also argues that the Mexican fleet would have to incur considerable financial and other costs by changing its fishing method to be able to label its products as dolphin-safe. In its second written submission, however, Mexico also contended that the nature and extent of such costs that would be incurred by the Mexican tuna fleet to change fishing areas or methods is immaterial to Mexico's *de facto* discrimination claims, given that access to the principal US distribution channels for Mexican tuna products is being denied by the US measures."
- 7.336 Mexico explained that there are regional differences within the ETP in terms of oceanic conditions, the tuna species that are available and the most efficient methods of harvesting. Mexico argues that substantial costs would derive from switching to alternative fishing methods such as FAD fishing. The held that because FAD fishing concentrates on less mature tuna, which are found near the Mexican coast the vessels would have to travel much farther, which would entail very significant additional costs. Mexico explained that its fishing efforts primarily concentrated in the waters adjacent to Mexico's coast (EEZ) where mature yellowfin tuna can be found in association with dolphins. The head of the social property of the
- 7.337 In that area, Mexico argues, the yellowfin tuna not associated with dolphins is found in lesser quantities and at juvenile stage of development and targeting juvenile yellowfin tuna not associated with dolphins would be neither commercially viable nor sustainable because it would result in significantly smaller catches that would have a negative impact on the tuna stocks. Mexico argues that to avoid destroying the yellowfin fishery close to Mexico, the fleet would have to cut production dramatically or move to another region to fish for a different species of tuna, skipjack, which is more commonly found in association with FADs. ⁵²⁸
- 7.338 Moving from the ETP would imply substantial financial costs, Mexico argues, as its vessels would have to travel greater distances (12 to 15 days) before they could even begin to fish. Moreover, vessels would have to focus on skipjack, which is less valuable. Mexico has specified that skipjack, even mature, is smaller than yellowfin tuna, that the market considers it to be of lower quality, and that its value is approximately 30 per cent lower than yellowfin tuna, per ton. Mexico concludes that it would therefore be disadvantaged with respect to other fleets that fish in their adjacent waters. ⁵²⁹
- 7.339 The United States responds that Mexico overstates the cost and difficulty of using other techniques to catch tuna insofar as the same boats and much of the same gear used to set on dolphins to catch tuna may be used to catch tuna using other techniques, specifically sets on floating objects and unassociated schools of tuna. The United States challenges Mexico's arguments that its proximity to the ETP gives it a competitive advantage relative to the United States and other countries in terms of fishing for tuna by setting on dolphins, observing that other countries, including the United States, are similarly close to the ETP, including those areas of the ETP where setting on dolphins to catch tuna occurs.

⁵²⁵ Mexico's response to Panel question No. 38, paras 84.

⁵²⁶ Mexico's response to Panel question No. 38, para. 86.

⁵²⁷ Mexico's response to Panel question No. 38, paras. 84-86.

⁵²⁸ Mexico's response to Panel question No. 38, para. 86.

⁵²⁹ Mexico's response to Panel question No. 38, para. 88.

- 7.340 Mexico also argues that there would be significant administrative complexities because of the need to seek licences from the Pacific island nations to fish in their Exclusive Economic Zones (EEZs) and to comply with the rules of the applicable fishery commission. Mexico refers to the Central and Western Pacific area, regulated by the Western and Central Pacific Commission (WCPFC), which has established regulations that prohibit fishing in the international waters. Mexico adds that in order to fish in that area it is necessary to fish in the EEZ's of the member nations of the Commission, which in turn implies requesting licences and paying fees. 530
- The European Union, as third party, asserts that there are no technical or legal barriers preventing any of the WTO Members bordering the ETP to fish for tuna outside the ETP. Mexico responds that this is not correct because licences would have to be obtained which are expensive and, to date, unobtainable.
- 7.342 We do not exclude that costs of adaptation to a technical regulation may be pertinent to an examination of whether less favourable treatment is being afforded with respect to a technical regulation. However, the existence of adaptation costs, in itself, does not, in our view, imply the existence of less favourable treatment. The existence of additional costs for some operators as a result of factors such as existing practices also does not necessarily, in our view, imply that less favourable treatment is being afforded in respect of the measures at issue, in violation of Article 2.1 of the TBT Agreement.
- 7.343 We first observe that, to the extent that Mexico suggests that there would be ecological costs associated with adapting to the US dolphin-safe provisions, in terms of sustainability of tuna stocks in the ETP, we do not understand the US measures to require or expect any vessel to adopt unsustainable fishing practices in order to comply with the requirements for dolphin-safe labelling.⁵
- 7.344 We recognize that, to the extent that the Mexican fleet would need to modify its fishing techniques, or relocate to other fisheries, in order to comply with the requirements of the US dolphinsafe provisions, this may entail some economic and financial costs, taking into account the fact that setting on dolphins is a particularly effective means of fishing for tuna in the ETP. 532 We also acknowledge that, to the extent that, due to its geographical proximity to an area within the ETP where setting on dolphins can easily be practiced, costs associated with travelling from one area of the ETP to the other to catch tuna by methods other than setting on dolphins and possibly targeting different types of tuna could be greater for Mexican vessels than for those of other fleets whose coasts are not similarly close to those areas. To that extent, adapting to the US dolphin-safe standard could in practice be more onerous for the Mexican fleet than for others who either did not exploit the association with dolphins in the first place or for whom relocating to another fishing area within the ETP or elsewhere implies less additional distance.
- 7.345 However, we are not persuaded that this implies that Mexican tuna products are being denied access to the advantage provided by the US dolphin safe labelling provisions or demonstrates the

531 The United States has noted that it is concerned with the issue of ecosystem protection and seeks to

address it through various means, both domestically and internationally. (United States' response to Panel question No. 64, para. 145).

⁵³⁰ Mexico's response to Panel question No. 38, para. 89.

⁵³² See Exhibit MEX-2, pp. 2 and 3 ("Dolphin fishing usually catches large fish that are often sexually mature and produces larger average catches than the other methods. Thus, redirecting the fishing away from tuna associated with dolphins would be less efficient and would have a negative effect on the yield of the fishery and perhaps on the conservation of tuna populations. Considering only the point of view of economics and harvesting tuna, large tuna should be sought; fishermen should fish on dolphins and be discouraged from fishing on logs or schools. However, dolphin fishing kills dolphins; to minimize the killing of dolphins, all fishing should be directed away from dolphins. This dichotomy is the basis of the tuna-dolphin problem").

existence of "less favourable treatment" being afforded to Mexican tuna products by the US measures. As observed in paragraph 7.376 below, it is possible that a technical regulation, by setting out certain requirements that must be complied with, would affect different operators on the market differently, depending on a range of factors such as their geographical circumstances, their existing practices or their technical capacities. Such factors may have an impact on how easily products of various origins will or will not be able to meet the requirements at issue. However, the existence of such differences does not necessarily imply, in our view, that the measures at issue discriminate against products of certain origins in violation of Article 2.1 of the TBT Agreement. This is especially the case, in our view, where the differential impact of the measures on products of different origins is the result of external factors other than the origin of the products itself.

7.346 In the present case, such costs would be felt also, though perhaps to varying degrees depending *inter alia* on the geographical location and existing practices of the fleets and their commercial choices, by vessels of other fleets wishing to catch tuna eligible for inclusion in a dolphin-safe tuna product for sale on the US market. These consequences would therefore not, in our view, be sufficient to amount to "less favourable treatment" within the meaning of Article 2.1 of the TBT Agreement. In addition, as discussed below, the practices of the national fleet do not necessarily dictate the dolphin-safe status of tuna products of the same origin.

Fishing practices of fleets and origin of tuna products

7.347 We also recall that, as discussed above, tuna products are not necessarily made from tuna of the same origin, so that processors of US, Mexican and other nationalities retain the choice of sourcing their tuna from vessels of any origin fishing for tuna in conditions that meet the requirements of the US measures, even if the Mexican fleet primarily choose not to fish in accordance with these requirements.

7.348 For that reason also, we are not persuaded that it necessarily follows from the fishing practices of the national fleet that the tuna *products* of the same origin are in the same situation. As observed above, Mexico explains that the great majority of Mexican tuna products are made from tuna caught by the Mexican fleet.⁵³³ However, the United States has noted that if tuna products from other countries, including the United States, were made with tuna caught by the Mexican vessels that set on dolphins to catch the tuna, those (non-Mexican) tuna products would also be ineligible to be labelled dolphin-safe. Indeed, the United States has also observed that, in 2009, two thirds of domestic canned tuna was sourced from foreign vessels. It thus concludes that 84 per cent of the US market for canned tuna in 2009 was accounted for by a combination of imported canned and domestic canned tuna that contains imported tuna.⁵³⁴

7.349 Conversely, the United States argues, if Mexican tuna producers choose to purchase tuna from vessels other than large purse seine vessels that set on dolphins to catch the tuna, including large purse seine vessels that use other gear types or fishing methods or small purse seine vessels that are prohibited from setting on dolphins, those (Mexican) tuna products could be eligible to use the dolphin-safe label. The United States observes that nothing requires the Mexican canneries to use tuna caught by Mexican vessels (indeed, Mexico is a net importer of tuna and tuna products), and that Mexican vessels use other methods beyond dolphin setting to catch tuna. ⁵³⁵

7.350 We agree with the United States' observations. The fact that the Mexican tuna fleet fishes in the ETP by setting on dolphins while the US fleet does not, in itself does not imply that tuna

⁵³³ Mexico's response to Panel question No. 144, para. 120.

⁵³⁴ United States response to Panel question No. 97, para. 21.

⁵³⁵ United States' comments to Mexico's response to Panel question No. 144, para. 64.

processors of Mexican and US origin are necessarily similarly affected, as Mexico argues, in such a manner that the relative situation of US and Mexican tuna *products* on the US market is affected. We now consider the situation of Mexican tuna products on the US market.

Mexican tuna products on the US market

7.351 As described above, Mexico contends that it has established that major grocery chains will not carry Mexican brands unless they have the dolphin-safe label and that the United States does not dispute that Mexican tuna products are not sold in major US distribution channels. In support for this assertion, Mexico has presented evidence that major retailers in the United States do not accept for sale tuna that is not eligible for the label, and that they would accept for sale such products if they were eligible for "dolphin-safe" labelling. Safe

7.352 The United States indicates that, although until 2002 Mexican tuna caught using methods other than setting on dolphins by small purse seine vessels in the ETP was incorporated into tuna products sold in the United States, and these products may have been sold in major US grocery retailers as dolphin-safe, there have been no imports of similar Mexican tuna products containing tuna caught by such vessels since that time. The United States points to specific instances of Mexican tuna products sold in specialty stores. The United States explains that Mexican tuna products not labelled as dolphin-safe are sold in speciality grocery stores and on the Internet and that tuna products originating in other countries that are not labelled as dolphin-safe are sold in major distribution channels, such as Walmart's Great Value Tuna, which is a product of Thailand. We note that the United States has also confirmed, however, that it is not aware of any major US grocery retailers that sell tuna products that contain tuna caught by setting on dolphins.

7.353 In the course of the proceedings, the United States presented evidence suggesting that tuna pouches (of the "Great Value" brand) that are not labelled dolphin-safe can be purchased from a major US retailer, Walmart. The United States argues that regardless of the source of the tuna, the fact that the largest US food distributor sells a product not labelled dolphin-safe belied Mexico's assertion that the label is necessary for tuna products to be sold in major distribution channels. Mexico observes however that the US Commerce Department lists Great Value on its "dolphinsafe.gov" website as a dolphin-safe brand, and that in any event Wal-Mart faces no market barrier in convincing itself to carry its own brand. The United States confirmed that the NMFS has verified that the pouches of that brand meet the conditions under the US dolphin-safe labelling provisions. The United States also argues that Mexico misconstrued the information on the US Commerce Department website, and that Great Value tuna products are sold both with and without dolphin-safe labels. In addition, the United States contends that the tuna contained in these Great Value pouches was not sourced from vessels flagged to Mexico. To the extent that the pouches at issue contain tuna eligible for labelling and thus meet the requirements for dolphin-safe labelling, however, we are not persuaded that their

⁵³⁶ Mexico's comments on the United States' response to Panel question No. 102(b), para. 24.

⁵³⁷ An affidavit from an officer of a US company that has the right to sell canned tuna from a Mexican producer in the US market submitted by Mexico states that for years this company attempted to sell the tuna product in major US grocery but the chains have been pressured not to sell this brand of canned tuna because it is not able to carry the "dolphin-safe" label in the U.S. market. It provides evidence that some of the major US chains have expressly indicated that if the tuna product qualified to be labeled "dolphin-safe" they would sell it and that the company's inability to place the "dolphin-safe" label on the cans has directly affected its ability to sell the Mexican brand in the U.S. Exhibits MEX-58, MEX-64 and MEX-100.

⁵³⁸ United States' response to Panel question No. 102 (a), para. 24.

⁵³⁹ United States' oral statement at the second substantive meeting, para. 19.

⁵⁴⁰ United States' response to Panel question No. 102(b), para. 25.

⁵⁴¹ United States' response to Panel question No. 95, para. 19.

⁵⁴² United States' response to Panel question No. 107, paras. 26-28.

availability calls into question Mexico's assertion that major retailers will not carry tuna products that are not eligible for the label.

7.354 More generally, the United States has explained that in 2009, US imports of fresh and frozen tuna totalled \$538 million (or 96 million kilos) and US imports of canned tuna totalled \$613 million (or 180.5 million kilos).⁵⁴³ It has specified that imports of tuna and tuna products from countries with purse seine vessels that fish for tuna in the ETP totalled \$139 million and that imports of tuna and tuna products from Mexico totalled \$13 million in 2009 of which \$7.5 million were tuna in cans, pouches and other air-tight containers.⁵⁴⁴ The United States has also observed that significant amounts of domestic canned tuna contained imported tuna. For example, in 2009, two thirds of domestic canned tuna was sourced from foreign vessels. It concludes that this means that 84 per cent of the US market for canned tuna in 2009 was accounted for by a combination of imported canned and domestic canned tuna that contains imported tuna.⁵⁴⁵

7.355 Mexico does not challenge the United States' import data, nor does it challenge the contention that the majority of imports of tuna products to the US market are not caught by setting on dolphins. However, it argues that it is precisely those facts that show the adverse impact the US measures have on Mexican tuna. Mexico observes that in 2009, US imports of tuna products totalled \$613 million. However, it notes that Mexico's exports have remained minimal throughout this period, and in 2009 US imports from Mexico were only \$7.5 million. Therefore, Mexican tuna products accounted for only 1 per cent of the US market for imported tuna products, and an even smaller percentage of the total US market. 546

7.356 Mexico also does not challenge the assertion that the majority of imported products on the US market are not caught by setting on dolphins but considers that the rise in the import share applies only with respect to tuna caught outside the ETP and/or using a fishing method other than the dolphin set method. For tuna caught in association with dolphins there has not been any significant growth or gains in market share. Mexico also asserts that because a substantial majority of the tuna products allowed to carry the dolphin-safe label in the US market are from tuna sourced from the Western and Central Pacific, there can be no doubt that the US provisions are for non-ETP tuna. S48

7.357 An examination of these import figures, which are not disputed, suggests that the United States imports a considerable proportion of the tuna that it consumes. These figures also suggest that the vast majority of the tuna and tuna products found on the US market is made from tuna caught without setting on dolphins, that is potentially eligible for dolphin-safe labelling under the US dolphin-safe labelling provisions, and that Mexican tuna products represent a very small proportion of the tuna products found on the US market. However, we are not persuaded that these elements, taken together, demonstrate that the measures place Mexican tuna products at a disadvantage, compared to US and other tuna products to the detriment of Mexican tuna products.

7.358 We highlight, in this respect, that the question to be considered in examining a claim of less favourable treatment is not simply whether the measure has some impact, or even some detrimental impact on imports, but, rather, whether the measures put the imported like products at issue at a

⁵⁴⁵ United States response to Panel question No. 97, para. 21.

⁵⁴³ US Imports of Tuna 2009 (all countries), Exhibit US-2.

⁵⁴⁴ United States first written submission, para. 89.

^{546 (}footnote original) NOAA, Fisheries of the United States, Fisheries Statistics and Economics Division, available at http://www.st.nmfs.noaa.gov/pls/webpls/trade_prdct_cntry_ind.results?qtype=IMP&qyearfrom=2000&qyearto=2010&qprod_name=TUNA+%25+ATC+%28%25&qcountry=%25&qsort=COUNTRY&qoutput=PRINTER. Exhibit MEX-88. (Mexico's second written submission, fn 19).

⁵⁴⁷ Mexico's second written submission, para. 38. ⁵⁴⁸ Mexico's second written submission, para. 23.

disadvantage compared to like products of national origin and like products originating in any other country. In other words, what we must determine in this case is whether the measures have modified the relative position on the market of US and Mexican tuna products, to the detriment of Mexican tuna products.

7.359 In this respect, the fact that Mexican imports represent only 1 per cent of tuna on the US market, in itself, only indicates that Mexico has a relatively limited penetration on the US tuna market. In the absence of further information as to what share of the US market Mexico might expect to secure in the absence of the measures at issue, we are not in a position to assess whether Mexico's level of participation in the US tuna market reflects a modification of the conditions of competition to the detriment of Mexican tuna products or whether it simply reflects Mexico's expected level of participation in the US market. 550

7.360 Similarly, that the vast majority of the tuna imported by the United States is caught by methods *other* than setting on dolphins support the view that it is more difficult to sell tuna that is caught by setting on dolphins in the United States. This is consistent with the elements described above relating to the value of access to the label on the market. However, this does not necessarily imply that the US dolphin-safe labelling provisions place Mexican tuna products at a disadvantage on the US market in this respect, as compared to tuna products of US or other origins. Rather, as observed earlier, this constraint would seem to apply equally to all tuna products on the market.

7.361 Furthermore, we note that the decisions by major processors of tuna products not to purchase tuna caught by setting on dolphins in fact predate the adoption of the first version of the DPCIA, enacted 28 November 1990, which first defined dolphin-safe tuna harvested by a vessel using purse seine nets in the ETP as tuna that is not caught on a trip involving intentional deployment on or encirclement of dolphins measures at issue. This suggest that their decision was not necessarily related to the adoption the measures.

7.362 One of the major tuna processors explains its dolphin-safe policy on its website as follows:

"Bumble Bee Foods, LLC remains fully committed to the 100% dolphin-safe policy we implemented in April 1990. ... The US Government Dolphin-Safe Regulations are in the process of being modified through the International Dolphin Conservation Program. The new regulations will allow for a less stringent US Dolphin-Safe Regulation. Despite the new, less stringent 'compliance' criteria, Bumble Bee remains committed to and in compliance with a 'no encirclement' policy. The commitment of Bumble Bee to dolphin-safety will remain unchanged regardless of any changes to the dolphin-safe law. We continue to strictly adhere to our 100% dolphin-safe policy." ⁵⁵¹

7.363 This statement suggests that even if the US measures were amended to allow the labelling of tuna caught in accordance with the terms of the AIDCP (which authorizes setting on dolphins subject to certain conditions and monitoring) certain tuna processors at least would not necessarily modify their purchasing policies. ⁵⁵² Indeed, according to the record of the hearing in the Senate before the

In response to a question from the Panel in this respect, Mexico asserted that its share of the US market should be higher and that latent demand should be taken into account, but it did not elaborate further on the extent of such demand. (See Mexico's response to Panel question No. 94).

⁵⁴⁹ See para. 7.273 above.

⁵⁵¹ Exhibit MEX-47 (1).

⁵⁵² The evidence presented with the Amicus curiae brief and referred to in the paragraphs to which the United States has referred in its answers to questions suggests that the opposition of the tuna processing companies to a change in the dolphin-safe standard is likely to be explained by the pressure that had been

Subcommittee on Oceans and Fisheries of the Committee on Commerce, Science and Transportation discussing the amendment of the DPCIA on 30 April 1996, which was cited in the paragraphs of the *amicus curiae* brief to which the United States has referred to, tuna processing companies which had taken "voluntary actions", despite not taking a formal position on any legislation, had clearly stated their firm support for the existing definition of "dolphin-safe". 553

7.364 To the extent that these companies would maintain their practices on the basis of their perception of consumer preferences, independently of any change in the US standard, the causal relationship between the US measures and the refusal of processors to purchase tuna caught by setting on dolphins is unclear. Indeed, these elements suggest that there is only a marginal relationship between the measures themselves and the practices of tuna processors, and that what these companies consider to be the determining factor in their decision is an absence of setting on dolphins, rather than compliance with the terms of the US measures.

7.365 Mexico argues that access to the principal US distribution channels is being denied to Mexican tuna products by the US measures, and that retailers, as well as consumers, would purchase such tuna products if they were eligible for dolphin-safe labelling.⁵⁵⁴ In support of this assertion, Mexico presents an affidavit, suggesting that retailers would buy Mexican tuna products if they were eligible for a label. This confirms the importance of dolphin-safe status to US retailers selling tuna products. However, the terms of the affidavit do not, in our view, make it clear, as Mexico suggests, that the AIDCP label would be acceptable to the retailers as an alternative dolphin-safe certification. The affidavit and supporting documents refer to eligibility for a "dolphin-safe" label. These statements may therefore be understood to mean that the retailers at issue would be prepared offer the products for sale if they met the conditions for dolphin-safe labelling under the existing US measures. Indeed, the affidavit explains that retailers who initially carried a Mexican brand withdrew it from sale following an intervention from an NGO. Another supporting document presented by Mexico expressly indicates that the retailer at issue is not concerned with legal issues but with consumer acceptance.⁵⁵⁵

7.366 We also note that Mexico itself has indicated that retailers are acutely aware of the dolphin-safe issue and of the risk of facing actions such as boycotts if they carry tuna not eligible for a label. This suggests that retailers are sensitive to the dolphin-safe issue in a manner comparable to that of the processors referred to above, i.e. that they do not wish to carry tuna products containing tuna

exerted since 1989-1990 by Earth Island Institute. According to a press article, official of Earth Island Institute charged that Bumblebee Seafoods had broken its pledge to sell only "dolphin-safe" tuna because, among other things, its parent company in Thailand had bought a cargo without getting documents to show it had been caught using techniques that did not kill dolphins and launched newspaper advertisements the week of 8 December 1990 to urge a consumer boycott of Bumblebee tuna. Exhibit attached to the Amicus curiae brief EX-2, NYT Article "A storm erupts over saving the dolphins". 8 December 1990.

Tuna processing companies listed (Starkist, Bumblebee and Chicken of the Sea) and announced that they would only accept tuna that had been caught without harassing dolphins. These voluntary actions dramatically and immediately transformed the two-decades long controversy over using dolphins to catch yellowfin tuna in the Eastern Tropical Pacific (ETP) ... Members of the U.S. tuna processing industry (canners) have not taken a formal position on any legislation, but StarKist, BumbleBee and Van Camp (Chicken of the Sea), which constitute 100% of the U.S. industry and 25% of the world market, have clearly stated their firm support for the current definition of 'dolphin-safe'." Exhibit attached to the Amicus curiae brief EX-4. (United States' response to Panel question No. 40(a), para. 98 which cites paragraphs 18-20, 25, 62-64 of the amicus curiae brief which in turn refer to exhibits 1, 2, 3 and 4 attached to the amicus submission.).

⁵⁵⁴ Mexico's first written submission, para. 112.

⁵⁵⁵ See Exhibit MEX-58.

⁵⁵⁶ Mexico's first written submission, para. 111.

caught in association with dolphins that may lead to pressure from NGOs and negative perception from consumers. A consideration of this evidence therefore does not modify our conclusion in paragraph 7.364 that the elements before us suggest that there is only a marginal relationship between the US measures and the practices of operators on the market, including retailers.

7.367 This conclusion is also not modified, in our view, by a consideration of the evidence presented by Mexico in relation to consumers' understanding of the term "dolphin-safe". As reflected in paragraph 4.141, Mexico submitted an opinion poll indicating that 48% of the public believes that dolphin-safe means no dolphins were injured or killed in the course of capturing the tuna (AIDCP standard), 22% believes it means there is no dolphin meat in the can, and only 12% believe it means dolphins were not encircled and then released in the capture of the tuna (US standard). According to this poll, 59% of the public think the definition of dolphin-safe should mean that no dolphins were injured or killed in the course of capturing tuna, whereas only 10% believe it should mean dolphins were not encircled and then released in the capture of the tuna. 557 However, the Panel is not persuaded that this supports Mexico's view that retailers and consumers would accept tuna products made of tuna labelled with the AIDCP label, or that those surveyed individuals who assume "dolphin-safe" to mean that no dolphins were injured or harmed would consider that no dolphin was injured or harmed if setting on dolphins was involved. We also note that the poll addresses consumer's perception of the meaning of the term "dolphin-safe", rather than their interest in purchasing one or another type of tuna based on its dolphin-safe status.⁵⁵⁸ The results of this poll would therefore not provide a sufficient basis, in our view, to conclude that consumers would consider tuna products made of tuna caught in conformity with AIDCP requirements to be "dolphin-safe", nor do they provide a basis for concluding that the US dolphin-safe provisions deny Mexican tuna products access to the US market.

7.368 We further note in this respect that some of the evidence presented to the Panel suggests that 90 per cent of the world's tuna companies have adopted a strict "no setting on dolphins" standard. 559 If this is the case, the proportion of tuna imported in the United States that is caught by other methods than setting on dolphins may simply reflect the general distribution of the products on the world market, rather than any specific features of the US market. Mexico observed that such claims were unverifiable insofar as the organization making this claim, whose standard is largely copied on the rules of the United States for the ETP, has no mechanism to authoritatively verify it. Mexico noted that the Earth Island Institute (EII) reportedly requires companies to pay annual or per-can/case fee to be listed as "dolphin-safe" on Earth Island's lists but that it had no mechanism of verification –other than to ask the companies to self-certify compliance. Mexico also emphasized that the United States had not disputed that there was no manner in which it could verify claims that no dolphins were killed or seriously injured during particular net sets outside the ETP because there are not trained, independent observers on board fishing vessels outside the ETP and because no fisheries management organization other than the IATTC requires vessels to keep tuna caught in sets that harmed dolphins in a separate well, and not mix them with tuna caught in sets that did not harm dolphins. Finally Mexico argued that the major three US distributors do not advertise that they comply with the EII standard but rather that they define "dolphin-safe" to mean not setting nets on dolphins. We acknowledge that the EII may not have an independent review mechanism to verify compliance with the EII standard, which is stricter than the US standard. However, compliance with the requirement of

⁵⁵⁷ Exhibit MEX-64.

⁵⁵⁸ The only question in the poll relating to purchasing preferences was expressed in terms of species rather than dolphin-safe status (44/52% of respondents indicated that they look for albacore tuna, 3/2% looked for skipjack and 6% for yellowfin). See Exhibit MEX-64.

Amicus Exhibit-28. As explained in paragraph 2.9 above, insofar as the Panel deemed this information to be relevant for the purposes of its assessment, it invited Mexico to comment on it in order to take full account of Mexico's right of response and defense in respect of due process considerations. Panel's question No. 88.

absence of setting, which is the common denominator of the US and the EII standard, is certified through the US procedures and the NOAA form. The lack of review mechanism alone therefore does not disprove the assertion that 90 per cent of the world's tuna companies catch tuna with methods other than setting on dolphins.

"Pressure" to adapt to the US standard

7.369 Finally, Mexico contends that the US measures unilaterally exert pressure on the Mexican fleet to change fishing areas and/or fishing methods, which Mexico argues is evidence of the *de facto* discriminatory effect of the measures. For Mexico, the fact that the US measures are aimed at encouraging foreign fishing fleets such as those of Mexico to change either their fishing areas or their methods is further evidence of the link between the US measures and the WTO-inconsistent discrimination. In its view, such an attempt at extraterritorial regulation by the United States is inconsistent with the national treatment and most-favoured nation obligations and the obligations in the WTO Agreements must not be interpreted so as to allow a WTO Member to condition access to its domestic market based on compliance with that Member's unilateral policy relating to actions outside its territory, including unincorporated processes and production methods (PPMs). Section 1.563

7.370 Mexico observes that a stated objective of the US measures is to use the US market to encourage fishing fleets such as those of Mexico to change their fishing areas and/or practices. Therefore it flows from that objective, according to Mexico, that the measures *de facto* condition access to the principal US distribution channels for tuna products on compliance with a fishing method that has been unilaterally imposed by the United States. Moreover, Mexico considers that merely because there may be alternative ways to obtain the dolphin-safe label and thereby access to the principal US distribution channels does not alter the fact that Mexican tuna products caught using the established fishing methods of the Mexican fleet are *de facto* treated less favourably. For Mexico, in its view, such an attempt at extraterritorial regulation by the United States is inconsistent with the national treatment and most-favoured nation obligations. Mexico acknowledges that conceivably, in some limited circumstances, a unilateral effort aimed at altering the behaviour of another sovereign country could be justified under one of the specific exceptions to the

⁵⁶⁰ Mexico contends that the certification on Form 370 that nets were not intentionally set on dolphins is not as verifiable for non-ETP vessels as for ETP vessels because certifications of the former are made only by the vessel's captain, while certifications of the latter must be ratified by an independent observer on board the vessel. See para. 4.127.

Mexico's second written submission, para. 151.

⁵⁶² Mexico's second written submission, para. 148.

⁵⁶³ (footnote original) Unincorporated process and production methods (PPMs), which are also referred to as non-product related PPMs, leave no trace in the final product and are not reflected in the physical attributes of the product. The fishing methods at issue in this dispute are an example. They leave no trace in the tuna products nor are they reflected in the physical attributes of the tuna products. (Mexico's second written submission, fn 80).

⁵⁶⁴ Mexico's second written submission, para. 116.

footnote original) In Canada – Wheat Exports and Grain Imports, the Panel found that the fact alternative commercially more attractive distribution channels were open to imports did not alter the fact that the Canadian measure treated imported grain less favourably than like domestic grain. See Panel Reports, Canada – Measures Relating to Exports of Wheat and Treatment of Imported Grain, WT/DS276/R, adopted 27 September 2004, as modified by the Appellate Body Report, WT/DS276/AB/R, paragraphs 6.213 and 6.295. In Canada – Autos, the Panel found that where an advantage accorded to the sale or use of domestic products but not to the sale or use of like imported products, it did not matter that the advantage could also be obtained by means other than the sale or use of domestic products. See Panel Report, Canada – Certain Measures Affecting the Automotive Industry, WT/DS139/R, WT/DS142/R, adopted 19 June 2000, as modified by the Appellate Body Report, WT/DS139/AB/R, WT/DS142/AB/R, paragraph 10.87. (Mexico's second written submission, fin 121).

⁵⁶⁶ Mexico's second written submission, para. 148.

WTO obligations, however it emphasizes that none of those exceptions has been invoked by the United States nor do any of them apply to the US measures. 567

7.371 In considering Mexico's arguments on this point, we recall the Appellate Body's observation, in US - Shrimp, that:

"It is not necessary to assume that requiring from exporting countries compliance with, or adoption of, certain policies (although covered in principle by one or another of the exceptions) prescribed by the importing country, renders a measure *a priori* incapable of justification under Article XX. Such an interpretation renders most, if not all, of the specific exceptions of Article XX inutile, a result abhorrent to the principles of interpretation we are bound to apply."⁵⁶⁸

Although this statement was made in the context of an analysis under Article XX, it implies that the fact that the measures at issue reflect a certain choice of policy or standard is not in itself a reason to assume that a measure is WTO-inconsistent. The same considerations are pertinent, in our view, in the context of Article 2.1 of the TBT Agreement.

- 7.372 We also note that the US dolphin-safe labelling provisions do not require the importing *Member* to comply with any particular fishing method (these measures do not state, for example, that no tuna may be imported if it originates in a country where tuna is caught by setting on dolphins). Rather, it is the products themselves that need to comply with the requirements of the labelling scheme, *if* they wish to benefit from the label and make dolphin-safe claims on the US market.
- 7.373 The United States describes one of the measures' objectives in terms of protecting dolphins by ensuring that the US market is not used to encourage fleets to catch tuna in a manner that adversely affects dolphins. However, we do not consider that in itself constitutes evidence of a discriminatory effect of the measures. As described above, to the extent that the measures at issue provide an incentive for fleets setting on dolphins to discontinue that practice to gain a commercial advantage on the US market, such incentive applies also to the US fleet and other tuna fleets operating in international waters.

Conclusion

- 7.374 In conclusion, based on the evidence presented to us, we are not persuaded that Mexico has demonstrated that the US dolphin-safe provisions afford less favourable treatment to Mexican tuna products within the meaning of Article 2.1 of the TBT Agreement.
- 7.375 That these measures may, through the operation of origin-neutral regulatory categories, have a detrimental impact on certain imports does not, in our view, necessarily imply that the measures afford less favourable treatment to such imported products within the meaning of Article 2.1. We acknowledge, in this respect, that different products of various origins may be affected differently by a measure that lays down certain product characteristics with which compliance is mandatory. However, as observed above, what matters for the purposes of determining whether there is a violation of Article 2.1 is not only the existence of some adverse impact on some imported products, but whether the group of imported products is placed at a disadvantage, in this respect, *compared to* the groups of like domestic and imported products originating in any other country. We also note, in this respect, the following findings of the Appellate Body in the context of Article III:4:

⁵⁶⁷ Mexico's second written submission, paras. 118 and 148.

⁵⁶⁸ Appellate Body Report, *US – Shrimp*, para. 121.

⁵⁶⁹ See the description of the US objectives in Section II.A below.

"[T]he existence of a detrimental effect on a given imported product resulting from a measure does not necessarily imply that this measure accords less favourable treatment to imports if the detrimental effect is explained by factors or circumstances unrelated to the foreign origin of the product." ⁵⁷⁰

7.376 In addition, as observed above with reference to the rulings of the Appellate Body in *Korea – Various Measures on Beef*, in our view, the form of the analysis is the treatment afforded by the measures themselves, rather than the consequences that arise as a result of the actions of private actors on the market.

7.377 In the present case, as discussed above, it appears to us that the measures at issue, in applying the same origin-neutral requirement to all tuna products, do not inherently discriminate on the basis of the origin of the products, and that they also do not make it impossible for Mexican tuna products to comply with this requirement, or, as Mexico puts it, "prohibit" the use of the label for Mexican tuna products.

7.378 Rather, on the basis of the elements presented to us in these proceedings, it appears to us that the impact of the US dolphin-safe provisions on different operators on the market and on tuna products of various origins depends on a number of factors that are not related to the nationality of the product, but to the fishing and purchasing practices, geographical location, relative integration of different segments of production, and economic and marketing choices. In this context, any particular adverse impact felt by Mexican tuna products on the US market is, in our view, primarily the result of "factors or circumstances unrelated to the foreign origin of the product", including the choices made by Mexico's own fishing fleet and canners.

3. Whether the US dolphin-safe labelling provisions are inconsistent with Article 2.2 of the TBT Agreement

7.379 Article 2.2 of the TBT Agreement establishes:

"Members shall ensure that technical regulations are not prepared, adopted or applied with a view to or with the effect of creating unnecessary obstacles to international trade. For this purpose, technical regulations shall not be more trade-restrictive than necessary to fulfil a legitimate objective, taking account of the risks non-fulfilment would create. Such legitimate objectives are, *inter alia:* national security requirements; the prevention of deceptive practices; protection of human health or safety, animal or plant life or health, or the environment. In assessing such risks, relevant elements of consideration are, *inter alia:* available scientific and technical information, related processing technology or intended end-uses of products."

7.380 Mexico submits that the US dolphin-safe provisions are inconsistent with Article 2.2 of the TBT Agreement because they do not fulfil a legitimate objective or, in the alternative, to the extent that the US measures fulfil any objectives, taking into account the risks of non-fulfilment, those objectives could be fulfilled using less trade-restrictive measures.⁵⁷¹

⁵⁷⁰ Appellate Body Report, *Dominican Republic - Import and Sale of Cigarettes*, para. 96.

⁵⁷¹ Mexico's first written submission, paras. 204-25; Mexico's first oral statement, para. 47; Mexico's second written submission, paras. 200-210; Mexico's second oral statement, paras. 85-115.

- 7.381 The United States contends Mexico's arguments should be rejected insofar as it holds the US dolphin-safe labelling provisions fulfil a legitimate objective and are not more trade restrictive than necessary to meet those objectives. 572
- 7.382 In order to address Mexico's claim, we must first clarify how a violation of this provision may be established. We note that Mexico's claim is based on an absence of compliance with the terms of the second sentence of Article 2.2. We note that the parties appear to agree that the second sentence of Article 2.2 gives meaning to the first sentence, but that they have proposed somewhat different approaches to the establishment of a violation of the second sentence of Article 2.2.
- 7.383 In Mexico's view, a technical regulation creates an "unnecessary obstacle to international trade" if its objective is not legitimate *or* if the regulation is more trade-restrictive than necessary to fulfil a legitimate objective taking account of the risks non-fulfilment would create. According to Mexico, two issues must be addressed under this provision: (i) whether the technical regulation fulfils a legitimate objective; and (ii) whether it is not more trade-restrictive than necessary to fulfil such objective taking account of the risks non-fulfilment would create.
- 7.384 In the United States' view, the second sentence of Article 2.2 of the TBT Agreement explains what the first sentence of this provision means.⁵⁷⁵ The United States submits that two elements must be shown for a measure to be considered more trade-restrictive than necessary: (i) the measure must be trade-restrictive; and (ii) the measure must restrict trade more than is necessary to fulfil the measure's legitimate objective.⁵⁷⁶
- 7.385 The Panel first notes that this provision embodies one of the core objectives of the TBT Agreement, namely "that technical regulations ... do not create unnecessary obstacles to international trade". The first sentence of Article 2.2 translates this general objective into a positive obligation by requiring Members to ensure that their technical regulations are not prepared, adopted or applied with a view to, or with the effect of, creating unnecessary obstacles to international trade.
- 7.386 The second sentence of Article 2.2 contains a more detailed obligation, i.e. that technical regulations must not be more trade-restrictive than necessary to fulfil a legitimate objective, taking account of the risks non-fulfilment would create. The Panel also notes that the first and the second sentences of this provision are connected by the expression "[f]or this purpose". In the Panel's view, this expression indicates that complying with the requirements contained in the second sentence of Article 2.2 serves the purpose of ensuring that technical regulations do not create unnecessary obstacles to international trade.
- 7.387 Thus, the Panel agrees with the United States that "the second sentence of Article 2.2 of the TBT Agreement explains what the first sentence of this provision means". In other words, the second sentence of Article 2.2 of the TBT Agreement establishes two requirements that technical regulations must comply with in order not to constitute unnecessary obstacles to international trade. A plain reading of the second sentence of Article 2.2 reveals that these requirements are:

⁵⁷² United States' first written submission, para. 145.

⁵⁷³ Mexico's first written submission, para. 205.

Mexico's first written submission, para. 205.

⁵⁷⁵ United States' first written submission, para. 143.

⁵⁷⁶ United States' first written submission, para. 163.

The fifth preambular paragraph of the TBT Agreement expresses the WTO Members' desire to:

[&]quot;[E]nsure that technical regulations and standards, including packaging, marking and labelling requirements, and procedures for assessment of conformity with technical regulations and standards do not create unnecessary obstacles to international trade".

⁵⁷⁸ United States' first written submission, para. 143.

- Technical regulations must pursue a legitimate objective; and
- They must not be more trade-restrictive than necessary to fulfil that legitimate objective, taking into account the risks non-fulfilment would create.

7.388 Therefore, we agree with Mexico that the analysis of its claims under Article 2.2 of the TBT Agreement may be conducted in two steps.⁵⁷⁹ First, to determine whether the US dolphin-safe provisions fulfil a legitimate objective; and, second, if that is the case, to determine whether those provisions are more trade-restrictive than necessary to fulfil such objective, taking account of the risks non-fulfilment would create.⁵⁸⁰ We also note that the burden rests on Mexico, as the complainant, to demonstrate that the conditions are met, to conclude that a violation of Article 2.2 of the TBT Agreement exists.

7.389 On the basis of this approach, we now consider Mexico's claims under Article 2.2 of the TBT Agreement.

- (a) Whether the US dolphin-safe labelling provisions pursue a legitimate objective
- 7.390 As observed above, the burden of demonstrating that the US dolphin-safe labelling provisions are inconsistent with Article 2.2 of the TBT Agreement, including that they are "more trade restrictive than necessary to fulfil *a legitimate objective*" (emphasis added) rests on Mexico.

7.391 We note in this respect that in the context of its analysis under Article 2.4 of the TBT Agreement, the panel in EC – Sardines expressed the view that "it is incumbent upon the respondent to advance the objectives of its technical regulation which it considers legitimate". We also note, however, that the Appellate Body observed in this respect that the burden of proof was not modified by considerations relating to the fact that it is the respondent that has adopted the technical regulation at issue and thus was in the best position to ascertain its objectives:

"The TBT Agreement acknowledges the right of every WTO Member to establish for itself the objectives of its technical regulations while affording every other Member adequate opportunities to obtain information about these objectives. That said, part of the reason why the Panel concluded that the burden of proof under Article 2.4 is on the respondent is because, in the Panel's view, the complainant cannot "spell out" the "legitimate objectives" of the technical regulation. In addition, the Panel reasoned that the assessment of the appropriateness of a relevant international standard involves considerations which are properly the province of the Member adopting or applying a technical regulation.

In our opinion, these two concerns are not justified."582

7.392 Similarly under Article 2.2 of the TBT Agreement, which also relies on the notion of "legitimate objective" pursued by the measures, the burden is on the complainant, i.e. in this case Mexico, to establish the existence of a violation of this provision, including that the measures are "more trade-restrictive than necessary to fulfil a legitimate objective", and this necessarily involves a determination of what such objective is and its legitimacy within the meaning of Article 2.2. ⁵⁸³ As we

⁵⁷⁹ Mexico's first written submission, para. 205.

⁵⁸⁰ Mexico's first written submission, para. 205.

⁵⁸¹ Panel Report, *EC – Sardines*, paras. 7.120-7.121.

⁵⁸² Appellate Body Report, *EC – Sardines*, paras. 276-277.

⁵⁸³ See Appellate Body Report, EC – Sardines, paras. 282.

understand it, the Appellate Body's observations with respect to the complainant's role in discharging its burden of proof are intended to make clear that a complainant is not relieved of its duty to present a *prima facie* case by the fact that some aspects of the provision at issue relate to the objectives of the measures, which are determined by the Member taking the measure. This does not imply, however, that the objectives of the measures as defined by that Member itself are not relevant to this determination. It only assumes that the complainant is expected to present its claims taking into account the information at its disposal in this respect.

7.393 In the present case, in order to determine whether the US dolphin-safe labelling provisions pursue a legitimate objective, we must identify such objective, and ascertain its "legitimacy" within the meaning of Article 2.2. In accordance with the burden of proof described above, it is appropriate, in our view, to consider in the first instance the objective of the measure as described by Mexico. Nonetheless, in order to make an objective assessment of the matter in accordance with Article 11 of the DSU and ensure that the right of the defending Member to adopt measures in pursuance of certain legitimate objectives in accordance with Article 2.2 is preserved, we must also consider in this respect, the United States' description of its own objectives, in order to clarify as necessary what the objectives of the measure are and the legitimacy of such objectives.

(i) The objectives pursued by the US measures

Arguments of the parties

7.394 Mexico argues that ascertaining the true objectives of a technical regulation rather than just the stated or acknowledged objectives is an important part of the interpretation and application of Article 2.2 of the TBT Agreement. According to Mexico, the objective assessment required by Article 11 of the DSU includes the facts pertaining to the objectives of the measures at issue. In Mexico's view, in ascertaining the objectives of the US measures the Panel should consider the "design, structure and characteristics of the measure". Mexico submits that "the U.S. measures do not protect animal life or health or the environment in the general sense". According to Mexico, the objective of the US measures is much narrower, namely, "to preserve dolphin stocks in the course of tuna fishing operations in the ETP". 585

7.395 The United States first submitted that the objectives of the US dolphin-safe labelling provisions are:

- ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins; and
- contributing to the protection of dolphins, by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins.⁵⁸⁶

7.396 According to the United States, these objectives are reflected both in the name of the Dolphin Protection Consumer Information Act itself as well as the findings of the US Congress specified in that statute. 587

Mexico's first written submission, para. 211; Mexico's response to Panel question No. 64, paras. 213, 218, 220.

⁵⁸⁵ Mexico's first written submission, paras. 207-208.

⁵⁸⁶ See United States' first written submission, para. 146; United States' first oral statement, para. 47.

⁵⁸⁷ United States' second written submission, para. 128.

- 7.397 The United States later described the second objective as ensuring that the US market is not used to encourage fishing fleets to catch tuna "by setting on dolphins". 588
- 7.398 Mexico submits that the way in which the United States has represented the objectives has created a moving target or evolving objectives. Mexico also argues that the United States' representation of the objectives confuses the objectives with the methods necessary to achieve them. As described above, Mexico further submits that the objective of the US measures is narrower than the protection of animal life or health or the environment. See 1919
- 7.399 The United States agrees with Mexico that the architecture and design of the US dolphin-safe labelling provisions as well as the findings of the US Congress are relevant to ascertaining the objectives of the US measures. The United States claims that the architecture, structure and design of the US dolphin-safe provisions also reflect their stated objectives. 592

Analysis by the Panel

7.400 As described above, there is some difference of views among the parties as to what objectives the US dolphin-safe labelling provisions seek to fulfil.

7.401 As described above, the United States itself initially identified two objectives of the US dolphin-safe provisions:

- ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins; and
- contributing to the protection of dolphins, by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins.⁵⁹³

7.402 The United States later expressed the second objective in terms of contributing to the protection of dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna "by setting on dolphins". 594

⁵⁸⁸ See United States' response to Panel question No. 27, para. 66; United States' response to Panel question No 64, paras. 146-147; United States' second written submission, paras. 36, 127-135, 156, 161, 188; United States' second oral statement, paras. 54-55; United States' response to Panel question Nos. 111 and 147, paras. 33 and 80.

⁵⁸⁹ Mexico's response to Panel question No. 64, para. 207.

Mexico's first written submission, para. 208; Mexico's response to Panel question No. 64, para. 214.
 United States' second written submission, para. 129.

For instance, the United States observes that the US dolphin-safe provisions establish conditions under which tuna products may be labelled dolphin-safe based on whether those products contain tuna that was caught in a manner that adversely affects dolphins. In addition, the United States claims that the US dolphin-safe provisions are designed in such a way that labelling tuna products dolphin-safe or with "any other term or symbol that falsely claims or suggests that the tuna contained in the product were harvested using a method of fishing that is not harmful to dolphins" when the conditions for labelling products in that way are not met is a violation of the US law prohibiting deceptive practices, United States' second written submission, para. 129.

^{7.593} See para. 7.394 above.

⁵⁹⁴ See para. 7.395 above.

7.403 Mexico described the objective of the US measures as being "narrower than the protection of animal life or health or the environment", and suggested that the actual objective of these measures is "to preserve dolphin stocks in the course of tuna fishing operations in the ETP". 595

7.404 Mexico then further clarified that its contention is not that the two objectives of the measures. as identified by the United States, "could not be found within the design and structure of the US measures". 596 Mexico rather argues that US dolphin-safe provisions pursue an additional objective, i.e. "protecting dolphins", and that they fail to fulfil this objective. ⁵⁹⁷

7.405 In light of these contentions, the Panel considers it necessary to seek to first clarify the objectives of the US dolphin-safe provisions. As the Appellate Body has recognized in the context of Article XIV of GATS, a panel's analysis is not bound by a Member's characterization of the objectives of its own measures. Such categorization must be made in an independent and objective fashion, based on the evidence in the record.⁵⁹⁸ The same considerations apply, in our view, in the context of Article 2.2 of the TBT Agreement.

7.406 In this task, the Panel's analysis will be guided by the description of the objectives of the measures by both parties, as well as by the structure and design of the US dolphin-safe provisions. We note in this respect that the parties both consider that the design, structure and characteristics of the US dolphin-safe provisions themselves are relevant to a clarification of their objectives. 599 We wish to make clear, however, that at this stage of the analysis our enquiry is limited to the question of what the *objective* or purpose of the measures is, and does not seek to address the separate question of what the measures actually do or do not do in pursuance of this objective.

7.407 We first note that Mexico has not suggested that the two objectives of the measures, as identified by the United States, "could not be found within the design and structure of the US measures". Accordingly, as we understand it, it is not disputed that the US dolphin-safe provisions pursue objectives relating both to consumer information and to dolphin protection, both in relation to the manner in which tuna is caught. We now consider further both aspects.

Consumer information objective

7.408 As described above, the United States has described the first objective of its measures as "ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins".

7.409 In relation to this objective, we first note that the title of the statute codifying the US dolphin-safe provisions is "Dolphin Protection Consumer Information Act". This title supports the contention that this piece of legislation is intended to protect consumers from being deceived by the information that is provided to them.

7.410 In addition, the US dolphin-safe provisions contain several findings of the US Congress that underline their enactment. Subsection 1385(b) of the DPCIA establishes:

⁵⁹⁵ Mexico's first written submission, para. 208.

⁵⁹⁶ Mexico's response to Panel question No. 137, para. 110.

⁵⁹⁷ Mexico stated that "it appears to Mexico that these two objectives [referring to the objectives identified by the United States] encompass a third incidental objective which relates to 'protecting dolphin'", see Mexico's response to Panel question No. 64, para. 214.

⁵⁹⁸ Appellate Body Report, *US – Gambling*, para. 304.
⁵⁹⁹ See Mexico's response to Panel question No. 64, paras. 213, 218-219 and United States' second written submission, para. 129.

"The Congress finds that-

- (1) dolphins and other marine mammals are frequently killed in the course of tuna fishing operations in the eastern tropical Pacific Ocean and high seas driftnet fishing in other parts of the world;
- (2) it is the policy of the United States to support a worldwide ban on high seas driftnet fishing, in part because of the harmful effects that such driftnets have on marine mammals, including dolphins; and
- (3) consumers would like to know if the tuna they purchase is falsely labelled as to the effect of the harvesting of the tuna on dolphins."
- 7.411 The third finding clearly refers to the intention of protecting consumers against deceitful information. The legislator's goal seems to be to create the necessary controls to guarantee that the information displayed by the labels used on tuna products is truthful about the effects on dolphins of the fishing techniques applied to catch the tuna contained in those products. We note that this finding does not make any express reference to any region or fishery in particular, nor does it refer to any specific fishing technique.
- 7.412 Therefore, the structure and design of the US dolphin-safe provisions support the view that one of their objectives is to ensure accurate information to consumers of tuna products concerning the harmful effects on dolphins resulting from fishing methods employed to catch the tuna contained in those products.
- 7.413 Consequently, the Panel accepts the United States' contention that its measures aim, *inter alia*, at "ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins".

Dolphin protection objective

- 7.414 As described above, the United States initially described its second objective as "contributing to the protection of dolphins, by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins" and later referred to it in terms of "contributing to the protection of dolphins, by ensuring that the US market is not used to encourage fishing fleets to catch tuna by setting on dolphins".
- 7.415 Mexico has suggested, however, that the US objective is "narrower than the protection of animal life or health or the environment", and that the actual objective of these measures is "to preserve dolphin stocks in the course of tuna fishing operations in the ETP"⁶⁰⁰ and that they pursue an "additional objective", i.e. "protecting dolphins".⁶⁰¹
- 7.416 In relation to this objective, we first note that the United States' own description of its operation implies an end-and-means relation between two different elements, i.e. ensuring that the US market is not used to encourage certain fishing techniques, *in order to* contribute to the protection of dolphins. In other words, as we understand it, the ulterior objective is contributing to the protection

⁶⁰⁰ Mexico's first written submission, para. 208.

Mexico stated that "it appears to Mexico that these two objectives [referring to the objectives identified by the United States] encompass a third incidental objective which relates to 'protecting dolphin'", see Mexico's response to Panel question No. 64, para. 214.

of dolphins, whereas the means chosen to achieve this objective is to ensure that the US market is not used to encourage certain fishing techniques.

7.417 With respect to the ulterior objective of contributing to the protection of dolphins, the Panel notes that the US dolphin-safe provisions appear under Subchapter II of Title 16, Section 1385 of the USC, entitled Conservation and Protection of Marine Mammals, which gives support to the United States' assertion that the measures at issue are part of its policies relating to the protection of marine mammals, including dolphins.⁶⁰²

7.418 Moreover, the first two findings of the US Congress cited above also reveal a preoccupation on the part of the US legislator with the protection of dolphins and other marine mammals. The US Congress seemed concerned in particular about the harmful effects suffered by these species arising from two sources: tuna fishing operations in the ETP, on one hand; and driftnet fishing in the high seas worldwide, on the other hand. In the Panel's view, these findings suggest the existence of a direct link between the enactment of the US dolphin-safe provisions and the US Congress' desire to protect dolphins and other marine mammals. Therefore, the structure and design of the US dolphin-safe provisions support the United States' argument that its measures are intended to "protect dolphins".

7.419 Mexico has suggested that the actual objective of these measures is "narrower" than the protection of animal life or health or the environment and is in fact "to preserve dolphin stocks in the course of tuna fishing operations in the ETP". We note in this respect that the United States has often referred in the course of the proceedings to its conviction that the dolphin populations in the ETP remain depleted. These references however, do not imply in our view that the US measures aim exclusively to preserve dolphin stocks in the ETP as suggested by Mexico. As we understand them, the United States' allegations that the dolphin stocks in the ETP remain depleted were rather meant to substantiate its contention that setting on dolphins is harmful to dolphins. Furthermore, the text of the US dolphin-safe provisions establishes certification requirements and conditions for access to the dolphin-safe label also with respect to tuna caught outside the ETP. Therefore, the design and structure of the US dolphin-safe provisions themselves support the view that they aim at protecting dolphins generally, in and outside the ETP. Thus we disagree with Mexico's allegation that the objectives of the measures, as reflected in the design and structure of the measures, are limited "to preserve dolphin stocks in the course of tuna fishing operations in the ETP".

7.420 We also note that, in response to a question by the Panel, the United States explained that its objective was not to achieve a specific number or rate of dolphin mortality, but rather generally to reduce the adverse effects of setting on dolphins to catch tuna by ensuring that the US market is not

604 United States first written submission, paras. 9, 47; United States' first oral statement, para. 7; United States' second written submission, paras. 2, 43; United States' second oral statement, para. 10.

We also note that Subsection 1835(h) of the US dolphin-safe provisions conditioned the change in the certification requirements for tuna caught in the ETP by using purse seine nets to certifying that no dolphins were killed or seriously injured, instead of that no purse seine net was used *and* that no dolphins were killed or seriously injured, to the existence of a finding by the Secretary of Commerce that setting on dolphins was not "having a significant adverse impact on any depleted <u>dolphin stock</u>" (emphasis added). However, such determination would have had effects only within the ETP. Thus, in principle, the requirement of certifying that no dolphins were set upon would have remained possible in other fisheries if there was, for instance, a determination that regular and significant association existed in that fishery.

⁶⁰² United States' first written submission, fn 69.

⁶⁰³ Mexico's first written submission, para. 208.

On the States' first written submission, paras. 46-59; United States' response to Panel question No. 65, para. 150; United States' second oral statement, para. 10.

⁶⁰⁶ Subsection 1835(d)(1)(B) and 1835(d)(1)(D).

used to encourage setting on dolphins. The United States added that "[s]eeking to protect dolphins from these adverse effects might also be considered as seeking to conserve dolphin populations". 608

7.421 In this respect, we note that, as explained above, the US description of the manner in which it sought to contribute to protecting dolphins through the measures has varied in the course of the proceedings. Initially, the United States depicted its objective as contributing to the protection of dolphins by "ensuring that the US market is not used to encourage fishing fleets to use fishing techniques that adversely affect dolphins" (emphasis added). 609 On other occasions, the United States defined it as contributing to the protection of dolphins by "ensuring that the US market is not used to encourage the practice of setting on dolphins to catch tuna" (emphasis added). These differences may have a bearing on later parts of our analysis. Hence, we consider it necessary to clarify this aspect, in light of the structure and design of the US dolphin-safe provisions.

7.422 The operative paragraphs of the US dolphin-safe provisions controlling the use of the dolphin-safe label bar its use on tuna that was caught using driftnet fishing in the high seas, regardless of whether any incidental killing or injury of dolphins or any other marine mammal occurred during the fishing trip. 611 As explained above in Section II of this Report, access to the dolphin-safe label is also denied to tuna caught by a large vessel (with 363 metric tons or more carrying capacity)⁶¹² in the ETP by intentionally setting on dolphins or in a fishing trip where dolphins were killed or seriously injured, regardless of the fishing method used. 613 The use of the label is also prohibited for tuna caught outside the ETP if dolphins were intentionally set upon⁶¹⁴; and in cases where an alternative label is used⁶¹⁵, if dolphins were killed or seriously injured, even if no setting on dolphins was involved.616

7.423 Consequently, the use of both driftnets in the high seas as well as the fishing technique known as setting on dolphins disqualify a tuna product from access to the label. It seems that the reason underlying this prohibition is the assumption that these two fishing techniques result in adverse effects on marine mammals including dolphins. 617 The measures also go further than addressing setting on dolphins by requiring that the tuna has not been caught in a set where dolphins were killed or seriously injured, when the tuna was caught using purse seine nets in the ETP. 618 In addition, the United States itself also describes the measures as conditioning access to an alternative dolphin-safe

 608 United States' response to Panel question No. 65(a), para. 151.
 609 United States first written submission, paras. 6-9, 60, 146; United States' response to Panel question No. 64, para. 149; United States' second written submission, para. 1.

⁶¹³ Subsections 1385(d)(1)(C), 1385(d)(2) and 1385(h) of the US dolphin-safe provisions.

⁶¹⁰ United States first written submission, para. 28; United States' response to Panel question Nos. 64 and 65, paras. 146 and 150, respectively; United States second written submission, paras. 7, 36, 133.

⁶¹¹ Subsection 1385(d)(1)(A) of the US dolphin-safe provisions.

⁶¹² See para. 2.12 above.

⁶¹⁴ Subsections 1385(d)(1)(B) of the US dolphin-safe provisions.

⁶¹⁵ As described in Section II.F of this Report, an "alternative label" under the DPCIA is a dolphin-safe label developed by private individuals or companies, that is in compliance with the dolphin-safe labelling requirements, but does not reproduce the official mark developed by the US Secretary of Commerce.

 ⁶¹⁶ Subsections 1385(d)(3)(A) of the US dolphin-safe provisions.
 617 The second finding of the US Congress expressly indicates that "the United States to support a worldwide ban on high seas driftnet fishing, in part because of the harmful effects that such driftnets have on marine mammals, including dolphins" (emphasis added). The United States has also clarified that "setting on dolphins to catch tuna adversely affects dolphins", United States first written submission, paras. 46-57; United States' first oral statement, para. 7; United States' second oral statement, paras. 7-13.

As described in para. 2.22 above, the same requirement would also apply outside the ETP if it were determined that a regular and significant tuna-dolphin association exists.

label on the tuna not being caught in a set where dolphins were killed or seriously injured (in addition to dolphins not having been set upon).

- 7.424 Thus, the structure and design of the US measures suggest that the US dolphin-safe provisions do not seek to discourage only setting on dolphins. They rather seem directed to discouraging, more generally, the use of fishing techniques that have harmful effects on dolphins, as the United States itself initially described it. This is also consistent with the manner in which the US objective relating to consumer information is formulated.
- 7.425 Therefore, for the purposes of our analysis of Mexico's claims under Article 2.2 of the TBT Agreement, we accept the United States' representation of the objectives of the US dolphin-safe provisions as described in paragraph 7.401 above.
- 7.426 Finally, we note that there is a direct correlation between the two objectives identified by the United States. As explained by the United States itself:

"The objectives of the U.S. dolphin-safe labeling provisions are (1) ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins; and (2) to the extent that consumers choose not to purchase tuna without the dolphin-safe label, the U.S. provisions ensure that the U.S. market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins" (emphasis added). 619

- 7.427 To the extent that, as described, the US dolphin-safe provisions operate on the basis of incentives created by consumer choice, achievement of the second objective seems to be dependant in large part on the achievement of the first objective. Only if consumers can and do accurately distinguish, under the measures at issue, tuna caught in conditions that are harmful to dolphins from tuna caught in conditions that are *not* harmful to dolphins, can the use of such harmful fishing techniques be discouraged on the US market through the use of the label.
- (ii) Whether the objectives pursued by the United States are legitimate

Arguments by the parties

7.428 Mexico submits that the US dolphin-safe provisions promote the use of alternative fishing methods that result in substantially higher bycatch and the depletion of ocean sealife. Mexico also argues that the US measures undermine the economic incentive for countries and fishing fleets to participate in the AIDCP, which has been recognized to be a very successful multilateral environmental agreement. 620

7.429 Accordingly, Mexico claims that the US measures "trade off" the protection of the life or health of other animals and the protection of the environment in general against the professed protection of the life or health of dolphins in the geographic confines of the ETP. According to Mexico, this trade-off undermines broader environmental objectives that are enshrined in the AIDCP.

⁶¹⁹ United States first written submission, paras. 146, 158.

⁶²⁰ Mexico cited to the fact that AIDCP was awarded the Margarita Lizárraga Medal by the Director General of the Food and Agriculture Organization (FAO) in November, 2005. The medal is awarded biennially to a person or organization that serves with distinction in the application of the Code of Conduct for Responsible Fisheries (Mexico's first written submission, para. 207). Mexico also cited support for the AIDCP from environmental NGOs (Mexico's first written submission, para. 90).

⁶²¹ Mexico's first written submission, para. 209.

Therefore, Mexico submits, the US dolphin-safe provisions cannot be found to "fulfil a legitimate objective" within the meaning of Article 2.2 of the TBT Agreement. 622

- 7.430 The United States claims that the objectives of the US dolphin-safe provisions fall within two of the objectives expressly listed in Article 2.2 of the TBT Agreement, namely, the "prevention of deceptive practices"; and the "protection of human health or safety, animal or plant life or health, or the environment". 623
- 7.431 Moreover, the United States submits that the preamble of the TBT Agreement provides relevant context for the words "legitimate objective" in Article 2.2. The United States considers that the preamble makes it clear that each Member has the right to decide for itself which legitimate objectives to pursue and to take measures to meet those objectives "at the levels it considers appropriate", including with respect to measures to protect animal life or health or the environment and to prevent deceptive practices. 624
- 7.432 The United States claims that since dolphins are animals, protecting them is a legitimate objective expressly contemplated under Article 2.2. Further, as dolphins comprise part of the environment, protecting dolphins also constitutes protecting the environment. 625
- 7.433 The United States disagrees with Mexico that the US dolphin-safe provisions "trade off" the protection of dolphin population in the ETP and the protection of the environment in general. In the first place, the United States argues that it is neither for Mexico nor for a WTO panel, to decide what policy objectives the United States should pursue. The United States adds that nothing in the TBT Agreement dictates that Members must prioritize one set of policy objectives over another.
- 7.434 Secondly, the United States disagrees with the implication of Mexico's suggestion that the United States does not take measures to protect other marine species or the environment. However, the United States submits, even assuming *arguendo* that such goals conflict, the decision over which to pursue, and to what level, is completely for the United States to make. According to the United States, the US dolphin-safe provisions seek to protect dolphins and they do not need also to protect every other marine species and the environment as a whole to serve a legitimate objective. The objectives of the US dolphin-safe provisions cannot be "illegitimate" simply because other environmental concerns also merit attention.
- 7.435 In addition, the United States argues that the objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna caught in a manner that adversely affects dolphins, also falls within the illustrative list in Article 2.2, which refers to the "prevention of deceptive practices". According to the United States, the US dolphin-safe provisions are intended to prevent deceptive practices by ensuring that tuna products are not falsely or misleadingly labelled dolphin-safe when they are caught using a fishing practice that adversely affects dolphins. 630

⁶²² Mexico's first written submission, para. 210.

⁶²³ United States' first oral statement, para. 47.

⁶²⁴ United States' first written submission, para. 145.

⁶²⁵ United States' first written submission, para. 151.

⁶²⁶ United States' first written submission, para. 148; United States' first oral statement, para. 47.

⁶²⁷ United States' first written submission, para. 149.

⁶²⁸ United States' first written submission, para. 150.

⁶²⁹ United States' first oral statement, para. 47.

⁶³⁰ United States' first written submission, para. 152.

Analysis by the Panel

7.436 Having clarified the objectives pursued by the US dolphin-safe provisions, we must now ascertain whether these objectives are "legitimate" within the meaning of Article 2.2 of the TBT Agreement. As observed by the panel in EC - Sardines, although the elaboration of the objectives of a measure is a prerogative of the Member establishing that measure, "[p]anels are required to determine the legitimacy of those objectives". 631 The Appellate Body supported this conclusion by stating that it shared the view of the Panel this part of the analysis "implies that there must be an examination and a determination on the legitimacy of the objectives of the measure". 632

7.437 Article 2.2 of the TBT Agreement provides a non-exhaustive list of legitimate objectives under this provision. 633 This list includes, as the United States has pointed out, the "prevention of deceptive practices" and the "protection of ... animal or plant life or health, or the environment". We are satisfied that the objectives of the US dolphin-safe provisions, as described in the previous section, fall within the scope of these two categories of legitimate objectives. The objective of preventing consumers of tuna products from being deceived by false dolphin-safe allegations falls within the broader goal of preventing deceptive practices. Similarly, the protection of dolphins may be understood as intended to protect animal life or health or the environment. In this respect, a measure that aims at the protection of animal life or health need not, in our view, be directed exclusively to endangered or depleted species or populations, to be legitimate. Article 2.2 refers to "animal life or health" in general terms, and does not require that such protection be tied to a broader conservation objective. We therefore read these terms as allowing Members to pursue policies that aim at also protecting individual animals or species whose sustainability as a group is not threatened.

7.438 Furthermore, for the reasons explained below, we also find that the US objectives relate to genuine concerns in relation to the protection of the life or health of dolphins and deception of consumers in this respect. The evidence presented by the parties amply demonstrates that dolphins and other marine mammals may be adversely affected by tuna fishing activities. 634 The Panel also notes that certain fishing techniques seem to pose greater risks to dolphins than others. It is undisputed, in particular, that the fishing method known as setting on dolphins may result in a substantial amount of dolphin mortalities and serious injuries, especially when used without applying certain fishing gear and procedures designed to reduce dolphin bycatch. The number of dolphins killed in the ETP before the adoption of the controls established by the AIDCP⁶³⁵, and the ensuing degradation of the dolphins stocks in this area, are illustrative of the potentially devastating consequences that tuna fishing activities may have on dolphins.

7.439 Both parties also share the view that consumers of tuna products in the United States are sensitive to these adverse consequences, and that this concern is reflected in the public's preference for tuna products that are labelled dolphin-safe. 636 Moreover, the parties have also pointed out that consumers have expectations about the harmfulness to dolphins of the fishing methods employed to

⁶³¹ Panel Report, *EC – Sardines*, para. 7.121.

⁶³² Appellate Body Report, *EC – Sardines*, para. 286. ⁶³³ Appellate Body Report, *EC – Sardines*, para. 286.

Appendic Body Report, 2c Surames, para 25.

634 See e.g. Exhibits MEX-2, MEX-3, MEX-4, MEX-5, MEX-67, MEX-84, MEX-93, MEX-97, MEX-98, MEX-99, MEX-105 (containing information relating to harm caused to dolphins resulting from setting on dolphins and other fishing techniques); Exhibits US-4, US-10, US-19, US-20, US-21, US-22, US-24, US-27, US-57, US-62, US-75 (referring to the negative effects on dolphins arising from the practice of setting purse-seine nets on them).

⁶³⁵ Mexico's first written submission, para. 50.

⁶³⁶ Mexico's second written submission, para. 165; United States' response to Panel question No. 40, para. 97; United States' second written submission, para. 61. The parties disagree, however, on what meaning consumers attribute to the terms "dolphin-safe".

catch tuna that is labelled dolphin-safe.⁶³⁷ Thus, there is a possibility that consumers may be deceived about whether the tuna contained in the products they purchase was caught using a method that is harmful to dolphins.

7.440 Moreover, nothing prevents Members from using the incentives created by consumer preferences to encourage or discourage particular behaviours that may have an impact on the protection of animal life or health. Hence, the Panel considers that regulating the information that appears on a label to ensure that consumers may safely exercise their preference is a legitimate mechanism to ensure this purpose. Consequently, we find the objective of contributing to the protection of dolphins by ensuring that the US market is not used to encourage fishing methods that adversely affect dolphins to be legitimate.

7.441 In relation to Mexico's argument that the objectives of the US dolphin-safe provisions conflict with broader environmental objectives, such as protecting other marine species or sealife in general, the Panel considers that it is a well-established principle underlying the provisions of the GATT 1994 and the covered agreements, that Members enjoy the right to determine the legitimate policies they want to pursue. In this regard, the Panel finds the following passage of the Appellate Body's decision in *US – Gasoline* illustrative:

"WTO Members have a large measure of autonomy to determine their own policies on the environment (including its relationship with trade), their environmental objectives and the environmental legislation they enact and implement. So far as concerns the WTO, that autonomy is circumscribed only by the need to respect the requirements of the *General Agreement* and the other covered agreements."

7.442 As established above, the US dolphin-safe provisions aim at protecting dolphins. That this particular piece of legislation does not afford protection to other animals or marine species should not be sufficient reason to consider the goal of the measure to be illegitimate. As the Appellate Body has recognized, "certain complex public health or environmental problems may be tackled only when a comprehensive policy comprising a multiplicity of interacting measures". The United States has indicated that its measures are part of a comprehensive policy to protect dolphins. Moreover, Mexico has never argued that the US dolphin-safe provisions are the only instrument the United States has put in place to protect marine life. It would be highly implausible to expect that a single regulatory instrument should serve all the purposes that may be legitimately pursued by the Members. Therefore, the Panel does not consider that the objective of contributing to the protection dolphins of the US dolphin-safe provisions should be considered illegitimate because it does not cover the protection of other marine species.

7.443 In the Panel's view, the objectives of protecting consumers from deceptive practices and contributing to protecting dolphins by discouraging certain fishing practices do not go against the object and purpose of the TBT Agreement, even in light of the existence of potentially conflicting objectives that could also be recognized as legitimate.

⁶⁴⁰ Appellate Body Report, *Brazil – Retreaded Tyres*, para. 151.

Mexico's first oral statement, para. 49; Mexico's second written submission, para. 177; United States' second written submission, para. 153.

⁶³⁸ Appellate Body Report, *US – Gasoline*, p. 28. See also, Panel Report, *US – Shrimp*, para. 7.26, fn 629; GATT Panel Report, *Japan – Alcoholic Beverages I*, para. 5.13.

⁶³⁹ Appellate Body Report, *US – Gasoline*, p. 29.

⁶⁴¹ United States' first written submission, para. 171; United States' second written submission, para. 157.

- 7.444 For these reasons, we find that the objectives of the US dolphin-safe provisions, as described by the United States and ascertained by the Panel, are legitimate within the meaning of Article 2.2 of the TBT Agreement.
- (b) Whether the US dolphin-safe provisions are more trade-restrictive than necessary to fulfil their objectives, taking into account the risks that non-fulfilment would create
- (i) Approach of the Panel

Arguments by the parties

7.445 As stated above, Mexico claims *in the alternative* that if the US measures are found to fulfil a legitimate objective, they are more trade-restrictive than necessary to fulfil that legitimate objective, taking into account the risks that non-fulfilment would create.⁶⁴²

7.446 Mexico observes that the term "necessary" has been interpreted in the context of Articles XX(b) and XX(d) of the GATT 1994 and Article XIV of the GATS. According to Mexico, these previous interpretations should guide the Panel's analysis of Mexico's claims under Article 2.1 of the TBT Agreement. Based on these previous interpretation of the word "necessary" in the above-mentioned provisions, Mexico argues that for a measure to be "necessary", the following factors must be examined, weighed and balanced: (i) the importance of the interests or values at stake; (ii) the extent of the contribution of the measure to the achievement of the measure's objective; (iii) the trade restrictiveness of the measure; and (iv) whether there are reasonably available alternative measures which may be less trade restrictive while providing an equivalent contribution to the achievement of the objective. 644

7.447 Therefore, Mexico argues that in the context of the facts of this dispute, the phrase "technical regulations shall not be more trade-restrictive than necessary to fulfil a legitimate objective, taking account of the risks non-fulfilment would create" means that the US measures shall not be more trade-restrictive (i.e. deny competitive opportunities to imports of Mexican tuna products and tuna) than necessary (i.e. in light of the importance of the objective of preserving dolphin stocks in the ETP tuna fishery and the contribution of the US measures to the achievement of that objective, there are no reasonably available less trade-restrictive measures that provide an equivalent contribution to the achievement of the objective), taking account of the risks non-fulfilment would create (i.e., in light of the available scientific and technical information, the chance or possibility of adverse consequences should the objective not be carried out).

7.448 The United States considers that a measure that is "more" trade-restrictive than "necessary" is a measure that restricts trade more than is needed or required to fulfil the measure's objective. The United States submits that the Panel should interpret Article 2.2 based on the ordinary meaning of its terms in their context in light of the object and purpose of the TBT Agreement, and look to supplementary means of interpretation to confirm that interpretation. According to the United States, Article 5.6 of the SPS Agreement provides relevant context for Article 2.2 of the TBT Agreement since the two provisions contain very similar language. According to the United States, the contextual reference to Article 5.6 of the SPS Agreement confirms that determining whether a measure is more trade-restrictive than necessary within the meaning of Article 2.2 of the TBT Agreement involves

⁶⁴² Mexico's first written submission, para. 213, Mexico's first oral statement, para. 56.

⁶⁴³ Mexico's first written submission, para. 219.

⁶⁴⁴ Mexico's first written submission, para. 219.

⁶⁴⁵ Mexico's first written submission, para. 222; Mexico's first oral statement, para. 84.

⁶⁴⁶ United States' first written submission, para. 166.

determining whether there is an alternative measure that could fulfil the measure's objective that is *significantly* less trade-restrictive. ⁶⁴⁷

7.449 The United States considers that the word "necessary" in Article 2.2 of the TBT Agreement appears in a very different context than the word "necessary" in Article XX of the GATT 1994. According to the United States, in Article XX(b) the question is whether it is necessary for a Member to breach its GATT obligations to protect human, animal or plant life or health. In Article 2.2, the question is whether an otherwise WTO-consistent measure restricts trade more than is necessary to fulfil the measure's objective. In Article 2.2 of the *TBT Agreement*, unlike under Article XX, it is the complaining party that has the burden of establishing that the measure is "more trade-restrictive than necessary."

7.450 The United States also argues that the analysis under Article 2.2 of the TBT Agreement entails comparing two alternatives that are WTO-consistent, while the alternatives being compared under Article XX of the GATT 1994 are an alternative that is WTO-inconsistent and another that is WTO-consistent. The United States concludes that in light of the different context in which the word "necessary" appears in Article 2.2 as compared to Article XX and the different circumstances surrounding conclusion of those provisions, it would not be appropriate to apply the same meaning or interpretive approach to both provisions. ⁶⁵¹

7.451 Mexico contends that when attempting to interpret the term "necessary", the United States erroneously relies on Article 5.6 of the SPS Agreement and its associated footnote. According to Mexico, the United States uses these provisions and their related jurisprudence to include in the definition of "necessary" in Article 2.2 of the TBT Agreement: (i) the concept that the measure must fulfil a legitimate objective "at the level that the Member imposing the measure has determined appropriate"; and (ii) the qualifying term "significantly" in the phrase "not more trade restrictive".

7.452 Although Mexico acknowledges that Article 5.6 and footnote 3 of the SPS Agreement provide guidance for the interpretation of the phrase "shall not be more trade restrictive than necessary to fulfil a legitimate objective" in Article 2.2 of the TBT Agreement, it observes that in Article 2.2 there is no language similar to that of footnote 3 to Article 5.6 of the SPS Agreement. 654

Analysis by the Panel

7.453 Having determined that the US dolphin-safe labelling provisions pursue legitimate objectives within the meaning of Article 2.2 of the TBT Agreement, we must now determine whether they are "more trade-restrictive than necessary" to fulfil such objectives, "taking account of the risks non-fulfilment would create".

7.454 The terms of Article 2.2 suggest that some restrictions on international trade may arise from the preparation, adoption or application of technical regulations that pursue legitimate objectives

⁶⁴⁷ United States' first written submission, paras. 166-69; United States' first oral statement, para. 157; United States' second written statement, para. 121.

⁶⁴⁸ United States' first oral statement, para. 49.

⁶⁴⁹ United States' response to question 69, para. 155.

⁶⁵⁰ United States' second written submission, para. 124.

United States' first oral statement, para. 49; United States' second written submission, paras. 123-125.

⁶⁵² Mexico's second oral statement, para. 79.

⁶⁵³ Mexico's second oral statement, para. 80.

⁶⁵⁴ Mexico's response to question 67 from the Panel, para. 227; Mexico's second oral statement, para. 82.

under the TBT Agreement. It further suggests that while a degree of "trade-restrictiveness" may be justified, where it is "necessary to fulfil a legitimate objective", a measure could not be justified under Article 2.2 if it is *more* trade restrictive than is necessary to achieve the objective at issue. What we must determine, therefore, is in what circumstances a measure may be considered to be "more trade-restrictive than necessary" under this provision. As observed above, a measure that would be "more trade-restrictive than necessary" within the meaning of the second sentence of Article 2.2 would create "unnecessary obstacles to trade" within the meaning of the first sentence.

7.455 Turning first to the question of what constitutes "trade-restrictiveness" in this context, we note that Mexico argues that measures that are "trade restrictive" include those that impose any form of limitation of imports, discriminate against imports or deny competitive opportunities to imports and that the United States agrees with Mexico that a measure that imposes limits on imports or discriminates against them would meet the definition of a measure that is "trade-restrictive". We also agree.

7.456 We further note that the wording of the provision in terms of the measure "not" being "more" trade restrictive than necessary implies that trade-restrictiveness is only permissible *to the extent that* it is necessary to the achievement of the objective. *A contrario*, where it would be possible to achieve the same objective through a *less* trade restrictive measure, then the measure at issue would be in violation of Article 2.2, because it would then be more trade restrictive than necessary to achieve the said objective.

7.457 We find support for this interpretation in past interpretations of the terms of Article XX of the GATT 1994. Most recently, in *China – Publications and Audiovisual Products*, the Appellate Body summarized the approach it had developed in previous cases under Article XX of the GATT 1994 and under Article XIV of GATS. It thus noted, *inter alia*, that in *Brazil – Retreaded Tyres*, it had described the process in the context of an analysis of "necessity" under Article XX(b) of the GATT 1994:

"The Appellate Body observed that a panel must consider the relevant factors, particularly the importance of the interests or values at stake, the extent of the contribution made by the measure to the achievement of the relevant objective, and the measure's trade restrictiveness. The Appellate Body stated that, if such an analysis 'yields a preliminary conclusion' that a measure is necessary, then the necessity of the measure must be 'confirmed' by comparing the measure with possible alternatives, in the light of the importance of the interests or values at stake."

7.458 Similarly, under Article 2.2 of the TBT Agreement, the analysis involves an assessment of the degree of trade-restrictiveness of the measure at issue in relation to what is "necessary" for the fulfilment of the legitimate objective being pursued, and this can be measured against possible alternative measures that would achieve the same result with a lesser degree of trade-restrictiveness. At the same time, we note that there are differences in the wording of Article 2.2 of the TBT Agreement, as compared to Article XX of the GATT 1994 or Article XIV of the GATS, which reflect also the different positions of the provisions within their respective agreements. In particular, we note that Article 2.2 of the TBT Agreement sets out a positive obligation, and is not formulated as an exception.

⁶⁵⁵ Mexico's first written submission, para. 217.

⁶⁵⁶ United States' first written submission, paras. 164-165.

⁶⁵⁷ Appellate Body Report, *China – Publications and Audiovisual Products*, para. 241.

7.459 With reference to Mexico's observation that the term "necessary" has been interpreted in the context of Articles XX(b) and XX(d) of the GATT 1994 and Article XIV of the GATS and that such interpretation should guide us in our interpretation of Article 2.2 of the TBT Agreement, we note that Article 2.2 of the TBT Agreement refers to technical regulations that are more *trade restrictive* than necessary to fulfil a legitimate objective, whereas Article XX of the GATT 1994 refers to "*measures* necessary" to protect public morals, to protect human, animal or plant life or health, to secure compliance with laws or regulations.

7.460 Thus, under Article 2.2 of the TBT Agreement, unlike in Article XX of the GATT 1994, the aspect of the measure to be justified as "necessary" is its trade restrictiveness rather than the necessity of the measure for the achievement of the objective. Given the fact that, under Article 2.2, the "necessity" to be assessed is that of the "trade-restrictiveness" of the measures rather than of the measures themselves⁶⁵⁸, we understand the term "necessary" in the second sentence of Article 2.2 to mean essentially that the trade-restrictiveness must be "required" for the fulfilment of the objective. At the same time, we note that this question is distinct from that of the level of protection that the Member seeks to achieve in relation to its objective. In this respect, we recall that the preamble of the TBT Agreement makes clear that a Member is entitled to take measures "at the level it consider appropriate", in pursuance of a legitimate objective under the Agreement. This implies, in our view, that an assessment of whether any trade-restrictiveness arising under the measures at issue is "necessary" within the meaning of Article 2.2 must be understood as an enquiry into whether such trade-restrictiveness is required to fulfil the legitimate objectives pursued by the Member *at its chosen level of protection*.

7.461 We find further support for our interpretation of the terms of Article 2.2 in the text of Article 5.6 of the SPS Agreement, which contains language very similar to that of Article 2.2 of the TBT Agreement:

"Without prejudice to paragraph 2 of Article 3, when establishing or maintaining sanitary or phytosanitary measures to achieve the appropriate level of sanitary or phytosanitary protection, Members shall ensure that such measures are not more

658 This is different from the situation under Article XX of GATT 1994. In *US – Gasoline*, the Appellate Body thus determined that "[o]bviously, there had to be a finding that the measure provided "less favourable treatment" under Article III:4 before the Panel examined the "General Exceptions" contained in Article XX. That, however, is a conclusion of law. The chapeau of Article XX makes it clear that it is the 'measures' which are to be examined under XX(g), and not the legal finding of 'less favourable treatment'." Appellate Body Report, *US – Gasoline*, p. 16. We also note the Appellate Body's observations in this respect in *China - Publications and Audiovisual Products*:

"We turn next to China's allegation that the Panel committed a "logical error" in finding that the provisions of its measures are not 'necessary' for essentially the same reasons as the ones for which the Panel found those provisions to be in violation of China's trading rights commitments. China suggests that the restrictive effect of a measure could be relevant to a panel's analysis of whether a measure is consistent with an obligation, or its analysis of whether that measure can be justified under an exception, but that it could not be relevant for both questions. We disagree. The fact that the restrictive effect of a measure is relevant in one context does not preclude that it may also be relevant in the other. In analyzing whether the provisions of China's measures are inconsistent with Article 5.1 of China's Accession Protocol, the Panel assessed *whether* the provisions restrict the enterprises that may engage in importing. Thereafter, in analyzing whether the provisions could be justified as 'necessary' under Article XX(a) of the GATT 1994, the Panel assessed *to what extent* the provisions restrict those wishing to engage in importing, as well as *how* the restrictive effect comports with the degree of contribution to the achievement of the objective, and the societal importance and value of the legitimate objective concerned. The restrictive effect of the provision was relevant to each of these distinct analytical inquiries. Therefore, we do not believe that the Panel's approach constitutes circular reasoning. On the contrary, it is the result of a proper sequential analysis.

trade-restrictive than required to achieve their appropriate level of sanitary or phytosanitary protection, taking into account technical and economic feasibility."

7.462 In addition, footnote 3 to Article 5.6 sets forth:

"For purposes of paragraph 6 of Article 5, a measure is not more trade-restrictive than 'required' unless there is another measure, reasonably available taking into account technical and economic feasibility, that achieves the appropriate level of sanitary or phytosanitary protection and is significantly less restrictive to trade."

7.463 As explained by the Appellate Body, Article 5.6 of the SPS Agreement establishes a three-pronged test of Article 5.6 in which it must be demonstrated, as one of the steps of such test, that the alternative is significantly less trade restrictive than the SPS measure contested.⁶⁵⁹

7.464 We recall that the United States suggested that this provision provides guidance for the interpretation of Article 2.2 of the TBT Agreement. We also note that Mexico recognizes this, although it warns against unduly incorporating into Article 2.2 of the TBT Agreement language that pertains to Article 5.6 of the SPS Agreement and its corresponding footnote. We are duly mindful of the fact that each provision must be interpreted in its proper context and that a similarly worded provision in a distinct covered agreement should not be assumed to have the same meaning as in another context. Nonetheless, we find that footnote 3 of the SPS Agreement, and the clarification that it provides concerning the meaning of the terms "not more trade restrictive than required" is pertinent for the purposes of confirming our understanding of the corresponding terms of Article 2.2 of the TBT Agreement, which play, in the context of this agreement, a comparable role. We note, however, that Article 2.2 of the TBT Agreement makes no reference to a technical regulation being "significantly" more trade restrictive than necessary. Without prejudice to what this term may imply in the context of Article 5.6 of the SPS Agreement, we note that Article 2.2 contains no such qualification.

7.465 In light of the above, we find that in order to determine whether a measure is more trade restrictive than necessary within the meaning of Article 2.2, we must assess the manner in which and the extent to which the measures at issue fulfil their objectives, taking into account Member's chosen level of protection, and compare this with a potential less trade restrictive alternative measure, in order to determine whether such alternative measure would similarly fulfil the objectives pursued by the technical regulation at the Member's chosen level of protection. To the extent that a measure is capable of contributing to its objective, it would be more trade-restrictive than necessary if an alternative measure that is less trade-restrictive is reasonably available, that would achieve the challenged measure's objective at the same level.

7.466 We also note that, in making this determination, we are required to take into account "the risks that non-fulfilment would create". The final sentence of Article 2.2 further clarifies that "[i]n assessing such risks, relevant elements of consideration are, *inter alia*: available scientific and technical information, related processing technology or intended end-used of products".

⁶⁵⁹ The Appellate Body in *Australia – Salmon* agreed with the panel that "Article 5.6 and, in particular, the footnote to this provision, clearly provides a three-pronged test to establish a violation of Article 5.6. As already noted, the three elements of this test under Article 5.6 are that there is an SPS measure which:

⁽¹⁾ is reasonably available taking into account technical and economic feasibility;

⁽²⁾ achieves the Member's appropriate level of sanitary or phytosanitary protection; and

⁽³⁾ is significantly less restrictive to trade than the SPS measure contested." Appellate Body Report, *Australia – Salmon*, para. 194. It consistently followed this approach in *Japan – Agricultural products II* and in *Australia – Apples* (para. 328).

⁶⁶⁰ Mexico's response to Panel question No. 67, para. 227; Mexico's second oral statement, para. 82.

7.467 As we understand it, this part of the text enjoins us to consider, as part of our analysis, both the likelihood and the gravity of potential risks (and any associated adverse consequences) that might arise in the event that the legitimate objective being pursued would not be fulfilled. We further understand this to imply that an alternative means of achieving the objective that would entail greater "risks of non-fulfilment" would not be a valid alternative, even if it were less trade-restrictive. This is consistent, in our view, with the fact that each Member is entitled, as expressed in the preamble of the TBT Agreement and as discussed above, to define its own level of protection.

7.468 Finally, we note that the burden on Mexico to demonstrate that the measures are more trade restrictive than necessary includes the identification of a reasonably available alternative that is capable of achieving the objective pursued by the challenged measure at the same level as the challenged measure, taking into account the risks non-fulfilment would create. As is the case under Article XX of the GATT 1994 and under Article 5.6 of the SPS Agreement, it is, in our view, for the complainant to make a prima facie case that the alternative meets the requirements of the provision at issue. As observed by the panel in *Australia – Apples*, "[t]he Appellate Body explained in *EC – Hormones* that the complainant must establish a prima facie case by presenting "evidence and legal arguments" sufficient to demonstrate that the defendant has breached its obligations with respect to a specific provision. ⁶⁶¹ In the context of Article 5.6 of the SPS Agreement, the Appellate Body confirmed in a subsequent dispute that "[p]ursuant to the rules on burden of proof ... it was for the [complainant] to establish a prima facie case that there is an alternative measure that meets all three elements under Article 5.6 in order to establish a prima facie case of inconsistency with Article 5.6". ⁶⁶² The same considerations similarly apply in the context of Article 2.2 of the TBT Agreement.

7.469 On the basis of the above, we now consider whether Mexico has demonstrated that the US dolphin-safe provisions are more trade restrictive than necessary, taking account of the risks nonfulfilment would create. In light of the fact that, as described above, the US measures pursue two related but distinct objectives, we consider this question from the perspective of each of these two objectives in turn.

- (ii) Whether the US dolphin-safe labelling provisions are more trade-restrictive than necessary to ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught in conditions that are harmful to dolphins
- 7.470 Mexico submits that the US dolphin-safe provisions fail to fulfil their stated objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught by a method that adversely affects dolphins because they allow tuna products to be labelled as dolphin-safe even if they contain tuna from fishing sets in which dolphins were killed or injured, when tuna was caught anywhere outside the ETP. 663
- 7.471 Mexico further claims that it has met the burden of presenting prima facie evidence that there is a reasonably available less trade-restrictive measure.⁶⁶⁴ In Mexico's view, the United States could create dolphin-safe standards rather than a technical regulation whereby the AIDCP standard could be recognized and a label complying with the AIDCP standard used. At the same time, the different US standard could be recognized and a label complying with that standard used. In this way, Mexico

⁶⁶¹ Panel Report, *Australia – Apples*, para. 7.1104 (original footnote omitted).

⁶⁶² Panel Report, Australia – Apples, para. 7.1104 (original footnote omitted).

⁶⁶³ Mexico's first oral statement, para. 48.

⁶⁶⁴ Mexico's response to question 67 from the Panel, paras. 234-36.

argues, US consumers would be fully informed of all aspects of dolphin-safe fishing methods and could choose accordingly when purchasing tuna products from US retailers.⁶⁶⁵

7.472 The United States submits that the US dolphin-safe provisions fulfil the objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins by establishing conditions under which tuna products may be labelled dolphin-safe that are based on whether the tuna product contains tuna that was caught in a manner that adversely affects dolphins. The United States further submits that specifically, tuna products that contain tuna that was caught by setting on dolphins – a technique known to harm dolphins – or in a set in which dolphins were observed killed or seriously injured may not be labelled dolphin-safe or with any other term or symbol that falsely claims or suggests that the tuna products do not contain tuna that was caught in a manner harmful to dolphins. According to the United States, by limiting use of the term dolphin-safe and any other term or symbol that claims or suggests that the tuna products do not contain tuna that was caught in a manner that is harmful to dolphins, to those products that contain tuna that was not caught in a manner that adversely affects dolphins, the U.S. provisions ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins.

7.473 As we understand it, Mexico argues both that the US dolphin-safe provisions as they currently exist and are applied fail to fulfil their objective, and that a less trade-restrictive alternative is reasonably available to the United States to fulfil its objective, by allowing the AIDCP certification to be recognized in addition to the existing US standard. As described above, to the extent that the measures are capable of contributing to their objective, they would be more trade-restrictive than necessary if an alternative measure that is less trade-restrictive is reasonably available, that would achieve the challenged measure's objective at the same level.

7.474 We recall that the United States has defined its consumer protection objective in terms of "ensuring that consumers are not misled or deceived about whether products contain tuna that was caught in a manner that adversely affects dolphins". While the articulation of this objective does not expressly state what the intended "level of protection" to be achieved is, a certain level of protection is embodied in the measure itself, and this is, in our view, what must form the basis for our assessment of the proposed alternative. We must therefore first clarify what the existing US measures achieve, in terms of "ensuring that consumers are not misled" with respect to the manner in which tuna is caught, and then compare that to what Mexico's proposed alternative would achieve in this respect.

7.475 On that basis, we first consider the manner and extent to which the US dolphin-safe labelling provisions fulfil the US objective of ensuring that consumers are not misled or deceived about whether tuna has been caught in a manner that adversely affects dolphins, and then whether, as Mexico claims, this objective could be similarly fulfilled by allowing the AIDCP standard to be applied in addition to the existing US standard.

The contribution of the US dolphin-safe provisions to the US objective of ensuring that consumers are not misled about whether the tuna contained in tuna products was caught in a manner that adversely affects dolphins

7.476 As observed above, Mexico claims that the US dolphin-safe provisions fail to achieve their objective of ensuring that the consumer is not misled about whether the tuna contained in tuna products was caught in a manner that adversely affects dolphins. In Mexico's view, the United States presumes that dolphins are always harmed by dolphin sets in the ETP, even if no dolphins are killed

⁶⁶⁵ Mexico's second written statement, para. 210; Mexico's second oral statement, para. 115.

⁶⁶⁶ United States second written statement, para. 136.

or seriously injured, while it presumes that no dolphins or other marine mammals are harmed outside the ETP. 667

7.477 According to Mexico, most consumers would expect the term "dolphin-safe" to mean that no dolphins were killed or injured in harvesting the tuna. Nevertheless, Mexico submits, the dolphin-safe provisions allow tuna products to be labelled as dolphin-safe even if they contain tuna from fishing sets in which dolphins were killed or injured, when the tuna was caught anywhere outside the ETP. According to Mexico, 48% of the public in the United States believe that the term dolphin-safe means no dolphins were injured or killed in the course of capturing tuna. However, according to Mexico, the US dolphin-safe label does not mean that for tuna caught outside the ETP. Mexico submits that for tuna products made from non-ETP tuna, the US dolphin-safe designation is meaningless and conveys no useful information to consumers. In Mexico's view, for non-ETP tuna products, consumers are not accurately informed about whether dolphins were "adversely affected". Thus, Mexico concludes, the US measures mislead consumers and for this reason cannot be fulfilling their stated objective of protecting consumers.

7.478 As we understand it, Mexico's argument is based on the fact that, under the US dolphin-provisions, no certification that no dolphin was killed or seriously injured in the sets is required with respect to tuna caught outside the ETP by methods other than setting on dolphins, ⁶⁷¹ even though such methods may in fact have resulted in significant harm to dolphins, and that this is misleading to consumers.

7.479 In our view, the objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins can be achieved if the dolphin-safe label allows consumers to accurately distinguish between tuna that was caught by using a fishing method that adversely affects dolphins and tuna that was *not* harvested in such manner, and to make decisions on the basis of accurate information in this respect, when purchasing tuna products. Therefore, the extent to which the US measures are able to fulfil this objective is directly related to the extent to which they accurately identify (and distinguish between) tuna that was caught in a manner that adversely affects dolphins and tuna that was caught in conditions that are not harmful to dolphins.

7.480 We must therefore determine whether the US dolphin-safe provisions grant access to the label to tuna products containing tuna that was caught by methods that may or do in fact adversely affect dolphins. This requires in the first instance a clarification of the meaning of the expression "tuna ... caught in a manner that adversely affects dolphins" in this context. Once this is clarified, we will need to consider the manner in which the measures at issue distinguish between different fishing methods in this respect.

7.481 The Panel notes that the only piece of evidence presented in these proceedings to ascertain what US consumers in fact understand the terms "dolphin-safe" to mean is an opinion poll submitted by Mexico.⁶⁷² This poll shows that 48 per cent of the 800 individuals surveyed believe that "dolphin safe" means that "no dolphins were killed or injured" while 12 per cent believe that it means that "dolphins were not encircled and then released to capture the tuna". The United States considers that

⁶⁶⁷ Mexico's response to Panel question Nos. 86, para. 3 and 136, para. 108.

⁶⁶⁸ Mexico's first oral statement, para. 48-49; Mexico's response to Panel question No. 110, para. 65.

⁶⁶⁹ Mexico's response to Panel question No. 86, para. 4.

Mexico's first oral statement, para. 48; Mexico's second written statement, para. 205; Mexico's second oral statement, para. 98; Mexico's response to question 86 from the Panel, para. 4.

⁶⁷¹ Section 1385 (d) of the AIDCP. See the description of the measures in Section II above.

⁶⁷² Exhibit MEX-64.

the poll includes misleading questions and therefore, does not support Mexico's contentions, because respondents indicating that dolphin-safe means that no dolphins were killed or injured could include respondents who believed that dolphin-safe means that dolphins were not set upon to catch the tuna. 673

7.482 We note, in light of this poll, that it is not clear that US consumers understand the term "dolphin-safe" to mean the same as what the US dolphin-safe provisions define it to mean. We also note that, to the extent that there are discrepancies between the meaning of this term under the measures and consumer perceptions, this may create confusion and undermine the ability of the measure to effectively ensure that consumers are not misled. However, as described above, our enquiry should focus on the manner in which the measures themselves allow or disallow access to a dolphin-safe label, and through this, define the meaning of the term "dolphin-safe" and the extent to which this accurately reflects a distinction between situations in which dolphins are adversely affected and situations in which this is not the case.

Meaning of the phrase "tuna caught in a manner that adversely affects dolphins"

7.483 The US dolphin-safe provisions do not define the notion of fishing methods that "adversely affect" dolphins. As described above, however, these terms have been used by the United States itself to describe the objectives of its measures and we have accepted the US characterization of its objectives. The arguments and evidence presented to the Panel suggest that there are different ways in which dolphins could be "adversely affected" by the manner in which tuna is caught, and the measures themselves also provide some indications as to the type of harmful effects that may arise and that are of concern in this context.

7.484 We first note that the Congress findings cited in paragraph 7.410 above refer to dolphins and other marine mammals being "frequently killed" in the course of fishing operation in the ETP and in other parts of the world. It is clear therefore that the killing of dolphins in the course of fishing operations is understood to constitute a situation in which dolphins are "adversely affected". In addition, in relation to the technique of setting on dolphins, the United States has identified certain negative impacts on dolphins beyond observed deaths and serious injuries. These may be generally described as *unobserved* consequences. The dolphins are "adversely affected".

7.485 We further note that, beyond injuries and deaths of individual dolphins, the evidence presented to the Panel shows that concerns arise or may arise in respect of the conservation of dolphin populations more generally. In this respect, in response to a request for clarification by the Panel as to whether its objective in terms of protection of dolphins could be expressed in terms of a certain level of mortality or whether it should be understood in terms of conservation of dolphin populations, the United States indicated that:

"The objective of the US dolphin-safe labelling provision that aims at protecting dolphins includes protecting dolphins from all of these adverse effects [observed mortalities and other effects of setting on dolphins]. It does not aim to achieve a

⁶⁷⁴ See for example United States' first written submission, paras. 46-51, 54-59, United States' response to the Panel questions Nos. 66, 34, 35, 36, 37, 5(a) and United States' second written submission, paras. 48-49.

⁶⁷³ United States' response to Panel question No. 42, paras. 108-09.

⁶⁷⁵ The Panel understands the United States' use of the terms "observed mortalities and injuries" as referring to dolphin killings or serious injuries that are reported during (or immediately after the conclusion of) dolphin-setting operations. Thus, to the extent that setting on dolphins also result in dolphin deaths or injuries that are *not* observed or taken into account as *observed* killings or serious injuries, the other adverse effects identified by the United States may be described as *unobserved* deaths of injuries of dolphins. Mexico disputes the extent and the consequences of such unobserved consequences. This is discussed in paragraphs 7.494 to 7.505 below.

specific number or rate of dolphin mortality, but rather generally to reduce the adverse effects of setting on dolphins to catch tuna by ensuring that the US market is not used to encourage setting on dolphins to catch tuna. Seeking to protect dolphins from these adverse effects *might also be considered as seeking to conserve dolphin populations*." (emphasis added)⁶⁷⁶

7.486 Therefore, the adverse effects on dolphins targeted by the US dolphin-safe provisions, as described by the United States, relate to observed and unobserved mortalities and serious injuries to individual dolphins in the course of tuna fishing operations. In addition, as described by the United States, to the extent that addressing such adverse effects "might also be considered as seeking to conserve dolphin populations", the US objectives also incorporate, at least indirectly, considerations regarding the conservation of dolphin stocks. Thus, in the Panel's view, the meaning of the terms "adversely affect" dolphins, the contribution of the US dolphin-safe provisions to the fulfilment of its stated objectives needs to be assessed in light of all three of these dimensions of the objectives pursued by the US dolphin-safe provisions.

7.487 With these clarifications in mind, we now consider the manner in which the US dolphin-safe provisions allow or deny access to the dolphin-safe label to tuna products, in consideration of the manner in which the tuna was caught and its potential to adversely affect dolphins.

7.488 As described in Section II:C above, the US dolphin-safe provisions deny access to a dolphin-safe label to tuna caught by setting on dolphins, wherever it is caught. They also deny access to the dolphin-safe label to tuna that was caught by using driftnets in the high seas. In addition, for tuna caught in the ETP by a vessel using a purse seine net with a carrying capacity of more than 362.8 metric tons, the United States requires a certificate that no dolphin was killed during the sets in which the tuna was caught. Outside the ETP, however, the US measures allow access to the label without such certification in all instances where the tuna was harvested by techniques other than setting on dolphins, unless the tuna was caught in a fishery where it has been determined that there is regular and significant tuna-dolphin association or regular and significant dolphin mortality or serious injury of dolphins. Since no such determinations have been made, access to the dolphin-safe label is in practice available to tuna caught outside the ETP by any method other than setting on dolphins or high seas driftnet without a requirement to certify that no dolphin was killed in the sets in which the tuna was caught.

7.489 It appears from these elements that two fishing methods (setting on dolphins and high seas driftnet fishing) are specifically identified in the US dolphin-safe provisions as disqualifying tuna for access to the label, thus reflecting an assumption that these methods "adversely affect" dolphins. For tuna caught by other methods, access to the label is subject to a certification that no dolphin was killed in the sets in which the tuna was caught for tuna caught in the ETP, but not elsewhere. The United States has also explained that if an alternative label rather than the official DOC label is used, this is conditioned on no dolphin having been killed or seriously injured, regardless of the type of fishery. 681

⁶⁷⁹ Subsection 1835(d)(1)(B)(i).

⁶⁷⁶ US response to Panel question No. 66(a), para. 151.

⁶⁷⁷ Subsections 1835(d)(1)(B) and 1835(d)(1)(C).

⁶⁷⁸ Subsection 1835(d)(1)(A).

⁶⁸⁰ Subsection 1835(d)(1)(D).

Subsection 1835(d)(3)(C)(i). See also the United States' own explanation of the operation of this provision, United States' second written submission, para. 40.

7.490 What we must consider, therefore, is the extent to which these distinctions, as contained in the US dolphin-safe labelling provisions and as applied by the United States, allow consumers to accurately distinguish between tuna that was caught in a manner that adversely affects dolphins and other tuna, by ensuring that the label is available exclusively to products containing tuna that was not caught "in a manner that adversely affects dolphins". As described above, we understand adverse effects on dolphins in this context to encompass observed as well as unobserved deaths and injuries, with the understanding that, as described by the United States, this may be considered to also seek to conserve dolphin populations.

Denial of access to the label for tuna caught by setting on dolphins (and high seas driftnet fishing)

7.491 We first consider the two fishing methods in relation to which access to the label is denied, i.e. setting on dolphins and high seas driftnet fishing. Mexico has not made any claim with respect to high seas driftnet fishing.

7.492 With respect to setting on dolphins, the United States argues that it is uncontested by Mexico that this fishing method adversely affects dolphins and that, therefore, by conditioning the labelling of tuna products to not containing tuna caught by setting on dolphins, the US provisions fulfil their objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna caught in a manner harmful to dolphins.⁶⁸²

7.493 We first note that both parties recognize that setting on dolphins may adversely affect dolphins. The number of dolphins killed in the ETP before the adoption of the controls established by the AIDCP, and the ensuing degradation of the dolphins stocks in this area, are well-documented. Indeed, Mexico does not deny that dolphins may be killed or seriously injured during purse-seine net fishing manoeuvres. We also agree with the United States that the existence of the DMLs established by the AIDCP shows that setting on dolphins, even in controlled conditions, may result in some dolphin mortality. In 2008, observed dolphin mortality in the ETP amounted to 1,168 dolphins, whereas in 2009, 1,239 dolphins were observed killed or seriously injured when set upon to catch tuna in the ETP. The parties to the AIDCP agreed to limit total incidental dolphin mortality in the purse-seine tuna fishery in the ETP to no more than five thousand dolphins annually. These limits have remained the same in recent years.

7.494 In addition, as described above, the United States has argued that setting on dolphins has adverse consequences for dolphins beyond observed mortalities or serious injuries, resulting from repeated chase and encirclement of dolphins (referred to in paragraph 7.484 above as "unobserved" consequences). Mexico disagrees with the United States about the magnitude and impact of such unobserved consequences.

⁶⁸² United States' second oral statement, para. 56.

⁶⁸³ Mexico's second written submission, para. 204; United States' first written submission, para. 52.

⁶⁸⁴ Mexico's first written submission, para. 50; Exhibit MEX-4, p. 4; Exhibit MEX-5, pp. 29-30; Exhibit US-61, p. 124.

⁶⁸⁵ Mexico's first oral statement, para. 9.

Although Mexico argues that the number of dolphins being killed annually in the ETP (around 1000) does not have a significant adverse effect on dolphins from a population recovery perspective, it does not deny that setting on dolphins even according to the AIDCP may still result in *observed* dolphin mortality or serious injury, Mexico's second written submission, para. 204.

⁶⁸⁷ Exhibit US-24, p. 50; Exhibit US-66, p. 3.

⁶⁸⁸ Exhibit MEX-11, Article V.

⁶⁸⁹ Exhibit US-50, p. 1.

7.495 The United States notes that dolphins are likely to be repeatedly exposed to the dangers of purse seine dolphin sets, and that millions of dolphins are chased and encircled each year as they are set upon to catch tuna in the ETP. 690 The United States refers to a number of reports and studies, including a report of the scientific research program mandated under the IDCPA which included abundance estimation, ecosystem studies, stress and other fishery effect studies and stock assessment. 691 The stress studies in that report were required to be undertaken by the IDCPA and consisted in four related research projects, a stress literature review, a necropsy study, a review of historical data, and a field study involving the repeated chasing and capturing of dolphins.

7.496 From the review of scientific literature on stress in mammals, the report concluded that tuna purse seine operations involve well-recognized stressors in other wild mammals, and it is plausible that stress resulting from chase and capture could compromise the health of at least some of the dolphins involved. With respect to the stress studies that included a combination of field experiments the report concludes that they support the possibility that purse seine fishing involving dolphins may have a negative impact on the health of some individuals. However, it also noted that several lines of research suggested potential physiological mechanisms of stress effects, but larger sample sizes and baseline data for the affected species are needed to fully interpret the findings. In particular, cow-calf separation and potential muscle injury leading to delayed death warrant future study.

7.497 As far as mother-calves separation is concerned, the study explains that unobserved calf mortality that has been reported and studied from 1973 to 1990 potentially could be large, and continuing at the present time, if mother-calf separation occurs during the chase portion of the fishing operation; however it also specifies that whether, and if so how often, such separations occur during the chase is unknown and that given these uncertainties, only a minimum estimate of unobserved calf mortality is possible, with the caveat that the actual mortality is likely to be larger by an unknown amount.

7.498 The United States has also provided evidence regarding mortality of nursing calves permanently separated from their mothers during fishery operations in the ETP in a scientific article published in 2007. This study finds that because dolphin calves are sometimes incapable of keeping up with their mothers during chase and encirclement, they are sometimes separated from them and vulnerable to starvation or predation. It also explains that despite the fact that observed mortality in the ETP is currently low, dolphin stocks are not recovering at expected rates because of the frequency of sets on dolphins and unobserved mortality of nursing calves. ⁶⁹² Finally, the United States has also presented a scientific article regarding evidence of acute pre-mortem stress in chased and purse-seine captured male dolphins. 693 This article concludes that a 2- to 3-h purse-seine set on dolphins to catch yellowfin tuna, including chase, net deployment, and closed net confinement, generates stressful stimuli.

7.499 These studies therefore suggest that various adverse impacts can arise from setting on dolphins, beyond observed mortalities, including cow-calf separation during the chasing and encirclement, threatening the subsistence of the calf and adding casualties to the number of observed

⁶⁹³ Exhibit US-11.

⁶⁹⁰ The United States observes that on average, each northeastern offshore spotted dolphin is chased 10.6 times and captured 0.7 times per year, and each coastal spotted dolphin chased twice a year. See United States' first written submission, para. 58; United States' second written submission, para. 49.

⁶⁹¹ Exhibit US-19., US-21, US-22; see also United States' first written submission, para. 54, United States' response to Panel question no. 34, paras. 82-83. The United States also submitted studies specifically addressing cow-calf separation (see Exhibits US-4, US-27 and US-28). Exhibit US-4.

mortalities⁶⁹⁴, as well as muscular damage, immune and reproductive systems failures and other adverse health consequences for dolphins, such as continuous acute stress.⁶⁹⁵

7.500 The Panel also notes that other studies question these conclusions. We note that, for instance, it has been suggested that the cow-calf bond remained intact, even after repeated sets. ⁶⁹⁶ It has also been suggested that there are no indications of compromised immune system, capture myopathy or adverse impacts due to heat stress occurring as a result of dolphin chase and encirclement. ⁶⁹⁷ One of the studies submitted by the United States, which in its opinion "represents the best information available at this time" ⁶⁹⁸, concludes that "[w]hile the physiological and behavioral changes [resulting from the chase and encirclement of dolphins] may affect some individuals, they have not been shown to be common enough to have population-level consequences". ⁶⁹⁹ The same study, citing a "summary of recent research" also concludes that "the purse-seine fishery has the capacity to affect dolphins beyond the direct mortality observed as bycatches".

7.501 Mexico presents a 1992 study by the Committee on Reducing Porpoise Mortality from Tuna Fishing, the Board on Environmental Studies and Toxicology, The Commission on Life Sciences and the National Research Council mandated by the MMPA, which found that:

"No specific information is available concerning the effects of the chase on the biology of dolphins. The chase is likely to result in stress. Some herds have developed strategies to avoid capture; others seem to have habituated to encirclement and seem to have developed behavioral patterns that reduce their risks once in the net. Further studies on physiological and behavioral impacts of the chase are obviously needed."

7.502 Mexico also refers to a report prepared by the IATTC in response to the SWFSC report mandated by the IDCPA.⁷⁰² This report concludes that "[t]he SWFSC Chase Encirclement Stress Studies (CHESS) cruise conducted some studies examining various aspects of stress, and the results of most of these studies were either negative (no evidence of significant stress-related injury) or equivocal."⁷⁰³

⁶⁹⁵ Exhibit US-11, pp. 191, 201; Exhibit US-19, p. 5.

The SWFSC Chase Encirclement Stress Studies (CHESS) cruise conducted some studies examining various aspects of stress, and the results of most of these studies were either negative (no evidence of significant stress-related injury) or equivocal. The studies showing no significant adverse impact are listed and described.

Behavioral studies – The study of the behavior of spotted dolphins encircled by purse seines noted that these dolphins are well-habituated to set operations (Santurtún and Galindo 2002).

⁶⁹⁴ Exhibit US-4, p. 24.

⁶⁹⁶ Exhibit MEX-67, p. 9 (containing a literature review on this subject).

⁶⁹⁷ Exhibit MEX-67, p. 9.

⁶⁹⁸ United States' second written submission, fn 70.

⁶⁹⁹ Exhibit US-21, p. 11.

⁷⁰⁰ Exhibit US-21, p. 11.

Tuna Fishing, the Board on Environmental Studies and Toxicology, The Commission on Life Sciences and the National Research Council mandated by the MMPA, (1992) Exhibit MEX-2, p. 114.

⁷⁰² Inter- American Tropical Tuna Commission Scientific Report on the Status of Dolphins Stocks in the Eastern Tropical Pacific Ocean. 2002. Exhibit MEX-67.

⁷⁰³ Inter- American Tropical Tuna Commission Scientific Report on the Status of Dolphins Stocks in the Eastern Tropical Pacific Ocean. 2002. Exhibit MEX-67. This report summarizes its findings in this respect as follows:

[&]quot;Evidence Refuting a Finding of a Significant Adverse Impact

7.503 Mexico further argued that the US position in this proceeding seemed to contradict the US actions in the multilateral forum of the AIDCP, and that the US theories of unobserved mortality presented in this case are scientifically unsupportable and not applied to any other fishery in the United States or the world.⁷⁰⁴

7.504 From the above, it appears that there is a degree of uncertainty in relation to the extent to which setting on dolphins may have an adverse impact on dolphins beyond observed mortality. Nonetheless, we consider that sufficient evidence has been put forward by the United States to raise a presumption that genuine concerns exist in this respect. The information presented to us in this respect also suggests that this is a field of research in which the collection and analysis of information is inherently difficult, but that efforts have been ongoing to better understand these issues, including in the context of the implementation of the DPCIA. We further note that such effects would arise as a result of the chase in itself, and would thus exist even if measures are taken in order to avoid the taking and killing of dolphins in the nets, as is the case under the AIDCP.

7.505 In light of the above, it appears that, to the extent that the US dolphin-safe provisions deny access to the label to products containing tuna caught by setting on dolphins, they enable the US consumer to avoid buying tuna caught in a manner involving the types of observed and unobserved adverse impact on dolphins associated with this method, as described above.

7.506 We also note, however, that, to the extent that the measures make no distinction between setting on dolphins in general and setting on dolphins under the controlled conditions developed in the context of the AIDCP, they would not allow the US consumer to be informed of any of the measures that have been taken in the context of this Programme. We note in this respect that the United States itself acknowledges that the implementation of the AIDCP has made an important contribution to the

This corroborates the findings of the IATTC (1986) and Pryor and Kang-Shallenberger (1991) that indicate that the dolphins have learned behaviors that reduce the risk of entanglement and allow them to anticipate events in the fishing operations.

Heat stress studies – One of the [SWFSC Chase Encirclement Stress Studies] studies (Pabst *et al.* 2002) examined the thermal stress by measuring deep-core temperatures and surface temperatures and heat flux from the dorsal fins of the dolphins (the dorsal fin functions as a radiator to release excess heat). None of these measurements showed any indications that adverse impacts due to heat stress were occurring. The data for one dolphin were puzzling because it displayed the highest deep-body temperature after being involved in the shortest chase (12 min.) of the study. It is most likely that the temperature reading was confounded by the long process (17 min.) required to capture this particular dolphin just prior to sampling". Lymphoid study – A study of the lymphoid organs conducted as part of the SWFSC Necropsy Program (Romano *et al.* 2002) showed no signs of stress or of a compromised immune system-10).

Blood analysis study – Analyses of the blood collected from dolphins captured during the CHESS studies were largely equivocal, being hampered by small sample sizes and few recaptures of tagged and sampled dolphins (St. Aubin 2002). In general, "certain changes were noted signaling a stress response, but none that suggested distress at the time of second capture." The SWFSC report suggests that capture myopathy may be a cause of delayed mortality (p. 68), but the blood analyses during the CHESS cruise reported by Forney *et al.* (2002) indicated that recaptured dolphins had lower values of the muscle-specific enzyme CK, rather than the higher IATTC Report – Oct 2002 10 values that would be encountered if capture myopathy were occurring" (Exhibit MEX-67, pp. 9-10).

⁷⁰⁴ Mexico's responses to the Panel's questions from the Second Substantive Meeting, paras. 55-56.

The state of the recognized that the "difficult logistics of studying pelagic dolphins have precluded detailed observations of [dolphin behaviour] during and after tuna purse-seine sets", Exhibit US-4, p. 20.

One study indicates for instance that "a 2- to 3-h purse-seine set on dolphins to catch yellowfin tuna, including chase, net deployment, and closed net confinement, generates stressful stimuli", Exhibit US-11, p. 191.

reduction in numbers of observed dolphin killings in the ETP. Moreover, the Panel recalls that the United States has acknowledged that "[s]ince the AIDCP's conclusion in 1998 and entry into force in 1999, all parties including Mexico have generally been abiding by their obligations under the AIDCP". The AIDCP ". The Panel recalls that the United States has acknowledged that "[s]ince the AIDCP's conclusion in 1998 and entry into force in 1999, all parties including Mexico have generally been abiding by their obligations under the AIDCP".

Conditions of access to the label for tuna caught by other methods

7.507 As described above, the US dolphin-safe labelling provisions contain further distinctions, for the granting or denial of the dolphin-safe label. We must also consider, therefore, the extent to which the requirements contained in the US dolphin-safe provisions and applied by the United States to tuna caught by methods other than setting on dolphins (and high seas driftnet fishing) allow consumers to accurately distinguish between tuna that was caught in a manner that adversely affects dolphins and other tuna, by ensuring that the label is available exclusively to products containing tuna that was not caught "in a manner that adversely affects dolphins".

7.508 As explained in paragraph 7.488 above, access to the label is available for tuna caught by methods other than setting on dolphins outside the ETP without certifying that no dolphins were killed or seriously injured, while tuna caught inside the ETP by the same methods requires a certificate that no dolphin was killed or seriously injured in the sets. Mexico observes in this respect that the US Form 370 (Fisheries Certificate of Origin) does not require any certification at all with respect to harm to dolphins in the capture of tuna outside the ETP⁷⁰⁹, and that even if it did, there is no system to verify certifications. To the capture of tuna outside the ETP⁷⁰⁹, and that even if it did, there is no system to verify certifications.

7.509 Mexico claims that there are substantial dolphin mortality rates associated with fishing techniques other than setting on dolphins⁷¹¹, and that it is well established that dolphins are regularly killed in gillnets, both in US waters and elsewhere.⁷¹² Mexico also argues that scientific research indicates that there are associations of tuna with dolphins outside the ETP and that a number of cetaceans species are affected in different fisheries and by different methods.⁷¹³

7.510 According to Mexico, notwithstanding the evidence that dolphins and other marine mammals are being killed in significant numbers in ocean regions other than the ETP, no measures have been taken in those other regions comparable to those taken for the ETP⁷¹⁴, and the United States tolerates such harm, even when the absolute numbers of marine mammals being killed are comparable to the ETP and when the ratios of marine mammals being harmed are higher in relation to their populations than in the ETP.⁷¹⁵ Thus, according to Mexico, the United States applies a standard for sustainability of marine mammal stocks to its own fisheries that tolerates level of mortalities, as a percentage of the potential biological removal (PBR), equivalent to or greater than those of dolphins in the ETP.⁷¹⁶The

⁷⁰⁷ The United States has accepted that the AIDCP "has made an important contribution to dolphin conservation in the ETP, including that it has fostered the continued reduction in *observed* dolphin mortalities in the fishery" (emphasis in the original), United States' first written submission, para. 9; United States' second written submission, para. 141.

⁷⁰⁸ United States' second written submission, fn 187.

Mexico's response to Panel question No. 103, paras. 47-48.

⁷¹⁰ Mexico's first oral statement, para. 54.

⁷¹¹ Mexico's first oral statement, para. 50.

⁷¹² Mexico's first oral statement, para. 52.

⁷¹³ Mexico's first written submission, paras. 11-18; Mexico's second written submission, paras. 98-110; Mexico's second oral statement, paras. 99-103.

⁷¹⁴ Mexico's first written submission, para. 19.

⁷¹⁵ Mexico's second oral statement, para. 105.

Mexico's response to Panel question No. 15, para. 31. Mexico has clarified that the PBR "is intended to be an estimate of the maximum level of takings that may be removed from a stock without affecting

United States considers, however, that Mexico's assertions are both factually incorrect and legally irrelevant, *inter alia* because the MMPA and the DPCIA are different measures with different objectives and, in the US view, comparisons between the tuna purse seine fisheries in the ETP and other fisheries that are not tuna fisheries are inappropriate.⁷¹⁷

- 7.511 The United States argues that the hypothetical situation of a tuna caught in a set where dolphins have been killed entering the US market with a dolphin-safe label might exist only in the limited scenario where three factors coincide, namely: (1) the tuna are caught in a fishery where there is no regular or significant association between tuna and dolphins and no regular or significant dolphin mortality; (2) a dolphin is accidentally killed when a purse seine net is accidentally set on a dolphin; and (3) the official dolphin-safe label is used. The United States claims that "the likelihood of any such products being on the US market is low" and that Mexico has presented no evidence that such hypothetical actually exists.⁷¹⁸
- 7.512 The United States also argues that dolphin mortalities in the ETP are fundamentally different from dolphin mortalities that may occasionally occur in other fisheries because, in the ETP, dolphins are intentionally exploited commercially and on a wide scale to catch tuna, and this results not only in observed mortalities but also in other unobserved effects, as described above. In the US view, outside the ETP there is no tuna-dolphin association similar to that in the ETP, and absent repeated intentional chase and encirclement of dolphins to catch tuna, there is no basis to infer that there are significant unobserved dolphin mortalities outside the ETP, much less so that such mortality would be anywhere close to the level of unobserved mortality in the ETP. The United States considers that observer programs exist in other fisheries, that are capable of detecting whether there is any indication of regular or significant mortality or regular and significant tuna-dolphin association, and that these programs have not reported any such indications. In the US view, the sources cited by Mexico rely on anecdotal information, from earlier reports not subsequently substantiated, are based on reporting of non-tuna fisheries, or are otherwise misrepresented.
- 7.513 Mexico refers to information from the websites of the other major tuna fisheries to show that they do not have observer programmes comparable to the ETP. It observes that the Indian Ocean Tuna Commission (IOTC) has no meaningful data on bycatch and no observer program. It also refers to the International Commission for the Conservation of Atlantic Tunas (ICCAT), which indicated it has a goal of achieving five percentage observer coverage on large longline vessels fishing for bigeye tuna and to establish observer coverage for large vessels fishing for bigeye in one small subregion. Mexico states that ICCAT reports include 26 species of dolphins and whales captured by fishing vessels in the Atlantic and Mediterranean, of which 15 were caught in purse seine nets. Mexico also noted that the Western and Central Pacific Fisheries Commission (WCPFC) has limited observer coverage and has not made data publicly available even projections on the overall interaction of

the ability of the stock to reach or maintain its optimum sustainable population". Mexico added that PBRs are developed for individual fisheries and stocks, Mexico's first written submission, fn 64.

United States' second written submission, paras. 143-146; United States' second oral statement, para. 56.

⁷¹⁷ See United States' second written submission, para. 53-58.

para. 56.

The second written submission, paras. 48-52; United States' response to Panel question No. 15, paras. 43-50; United States' second oral statement, paras. 14-18; United States comments on Mexico's response to Panel question Nos. 86 and 100, paras. 9 and 25; United States response to Panel question No. 37, para. 92.

⁷²⁰ United States' comments on Mexico's response to Panel question No. 86, para. 3.

⁷²¹ United States' comments on Mexico's response to Panel question No. 86, para. 9.

United States' second written submission, paras. 50-51; 56-57; United States second oral statement, paras. 15-16, United States' response to Panel question No. 108, para. 31, United States' comments on Mexico's response to Panel question No. 136, para. 57 and fn 83.

this fishery with marine mammals, or attempted to estimate the size of dolphin stocks but that nonetheless, observers witnessed intentional setting of nets on dolphin and whales, indicating that there is an association between tuna and marine mammals in at least some parts of the Western and Central Pacific.

7.514 Thus, according to Mexico, these other regional fisheries management organizations are still in the early stages of beginning to monitor the impact of fishing operations on marine mammals and evidence of dolphin mortalities in these other fisheries already presented by Mexico cannot be refuted by citing to inaction by those fisheries management organizations. Hexico adds that the United States does not make the same projections of multiplier effects on populations that it applies to the ETP when evaluating observed marine mammal mortalities in other fisheries, even though dolphins killed in those fisheries are separated from their calves, and escaping dolphins are likely injured by destructive fishing techniques such as FAD and longline fishing. For Mexico, there is no reason to believe that the indirect ("unobserved") effects on dolphins outside the ETP are less than those within the ETP because the evidence indicates that dolphin mortalities – both "observed" and "unobserved" – in these other ocean regions are likely higher than in the ETP, as in those other regions there is no regulation of fishing designed to protect dolphins and other marine mammals, and no one is even monitoring the impact of fishing on marine mammals.

7.515 In light of these arguments, we now consider the manner in which the requirements of the US dolphin-safe provisions operate in respect of tuna caught by methods other than setting on dolphins, and the extent to which they allow the consumer to accurately distinguish, based on the label, between tuna caught in a manner harmful to dolphins and other tuna. This question, in our view, turns in the first instance on the conditions of eligibility for the label, rather than on the "likelihood" of the situation arising on the market.

7.516 As we understand it, there are two related aspects to Mexico' argument: first, if access to the label *is* available to tuna caught in conditions where dolphins may in fact have been harmed, this would be misleading to the consumer, in that it would lead it to believe incorrectly that such tuna was caught in conditions that are not harmful to dolphins; secondly, to the extent that some tuna caught in conditions that are equally harmful to dolphins are *denied* access to the label, this is also misleading in that the consumer would not be in a position to accurately identify these products as equally harmful or not harmful to dolphins.

Whether fishing techniques other than setting on dolphins may adversely affect them

7.517 In order to address this argument, we must first consider whether, as Mexico argues, fishing for tuna by methods *other* than setting on dolphins outside the ETP may adversely affect dolphins, such that access to the US label is available to tuna that has in fact been caught in a manner that adversely affects dolphins.

7.518 We note at the outset that, as several of the studies submitted by the parties observed, information is lacking to evaluate the existence and extent of the threats faced by different species of dolphins in different areas around the globe, especially outside the ETP.⁷²⁶ A report commissioned by

⁷²³ Mexico's second written submission, paras. 86–110.

See Mexico's responses to the panel's questions from the first substantive meeting, para. 26; Mexico's second written submission, para. 69.

⁷²⁵ See Mexico's second written submission, para. 104.

⁷²⁶ E.g. Exhibit MEX-2 (suggesting more research "on behaviour of tuna and dolphins" in the ETP); Exhibit MEX-5, p. vii (recommending research globally); Exhibit US-10, p. 38 (stating that "data collection by

the US National Oceanic and Atmospheric Administration (NOAA) Fisheries cautions in this respect that the lack of evidence that dolphins are being affected by fishery-related activities in certain fisheries should not be wrongfully taken as evidence of the absence of a problem:

"There are large areas of the world where it seems likely there may well be interactions between cetaceans and fisheries, but for which there are, as yet, no data, and no idea of any impact that such fisheries may cause. This lack of information on the impacts of a fishery does not imply, however, that there is no problem, especially since reporting of just a few individuals in a specific fishery may be indicative of a larger interaction. Only when scientists can accomplish a detailed study of the cetacean stock abundance, the fishing effort, and the bycatch rate in each fishery can a thorough and accurate assessment be made".

7.519 In contrast, due in particular to the AIDCP On-Board Observers Program and the AIDCP System for Tracking and Verifying Tuna, detailed information is available about dolphin mortalities resulting from tuna fishing activities *in* the ETP. Hence, the Panel's analysis of the existence of dolphin bycatch during tuna fishing operations *outside* the ETP is based on the evidence contained in a limited amount of *ad hoc* studies. In this regard, we find the following observation of the Appellate Body in the context of an analysis under Article XIV of the GATS pertinent for the purposes of our own assessment of the evidence in this case:

"The selection of a methodology to assess a measure's contribution is a function of the nature of the risk, the objective pursued, and the level of protection sought. It ultimately also depends on the nature, quantity, and quality of evidence existing at the time the analysis is made". 728

7.520 The evidence submitted to the Panel suggests that the association between schools of tunas and dolphins does not occur outside the ETP as *frequently* as it does within the ETP.⁷²⁹ This evidence further suggests that although there are indications that intentional setting on dolphins occurs outside

objective scientific agencies may be the only route to a truly unbiased picture of dolphin mortality incidental to purse-seine operations" in the western Pacific Ocean).

⁷²⁷ Exhibit MEX-5, p. vii. The Panel notes that, as indicated above, Exhibit MEX-5 is a report commissioned by the United States National Oceanic and Atmospheric Administration (NOAA) Fisheries. We also recall that the United States has submitted that NOAA neither edited nor endorsed this report. Moreover, the United States highlights that the "views or opinions expressed in this report are those of the authors" and not necessarily those of NOAA, United States' second written submission, fn 75. Despite these observations, the Panel considers that the fact that this study was commissioned to outside contractors and that it reflects their views and not those of the US agency does not diminish its probative value. The fact that NOAA commissioned this study to these contractors in particular suggests that they deserved a certain level of credibility in the eyes of agency itself. Moreover, the Panel notes that the authors of this report base their conclusions on previous studies, estimates and field-research. Thus, the Panel sees no reason to disregard this evidence as untrustworthy. See also Exhibit MEX-99, p. Ev 26, which states that:

"During the 1990s, observer studies of By-Catch in pelagic trawl fisheries recorded dolphin catches in four fisheries targeting sea bass, hake, tuna and horse mackerel. The report notes that, given the size of the European fleet and the amount of fishing effort, the total number of animals caught may be significant. It also observes that the By-Catch estimate must be treated as a minimum because, for instance, some fishing fleets refused to take observers on board".

⁷²⁸ Appellate Body Report, *Brazil – Retreaded Tyres*, para. 145.

United States' first written submission, para. 58._See e.g. Exhibits MEX-2, p. 45 and US-61 (describing the type of tuna-dolphin association that takes place in the ETP); Exhibits MEX-4, p. 16-17 and Exhibit US-10, p. 38 (stating that such association occurs rarely outside the ETP).

the ETP⁷³⁰, there are "no records of consistent or widespread fishing effort on tuna-dolphin associations anywhere other than in the ETP". 731 However, there are clear indications that the use of certain tuna fishing techniques other than setting on dolphins may also cause harm to dolphins. 732

7.521 For instance, the use of unassociated purse-seine sets or FADs to catch tuna may result, in certain cases, in substantial dolphin bycatch. Trawl fishing is another method that may be employed to harvest tuna, and that may also produce dolphin bycatch. Similarly, the use of driftnets to catch tuna in coastal areas within Exclusive Economic Zones is considered "a highly destructive practice" and one of "the greatest threats to populations of small cetaceans" in certain areas of the world. 735 A report prepared by the Committee on Porpoise Mortality from Tuna Fishing established under auspices of the National Research Council thus observes that "[a]lthough purse seining for yellowfin tuna associated with dolphins in the ETP is the primary focus of this report, other methods, including additional purse-seining modes, are used for catching yellowfin. Some of these methods are known to kill dolphins, as are other techniques of fishing for other fish species (Northridge, 1984, 1991)." The report identifies longline, log and school fishing as the most important of the other fishing methods known to kill dolphins.

7.522 There are reports on dolphin bycatch resulting from tuna fishing operations in European fisheries. 737 There are also reports on dolphin mortalities arising from tuna fishing activities in

See also Exhibit MEX-5, p. 100, stating:

"In the Northeast Atlantic, within the framework for cooperation and research of the [Agreement on the Conservation of Small Cetaceans of the Baltic and North Seas, ASCOBANS], does not provide authority for actual regulation of fishing operations, even though it has documented how those operations affect cetacean bycatch. Action is up to individual parties of ASCOBANS for measures within the EEZs".

In relation to driftnets to catch tuna, the Panel observes that the US dolphin-safe provisions prohibit the use of the dolphin-safe label for tuna caught "on the high seas by a vessel engaged in driftnet fishing" (emphasis added), but they do not impose the same restriction in relation to driftnet fishing for tuna within EEZs. This implies that tuna caught by applying this method within territorial waters is eligible to be labelled dolphin-safe, see US dolphin-safe provisions subsection 1835(d)(1)(A), Exhibit US-05.

⁷³⁰ See e.g. Exhibit MEX-98, page. 59; Exhibit MEX-105. See also United States' response to Panel question No. 108, para 30, where it acknowledges that, in oceans other than the ETP, it may be possible to intentionally set a purse seine net on dolphins but if this occurs, it does not involve the intentional chase and encirclement of dolphins as a means to catch tuna.

The state of the s

⁷³³ Exhibit MEX-5, pages 26, 112, 131 (reporting on an estimate annual bycatch of 2000 dolphins as a result of tuna fishing operations by a fleet of five purse seiners using fish-aggregating devices or FADs).

According to the UK Department of Environment, Food and Rural Affairs a study of a *pair-trawl*

for tuna, conducted by the Republic of Ireland in 1998 and 1999, recorded a total bycatch of 180 cetaceans, including the Atlantic white-sided dolphin and the striped dolphin, Exhibit MEX-99, pages 13, Ev 27. See also the information on European fisheries using pelagic trawl Exhibit MEX-99, p. Ev 26.

⁷³⁵ See e.g. "[t]he By-Catch of cetaceans in fisheries is recognised to be one of the greatest threats to populations of small cetaceans and has been highlighted by various international fora including the Agreement on the Conservation of Small Cetaceans of the Baltic and North Seas (ASCOBANS). Several fisheries and sea areas have already been identified where By-Catch presents a serious and unsustainable problem. The case of pelagic driftnets used in tuna and swordfish fisheries is an example of a highly destructive practice that has now been addressed by the EU in the form of the driftnet ban that came into effect in January 2002. However, there is ample evidence of problems in other fisheries that have yet to be addressed. Moreover, many fisheries in the EU that present a threat to cetaceans are not yet being monitored for their By-Catch. Therefore, the data that are available represent only a minimum estimate of the scale of the problem". Exhibit MEX-99, p. Ev 26.

⁷³⁶ Exhibit MEX-2, p. 37.

737 "There is some evidence that other <u>pelagic fisheries</u> may also be responsible for some <u>by-catch of</u> common dolphins, although few observer studies of by-catch in these fisheries appear to have been carried out. The Whale and Dolphin Conservation Society (WDCS) and Nick Tregenza both note that, during the 1990s,

fisheries in Asia⁷³⁸ and Africa.⁷³⁹ For instance, a study commissioned by the NOAA of the US Department of Commerce refers to the case of 1,700 bottlenose dolphins and 1,000 spinner

observer studies of by-catch in pelagic trawl fisheries recorded <u>dolphin catches</u> in three other fisheries, those for mackerel, horse mackerel, hake and <u>tuna</u>" (emphasis added), Exhibit MEX-99, p. 31.

"The [Whale and Dolphin Conservation Society] also pointed out that a number of other pelagic fisheries share common characteristics with the pelagic sea bass fishery: other fisheries also use <u>trawling and pair trawling gear</u> and operate in the Celtic Sea/Bay of Biscay area during the winter months, when dolphin strandings occur on the south-west coast. In addition to the fisheries named above, they also cited the herring, blue whiting, pilchard, sardine and anchovy fisheries, and the <u>albacore tuna fishery</u>, which operates during the summer months but <u>uses pair trawling gear</u>. The WDCS considers that, <u>until these fisheries are properly monitored</u>, it is reasonable to assume that some, if not all, may be responsible for some cetacean by-catch" (emphasis added), Exhibit MEX-99, p. 32.

"Goujon estimates that the French <u>driftnet fishery for tuna</u> caught <u>1,722</u> (1365-2079) common, striped and bottlenose dolphins, and long-finned pilot whales in 1992; and <u>1,654</u> (1115-2393) common, striped and bottlenose dolphins, and long-finned pilot whales in 1993" (emphasis added), Exhibit MEX-5, p. XX-16, fn 89.

"In 1995, the UK <u>driftnet fishery for albacore</u> caught <u>104</u> striped dolphins (38 – 169)" (emphasis added), Exhibit MEX-5, p. AA-16.

"In 1996 and 1998 respectively, the Irish <u>driftnet fishery for albacore</u> caught <u>136</u> and <u>964</u> striped dolphins" (emphasis added), Exhibit MEX-5, p. AA-16.

738 "Roughly <u>1,700 bottlenose dolphins and 1,000 spinner dolphins</u> are incidentally caught in <u>gillnet</u>,

Task region's fisheries are diverse and poorly documented. Nevertheless, coastal gillnets, especially driftnets for tunas and mackerels, are widely used. After a closure in Australian waters, the Taiwanese driftnet fishery relocated and continued fishing in Indonesian waters in the Arafura Sea. With no reduction in effort, high cetacean bycatch rates are probable" (emphasis added), Exhibit MEX-5, p. 26.

"Spinner and Fraser's dolphins experience substantial bycatch in Philippine fisheries. In the Philippines, scientists estimated that about 2,000 dolphins—primarily spinner, pan-tropical spotted, and Fraser's—were being killed each year by a fleet of five tuna purse seiners using fish-aggregating devices" (emphasis added), Exhibit MEX-5, pages 26, 112, 131.

"Few recent studies appear to have been made in this area. The recent revelation that a driftnet fishery has been operating off Tristan da Cunha <u>for tuna</u>, with concomitant incidental mortality of small whales and dolphins, suggests that there may also be considerable mortality to some as yet unidentified species. Incidental mortality to Heaviside's dolphin, which is restricted to the coastal zone of South Africa and Namibia, may also be an <u>important interaction</u>, but recent data on bycatch and population size are lacking" (emphasis added), Exhibit MEX-5, p. 18.

"A now-terminated Taiwanese shark and tuna gillnet fishery operated off Northern Australia and caught bottlenose dolphins, spinner dolphins, spotted dolphins, humpback dolphins and false killer whales, a proportion of which are in this area. The fishery was mainly located in Area 71 and is discussed under that section. Given the amount of gillnetting likely to occur in this region, accidental catches may adversely affect small coastal species such as the finless porpoise and Irrawaddy dolphin to some extent. The driftnet fisheries operating farther offshore—in the Bay of Bengal, for example—might be expected to catch spinner and spotted dolphins, at least, and perhaps other species. Driftnet fisheries in the southern Indian Ocean may catch a variety of species such as the spectacled porpoise, the southern right whale dolphin, and common dolphin. All of these fisheries require more detailed information on non-target catches" (emphasis added), Exhibit MEX-5, p. 23.

"Catches in India are reported quite frequently, and formed 33% of the total catch of cetaceans recorded in the gillnet fishery at Calicut. Bottlenose dolphins are one of the commonly caught dolphins in seerfish and <u>tuna driftnet fisheries</u> on the <u>west coast of India</u>, and in coastal gillnet fisheries for pomfrets and other species too. In Sri Lanka, this species was found to consist of between 5 and 25% of the <u>total cetacean catch</u> in four different surveys amounting to <u>1,250 to 10,000 animals</u>" (emphasis added), Exhibit MEX-5, p. AA-40.

"Spinner dolphins are caught in Sri Lankan coastal gillnet and driftnet fisheries. This species is caught in Pakistani offshore deepwater gillnet fisheries and is commonly entangled in coastal driftnet fisheries for seerfish and tunas on the west coast of India, and is also entangled in other gillnet fisheries for sharks, pomfrets and other species" (emphasis added), Exhibit MEX-5, p. AA-41.

dolphins being caught in one year in the western central Pacific by gillnet, driftnets and purse-seine fisheries, including coastal gillnets fishing for tuna. The study also states that "[i]n the Philippines, scientists estimated that about 2000 dolphins –primarily spinner, pan-tropical spotted, and Fraser's-were being killed each year by a fleet of five tuna purse seiners using fish aggregating devices". Moreover, the study also stresses the "need to continue efforts to assess incidental catch in the tuna purse seine and drift gillnet fisheries" in the western central Pacific. 741

7.523 The Panel also notes that, based on the number of documented incidental bycatch of dolphins and other small cetaceans in the western central Pacific, the same study identified the need to address the western central Pacific tuna-dolphin interactions as a second tier priority for agency action. This study refers to examples of observed dolphin mortalities in the western central Pacific tuna-dolphin observed mortalities in the ETP in recent years (which amount to approximately 1000 to 1200 deaths per year).

7.524 We note that the United States has submitted that observer programmes outside the ETP exist and that they have not reported significant mortalities. Mexico observes however that there is limited observer coverage outside of the ETP⁷⁴⁶ and argues that the United States greatly overstated

"Spinner dolphins are the most frequently caught species in the Sri Lankan fishery, where they formed between 33 and 47% of the total cetacean catch in for different surveys, or roughly <u>7,050-11,750 dolphins per</u> year" (emphasis added), Exhibit MEX-5, p. AA-41.

"Finless <u>porpoise</u> are entangled in Sri Lankan coastal gillnet and driftnet fisheries, shark nets in Australian, and Indian ocean coastal gillnets. This species is commonly caught in seerfish and <u>tuna driftnet fisheries</u> throughout the west coast of India. Finless porpoises have been caught in a shrimp trawl in Pakistan in 1989, entangled in beach seines and stake nets for shrimp, and entangled in small and medium mesh finfish gillnets in shallow inshore waters of Pakistan" (emphasis added), Exhibit MEX-5, p. AA-43.

"The annual bycatch of small cetaceans in a single tuna driftnet fishery in Negros Oriental was estimated at about 400" (emphasis added), Exhibit MEX-5, pages 26, 27, 131.

739 "In the Eastern Central Atlantic, the <u>clymene dolphin</u> (Ghanaians call it the "common dolphin"), bottlenose, pantropical spotted, Risso's, long-beaked common, and rough-toothed dolphins; short-finned pilot <u>whale</u>, melon-headed whale, dwarf sperm, and Cuvier's beaked whale may all be caught in <u>large-meshed drift</u> <u>gillnets targeting tuna</u>, sharks, billfish, manta rays, and dolphins", (emphasis added), Exhibit MEX-5, p. 102.

"In 1997, the IWC Scientific Committee concluded that information on small cetaceans in Africa (outside southern Africa) is very sparse and that issues of cetacean fishery bycatch must be addressed. Projects that have sampled landing sites of small scale coastal fisheries in Ghana since 1998 show that bycatch and directed harvests of small cetaceans are commonplace and possibly increasing. The largest catches, by far, are the result of deployment of large meshed drift gillnets targeting tuna, sharks billfish, manta rays, and dolphins. The species most frequently caught are clymene (Ghanaians call it the "common dolphin"), bottlenose, pantropical spotted, Risso's, long-beaked common, and rough-toothed dolphins, together with short-finned pilot and melon-headed whales" (emphasis added), Exhibit MEX-5, p. 9.

The United States criticises this information by pointing out that it "was based on data from a report from 1994" and that "there has been no substantiation of these claims in the 16 plus years since they were made". In addition the United States claims that these estimates "do not appear to be supported by robust scientific method", United States' second written submission, para. 50. The Panel considers that while the information seems to be based on a study prepared in 1994, the United States did not submit any previous or more recent piece of evidence showing that these estimates were wrongfully calculated. Furthermore, the United States did not elaborate on why, in its opinion, the scientific basis of this study was not "robust".

- ⁷⁴¹ Exhibit MEX-5, p. 27.
- ⁷⁴² Exhibit MEX-5, pp. xxi-xxii.
- ⁷⁴³ Exhibit MEX-5, pp. 26-29, AA-60-63.
- The United States claims that dolphin observed mortalities in the ETP has been reduced to approximately 1200 dolphins per year, United States' second written submission, para. 2.
 - ⁷⁴⁵ See para. 7.512 above.
 - ⁷⁴⁶ Mexico's second written submission, para. 99.

the monitoring activities of the non-ETP regional fisheries management organizations. America also argues that it has produced evidence that no regional fisheries management organization other than the IATTC has had a comprehensive program for many years to monitor bycatch of marine mammals. We note in this respect that the United States has not challenged Mexico's assertion that observer coverage outside of the ETP is not as comprehensive as it is in the ETP. Instead, the United States has argued that Mexico's conclusion that the lack of 100 per cent observer coverage in fisheries outside of the ETP means that the risk of dolphin mortality or serious injury is unknown should be rejected. The United States suggests that if there were reasons to believe that there was a regular and significant association between tuna and dolphins in purse seine tuna fisheries outside the ETP or regular and significant dolphin mortality or serious injury in other fisheries outside of the ETP, the relevant regional fisheries management organizations would be aware of the issue and it would be their responsibility to address such issue. However, as described above, some studies warn that absence of information should not be a basis for assuming that no problem exists and suggested rather that the level of information should be increased.

7.525 The United States also argues that there are very few interactions between US vessels fishing for tuna and dolphins in the WCPO and that reports on that fishery reveal that marine mammal take is "relatively small". In support of this assertion, the United States states that "based on reporting from independent observers on US vessels fishing for tuna in the Western Central Pacific Ocean (WCPO), the United States noted that of the 1500 sets observed in 2008, there were 5 interactions with false killer whales and one interaction with a short-finned pilot whale observed; there were no observed interactions with other marine mammals". The United States cites as source for this assertion a communication of a NOAA official, without however providing such communication or any supporting evidence. As observed by Mexico, this leaves unexplained the basis for this assertion, including whether the observers were specifically trained to watch for dolphin interactions, where the observed vessels were fishing, or how the vessels were fishing. We also note Mexico's observation that the observation of 1500 sets represents a sample of 1.5% of the total number of sets in the WCPO, making it meaningless. We further note that this information relates only to US vessels.

⁷⁴⁷ Mexico's second written submission, para. 89.

⁷⁴⁸ Mexico noted that the Indian Ocean Tuna Commission (IOTC) is still in the initial stages establishing programs to monitor bycatch quoting a report that stated that the IOTC had resolved to implement a regional observer scheme (ROS) beginning in July 2010 that would target a coverage of 5% of fishing operations for vessels larger than 24m from 1 July 2010 which would gradually increase towards a 5% coverage for vessels smaller than 24m fishing outside their EEZ and develop sampling programs to monitor artisanal fisheries at landings. The report quoted by Mexico noted that although the ROS started on 1 July 2010, to date, no observer trip reports had been sent to the Secretariat. Mexico also showed that the International Commission for the Conservation of Atlantic Tunas (ICCAT) also has very limited information on bycatch by referring to its latest biennial report published in 2010, includes the following, which with respect to observer coverage noted that "In order to obtain data on the composition of the catches, particularly those of spawners, relative to the fishing areas and seasons, there shall be observers on board at least 5% of longline vessels over 24 meters fishing for bigeye." Mexico also explained that the ICCAT Regional Observer Program shall be established to ensure observer coverage of 100% of all fishing vessels over 20 m fishing for bigeye tuna in a specific area is nonetheless limited to large vessels fishing for bigeye tuna in a specific region and the specific tasks of the observers do not appear to include recording data on bycatch. See Mexico's second written submission, paras. 90-96.

⁷⁴⁹ United States' second written submission, para. 47.

⁷⁵⁰ United States' second written submission, para. 47.

⁷⁵¹ United States' comments on Mexico's answers to Panel question No. 101, para. 29.

⁷⁵² See paragraph 7.518 above, fn 725 referring to MEX-5.

United States' response to the Panel question No. 15, para. 49.

⁷⁵⁴ Mexico's second written submission, para. 98.

- 7.526 The United States further refers to a report of the Secretariat of the Pacific Community in the framework of the Oceanic Fisheries Program, assessing the status of stocks in the western and central pacific tuna fishery, which was provided as evidence by Mexico. The United States notes that this report shows that over an eleven year period, 33,319 tuna purse seine sets were observed and bycatch of 1315 marine mammals was reported, 46 of which were reported killed.
- 7.527 We note that the figures at issue relate specifically to the equatorial purse-seine fishery in the WCPO. The report takes note of a "relatively high level of observer coverage", with one-third of sets observed over the period, and observes that marine mammals were caught in a very small proportion of these observed sets, mainly from sets targeting tuna schools associated with either whales or dolphins. We also note that the full data set reflects, in addition to the 46 marine mammals identified as killed referred to above, 36 alive and 1233 whose fate remained "unknown".
- 7.528 We also note that the authors of the report themselves do not consider these figures to be a sufficient basis for overall estimates for the fishery. Rather, the authors suggest that a more detailed analysis would be required for this purpose:

"For the WCP-CA, incidences of capture of marine mammals in the equatorial purseseine and area-specific longline fisheries were summarised from observer data to identify (where possible) the main species caught and provide an indication of the level of fishery interaction. However, it is important to note that no attempt has been made to apply these data to provide overall estimates for the fishery. This work would require a more detailed analysis of observer coverage rates of individual fleets and consideration of appropriate spatial and temporal stratification. It is intended to undertake such an analysis during the next year."

- 7.529 We are therefore not persuaded that these figures demonstrate, as the United States suggests, that there is no or only insignificant risk of dolphin mortality or injury arising from tuna fishing operations outside the ETP or call into question the other evidence referred to by Mexico and cited above. Rather, the observations reflected in this report confirm the existence of interaction outside the ETP between purse seine (and longline) tuna fisheries and marine mammals, including dolphins, and the existence of some bycatch and mortality in this context in the WCPO.
- 7.530 Furthermore, the authors' conclusion that further study would be required in order to draw overall conclusions for the fishery confirms that the information available in this respect is incomplete and that this issue warrants further analysis. This observation is consistent with other evidence

Overview And Status Of Stocks," 2008, ps. 59-60. Exhibit MEX-98. The Secretariat of the Pacific Community (SPC) is an international organisation that provides technical and policy advice and assistance, training and research services to its Pacific Island members. The Fisheries, Aquaculture and Marine Ecosystems Division assists members to sustainably develop and manage their coastal, aquaculture and oceanic (tuna) fisheries by providing technical advice, assistance and training; conducting research and monitoring; and disseminating information through two programs mainly, the Oceanic Fisheries Program and the Coastal Fisheries Program. The Oceanic Fisheries Program provides data products, such as stock assessment and evaluation of species. It is in this context that the 2006 report on stock assessment was issued, with a view primarily to assessing tuna stocks. The SPC is not responsible for international management of tuna fisheries throughout the region; this is the responsibility of the Western and Central Pacific Fisheries Commission (WCPFC) whose membership includes all 26 SPC members, as well as Canada, China, Chinese Taipei, the European Union, Japan, Korea and Philippines. The OFP provides data management and stock assessment services and advice to WCPFC under an annual service agreement.

⁷⁵⁶ See Exhibit MEX-98, p. 59-60.

presented by Mexico and discussed above.⁷⁵⁷ We also recall in this respect the observations of the report commissioned by NOAA and referred to above, that the absence of information should not be assumed to reflect an absence of problem.⁷⁵⁸

7.531 Therefore, based on the evidence provided to us, we conclude that Mexico has demonstrated that certain tuna fishing methods other than setting on dolphins have the potential of adversely affecting dolphins, and that the use of these other techniques outside the ETP may produce and has produced significant levels of dolphin bycatch, during the period over which the US dolphin-safe provisions have been in force.

Whether access to the US dolphin-safe label is available to tuna caught in conditions that may adversely affect dolphins

7.532 We note that where such tuna is caught outside the ETP, it would be eligible for the US official label, even if dolphins have in fact been caught or seriously injured during the trip, since there is, under the US measures as currently applied, no requirement for a certificate to the effect that no dolphins have been killed or seriously injured outside the ETP. The seriously injured outside the ETP.

7.533 The Panel further observes that some of the examples referred to above involve dolphin mortalities occurring as a result of tuna fishing operations in the western central Pacific Ocean (WCPO)⁷⁶⁰ where, according to the United States, "almost all U.S. purse seine vessels fish for tuna". The United States has further clarified that the US tuna canning industry uses tuna caught by both, domestic and foreign vessels. According to the information provided by the United States, during 2009, 99 per cent of the tuna used by US canners was caught in the western Pacific by US vessels. Similarly, 80 per cent of the tuna US canners sourced from foreign vessels was also caught in the western Pacific. Thus, according to these figures, approximately 86% of the tuna packed by US canners in 2009 was fished in the western Pacific Ocean. The Panel also notes that

Moreover, there are indications that the three major tuna canners have processing facilities in locations situated near the western Pacific Ocean. Indeed, the study conducted by the Committee on Reducing Porpoise Mortality from Tuna Fishing established by the National Research Council, and funded by the US Department of Commerce narrates that:

"During the 1960s and 1970s, U.S. tuna companies relocated most of the canning operations from the U.S. west coast to offshore U.S. sites in <u>American Samoa</u> and Puerto Rico to take advantage of favorable economic conditions including relatively inexpensive labor and tax advantages offered by commonwealth and territorial governments. During the 1980s, these companies began shifting from these offshore U.S. territories to <u>Asian sites</u> to take advantage of even cheaper labor and less costly worker benefits and environmental restrictions" (emphasis added).

The same study reports that, at the moment it was concluded, Starkist was the only US company still operating in American Samoa or Puerto Rico, while the companies that owned the brands Chicken of the Sea and Bumble Bee had been sold to Indonesian and Thai companies, respectively. Exhibit MEX-5, p. 31. This suggests the possibility that all three major tuna canners participating in the US market, including those that have been sold to foreign companies, could source the tuna contained in their products from fleets operating in near-by waters, including the western Pacific Ocean.

⁷⁵⁷ See also Exhibit MEX-5, p.27.

⁷⁵⁸ See para. 7.518 above and fn 725 referring to MEX-5.

⁷⁵⁹ We recall in this respect that, as described in para. 2.25 above, such a requirement would exist in the event of a determination by the Secretary of Commerce that there is regular and significant dolphin mortality or serious injury of dolphins, but that no such determination has in fact been made.

⁷⁶⁰ Exhibit MEX-5, pages 18, 23, 26, 27, 112, 131, AA-40, AA-41, AA-43.

United States' response to Panel question No. 15(b), para. 49.

⁷⁶² United States' second written submission, para. 22; see also Exhibit US-61, p. 151.

⁷⁶³ Exhibit US-55, p. 2.

Thailand's tuna canners, which dominate the US market of imported tuna products⁷⁶⁵, also obtain most of their tuna supply from companies operating in the WCPO. ⁷⁶⁶

7.534 In addition, the United States has explained that the techniques currently used by US and foreign vessels to catch tuna for products sold on the US market include "purse seine sets on dolphins, unassociated purse seine sets (sets on floating objects such as FADs and free swimming schools), longline, troll, pole and line, gillnet, harpoon and handline" (emphasis added). 767 The United States has also stated that "most tuna products sold in the United States are eligible to be labelled dolphin-safe", and that for instance, "98.5% of total US imports (by volume) of canned tuna products were eligible to be labelled dolphin-safe in 2009". This means the vast majority of tuna products containing tuna caught in the western Pacific Ocean by using, for instance, FADs, trolls or gillnets, sold in the US market are eligible to be labelled dolphin-safe. ⁷⁶⁹Based on this information, the Panel considers that Mexico has sufficiently demonstrated that tuna caught during a trip where dolphins were killed or seriously injured using a method of fishing other than setting on dolphins outside the ETP may be contained in the tuna products sold in the US market under the dolphin-safe label. This is particularly true considering that no determination of existence of regular and significant tunadolphin association⁷⁷⁰ or regular and significant mortality⁷⁷¹ has been made for any fishery outside the ETP, which means that under the DPCIA provisions that are currently applicable all tuna products containing tuna caught in a non-ETP fishery using a method other than setting on dolphins are eligible to be labelled dolphin-safe without certifying that no dolphin was killed or seriously injured in the set.

7.535 We also note that the United States has explained that Subsection 1385(d)(3)(C)(i) of the US dolphin-safe provisions establishes conditions that must be met in addition to those contained in Subsections (d)(1) and (d)(2), to label tuna products dolphin-safe with an alternative mark (rather than the official DOC mark). Namely, if an alternative label rather than the official DOC label is used, no dolphin may have been killed or seriously injured in the sets in which the tuna was caught regardless of the type of fishery. Thus, according to the United States, tuna products using an alternative dolphin-safe mark may not contain tuna caught in a trip where dolphins were killed or seriously injured even if there is no determination that there is regular and significant tuna-dolphin association or dolphin mortality in the fishery where that tuna was caught. The United States further

⁷⁶⁵ Mexico's first written submission, para. 37.

⁷⁶⁶ These companies are FCF, Tri Marine and Itochu, which specialize in the procurement of canning-grade tuna, Exhibit MEX-111.

⁷⁶⁷ United States' response to Panel question 90, para. 10. The Panel also recalls that the United States has clarified that techniques used to catch tuna used in tuna products exported to the United States by vessels flagged to Thailand, the Philippines, Ecuador, Vietnam and China since 2000, include gillnet, longline, pole and line, purse seine, troll, etc. The Thai fleet for instance uses, *inter alia*, gillnets and trolls to fish for tuna, United States' response to Panel question No. 98, para. 22.

⁷⁶⁸ United States' response to Panel question No. 95, para. 18.

This is particularly true if we consider that approximately 1% of the imports of tuna products in the United States come from Mexico, and that most of these imports do not qualify to be labelled as dolphin-safe. See Exhibits MEX-86(A), MEX-86(B), MEX-86(C) and MEX-66. The United States has stated that "of the \$1.2 billion of U.S. imports of tuna and tuna products, the <u>vast majority</u> contained tuna that was caught by methods other than setting on dolphins and are eligible to be labelled dolphin-safe" (emphasis added). The United States added that "[f]or example, of the over 10,000 entries of canned tuna products in 2009, all but 137 entries were dolphin-safe", United States' response to Panel question No. 95, para. 18.

⁷⁷⁰ United States' response to Panel question No. 12(a), para. 29.

⁷⁷¹ United States' response to Panel question No. 85(a), para. 1.

⁷⁷² See paragraph 7.489 above.

argues that none of the US canners uses the official dolphin-safe label mark, since they have developed their own dolphin-safe logos. 773

7.536 The Panel recognizes that Subsection 1385(d)(3)(C)(i) suggests that tuna products using alternative labels must not contain tuna caught in a set where dolphins were killed or seriously injured even if no determination of regular and significant tuna-dolphin association or dolphin mortality has been made in relation to the fishery in question.⁷⁷⁴ Although the Panel fails to understand the reasons behind the US regulator's decision to establish a difference in this regard between the official and alternative dolphin-safe labels, it defers to the United States' interpretation of its own regulations and accepts its explanation that these requirements apply *in addition* to the general requirements of Subsections (d)1 and (d)2.⁷⁷⁵

7.537 At the same time, we also note that according to NOAA Form 370, which is the instrument that "reflects the documentation requirements necessary to support dolphin-safe claims for tuna products as set forth in the [US dolphin-safe provisions]"⁷⁷⁶, tuna products imported to United States containing tuna caught in a non-ETP fishery where there is no regular and significant tuna-dolphin association or dolphin mortality, using a method other than setting on dolphins, do not need to demonstrate (or even declare) that no dolphins were killed or seriously injured in order to use the official DOC mark. NOAA Form 370 also does not appear to provide any particular space in which such declaration might be made.

7.538 Therefore, although the text of the US dolphin-safe provisions seems to require that tuna products using an alternative label must not contain tuna caught in a set where dolphins were killed or seriously injured, regardless of whether there is a determination of regular and significant tuna-dolphin association or dolphin mortality, it is not clear that the application of these provisions through the use of the NOAA Form 370, as described by the United States itself, involves any enforcement of such requirement. This seems to considerably increase the chances of tuna caught during trips involving dolphin killings or serious injuries, being sold as dolphin-safe in the US market.

7.539 In addition, we note that to the extent that the use of an "alternative" label is subjected to different – and more stringent – requirements than the official label, the official label and alternative labels would have different meanings, leading, as Mexico points out, to further confusion for consumers. Furthermore, that the requirements applicable to the use of an alternative label may be more stringent than those for access to the official label does not modify the fact that access to the official label is available to tuna products caught outside the ETP by methods other than setting on dolphins, regardless of whether dolphins have been killed or seriously injured in the sets.

7.540 Finally, we note that the United States also argues that in adopting measures to fulfil legitimate objectives, it is appropriate for Members to consider the costs of such measures in light of their benefits and that, where the risk of a product containing tuna caught in a set in which a dolphin was killed or seriously injured is low, the dolphin-safe provisions do not condition dolphin-safe claims on the provision of an observer's statement that no dolphins were killed or seriously injured. The United States argues that the cost of requiring a certification for tuna caught in the ETP using purse seine nets that no dolphins were killed or seriously injured in the set in which the tuna was

⁷⁷⁵ United States' response to Panel question No. 8; United States' second written submission, paras. 40-41. See also Section II.F above.

 $^{^{773}}$ United States' response to Panel question No. 11, para. 28.

⁷⁷⁴ Exhibit US-05.

⁷⁷⁶ United States' response to Panel question No. 7, para. 16.

⁷⁷⁷ Mexico's response to Panel question No. 86, para. 3.

⁷⁷⁸ United States' second oral statement, para. 56.

caught is relatively low as compared to the cost of imposing the same requirement for tuna caught outside the ETP, because if it required certification that no dolphins were killed or seriously injured for tuna caught outside the ETP, this would impose the additional cost of maintaining 100 per cent observer coverage on vessels in that fishery, for which there is no intergovernmental agreement that such observer coverage would be warranted. 779

7.541 We note that this explanation is not consistent with the United States' own explanation that Subsection 1385(d)(3)(C)(i) imposes a requirement that no dolphin be killed or seriously injured if an alternative label is used, as discussed in the preceding paragraphs. We understand the United States to suggest that the requirement that no dolphin has been killed or seriously injured is in fact not enforced through a certification requirement with respect to the alternative label, while a statement by an observer is required with respect to the same substantive requirement, for the use of the official label in the ETP. We fail to see, however, how the cost of demonstrating compliance with the same requirement (i.e. that no dolphin was killed or seriously injured) would justify that no such requirement be imposed with respect to the use of an official label, while it would be imposed for the same tuna caught in the same conditions in the same fisheries, in the case of use of an alternative label. It is also not clear to us what the imposition of this additional requirement means in practice in respect of the alternative label, if it is assumed that it cannot be verified and that this is a reason not to impose it for the use of the official label.

7.542 In light of the above, we find that, to the extent that the US dolphin-safe measures grant access to the label to certain tuna products that may have been caught in a manner that adversely affects dolphins (i.e. tuna products containing tuna caught outside the ETP by methods other than setting on dolphins, regardless of whether dolphins may have been killed on the trip), they do not allow the consumer to accurately distinguish between tuna caught in a manner that adversely affects dolphins and other tuna. In addition, to the extent that, depending on whether such tuna is caught in the ETP or outside the ETP, the measures require or do not require a certification that no dolphin has been killed or seriously injured, we find that this also does not allow the consumer to accurately identify tuna caught in conditions that are similarly harmful to dolphins.

7.543 We also note that the provisions of the DPCIA themselves envisage the possibility that a fishery outside the ETP would be identified as one "having a regular and significant mortality, or serious injury of dolphins", which would then lead to the application in such fishery of requirements comparable to those applied in the ETP, including a requirement to certify that no dolphin has been killed or seriously injured on the trip. However, in the absence of such determination – despite strong evidence that regular and significant mortality and serious injury of dolphins also exists outside of the ETP, as described in paragraph 2.25 and explained in paragraph 7.532 above – no such requirement currently applies outside the ETP. We note in this respect that we are not aware of any process or procedure having been established or initiated, under the US dolphin-safe measures, in order to trigger such determination.

7.544 To summarize, therefore, the US dolphin-safe labelling provisions, as currently applied, address observed and unobserved mortality resulting from setting on dolphins, in any fishery, as well as observed mortality from other fishing methods within the ETP. However, they do not address mortality (observed or unobserved) arising from fishing methods other than setting on dolphins outside the ETP. In addition, the measures also do not allow the consumer to be informed in any manner with respect to measures taken in the context of the AIDCP, to minimize the incidental taking and killing of dolphins when setting on them in the ETP.

⁷⁷⁹ United States' second written submission, paras. 46, 149.

7.545 Consequently, when consumers buy a tuna product labelled dolphin-safe under the US measures, they may be completely assured that no dolphin was adversely affected during the catching of that tuna in the ETP. However, consumers would not have equal certainty that no dolphin was killed or injured or that dolphins were not otherwise adversely affected in respect of tuna caught outside the ETP.

Adverse effects of fishing activities on dolphins in and outside of the ETP

7.546 The Panel would like now to address the United States' suggestion that granting access to the dolphin-safe label to tuna caught outside the ETP using a method other than setting on dolphins does not represent a considerable risk of confusion for the consumers, because the probability that dolphins were hurt during the fishing of that tuna is very low. The United States has explained that the difference in certification requirements between tuna coming from the ETP and tuna harvested in other areas reflects a calibration of the likelihood that any given dolphin would interact with fishing gear and be accidentally killed or seriously injured as a result.⁷⁸⁰

7.547 The United States has suggested that the rationale behind these differences in requirements for tuna harvested in the ETP is the existence of special conditions of this area of the Pacific Ocean. First, according to the United States, in the ETP "there is regular and significant tuna-dolphin association that is commercially exploited on a wide scale commercial basis to catch tuna" and for which dolphin mortality and serious injury are "a regular, foreseeable and expected consequence of exploiting that association". Second, in the United States' view, "dolphin populations in the ETP are depleted with abundance levels at less than 30 per cent of the levels they were at before the practice of setting on dolphins to catch tuna began".

7.548 The United States submits that these differences mean that outside the ETP there is much lower likelihood that dolphins will be killed or seriously injured as a result of tuna fishing activities. However, the United States adds that "to the extent that unintentional deaths of dolphins in purse seine fishery other than the ETP or in non-purse seine fishery does indicate a systemic issue", the US measures contemplate "that dolphin-safe claims on tuna products that contain tuna caught in such fisheries be supported by an observer's statement that no dolphins were killed or seriously injured in the set". The seriously injured in the set".

7.549 We understand the United States' explanation to suggest that the US dolphin-safe provisions aim at establishing different requirements based on the likelihood that dolphins will be killed or seriously injured. Thus, the greater the risk, the stricter the certification requirements are for tuna products to be labelled dolphin-safe. We acknowledge that it may be legitimate to provide for a different degree of stringency or of tolerance to adverse impacts, depending on the extent of the threat. This is consistent, in our view, with the requirement to take into account the risks that non-fulfilment of the objective would create, as foreseen in Article 2.2. However, we are not persuaded that this is what the US dolphin-safe measures achieve, in respect of the different treatment of tuna caught within and outside of the ETP.

7.550 We first observe that, as described by the United States itself, its measures seek to address a range of adverse effects of fishing techniques on dolphins. As established in paragraphs 7.483 to 7.486 above, this includes situations in which dolphins are killed or seriously injured. It is also clear,

⁷⁸⁰ United States' second written submission, paras. 39-47.

⁷⁸¹ United States' second written submission, para. 42; United States' second oral statement, para. 14

⁷⁸² United States' second written submission, para. 43.

⁷⁸³ United States' second written submission, para. 44.

⁷⁸⁴ United States' second written submission, para. 44.

from the United States' own description of its objectives, that its intention to address adverse effects of fishing techniques on dolphins in the form of observed or unobserved mortality is not subordinated to considerations relating to the conservation of depleted dolphin stocks. This is made clear by its clarification, in response to a question by the Panel, that addressing such impacts "*might also be considered* to seek to conserve dolphin populations". In these conditions, we are not persuaded that differences with respect to the depletion status of the stocks at issue would be sufficient to justify the absence of any requirement to certify that no dolphin has been killed or seriously injured, in situations where dolphins may in fact have been be killed or seriously injured.

7.551 Notwithstanding this conclusion, we also note that, although the United States did not identify the conservation of dolphin stocks or populations as a direct objective of its measures, the recovery of dolphin populations, as well as the likelihood of mortality or serious injury in a particular fishery, are among the considerations reflected in the design and structure of the US dolphin-safe labelling provisions. We therefore consider further the extent to which such considerations could explain the absence of a requirement to certify that no dolphin was killed for tuna caught outside the ETP.

7.552 We first note that, even assuming that the United States' contention that certain environmental conditions in the ETP (such as the intensity of tuna-dolphin association) are unique,⁷⁸⁷ the evidence submitted to the Panel suggests that the *risks* faced by dolphin populations in the ETP are *not*. As discussed above, a study commissioned by NOAA entitled "An evaluation of the most significant threats to cetaceans, the affected species and the geographic areas of high risk, and the recommended actions from various independent institutions", contains multiple examples of numerous dolphins being killed annually in other fisheries as a result of tuna fishing activities and recommends further action being taken in this respect.⁷⁸⁸

7.553 In some cases, cetacean bycatch in other fisheries targeting different species including tuna, has been considered "a serious and unsustainable problem". For instance, the above-mentioned study recommends that the United States collaborates with international organizations to "investigate the interaction between tuna purse seine vessels fishing for tuna off the coast of West Africa and whales and dolphins". According to this study, "allegations and sparse documentation of these interactions have existed for more than twenty years". Thus the study identifies the need to "plac[e] observers on tuna vessels fishing in these areas [to] help document the occurrence of association of

⁷⁸⁵ See paragraphs 7.485-7.486.

⁷⁸⁶ As noted in footnote 162 above, the change in the US dolphin-safe standard from a standard of absence of setting on dolphins to a standard of absence of killing or seriously injury dolphins was made contingent upon secretarial findings regarding whether the intentional deployment on or encirclement of dolphins with purse seine nets is having a significant adverse impact on any depleted dolphin stock in the eastern tropical Pacific Ocean. The type of scientific research mandated to substantiate the secretarial finding also evidenced the preoccupation for conservation matters insofar as it required, inter alia, population abundance surveys of depleted stocks and research into whether the physiological stress effects of using purse seine nets to chase and encircle dolphins is adversely affecting depleted dolphin populations prior to implementing any change in the dolphin-safe label. However, the court rulings overruled the findings, emphasizing among other things the recovery issue and stating that all of the best available evidence pointed towards the fishery as the cause of the dolphin's failure to recover (See for instance Earth Island Institute v Evans). No. 03-0007, 2004 U.S. Dist. LEXIS 15729, 2004 WL 1774221, at *30-31 (N.D. Cal. Aug. 9, 2004). In addition, the DPCIP foresees the possibility of requiring a certification that no dolphins are killed or seriously injured, if a determination is made that serious mortality or tuna-dolphin association exists in the fishing. In addition, as described in Section III above, the US dolphin-safe provisions foresee the possibility of more onerous certification requirements outside the ETP in the event of a determination of regular mortality, or of association between tuna and dolphins, in a fishery.

⁷⁸⁷ United States' second oral statement, paras. 14-18

⁷⁸⁸ E.g. Exhibit MEX-5, pp. 26, 27, 112, 131AA-16, fn 89, AA-40, AA-41.

⁷⁸⁹ Exhibit MEX-99, p. Ev 26.

tuna schools with whales and dolphins and the frequency of encirclement and magnitude of any bycatch. At the same time, the study reports that the bycatch in this area "threatens the continued existence of Atlantic humpback dolphins" and reports that populations of this species of dolphin near the coast of Ghana and Togo are "severely depleted".

7.554 The Panel also observes that in the western Pacific Ocean, where most of the tuna sold in the US market is sourced from, there are also examples of incidental dolphin mortalities which affect a percentage of the dolphin populations in that area that is higher than the percentage of dolphins observed to be affected in the ETP (which is less than 0.1 per cent) under the controlled conditions of the AIDCP. The Panel notes in particular, that the same study indicates that in the Philippines, Indonesia, Thailand, and elsewhere in the western central Pacific, relatively little is known about abundance, distribution, and bycatch levels of cetaceans such as the Irrawaddy dolphin, Indo-Pacific humpback dolphin, Indo-Pacific bottlenose dolphin, finless porpoise, and spinner dolphin (and its dwarf form) and that "comprehensive cetacean abundance and bycatch surveys are needed to develop effective mitigation strategies", including assessments on "incidental catch in the tuna purse seine and drift gillnet fisheries". The Panel also notes that the study reports that Irrawaddy dolphin populations are seriously depleted in parts of Thailand and the Philippines.

7.555 Thus, based on the evidence, we are not persuaded that the US statement that "in the course of using other fishing methods for tuna outside the ETP, *a* dolphin may be killed or seriously injured"⁷⁹⁶, accurately represents the magnitude of the threat posed by tuna fishing to dolphins outside of the ETP.

7.556 Again, the Panel recognizes that the evidence concerning dolphin bycatch outside the ETP is not as detailed and as abundant as the information in relation to the ETP. As mentioned above, it has been observed in this respect that the absence of information is not indicative of absence of a problem and that the scarce information available should be considered as an underestimation of the bycatch. The is also known that some fishing fleets refuse to take observers on board. Therefore, we disagree with the United States that "[i]f anything can be inferred from the absence of a 100 per cent observer requirement to monitor marine mammal interactions in other fisheries is that in those fisheries there is no problem such as the one in the ETP".

7.557 We also note that the evidence suggests that observed dolphin mortality in the ETP, by contrast, is low relative to population size. Indeed, the United States acknowledges that the observed annual dolphin mortality in the ETP "is not believed to be significant from a population recovery perspective". In addition, most of the studies submitted by the parties also suggest that the

⁷⁹⁰ Exhibit MEX-5, p. xxi.

⁷⁹¹ Exhibit MEX-5, p. 12.

Figure 1992 Exhibit MEX-5, p. 30 (reporting that the percentage of the population of dolphins observed killed is less than 0.1 per cent), p. 96 (indicating for instance that the estimate bycatch mortality for spinner dolphins in this area is 1,500-3,000 dolphins which represents between 5 and 10 per cent of the population of this type of dolphins in this area); see also, e.g., United States' first written submission, paras. 62-64 (noting that bycatch of dolphins in the ETP stands apart from other fisheries in that three to six million dolphins are chased and encircled each year when purse seine nets are intentionally set on them to catch tuna).

⁷⁹³ Exhibit MEX-5, pp. 29 and 114.

⁷⁹⁴ Exhibit MEX-5, p. 27.

⁷⁹⁵ Exhibit MEX-5, p. 28-29, 113.

⁷⁹⁶ United States' second written submission, para. 44 (emphasis added).

⁷⁹⁷ Exhibit MEX-5, p. vii.

⁷⁹⁸ Exhibit MEX-99, p. Ev 26.

⁷⁹⁹ United States' second oral statement para. 18.

⁸⁰⁰ Exhibit US-20, p. 12; Exhibit US-21, p. 10.

⁸⁰¹ United States' response to Panel question No. 36, para. 89.

dolphin populations identified by the United States as depleted are recovering, ⁸⁰² although they disagree as to whether they are growing at sufficiently high rates. Nevertheless, the most recent estimates show that "[o]ver the 8-year period from 1998-2006 all 3 of the officially depleted dolphin stocks (coastal and northeastern offshore spotted dolphins and eastern spinner dolphins) were estimated to be growing at rates considered to be near the 4-8 per cent maximum possible for dolphins". ⁸⁰³ The United States, while suggesting the possibility that uncertainty in abundance estimates may mean that dolphin stocks are actually declining, notes that the most recent assessment model estimates median annual population growth rates of northeastern offshore spotted dolphins and eastern spinner dolphins of 1.7 per cent and 1.4 per cent per year, respectively. The United States also notes, however, that this estimate is 2.3 per cent and 2.6 per cent below the expected annual population growth rate, i.e. 4 per cent. ⁸⁰⁴

7.558 We also note that it has been suggested that factors other than purse-seine fishery-related effects are hindering the growth rate of dolphins stocks in the ETP. Evidence submitted by the United States indicates that "it is unknown whether such low intrinsic growth rates are natural to these species or whether the low rates are due to other factors impeding the recovery of these stocks". 805

⁸⁰² Exhibit US-4, p. 1; Exhibit US-19, p. 32.

⁸⁰³ Exhibit US-20, p. 12. The Panel notes that the United States has questioned Mexico's assertion that this report stands for the proposition that the relevant dolphin populations are growing at rates near 4-8 per cent and pointed out that the report gave several caveats that precluded drawing that conclusion. First, the United States submits that statistical and methodological constraints preclude a definitive conclusion as to whether the higher estimates mean the populations have actually increased. Given the 95% confidence intervals on the estimates of growth rate, it is possible that these stocks are declining, United States' first written submission, para. 48, United States' second oral statement, para. 12, Exhibit US-20, p. 12-13. Second, the United States also points out that this study indicates that the apparent decrease in abundance of the western/southern stock of offshore spotted dolphin is coincident with the apparent increase abundance of northeastern offshore spotted dolphins, suggesting that dolphins could be moving across the geographic boundaries, United States' first written submission, fn 49, United States' second oral statement, para. 12, Exhibit US-20, p. 13. Finally, the United States submits that this report contains an abundance estimate and is not an assessment model, United States' second oral statement, para. 11. The United States explains that because of variability in and uncertainty surrounding point estimates of abundance, it is not possible to infer population growth rates by looking at perceived trends in these point estimates over time. The United States further explains that the rate at which dolphin populations are growing should be estimated using assessment models, which can condition based on realistic population dynamics, United States' response to Panel question No. 37, para, 92, United States second oral statement, paras, 11-12, Exhibit US-20, p. 13. In this regard the Panel observes that, because of the very nature of the issues this type of studies deal with, it appears to be extremely difficult to arrive to an optimum level of absolute certainty. We are aware that the conclusions contained in most of the studies submitted by the parties imply a certain margin of error. For instance, the study used by the United States to support its argument that dolphin stocks in the ETP are not recovering, also includes certain caveats and tempers its conclusions by "admitting that there is small probability that, relative to their respective estimates ... neither [the northeastern offshore spotted stock nor eastern spinner dolphins] stock was depleted in 2002", Exhibit US-21, p. 9. The Panel also observes that besides pointing out that this study is not an assessment model, the United States did not submit any other reason indicating that the results provided in this study were not reliable. Nor did the United States provide any evidence supporting the hypothesis that the increase in the numbers of northeastern offshore spotted dolphins is due to a relocation of the western/southern stock. Finally, the Panel notes the United States itself has also relied on it to support its own arguments, see e.g. United States' first written submission, para. 47.

⁸⁰⁴ United States' response to Panel question No. 37, paras. 90-92. According to the United States northeastern offshore spotted dolphins are growing at rates of 1.7% while eastern spinner dolphins growth rate is 1.4 per cent.

Exhibit US-21, page 10. One hypothesis is that northeastern offshore spotted and eastern spinner dolphin populations are already at, or near, their carrying capacities (Exhibit US-19, page 32). Another hypothesis is that after bycatch due to the purse-seine fishery has been reduced or eliminated, there is a lag

- 7.559 As described above, the United States has argued that any differences in documentation to substantiate dolphin-safe claims are calibrated to the risk that dolphins will be killed or seriously injured when tuna is caught. In this context, the United States has emphasized the uniqueness of the ETP in terms of the phenomenon of tuna-dolphin association, which is such that it is used widely on a commercial basis and causes observed and unobserved mortalities, which the United States argues are fundamentally different from dolphin mortalities occurring in other oceans. We are not persuaded, however, that the United States has demonstrated that the requirements of the US dolphin-safe labelling provisions are "calibrated" to the likelihood of injury, as it suggests.
- 7.560 We recall that under the US measures, intentionally setting on dolphins disqualifies tuna products from access to a dolphin-safe label, both in and outside the ETP. In these circumstances, it is unclear to us why the requirement that no dolphin be killed or seriously injured in the sets in which the tuna was caught would apply in respect of tuna caught by methods other than setting on dolphins in the ETP but not outside the ETP, for access to the official label. Even assuming that, as we have accepted earlier, setting on dolphins may result in observed and unobserved harmful effects on dolphins, these are not the effects targeted by the additional requirement imposed, in respect of tuna caught in the ETP, that no dolphin have been killed or seriously injured during the set (since, by hypothesis, no setting on dolphins must have taken place, in order to qualify for the label). Accordingly, the only types of adverse effects on dolphins that can be targeted by the additional requirement applied in the ETP that no dolphin have been killed or seriously injured (applied in addition to the requirement of not setting on dolphins) are those arising from incidental bycatch as a result of fishing methods other than setting on dolphins. The United States does not explain, however, why such incidental bycatch would be more likely to take place in the ETP than outside the ETP, and justify a requirement that no dolphin be killed or seriously injured with respect only to the ETP.
- 7.561 Conversely, assuming for the sake of argument that the United States is correct in its assertion that tuna and dolphins do not associate in other oceans as they do in the ETP, the need for maintaining the requirement of absence of setting on dolphins would not be justified for fisheries other than the ETP on the basis of the risks for dolphins, since there would be no possibility of exploiting such interaction outside of the ETP on a commercial basis. To the extent that an incidental risk of death or injury remains, due to other fishing methods, such risk could only be monitored by imposing a different substantive requirement, i.e. that no dolphins were killed or seriously injured in the sets in which the tuna was caught. Yet, the measures impose, for fisheries other than the ETP, a requirement not to set on dolphins but no requirement that no dolphin have been killed or seriously injured, at least for access to the official label. In light of the above, we are not persuaded that the requirements applicable in different fisheries under the US dolphin safe measures are "calibrated", as the United States suggests, to the likelihood of dolphins being killed or seriously injured.
- 7.562 Therefore, in light of the evidence presented in these proceedings, the Panel is not persuaded that the threats arising from the use of fishing methods other than setting on dolphins to catch tuna outside the ETP are insignificant, as the United States suggests, be it in terms of observed mortality or serious injury, or even, in at least some cases, in terms of sustainability of the populations. Nor are we persuaded that they are demonstrated to be lower than the similar threats faced by dolphins in the ETP. As a result, granting access to dolphin-safe label to tuna products containing tuna caught in this manner creates a genuine risk that consumers may be misled about whether that tuna was caught by using a technique that does not adversely affect dolphins.

7.563 From these elements, we conclude that in relation to the first objective pursued by the US dolphin-safe provisions, the US measures can only *partially* ensure that consumers are informed about whether tuna was caught by using a method that adversely affects dolphins. 806

7.564 This is in part because they could be misled into thinking that a tuna product did not involve injury or killing of a dolphin when this may in fact have been the case (if caught outside the ETP), and also because, to the extent that the measures would guarantee an absence of killing with respect to some tuna (caught in the ETP) but not others (caught outside the ETP), this creates ambiguity as to the meaning of the guarantee provided by the label. Furthermore, although the United States submits that the requirements for the alternative label differ from those of the official label, this difference would also not allow the consumer to accurately distinguish between tuna that has and has not been caught in a manner that adversely affects dolphins and create further confusion as to the meaning of the label and thus create the potential to mislead the consumer.

Whether the alternative measure proposed by Mexico provides a reasonably available less-trade restrictive means of achieving the same level of protection

7.565 The parties agree, and we concur, that the burden of establishing prima facie the existence of an alternative measure that would similarly fulfil the objectives of the United States rests on Mexico as the complainant. 807

7.566 Mexico has suggested that "a 'reasonably available alternative measure' for the United States would be to permit the use in the US market of the AIDCP 'dolphin-safe label'". According to Mexico, the objective of informing the consumers can be accomplished by the AIDCP, since the monitoring, tracking, verification and certification system under the AIDCP would allow the consumers to have the assurances that no dolphins were injured or killed during the capturing of tuna.

7.567 In the United States' view, neither the AIDCP (or measures implemented pursuant to it) nor the AIDCP label constitutes a "reasonably available alternative" that could fulfil the legitimate objectives of the DPCIA provisions. According to the United States, eliminating the DPCIA provisions *in lieu* of the AIDCP would not fulfil the objective of contributing to the protection of dolphins or ensuring that consumers are not misled or deceived about whether or not tuna products contain tuna caught in a manner that adversely affects dolphins. The United States explains that while the AIDCP has made an important contribution to dolphin protection, the US provisions further contribute to protecting dolphins by ensuring the U.S. market is not used to encourage fishing fleets to set on dolphins to catch tuna, which is a fishing technique that, in the United States' view, adversely affects dolphins.

⁸⁰⁶ According to the evidence provided by Mexico, at least around 12% of the public in the United States believes that dolphin-safe means that setting on dolphins were encircled and then released in the capture of the tuna, Exhibit MEX-64.

⁸⁰⁷ United States' first written submission, para. 169; Mexico's response to Panel question No. 67, para. 234. See also para. 7.468 above.

⁸⁰⁸ Mexico's response to Panel question No. 134, para. 52.

⁸⁰⁹ Mexico's response to question 67 from the Panel, para. 238.

United States' first written submission, para. 170; United States' first oral statement, para. 48; United States' response to question 37 from the Panel, paras. 91-2; United States' second written submission, paras. 140 and 155.

United States' first written submission, paras. 170-172; United States' first oral statement, para. 48. United States' second written submission, paras. 160-162.

- 7.568 We first note that the US dolphin-safe provisions do not formally restrict the importation or sale of tuna or tuna products that are not labelled dolphin-safe. However, as noted above, the parties agree that the US public has a preference for tuna products that are dolphin-safe, and access to the label is therefore a valuable advantage on the US market. To the extent that the proposed alternative would provide access to the label, and thus to this advantage, to a greater range of tuna products, including imported tuna products, it would be less-trade restrictive than the existing US measures, in that it would allow greater competitive opportunities on the US market to those products.
- 7.569 We now consider whether this proposed alternative is reasonably available to the United States, to achieve the same level of protection as it currently achieves with the existing US dolphin-safe labelling provisions.
- 7.570 In the Panel's view, Mexico has not suggested, as described by the United States, "eliminating the DPCIA provisions *in lieu* of the AIDCP". As we understand it, what Mexico suggests is that the US dolphin-safe label and the AIDCP label should be allowed to coexist on the US market in order to provide fuller information to US consumers. In practical terms, this would entail that tuna caught in the ETP by setting on dolphins could be labelled as complying with the AIDCP requirements for dolphin-safe labelling, and sold in the US market, as long as it was harvested in accordance with the AIDCP.
- 7.571 The AIDCP requirements for dolphin-safe labelling are based on compliance with the Procedures for AIDCP Dolphin Safe Tuna Certification, which includes submitting an independent observer's statement that no dolphin was killed or seriously injured during the sets in which the tuna were caught. Therefore, if tuna products labelled dolphin-safe according to the AIDCP were allowed in the US market, US consumers buying these products would have the assurance that no *observed* dolphin deaths or serious injuries occurred during the sets in which that tuna was caught. They would also been assured that certain specific measures have been taken, in the process of setting on dolphins to catch the tuna, in order to minimize incidental catches of dolphins. However, consumers would not be assured that tuna products using this label do not contain tuna caught by setting on dolphins, or that no *unobserved* negative effects on dolphins arose in this context.
- 7.572 As observed in the preceding section, insofar as the US dolphin-safe provisions do not require a certification that no dolphin was killed or seriously injured in respect of tuna caught outside the ETP, consumers can only be assured that such tuna, when labelled dolphin-safe, did not result in *unobserved* adverse effects on dolphins. However, as discussed above, they would remain unsure whether no dolphins were observed being killed or seriously injured, insofar as the US dolphin-safe labelling provisions fail to impose the obligation to show that tuna caught outside of the ETP without setting on dolphins involved no dolphin mortality or serious injury.
- 7.573 Thus, under both, the US measures and the AIDCP regime, consumers of tuna products would bear a certain level of uncertainty whether dolphins were adversely affected during the catching of the tuna contained in those products. Considering that under the US measures, as we concluded above, it is possible that tuna caught during a trip where dolphins were in fact killed or injured may be labelled dolphin-safe; while under the AIDCP, a label would only be granted where no dolphins was killed, but where certain unobserved adverse effects could nonetheless have been caused to dolphins, we consider that the extent to which consumers would be misled as to the implications of the manner in which the tuna was caught would not be greater if the AIDCP label were allowed to co-exist with the US dolphin-safe provisions, than it currently is under the existing measures.

_

⁸¹² United States' first written submission, para. 172.

⁸¹³ Exhibit MEX-56.

7.574 In this respect, and bearing in mind the United States' submission that its measures calibrate the applicable requirements for the use of the dolphin-safe label in accordance with the level of risk faced by dolphins, we are not persuaded that allowing consumers to be fully informed about the efforts made in the context of the AIDCP for the protection of dolphins in fishing for tuna and the monitoring of dolphin stocks in the ETP would be more misleading than allowing a dolphin-safe label to be applied to tuna caught outside the ETP in the absence of any monitoring of observed or unobserved killing or of dolphin populations.

7.575 Moreover, the Panel notes that Mexico goes beyond simply suggesting that the use of the AIDCP dolphin-safe logo be allowed in the US market. Mexico submits that the coexistence of the US and the AIDCP dolphin-safe labels, would permit US consumers to be "fully informed of *all aspects* of dolphin-safe fishing methods" (emphasis added) so that "they [could] choose accordingly when purchasing tuna products from US retailers". The Panel understands Mexico's suggestion as advocating the inclusion of more information on the dolphin-safe labels allowed in the US market on the actual and potential fishery-related adverse effects on dolphins. The Panel considers that such addition would contribute to informing consumers about the precise dolphin-safe characteristics of the various techniques used to harvest tuna. In our view, this would enhance the ability of the dolphin-safe labels to remedy market failures arising from asymmetries of information between tuna producers, retailers and final consumers in the US market. Well-informed consumers would be in a better position to use their purchasing power to influence the way tuna fisheries and canners operate.

7.576 In such conditions, accepting the use of the AIDCP dolphin-safe label as a tool to provide more precise information to US consumers would not imply that the United States should forego its own dolphin-safe label. To the extent that the alternative suggested by Mexico implies conveying more complete information to consumers, such alternative may have the potential to reduce the possibilities of consumer deception *more* than the current US dolphin-safe label.

7.577 Therefore, the Panel concludes that allowing compliance with the dolphin-safe requirements of the AIDCP dolphin-safe label in conjunction with the existing standard embodied in the US dolphin-safe provisions would be at least as apt to contribute to the objective of ensuring that consumers are not misled about whether tuna has been caught in a manner that adversely affects dolphins. Both the existing US measures and the alternative suggested by Mexico may reduce to some extent, but do not eliminate, the possibilities of the US consumers being misled. Taking into account the risks of non-fulfilment of the United States' objective if the AIDCP label was allowed, the Panel concludes, accordingly, that the alternative measure suggested by Mexico does not entail a greater risk of consumers being deceived than the US dolphin-safe provisions, as currently applied.

7.578 In light of all the above, we find that, in relation to the objective of consumer information, Mexico has identified a less trade-restrictive alternative that would achieve a level of protection equivalent to that achieved by the US measures, taking account of the risks non-fulfilment would create. Thus, the Panel concludes that Mexico has demonstrated that the US dolphin-safe provisions are more trade restrictive than necessary to fulfil their objective of ensuring that consumers are not misled about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins, and the United States has not successfully rebutted this claim. In reaching this conclusion, the Panel has considered the available scientific and technical information, related processing technology and the intended end-uses of tuna products.

Mexico's second written statement, para. 210; Mexico's second oral statement, para. 115.

⁸¹⁵ The Panel recognizes that without additional information in the labels, the US and the AIDCP dolphin-safe logos may be confused with one another.

(iii) Whether the US dolphin-safe provisions are more trade-restrictive than necessary to contribute to the protection of dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins

Arguments by the parties

- 7.579 Mexico claims that the US measures have no effect on encouraging foreign fishing fleets not to set upon dolphins. Mexico argues that the US measures will not influence or modify the conduct of the ETP tuna fishery so as to further preserve dolphin stocks. According to Mexico, the US measures do not add to that protection, particularly in respect of the actions of fishing fleets from countries that are signatories to the AIDCP and that are already fishing in what Mexico considers as the most environmentally responsible manner.⁸¹⁶
- 7.580 According to Mexico, all the US measures accomplish is to block imports of tuna products and tuna from those countries and, thereby, protect the US tuna industry. In Mexico's view, by withdrawing the economic incentive for countries and fishing fleets to comply with the AIDCP, the US measures undermine the viability of the AIDCP. Thus, Mexico concludes that undermining the AIDCP does *not* fulfil the objective of preserving dolphin stocks in the ETP.⁸¹⁷
- 7.581 Mexico also submits that the key underlying factual premise of the *Hogarth* ruling is the belief that dolphin stocks in the ETP were not recovering. According to Mexico, a recent study commissioned by the US Department of Commerce shows that dolphin populations in the ETP are recovering. Moreover, Mexico submits that no scientific evidence supports the view that setting on dolphins in a manner consistent with the AIDCP adversely affects dolphins from a stock sustainability perspective. Therefore, Mexico concludes the court ruling does not fulfil the objective of preserving dolphin stocks in the ETP. **
- 7.582 Thus, Mexico argues that the US measures are based on conditions that existed in the late 1980s but which ceased to exist many years ago and that they provide no mechanism for the United States to re-evaluate the status of marine mammal protection in the ETP based on the best available evidence. 821
- 7.583 Mexico also rejects the US argument that besides the observed dolphin injuries and mortalities in the ETP, there are unobserved mortalities of dolphins due to the fishing practice of setting on them. According to Mexico, the United States does not make projections about unobserved mortalities for any fishery other than the ETP, even when other fisheries have been identified by the United States as having high levels of mortality.⁸²²
- 7.584 Mexico claims that the United States does not apply the same multipliers to account for alleged unseen mortalities in its own fisheries as it does for the ETP. According to Mexico, the US analytical approach to evaluating dolphin populations incorporates a considerable amount of speculation and unreliable assumptions. Mexico submits that following the United States approach would mean that unobserved mortalities are over 28 times greater than the observed mortalities of about 1,200. Nonetheless, in Mexico's view, even if the US approach is applied, the best available

⁸¹⁶ Mexico's first written submission, para. 211; Mexico's second written statement, para. 206.

⁸¹⁷ Mexico's first written submission, para. 211; Mexico's second written statement, para. 206.

⁸¹⁸ Mexico's first written submission, para. 211.

⁸¹⁹ Mexico's second written submission, para. 204.

⁸²⁰ Mexico's first written submission, para. 211.

⁸²¹ Mexico's second oral statement, para. 111; Mexico's response to question 86, para. 4.

⁸²² Mexico's response to question 100 from the Panel, para. 43.

evidence is that dolphin mortalities in the ETP are negligible and do not affect populations of any of the dolphin stocks.⁸²³

7.585 According to the United States, the US dolphin-safe provisions help protect dolphin populations, by ensuring that the "dolphin-safe" label is not used on tuna products that contain tuna caught by setting on dolphins. According to the United States, to the extent customers choose not to purchase tuna products without the dolphin-safe label, the US dolphin-safe provisions help ensure that the US market is not used to encourage fishing fleets to set on dolphins. In the United States' view, as the practice of setting on dolphins to catch tuna in the ETP decreases in frequency, the associated adverse effects on dolphin populations decrease as well.⁸²⁴

7.586 The United States argues that the most probable reason dolphin populations remain depleted and show no clear signs of recovery is the practice of setting dolphins to catch tuna and the many adverse effects beyond immediate mortality or serious injury in the nets. 825

7.587 The United States submits that setting on dolphins has significant adverse effects on individual dolphins. 826 First, the United States argues that dolphin deaths are a foreseeable and expected consequence of setting on dolphins to catch tuna, and this is why under the AIDCP each vessel that wishes to fish for tuna in this way is assigned a set number of dolphins that it may be observed killing each year. 827 Therefore, the United States claims that "it cannot be disputed that not setting on dolphins to catch tuna affords greater protection to dolphins than setting on them". 828

7.588 Second, the United States submits that the observed dolphin mortalities and serious injuries are only one of the many adverse effects of setting on dolphins to catch tuna. In the United States' view, as long as dolphins are set upon to catch tuna, they will suffer the adverse effects of that practice. 829 The United States accepts that the magnitude of observed dolphin mortality due to setting on dolphins with purse seine nets has decreased significantly.⁸³⁰ However, according to the United States, the AIDCP does not address the many other adverse effects of setting on dolphins to catch tuna, such as separation of dependent calves from their mothers, reduced reproductive success due to stress induced foetal mortality, acute cardiac and muscle damage, cumulative organ damage, compromised immune function and increased predation.⁸³¹ The United States points out that speedboats and helicopters are used to encircle dolphins during setting on dolphins operations.⁸³² Thus, the US dolphin-safe provisions further contribute to protecting dolphins by ensuring that the US market is not an incentive for fishing fleets to use fishing techniques that adversely affect dolphins.833

⁸²³ Mexico's second oral statement, paras. 105-107.

⁸²⁴ United States' first written submission, para. 158.

⁸²⁵ United States' response to question 66 from the Panel, para. 154.

⁸²⁶ United States' first written submission, paras. 49-59; United States' second oral statement, paras. 7-9.

⁸²⁷ United States' second written submission, para. 139.

⁸²⁸ United States' second written submission, para. 157.

⁸²⁹ United States' second written submission, para. 157.

⁸³⁰ United States' first written submission, para. 54.

United States' first written submission, para. 56; United States' second written submission,

⁸³² United States' second oral statement, para. 7; Mexico's response to question 99 from the Panel, para. 40.

833 United States' first written submission, para. 170; United States' first oral statement, para. 48

Analysis by the Panel

7.589 For the purposes of determining whether the US dolphin-safe provisions are more trade-restrictive than necessary to contribute to protecting dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins", we need first to ascertain the manner and extent to which these measures contribute to this objective, and then consider whether the same level of protection could be achieved by the alternative measure proposed by Mexico.

The contribution of the US dolphin-safe provisions to protecting dolphins, by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins

- 7.590 In considering the contribution of the US dolphin-safe provisions to protecting dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins, we first note that, as described earlier, the achievement of this objective is in large part dependent on the successful achievement of the first objective relating to consumer information. Only if consumers are correctly guided by the dolphin-safe label towards products that contain tuna not caught using a fishing method that adversely affects dolphins, can the US measures successfully avoid encouraging those fishing techniques.
- 7.591 We also note in this respect that this US objective goes beyond mere information, as it seeks to provide a disincentive for certain behaviour, based on consumer choice. As described by the United States itself, it is only "to the extent that consumers choose" to buy dolphin-safe products that the measures will be able to ensure that the US market is not used to encourage the use of fishing techniques harmful to dolphins. 834
- 7.592 At the outset, we recall that we have already determined above that the US dolphin-safe provisions can only partially achieve the objective of ensuring that consumers are not misled about whether tuna was harvested by a method that adversely affects dolphins. It follows that the second objective can only be partially fulfilled, at best: to the extent that the measures fail to allow the consumer to accurately distinguish between tuna caught in conditions harmful to dolphins and other tuna, the related fishing techniques cannot be discouraged or encouraged in a manner that accurately reflects this distinction. With this in mind, we consider further the extent to which the US dolphin-safe provisions contribute to protecting dolphins.
- 7.593 The Panel observes that the United States has articulated its objective in general terms, without making any distinctions among fisheries, suggesting its intention to discourage the use of harmful techniques in *all* tuna fisheries regardless of their location.
- 7.594 We recall that under the US measures, tuna caught in the ETP may only be labelled dolphin-safe if a certification is provided that no dolphins were set upon and, *in addition*, that no dolphins were killed or seriously injured. We also recall that the US measures allow virtually *all* tuna harvested outside the ETP by using a method other than setting on dolphins to be labelled dolphin-safe, without requiring a statement that no dolphins were killed or seriously injured, despite indications that dolphins may be adversely affected by those fisheries.
- 7.595 We further recall that, as described above, the United States has also explained that in relation to tuna that is intended to be labelled with an alternative label, it is necessary to certify that no dolphins were killed or seriously injured, wherever it was caught. However, as discussed earlier, it is

⁸³⁴ United States first written submission, paras. 146, 158.

not clear that this requirement is in fact enforced through the NOAA Form 370. In addition, as also previously observed, if such a requirement is enforced, it would not modify the fact that access to the official label is available without such certification.

7.596 As described above, Mexico considers that the US measures effectively discourage rather than encourage the efforts undertaken by signatories of the AIDCP to prevent the harmful effects of setting in dolphins in the ETP. Mexico argues in particular that the US measures mistakenly assume that dolphin populations in the ETP are depleted. As discussed previously, however, the US objectives in relation to dolphin protection are not limited to the conservation of dolphin populations or to the avoidance of direct mortality. They also encompass additional unobserved incidental impacts of setting on dolphins. To the extent that denying a dolphin-safe label to any tuna caught by setting on dolphins ensures that these *unobserved* impacts of setting on dolphins are not encouraged, the US dolphin-safe provisions thus contribute to the United States' objective of not encouraging fleets to catch tuna in a manner that adversely affects dolphins.

7.597 However, to the extent that the US measures allow the use of a dolphin-safe label for tuna caught by any method other than setting on dolphins outside the ETP, without requiring any certification in relation to whether a dolphin may have been killed or seriously injured, the US dolphin-safe provisions do not ensure that the US market is not used to encourage fishing techniques that adversely affect dolphins. As determined above, the potential for dolphin mortality or serious injury outside the ETP, in the course of fishing operations using techniques other than setting on dolphins, is not insignificant.

7.598 Moreover, the Panel notes that by granting access to the dolphin-safe label to tuna caught by using fishing techniques that adversely affect dolphins outside the ETP, while at the same time denying it to tuna caught in the controlled conditions of the AIDCP in the ETP, the US dolphin-safe provisions may not only fail to discourage such harmful fishing techniques, but may actually have the opposite effect. Indeed, it seems that under the current implementation of the US measures, virtually all tuna caught outside the ETP by using a method other than setting on dolphins may be labelled dolphin-safe. Fisheries using these other techniques, knowing that their tuna is granted virtually unrestricted access to the dolphin-safe label, have no incentive to adapt their fishing gear and practices to address dolphin protection concerns. As we established earlier, the application of some of these other techniques may result in the death of a considerable number of dolphins and other marine mammals around the world. 835 This means that, in practice, fleets wishing to have access to the US market could be discouraged from setting on dolphins in the ETP under the regulated conditions of the AIDCP and encouraged to relocate instead to other fisheries in which dolphins populations are less closely monitored and where the incidental killing of dolphins is not monitored. We note in this respect the following observations made in the report commissioned by NOAA, referred to previously:

Cetaceans are "migratory." They spend several months each year traveling from one area to another, often covering vast distances in search of food, a particular climate, or a safe breeding ground. From a conservation and management perspective migratory species are exposed to an array of threats because they do not confine themselves to one location. Moreover, because they periodically cross through a number of jurisdictions, the level of protection afforded to cetaceans fluctuates according to their geographical location. Inevitably, migrating animals will pass

⁸³⁵ See paragraphs 7.520 to 7.531 above.

through jurisdictions where cetacean conservation is less of a priority than in other areas.836

For these reasons, the Panel concludes that the US dolphin-safe provisions are capable of protecting dolphins by ensuring that the US market is not used to encourage fishing practices that may kill or seriously injure dolphins, only within the ETP. In other fisheries, the US measures are capable of achieving their objective only in relation to the practices of setting on dolphins and using high seas driftnets. However, in relation to all other fishing techniques used outside the ETP, the US measures are not able to contribute to the protection of dolphins. Therefore, the Panel concludes that US measures may, at best, only partially fulfil their stated objective of protecting dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins.

7.600 As in relation to the first objective, we note that the United States has explained that the difference in certification requirements between tuna caught in the ETP and tuna harvested in other areas reflects a calibration of the likelihood that any given dolphin would interact with fishing gear and accidentally be killed or seriously injured⁸³⁷ and that the rationale behind these differences is the existence of special conditions, such as regular and significant tuna-dolphin association and the fact that dolphin populations in the ETP do not show clear signs of recovery in the ETP. 838 However, as discussed in relation to the first objective, in light of the evidence presented in these proceedings, the Panel is not persuaded the threats from tuna fishing for dolphins outside the ETP are insignificant, either in terms of observed mortality or serious injury, or even, for at least some of populations, in terms of sustainability of the populations or that the US measures calibrate the requirements to the likelihood of interaction and harmful effects to dolphins.

Whether the alternative suggested by Mexico provides a reasonably available less trade restrictive means of achieving the same level of protection

Having clarified the extent to which the US dolphin-safe provisions are able to contribute to the protection of dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins, we must now consider whether the alternative measure proposed by Mexico would provide a less trade-restrictive means of achieving the same level of protection.

7.602 Mexico submits that the objective of discouraging fishing techniques that affect dolphins can also be accomplished by the AIDCP. According to Mexico, the dolphin set method administered by the AIDCP provides the most comprehensive and tightly regulated protection for dolphins and other bycatch of any fleet in the world, including the US fleet. In addition, Mexico claims that the objective of protecting dolphins has long been accomplished in the ETP through the AIDCP, which Mexico considers to be the most successful multilateral agreement regarding effective dolphin protection.⁸³⁹

7.603 In the United States' view, neither the AIDCP itself nor measures implemented pursuant to it constitute a "reasonably available alternative" that could fulfil the legitimate objectives of the DPCIA provisions. While the United States agrees that the AIDCP has made an important contribution to

⁸³⁶ Exhibit MEX-5, p. v.

⁸³⁷ United States' second written submission, paras. 39-47. See the discussion of the adverse effects of setting on dolphins in paragraphs 7.485-7.486, 7.587-7.588 above.

United States' second written submission, para. 42-43; United States' second oral statement,

para. 14 839 Mexico's response to question 67 from the Panel, paras. 239-40; Mexico's first oral statement, para. 56.

dolphin protection in the ETP, in its view, despite the conservation measures called for under the AIDCP, dolphin populations remain depleted and have not recovered. 840

7.604 The United States accepts that the magnitude of observed dolphin mortality due to setting on dolphins with purse seine nets has decreased significantly. However, according to the United States, the AIDCP does not address the many other adverse effects of setting on dolphins to catch tuna, such as separation of dependent calves from their mothers, reduced reproductive success due to stress induced foetal mortality, acute cardiac and muscle damage, cumulative organ damage, compromised immune function and increased predation. The United States points out that speedboats and helicopters are used to encircle dolphins during setting on dolphins operations. As discussed earlier, the United States submits that setting on dolphins has significant adverse effects on individual dolphins. The United States claims that "it cannot be disputed that not setting on dolphins to catch tuna affords greater protection to dolphins than setting on them". In the United States' view, as long as dolphins are set upon to catch tuna, they will suffer the adverse effects of that practice.

7.605 The United States observes that the AIDCP does not contain provisions aimed at discouraging the practice of setting on dolphins to catch tuna. Thus, in the United States' view, the AIDCP is not a substitute for provisions that seek to protect dolphins by discouraging the practice of setting on them to catch tuna. 847

7.606 The United States claims that the DPCIA provisions together with the measures called for under the AIDCP and other provisions of US law form part of a comprehensive US strategy to protect dolphins. Thus, the United States claims that substituting the DPCIA provisions for measures called for under the AIDCP would reduce the overall ability of the United States to protect dolphins. 848

7.607 The Panel first recalls that the alternative measure suggested by Mexico would allow the coexistence of the US and the AIDCP dolphin-safe labels in the US market, with the aim of providing the US consumers with more detailed information about the dolphin-safe implications of the fishing methods employed to catch the tuna contained in those products. As noted above, Mexico does not suggest replacing the US dolphin-safe provisions in their entirety with the AIDCP regime. Nor could the provisions of the AIDCP be applicable in other fisheries to countries that are not parties to this treaty.

7.608 We have already concluded that the US dolphin-safe provisions are capable of contributing to the protection of dolphins in the ETP by discouraging the practice of setting on dolphins and other

United States' first written submission, para. 170; United States' first oral statement, para. 48; United States' response to question 37 from the Panel, paras. 91-2; United States' second written submission, paras. 140 and 155.

⁸⁴¹ United States' first written submission, para. 54.

⁸⁴² United States' first written submission, para. 56; United States' second written submission, paras. 139-40.

⁸⁴³ United States' second oral statement, para. 7; Mexico's response to question 99 from the Panel, para. 40.

para. 40.

844 United States' first written submission, paras. 49-59; United States' second oral statement, paras. 7-9. See the discussion of the adverse effects of setting on dolphins in paras. 7.485-7.486 and 7.587-7.588.

⁸⁴⁵ United States' second written submission, para. 157.

⁸⁴⁶ United States' second written submission, para. 157.

⁸⁴⁷ United States' second written submission, para. 157.

⁸⁴⁸ United States' first written submission, para. 171; United States' second written submission, para. 157.

fishing methods that may cause dolphin deaths or injuries. On the other hand, since tuna caught by setting on dolphins may be labelled dolphin-safe under the AIDCP, Mexico's alternative would not discourage the use of this practice in the ETP. To that extent, any adverse effects on dolphins of setting on them that are not addressed by the AIDCP regime would not be discouraged.

7.609 It is undisputed that the application of the AIDCP controls has played a decisive role in the considerable reduction of dolphin mortalities in the ETP in the last decades. 849 Annual estimate of dolphin mortality in the ETP in 1986 was 132,169 marine mammals, whereas the estimate in 2008 was 1,169 individuals. 850 As noted above, the United States recognizes that the observed dolphin mortalities as a result of setting on dolphins in the ETP are approximately 1,200 dolphins per year.⁸⁵¹ Furthermore, the Panel observes that the majority of sets on dolphins in the ETP these days produce zero observed dolphin mortalities. In 2008, for instance, in 92 per cent of the sets on dolphins in the ETP no incidental mortalities of dolphins were observed to have occurred.⁸⁵²

7.610 Improvements in the fishing gear and methods may considerably reduce the risks arising from setting on dolphins.⁸⁵³ Mexico has explained that some of these adaptations have already been adopted in the ETP under the AIDCP. 854 According to the Committee on Reducing Porpoise Mortality from Tuna Fishing established by the US National Academy of Science, small and major modifications to the gear and techniques may be adopted to further reduce dolphin mortality. 855 Moreover, the same Committee has indicated that "improvement in captain performance is the single most important step that can be taken to reduce dolphin mortality in the ETP purse-seine fishery".856

7.611 Moreover, the AIDCP establishes overall dolphin mortality limits (DMLs) for the fleets fishing for tuna in the ETP (5,000 individuals). In recent years, these limits have not been met.⁸⁵⁷ The Panel also notes that the AIDCP has established certain mechanisms to enforce these limits. Under the AIDCP observer program 100 per cent of the trips are monitored by an independent observer. 858 The AIDCP has also established an International Review Panel (IRP) that monitors the compliance by the vessels operating in the ETP with requirements and controls established by the AIDCP. 859 Finally, under the System for Tracking and Verifying Tuna of the AIDCP, each party establishes its own tracking and verification programme, implemented and operated by a designated national authority, which includes periodic audits and spot checks for caught, landed, and processed

Exhibit US-24, p. 52.

850 Exhibit US-24, p. 52.

851 United States' second written submission, para. 2.

⁸⁴⁹ Exhibit MEX-5, p. 31, Exhibit US-24, pp. 22-23.

⁸⁵² Exhibit US-24, page 23. Already in 1992, only 0.5% of dolphins encircled in the purse-seine net sets monitored by the NMFS were injured or killed, Exhibit MEX-2, page 19.

⁸⁵³ Exhibit MEX-2, page 110.

This methods include: (i) backing down the edges of the nets to create a channel through which the dolphins may escape, (ii) using divers and personnel in small boats to assist the dolphins, (iii) covering the top portion of the purse seine with a skirt of mesh that prevents dolphins from catching their beaks or heads in the net in the event of a net "canopy", and (iv) prohibiting fishing after sundown to ensure that dolphins within the net can be seen, Mexico's first written submission para. 29.

⁸⁵⁵ Exhibit MEX-2, pages 72-90.

Exhibit Mex-02, page 10.

Exhibit US-24, page 26.

The IATTC's international observer program and the national observer programs of Colombia, Ecuador, the European Union, Mexico, Nicaragua, Panama and Venezuela constitute the AIDCP On-Board Observer Program, Exhibit US-24, P. 23.

⁸⁵⁹ During 2008, the IRP consisted of 20 members; the 14 participating governments that have accepted the Agreement, and 6 representatives of non-governmental organizations (NGOs), 3 from environmental organizations and 3 from the tuna industry, Exhibit US-24, p. 26.

tuna products. 860 Should the mortalities of the fleet of any party to the AIDCP meet or exceed the total amount of DML distributed, fishing for tuna in association with dolphins must cease for all vessels operating under the jurisdiction of that party. 861

7.612 On the basis of the above, we conclude that, when conducted under "controlled" circumstances, e.g. in accordance with the requirements of the AIDCP, the adverse effects of setting on dolphins in the form of observed dolphin mortality or serious injury may be considerably reduced. In addition, compliance with the AIDCP dolphin-safe labelling requirements includes a requirement that no dolphin has been killed or seriously injured in sets in which the tuna was caught. Therefore, allowing compliance with the AIDCP labelling requirements to be advertised on the US market would discourage observed dolphin mortality resulting from setting on dolphins to the same extent as the existing US dolphin-safe provisions do and would involve no reduction in the level of protection in this respect.

7.613 However, as described earlier, the US objectives also include an intention to minimize unobserved consequences of setting on dolphins. Such adverse effects would not be discouraged through the AIDCP labelling requirements, which allows setting on dolphins. We also note that, according to the United States, these unobserved effects are responsible for the fact that the dolphin populations in the ETP remain depleted. Therefore, to the extent that it would not discourage these unobserved effects of setting on dolphins and their potential consequences on dolphin populations within the ETP, the use of the AIDCP labelling requirements as the exclusive basis for a dolphin-safe label could potentially provide a lesser degree of protection than the existing US dolphin-safe provisions, which disqualify setting on dolphins entirely for access to the dolphin-safe label. However, we note that, as observed above, other fishing techniques may also have adverse consequences on dolphins. As discussed above, the evidence before the Panel suggests that significant dolphin mortality also arises outside the ETP from the use of other techniques than setting on dolphins, and that some of the affected dolphin populations may be at risk as a result. The Panel notes in this respect the example of a Japanese driftnet fishery for albacore that was observed to have a dolphin mortality rate of three animals per net. 862 In contrast, as previously mentioned, the vast majority of the dolphin sets in the ETP are zero-kills.⁸⁶³

7.614 As also discussed above, the US dolphin-safe measures do not address such adverse impacts, and thus do not discourage the use of such fishing techniques, in respect of tuna caught outside the ETP. To the extent that they may encourage relocation from the ETP to other areas where such adverse effects arise but where access to the label is not subjected to certification that no dolphin was killed or seriously injured in the course of the trip, the US measures as currently applied (i.e. in the absence of any determination of regular and significant mortality outside the ETP pursuant to Section 1385(d)(1)(D) of the DPCIA) may even have the effect of encouraging the development of such fishing techniques without seeking to monitor and reduce the level of incidental killings and other harmful effects on dolphins.

7.615 In these conditions, we are not persuaded that allowing compliance with the AIDCP requirements to be advertised in addition to the existing US standard would lead to a lower level of protection than is currently provided under the US dolphin-safe provisions. As established above, in some cases, the risks arising from setting on dolphins under controlled circumstances may be lower

⁸⁶⁰ This system also includes mechanisms for communication and cooperation between and among national authorities, Exhibit US-24, p. 26.

⁸⁶¹ See Mexico's second written submission, para.67 and MEX-110

⁸⁶² See Exhibit MEX-2, p. 101.

⁸⁶³ See para. 7.574 and footnote 770 above.

than the risks arising from other fishing techniques applied without controlling for dolphin mortality or other adverse impacts.

7.616 This conclusion is not modified by the fact that access to alternative labels under the US measures are subjected to the additional requirement that no dolphin was killed even for tuna caught outside the ETP, since this would not modify the fact that access to the US official label, which provides the same advantage on the US market as an alternative label, is available without a requirement that no dolphin was killed or seriously injured.

7.617 We are mindful of the United States' submission that its measures calibrate the applicable requirements for the use of the dolphin-safe label in accordance with the level of risk faced by dolphins. However, as discussed in the context of our earlier consideration of this issue in relation to the first US objective, we are not persuaded, based on the evidence presented to us, that at least some of the dolphin populations affected by fishing techniques other than setting on dolphins are not facing risks at least equivalent to those currently faced by dolphin populations in the ETP under AIDCP monitoring.

7.618 In light of all the above, the alternative suggested by Mexico does not seem to create greater risks to dolphins in the ETP than those accepted by the United States under the challenged measures in relation to other fishing techniques used outside the ETP. Thus, we consider that Mexico's alternative would achieve a level of protection equal to that achieved by the US dolphin-safe provisions outside the ETP, as currently applied. As mentioned above in relation to the objective of consumer information, we note that this does not imply, in our view, that the different conditions for compliance with the US and the AIDCP label could not be made clear, to allow the consumer the benefit of full information in this respect.

7.619 Recalling our conclusion that the alternative submitted by Mexico is less trade-restrictive than the US dolphin-safe provisions, we therefore conclude that Mexico has identified a reasonably available less-trade restrictive alternative that would meet the level of protection achieved by the US measures in relation to the objective of contributing to the protection of dolphins by ensuring that the US market is not used to encourage fishing to catch tuna in a manner that adversely affects dolphins. In reaching this conclusion, we have taken into account the available scientific and technical information, related processing technology and the end-uses of tuna products.

(iv) Conclusion

7.620 In light of our determinations above in relation to both objectives of the US dolphin-safe provisions, we find that these measures are more trade-restrictive than necessary to fulfil their legitimate objectives, taking account of the risks non-fulfilment would create. Consequently, the Panel finds that the US dolphin-safe provisions are inconsistent with Article 2.2 of the TBT Agreement.

7.621 As described above, we make this determination taking into account our finding that the US dolphin-safe measures, as applied, only partly address the adverse effects on dolphins of tuna fishing that the United States has identified as relevant it the context of its objectives of informing consumers and contributing to the protection of dolphins in relation to the impact of such fishing techniques. Specifically, the US dolphin-safe provisions do not address observed mortality, and any resulting adverse effects on dolphin populations, for tuna not caught by setting on dolphins or high seas driftnet fishing outside the ETP. Similarly, the proposed AIDCP dolphin-safe standard which Mexico identified as part of its proposed alternative measure would also not address the entirety of the adverse effects identified by the United States, insofar as it would not address unobserved mortalities from setting on dolphins, and any resulting adverse effects on dolphin populations.

7.622 We also recall, in this context, our determination that the choice of the level of protection to be achieved in pursuance of the legitimate objectives identified is the prerogative of the Member taking the measures, and we therefore make no determination as to what might be an appropriate level of protection to achieve in relation to the objectives identified by the United States for the information of consumers and the protection of dolphins in relation to the manner in which tuna is caught.

7.623 Finally, we note that, as reflected in our analysis above, our findings take into account the information, including scientific information concerning the effects of tuna fishing on dolphins that is available to us for the purposes of these proceedings. From these elements, it appears that a number of aspects of this issue are not fully documented and that further research may be necessary in order to ascertain the exact situation in various areas.

4. Whether the US dolphin-safe labelling provisions are inconsistent with Article 2.4 of the TBT Agreement

7.624 Article 2.4 of the TBT Agreement provides that:

"Where technical regulations are required and relevant international standards exist or their completion is imminent, Members shall use them, or the relevant parts of them, as a basis for their technical regulations except when such international standards or relevant parts would be an ineffective or inappropriate means for the fulfilment of the legitimate objectives pursued, for instance because of fundamental climatic or geographical factors or fundamental technological problems."

7.625 Mexico argues the US dolphin-safe labelling measures are not based on an existing relevant international standard for the following reasons: (i) a relevant international standard exists (the "AIDCP standard"); (ii) the United States has failed to base its regulation on that international standard; and (iii) the relevant international standard is not an ineffective or inappropriate means for the fulfilment of the legitimate objective pursued. Reason the United States has failed to base its measures on the international standard, Mexico argues, because, it contemplated the application of the AIDCP Standard but rejected its application in favour of a unilateral standard, a "setting of nets on dolphins" standard.

7.626 The United States first considers that the AIDCP resolutions cited by Mexico do not set out a relevant international standard. Secondly, it argues that the use of the definition of dolphin-safe in the AIDCP resolution would neither be effective nor appropriate to fulfil the objectives of the US labelling provisions, clarifying that it is for Mexico to show that the relevant international standard it identifies would be effective and appropriate in meeting the objective pursued by the US dolphin-safe labelling provisions. 866

(a) Overall approach

7.627 The elements of a violation of Article 2.4 of the TBT Agreement were addressed by the Appellate Body in *EC – Sardines*. In that case, although the panel and the Appellate Body followed the same approach in their assessment of Peru's claim under Article 2.4. They first considered whether the alleged international standard was indeed a "relevant international standard" within the meaning of Article 2.4, then they analysed whether the relevant international standard had been used "as a basis for" the EC regulation challenged by Peru; finally the third element considered was the

⁸⁶⁴ Mexico's first written submission, para. 228.

⁸⁶⁵ Mexico's first written submission, para. 245.

⁸⁶⁶ United States' first written submission para. 178.

"ineffectiveness or inappropriateness" of the relevant international standard for the fulfilment of the legitimate objectives pursued. We find this approach to be consistent with the terms and structure of Article 2.4. Accordingly, we find it appropriate to examine Mexico's claim under Article 2.4 of the TBT Agreement on the basis of the following three elements:

- the existence or imminent completion of a relevant international standard;
- whether the international standard has been used as a basis for the technical regulation; and
- whether the international standard is an ineffective or inappropriate means for the fulfilment of the legitimate objectives pursued, taking into account fundamental climatic or geographical factors or fundamental technological problems.

7.628 We also note that, in addressing the allocation of burden of proof under this provision, the Appellate Body held that the complaining party had to prove that the alleged international standard had not been used as a basis for the challenged regulation and that it also had to show that the alleged international standard was effective and appropriate to fulfil the legitimate objectives pursued by the regulation. 868

7.629 We therefore now consider the three elements identified above in order to determine whether Mexico has demonstrated that the US dolphin-safe labelling provisions are not based on a relevant international standard in violation of Article 2.4. Specifically, we will consider:

- whether the "AIDCP standard" constitutes a relevant international standard;
- whether the United States has used it as a basis for its dolphin-safe labelling provisions; and
- whether it is an effective and appropriate means for the fulfilment of the legitimate objectives pursued by the United States.
- (b) Whether the "dolphin-safe" definition contained in the AIDCP Tuna Tracking System Resolution is a relevant international standard
- (i) Arguments of the parties

7.630 In its first written submission, Mexico explained that the parties to the AIDCP adopted in 2001 the "Resolution to Adopt the Modified System for Tracking and Verification of Tuna" which contains the following definitions:

⁸⁶⁷ See Panel Report, EC – Sardines, paras. 7.61-7.139 and Appellate Body Report, paras. 217-291.

^{868 &}quot;Similarly, the circumstances envisaged in the second part of Article 2.4 are excluded from the scope of application of the first part of Article 2.4. Accordingly, as with Articles 3.1 and 3.3 of the SPS Agreement, there is no 'general rule-exception' relationship between the first and the second parts of Article 2.4. Hence, in this case, it is for Peru – as the complaining Member seeking a ruling on the inconsistency with Article 2.4 of the *TBT Agreement* of the measure applied by the European Communities – to bear the burden of proving its claim. This burden includes establishing that Codex Stan 94 has not been used 'as a basis for' the EC Regulation, as well as establishing that Codex Stan 94 is effective and appropriate to fulfil the 'legitimate objectives' pursued by the European Communities through the EC Regulation." Appellate Body Report, *EC – Sardines*, paras. 274-275.

- Dolphin-safe tuna is tuna captured in sets in which there is no mortality or serious iniury of dolphins:
- Non-dolphin-safe tuna is tuna captured in sets in which mortality or serious injury of dolphins occurs.869

7.631 Further, Mexico explains that the parties to the AIDCP also adopted the "Resolution To Establish Procedures For AIDCP Dolphin Safe Tuna Certification" and that that document states the "[t]he terms used in this document are as defined in the AIDCP System for Tracking and Verification of Tuna"; therefore Mexico notes that the AIDCP's definitions of "dolphin-safe" and "nondolphin-safe" apply with respect to the AIDCP's rules on dolphin-safe certification. 870

7.632 Following the Panel's reasoning in EC – Sardines, Mexico observes that before addressing the question of whether an international standard is relevant, it must first be determined that an international standard exists.⁸⁷¹ Mexico referred to the definition of "standard" provided in Annex 1.2 of the TBT Agreement:

"Document approved by a recognized body, that provides, for common and repeated use, rules, guidelines or characteristics for products or related processes and production methods, with which compliance is not mandatory. It may also include or deal exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method."

According to Mexico the definition of "dolphin-safe" contained in the AIDCP Resolution to Adopt the Modified System for Tracking and Verification of Tuna⁸⁷² is an international standard within the meaning of Annex 1.2 of the TBT Agreement for of the following reasons: (i) the AIDCP provides for common and repeated use, rules, guidelines or characteristics for products or related processes and production methods because it provides rules expressly dealing with the characterization of tuna as dolphin-safe and non-dolphin-safe⁸⁷³, (ii) compliance with the AIDCP dolphin-safe certification is not mandatory and Section 2.1 of the Resolution To Establish Procedures For the AIDCP Dolphin Safe Tuna Certification which provides that "Application of the procedures for the use of the AIDCP Dolphin Safe Tuna Certificate shall be voluntary for each Party, especially in the event that they may be inconsistent with the national laws of a Party"874 and (iii) the AIDCP Standard was prepared and approved by the AIDCP member governments, which constitute a recognized body according to the sixth edition of the ISO/IEC Guide 2: 1991 which defines the terms used in the TBT Agreement according to Annex 1 of the TBT Agreement. 875 Mexico notes that:

"[T]he 1991 Guide states in Article 4.1 that a 'body' is defined as a '[l]egal or administrative entity that has specific tasks and composition.' The AIDCP fits this definition. Article 4.2 states that an 'organization' is defined as a '[b]ody that is based

⁸⁶⁹ Mexico's first written submission, para. 229.

⁸⁷⁰ Mexico's first written submission, para. 230.

⁽footnote original) Panel Report, European Communities – Trade Description of Sardines, WT/DS231/R and Corr.1, adopted. (Mexico's first written submission, fn 151).

²³ October 2002, as modified by the Appellate Body Report, WT/DS231/AB/R, para. 7.63.

872 Mexico refers to the definition as the "AIDCP Standard". See Mexico's first written submission, fn 149.

⁸⁷³ Mexico's first written submission, para. 234.

⁸⁷⁴ Mexico's first written submission, para. 235.

⁸⁷⁵ Mexico notes that the TBT Agreement does not defined "recognized body" but that Annex I of the TBT Agreement establishes that the terms of the sixth edition of the ISO/IEC Guide 2: 1991when used in the TBT Agreement shall be given the same meaning.

on the membership of other bodies or individuals and has an established constitution and its own administration.' The AIDCP also fits this definition, as it is composed of governments and is administered by the secretariat of the IATTC."⁸⁷⁶

7.634 Mexico notes that a "standardizing body" under Article 4.3, is a "[b]ody that has recognized activities in standardization" and that under Article 1.1, "standardization" is "[a]ctivity of establishing, with regard to actual or potential problems, provisions for common and repeated use, aimed at the achievement of an optimum degree of order in a given context."⁸⁷⁷ Mexico also recalls the definition of a "provision" of Article 7.1 of the ISO/IEC Guide which is an "[e]xpression in the content of a normative document, that takes the form of a statement, an instruction, a recommendation, or a requirement."⁸⁷⁸

7.635 Taking all these elements into account, Mexico concludes that "the characterization of tuna as dolphin-safe is an activity that falls within the definition of a 'provision' suitable for 'standardization', and the AIDCP organization is a 'recognized body' for purposes of the definition of a 'standard' in Annex I.2 of the TBT Agreement. For these reasons, the AIDCP Standard falls within the definition of "standard" contained in Annex 1.2 of the TBT Agreement."⁸⁷⁹

7.636 The United States in turn argues that "the definition of 'dolphin-safe' in the AIDCP tuna tracking resolution Mexico cites does not constitute a relevant international standard within the meaning of Article 2.4 of the TBT Agreement as it is not (1) a standard; (2) international; or (3) relevant."

7.637 The United States observes that as the "relevant international standard" for the purpose of its claims under Article 2.4, Mexico cites the definition of "dolphin-safe" in an AIDCP resolution: Resolution to Adopt the Modified System for Tracking and Verification of Tuna which states that "the terms used in this document are defined as follows" and includes in the list of those terms a definition for "dolphin-safe". The United States also observes that Mexico cites a second AIDCP resolution: "Resolution to Establish Procedures for AIDCP Dolphin Safe Tuna Certification" and that it characterizes this resolution as setting out the "AIDCP's rules on dolphin-safe certification". The United States specifies that this resolution defines the term "AIDCP Dolphin Safe Tuna Certificate" for the purposes of the resolution as a "[d]ocument issued by the competent national authority, evidence of the dolphin-safe status of tuna and tuna products, in accordance with the AIDCP System for Tracking and Verification of Tuna."

7.638 The United States first notes that the definition in the AIDCP tuna tracking resolution does not meet the definition of a "standard" in Annex 1 of the TBT Agreement because it does not set out "rules, guidelines or characteristics for products or related processes and production methods"; it sets out a definition for purposes of an intergovernmental agreement. The United States contends that the definition does not itself establish any rules, guidelines or characteristics for products or their related processes and production methods, or aspects covered under the second sentence of the definition of a

⁸⁷⁶ Mexico's first written submission, paras. 233-237.

⁸⁷⁷ Mexico's first written submission, para. 238.

⁸⁷⁸ (footnote original) Pursuant to Article 3.1, a "normative document" is a "Document that provides rules, guidelines or characteristics for activities or their results." Article 7.2 provides that a "statement" is a "Provision" that conveys information, while Article 7.3 provides that an "instruction" is a "Provision that conveys an action to be performed." Pursuant to Article 7.4, a "recommendation" is a "Provision that conveys advice or guidance," while Article 7.5 provides that a "requirement" is a "Provision that conveys criteria to be fulfiled." (Mexico's first written submission, fn 153).

⁸⁷⁹ Mexico's first written submission, paras. 238-239.

⁸⁸⁰ United States' first written submission, para. 182.

⁸⁸¹ United States' first written submission, para. 181.

technical regulation, such as labelling requirements.⁸⁸² The United States argues that the definition in the AIDCP resolution, simply defines a term and while the tuna tracking resolution more broadly seeks to establish procedures to track tuna, Mexico does not appear to be arguing that the resolution constitutes the "rules" at issue but that the definition itself sets out such "rules".⁸⁸³

7.639 Secondly, the United States argues that the "dolphin-safe" definition in the tuna tracking resolution is not "for common and repeated use". The United States further argued that to construe the definition of "dolphin safe" in the AIDCP resolution as a standard for "common and repeated use" would expand the scope of the definition of a "standard" and have serious implications with respect to Members' rights and obligations under any intergovernmental agreement. 885

7.640 The third reason the United States invokes in support of its argument is that the "dolphin-safe" definition in the tuna tracking resolution is not contained in a "document approved by a ... body". It holds that the AIDCP tuna tracking resolution (as well the AIDCP dolphin-safe certification resolution) is a document approved by the parties to the AIDCP, and neither the AIDCP nor the parties to it constitute a "body" (i.e. a "legal or administrative entity that has specific tasks and composition"). The United States points out that the AIDCP is an inter-governmental agreement and that countries are parties to, not members of, the AIDCP. The United States asserts that, even assuming *arguendo* that the AIDCP was a "body", it does not have recognized activities in standardization and therefore would not constitute a "recognized" body. 887

7.641 Finally, the United States argues that the definition in the AIDCP tuna tracking resolution does not qualify as "international" insofar as the resolution has only been adopted by the 14 countries parties to the AIDCP. It notes that despite the AIDCP being open to ratification by other countries, its accession is still limited to countries meeting certain criteria. Recalling that the term "international standard" is defined in the ISO/IEC Guide 2: 1991 as "[s]tandard that is adopted by an international standardizing/standards organization and made available to the public," and that the TBT Agreement defines "international body" as a "body ... whose membership is open to the relevant bodies of at least all Members", the United States holds that read together, an international standard is adopted by a body whose membership is open to the relevant bodies of at least all Members, and yet the AIDCP is not open for any Member to join. Assuming arguendo that the AIDCP were a "body", neither the

The United States explains that the word "common" in the definition of a standard refers to a rule, guideline or characteristic of general application and a rule, guideline, or characteristic that is for common and repeated use would not be one that was drafted for the specific purpose of defining a term in an international agreement, and that the AIDCP does not purport to establish a definition of "dolphin safe" for application outside the context of the AIDCP resolutions. United States' second written submission, paras. 171-172; United States' answers to Panel question No. 33, paras. 74-76

887 It stresses that the objectives of the AIDCP and the activities parties take pursuant to it are fundamentally different from those of bodies such as the Codex Alimentarius Commission (Codex) or ASTM International that have as their core function the development of standards. United States' first written submission, para. 185.

⁸⁸⁸ (i) with a coastline bordering the ETP; (ii) that are a member of the IATTC; or (iii) whose vessels fished for tuna in the ETP between 21 May 1998 and 14 May 1999.

⁸⁸² United States' second written submission, para. 170.

⁸⁸³ United States' first written submission, para. 183.

⁸⁸⁵ For example, the United States argues, it would mean that any time a Member agreed to the definition of a term for purposes of an intergovernmental agreement, it would have an obligation to use that definition as the basis for its technical regulations, even for example, where the intergovernmental agreement expressly states that no party has an obligation to implement that definition for purposes of domestic law. United States' second written submission, para. 172; United States' answers to Panel question No. 33, para. 77

⁸⁸⁶ United States' first written submission, fn 197.)

⁸⁸⁹ United States' first written submission, paras. 186-187.

AIDCP nor resolutions adopted by its parties qualify as international within the meaning of Article 2.4 of the TBT Agreement.

7.642 In the absence of a definition of the term "international standard" in the TBT Agreement, the Panel invited the parties to clarify this term for the purposes of Article 2.4 of the TBT Agreement. The United States explained that, by virtue of the chapeau of Annex 1 of the TBT Agreement the terms of the TBT Agreement had the same meaning as given in the sixth edition of the ISO/IEC Guide 2: 1991. It also noted that the ISO/IEC Guide in turn defines an "international standardizing organization" as a standardizing organization "whose membership is open to the relevant national body from every country"890 and "organization" as a "body" and that Annex 1 of the TBT Agreement provides in paragraph 4 that an "international body" is a "body ... whose membership is open to the relevant bodies of at least all Members". The United States also observed that in paragraph 2 (explanatory note) that "[s]tandards prepared by the international standardization community are based on consensus". In light of these definitions the United States concluded that "an international standard is a standard that is (i) adopted by a body whose membership is open to the relevant bodies of at least all Members, (ii) based on consensus and (iii) made available to the public". 891 It further noted that this interpretation of the term "international standard" was consistent with the work of Members since the conclusion of the Uruguay Round and particularly with the 2000 Decision of the Committee on Technical Barriers to Trade on Principles for the Development of International Standards, Guides and Recommendations with Relation to Articles 2, 5 and Annex 3 of the [TBT] Agreement⁸⁹² that sets out principles and procedures that standardizing bodies should observe when developing international standards, guides and recommendations.

7.643 Mexico agrees that the ISO/IEC Guide definition of "international standard" is applicable by virtue of Annex I of the TBT Agreement. 893 It further notes that the 1991 Guide defines "international standardizing organization" as a "Standardizing organization whose membership is open to the relevant national body from every country" and that a standardizing body is a "Body that has recognized activities in standardization". 894 However, Mexico submits that the concept of "international standard" should be applied flexibly to suit the circumstances of a particular situation. It illustrates that statement with the example of the Institute of Electronics Engineers and SAE International, Mexico asserts that these organizations are engaged in setting international standards despite the fact that they have as members individuals and not governments. Mexico adds that it is not clear that the TBT Agreement was intended to exclude the standards of those organizations from being treated as "international standards" even though those entities are not open to WTO members as such. 895 Finally Mexico points out to that the use of national standards by developed countries to create non-tariff barriers to import from developing countries is a growing source of concern for the WTO system and that Article 2.4 is an important tool in ensuring that standards are adopted and applied in a trade-neutral manner. 896

7.644 With regard to the United States' contention that the AIDCP should not be considered as a "recognized body" insofar as it does not have recognized activities in standardization, Mexico replied that the United States had not explained what it considered to be "recognized activities" in standardization and that the AIDCP is not only *a* recognized body but *the* recognized body for the

⁸⁹⁰ ISO/IEC Guide 2: 1991, para. 4.3.2.

⁸⁹¹ United States' response to Panel question No. 59, para. 135.

⁸⁹² Decision of the Committee on Principles for the Development of International Standards, Guides and Recommendations with Relation to Articles 2, 5 and Annex 3 of the Agreement, G/TBT/1/Rev.9, pp. 37-39.

⁸⁹³ Mexico's response to Panel question No. 59, paras. 184-185.

⁸⁹⁴ Mexico's response to Panel question No. 59, para. 186.

⁸⁹⁵ Mexico's response to Panel question No. 59, para. 187.

⁸⁹⁶ Mexico's response to Panel question No. 59, para. 188.

setting of standards relating to the protection of dolphins in the ETP insofar as its main role is to establish rules and procedures – i.e., standards – governing the interaction between fishing and dolphins. It notes that the members of the AIDCP have issued the Procedures for AIDCP Dolphin Safe Tuna Certification which are directly applicable to the subject of this dispute. It also holds that the AIDCP has issued a number of other standards, including Procedures for Maintaining the AIDCP List of Qualified Captains, Technical Guidelines to Prevent High Mortality During Sets on Large Dolphin Herds, and Guidelines for Required Raft for the Observation and Rescue of Dolphins and that the core function of the AIDCP is to establish procedures to be followed by the commercial fleets of the individual nations that are members of the AIDCP.

7.645 As for the characteristics that a body must meet to produce "international standards" within the meaning of Article 2.4 of the TBT Agreement Mexico, recalling that Annex 1 of the TBT Agreement defines an "international body or system" to be a "[b]ody or system whose membership is open to the relevant bodies of at least all Members", notes that Annex 1 also defines "regional body", "central government body", "local government body" and "non-governmental body" and that, accordingly, participation in the "body or system" by central government – as is the case for the AIDCP - is sufficient to meet the qualification test for "relevant bodies". 898 Noting that "body" is not defined in the TBT Agreement, Mexico recalls that the 1991 Guide states in Article 4.1 that a "body" is defined as a "[1]egal or administrative entity that has specific tasks and composition." It argues that Article 4.2 states that an "organization" is defined as a "[b]ody that is based on the membership of other bodies or individuals and has an established constitution and its own administration." According to Mexico national governments of course are "bodies", and the AIDCP is based on government's membership. It explains the AIDCP has its own administration, which is implemented by the Secretariat of the IATTC and that AIDCP system operates in conformity with the 2000 TBT Committee decision. 899 Mexico concludes that the AIDCP has the characteristics appropriate for a body to produce international standards. 900

7.646 In light of the definition of international standard, the United States contends that in order for a standard developed by a body to be international that body must (i) permit at least all Members to be members of that body, (ii) have developed the standard based on consensus, and (iii) made the standard publicly available. Therefore, determining whether a particular standard is "international" involves determining, in its view, not only whether the body that developed the standard is "international" but also whether the standard was adopted by consensus and made publicly available. ⁹⁰¹

7.647 In response to a question by the Panel, Mexico observed that if negotiators of the TBT Agreement had intended that "recognized body" have the exact same meaning as "standardizing body" as defined in the ISO/IEC Guide 2: 1991, they would have used the same term. 902 It also asserted that the core purpose of the AIDCP organization – which is composed of the member central governments acting collectively, and is administered by the secretariat of the IATTC – is to establish standards for tuna fishing to protect marine mammals and that there is no support for the US claim that the AIDCP organization is not a "recognized" or "standardizing" body for establishing when tuna caught in the ETP is dolphin-safe. 903 In contrast, the United States is of the view that a reasonable

⁸⁹⁷ Mexico's response to Panel question No. 60, paras. 190-191.

⁸⁹⁸ Mexico's response to Panel question No. 61, paras. 192-193.

Mexico's response to Panel question No. 61, paras. 195-196 and Mexico's second written submission, para. 212.

⁹⁰⁰ Mexico's response to Panel question No. 61, para. 197.

⁹⁰¹ United States' response to Panel question No. 61, para. 137.

⁹⁰² Mexico's response to Panel question No. 62, para. 199.

⁹⁰³ Mexico's second written submission, para. 213.

interpretation of "recognized" in the context of a standard would be that the body has recognized activities in standardization insofar as neither the TBT Agreement nor the ISO Guide contain a definition of "recognized body" but the ISO Guide does include a definition of the term "standardizing body". Mexico however contends that the United States' proposed interpretation would conflict with the wording of Annex 1.2.

7.648 The United States also commented on Canada's argument in its third party submission that the question of whether a body is "recognized" within the meaning of Annex 1 should be determined based on whether the body applies the six principles set out in the TBT Committee Decision on Principles for the Development of International Standards, Guides and Recommendations stating that given that the Committee Decision sets out principles and procedures that standardizing bodies should observe when developing international standards it would seem to support the position that a "recognized" body means a body that has recognized activities in standards development. However, the United States acknowledged Canada's suggestion that a body may be "recognized" not simply because it develops standards or has recognized activities in standardization, but because it develops standards or engages in standardization activities in accordance with certain recognized principles, for example, those in the Committee Decision. 905 Finally the United States concludes that under either interpretation, however, the AIDCP resolutions were not adopted by a "recognized body" because neither the AIDCP nor the parties to it constitute a "body" within the meaning of Annex 1 of the TBT Agreement and neither the AIDCP or the parties to it have recognized activities in standardization, or under Canada's interpretation, develop standards in accordance with the TBT Committee Decision principles, since only parties to the AIDCP participate in activities related to the AIDCP and only certain Members may be parties to the AIDCP.90

7.649 Concerning the meaning of the term "international body" contained in Annex 1.4 of the TBT Agreement, Mexico is of the view that the definition should be interpreted to mean that membership is open to those WTO Members who have an interest in the subject matter of the regulation. 907 In that regard it notes that first the AIDCP is not closed to additional members and that it remains open to accession by any State or regional economic integration organization that meets the requirements set out in the agreement, or is otherwise invited to accede to the Agreement on the basis of a decision by the Parties. It explains that no additional countries or regional economic integration organizations have expressed interest in joining the AIDCP and that further it is common that during the AIDCP meetings, Parties invite observer countries that regularly attend such meetings with the intention in the future to become Parties. In addition it stresses that unlike the SPS Agreement which specifically identifies three specific sources of international standards, the TBT Agreement does not and therefore retains flexibility for determining what are the relevant and applicable international standards on a case-by-case basis. Mexico also notes that in this case, the United States – a founding member of the AIDCP – has asserted that it can ignore an international standard that it created, to the disadvantage of Mexico, a developing country that expressly agreed to the international standard. Finally it observes that the US measure at issue is a standard specially designed to have an impact only on the ETP, where Mexico fishes and that this is reflected in the fact that the United States applies a different, less restrictive standard to other fisheries. Mexico finally contends that it cannot reasonably be expected that countries that are unaffected would be interested in participating. 908

7.650 On the contrary the United States notes that Annex 1 of the TBT Agreement states that an "international body or system" is a "body or system whose membership is open to the relevant bodies

⁹⁰⁴ United States' response to Panel question No. 62, para. 138.

⁹⁰⁵ United States' response to Panel question No. 62, paras. 138-139.

⁹⁰⁶ United States' response to Panel question No. 62, paras. 140.

⁹⁰⁷ Mexico's response to Panel question No. 63, para. 200.

⁹⁰⁸ Mexico's response to Panel question No. 63, paras. 201-206.

of at least all Members" and that nothing in this text supports reading the reference to "at least all Members" as meaning a subset of Members, for example, those that have interest in some or all of the work of that body or system. In fact, the context in which the definition of "international body or system" appears supports the opposite conclusion. In particular, the definition of a "regional body or system" is "body or system whose membership is open to the relevant bodies of only some of the Members". Thus, a body whose membership is open to only some Members, for example those that have an interest in the work of the body, would be a regional body; it would not be an international body.

7.651 In the United States' view, if an international body included bodies whose membership was only open to a subset of Members, it would mean that bodies that are not open to all Members may develop "international standards". This would have two important implications: (i) pursuant to Article 2.4 of the TBT Agreement, Members would have an obligation to use such "international standards" as the basis for their technical regulations (unless ineffective or inappropriate to meet a legitimate objective); (ii) it would also, pursuant to Article 2.5 of the TBT Agreement, give any Member that based a technical regulation on such an "international standard" a presumption that the technical regulation was not an unnecessary obstacle to trade. In other words, defining an international body to include bodies whose membership is open to only a subset of Members, would result in that subset of Members determining for other Members the "international standards" that must be the basis of their technical regulations.

7.652 Conversely, the United States contends that defining an international body whose membership is open to all Members ensures that all Members have the opportunity to participate in the development of any "international standard" that by virtue of Article 2.4 of the TBT Agreement they are obliged to use as the basis for their technical regulations (unless ineffective or inappropriate to fulfil the legitimate objective pursued). It also ensures that Members do not benefit from a presumption that their technical regulations are no more trade restrictive than necessary when based on standards developed by bodies that permitted only a subset of Members to participate.

7.653 The United States also contends that when standards are developed by bodies not open to at least all Members, the likelihood that they will develop standards that are relevant, effective and appropriate in fulfilling the objectives pursued by Members that were not permitted to participate in their development is greatly reduced along with the likelihood that those standards will be used as the basis for the Member's technical regulations.

7.654 The Panel asked the United States whether it agreed that the concept of "international standard" in Article 2.4 "should be applied flexibly". The United States responded that if this meant the term should be interpreted in a manner that departs from the text of the TBT Agreement, it did not agree with Mexico's assertion. It reiterated, for the reasons set in paragraph 7.646 above, the meaning of the term "international standard" is a standard that is (i) adopted by a body whose membership is open to the relevant bodies of at least all Members; (ii) based on consensus and (iii) made available to the public. The United States said it based this interpretation on the text of the TBT Agreement, in particular the definition of the term "international body" together with the ISO/IEC Guide 2:1991 definition of an "international standard" and the language in paragraph 2 of

⁹⁰⁹ United States' response to Panel question No. 63, para. 141.

⁹¹⁰ United States' response to Panel question No. 63, para. 142.

⁹¹¹ United States' response to Panel question No. 63, para. 143.

United States' response to Panel question 59, paras. 134-135 and United States' second written submission, paras. 176-177.

Annex 1 of the TBT Agreement: "Standard prepared by the international standardization community are based on consensus". 913

7.655 As for Mexico's comment that certain organizations despite being composed of individuals and not governments as members are nonetheless engaged in standard setting activities on the United States agrees that bodies such as IEEE and SAE may develop international standards based on the meaning of the term "international standards" derived from the text of the TBT Agreement. This is because TBT Agreement defines an international body as a body whose membership is open to the relevant bodies of at least all Members, and the ISO/IEC Guide 2 defines "international standardizing organization" as a "standardizing organization whose membership is open to the relevant national body from every country". The United States specifies that nothing in these definitions indicate that membership is limited to governments. A body such as SAE that allows individuals as well as bodies of at least all WTO Members to be members would fall within the definition of a body whose membership is open to the relevant bodies of at least all WTO Members, the United States says. The same standard setting activities of individuals and standard setting activities of individuals as well as bodies of at least all WTO Members, the United States says.

7.656 The United States also noted that there is an important policy reason to not interpret the term "international standard" "flexibly" as Mexico suggests or in a way that departs from the text of the TBT Agreement: under Article 2.4 of the TBT Agreement, Members have an obligation to base their technical regulations on relevant international standards, unless ineffective or inappropriate to fulfil a legitimate objective. Accordingly, it is important for Members to know which standards are "international" and therefore standards on which they have an obligation to base their technical regulations. If the term "international standard" is given a "flexible" meaning that departs from the text of the TBT Agreement, it is not clear how Members could be expected to know which standards are international and in turn how to comply with their obligation under Article 2.4 of the TBT Agreement it argues. 917

7.657 The Panel asked whether it should take into account the fact that the United States is a member of the IATTC and a signatory party to the AIDCP, Mexico argued that the AIDCP's role is to establish standards governing the interaction between fishing and dolphins, which then must be implemented and enforced by its member nations. It further noted that the AIDCP has promulgated the Procedures for AIDCP Dolphin Safe Tuna Certification which incorporates a definition of "dolphin-safe" which is the multilateral standard for dolphin-safe for tuna harvested in the ETP. According to Mexico, this meant that the United States, as a founding and fully participating member of the AIDCP, has "recognized" the AIDCP's standardizing activities. 918

7.658 The United States considers that the AIDCP is an intergovernmental agreement, not a body. It further says that the AIDCP resolutions were adopted by the parties to the AIDCP and not by a "body" as that term is defined for the purposes of the TBT Agreement. While the IATTC is a body, neither the AIDCP nor the AIDCP resolutions were adopted by the IATTC. Finally it holds that the fact that the United States is a signatory to the AIDCP and a member of IATTC does not change these facts. It also emphasizes that decisions regarding implementation of the international dolphin conservation program established under the AIDCP are taken by the parties to the AIDCP, not the

915 ISO/IEC Guide 2:1991 defines a "body" as a "legal or administrative entity that has specific tasks and composition", for example a company, organization or authority. ISO/IEC Guide 2:1991, para. 4.1; *see also* ISO/IEC Guide 2:1991, para. 4.5 (defining "authority" as "body that has legal powers and rights").

⁹¹³ United States' response to Panel's question No. 139, para. 66.

⁹¹⁴ See paragraph 7.643.

⁹¹⁶ United States' response to Panel question No. 139, para. 67.

⁹¹⁷ United States' response to Panel question No. 139, para. 68.

⁹¹⁸ Mexico's response to Panel question No. 141, paras. 114-115.

⁹¹⁹ United States' response to Panel question No. 141, para. 70.

IATTC and that while meetings of the parties to the AIDCP typically take place in conjunction with IATTC meetings and the IATTC is to provide Secretariat support for the AIDCP, decisions regarding implementation of the international dolphin conservation program including any resolutions thereunder remain decisions of the parties to the AIDCP. It further specifies that the IATTC has no legal authority to undertake activities or make decisions regarding the international dolphin conservation program established by the AIDCP and that in addition the 1949 IATTC Convention and its successor agreement the Antigua Convention, which establish the IATTC, and the AIDCP are distinct legal instruments, each with their own objectives and obligations.

(ii) Analysis by the Panel

7.659 We must determine, in this part of our analysis, whether the "dolphin-safe" definitions and labelling provisions contained in the AIDCP resolutions identified by Mexico constitute a "relevant international standard" in relation to the US dolphin-safe labelling provisions within the meaning of Article 2.4 of the TBT Agreement.

7.660 For that purpose, we must first clarify the notion of "international standard" for the purposes of this provision, and then determine whether the AIDCP resolution falls within the scope of this notion, and whether it is "relevant" in this dispute.

The notion of "international standard" in Article 2.4 of the TBT Agreement

7.661 Article 1.1 of the TBT Agreement provides that "[g]eneral terms for standardization and procedures for assessment of conformity shall normally have the meaning given to them by definitions adopted within the United Nations system and by international standardizing bodies taking into account their context and in the light of the object and purpose of this Agreement. In addition, Annex 1 of the TBT Agreement, entitled "Terms and their definitions for the purpose of this Agreement", provides that:

"The terms presented in the sixth edition of the ISO/IEC Guide 2: 1991, General Terms and Their Definitions Concerning Standardization and Related Activities, shall, when used in this Agreement, have the same meaning as given in the definitions in the said Guide taking into account that services are excluded from the coverage of this Agreement."

7.662 Annex 1 also provides specific definitions for the terms: "technical regulation", "standard", "conformity assessment procedures", "international body or system", "regional body or system", "central government body", "local government body" and "non-governmental body". As expressed by the Appellate Body in *EC – Sardines*: "[A]ccording to the Chapeau [of Annex 1], the terms defined in Annex 1 apply for the purposes of the TBT Agreement only if their definitions depart from those in the ISO/IEC Guide 2: 1991 (the 'ISO/IEC Guide'). This is underscored by the word 'however'".

7.663 The term "international standard" is not defined in Annex 1 of the TBT Agreement, but is defined in the ISO/IEC Guide 2. In accordance with the terms of Annex 1, in the absence of a specific definition of this term in Annex 1, the term "international standard" should be understood to have the same meaning in the TBT Agreement as in the ISO/IEC Guide 2, which defines it as a "standard that is adopted by an international standardizing/standards organization and made available to the public".

⁹²⁰ United States' response to Panel question No. 141, para. 71.

⁹²¹ Appellate Body Report, EC – Sardines, para. 224.

7.664 An "international standard" is thus composed of three elements: (i) a standard; (ii) adopted by an international standardizing/standards organization; and (iii) made available to the public. We must therefore consider whether the provisions of the AIDCP tuna tracking and verification resolution (which contain a definition of dolphin-safe) and of the AIDCP dolphin-safe certification resolution (which provides for the AIDCP dolphin-safe label) meet each of these components and thus constitute an "international standard".

7.665 Finally, we note that both parties have referred to the TBT Committee Decision on *Principles for the Development of International Standards, Guides and Recommendations* that sets out principles and procedures that standardizing bodies should observe when developing international standards. We consider it appropriate to take into account the principles contained in this decision where they may inform our understanding of certain aspects of the ISO/IEC Guide definitions such as the terms "international standardizing/standards organization" and "made available to the public" in the definition of "international standard". However we note that the Panel in EC - Sardines, in a statement not addressed by the Appellate Body, observed that the TBT decision "is a policy statement of preference and not the controlling provision in interpreting the expression 'relevant international standard' as set out in Article 2.4 of the TBT Agreement". 922

Whether the AIDCP dolphin-safe definition and certification constitutes a "standard" for the purposes of Article 2.4

7.666 The term "standard" is defined both in the ISO/IEC Guide 2 and in Annex 1 of the TBT Agreement.

7.667 The ISO/IEC Guide 2 defines a "standard" as a:

"[D]ocument, established by consensus and approved by a recognized body, that provides, for common and repeated use, rules, guidelines or characteristics for activities or their results, aimed at the achievement of the optimum degree of order in a given context".

7.668 Annex 1.2 of the TBT Agreement defines a "standard" as a:

"Document approved by a recognized body, that provides, for common and repeated use, rules, guidelines or characteristics for products or related processes and production methods, with which compliance is not mandatory. It may also include or deal exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method.

7.669 We note that the existence of differences between the ISO/IEC definition and the definition of a "standard" in Annex 1 of the TBT Agreement is acknowledged in the Explanatory Note, which distinguishes between standards "prepared by the international standardizing community" and standards covered by the TBT Agreement, i.e. those standards the preparation, adoption of application of which is regulated in the relevant provisions of the TBT Agreement. The Explanatory Note suggests that standards prepared by the international standardizing community are based on consensus while the TBT Agreement also covers "documents that are not based on consensus", and that standards as defined in the ISO/IEC Guide may be voluntary or mandatory, whereas those covered by the TBT Agreement are defined as voluntary.

⁹²² Panel Report, *EC – Sardines*, para. 7.91.

⁹²³ The Explanatory Note to Annex 1.2 provides as follows:

7.670 We see a difference between the notion of "standard", as defined in Annex 1.2 of the TBT Agreement for the purposes of defining the scope of application of the provisions of the TBT Agreement on standards (such as Articles 4), and the use of the term "standard" in the definition of the composite term "international standard" in the ISO/IEC Guide 2.

7.671 We acknowledge that, as noted by the Appellate Body, the terms defined in Annex 1 apply for the purposes of the TBT Agreement if these definitions depart from those in the ISO/IEC Guide. 924 Nonetheless, in our view, the term "standard" as used in the definition of an "international standard" in the ISO/IEC Guide 2 must be read in its proper context, i.e. as it is defined in the ISO/IEC Guide itself, in order to assign it the meaning intended in that definition. This is consistent with the terms of Article 1.1 and with Annex 1 of the TBT Agreement, which, as described above, provides that "[t]he terms presented in the sixth edition of the ISO/IEC Guide 2: 1991 (...) shall, when used in this Agreement, have the same meaning as given in the definitions in the said Guide".

7.672 Accordingly, pursuant to the definition of the ISO/IEC Guide 2, we must consider whether the AIDCP "dolphin-safe" provisions of the AIDCP constitute a "document, established by consensus and approved by a recognized body, that provides, for common and repeated use, rules, guidelines or characteristics for activities or their results, aimed at the achievement of the optimum degree of order in a given context".

7.673 We first note that the AIDCP tuna tracking and verification resolution containing the dolphin-safe definition is a document that describes a system for tracking tuna caught in the Agreement Area by vessels fishing under the AIDCP. The purpose of this system is to enable dolphin-safe tuna to be distinguished from non-dolphin-safe tuna, from the time of capture to the time it is ready for retail sale, on the basis of tuna tracking forms and additional verification procedures described in this document. Therefore, insofar as it contains provisions that relate to the capture, unloading, storage, transfer and processing of tuna, the AIDCP tuna tracking and verification resolution is a document that provides rules, guidelines or characteristics for tuna fishing and tuna.

7.674 The United States asserted that "one element necessary for a measure to constitute a standard is that it provide rules, guidelines etc. for 'common and repeated' use" and while noting that the ordinary meaning of "common" is "shared ... of general application" and of "repeated" is "frequent", the United States deduces that "common' addresses the shared or general nature of the measure, while 'repeated' addresses the frequency of the measure is to be used". In its view, a rule, guideline etc. that is for common and repeated use would not be one that was drafted for the specific purpose of defining a term in an international agreement. We agree with the United States that one component of

"The terms as defined in ISO/IEC Guide 2 cover products, processes and services. This Agreement deals only with technical regulations, standards and conformity assessment procedures related to products or processes and production methods. Standards as defined by ISO/IEC Guide 2 may be mandatory or voluntary. For the purpose of this Agreement standards are defined as voluntary and technical regulations as mandatory documents. Standards prepared by the international standardization community are based on consensus. This Agreement covers also documents that are not based on consensus."

These language differences were emphasized by the Appellate Body in *EC – Sardines*: "[t]he definition of a 'standard' in the ISO/IEC Guide expressly includes a consensus requirement" (Appellate Body Report, *EC – Sardines*, para. 225). According to the Appellate Body, the omission of a consensus requirement in Annex 1.2 was a deliberate choice on the part of the drafters of the TBT Agreement and the last two sentences of the Explanatory note give effect to this choice. The Appellate Body considered had the negotiators considered consensus to be necessary to satisfy the definition of "standard", they would have said so explicitly in the definition as is the case in the ISO/IEC Guide (ibid).

⁹²⁴ Appellate Body Report, *EC – Sardines*, para. 224.

United States' response to Panel question No. 33, para. 75.

the content of an international standard concerns the general applicability of the rules, guidelines or characteristics therein provided. This in turn has a bearing on the frequency of the use of such rules, guidelines or characteristics. We observe that to the extent that the rules and guidelines of the tracking system need to be observed and the tuna tracking forms completed for each fishing, unloading, storage, processing and marketing operation, the AIDCP tuna tracking and verification resolution provides rules "for common and repeated use".

7.675 In addition, the AIDCP dolphin-safe certification resolution is a document that establishes procedures to obtain the dolphin-safe certificate, provides what constitutes an invalid dolphin-safe certificate and it regulates the content of such certificate. This resolution defines the AIDCP Dolphin Safe Certificate as a "[d]ocument issued by the competent national authority, evidence of the dolphin-safe status of tuna and tuna products, in accordance with the definition of the AIDCP System for Tracking and Verification of Tuna". It also provides a definition of AIDCP Dolphin Safe Tuna Label as: "[g]raphic representation which distinguishes *dolphin-safe* tuna and tuna products, as defined in the System for Tracking and Verifying Tuna, which can be used on the packaging of tuna certified under this resolution". We infer from this description that the AIDCP dolphin-safe certification resolution is a document which includes symbols, packaging, marking or labelling requirements as they apply to tuna and tuna products.

7.676 We acknowledge the Appellate Body's statement that consensus is not an element of the definition of an international standard. Nonetheless, we observe that the two AIDCP resolutions cited by Mexico are documents that, as the United States pointed out, are approved by the Parties of the AIDCP. AIDCP resolutions are adopted in ordinary or extraordinary meetings of the AIDCP where quorum is reached when a majority of parties are present and are adopted by consensus. Specifically, the parties to the AIDCP agreed, at the 5th Meeting of the Parties, held on 15 June 2001 in San Salvador (El Salvador), to "adopt the following Procedures for AIDCP Dolphin-Safe Tuna Certification" and the "modified System for Tracking and Verification of Tuna developed by the Permanent Working Group on Tuna Tracking and recommended by the International Review Panel". The Parties to the AIDCP that were present at the meeting and approved these resolutions are States which have ratified or acceded to the Agreement (Colombia, Costa Rica, Ecuador, El Salvador, the European Union, Mexico, Nicaragua, Panama, Peru, the United States, Vanuatu and Venezuela). Both resolutions took effect thirty days after their adoption date. We believe that these documents therefore meet the ISO/IEC Guide 2 definition of "consensus", which is described as a "[g]eneral agreement, characterized by the absence of sustained opposition to substantial issues by any important part of the concerned interests and by a process that involves seeking to take into account the views of all parties concerned and to reconcile any conflicting arguments".

7.677 In conclusion, from and analysis of the content of the AIDCP resolutions we conclude that the dolphin-safe definition in the AIDCP tuna tracking and verification resolution cross-referenced in the AIDCP dolphin-safe certification resolution provides, for common and repeated use, rules, guidelines or characteristics for tuna fishing and tuna products, and thus constitutes a "standard" for the purposes of Article 2.4 of the TBT Agreement. With this preliminary determination in mind, we now consider whether it constitutes an "*international* standard" within the meaning of that provision.

⁹²⁶ See para. 7.669 above.

Whether the dolphin-safe definition in the AIDCP tuna tracking and verification resolution has been approved by an international standardizing or standards organization and made available to the public

7.678 As described above, the ISO/IEC Guide defines an "international standard" as a "[s]tandard that is adopted by an international standardizing/standards organization and made available to the public". Further, a standard is defined, inter alia, as a document approved by a "recognized body".

7.679 We must therefore now consider whether the AIDCP dolphin-safe definition has been approved by an "international standardizing/standards organization". We note that the ISO/IEC Guide 2 defines an "international standardizing organization" as a "standardizing organization whose membership is open to the relevant national body from every country". A "standardizing body", in turn, is defined as a "body that has recognized activities in standardization". The term "organization" is defined as a "body that is based on the membership of other bodies or individuals and has an established constitution and its own administration". (emphasis added) The Guide also defines the term "body" as a "legal or administrative entity that has specific tasks and composition". The TBT Agreement does not define these various terms, but does contain definitions of "international body or system", "regional body or system" and "central government body". 927

7.680 In light of these definitions, we must therefore determine whether the AIDCP dolphin-safe definition and labelling provisions were adopted by a legal or administrative entity based on the membership of other bodies or individuals that has an established constitution and its own administration, has recognized activities in standardization, and whose membership is open to the relevant national body of every country.

We first note that the two relevant AIDCP resolutions cited by Mexico are documents that are approved by the Parties of the AIDCP. AIDCP resolutions are adopted in ordinary or extraordinary meetings of the AIDCP where quorum is reached when a majority of parties are present and are adopted by consensus. 928 The parties to the AIDCP adopted the Procedures for AIDCP Dolphin-Safe Tuna Certification and the modified System for Tracking and Verification of Tuna on 15 June 2001. The Parties to the AIDCP that were present at the meeting and approved these resolutions are States which have ratified or acceded to the Agreement (Colombia, Costa Rica, Ecuador, El Salvador, the European Union, Mexico, Nicaragua, Panama, Peru, the United States, Vanuatu and Venezuela⁹²⁹). Both resolutions took effect thirty days after their adoption date.

7.682 The AIDCP is an international agreement concluded among States, and it does not as such have an established constitution or its own administration as such. However, the Parties to the convention acting jointly accomplish specific tasks in fulfilment of its objectives, as illustrated by the two resolutions at issue. Article II of the AIDCP thus provides that:

"The objectives of this Agreement are:

⁹²⁷ Annex 1.4 of the TBT Agreement defines an "international body or system" as a "[b]ody or system whose membership is open to the relevant bodies of at least all Members". "Regional body or system" is defined in Annex 1.5 as a:"[b]ody or system whose membership is open to the relevant bodies of only some of the Members". Finally "central government body" is, according to Annex 1.6, a "[c]entral government, its ministries and departments or any body subject to the control of the central government in respect of the activity in question".

928 Article VIII and Article IX of the AIDCP.

⁹²⁹ Guatemala and Honduras did not attend the AIDCP meeting.

- To progressively reduce incidental dolphin mortalities in the tuna purse-seine fishery in the Agreement Area to levels approaching zero, through the setting of annual limits:
- With the goal of eliminating dolphin mortality in this fishery, to seek 2. ecologically sound means of capturing large yellowfin tunas not in association with dolphins; and
- To ensure the long-term sustainability of the tuna stocks in the Agreement 3. Area, as well as that of the marine resources related to this fishery, taking into consideration the interrelationship among species in the ecosystem, with special emphasis on, inter alia, avoiding, reducing and minimizing bycatch and discards of juvenile tunas and non-target species."
- In addition, the implementation of the IDCP (the programme undertaken under the AIDCP), is supported by the IATTC, which has significant responsibilities in this respect and provides the Secretariat for the programme. The IATTC has an institutional structure composed of a principal body (the Commission), advisory committees, permanent working groups and two other types of committees (the committee for the review of implementation of measures adopted by the Commission and the scientific advisory committee. It also has a constitution, the Antigua Convention⁹³⁰ as well as its own administration, insofar as it is governed by its own rules of procedures and financial regulations and has a Secretariat.
- 7.684 The institutional link between the AIDCP and the organization is strengthened by the fact that all the States that are parties to the AIDCP are members of the IATTC (with the exception of Honduras). The AIDCP is open for signature, inter alia by States or regional economic integration organizations that are members of the IATTC. As for the ordinary annual meeting of the Parties of the AIDCP where decisions related the implementation of the agreement are taken, the AIDCP provides that it should take place preferably in conjunction with the IATTC meeting.⁹³¹ In light of these facts, we conclude that the parties of the AIDCP acting within the institutional framework of the IATTC constitute an "organization" for the purposes of the application of Article 2.4 of the TBT Agreement.
- 7.685 We now consider whether the parties of the AIDCP collectively act as a "standardizing body", that is, a "[b]ody that has recognized activities in standardization" (emphasis added) as defined by the ISO/IEC Guide 2. "Standardization" is defined as the "[a]ctivity of establishing, with regard to actual or potential problems, provisions for common and repeated use, aimed at the achievement of the optimum degree of order in a given context". 932 As concluded above, the AIDCP resolutions contain provisions, for common and repeated use, that concern tuna and tuna products and their related processes and production methods and that also deal with marking and labelling requirements. We therefore conclude that the parties of the AIDCP, within the institutional framework of the IATTC, develop and establish, with regard to dolphin mortality and tuna-stock sustainability problems, provisions for common and repeated use, aimed at the achievement of the optimum degree of dolphin

⁹³⁰ Which was negotiated to strengthen and replace the 1949 Convention establishing the IATTC, and entered into force on 27 August 2010.

 ⁹³¹ Article VIII.2 of the AIDCP.
 932 1.1 of the ISO/IEC Guide 2. We note in this respect that it is not necessary for the standardizing activity to be the single or even a "principal function" of the organization, in order for it to be a "standardizing body". This contrasts with the definitions of a "standards body", which is defined by the ISO/IEC Guide 2 as a "standardizing body" "that has as a principal function, the preparation, approval or adoption of standards".

protection and rational use of tuna resources in the context of the ETP tuna purse-seine fishery. ⁹³³ The AIDCP has *activities* in standardization.

7.686 Mexico has also stressed that "[t]here can be no doubt that the United States, as a founding and fully participating member of the AIDCP, has 'recognized' the AIDCP's standardizing activities". 934 We agree that the term "recognized" refers to the body's activities in standards development, and that the participation in these activities of the countries that are parties to the Agreement is evidence of their recognition. In our view, such recognition may also be inferred from the recognition of the resulting standard, i.e. when its existence, legality and validity has been acknowledged. In this respect, it appears that the AIDCP dolphin-safe definition and certification have been recognized by the parties to the Agreement, including the United States. This is reflected in one of the court rulings that led to the final *Hogarth* ruling that Mexico has challenged. The District Court for the Northern District of California in Earth Island Institute v. Evans, ruled that:

"Accordingly, Congress rejected the Panama Declaration on this point and instead adopted a compromise which provided that any change from the existing standard to the *less protective standard* called for by the Panama Declaration would turn on the scientific question of 'whether the intentional deployment on or encirclement of dolphins with purse seine nets is having a significant adverse impact on any depleted dolphin stock in the [ETP].""⁹³⁶ (emphasis added).

7.687 In light of the above, we conclude that the AIDCP has "recognized activities in standardization" and thus constitutes a "standardizing body". We must now consider whether membership in the AIDCP is open to the relevant national body of every country, such that it may be considered to be an "international standardization organization" as defined in the ISO/IEC Guide 2.

7.688 Article XXVI of the AIDCP provides that:

"This Agreement shall remain open to accession by any State or regional economic integration organization that meets the requirements in Article XXIV, or is otherwise invited to accede to the Agreement on the basis of a decision by the Parties."

7.689 The requirements of Article XXIV are temporal rather than geographical. The Article reads as follows:

"This Agreement is open for signature at Washington from May 21, 1998, until May 14, 1999 by States with a coastline bordering the Agreement Area and by States or regional economic integration organizations which are members of the IATTC or whose vessels fish for tuna in the Agreement Area while the Agreement is open for signature."

7.690 In addition, given that the AIDCP was open for signature by Members of the IATTC it may be useful to note that IATTC membership is open to the following: (a) the Parties to the 1949 Convention (i.e. the United States and Costa Rica); (b) States not Party to the 1949 Convention with a coastline bordering the Convention Area; (c) States whose vessels fish for fish stocks covered

936 Exhibit MEX-29.

⁹³³ See AIDCP Preamble.

⁹³⁴ Mexico's response to Panel question No. 141, para. 115.

⁹³⁵ The Shorter Oxford English Dictionary, 5th ed. (Oxford University Press), Volume II, p. 2490.

by the Convention, following consultation with the Parties; or (d) States that are otherwise invited to join on the basis of a decision by the Parties.⁹³⁷

7.691 Furthermore, we consider that the principle of openness embodied in Section C of the TBT decision on *Principles and Procedures for the Development of International Standards, Guides and Recommendations* which establishes that membership should be open on a non-discriminatory basis to relevant bodies of at least all WTO Members may assist us in interpreting the term "international standardizing/standards organization". The AIDCP membership was open for signature from 21 May, 1998 until 14May, 1999 to States whose vessels fished for tuna in the Agreement Area. Given that there were no limitations to or prohibitions of fishing in the agreement area, provided that the vessels did not operate in the EEZ of one of the coastline countries of the agreement area, any country whose fishing fleet was operating in the ETP could have signed the AIDCP. Thus the AIDCP was indeed open to signature on a non-discriminatory basis to the relevant bodies of at least all WTO Members in accordance with the principle of openness as described in the TBT decision. In addition, the AIDCP remains open to accession to any States or regional economic integration organization that is invited to accede to the Agreement on the basis of the parties' decision. To this day, the AIDCP membership is therefore open on a non-discriminatory basis to the relevant bodies of at least all WTO Members in accordance with the principle of openness as described in the TBT Committee Decision.

7.692 We find that the AIDCP is therefore an international standardizing organization for the purpose of Article 2.4 of the TBT Agreement.

7.693 In light of the above, we conclude that the AIDCP is open to the relevant body of every country and is therefore an international standardizing organization for the purposes of Article 2.4 of the TBT Agreement.

7.694 The final point to be addressed is whether the AIDCP dolphin-safe definitions and certification were made available to the public. We note that the dolphin-safe certification resolution, entitled "Public education", provides for transparency procedures. It establishes that:

- "a. Each Party, as appropriate, and the Secretariat shall give due publicity to the new *AIDCP Dolphin Safe* Certificate and *AIDCP Dolphin Safe* Label through relevant communications to interested governmental and non-governmental entities.
- b. The Parties agree to disseminate objective information to, *inter alia*, importers, fishermen's organizations, and non-governmental organizations, using their own capabilities within their national and international markets, to support an accurate public perception of the AIDCP, in order to increase the broad understanding of the AIDCP and its objectives.
- c. The Parties, through the Secretariat, may support the design and implementation of an international public education campaign to accomplish the objectives set forth within this section."

7.695 These transparency procedures aim at informing market operators about the AIDCP and its objective and the procedures for the dolphin-safe certificate and the dolphin-safe label which, by the same token, makes it possible to obtain the certificate and the label. Thus, these procedures are "made available to the public". We are therefore satisfied that the IADCP dolphin-safe definition and certification are made available to the public.

⁹³⁷ See Article XXVII of the Antigua Convention.

7.696 In addition, we consider that the principle of transparency contained in Section B of the TBT Committee Decision on *Principles for the Development of International Standards, Guides and Recommendations* and which informs the reading of the terms "made available to the public" supports this finding. We agree with Mexico that the AIDCP system operates in conformity with the principle of transparency as described in the TBT decision. As Mexico noted AIDCP Article XVII provides:

- "1. The Parties shall promote transparency in the implementation of this Agreement, including through public participation, as appropriate.
- 2. Representatives from intergovernmental organizations and representatives from non-governmental organizations concerned with matters relevant to the implementation of this Agreement shall be afforded the opportunity to take part in meetings of the Parties convened pursuant to Article VIII as observers or otherwise, as appropriate, in accordance with the guidelines and criteria set forth in Annex X. Such intergovernmental organizations and nongovernmental organizations shall have timely access to relevant information, subject to procedural rules on access to such information that the Parties may adopt."

7.697 This provision ensures that information on the procedures for the dolphin-safe certificate and the dolphin-safe label is effectively disseminated to all interested parties in the territories of the parties of the AIDCP, in accordance with Section B of the TBT decision on *Principles for the Development of International Standards*, *Guides and Recommendations* which defines transparency in terms of accessibility of standards to at least all interested parties in the territories of at least all WTO Members and effective dissemination of information concerning transparency procedures which should include *inter alia*, at a minimum, the prompt publication of a standard upon adoption.

Whether the AIDCP dolphin-safe definition and certification are "relevant"

7.698 Mexico considers that the AIDCP standard is relevant "because it serves the exact same purpose as the United States dolphin-safe labelling provisions" insofar as it is used to determine when tuna and tuna product can be certified as dolphin-safe and bear a dolphin-safe label. 939

7.699 On the contrary, in the United States' view, the definition Mexico cites "does not bear upon, relate to or pertain to the U.S. dolphin-safe labeling provisions". The United States reiterates that the definition is just that – a definition and that it does not set out any rules, guidelines or characteristics for the *labelling* of tuna products and therefore, concludes that the definition does not bear upon, relate to or pertain to the US dolphin-safe labelling provisions. In addition, the United States argues, it is also clear from the text of the tuna tracking resolution that the relevance of its definition of dolphin-safe is limited to defining that term for purposes of the resolution. Finally, the United States contends that the definition of dolphin-safe in the tuna tracking resolution does not relate or pertain to the objective of the United States dolphin-safe labelling provisions because, whilst it defines dolphin-safe as tuna caught in a set in which no dolphins were observed killed or seriously injured, the United States dolphin-safe labelling provisions, however, seek to ensure that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins, which includes not only whether the tuna was caught in a set in which

⁹³⁸ See Mexico's response to Panel question No. 61, para. 196.

⁹³⁹ Mexico's first written submission, para. 241.

⁹⁴⁰ United States' first written submission, para. 189.

⁹⁴¹ United States' first written submission, para. 189.

dolphins were observed killed or seriously injured but whether dolphins were otherwise adversely affected 942

7.700 As noted by both parties, in EC-Sardines, the Appellate Body agreed with the panel's statement that the ordinary meaning of the term "relevant" is "bearing upon or relating to the matter in hand; pertinent". According to the panel's reasoning, to be a "relevant international standard", the standard at issue in the dispute – Codex Stan 94 – would have to "bear upon, relate to, or be pertinent to the EC Regulation".

7.701 In the present case, we need therefore to determine whether the AIDCP dolphin-safe definition and certification bear upon, relate to, or are pertinent to the US dolphin-safe labelling provisions. As noted by Mexico "the measures apply to the same product, i.e. tuna, which are caught in the same area, and are then processed into canned tuna and bear a dolphin-safe label". He US dolphin-safe labelling provisions and the AIDCP resolutions both deal with the same products: tuna and tuna products. In addition, the AIDCP dolphin-safe certification resolution establishes the "AIDCP Dolphin Safe Tuna Certificate" and the "AIDCP Dolphin Safe Tuna Label", respectively defined as "Document issued by the competent national authority, evidence of the dolphin-safe status of tuna and tuna products, in accordance with the definition of the AIDCP System for Tracking and Verification of Tuna" and "Graphic representation which distinguishes *dolphin-safe* tuna and tuna products, as defined in the System for Tracking and Verifying Tuna, which can be used on the packaging of tuna certified under this resolution". The US dolphin-safe labelling provisions and the later AIDCP resolution therefore regulate the same subject matter, insofar as they both deal with the definition of criteria for identifying tuna as "dolphin-safe" and means for identifying dolphin-safe tuna, in the form of labelling requirements.

7.702 In our view, the dolphin-safe definition in the AIDCP tuna tracking and verification resolution cross-referenced in the AIDCP dolphin-safe certification resolution is relevant for the purposes of the US dolphin-safe labelling provisions, for the reasons explained below.

7.703 The US dolphin-safe labelling provisions establish regulatory categories based on a dual criterion: the area where the tuna fishery takes place and the fishing method used. The AIDCP regulates the ETP, which is an area regulated also by the US dolphin-safe provisions. In addition, the AIDCP is also pertinent to the extent that it addresses the fishing method to which the United States' measures foreclose access to the US dolphin-safe label: setting on dolphins. Because of the combination of these two elements, the AIDCP dolphin-safe standard bears upon the matter that is regulated by the US dolphin-safe label, insofar as both instruments address the consequences of the

⁹⁴³ Panel report, *EC – Sardines*, para. 7.68, quoting Webster's New World Dictionary (William Collins & World Publishing Co., Inc. 1976), p. 1199; Appellate Body Report on *EC – Sardines*, para. 230.

⁹⁴² United States' first written submission, para. 190.

Sardines and Sardine-type Products' and the EC Regulation lays down common marketing standards for preserved sardines. The European Communities indicated in its response that the term 'canned sardines' and 'preserved sardines' are essentially identical. Therefore, it is apparent that both the EC Regulation and Codex Stan 94 deal with the same product, namely preserved sardines. The scope of Codex Stan 94 covers various species of fish, including *Sardina pilchardus* which the EC Regulation covers, and includes, *inter alia*, provisions on presentation (Article 2.3), packing medium (Article 3.2), labelling, including a requirement that the packing medium is to form part of the name of the food (Article 6), determination of net weight (Article 7.3), foreign matter (Article 8.1) and odour and flavour (Article 8.2). The EC Regulation contains these corresponding provisions set out in Codex Stan 94, including the section on labelling requirement." Panel report, *EC – Sardines*, para. 7.69.

⁹⁴⁵ Mexico's first written submission, para. 241.

tuna-dolphin association in the ETP tuna purse-seine fishery and define "dolphin-safe" tuna with a view to promoting fishing methods that minimize harm to dolphins.

- 7.704 The AIDCP also has a bearing on the documentary evidence required to obtain the dolphin-safe label under the US measures, insofar as Annex II establishes an on-board observer program aimed at certifying the manner in which tuna was harvested; the US dolphin-safe label is then, by virtue of the US dolphin-safe labelling provisions, contingent upon this type of certification. For instance, for tuna caught in the ETP to be labelled dolphin-safe, the DPCIA provides that it must be accompanied, *inter alia*, by a written statement executed by:
 - "(I) the Secretary or the Secretary's designee;
 - (II) a representative of the Inter-American Tropical Tuna Commission; or
 - (III) an authorized representative of a participating nation whose national program meets the requirements of the International Dolphin Conservation Program, which states that there was an observer approved by the International Dolphin Conservation Program on board the vessel during the entire trip and that such observer provided the certification required under subsection (h); and"
- 7.705 This provision confirms, in our view, the relevance of the AIDCP and the IATTC as competent authorities that regulate the ETP tuna purse seine fishery.
- 7.706 In addition, for fisheries located outside of the ETP where regular and significant tuna-dolphin association has been determined, the US dolphins safe labelling provision also require a written statement of "an observer participating in a national or international program acceptable to the Secretary"; this evidences, in our view, the pertinence of this mechanism put in place by the AIDCP that serves as a reference for other fisheries. The US dolphin-safe labelling provisions build on the institutional framework established by the AIDCP which signals, not only the matters regulated by the two types of provisions are closely connected, but that in addition the US dolphin-safe labelling provisions heavily rely on the AIDCP system. In light of the above, we find that the dolphin-safe definition in the AIDCP tuna tracking and verification resolution cross-referenced in the AIDCP dolphin-safe certification resolution is relevant for the US dolphin-safe labelling provisions within the meaning of Article 2.4 of the TBT Agreement.
- 7.707 In light of the above, we find that the AIDCP dolphin-safe definition and certification constitute a "relevant international standard" within the meaning of Article 2.4 of the TBT Agreement, for the purpose of the US dolphin-safe labelling provisions.
- (c) Whether the United States used the AIDCP standard as a basis for its dolphin-safe labelling provisions
- (i) Arguments of the parties
- 7.708 Article 2.4 of the TBT Agreement provides that, where international standards exist or their completion is imminent Members shall use them, or the relevant parts of them, as a basis for their technical regulations.
- 7.709 We have already determined that the dolphin-safe definition in the AIDCP tuna tracking and verification resolution cross-referenced in the AIDCP dolphin-safe certification resolution is the relevant international standard for the purposes of the US dolphin-safe labelling provisions. The next step in the enquiry is to determine whether it was used as a basis for the US dolphin-safe labelling provisions.

7.710 Mexico contends that the United States has not based the US labelling provisions on the AIDCP standard, but on the contrary it "contemplated the application of the AIDCP Standard – which is incorporated into the U.S. national statute - but rejected its application in favour of a unilateral standard, a setting of nets on dolphins standard". 946 Under these circumstances, it is obvious that the current US standard is not based on the AIDCP Standard, Mexico states. 947

(ii) Analysis by the Panel

In EC — Sardines, the Appellate Body agreed with the panel that an international standard is used "as a basis for" a technical regulation "when it is used as the principal constituent or fundamental principle for the purpose of enacting the technical regulation". As Mexico noted the Appellate Body cited certain definitions of the term "basis", and concluded that:

"From these various definitions, we would highlight the similar terms 'principal constituent', 'fundamental principle', 'main constituent', and 'determining principle' all of which lend credence to the conclusion that there must be a very strong and very close relationship between two things in order to be able to say that one is 'the basis for' the other".94

7.712 As noted above, we consider that there the US dolphin-safe labelling provisions and the AIDCP resolutions are closely connected. In our view, the US legislator has constructed the US dolphin-safe labelling scheme building on the AIDCP foundations. However, the strong relationship between the two bodies of rules appears to be insufficient to infer that the AIDCP standard was used as a basis for the technical regulation.

7.713 In EC – Sardines, the Appellate Body rejected the European Communities' assertion that a "rational relationship" between an international standard and a technical regulation was sufficient to find that the former is used "as a basis for" the latter:

"[W]e see nothing in the text of Article 2.4 to support the European Communities' view, nor has the European Communities pointed to any such support. Moreover, the European Communities does not offer any arguments relating to the context or the object and purpose of that provision that would support its argument that the existence of a 'rational relationship' is the appropriate criterion for determining whether something has been used 'as a basis for' something else.

We see no need here to define in general the nature of the relationship that must exist for an international standard to serve 'as a basis for' a technical regulation. Here we need only examine this measure to determine if it fulfils this obligation."⁹⁵⁰

7.714 Finally, as Mexico points out, the Appellate Body also stated that:

"In our view, it can certainly be said – at a minimum – that something cannot be considered a 'basis' for something else if the two are *contradictory*. Therefore, under Article 2.4, if the technical regulation and the international standard contradict each

⁹⁴⁶ Mexico's first written submission, para. 245.

⁹⁴⁷ Mexico's first written submission, para. 245.

⁹⁴⁸ Appellate Body Report, *EC – Sardines*, para. 240–245.

⁹⁴⁹ (footnote original) Appellate Body Report, European Communities – Trade Description of Sardines, WT/DS231/AB/R, adopted 23 October 2002, para. 245. (Mexico's first written submission, fn 157).

950 Appellate Body Report, *EC – Sardines*, paras. 247–248.

other, it cannot properly be concluded that the international standard has been used 'as a basis for' the technical regulation." ⁹⁵¹

7.715 In the case at hand, the departure from the AIDCP standard was formally stated by the Court rulings, and in particular the *Hogarth* ruling which describes it as an explicit refusal to adopt the standard:

"The program was formalized into a legally-binding agreement known as the Panama Declaration, pursuant to which the United States' delegation agreed to seek a weakening of the dolphin-safe labeling standard and allow such a label to be affixed to tuna caught with purseseine nets as long as no dolphins were observed to be killed or seriously injured during the set. ...

When the delegation asked Congress to change the standard, however, Congress refused to relax its strict requirements without affirmative evidence that the tuna fishery was not significantly contributing to the slowness of the recovery rate of already depleted dolphin stocks."

- 7.716 In light of this evidence, we conclude that the United States failed to base the US dolphin-safe labelling provisions on the relevant international standard of the AIDCP.
- (d) Whether the AIDCP dolphin-safe standard would be an ineffective or inappropriate means for the fulfilment of the legitimate objective(s) pursued by the United States
- (i) Arguments of the parties

7.717 Mexico considers the AIDCP standard to be an effective means for achieving the pursued objective of protecting dolphins given that dolphin mortality in the ETP has decreased by 99 per cent. In addition, Mexico infers from the fact that the United States agreed to the standard itself in the context of the AIDCP that the United States considered it as "an effective means for accomplishing the objective of informing consumers that tuna was fished in a manner safe for dolphins."

7.718 The United States argues that the use of the definition of dolphin-safe in the AIDCP resolution would neither be effective nor appropriate to fulfil the objectives of the US labelling provisions, in particular, the definition would not ensure that consumers are not misled or deceived about whether the tuna products contain tuna that was caught in a manner that adversely affects dolphins. It adds that that the use of the definition would not be effective at protecting dolphins at the level the United States considers appropriate. Although the United States agrees with Mexico that the AIDCP has made an important contribution to protecting dolphins in the ETP, it considers however that it has only addressed part of the problem aiming at reducing observed dolphin mortality and serious injury caused by setting on dolphins to harvest tuna but not addressing other adverse effects of setting on dolphins. Furthermore, it argues that the US dolphin-safe labelling provisions have addressed both observed dolphin mortality and serious injury but also other adverse effects by encouraging fishing fleet to use other fishing techniques. It concludes that the US provisions have the objective to protect dolphins in ways that go beyond the protection provided under the AIDCP. In

⁹⁵¹ Appellate Body Report, *EC – Sardines*, para. 248.

⁹⁵² Earth Island Institute v. Hogarth, Exhibit MEX-30.

⁹⁵³ Mexico's first written submission, para. 251.

⁹⁵⁴ United States' first written submission, paras. 191-192.

⁹⁵⁵ United States' first written submission, para. 193.

⁹⁵⁶ United States' first written submission, para. 194.

that sense, the United States considers that the AIDCP only addresses partially the first objective targeted by the US labelling provisions, that is, the protection of dolphin populations. In its first oral statement the US asserted that allowing tuna products to be labelled dolphin-safe based on the AIDCP resolution definitions would defeat the objective of ensuring consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins and also defeat the objective of ensuring that the US market is not used to encourage the technique of setting on dolphins to catch tuna.

7.719 The Parties were asked whether the United States, acting as member of the IATTC and within the context of this organization had ever objected the appropriateness or effectiveness of the AIDCP regime to protect the dolphin populations in the ETP and whether it had, within the same context, has the United States ever expressed concerns about the consumer-deceiving potential of the AIDCP "dolphin-safe" designation. The United States confirmed it had been and continues to be a strong supporter of the AIDCP, whether at meetings of the parties to the AIDCP, IATTC meetings or in other contexts because it recognizes that setting on dolphins to catch tuna occurs and that the conservation measures called for under the AIDCP are an effective means to reduce observed dolphin mortalities when dolphins are set upon to catch tuna. In the context of the AIDCP, the United States said it had periodically proposed or supported efforts to strengthen implementation of the AIDCP. However it also observed that its strong support for the AIDCP should not be understood to mean that the United States supports the practice of setting on dolphins to catch tuna or that the United States believes the measures called for under the AIDCP are sufficient to protect dolphins from the harms associated with setting on dolphins to catch tuna. Regarding its concerns about other countries' use of the AIDCP dolphin-safe designation for their markets in the IATTC or AIDCP party meetings, the United States responded it never expressed any. 957

7.720 Mexico also confirmed the United States had never expressed objections to the AIDCP regime to protect dolphins and that it was also not aware that the United States has expressed concern over the AIDCP designation being deceptive. It noted that if the United States had concerns about the AIDCP, it should have raised those concerns in the IATTC rather than imposing a unilateral measure and emphasized that a key point is that the decisions, resolutions and regulations of the AIDCP are adopted by consensus which means that the United States has specifically agreed to all aspects of the AIDCP regime, including the AIDCP dolphin-safe certification program.

(ii) Analysis by the Panel

7.721 We first recall the Appellate Body's clarification in EC – Sardines that the burden of proving that the international standard is effective and appropriate to fulfil the legitimate objectives pursued by the challenged measures rests on the complaining party:

"There are strong conceptual similarities between, on the one hand, Article 2.4 of the *TBT Agreement* and, on the other hand, Articles 3.1 and 3.3 of the SPS Agreement, and our reasoning in *EC — Hormones* is equally apposite for this case. ... Accordingly, as Articles 3.1 and 3.3 of the SPS Agreement, there is no 'general rule-exception' relationship between the first and the second parts of Article 2.4. Hence, in this case, it is for Peru — as the complaining Member seeking a ruling on the inconsistency with Article 2.4 of the *TBT Agreement* of the measure applied by the European Communities — to bear the burden of proving its claim. This burden includes establishing that Codex Stan 94 has not been used 'as a basis for' the EC Regulation, as well as establishing that Codex Stan 94 is effective and appropriate to

958 Mexico's response to Panel question No. 142, paras. 117-118.

⁹⁵⁷ United States' response to Panel question No. 142.

fulfil the 'legitimate objectives' pursued by the European Communities through the EC Regulation."959

- Accordingly, in the present case, it is for Mexico to demonstrate that the AIDCP dolphin-safe standard is effective and appropriate to fulfil the legitimate objectives pursued by the United States through its dolphin-safe labelling provisions.
- 7.723 With respect to the meaning of the term "ineffective or inappropriate means", we note that the Appellate Body in EC - Sardines agreed with the Panel's view that the term "ineffective or inappropriate means" refers to two questions, the question of the effectiveness of the measure and the question of the appropriateness of the measure and that these two questions, although closely related, are different in nature. 960 The Panel had interpreted the term "ineffective" as referring to something which is not "having the function of accomplishing", "having as a result", or "brought to bear", whilst it had interpreted the term "inappropriate" as referring to something which is not "specially suitable", "proper" or "fitting". 961 In sum, the Panel had noted that: "... in the context of Article 2.4, an ineffective means is a means which does not have the function of accomplishing the legitimate objective pursued, whereas an inappropriate means is a means which is not specially suitable for the fulfilment of the legitimate objective pursued. ... The question of effectiveness bears upon de results of the means employed, whereas the question of appropriateness relates more to the nature of the means employed. (original emphasis)."962
- 7.724 In light of these determinations, we now consider whether Mexico has discharged its burden of showing that the AIDCP standard is appropriate and effective to fulfil the US dolphin-safe labelling provisions' objectives. We note that this enquiry differs from that conducted earlier under Article 2.2. In that context, as discussed earlier, Mexico had suggested that a label complying with the AIDCP standard could be allowed to coexist with the existing US standard. 963 Under Article 2.4 of the TBT Agreement, the Panel's point of enquiry is whether this international standard in itself would fulfil the legitimate objectives of the United States.
- 7.725 In accordance with the clarifications provided by the Appellate Body as described above, we consider that the AIDCP standard would be effective if it had the capacity to accomplish the two legitimate objectives defined by the United States, and it would be appropriate if it were suitable for the fulfilment of both of these objectives. 964 In addition, as noted by the Panel and the Appellate Body, insofar as the terms "ineffective" and "inappropriate" have different meaning and that it is conceptually possible that a measure could be effective but inappropriate, Mexico bears the burden of showing that the AIDCP standard is both effective and appropriate. 965 Mexico therefore has the duty to adduce sufficient evidence that the AIDCP standard meets the legal requirements of effectiveness and appropriateness set out in Article 2.4 of the TBT Agreement. 966
- 7.726 As described in Section (i) above, the US dolphin-safe labelling provisions have two stated objectives: (1) ensuring that consumers are not misled or deceived about whether tuna products

⁹⁵⁹ Appellate Body Report, *EC – Sardines*, paras. 274–275

⁹⁶⁰ Appellate Body Report, EC – Sardines, para. 285 citing the Panel Report, EC – Sardines, para. 261. Appellate Body Report, EC - Sardines, para 285 citing the Panel Report, EC - Sardines

para. 7.116. ⁹⁶² Appellate Body Report, EC - Sardines, para 285 citing the Panel Report, EC - Sardines para. 7.116.

963 Mexico's second written submission, para. 210.

FC Sardines para. 28

⁹⁶⁴ Appellate Body Report, EC – Sardines, para. 288.

Appellate Body Report, EC - Sardines, para. 289 citing the Panel Report, EC - Sardines,

⁹⁶⁶ Appellate Body Report, EC – Sardines, para. 290.

contain tuna that was caught in a manner that adversely affects dolphins; and (2) contributing to the protection of dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins. 967 We recall that, as discussed in Section 3(b)(ii)above (in particular in paragraphs 7.484 to 7.486), the scope of these objectives is not limited to the ETP, and encompasses a consideration of direct consequences of fishing techniques such as killing or serious injuries, and of unobserved consequences of setting on dolphins. We also recall the United States' observation that reducing such adverse effects "might also be considered to contribute to the protection of dolphin populations". 968

7.727 We first note that, to the extent that the US objectives are not limited to the ETP, and that the AIDCP standard addresses fishing conditions in the ETP and not in any other fishery, the AIDCP standard alone would not have the capacity to address US concerns in relation to the manner in which tuna is caught beyond the ETP. Although, as observed above, the types of mechanisms developed and implemented under the AIDCP may provide guidance in addressing the dolphin-safe issue beyond the ETP, we cannot assume that this would necessarily lead to the achievement of US objectives in these fisheries. Nonetheless, we consider further whether the requirements of the AIDCP for dolphin-safe labelling would be appropriate or effective to fulfil the US objectives with respect to the area that it addresses, i.e. the ETP.

7.728 With respect to the objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins, the United States holds that the AIDCP standard would not be effective because tuna caught in accordance with the AIDCP may be caught by setting on dolphins and nonetheless is certified as dolphin-safe under the AIDCP and the AIDCP label would not ensure that consumer have accurate information about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins and are not misled or deceived. 969

7.729 As described earlier, compliance with the AIDCP standard allows the use of setting on dolphins as a fishing technique, but subjects it to a number of conditions and a monitoring programme, to reduce the number of incidental bycatches of dolphins. In addition, the AIDCP dolphin-safe standard is only accorded to tuna that was caught on a trip in which no dolphin was killed or seriously injured. Therefore, with the AIDCP label alone, consumers will not be misled or deceived about whether dolphins were killed during the sets in which the tuna is caught. However, to the extent that there might be other adverse effects deriving from that fishing method, the AIDCP standard alone would not address them. It would not also, in itself, convey any information in this respect. The AIDCP dolphin-safe label itself does not convey any information on the fishing method that has been used for harvesting tuna contained in the product that bears the AIDCP logo, or on the impact that such method may have on dolphins. Even assuming that the parties to the AIDCP comply strictly with their transparency obligations, this would only inform the interested actors that the meaning of the term "dolphin-safe" in the context of the AIDCP framework refers to tuna captured in association with dolphins, in sets in which mortality or serious injury of dolphins does not occur.

7.730 Therefore, to the extent that it would not allow consumers to be informed of the fact that dolphins were chased in the context of catching the tuna at issue, and of the existence of potential unobserved consequences from such setting, the AIDCP label would not address some of the adverse effects on dolphins that the United States has identified as part of its objectives.

⁹⁶⁹ United States' first written submission, paras. 195-196.

⁹⁶⁷ See e.g. United States' first written submission, para. 146; United States' first oral statement, para. 47.

968 See para. 7.484 and fn 617.

7.731 We therefore conclude that the AIDCP standard, applied alone, would not be an effective or appropriate means of fulfilling the US objective of ensuring that consumers are not misled or deceived about whether tuna products contain tuna that was caught in a manner that adversely affects dolphins.

7.732 We now turn to the objective of contributing to protecting dolphins by ensuring that the US market is not used to encourage fishing fleets to catch tuna in a manner that adversely affects dolphins. In the United States' view, while the AIDCP has made an important contribution to protecting dolphins in the ETP, it only addresses part of the problem – that is how to reduce dolphin mortality when setting on dolphins to catch tuna. However, because it does not prohibit setting on dolphins to catch tuna, it does not ensure that no dolphins are in fact killed or seriously injured when dolphins are used to catch tuna and it does not address other adverse effects of setting on dolphins to catch tuna, the United States says. It emphasizes that the AIDCP contemplates that up to 5000 dolphins may be killed using this technique per year. And yet, the United States explains that its measures seek not only to reduce observed dolphin mortality and serious injury, but also to address other adverse effects of setting on dolphins to catch tuna by encouraging fishing fleets to transition to techniques to catch tuna that do not involve setting on dolphins by prohibiting use of the dolphin-safe label for tuna products that contain tuna that was caught during a trip in which purse seine nets were deployed or used to encircle dolphins. 971

7.733 The United States therefore concludes that the US dolphin-safe labelling provisions have as their objective to protect dolphins in ways that go beyond the protections provided for under the AIDCP and that relying solely on the AIDCP or its resolutions would not be an effective means of fulfilling the objective of the US dolphin-safe labelling provisions "to protect dolphins above and beyond minimizing observed mortalities and serious injuries as a consequence of setting on dolphins to catch tuna". 972

7.734 Mexico's main argument in support of its assertion that the AIDCP standard is an effective means for achieving the objective of protecting dolphins is that the objective of the US dolphin-safe labelling provisions relate solely to adverse effect on dolphins that occur when nets are set upon dolphins, but that the measures have no objectives concerning adverse effects on dolphins resulting from other fishing methods or occurring in ocean regions other than the ETP. 973 Mexico also argues that the US dolphin-safe labelling provisions are based on the underlying assumption that the fishing method used by the Mexican fleet and regulated by the AIDCP adversely affects dolphins, which it deems is unsupported by reliable evidence. Mexico focuses on the dolphin stocks recovery and affirms that the best available scientific evidence shows that dolphin mortalities in the ETP are negligible and are not affecting populations of any of the dolphin stocks. Mexico underlines the fact that the IATTC has also questioned whether the overall United States' analytical approach to evaluating dolphin populations is sound. 974 To summarize, Mexico argues that there is "no scientific evidence that setting upon dolphins in a manner consistent with the AIDCP adversely affects dolphins from a stock sustainability perspective."⁹⁷⁵ It holds that the most recent study, a 2008 US DOC study, indicated that dolphin stocks are recovering, indicating that the fishing methods are not having adverse effect on dolphins.⁹⁷⁶

⁹⁷⁰ United States' first written submission, para. 193.

⁹⁷¹ United States' first written submission, para. 194.

⁹⁷² United States' first written submission, para. 194.

⁹⁷³ Mexico's oral statement of the second substantive meeting, para. 88.

⁹⁷⁴ Mexico's oral statement of the second substantive meeting, para. 106.

⁹⁷⁵ Mexico's second written submission, para. 204.

⁹⁷⁶ Mexico's second written submission, para. 204.

7.735 Mexico argues that the US dolphin-safe labelling provisions are based on the assumption that one of the fishing method used by the ETP fishing fleets adversely affects dolphins which is, in turn, based on the assumption that dolphin stocks are not recovering. As explained in paragraph 7.550 above, however, we are not persuaded that the objective of protecting dolphins through the US dolphin-safe provisions is to be understood exclusively, or even primarily, in terms of dolphin population recovery. Rather, both US objectives are defined in terms of "adverse effects" of fishing practices on dolphins. As described above, this includes observed mortality from tuna fishing as well as unobserved consequences of setting on dolphins. As also described above, the United States has indicated that this "may also be considered as seeking to conserve dolphin populations". This suggests to us that the US objective of seeking to minimize observed and unobserved mortality and injury to dolphins is not conditioned upon or dependent on dolphin populations being depleted.

7.736 As we understand it therefore, the United States' assumption that setting on dolphins is harmful to dolphins is not premised only on the lack of dolphin stocks recovery. As mentioned above, the United States has referred to a diversity of adverse effects of setting on dolphins, emphasizing both the individual dolphin mortality (observed or delayed) as well as the issue of the recovery of dolphin stocks. As a study presented as evidence by Mexico itself states describes, there are ecological but also other concerns for dolphins and various target levels of either dolphin population size or mortality. We note that this study also suggests that the difficulty presented by the tunadolphin issue as an international problem is due to differing conservation ethics, and suggests that the United States' laws and policies have the goal of preventing all dolphin mortality from tuna fishing, while the laws and policies of other nations and more often directed toward conserving dolphin populations but not necessarily preventing all mortality.

7.737 As explained earlier in the context of our determinations under Article 2.2 in relation to the legitimate objectives pursued by the United States, the Panel has considered that despite the existence of a degree of uncertainty in relation to the extent to which setting on dolphins may have adverse impact on dolphins beyond observed mortality, sufficient evidence has been put forward by the United States to raise a presumption that genuine concerns exist in this respect and that the method of setting on dolphins "has the capacity" of resulting in observed and unobserved adverse effects on dolphins .

7.738 We acknowledge that the AIDCP standard contributes importantly, as the United States itself observes, to the reduction of dolphin mortality from setting on dolphins within the ETP. The may even contribute to the protection of dolphin stocks and progressive recovery of depleted populations. However, taken alone, it fails to address unobserved adverse effects derived from repeated chasing, encircling and deploying purse seine nets on dolphins, such as separation of mothers and their dependent calves, killing of lactating females resulting in higher indirect mortality of dependent calves and reduced reproductive success due to acute stress caused by the use of helicopters and speedboats during the chase. The standard contributes importantly, as the United States itself observes, to the reduction of dolphin mortality from setting on dolphins within the ETP. The may even contribute to the protection of dolphin stocks and progressive recovery of depleted populations. However, taken alone, it fails to address unobserved adverse effects derived from repeated chasing, encircling and deploying purse seine nets on dolphins, such as separation of mothers and their dependent calves, killing of lactating females resulting in higher indirect mortality of dependent calves and reduced reproductive success due to acute stress caused by the use of helicopters and speedboats during the chase.

7.739 We also note that, to the extent that the AIDCP standard addresses setting on dolphins and not other fishing techniques that may also result in adverse effects on dolphins, it would also not provide an effective or appropriate means of fulfilling the US objectives in this respect.

⁹⁷⁷ See Mexico's oral statement at the second substantive meeting, para.104.

⁹⁷⁸ *Dolphins and the Tuna Industry*, a study by the Committee on Reducing Porpoise Mortality from Tuna Fishing, the Board on Environmental Studies and Toxicology, The Commission on Life Sciences and the National Research Council mandated by the MMPA, (1992) p. 3, 5 and 21. MEX-2

⁹⁷⁹ United States' first written submission, para. 170.

⁹⁸⁰ Exhibits US-4 and US-11.

7.740 For all these reasons, we find that Mexico has failed to demonstrate that the AIDCP dolphin-safe standard is an effective and appropriate means to fulfil the US objectives at the United States' chosen level of protection.

C. MEXICO'S CLAIMS UNDER THE GATT 1994

- 7.741 In addition to its claims under the TBT Agreement, Mexico has also raised claims under Articles I:1 and III:4 of the GATT 1994. Given the strong commonalities between these claims and some of Mexico's claim under the TBT Agreement, we must consider whether it is necessary, for the full resolution of the dispute, to consider also these claims under GATT 1994.
- 7.742 We note that, in response to a question by the Panel⁹⁸¹, Mexico has argued that it was necessary and essential to the effective resolution of this dispute that the Panel rule on all of the claims raised by Mexico under Articles I:1 and III:4 of the GATT 1994 and Articles 2.1, 2.2, and 2.4 of the TBT Agreement because of: "(i) the nature of the measures at issue; (ii) the fact that this is the first time that such measures have been subject to dispute settlement under the DSU; (iii) the differences in the wording and potential application of the provisions of the GATT 1994 and the TBT Agreement to the measures; and (iv) the importance of the effective discipline of such non-tariff measures to developing country Members such as Mexico. This final reason the importance of these disciplines to developing country Members is particularly important because developing country Members may be most likely to be exposed to the adverse effects of non-tariff measures such as those at issue in this dispute."
- 7.743 We agree with Mexico that, should the Panel fail to make findings that are necessary to resolve the dispute this would constitute a false judicial economy and an error of law. However, we also note that if the panel finds that the matter in dispute is sufficiently resolved by the findings on the first claims examined, there is no need to examine additional claims. As explicitly stated by the Appellate Body, "[n]othing in [Article 11 of the DSU] or in previous GATT practice *requires* a panel to examine *all* legal claims made by the complaining party. 984
- 7.744 In this respect, we note that three of the reasons invoked by Mexico in support of its view that the Panel should not exercise judicial economy with respect to any of its claims (reasons (i), (ii) and (iv)) do not appear to directly relate to the question of whether the dispute would be fully resolved. These considerations therefore have little, if any, bearing on the question of whether we may exercise judicial economy.
- 7.745 The third reason invoked by Mexico, i.e. "differences in the wording and potential application of the provisions of the GATT 1994 and the TBT Agreement to the measures", does, however, have a bearing on the question at hand. To the extent that the legal and factual bases of different claims are distinct, this may affect the question of whether all the issues before the Panel have been properly addressed through an examination of some of these claims only.
- 7.746 We note that Mexico's claims under the GATT 1994 are non-discrimination claims under Articles I:1 and III:4. We also note that in the context of considering Mexico's claims under the TBT Agreement, we have considered among others, Mexico's non-discrimination claims under Article 2.1 of that Agreement.

⁹⁸¹ Mexico's response to Panel question No. 1 paras. 3-4.

⁹⁸² Mexico's response to Panel question No. 114 para. 67.

⁹⁸³ Appellate Body Report, *Canada – Wheat Exports and Grain Imports*, para. 133.

⁹⁸⁴ Appellate Body Report, *US – Wool Shirts and Blouses*, para. 339.

7.747 We further recall that, in the presentation of its arguments to the Panel under the TBT Agreement, Mexico consistently referred the Panel to its arguments under Articles I.1 and III:4 of the GATT 1994 (and the United States similarly referred to its own responses under the GATT 1994). In that context, Mexico argued that Article 2.1 of the TBT Agreement contains two non-discrimination obligations applicable to technical regulations, one that is similar to the national treatment obligation in Article III:4 and the other that is similar to the most-favoured-nation obligation in Article I:1, and that although language used in Article 2.1 is different from that used in Articles III:4 and I:1, both of these GATT 1994 provisions offer guidance on how to interpret Article 2.1. Mexico has not provided any explanation for its contrary view expressed in the context of its request that the Panel refrain from exercising judicial economy, that it was necessary to rule on these claims under both agreements and under both contexts (national treatment and MFN) because the nature, scope and application of the claims under Articles I:1 and III:4 of the GATT 1994, and Article 2.1 of the TBT Agreement, are different, and address different rights and obligations which, in turn, will have different implications during the implementation phase of this dispute.

7.748 In light of the fact that we have addressed, in the context of our examination of Mexico's claims under the TBT Agreement, all aspects of Mexico's claims, including non-discrimination aspects under Article 2.1, and other aspects under Article 2.2 and 2.4, and in light of our findings under these provisions, we are not persuaded that it is necessary for us to consider separately and additionally Mexico's claims under Articles I:1 and III:4 of the GATT 1994. Accordingly, we exercise judicial economy in respect of these claims and decline to rule on them.

VIII. RULINGS AND RECOMMENDATIONS

- 8.1 In light of the above findings, the Panel finds that the US dolphin-safe provisions:
 - (a) are not inconsistent with Article 2.1 of the TBT Agreement;
 - (b) are inconsistent with Article 2.2 of the TBT Agreement because they are more traderestrictive than necessary to achieve a legitimate objective, taking into account the risks that non-fulfilment would create;
 - (c) are not inconsistent with Article 2.4 of the TBT Agreement.

⁹⁸⁷ We note in this respect the following determinations of the Appellate Body in relation to the completion of the analysis in the context of Peru's claims under Articles 2.1, 2.2 and 2.4 of the TBT Agreement and Article III:4 of GATT 1994 in *EC – Sardines*:

"Peru submits that, if we conclude that the EC Regulation is consistent with Article 2.4, it would be appropriate for us to complete the Panel's analysis and resolve the dispute by making findings on those provisions of Article 2 of the *TBT Agreement* on which the Panel did not make any findings, namely Articles 2.2 and 2.1 of the *TBT Agreement*. Although Peru made a claim before the Panel under Article III:4 of the GATT 1994, Peru does not ask us to complete the analysis by addressing that provision. The European Communities objects to the completion of the analysis, expressing the view that there are not sufficient undisputed facts in the record to do so.

Because we have found that the EC Regulation is *not* consistent with Article 2.4 of the *TBT Agreement*, the conditions to Peru's request have not been met, and, therefore, we do not think it is necessary for us to make a finding under Articles 2.2 and 2.1 of the *TBT Agreement* in order to resolve this dispute. Equally, we do not think it is necessary to make a finding under Article III:4 of the GATT 1994 in order to resolve this dispute. Therefore, we decline to make findings on Articles 2.2 and 2.1 of the *TBT Agreement*, or on Article III:4 of the GATT 1994." (Appellate Body Report, *EC – Sardines*, paras. 312-313).

⁹⁸⁵ Mexico's response to Panel question No. 58 para. 172.

⁹⁸⁶ Mexico's response to Panel question No. 114 para. 69.

- 8.2 For the reasons explained in Section VII of this Report, the Panel exercises judicial economy with respect to Mexico's claims under Articles I:1 and III:4 of the GATT 1994.
- 8.3 Under Article 3.8 of the DSU, in cases where there is an infringement of the obligations assumed under a covered agreement, the action is considered prima facie to constitute a case of nullification or impairment of benefits under that agreement. Accordingly, we conclude that, to the extent that the United States has acted inconsistently with the provisions of the TBT Agreement, it has nullified or impaired benefits accruing to Mexico under that Agreement. We therefore recommend that the DSB request the United States to bring its measures into conformity with its obligations under the TBT Agreement.